Shared and the second second second



Department of Energy

Richland Operations Office P.O. Box 550 Richland, Washington 99352 RECEIVED 1996 SEP 30 AM 11: 19 DNF SAFETY BOARD

96-WSD-234

SEP 2 7 1996

Mr. John T. Conway, Chairman Defense Nuclear Facilities Safety Board 625 Indiana Avenue N.W., Suite 700 Washington, D.C. 20004

Dear Mr. Conway:

TRANSMITTAL OF INFORMATION TO COMPLETE DEFENSE NUCLEAR FACILITIES SAFETY BOARD (DNFSB) RECOMMENDATION 93-5 MILESTONE 5.4.3.5c

This letter constitutes completion of DNFSB Recommendation 93-5 Implementation Plan, Revision 1, Milestone 5.4.3.5c, "Letter Reporting Approval of Safety Assessment for Rotary Mode Core Sampling in Flammable Gas Tanks and Documenting Incorporation into the Authorization Basis." The due date for Milestone 5.4.3.5c is September 1996. The U.S. Department of Energy, Richland Operations Office (RL) has completed the actions identified under this milestone and proposes closure of the milestone.

Enclosure 1 is the Safety Assessment (SA) of Rotary Mode Core Sampling in Flammable Gas Single-Shell Tanks developed by the Los Alamos National Laboratory for the Westinghouse Hanford Company (WHC). Enclosure 2 (WHC Engineering Change Notice [ECN] #609990) contains the Interim Operational Safety Requirements (IOSRs) that were developed to implement the SA. Both of these documents were intensively reviewed by the Independent Review Team from Idaho National Engineering Laboratory and my staff with the participation of your staff. The product of the review was the Safety Evaluation Report (SER [Enclosure 3]). This SER was approved by RL in Enclosure 4. The WHC ECN #609990 (Enclosure 2) incorporated the SA and IOSRs into the Interim Safety Basis, which is the Authorization Basis for the Tank Waste Remediation System.

If you have any questions, please contact me or your staff may contact Jackson Kinzer, Assistant Manager for Tank Waste Remediation System, on (509) 376-7591.

Sincerely.

/John D. Wagoner/ Manager

WSD:PRH

Enclosures (4)

cc w/encls: A. Alm, EM-1 R. Guimond, EM-2

R. Izatt, ÉM-2

J. Tseng, EM-4

S. Cowan, EM-30 M. Hunemuller, EM-38 M. Whitaker, S-3.1 S. Trine, RL DNFSB Liaison cc w/o encls: R. F. Bacon, WHC L. F. Ermold, WHC



P.O. Box 1970 Richland, WA 99352

96/3861

RECEIVED 1906 SEP 30 AN 11: 19 DMF SAFETY BOARD

9654064

Mr. J. K. McClusky, Director Waste Storage Division U.S. Department of Energy Richland Operations Office Richland, Washington 99352

Dear Mr. McClusky:

September 12, 1996

COMPLETION OF DEFENSE NUCLEAR FACILITIES SAFETY BOARD RECOMMENDATION 93-5 IMPLEMENTATION PLAN, REVISION 1, COMMITMENT 5.4.3.5c, LETTER REPORTING APPROVAL OF SAFETY ASSESSMENT FOR ROTARY MODE CORE SAMPLING IN FLAMMABLE GAS TANKS AND DOCUMENTING INCORPORATION INTO THE INTERIM SAFETY BASIS

- References: (1) Letter, J. D. Wagoner, RL, to A. L. Trego, WHC, "Authorization of the Safety Assessment of Rotary Mode Core Sampling in Flammable Gas Single-Shell Tanks, WHC-SD-WM-SAD-035, Revision Oa, and Interim Operation Safety Requirements," 96-QSH-042, dated August 30, 1996.
 - (2) WHC-SD-WM-SAD-035, "A Safety Assessment of Rotary Mode Core Sampling in Flammable Gas Single Shell Tanks: Hanford Site, Richland, Washington," dated August 8, 1996.
 - (3) WHC-SD-WM-ISB-001, "Hanford Site Tank Farm Facilities Interim Safety Basis," Rev O-K, dated July 19, 1996.
 - (4) "Safety Evaluation Report of the Safety Assessment document titled 'A Safety Assessment of Rotary Mode Core Sampling in Flammable Gas Single Shell Tanks: Hanford Site, Richland, Washington,'" Lockheed Idaho Technologies Company, dated July 18, 1996.
 - (5) WHC-SD-WM-OSR-005, "Single Shell Tank Interim Operational Safety Requirements," Rev O-E, dated July 18, 1996.

This letter reports completion of Defense Nuclear Facilities Safety Board (DNFSB) Recommendation 93-5 Implementation Plan Milestone 5.4.3.5c, Letter Reporting Approval of Safety Assessment for Rotary Mode Core Sampling in Flammable Gas Tanks and Documenting Incorporation into the ISB.

By Reference 1, the U.S. Department of Energy, Richland Operations Office, authorized the Safety Evaluation Report (Reference 4) documenting the review and approval of the Rotary Mode Core Sampling System Safety Assessment Mr. J. K. McClusky Page 2 September 12, 1996

ł

(Reference 2) and the related Interim Operational Safety Requirements (Reference 5). The attachment to this letter (Engineering Change Notice #609990) incorporates the Safety Assessment and the supporting Operational Safety Requirements into the Tank Farm Interim Safety Basis (Reference 3).

The due date for submittal of this milestone to DNFSB is September 30, 1996.

Very truly yours,

m fur

L. F. Ekodold, Director TWRS Characterization Project Tank Waste Remediation System

srb

Attachment

- HQ J. Poppiti M. A. Mikolanis
- RL R. E. Gerton
 - P. R. Hernandez
 - J. F. Thompson, Jr.
 - S. L. Trine
 - N. W. Willis
 - A. H. Wirkkala (w/o attachment)

PNNL - A. F. Noonan

SAIC - H. G. Sutter

A Safety Assessment of Rotary Mode Core Sampling in Flammable Gas Single Shell Tanks: Hanford Site, Richland, Washington

Prepared by

Nuclear Systems Design and Analysis Group Technology and Safety Assessment Division Los Alamos National Laboratory





96/3861

A SAFETY ASSESSMENT OF ROTARY MODE CORE SAMPLING IN FLAMMABLE GAS SINGLE SHELL TANKS: HANFORD SITE, RICHLAND, WASHINGTON

by

Waste Tank Safety Analysis Team

Nuclear Systems Design and Analysis Group Technology and Safety Assessment Division Los Alamos National Laboratory Los Alamos, New Mexico

WHC-SD-WM-SAD-035, Rev. 0-a

4

U.S. DEPARTMENT OF ENERGY

August 8, 1996

LOS ALAMOS NATIONAL LABORATORY

ACKNOWLEDGMENTS

Contributors to the development of this SA include the listed personnel.

Project Manager

James H. Scott

Team Leader

K. O. Pasamehmetoglu

Principal Investigator

🔍 C. Unal

Technical Contributors

| K. O. Pasamehmetoglu | R. T. Perry | W. O. Miller ¹ |
|----------------------|---------------------------|---------------------------|
| C. Unal | E. A. Rodriguez | D. A. Siebe |
| J. R. White | J. Spore | D. I. Poston |
| D. R. Bennett | P. Šadasivan | P. Martinez |
| W. R. Bohl | J. Lin | D. Hamilton ² |
| P. McClure | A. S. Neuls | J. Smalley ² |
| J. M. Butner | M. K. Sasser | J. Criddle ² |
| K. Stroh | A. Espinoza | J. Robinson ² |
| J. N. Edwards | J. Darby | A. Mousel ² |
| W. L. Kubic | B. Weinstein ¹ | H.H. Ziada² |
| M. Amador | J. Heneage | K. Witwer ² |

¹ Ogden Environmental and Energy Services Co.

² Westinghouse Hanford Company

1

Editor

M. Amador

Word Processor

A. Espinoza

EXECUTIVE SUMMARY

This safety assessment (SA) addresses each of the required elements associated with the installation, operation, and removal of a rotary-mode core sampling (RMCS) device in flammable-gas (FG) single-shell tanks (SSTs). The RMCS operations are needed to retrieve waste samples from SSTs with hard layers of waste for which push-mode sampling is not adequate for sampling.

This SA was prepared using the "Interim Guidance for Preparing Safety Assessments," which was documented in Appendix A of Westinghouse Hanford Company (WHC) report WHC-CM-6-32. The contents of this SA address most of the elements required in the Department of Energy (DOE) Standard (DOE-STD-3011-94) "Guidance for Preparation and Submittal of Basis for Interim Operation (BIO) for DOE Nonreactor Nuclear Facilities." The hazard analysis contained in this SA was performed using the guidance provided in Chapter 2 of the DOE Standard (DOE-STD-3009-94) "Preparation Guide for U. S. Department of Energy Nonreactor Nuclear Facility Safety Analysis Reports." However, these standards generally apply to a facility as opposed to a system or activity, which is the subject of the current SA. The hazard analysis for the system of interest was performed in parallel with the design. All required design changes (with the required protective equipment credited in the accident analysis) are documented in this SA.

In this SA, potential hazards associated with the proposed action were identified and evaluated systematically. Several potential accident cases that could result in radiological or toxicological gas releases were identified and analyzed and their consequences assessed. Administrative controls, procedures, and design changes required to eliminate or reduce the potential of hazards were identified.

The accidents were analyzed under nine categories, four of which were burn scenarios. In SSTs, burn accidents result in unacceptable consequences because of a potential dome collapse. The accidents in which an aboveground burn propagates into the dome space were shown to be in the "beyond extremely unlikely" frequency category. Given the unknown nature of the gas-release behavior in the SSTs, many design changes and administrative controls were implemented to achieve these low frequencies. Likewise, drill string fires and dome space fires were shown to be very low frequency accidents (<1.0E-6/yr) by taking credit for the design changes, controls, and available experimental and analytical data.

Under the category of waste fires, the possibility of igniting the entrapped gases and the waste itself were analyzed. Experiments were conducted at the BOM to demonstrate that the drill bit is not capable of igniting the trapped gas in the waste. Laboratory testing and thermal analysis demonstrated that, under normal operating conditions, the drill bit will not create high enough temperatures to initiate a propagating reaction in the waste. However, system failure that coincides in a waste layer with high organic content and low moisture may initiate an exothermic reaction in the waste. Consequently, a conservative approach based on the current state of the knowledge resulted in limiting the drilling process to a subset of the FG tanks.

Accidents from the chemical reactions and criticality category are shown to result in acceptable risk. Many accidents are shown to result potentially in containment (tank liner) breach below the waste level. Mitigative features are provided for these accidents. Gas-release events (GREs) without burn also are analyzed, and radiological and toxicological consequences are shown to be within risk guidelines. Finally, the consequences of potential spills are shown to be within the risk guidelines.

Accidents associated with external events also are addressed in this SA. For the SSTs, large seismic events with low frequency of occurrence may result in catastrophic dome failure. However, such events and their consequences are independent of the RMCS operations. Lightning is considered a potential initiator for burn accidents.

The conservative consequences of the accidents are compared with the WHC risk guidelines using accident frequencies obtained on a per-tank and per-year basis. All of the accidents analyzed in this SA are shown to meet the radiological and toxicological risk guidelines. The on-site and off-site consequences of a burn in an SST dome space are high because of a potential dome collapse and do not meet the risk guidelines if not mitigated. Mitigated frequency of the dome collapse accident is shown to be $<10^{-6}/yr$.

This SA is written to cover all FG tanks. As discussed in Section 1, a bounding tank is chosen and a bounding set of parameters are used in the analyses. However, all the SSTs are not screened in determining the bounding set of parameters. To address the issue associated with organic reaction, RMCS is currently allowed in a limited number of tanks. These tanks are explicitly identified in this SA. To encompass the flammable gas issues, a checklist is prepared and included in Section 7 of this SA. The checklist includes tank specific parameters that must be screened against the assumptions made in this SA. This checklist is aimed at complementing the Unreviewed Safety Question (USQ) screening process which would be required to apply this SA to any given tank.

EXECUTIVE SUMMARY

This safety assessment (SA) addresses each of the required elements associated with the installation, operation, and removal of a rotary-mode core sampling (RMCS) device in flammable-gas (FG) single-shell tanks (SSTs). The RMCS operations are needed to retrieve waste samples from SSTs with hard layers of waste for which push-mode sampling is not adequate for sampling.

This SA was prepared using the "Interim Guidance for Preparing Safety Assessments," which was documented in Appendix A of Westinghouse Hanford Company (WHC) report WHC-CM-6-32. The contents of this SA address most of the elements required in the Department of Energy (DOE) Standard (DOE-STD-3011-94) "Guidance for Preparation and Submittal of Basis for Interim Operation (BIO) for DOE Nonreactor Nuclear Facilities." The hazard analysis contained in this SA was performed using the guidance provided in Chapter 2 of the DOE Standard (DOE-STD-3009-94) "Preparation Guide for U. S. Department of Energy Nonreactor Nuclear Facility Safety Analysis Reports." However, these standards generally apply to a facility as opposed to a system or activity, which is the subject of the current SA. The hazard analysis for the system of interest was performed in parallel with the design. All required design changes (with the required protective equipment credited in the accident analysis) are documented in this SA.

In this SA, potential hazards associated with the proposed action were identified and evaluated systematically. Several potential accident cases that could result in radiological or toxicological gas releases were identified and analyzed and their consequences assessed. Administrative controls, procedures, and design changes required to eliminate or reduce the potential of hazards were identified.

The accidents were analyzed under nine categories, four of which were burn scenarios. In SSTs, burn accidents result in unacceptable consequences because of a potential dome collapse. The accidents in which an aboveground burn propagates into the dome space were shown to be in the "beyond extremely unlikely" frequency category. Given the unknown nature of the gas-release behavior in the SSTs, many design changes and administrative controls were implemented to achieve these low frequencies. Likewise, drill string fires and dome space fires were shown to be very low frequency accidents (<1.0E-6/yr) by taking credit for the design changes, controls, and available experimental and analytical data.

Under the category of waste fires, the possibility of igniting the entrapped gases and the waste itself were analyzed. Experiments were conducted at the BOM to demonstrate that the drill bit is not capable of igniting the trapped gas in the waste. Laboratory testing and thermal analysis demonstrated that, under normal operating conditions, the drill bit will not create high enough temperatures to initiate a propagating reaction in the waste. However, system failure that coincides in a waste layer with high organic content and low moisture may initiate an exothermic reaction in the waste. Consequently, a conservative approach based on the current

state of the knowledge resulted in limiting the drilling process to a subset of the FG tanks.

Accidents from the chemical reactions and criticality category are shown to result in acceptable risk. Many accidents are shown to result potentially in containment (tank liner) breach below the waste level. Mitigative features are provided for these accidents. Gas-release events (GREs) without burn also are analyzed, and radiological and toxicological consequences are shown to be within risk guidelines. Finally, the consequences of potential spills are shown to be within the risk guidelines.

Accidents associated with external events also are addressed in this SA. For the SSTs, large seismic events with low frequency of occurrence may result in catastrophic dome failure. However, such events and their consequences are independent of the RMCS operations. Lightning is considered a potential initiator for burn accidents.

The conservative consequences of the accidents are compared with the WHC risk guidelines using accident frequencies obtained on a per-tank and per-year basis. All of the accidents analyzed in this SA are shown to meet the radiological and toxicological risk guidelines. The on-site and off-site consequences of a burn in an SST dome space are high because of a potential dome collapse and do not meet the risk guidelines if not mitigated. Mitigated frequency of the dome collapse accident is shown to be $<10^{-6}/yr$.

This SA is written to cover all flammable gas tanks. As discussed in Section 1, a bounding tank is chosen and a bounding set of parameters are used in the analyses. However, all the single-shell tanks are not screened in determining the bounding set of parameters. To address the issue associated with organic reaction, rotary mode core sampling is currently allowed in a limited number of tanks. These tanks are explicitly identified in this SA. To encompass the flammable gas issues, a checklist is prepared and included in Section 7 of this SA. The checklist includes tank specific parameters that must be screened against the assumptions made in this SA. This checklist is aimed at complementing the Unreviewed Safety Question (USQ) screening process which would be required to apply this SA to any given tank.

| | TABLE OF CONTENTS | Page |
|---------------|--|-------|
| EXE | ECUTIVE SUMMARY | 1 |
| AC | KNOWLEDGMENTS | ••••• |
| LIST | T OF FIGURES | xiv |
| LIST | I OF TABLES | xxiv |
| ACI | RONYMS | xx |
| DEF | FINITIONS | xxxiv |
| | | |
| 1.0. | SCOPE | 1-1 |
| 1.1. | BACKGROUND AND PURPOSE OF THE PROPOSED ACTION | 1-1 |
| 1 .2 . | SAFETY ASSESSMENT-SCOPE AND APPROACH | 1-2 |
| 1.3. | NO-ACTION ALTERNATIVE | |
| 1.4. | SUMMARY OF SIGNIFICANT CHARACTERISTICS OF THE SST FARMS AND THEIR ENVIRONMENT | |
| 1.5. | REFERENCES | 1-10 |
| | | |
| 2.0. | DESCRIPTION OF ACTION | 2-1 |
| 2.1. | PRINCIPAL SAFETY CRITERIA | 2-2 |
| | 2.1.1. DOE Safety and Design Requirements | 2-2 |
| | 2.1.2. WHC General Safety Criteria | 2-3 |
| | 2.1.3. Safety Criteria for Tanks on the FGWL | 2-3 |
| | 2.1.4. Safety Criteria for Tanks on the Organic Watch List | 2-4 |
| 2.2. | PHENOMENOLOGY | 2-4 |
| | 2.2.1. Gas Storage and Release Mechanisms in SSTs | 2-5 |
| | 2.2.2. Gas Composition | 2-5 |

i

| | | | • |
|------|--------|--|--------------|
| | 2.2.3. | Waste Characteristics | 2-6 |
| | 2.2.4. | Flammability and Ignition | 2-7 |
| 2.3. | FLAM | MABLE-GAS TANKS | |
| 2.4. | RMCS | SYSTEM AND ASSOCIATED EQUIPMENT | 2-10 |
| | 2.4.1. | RMCS Trucks | 2-10 |
| | | 2.4.1.1. Platform Hoist | 2-11 |
| | | 2.4.1.2. Grapple Hoist Assembly | 2- 11 |
| | | 2.4.1.3. Drill Unit and Quill Rod | 2-13 |
| | | 2.4.1.4. Drill String Assembly | 2-13 |
| | | 2.4.1.5. Universal Sampler | 2-14 |
| | | 2.4.1.6. Shielded Receiver Assembly | 2- 15 |
| | | 2.4.1.7. Purge Gas Enclosure Assembly | 2-16 |
| | | 2.4.1.8. Instrumentation and Control System | 2-17 |
| | | 2.4.1.8.1. Programmable Logic Controllers (PLC) | 2-17 |
| | | 2.4.1.8.2. Engine Shutdown | 2-17 |
| | | 2.4.1.8.3. Instrumentation Cabinet | 2-18 |
| | 2.4.2. | 2.4.1.8.4. Hydraulic Button Detector Exhauster Assembly | 2-19 2-19 |
| | 2.4.3. | Flammable Gas Detection System | 2-21 |
| | 2.4.4. | Riser and Adapter Equipment | 2-22 |
| | 2.4.5. | Cable Spray Washer Assembly and Change-out Assembly | 2-23 |
| | 2.4.6. | Primary Support Equipment | 2-24 |
| | | 2.4.6.1. Liquid Nitrogen Supply Trailer and Heater | 2-24 |
| | | 2.4.6.2. Mobile X-ray Imaging System | 2-25 |
| | | 2.4.6.3. Casks and Sample Transfer Truck | 2-25 |

| | | • |
|------|--|---------------|
| | 2.4.6.4. Portable Generator Set | |
| | 2.4.6.5. Power Distribution Trailer | |
| | 2.4.6.6. Breathing Air Trailer/Compre | ssor2-25 |
| | 2.4.6.7. Support Truck and Trailer | |
| | 2.4.6.8. Crane | 2-26 |
| | 2.4.6.9. Light Units | 2-26 |
| | 2.4.6.10. Tent | 2-26 |
| | 2.4.6.11. Video Vehicle | 2-26 |
| | 2.4.7. RMCS System Weights | 2-26 |
| 2.5. | RMCS OPERATIONS | |
| | 2.5.1. Preinstallation | |
| | 2.5.2. Installation | 2-30 |
| | 2.5.3. Sampling Operations | 2-31 |
| | 2.5.3.1 Drilling | |
| | 2.5.3.2. Removing the Universal Sampler | 2-33 |
| | 2.5.3.3. Mobile x-ray Image System Operation | 52-34 |
| | 2.5.3.4. Sample into Receiving Cask | 2-34 |
| • | 2.5.3.5. New Sampler Preparation | 2-34 |
| | 2.5.3.6. New Sample Insertion into Drill Strin | ıg2-35 |
| | 2.5.4. DS Removal | 2-35 |
| | 2.5.5. Operational Conditions and Characteri | stics2-36 |
| | 2.5.5.1. Normal Operations | 2-36 |
| | 2.5.5.1.1. Normal Drilling and Sample | Retrieval2-36 |
| | 2.5.5.1.2. Walkdown Mode | 2-37 |
| | 2.5.5.1.3. Bottom Detection | 2-37 |

| 2.5.5.1.4. Stabilized Mode | 2-38 |
|--|------|
| 2.5.5.2. Abnormal Drilling Conditions | 2-38 |
| 2.5.5.2.1. Reduced Nitrogen Flow | 2-38 |
| 2.5.5.2.2. Excess rpm or Downforce on Drill String | 2-38 |
| 2.5.5.2.3. Penetration Rate | 2-39 |
| 2.5.5.2.4. Drill Bit Plug In | 2-39 |
| 2.5.5.2.5. Stuck Drill Bit | 2-39 |
| 2.5.5.2.6. DS Flooding or Structural Failure | 2-39 |
| 2.5.5.3. Loss of exhauster | 2-39 |
| 2.5.5.4. Gas-release Event | 2-40 |
| 2.5.5.5. Emergency Response | 2-40 |
| 2.5.6. Restart | 2-40 |
| 2.6. REFERENCES | 2-42 |
| 3.0. IDENTIFICATION OF HAZARDS | 3-1 |
| 3.1. METHODOLOGY | 3-1 |
| 3.2. RESULTS OF THE ROTARY-CORE DRILLING SYSTEM HAZARDS IDENTIFICATION | 3-4 |
| 3.3. REFERENCES | 3-16 |
| 4.0. HAZARD ANALYSIS | 4-1 |
| 4.1. ABOVEGROUND FIRE ACCIDENTS | 4-5 |
| 4.1.1. Flammable-Gas Release to Exhauster and Burn (Ignition in the Exhauster, Electrostatic and Electrical/Mechanical, and Other Spark Sources, Operation, and Removal) | .4-5 |
| 4.1.1.1. Static Electricity | .4-7 |
| 4.1.1.2. Lightning | .4-7 |

| | 4113 Electrical Spark from Nonqualified Electrical | • |
|------------|---|------|
| | Equipment | 4-7 |
| | 4.1.1.4. Mechanical Sparks | 4-8 |
| | 4.1.1.5. Summary of Exhauster Fires | 4-9 |
| 4.1.2. | Flammable-Gas Release Through Torn Duct and Burn (Operation) | 4-9 |
| | 4.1.2.1. Flammable-Gas Release through Open Riser (or Possible Leak Paths) Driven by Gas Release Event and Burn (Operation and Removal) | 4-11 |
| | 4.1.2.2. Flammable-Gas Release through Shielded Receiver (SR) and Ignition in the Shielded Receiver (Operation) | 4-14 |
| 4.1.3. | Flammable Gas and Ignition in X-ray or Cask | 4-19 |
| 4.1.4. | Flammable Gas Accumulation or Release Caused by the Loss of Electrical Power and Ignition Source (Installation, Operation, and Removal) | 4-19 |
| 4.1.5. | Propane Release from Refueling of Nitrogen Trailer Propane Tank and Fire (Installation, Operation, and Removal) | 4-20 |
| 4.1.6. | Flammable Diesel and Gasoline Fuel Fire (Installation, Operation, and Removal) | 4-20 |
| 4.1.7. | Collision Caused by Trucks and Other Equipment (Installation, Operation, and Removal) | 4-21 |
| 4.1.8. | Summary of Aboveground Fire Accident Category | 4-22 |
| 4.2. GAS R | ELEASE AND DOME FIRE ACCIDENTS | |
| 4.2.1. | Ignition of Flammable Gas as a Result of a Drill String Break (Operation) | 4-22 |
| 4.2.2. | Ignition of Flammable Gas in the Riser Caused by the Drill-String, Equipment or Tool Drop (Installation and Removal) | 4-23 |
| 4.2.3. | Flammable-Gas Release and Burn Caused by the Drill String or Tool Drop on Crust (Installation, Operation, and Removal) | 4-23 |

| | 4.2.4. | Ignition of Flammable Gas in the Riser Caused by a Frictional Spark (Installation, Operation, and Removal) | |
|------|--------|---|------|
| | 4.2.5. | Ignition of Flammable Gas in the Riser Caused by Electrostatic Spark (Installation, Operation, and Removal) (Rubber Frisbee and Drill String) | 4-28 |
| | 4.2.6. | Ignition of Flammable-Gas Release Caused by Crust Penetration and Frictional Sparks Caused by the Drill Bit (Operation) | 4-28 |
| | 4.2.7. | Ignition of Flammable Gas in the Dome (Operation) | 4-30 |
| | 4.2.8. | Ignition Of Dust and Flammable Gas in the Dome (Operation and Removal) (Static Electrical Charges and Other Causes) | 4-30 |
| | 4.2.9. | Summary of the Dome Fire Accident | 4-30 |
| 4.3. | DRILL | STRING FIRE ACCIDENTS | 4-31 |
| | 4.3.1. | Flammable-Gas Penetration into the Drill String (Operation) | 4-31 |
| | 4.3.2. | Ignition of Flammable Gas in the Drill String Caused by Drops (Operation) | 4-33 |
| | 4.3.3. | Ignition of Flammable Gas in the Drill String Caused by Assembly/Disassembly of Drill Rods or Drill Rod-Quill Rod Adapter Impact | 4-35 |
| | 4.3.4. | Ignition of Flammable Gas in the Drill String Caused by Unqualified Electrical Equipment in the Drill Head or Shielded Receiver (Operation) | 4-38 |
| | 4.3.5. | Ignition of Flammable Gas in the Drill String Caused by Drill-String Failure (Operation) | 4-39 |
| | 4.3.6. | Ignition of Flammable Gas in the Drill String Caused by Frictional Sparks (Operation) | 4-39 |
| | 4.3.7. | Ignition of the Flammable Gas in the Drill String Caused by Frictional Sparks from the Shear Pin Break (Operation) | 4-40 |
| | 4.3.8. | Detonation in the Drill String and Drill-String Ejection (Operation) | 4-40 |
| | 4.3.9. | Ignition in the Drill String Caused by Lightning | 4-41 |

August 8, 1996

| | 4.3.10 |). Ignition in the Drill String Caused by Static Electricity Between Seals and Rotating Parts | 4-41 |
|------|--------|---|------|
| | 4.3.11 | . Summary of Drill-String Fires | 4-41 |
| 4.4. | WAST | E FIRE ACCIDENTS | |
| | 4.4.1. | Waste Fire Caused by Drill Bit Over-Temperature (Operation) | 4-42 |
| | | 4.4.1.1. Summary of Controls for Drilling Operation | 4-45 |
| | | 4.4.1.2. Accident Frequency of Waste Ignition | 4-45 |
| | 4.4.2. | Waste Fire as a Result of Exothermic Reaction Caused by Incompatible Materials (Installation, Operation, and Removal) | 4-48 |
| · | 4.4.3. | Crust Burn Caused by Drop Impact (Installation, Operation, and Removal) | 4-49 |
| | 4.4.4. | Flammable-Gas Ignition under the Waste Surface Caused by Friction From the Rotation or Penetration of the Drill Bit (Operation) | 4-49 |
| | 4.4.5. | Summary of Waste Fire Accidents | 4-52 |
| 4.5. | CHEM | ICAL REACTIONS AND CRITICALITY ACCIDENTS | 4-52 |
| | 4.5.1. | Criticality Caused by Drilling (Operation and Removal) | 4-52 |
| | 4.5.2. | Water Addition and Temperature Limits (Operation and Removal) | 4-52 |
| | 4.5.3. | Waste Ignition and Melting (Operation) | 4-52 |
| | 4.5.4. | Energy Addition to the Tank (Operation) | 4-53 |
| | 4.5.5. | Impact-Sensitive Compounds | 4-53 |
| 4.6. | CONT | AINMENT BREACH ACCIDENTS | 4-53 |
| | 4.6.1. | Excessive Static and Dynamic Tank Dome Loading (Installation, Operation and Removal) | 4-54 |
| | 4.6.2. | Penetration of Tank Bottom, Drill String Drop, and Drilling against the Liner (Installation, Operation and Removal) | 4-54 |

 $\frac{1}{2}$

| | 4.6.3 . | Failure of the Drill String (Installation, Operation, and Removal) | 4-55 |
|------|------------------|---|------|
| | 4.6.4. | Riser Damage (Installation, Operation, and Removal) | 4-56 |
| | 4.6.5. | Tank Wall Penetration (Operation) | 4-57 |
| 4.7. | GAS R | ELEASES WITHOUT BURN | 58 |
| | 4.7.1. | Toxic-Gas Releases (Installation, Operation, and Removal) | 4-58 |
| | 4.7.2. | Unfiltered Releases | 4-58 |
| | 4.7.3. | Steam Release and Moisture Removal from Tank (Operation) | 4-58 |
| | 4.7.4. | Undetected Gas Release (Operation) | 4-58 |
| | 4.7.5. | Nitrogen Additions (Installation, Operation, and Removal) | 4-58 |
| 4.8. | SPILLS, RADIA | , RELEASES, AND HAZARDOUS MATERIAL AND TION EXPOSURES | 4-58 |
| | 4.8.1. | HEPA Releases as a Result of Excessive Loading and HEPA Failures (Operation and Removal) | 4-59 |
| | 4.8.2. | Exhauster Continuous Release after Filter Failure (Operatio and Removal) | 4-59 |
| | 4.8.3. | Inlet Duct Failure (Operation and Removal) | 4-59 |
| | 4.8.4. | Releases from Core Sampler Drops (Operation) | 4-59 |
| | 4.8.5. | Open Sampler Spills (Operation) | 4-62 |
| | 4.8 .6 . | Spill from Accumulated Waste in Core Barrel | 4-62 |
| | 4.8.7. | Ineffective Decontamination and Drop of Drill String | 4-62 |
| | 4.8.8. | Radiation Exposures (Installation, Operation, and Removal) | 4-62 |
| | | 4.8.8.1. Exposures from HEPA filters | 4-62 |
| | | 4.8.8.2. Radiation Doses from the Shielded Receiver | 4-63 |
| | | 4.8.8.3. Radiation from the X-ray Machine | 4-63 |

| | WHC-SD-WM-SAD-035, Rev. 0-a |
|-------------|---|
| | 4.8.8.4. Excessive Radiation Exposure as a Result of Personnel Being in a Direct Line of Sight to the Tank Waste (Installation and Removal)4-63 |
| | 4.8.8.5. Excessive Personnel Radiation Exposure during FG/RMCS Decontamination4-63 |
| 4.9. EXTE | RNAL EVENTS |
| 4.9.1 | Lightning4-64 |
| 4.9.2 | Wind4-66 |
| 4.9.3 | Fire |
| 4.9.4 | Earthquake4-67 |
| 4.9.5 | Tornado4-67 |
| 4.9.6 | Flood and High Rainfall |
| 4.9.7 | Volcanism4-67 |
| 4.9.8 | Dust Devils |
| 4.10. REFEI | XENCES |
| 5.0. CONS | EQUENCES OF ACCIDENTS |
| 5.1. RISK (| GUIDELINES |
| 5.1.1 | Radiological Risk Criteria |
| 5.1.2 | Toxicological Risk Guidelines5-2 |
| 5.2. ACCII | DENT CONSEQUENCES |
| 5.3. CONS | EQUENCES OF DOME COLLAPSE ACCIDENTS |
| 5.3.1 | Radiological Consequences5-8 |
| 5.3.2 | Toxicological Consequences5-8 |
| | 5.3.2.1. Waste Material5-8 |
| | 5.3.2.2. Toxic Gases |
| 5.4. GRE V | VITHOUT A BURN |

August 8, 1996

Ş

ix

| | 5.4.1. | Toxic-Gas Releases | .5-9 |
|--------------|-----------------|--|---------------|
| | | 5.4.1.1. Toxicological Consequences | 5-12 |
| | | 5.4.1.2. Radiological Consequences | 5-14 |
| | 5. 4.2 . | Unfiltered Material Releases | 5-14 |
| | | 5.4.2.1. Radiological Consequences | 5-16 |
| | 5.4.3. | Toxicological Consequences | 5-16 |
| 5.5. | SPILL A | ACCIDENTS | 5-17 |
| | 5.5.1. | Radiological Consequences | 5-17 |
| | 5.5.2. | Toxicological Consequences5 | 5-18 |
| 5.6. | COMP. THE RO | ARISON OF THE ACCIDENT CONSEQUENCES WITH GS | 5-19 |
| | 5.6.1. | Accidents with Radiological Consequences | 5- 2 0 |
| | 5.6.2. | Toxic-Gas Releases | 5-20 |
| | 5.6.3. | Toxic Liquid/Solid Releases5 | 5-20 |
| 5.7. | REFER | ENCES | 5-2 8 |
| 6 .0. | CONTR | OLS | 6-1 |
| 6.1. | ADMIN | VISTRATIVE CONTROLS | 6-1 |
| 6.2. | LEVEL | DESIGNATION PROCESS | 6-2 |
| 6.3. | FG/RM | ICS SAFETY DESIGN FEATURES | 6-3 |
| | 6.3.1. | Worker Health and Safety Requirements | .6-3 |
| 6.4. | FG/RM | ICS CONTROLS | 6-9 |
| | 6.4.1. | FG/RMCS Controls Phases | .6-9 |
| | 6.4.2. | Control Table Columns6 | j -10 |
| 6.5. | ALL PH | HASES | 6-11 |
| | 6.5.1. | Use of FG/RMCS Controls6 | 5-11 |

August 8, 1996

| | 6.5.1.1. | Control Applicability | 6-11 |
|--------|----------|--|------|
| | 6.5.1.2. | Existing procedures and controls | 6-11 |
| 6.5.2. | Flamm | able Gas Detection System | 6-11 |
| | 6.5.2.1. | Flammable gas detection system calibration | 6-11 |
| | 6.5.2.2. | Flammable gas detection system operation | 6-12 |
| | 6.5.2.3. | Tank gas pressure detection system | 6-13 |
| 6.5.3. | Spark-f | free environment | 6-13 |
| | 6.5.3.1. | Spark-free environment | 6-13 |
| 6.5.4. | Materia | al compatibility | 6-13 |
| | 6.5.4.1. | Material compatibility | 6-13 |
| 6.5.5. | Exhaus | ster emissions | 6-14 |
| 6.5.6. | Simulta | aneous activities | 6-14 |
| | 6.5.6.1. | FG/RMCS vehicle operations on tank | 6-14 |
| | 6.5.6.2. | Simultaneous activities on FG/RMCS tank | 6-14 |
| | 6.5.6.3. | Inter-connected tanks | 6-15 |
| 6.5.7. | Refueli | ng Activities | 6-15 |
| | 6.5.7.1. | Flammable gas concentrations | 6-15 |
| | 6.5.7.2. | Equipment refueling | 6-19 |
| 6.5.8. | Drill St | tring Support | 6-16 |
| | 6.5.8.1. | Locking wench | 6-16 |
| 6.5.9. | Externa | al events | 6-16 |
| | 6.5.9.1. | Meteoro-logical verification | 6-16 |
| | 6.5.9.2. | Range fires | 6-16 |
| | 6.5.9.3 | Thunder-storm & lightning, high winds, dust devils and tornadoes | 6-16 |

| | | 6.5.9.4. High rain fall or flooding, seismic activity, and volcanic activity/ashfall | 6-16 |
|------|-----------------|--|---------------|
| 6.6. | PRE-IN | ISTALLATION PHASE | 6-18 |
| | 6.6.1. | Waste Reactivity Fraluation | 6-18 |
| | | 6.6.1.1. Waste reactivity | 6-18 |
| | 6.6 .2 . | Tank and riser conditions | 6-18 |
| | | 6.6.2.1. Gas leak paths | 6-18 |
| | | 6.6.2.2. Open riser exclusion zone | 6-19 |
| | 6.6.3. | Grounding and bonding | 6-19 |
| | | 6.6.3.1. Equipment attached to tank | 6-19 |
| | | 6.6.3.2. Equipment not attached to tank | 6-20 |
| | 6.6.4. | Energized equipment | 6-20 |
| | | 6.6.4.1. Energized equipment in the dome | 6-20 |
| | | 6.6.4.2. Energized equipment in open riser exclusion zones | 6-20 |
| | 6.6.5. | Equipment positioning and Setup | 6-21 |
| | | 6.6.5.1. Tank loading | 6-21 |
| | | 6.6.5.2. Vehicle operations on tank | 6-21 |
| | | 6.6.5.3. Truck ramp and jacks | 6-21 |
| | | 6.6.5.4. Truck position | 6 -2 1 |
| | | 6.6.5.5. Lift operations | 6-22 |
| | 6.6.6. | Exhauster setup | 6-22 |
| | | 6.6.6.1. Exhauster installation | 6-22 |
| | | 6.6.6.2. Exhauster duct | 6-22 |
| | | 6.6.6.3. Duct exclusion zone | 6-23 |
| | | 6.6.6.4. Exhauster fan | 6-23 |

 $\overline{\tau}$

| | 6.6.7. | Tank Breather Inlet HEPA Fil | ter | 6-23 |
|---------------|--------|---|---------------------|---------------|
| | | 6.6.7.1. Portable Inlet Stack | | 6-23 |
| 6.7. | INSTA | LATION PHASE | | 6-25 |
| | 6.7.1. | Riser preparation | | 6-25 |
| | | 6.7.1.1. Gas concentrations containment | while breaching tan | k 6-25 |
| | | 6.7.1.2. Riser exposure | | 6 -2 5 |
| | | 6.7.1.3. Riser equipment | | 6 -2 5 |
| | | 6.7.1.4. Sleeve insertion | | 6-25 |
| | | 6.7.1.5. Frisbee installation | | 6-26 |
| | | 6.7.1.6. Frisbee/DS interface l | ubrication | 6-26 |
| 6 <i>.</i> 8. | OPERA | TIONS PHASE | | 6-27 |
| | 6.8.1. | Drill Bit | •••••• | 6-27 |
| | | 6.8.1.1. Drill bit qualification. | | 6-27 |
| | 6.8.2. | Core Sampler | | 6-27 |
| | | 6.8.2.1. Core sampler qualific | ation | 6-27 |
| | | 6.8.2.2. Leakage | | 6-27 |
| | 6.8.3. | Drill string | | 6-27 |
| | | 6.8.3.1. Drill bit, core barrel ar | d drill rods | 6-27 |
| | | 6.8.3.2. DS open to atmosphe | re | 6-28 |
| | | 6.8.3.3. Gas accumulation | | 6-28 |
| | | 6.8.3.4. Waste flooding | | 6-28 |
| | 6.8.4. | Shielded Receiver | | 6-29 |
| | | 6.8.4.1. Sampler retrieval | | 6-29 |
| | | 6.8.4.2. Waste accumulation. | | 6-29 |

| | 6.8.4.3. | Waste contamination | 6-29 |
|----------------|--------------------------|---|------|
| 6.8.5. | RLU/G | Grapple | 6-29 |
| | 6.8.5.1. | Travel rate | 6-29 |
| | 6.8.5.2 . | Grapple Load | |
| 6.8.6. | X-ray | | |
| | 6.8.6.1. | Sampler insertion | 6-30 |
| | 6.8.6.2. | X-ray sample liner | 6-30 |
| 6.8.7. | Exhaus | ter | |
| | 6.8.7.1. | Duct inspections | 6-30 |
| | 6.8.7.2. | Operations | 6-30 |
| | 6.8.7.3. | Riser aerosols | 6-31 |
| | 6.8.7.4. | Exhauster-induced tank pressure | 6-31 |
| | 6.8.7.5. | HEPA radiological monitoring | |
| | 6 .8.7. 6. | Operational-induced exhauster shut down | 6-32 |
| 6.8.8. | DS Nitr | rogen purge system | 6-32 |
| | 6.8.8.1. | Bypass leakage verification | 6-32 |
| 6.8.9 . | Riser/s | leeve Purge Flow | 6-33 |
| | 6.8.9.1. | Bypass leakage verification | 6-33 |
| 6.8.10 | . Hydros | tatic Head | 6-33 |
| | 6 .8.1 0.1. | . Bypass leakage verification | 6-33 |
| 6.8.11 | . Instruπ | nentation and controls | 6-33 |
| | 6.8.11.1 | . Truck I&C | 6-33 |
| | 6. 8.11 .2 | . Exhauster I&C | |
| 6.8.12 | . Rotary | Drilling Operations | 6-34 |

| | | 6.8.12.1. Installation and removal of drill rod spray | |
|--------------|--------|---|---------------|
| | | washer and change-out assembly | 6-34 |
| | | 6.8.12.2. Last sampler | 6-34 |
| | 6.8.13 | Rotary Drilling Operational Parameters | 6-34 |
| | | 6.8.13.1. Down force | 6-34 |
| | | 6.8.13.2. Down Force with DS length > 45 ft | 6-35 |
| | | 6.8.13.3. RPM | 6-35 |
| | | 6.8.13.4. RPM with DS length > 45 ft | 6-36 |
| | | 6.8.13.5. Nitrogen purge flow | 6- 36 |
| | | 6.8.13.6. Riser/sleeve purge operations | 6-37 |
| | | 6.8.13.7. Slow penetration rate | 6-37 |
| | 6.8.14 | . Shut down and Restart Conditions | 6-38 |
| | | 6.8.14.1. Drill Rig shut down | 6-38 |
| | | 6.8.14.2. GRE shut down | 6 -3 9 |
| | | 6.8.14.3. GRE restart | 5- 3 9 |
| | | 6.8.14.4. Power loss restart | 5 -39 |
| | 6.8.15 | . Water addition | 6-40 |
| | | 6.8.15.1. Quantity | 6-40 |
| | | 6.8.15.2. Temperature | 5-40 |
| 6.9 . | REMO | VAL PHASE | 6-41 |
| | 6.9.1. | DS Removal | 5-41 |
| | | 6.9.1.1. DS removal | 5-41 |
| | 6.9.2. | Decontamination | 5-41 |
| | | 6.9.2.1. Decontamination system | 5-41 |
| | | 6.9.2.2. Riser exposure | 5-41 |

| 6.9.2.3. | Drill rod removal | 1 |
|--------------------|-------------------|---|
| 6.10. REFERENCES | | 2 |
| 7.0. CHECKLIST ITE | MS | |

| APPENDIX A | HAZARD IDENTIFICATION | A-1 |
|------------|---|-----|
| APPENDIX B | EQUIPMENT RATING AND AUTOMATIC SHUT-DOWN REQUIREMENTS | B-1 |
| APPENDIX C | BOUNDING GAS COMPOSITION, LEAN FLAMMABILITY LIMIT, AND ALLOWABLE GAS RELEASE AND BURN VOLUMES | C-1 |
| APPENDIX D | RELIABILITY OF SYSTEMS | D-1 |
| APPENDIX E | ACCIDENT SEQUENCES | E-1 |
| APPENDIX F | THERMAL ANALYSIS OF ROTARY DRILLING | F-1 |
| APPENDIX G | WASTE AND CRUST IGNITION | G-1 |
| APPENDIX H | DUST EXPLOSION IN THE DOME | H-1 |
| APPENDIX I | DOME COLLAPSE ACCIDENTS | I-1 |
| APPENDIX J | HYDROGEN DIFFUSION INTO DRILL STRING AND BURN/DETONATION ANALYSIS | J-1 |
| APPENDIX K | TOXICOLOGICAL CONSEQUENCE ANALYSIS METHODOLOGY | K-1 |
| APPENDIX L | PROBABILITY OF GAS-RELEASE EVENTS DURING INTRUSION | L-1 |
| APPENDIX M | UNFILTERED RELEASES AND NITROGEN ADDITIONS | M-1 |
| APPENDIX N | STRUCTURAL ISSUES ASSOCIATED WITH ROTARY MODE CORE DRILLING OPERATIONS | N-1 |
| APPENDIX O | ENERGY AND WATER ADDITIONS/REMOVALS | 0-1 |
| APPENDIX P | ANALYSIS OF FRICTIONAL HEATING IN ROTARY-MODE CORE SAMPLING | P-1 |
| APPENDIX Q | SAMPLE PROBLEM FOR THE COMPUTATION OF THE SITE-WIDE RISK | Q-1 |
| APPENDIX R | SOURCE TERM, CRITICALITY, RADIATION EXPOSURES AND CONSEQUENCE CALCULATION METHODOLOGY | R-1 |

| APPENDIX S | SPILL RELEASE FRACTIONS |
|------------|---|
| APPENDIX T | BUREAU OF MINES IGNITION TEST PROGRAM FUNCTIONAL REQUIREMENTS AND ACCEPTANCE CRITERIA |
| APPENDIX U | FUNCTIONAL REQUIREMENTS FOR FLAMMABLE AND TOXIC GAS (HYDROGEN AND AMMONIA) SENSORSU-1 |

| LIST | OF | FIG | URES |
|------|----|-----|------|
| ۰. | | | |

Page

| Fig. 1 - 1. | Location of the Tank Farms within the 200 Area1-4 |
|--------------------|--|
| Fig. 1-2. | Location of 200 Area within the Hanford Site1-5 |
| Fig. 1-3. | Locations of surface-water ponds, ditches, and ephemeral streams on the Hanford Site1-7 |
| Fig. 2-1. | Typical A-Farm SST cross section2-9 |
| Fig. 2-2. | Sampling truck configuration2-11 |
| Fig. 2-3. | Grapple hoist assembly2-12 |
| Fig. 2-4. | Grapple Schematic2-12 |
| Fig. 2-5. | Universal sampler2-14 |
| Fig. 2-6. | Shielded receiver assembly2-16 |
| Fig. 2 - 7. | Mechanical remote latch unit configuration2-17 |
| Fig. 2-8. | Exhauster system2-20 |
| Fig. 2-9. | Riser adapter, washer/frisbee wiper assembly and foot clamp2-23 |
| Fig. 2-10. | Change-out assembly2-24 |
| Fig. 2- 11. | FG/RMCS process2-28 |
| Fig. 2-12. | Relative placement of equipment required for core sampling in the rotary mode |
| Fig. 2-13. | Pre-drilling configuration2-32 |
| | |
| Fig. 5-1. | Comparison of calculated radiological doses with the on-site RGs5-22 |
| Fig. 5 -2 . | Comparison of calculated radiological doses with the off-site RGs5-22 |
| Fig. 5-3. | On-site ammonia concentrations for gas release accidents |

Fig. 5-4. Off-site ammonia concentrations for the GRE accidents......5-24

| Fig. 5-5. | Consequence of toxic waste releases at the on-site receptor location5-26 |
|------------------------|--|
| Fig. 5-6. | Consequence of toxic waste release at the off-site receptor location5-26 |
| Fig. B-1. | Probability of exceeding 2.4% H ₂ B-5 |
| Fig. B-2. | Necessary trip points to shut down the equipment before the concentration reaches 2.4% H ₂ during constant rate of concentration increase |
| Fig. C-1. | Schematic of an SST. (Dish bottom is not represented in this figure.) C-2 |
| Fig. F-1. | Sketch illustrating the cutting process of the drill bitF-1 |
| Fig. F-2. | Drilling energy flow diagramF-7 |
| Fig. F-3. | Drawing of Drill BitF-8 |
| Fig. F-4. | Free body diagram of drill bit toothF-9 |
| Fig. F-5. | Test 15 - Measured bit and calculated bit and chip temperaturesF-20 |
| Fig. F-6. Fig. F-7. | Test 16 - Measured temperature dataF-25 Average chip temperature decrease predicted using (a) forced convection correlation and (b) natural convection correlationF-27 |
| Fig. F-8(a |). Comparison of calculated and measured temperatures of Test 16F-28 |
| Fig. F-8(b |). Comparison of calculated and measured temperatures of Test 23F-28 |
| Fig. F-8(c |). Comparison of calculated and measured temperatures of Test 15F-29 |
| Fig. F-8(d | Comparison of calculated and measured temperatures of Test 17F-29 |
| Fig. F-9. | Time to reach 150°C as a function of torque for various penetration rates and an initial temperature of 22°CF-30 |
| Fig. F-10. | Time to reach 150°C as a function of torque for various penetration rates and an initial temperature of 90°CF-30 |
| Fig. F-11. | Torque at which waste temperature reaches 150°C as a function of penetration rate |

| Fig. F-12. | Temperature histories during an abnormal operation; speed is increasedF-33 |
|--------------------|--|
| Fig. F-13. | Temperature histories during an abnormal operation force is increasedF-34 |
| Fig. G - 1. | Proposed criteria for no propagation of exothermic reactions in the waste |
| Fig. G-2. | Event tree for propagating exothermic reactionsG-12 |
| Fig. I-1. | Concentration for 1E+6 particles/cm ³ using a lognormal distributionI-5 |
| Fig. I-2. | Schematic of dome collapse on salt cakeI-6 |
| Fig. I-3. | Coherent collapsing mass and associated respirable releaseI-7 |
| Fig. I-4. | Respirable release treating a burn using an explosion prescriptionI-8 |
| Fig. I-5 | Amount of aerosolized material as a function of an energy- absorbing layerI-9 |
| Fig. I-6. | Incoherent model of the collapsing domeI-10 |
| Fig. J-1 . | Average hydrogen concentration in the drill stringJ-5 |
| Fig. J-2. | Hydrogen volume fraction versus axial location for small flow area caseJ-6 |
| Fig. J-3. | Hydrogen volume fraction versus axial position for big flow area caseJ-7 |
| Fig. L-1. | Gas-release rates as a function of timeL-10 |
| Fig. L-2. | Peak dome concentration as a function of gas-release rateL-11 |
| Fig. L-3. | Dome space concentration as a function of release volume and timeL-13 |
| Fig. M-1. | Bubble-diameter to yield-stress relation at the initiation of a bubble motion |

| Fig. N-1 | . Buckling limits for drill strings considering three boundary conditions |
|-----------|--|
| Fig. N-2 | . Sketch of a drill string in riserN-5 |
| Fig. N-3 | . Resonance speed caused by torsional loadN-10 |
| Fig. N-4 | . Transverse resonance rotation speed as a function of the drill string length |
| Fig. N-5 | Cross-sectional view of A-101N-12 |
| Fig. P-1. | Schematic view of a sliding interface between real materials, showing the apparent and real areas of contactP-2 |
| Fig. P-2. | Cross-sectional view of the drill string in the conducting sleeve (to scale)P-4 |
| Fig. P-3. | Vertical section of the drill string in the conducting sleeve (to scale)P-5 |
| Fig. P-4. | Azimuthal noding of the sleeve. Nine azimuthal nodes are actually calculated (symmetry gives the effect of 17 nodes with an adiabatic boundary opposite the contact region)P-9 |
| Fig. P-5 | Estimated peak surface temperature as a function of time after initial contact. The lower curve is calculated with the numerical model described. The upper curve results when the conservative "flash temperature rise" is added to the numerical results |
| Fig. P-6. | Transient cooling curve for a "worst-case" hot spot. It can be seen that the very small (13.1 mm ²) hot spot cools to below 400°C in less than 0.6 ms when exposed from under the contact areaP-11 |
| Fig. P-7. | Transient cooling curve for a 20- μ m-thick hot-melt smear as a function of time after the melt is swept out of the contact zone. At that point, heating stops and conduction to the cool surface begins. The plateau represents the time required to remove the heat of fusion from the melt |
| Fig. P-8. | Transient cooling curve for a 200- μ m-thick (10 x expected) hot-melt smear as a function of time after the melt is swept out of the contact zone. At that point, heating stops and conduction to the cool surface begins. The plateau represents the time required to remove the heat of fusion from the meltP-16 |

| Fig. T-1. | The photograph of three sharp, hardened 4140-steel shapes before drilling. | . T-14 |
|-------------------|--|----------------|
| Fig. T-2. | The photograph illustrating the three sharp, hardened 4140-steel shapes during drilling. | . T- 14 |
| Fig. T- 3. | The photograph of the drill bit after drilling on a hardened 4140-steel shape | . T-15 |

| | LIST OF TABLES | Page |
|------------|--|---------------|
| TABLE 2-1 | RELEVANT DOE ORDERS | 2-2 |
| TABLE 2-2 | GAS DATA FOR SSTS ON THE FGWL | 2- 6 |
| TABLE 2-3 | WASTE INFORMATION FOR FGWL SSTs | 2-7 |
| TABLE 2-4 | RMCS COMPONENT WEIGHT BREAKDOWN | 2-27 |
| TABLE 2-5 | EQUIPMENT PROCEDURES USED IN SA | 2 -3 1 |
| TABLE 2-6 | OPERATIONAL PARAMETERS FOR ROTARY-MODE CORE SAMPLING METHOD | 2-37 |
| TABLE 3-1 | SUMMARY OF POTENTIAL HAZARDS | 3-2 |
| TABLE 3-2 | ABOVEGROUND FIRE HAZARD ASSESSMENT RESULTS | 3-7 |
| TABLE 3-3 | GAS RELEASE AND DOME FIRE HAZARD ASSESSMENT RESULTS | 3-8 |
| TABLE 3-4 | DRILL STRING FIRE HAZARD ASSESSMENT RESULTS | .3-10 |
| TABLE 3-5 | WASTE FIRE-HAZARD ASSESSMENT RESULTS | .3-11 |
| TABLE 3-6 | CHEMICAL REACTIONS AND CRITICALITY HAZARD RESULTS | .3-12 |
| TABLE 3-7 | CONTAINMENT BREACH HAZARD ASSESSMENT RESULTS | .3-13 |
| TABLE 3-8 | GAS RELEASE WITHOUT BURN HAZARD ASSESSMENT RESULTS | .3-14 |
| TABLE 3-9 | SPILLS AND RADIATION EXPOSURE HAZARD ASSESSMENT RESULTS | .3-14 |
| TABLE 3-10 | RESULTS OF HAZARD ASSESSMENT FOR EXTERNAL EVENTS | .3-15 |
| TABLE 4-1 | SUMMARY OF SYSTEM RELIABILITY QUANTIFICATION | 4 |
| TABLE 4-2 | SUMMARY OF FIRE ACCIDENTS IN EXHAUSTER | 10 |
| TABLE 4-3 | ABOVE-TANK FIRE ACCIDENTS CAUSED BY TORN EXHAUSTER DUCT | 11 |

August 8, 1996

| TABLE 4-4 | ABOVE-TANK FIRE ACCIDENTS CAUSED BY GAS RELEASES FROM OPEN RISERS |
|-------------------|--|
| TABLE 4-5 | ABOVE-TANK FIRE ACCIDENTS IN SHIELDED RECEIVER |
| TABLE 4-6 | ABOVE-TANK FIRE ACCIDENTS IN X-RAY MACHINE AND CASK |
| TABLE 4-7 | DOME FIRE ACCIDENTS CAUSED BY DRILL STRING BREAKAGE |
| TABLE 4-8 | DOME FIRE ACCIDENTS CAUSED BY EQUIPMENT AND TOOL DROPS INTO RISER |
| TABLE 4-9 | DOME FIRE ACCIDENTS CAUSED BY FRICTIONAL SPARKS IN THE RISER |
| TABLE 4-10 | DOME FIRE ACCIDENTS CAUSED BY DRILL BIT FRICTIONAL SPARKS |
| TABLE 4-11 | DRILL STRING FIRE ACCIDENTS-DRILL STRING IMPACT, GRAPPLE, AND CORE SAMPLER DROP |
| TABLE 4-12 | DRILL STRING FIRE ACCIDENTS-CONDITIONAL FREQUENCIES OF HYDROGEN PENETRATION INTO THE DRILL STRING |
| TABLE 4-13 | WASTE FIRE CAUSED BY FRICTIONAL HEATING FROM THE DRILL BIT |
| TABLE 4-14 | FLAMMABLE-GAS IGNITION UNDER THE WASTE SURFACE CAUSED BY FRICTION FROM THE ROTATION OF THE DRILL BIT |
| TABLE 4-15 | SPILL ACCIDENTS CONTROLS |
| TABLE 5-1A | RADIOLOGICAL RISK GUIDELINES (REF. 1, WHC-CM-4-46, REV. 3) |
| TABLE 5-2 | TOXICOLOGICAL RISK GUIDELINES |
| TABLE 5-3 | ESTABLISHED RISK GUIDELINES FOR SPECIFIC GASES |
| TABLE 5-4 | MAXIMUM ACCEPTABLE WASTE MATERIAL RELEASES ⁵ 5-5 |
| TABLE 5-5 | SUMMARY OF CONSEQUENCE ANALYSIS |
| TABLE 5-6 | BOUNDING RADIOLOGICAL CONSEQUENCES OF DOME COLLAPSE ACCIDENT | 5-9 |
|------------|--|------------|
| TABLE 5-7 | ANTICIPATED AND UNLIKELY GAS-RELEASE EVENTS | 5-10 |
| TABLE 5-8 | SUMMARY OF GAS-RELEASE ACCIDENTS | 5-11 |
| TABLE 5-9 | AMMONIA CONCENTRATIONS AT ON-SITE AND OFF-SITE RECEPTORS FOR GROUND AND STACK RELEASES FOR ALL GAS RELEASE ACCIDENTS | 5-13 |
| TABLE 5-10 | UNFILTERED WASTE RELEASE ACCIDENTS | 5-15 |
| TABLE 5-11 | RADIATION DOSES AT ON-SITE AND OFF-SITE RECEPTORS FOR GRE ACCIDENTS | 5-16 |
| TABLE 5-12 | ONSITE AND OFFSITE TOXIC SOLID WASTE RELEASES | 5-17 |
| TABLE 5-13 | SUMMARY OF SPILL ACCIDENTS | 5-18 |
| TABLE 5-14 | ONSITE AND OFFSITE RADIOLOGICAL DOSES FOR SPILLS | 5-18 |
| TABLE 5-15 | SUMMARY OF ALL ACCIDENTS RESULTING IN RADIOLOGICAL RELEASES | 5-21 |
| TABLE 5-16 | SUMMARY OF GAS-RELEASE ACCIDENTS | 5-23 |
| TABLE 5-17 | SUMMARY OF TOXIC SOLID WASTE RELEASES | 5-25 |
| TABLE 6-1 | FG/RMCS CREDITED DESIGN SAFETY FEATURES | .6-5 |
| TABLE A-1 | LIKELIHOOD CLASSIFICATION TABLE | A-4 |
| TABLE A-2 | CONSEQUENCE SEVERITY TABLE | A-5 |
| TABLE A-3 | RISK-RANKING MATRIX | A-5 |
| TABLE A-4 | DESIGN-BASIS ACCIDENTSA | -13 |
| TABLE A-5 | ROTARY MODE CORE DRILLING HA ACCIDENT DATABASE A | -14 |
| TABLE A-6 | ABOVEGROUND FIRE EVENT TREE DESCRIPTIONS | -28 |
| TABLE A-7 | DOME-FIRE EVENT TREE DESCRIPTIONS | -29 |

| TABLE A-8 | DRILL-STRING FIRE EVENT TREE DESCRIPTIONS |
|------------|--|
| TABLE A-9 | WASTE FIRE EVENT TREE DESCRIPTIONS |
| TABLE A-10 | GAS RELEASE EVENT TREE DESCRIPTIONS |
| TABLE A-11 | TANK LOADING (CONTAINMENT BREACH) EVENT TREE DESCRIPTIONS |
| TABLE A-12 | CONSEQUENCE ASSIGNMENTS FOR HAZARD IDENTIFICATION DATABASE |
| TABLE C-1 | CAPACITY AND CHARACTERISTIC DIMENSIONS OF THE SSTs OF INTEREST |
| TABLE C-2 | DOME VOLUMES AND IMPORTANT DIMENSIONS |
| TABLE C-3 | HYDROGEN, NITROUS OXIDE, AND METHANE FRACTION IN THE NONDENSABLE RELEASE GAS |
| TABLE C-4 | GAS CONCENTRATION DATA REPORTED IN REF. 10 |
| TABLE C-5 | GAS CONCENTRATION DATA REPORTED IN REF. 10 |
| TABLE D-1 | RESULTS OF SYSTEM RELIABILITY QUANTIFICATION |
| TABLE E-1 | SUMMARY OF ACCIDENT QUANTIFICATION RESULTSE-3 |
| TABLE E-2 | SUMMARY OF ACCIDENT QUANTIFICATION RESULTS INCLUDING PHENOMENOLOGICAL FACTORSE-10 |
| TABLE F-1 | LITERATURE VALUES FOR SPECIFIC ENERGY AND UNIT HORSEPOWERF-4 |
| TABLE F-2 | ENVELOPE TEST CONDITIONSF-6 |
| TABLE F-3 | ANALYSIS RESULTS, TESTS 6, 12, AND 16F-18 |
| TABLE F-4 | COMPARISON OF WASTE SIMULANT AND PUMICE TEST MATERIAL |
| TABLE F-5 | TEMPERATURE INCREASE CALCULATIONSF-31 |

| TABLE G-1 | DEGREE OF HAZARD AS A FUNCTION OF ENTHALPY OF DECOMPOSITION OR REACTION |
|------------|---|
| TABLE G-2 | ANOVA MODEL PREDEICTIONS OF THE FRACTION OF WASTE THAT IS REACTIVE |
| TABLE G-3 | FREQUENCY OF A PROPAGATING EXOTHERMIC REACTION AS A RESULT OF ROTARY MODE CORE SAMPLING |
| TABLE K-1 | DISPERSION COEFFICIENTSK-2 |
| TABLE K-2 | ATMOSPHERIC DISPERSION COEFFICIENTS OBTAINED FROM AI-RISK |
| TABLE K-3 | RECEPTOR DOSES FOR A 1-M ³ OR 1-M ³ /S TOXIC-GAS RELEASE |
| TABLE L-1 | QUALITATIVE LIKELIHOOD MATRIX FOR GAS-RELEASE VOLUMES AND RATESL-8 |
| TABLE L-2 | PROBABILISTIC RESULTS FOR DOME SPACE AMMONIA CONCENTRATIONSL-15 |
| TABLE M-1 | GRAMS OF AEROSOL IN THE TANK DOME AND GRAMS TRANSPORTED UP TO STACK FOR 700-SCFM FLOW |
| TABLE M-2 | UNFILTERED AEROSOL RELEASE AS A FUNCTION OF GAS RELEASE |
| TABLE N-1 | RMCS COMPONENT WEIGHT BREAK-DOWNN-2 |
| TABLE N-2 | ROTATIONAL SPEED LIMITS |
| TABLE Q-1 | INDIVIDUAL TANK DATAQ-6 |
| TABLE Q- 2 | SAMPLE PROBLEM RESULTSQ-8 |
| TABLE R-1 | ACCIDENT SOURCE TERM ACTIVITY CONCENTRATIONS |

xxviii

| TABLE R-2 | MATERIALS, DIMENSIONS, AND DENSITY OF THE SAMPLE RECEIVER |
|-----------|--|
| TABLE R-3 | DISPERSION COEFFICIENTSR-8 |
| TABLE R-4 | SITE BOUNDARY DISTANCES FOR THE TANK FARMS ¹⁵ R-9 |
| TABLE R-5 | RECEPTOR DOSES FOR A 1-LITER RELEASE |
| TABLE T-1 | SUMMARY OF ACCIDENT CASES FOR WHICH TEST RESULTS ARE NEEDED |
| TABLE T-2 | GAS MIXTURES |



·.

| ACRONYM | DEFINITION | | | |
|---------|--|--|--|--|
| 1-D | One-dimensional | | | |
| 2-D | Two-dimensional | | | |
| AED | Eerodynamic equivalent diameter | | | |
| AIChE | American Institute of Chemical Engineers | | | |
| ALARA | As low as reasonably achievable | | | |
| АМСА | Air Movement and Control Association | | | |
| ANL | Argonne National Laboratory | | | |
| ANSI | American National Standards Institute | | | |
| ARC | Accelerated rate calorimetry | | | |
| ASA | Accelerated Safety Analysis | | | |
| ASTM | American Society for Testing and Materials | | | |
| BOM | Bureau of Mines | | | |
| САА | Chronic annual average | | | |
| CANDU | Canadian reactor licensing | | | |
| СВ | Containment Breach | | | |
| 20 | Concentrated complexant | | | |
| CEDES | Committed effective dose equivalents | | | |
| CEL | Chemical Engineering Laboratory | | | |
| CGM | Combustible Gas Meter | | | |
| CL | Convective layer | | | |
| DBA | Design Basis Accident | | | |
| DBE | Design basis earthquake | | | |
| DC | Dilute complexed (waste) | | | |
| DCRT | Double-contained receiver tank | | | |
| DDT | Deflagration-to-Detonation | | | |
| DIP | Differential Indicating Probe | | | |
| DOE | Department of Energy | | | |
| DOE-RL | Department of Energy-Richland Area Office | | | |
| DR | Damage ratio | | | |
| DS | Drill string | | | |
| DSC | Differential Scanning Calorimetry | | | |

| ACRONYM | DEFINITION | | | |
|---------|---|--|--|--|
| DSF | Dome space fire | | | |
| DSSF | Double-shell slurry feed | | | |
| DST | Double-shell tank | | | |
| DTA | Differential Thermal Analysis | | | |
| EDES | Effective dose equivalents | | | |
| EDTA | Ethylenediaminetetra-acetic acid | | | |
| ERDA | (US) Energy Research & Development Administration | | | |
| ERP | Emergency Response Planning | | | |
| EXF | External fire | | | |
| FGWL | Flammable Gas Watch List | | | |
| GRE | Gas-release event | | | |
| HA | Hazard analysis | | | |
| HASP | Health and Safety Plan | | | |
| HAZOP | Hazards and operability | | | |
| HBD | Hydraulic bottom detector | | | |
| HEDTA | Hydroxyethyl-ethylenediaminetriacetic acid | | | |
| HEPA | High-efficiency particulate air (filter) | | | |
| HIS | Hazards Identification Study | | | |
| HMS | Hanford Meteorological Station | | | |
| HTWRS | Hanford tank waste remediation system | | | |
| I&C | Instrumentation and control | | | |
| IEEE | Institute of Electrical and Electronic Engineers | | | |
| IH | Industrial Hygience | | | |
| IRRAS | Interim Reliabililty Risk Assessment System | | | |
| ISB | Interim Safety Basis | | | |
| L/D | Length-to-diameter | | | |
| LANL | Los Alamos National Laboratory | | | |
| LASAN | Los Alamos Systems Analysis | | | |
| LCO | Limiting Conditions for Operation | | | |
| LFL | Lower (or lean) flammability limit | | | |
| LOW | Liquid observation well | | | |
| LPF | Leak path factor | | | |
| MAF | Mitigated accident frequency | | | |

August 8, 1996

| ACRONYM | DEFINITION | | | |
|---------|--|--|--|--|
| MAR | Material at risk | | | |
| MEI | Maximum Exposed Individual | | | |
| MIST | Minimum ignition surface temperature | | | |
| MMD | Mass median diameter | | | |
| NCPLX | Noncomplexed waste | | | |
| NEC | National Electric Code | | | |
| NFPA | National Fire Protection Association | | | |
| NRC | Nuclear Regulatory Commission | | | |
| NSSFC | National Severe Storms Forecast Center | | | |
| NUREG | Nuclear Regulatory Commission Regulation | | | |
| OSD | Operational safety document | | | |
| OSHA | Occupational Safety and Health Administration | | | |
| OSR | Operational Safety Requirement | | | |
| PEL-TWA | Permissable exposure limit time-weighted average | | | |
| PG | Purge gas | | | |
| PIC | Person in charge | | | |
| PLC | Programmable Logic Controller | | | |
| РМ | Plume meander | | | |
| PNNL | Pacific Northwest National Laboratory | | | |
| PPF | Pump pit fire | | | |
| PRC | Plant Review Committee | | | |
| PVC | Polyvinyl chloride | | | |
| RCRA | Resource Conservation and Recovery Act | | | |
| RG | Risk Guidelines | | | |
| RCV | Ram control valve | | | |
| RF | Respirable fraction | | | |
| RLU | Remote latch unit | | | |
| RMCS | Rotary-mode core sampling | | | |
| RSST | Reactive Systems Screening Tool | | | |
| SA | Safety assessment | | | |
| SC I | Safety Class I | | | |
| SE | Seismic event | | | |
| SHMS | Standard Hydrogen Monitoring Systems | | | |

August 8, 1996

۰.

| ACRONYM | I NITION | | | | |
|---------|--|--|--|--|--|
| SMC | Sierra Monte Corporte on | | | | |
| SOV | Solenoid-op rated valve | | | | |
| SR | Shielded receiver | | | | |
| SSFGWLT | Single-Shell Flammable Gas Watch List Tank | | | | |
| SMM | Supernate mixing model | | | | |
| SST | Single-shell tank | | | | |
| TC | Thermocouple | | | | |
| TEDE | Total Effective Dose Equivalent | | | | |
| TGR | Toxic gas release | | | | |
| TI | Total Inventory | | | | |
| TLM | Tank layer model | | | | |
| TOC | Total organic compound | | | | |
| TRG | Test Review Group | | | | |
| TSD | Treatment, Storage and Disposal | | | | |
| TWRS | Tank Waste Remediation System | | | | |
| UAF | Unmitigated accident frequency | | | | |
| ULD | Unit-liter dose | | | | |
| UOR | Unusual occurrence report | | | | |
| USQ | Unreviewed safety question | | | | |
| WHC | Westinghouse Hanford Company | | | | |
| ZPA | Zero-Period Acceleration | | | | |

DEFINITIONS

Dust Devil. A dust devil is a localized wind pattern that moves in a circular motion which spawns and decays quickly and travels at relatively low velocities.

Immediate Shutdown. Immediate shutdown is defined as the time it takes for the PLC to send a shutdown signal to the drill engine upon receipt of a valid shutdown signal with no additional programmed-delay. It is understood that the determination of a valid alarm signal requires approximately 2 seconds.

Independent Verification. Independent task verification is defined as requiring that either a second person verify whether a task is performed correctly after a task is completed or whether the original task performer verifies a task correctly performed at a different time and location.

Rotary Drilling. Rotary drilling is defined as rotation of the drill string greater than 2 rpm, while the drill string is in the tank waste.

Waste-intrusive activities. Waste-intrusive activities are defined to include all actions in which motion of, or motion in, the drill string occurs, while the drill string is in the tank waste, including drilling, gas flows and sample insertion and recovery, while the drill string is in the tank waste. Wasteintrusive also includes the four hours following termination of these activities.

. •. .

1.0. SCOPE

This safety assessment (SA) addresses each of the proposed elements required to evaluate the installation, operation, and removal of rotary-mode core sampling (RMCS) equipment in single-shell tanks (SSTs) on the Flammable Gas Watch List (FGWL) or those tanks recommended by the contractor to be included on the FGWL, hence referred to as FG/RMCS operations. These tanks, referred to as flammable-gas tanks (FGTs), are located within the 200 Area in Hanford Site, Richland, Washington. Specifically, this SA addresses the proposed action to install, operate, and remove an FG/RMCS device in the subject tanks.

The objective of this SA is to (1) systematically identify each of the potential hazards associated with these proposed sampling actions, (2) analyze each of the resultant accident sequences, (3) assess the consequences of the accident sequences, and (4) identify the controls and procedures necessary to eliminate or reduce the potential hazards. Section 1 of this SA provides the background information for the proposed actions, discusses the no-action alternative, and outlines the safety assessment approach and scope. Also included in Section 1 is a summary of the significant characteristics of the SST farms.

1.1. BACKGROUND AND PURPOSE OF THE PROPOSED ACTION

The Department of Energy (DOE) is responsible for the management and storage of the waste accumulated from processing defense reactor irradiated fuels for plutonium recovery at the Hanford Site. Currently, there are 177 waste tanks, including 149 SSTs and 28 double-shell tanks (DSTs) located in the 200 area of the Hanford reservation. These wastes, consisting of liquids and solids, are stored in underground storage tanks pending final disposition. A systems approach, managed as part of the tank waste remediation system (TWRS), has been adopted to address the complex and interrelated activities associated with the management and disposal of Hanford tank wastes. The goal of the TWRS is to reduce the environmental, safety, and health risks inherent in the Hanford tank waste operation and remediation. The highest priority for this program is to identify a corrective action strategy for each waste tank safety issue and to mitigate known safety concerns. The four safety issues include (1) flammable-gas generation and concentrations that exceed the lower flammability level (LFL); (2) tanks containing mixtures of ferrocyanide compounds and nitrate/nitrite materials that could, if specific concentrations and conditions were to occur, support an exothermic reaction leading to an explosion; (3) tanks containing organic compounds that could, if locally concentrated, support an exothermic reaction; and (4) Tank 241-106-C, which contains a strontium source generating high heat that requires periodic cooling.

This SA is concerned primarily with SSTs that are on the FGWL or have been recommended by the contractor for inclusion on the FGWL. These tanks are listed and discussed in Section 2.

P

Rotary-mode drilling is necessary only for the SSTs with hard waste layers where waste samples cannot be obtained using the push-mode sampler. The rotary core sampling yields certain hazards that, if not mitigated, result in consequences beyond those analyzed in the push-mode sampling SA.¹ Therefore, an Unreviewed Safety Question (USQ) evaluation was performed. It was concluded that the FG/RMCS operations are not covered by the current authorization and that a separate SA was needed to perform rotary-mode core sampling operations on FGTs. This SA fulfills that need.

1.2. SAFETY ASSESSMENT—SCOPE AND APPROACH

The scope of this SA is to provide a safety basis for the FG/RMCS operations in the single-shell FGTs. To develop an SA that aims at bounding FG/RMCS operations on all FGTs, the following methodology was implemented. The identification of hazards associated with the installation, operation, and removal of the FG/RMCS was performed in a generic tank. Accident sequences also are developed from the evaluated hazards in a generic way. Accident analysis and resulting controls required the discussion of specific parameters pertinent to each SST. A set of bounding tank parameters was not determined through detailed analysis. Instead, a representative tank was chosen that was shown to have bounding tank parameters by performing a preliminary screening process. The screening process considered important tank parameters, such as the retained-gas amount, measured dome flammable- and toxic-gas concentrations, the observed or anticipated gas-release amount, and the waste type. Among the SSTs on the FGWL, Tank A-101 was found to maximize the parameters of interest. The total waste stored in Tank A-101 is 953 Kgal and is mostly consisted of salt cake (950 Kgal). Tank A-101 waste is classified as double-shell slurry feed (DSSF). Estimated radionuclides in Tank A-101 are Cesium (Cs-137), Strontium (S-90), Plutonium (Pu), and Uranium (U).

Accident analyses were performed with this anticipated set of bounding tank parameters. When the first revision of this SA was issued, the anticipated bounding tank parameters were not screened in detail for all tanks of interest. Furthermore, it can be anticipated that additional tanks will be designated as FGTs in the near future after the completion of this SA. Care must be taken in applying this SA to tanks that are on other watch lists (organic, ferrocyanide, etc.). This SA was concerned primarily with the FG issues; hazards specific to tanks on other watch lists may not have been properly addressed in this document. Thus, a screening process with a checklist of items was developed. The controls produced in this SA will require the review and approval of the screening results against the checklist for performance of the FG/RMCS in specific tanks by the Plant Review Committee (PRC). The PRC may charter a separate technical review group to perform the review and approval responsibilities of the PRC. Also, the Westinghouse Hanford Company (WHC) Design Authority is responsible for all aspects of equipment design.

This SA is developed using the guidelines provided in Ref. 2. The approach implemented in this SA incorporates a systematic evaluation of the potential

1-2

hazards related to rotary-mode core sampling in tanks with a flammable environment and the activities required for the installation, operation, and removal of the equipment. For the potential hazards identified, evaluations were completed to establish their potential severity and the resultant consequences of accidents that may occur in response to these hazards. These evaluations consisted of detailed analyses and evaluations using analytical and numerical techniques, routine engineering calculations, and/or a review of existing information to establish the consequences, if any, of these hazards. Finally, this SA identifies the procedures and controls implemented to prevent or mitigate the consequences of these hazards.

Commensurate with this approach, an SA format is developed in Ref. 3. Section 2 of the SA describes the equipment, subsystems, and procedures used during FG/RMCS operations. Section 3 then systematically defines the hazards, causes, and potential accident sequences anticipated, not only with the FG/RMCS operational phase, but also with installation and removal activities. Section 4 assesses the identified hazards in Section 3, followed by a consequence analysis of the postulated accidents in Section 5. Section 6 defines both design features and administrative/ procedural controls that are required to ensure an acceptable level of safety during FG/RMCS operations, especially in a flammable environment. In Section 7, a checklist is provided that contains the items that must be addressed in applying this SA to all FGTs.

This format addresses all activity-related elements listed in US Department of Energy (DOE) Guidance document 3011-94 (Ref. 4). Reference 4 is aimed primarily at safety documents developed for a facility, this SA is aimed at a specific activity, namely the RMCS operations in FGTs. Thus, most of the facility-related sections of the 3011 Guidance are not addressed in this SA. However, a brief summary of the significant characteristics of the SST farms and their environment are provided in Section 1.4 of this SA.

1.3. NO-ACTION ALTERNATIVE

Tank characterization is a high-priority activity at the Hanford site, and FG/RMCS is the proposed retrieval technique for SST samples. A "no-action" alternative would prevent or delay full-depth core sampling activities in SSTs that are FGTs. The analysis of these waste samples is very important for the following reasons:

- Addressing the issues associated with the safe storage of the waste, and
- Developing sound strategies for the retrieval and ultimate disposal of the waste.

Currently, there is no engineered or conceptualized design that can replace the FG/RMCS equipment for obtaining samples from hard waste layers for which the push mode sampling is not adequate.

1.4. SUMMARY OF SIGNIFICANT CHARACTERISTICS OF THE SST FARMS AND THEIR ENVIRONMENT

All the SSTs of interest are located in the 200 Area at the Hanford Reservation, as shown in Fig. 1-1. Detailed descriptions of the SSTs and SST tank farms are provided in Refs. 5 and 6. Specific characteristics of the SSTs pertinent to this SA are discussed in Section 2 and other parts of the SA when needed. This section provides a summary of the descriptive information for the site.

A detailed and comprehensive description of the Hanford Site is presented in documents developed by the (US) Energy Research and Development Administration (ERDA), DOE, and Pacific Northwest Laboratory (PNL).^{7,9} This section summarizes results presented in these references and others as they apply to the 200 Areas. The DOE Hanford Site lies within the semiarid Pasco Basin of the



Fig. 1-1. Location of the Tank Farms within the 200 Area.

Columbia Plateau in southeastern Washington State. The Hanford Site occupies an area ~1450 km² (570 mi²) north of the confluence of the Snake, Yakima, and Columbia Rivers. This land, with restricted public access, provides a buffer for the smaller areas currently used for the production of nuclear materials, waste storage, and waste disposal; only ~6% of the land area has been disturbed and is actively used. The Columbia River flows through the northern part of the Hanford Site; turning south, it forms part of the site's eastern boundary (Fig. 1-2).¹⁰The terrain of the central and eastern parts of the Hanford Site is relatively flat, with evidence in the central part of the Site (including the 200-Area Plateau) of minimal erosion since the deposition of Hanford Formation sediments by glacial floodwaters ~13,000 yr ago. The soil beneath the tank farm consists of silt, sand, and gravel. The principal geologic units beneath much of the 200-West Area are, in ascending order: (1) the Columbia River Basalt Group, with interbedded sediments of the Ellensburg Formation; (2) the Ringold Formation; (3) the Plio-Pleistocene unit; and (4) the Hanford Formation. The Ringold Formation is ~47.2 m (155 ft) below the surface of the SY Tank Farm.¹¹



Fig. 1-2. Location of 200 Area within the Hanford Site.

August 8, 1996

Two areas of shallow, swarm seismic activity, Coyote Rapids and Cold Creek, are located within 16.1 km (10 mi) of the 200-West Area. The Coyote Rapids swarm area has been the site of 8 swarms consisting of 91 shallow seismic events during the period between 1969 and 1986. The depth distribution of these seismic events is bimodal, with maximum activity occurring near the surface and at a depth of 4.0 to 6.9 km (2.5 to 4.3 mi). The Cold Creek swarm area, located 12.9 km (8 mi) south of the 200-West Area, includes 32 events from 1979 to 1986 that occurred at depths up to 4.8 km (3 mi).

Several surface ponds and ditches associated with fuel and waste processing activities are present within the 200 Area (Fig. 1-3).¹² These ponds and ditches are used primarily as wasteways for process and cooling water and sometimes contain small quantities of radionuclides (both fission products and transuranic elements). Cold Creek and its tributary, Dry Creek, are ephemeral streams within the Yakima River drainage system along the southern boundary of the Hanford Site. Both streams drain areas to the west of the Hanford Site and cross the southwestern part of the Site toward the Yakima River. The potential for flash flooding from the Cold Creek drainage system has been examined, and a maximum flood depth of 2.1 m (7 ft) was estimated along the southwestern part of the 200-Area plateau. However, the maximum probable flood has not been well defined for the Cold Creek drainage system. A 100-yr peak stage flood, estimated to be ~0.9 m (3 ft) above the Cold Creek Valley floor, would not reach the 200-West Area.¹³

Wastewater ponds on the Hanford Site have recharged the unconfined aquifer below the 200-Area artificially. The increase in water-table elevations was most pronounced from 1950 to 1960 and had approached equilibrium between the unconfined aquifer and the recharge between 1970 and 1980, when only small increases in water-table elevations occurred. Wastewater discharges from the 200 Area were reduced significantly in 1984 (Ref. 14), with an accompanying decline in water-table elevations. The depth to groundwater currently is ~50 to 60 m (164 to 197 ft) in the 200 Area. Groundwater flow direction is generally in an easterly and southeasterly direction, toward the Columbia River.

Lateral groundwater movement occurs within a shallow, unconfined aquifer consisting of fluvial and lacustrine sediments lying on top of the basalts and within deeper confined-to-semiconfined aquifers consisting of basalt flow tops, flow bottom cones, and sedimentary interbeds.¹⁵ Sources of natural recharge to the unconfined aquifer are rainfall and runoff from the higher bordering elevations, water infiltrating from small ephemeral streams, and river water along influent reaches of the Yakima and Columbia Rivers. Artificial recharge to the unconfined aquifer results from the disposal of wastewater to the ground below the 200 Areas from the surrounding highlands. This recharge to the aquifer [5.5E04 m³/d (1.5E07 gal./d)] is ~10 times the natural recharge entering the unconfined aquifer below the 200 Areas.¹⁵ Beneath the disposal ponds, groundwater mounds have developed in response to the artificial recharge. Beneath U Pond, located in the 200-West Area, the water table rose ~24.4 m (80 ft) from the start of disposal operations in 1944

(Refs. 16 and 17) until U Pond was decommissioned in 1985. From the recharge areas to the west, the groundwater flows down the gradient to the discharge areas along the Columbia River, interrupted locally by the groundwater mounds in the 200 Areas. The horizontal and vertical extent of these mounds appears to be related directly to the surface discharge of wastewater from facilities in this area.¹⁸



Fig. 1-3. Locations of surface-water ponds, ditches, and ephemeral streams on the Hanford Site.

Climatological data are available from the Hanford Meteorological Station (HMS), which is located between the 200-East and 200-West Areas. Data have been collected at this location since 1945. Temperature and precipitation data also are available from nearby locations for the period of 1912 through 1943. A summary of these data through 1980 has been published in Ref. 19. Data from the HMS are representative of the general climatic conditions for the region and describe the specific climate of the 200-Area Plateau.

The prevailing winds on the 200-Area Plateau are from the northwest. Secondary maxima occur for southwesterly winds. Diurnal and monthly averages and extremes of temperature, dew point, and humidity are contained in Stone et al.¹⁴

Ranges of daily maximum temperatures vary from a normal 2° C (36° F) in early Janu v to 35° C (95° F) in late July. The record maximum temperature is 46° C (115° and the record minimum temperature is -32.8° C (-27.0° F). Relative bunically/dew-point temperature measurements are made at the HMS and at the ree 61.0-m (200-ft) monitoring tower locations. The annual average relative humidity at the HMS is 45%. It is highest during the winter months (averaging ~75%) and lowest during the summer (averaging ~35%). At the Hanford Site, the severe-weather phenomenon that occurs most frequently and has the greatest effect

the dust storm.¹⁹ The maximum recorded peak gust at 15 m (50 ft) aboveground was 128 km/h (80 mi/h), which occurred in January 1972. A 100-yr return period peak gust of 138 km/h (86 mi/h) has been calculated at the 15-m (50-ft) elevation.

Precipitation measurements have been made at the HMS since 1945. Average annual precipitation at the HMS is 16 cm (6.3 in.). Most of the precipitation occurs during the winter, with nearly half of the annual amount occurring in the months of November through February. Rainfall intensities of 1.3 cm/h (0.5 in./h) and persisting for 1 h are expected once every 10 yr. Rainfall intensities of 2.5 cm/h (1.0 in./h) for 1 h are expected only once every 500 yr. The Hanford Site is not a major thunderstorm area. On average, only about 10 thunderstorm days per year are recorded at the Hanford Site, although this number has varied from a low of 3 to a high of 23 thunderstorm days per year. Thunderstorms theoretically can occur during any month of the year; however, they occur most frequently from April through September. The largest number of thunderstorm days recorded in a single month is eight, which has occurred in both June and August. Large differences in electric potential can occur during thunderstorms, which, in turn, can lead to lightning strikes. In general, ~20% of lightning strikes are cloud-toground/ground-to-cloud discharges. Lightning strikes in the summer have occasionally ignited range fires in the Hanford Site region. Estimates of the extreme thunderstorm winds, based on peak gusts observed from 1945 through 1980, are given in Ref. 19. Using the National Weather Service criteria for classifying a thunderstorm as "severe" [i.e., hail with a diameter ≥ 20 mm (0.8 in.) or wind gusts \geq 93 km/h (84.8 ft/s)], only 1.9% of all thunderstorm events observed at the HMS have been "severe" storms; all met the criteria based on wind gusts.

The nearest volcano is in the Cascade Range, more than 100 km (62 mi) from the Hanford Site, and most eruption products are deposited within 50 km (31 mi) of their source. There is no evidence that volcanic lava flows, debris flows, or mudflows from the Cascade Range volcanoes reached the Pasco Basin during the Quaternary period.

Flows of lava, debris, and mud tend to be confined to existing drainage channels, and because no streams flow directly from the Cascade Range to the Hanford Site, these types of volcanic deposits are not considered likely at the 200 Area.

Tornadoes are infrequent and generally small in the northwest portion of the United States. The HMS climatological summary and the National Severe Storms

Forecast Center (NSSFC) database list 22 separate tornado occurrences within 161 km (100 mi) of the Hanford Site from 1916 through August 1982. Two additional tornadoes have been reported since August 1982.

1.5. **REFERENCES**

- N. J. Milliken and G. R. Geschke "Safety Analysis for Push-Mode and Rotary-Mode Core Sampling," Westinghouse Hanford Company report, WHC-SD-WM-SARR-031, Rev. 2 (July 1995).
- 2. Westinghouse Hanford Company, "Safety Analysis Manual," Westinghouse Hanford Company report WHC-CM-4-46 Rev. 4 (September 30, 1995).
- 3. Westinghouse Hanford Company, "Interim Guidance for Preparing Safety Assessments," in Safety Analysis and Engineering Manual: Work Procedures, Westinghouse Hanford Company report WHC-CM-6-32, Rev. 1 (November 1992).
- 4. US Department of Energy, "Guidance for Preparation of DOE 5480.22 (TSR) and DOE 5480.23 (SAR) Implementation Plans," DOE Standard 3011-94 (November 1994).
- 5. H. R. Brager, "Summary of Information on Flammable Gas Watch List Tanks," Westinghouse Hanford Company report WHC-EP-0711 (January 1994).
- 6. Westinghouse Hanford Company, "Hanford Site Tank Farm Facilities Interim Safety Basis," WHC-SD-WM-ISB-001, Rev. 0, Vol. 1 (August 1993).
- 7. US Energy Research and Development Administration, "Final Environmental Impact Statement of Waste Management Operations, Hanford Reservation, Richland, Washington," US Energy Research and Development Administration report ERDA-1538, 2 vols. (1975).
- 8. US Department of Energy, "Draft Environmental Assessment, Reference Repository Location, Hanford Site, Washington," US Department of Energy report DOE/RW-0017 (1984).
- 9. Pacific Northwest Laboratory, "Hanford Site National Environmental Policy Act (NEPA) Characterization," Pacific Northwest Laboratory report PNL-6451 (1990).
- 10. M. Gerber, "Legend and Legacy: Fifty Years of Defense Production at the Hanford Site," Westinghouse Hanford Company report WHC-MR-0293, Rev. 1 (1992).
- 11. "Investigation to Determine Dynamic Soil Properties at the 241-SY-Tank Site," URS/John A. Blume & Associates, Atlantic Richfield Hanford Company (1974).

August 8, 1996

1-10

- 12. "Environmental Impact Statement, Disposal of Hanford High-Level and Transuranic and Tank Wastes, Hanford Site, Richland, Washington," US Department of Energy report DOE/EIS-0113, Vol. 1-III (1987).
- 13. R. L. Skaggs and W. H. Walters, "Flood Risk Analysis of Cold Creek near the Hanford Site," Pacific Northwest Laboratory for Rockwell Hanford Operations report RHO-BWI-C-120 (PNL-4219) (1981).
- 14. W. A. Stone, J. M. Thorp, O. P. Gifford, and D. J. Hoitink, "Climatological Summary for the Hanford Area," Pacific Northwest Laboratory report PNL-4622 (1983).
- 15. US Department of Energy, "Consultation Draft: Site Characterization Plan, Reference Repository Location, Hanford Site, Washington," US Department of Energy report DOE/RW-0164, Vol. 2 of 9 (1988).
- 16. C. C. Meinhardt and J. C. Frostenson, "Current Status of 200 Area Ponds," Rockwell Hanford Operations report RHO-CK-798 (1979).
- 17. J. D. Jamison, "Standardized Input for Hanford Environmental Impact Statements Part II: Site Descriptions," Pacific Northwest Laboratory report PNL-3509, Part 2 (1982).
- 18. D. A. Zimmerman, A. E. Reisenauer, G. D. Black, and M. A. Young, "Hanford Site Water Table Changes 1950 through 1980—Data Observations and Evaluation," Pacific Northwest Laboratory report PNL-5506 (1986).
- 19. D. I. Herborn et al., "Hanford Waste Vitrification Plant Preliminary Safety Analysis Report," Westinghouse Hanford Company report WHC-EP-0250 (1991).

the age of the

2.0 DESCRIPTION OF ACTION

The section presents the detailed descriptions required to evaluate the installation, operation, and removal of rotary-mode core sampling (RMCS) equipment in singlesection tanks (SSTs) on the Flammable Gas Watch List (FGWL) or those tanks recommended by the contractor to be included on the FGWL, hence referred to as FG/RMCS operations. The safety of RMCS operations in flammable gas tanks has been questioned because of the potential to induce a spark within the flammable environment of the tanks. The descriptions reflect an understanding of the FG/RMCS equipment and processes at the time of the safety assessment (SA) and are provided for the information of the reader only.

This section details the safety criteria surrounding tanks on the FGWL and tanks recommended by the contractor for inclusion on the FGWL, collectively to referred to as FGTs, the gas and ignition phenomenology anticipated during sampling operations in these tanks, and descriptions of the tanks and their characteristics. Descriptions of the equipment and systems required for rotary-mode core sampling follow, along with a summary of the drilling operations under normal, and certain abnormal conditions.

Because tank characterization and sampling are of the highest priority at Hanford as stated in Public Law 101-510, Section 3137, "Safety Measures for Waste Tanks at Hanford Nuclear Reservation," (1990), a suite of sampling methods have been incorporated into characterization and stabilization strategies, including grab sampling, vapor sampling, auger operations, and push-mode sampling. However, rotary-mode core sampling is used for obtaining full-depth core samples in tanks with salt cake waste.

Rotary-mode sampling operations and procedures are similar to push-mode sampling techniques with several differences: the drill bit differs, a nitrogen purge system is activated to cool the drill bit as it rotates, and an exhauster is used to compensate for the nitrogen purge flow and aerosol introduced into the tank dome. In general, a sampling truck capable of rotary-mode sampling, a nitrogen supply system, an exhauster, and a variety of support equipment is set up on or near the tank. The sampling truck is located at the appropriate riser, and the drill string with a universal sampler is inserted into the tank. With the nitrogen purge and exhauster systems activated, the rotary drilling collects a cylindrical waste sample that is withdrawn from the tank, transferred in a shielded receiver to a mobile X-ray system for preliminary examination, then transferred into a cask for transport to the analytical laboratories for full characterization. The drilling/sampling sequence is repeated until a set of samples representing a full-depth core is acquired.

Specific design features and assumptions, provided in Section 6, shall be used to assess the extent to which changes or modifications alter the functions or operational characteristics of the FG/RMCS processes, systems, or components. The Department of Energy (DOE) Order 5480.21, Unreviewed Safety Questions (USQs) process shall be used to assess which SA results could be altered or negated by said changes or modifications, and to what extent SA revisions could be required.

2.1. PRINCIPAL SAFETY CRITERIA

Safety criteria for this section include a consideration of DOE Orders, existing Westinghouse Hanford Company (WHC) documents and procedures, and the more specific criteria associated with FGWL tanks. Because several of the tanks on the FGWL are also on the Organic Watch List, the organic criteria are also provided.

2.1.1. DOE Safety and Design Requirements

The DOE Orders cited in TABLE 2-1, are presently applicable to the design of the rotary-mode core sampling equipment. They are helpful in developing the criteria outlined in Sections 2.1.2, 2.1.3, 2.1.4 and 2.2. The risk criteria are given in Section 5 of this SA.

| DOE Order | Title |
|-------------------|--|
| DOE Order 1540.2 | "Hazardous Material Packaging" |
| DOE Order 4330.4A | "Maintenance Management Program" |
| DOE Order 5000.3B | "Occurrence Reporting and Processing of Operations Information" |
| DOE Order 5400.1 | "General Environmental Protection Program" |
| DOE Order 5400.5 | "Radiation Protection of the Public and the Environment" |
| DOE Order 5480.4 | "Environmental Protection, Safety, and Health Protection Standards" |
| DOE Order 5480.5 | "Safety of Nuclear Facilities" |
| DOE Order 5480.7 | "Fire Protection" |
| DOE Order 5480.10 | "Contractor Industrial Hygiene Program" |
| DOE Order 5480.11 | "Radiation Protection for Occupational Workers" |
| DOE Order 5480.19 | "Conduct of Operations Requirements for DOE Facilities" |
| DOE Order 5480.20 | "Personnel Selection, Qualification, Training, and Staffing Requirements at DOE Reactor and Non-Reactor Nuclear Facilities" |
| DOE Order 5480.21 | "Unreviewed Safety Questions" |
| DOE Order 5480.22 | "Technical Safety Requirements" |
| DOE Order 5480.23 | "Nuclear Safety Analysis Report" |
| DOE Order 5480.31 | "Start-up and Re-start of Nuclear Facilities" |
| DOE Order 5483.1A | "Occupational Safety and Health Program for DOE Contractor Employees at Government Owned Contractor Operated Facilities" |
| DOE Order 5500.2B | "Emergency Categories, Classes and Notification and Reporting Requirements" |
| DOE Order 5500.3A | "Planning and Preparedness for Operational Emergencies" |
| DOE Order 5700.6C | "Quality Assurance" |
| DOE Order 5820.2A | "Radioactive Waste Management" |
| DOE Order 6430.1A | "General Design Criteria" |

TABLE 2-1 RELEVANT DOE ORDERS

2.1.2. WHC General Safety Criteria

WHC standard controls include a series of WHC documents that define the safety envelope for the tank farm. The primary documents include the following:

- "Tank Farm Health and Safety Plan,"¹
- "Hanford Site Tank Farm Facilities Interim Safety Basis."²

Other pertinent WHC documents are referenced as necessary throughout the SA.

2.1.3. Safety Criteria for Tanks on the FGWL

Initially, Hanford tanks were identified for inclusion on the FGWL by qualitatively considering the following:³

- The presence of specific types of waste in the tank,
- The presence of a high radiation field in the tank,
- Observation of certain gaseous components in the dome space,
- Observation of pattern changes in the surface level,
- Observation of periodic pressure surges in the dome space, and
- Observations of the axial temperature profiles in the waste.

There are currently 19 SSTs on the FGWL with flammable gas concentrations that can exceed 25% LFL in the dome of a full tank with a decay heat generation of 30,000 Btu/h. They were placed on the watch list more for the *potential* of containing flammable gas rather than the verified presence of hazardous concentrations.

Further details of these criteria are provided in Ref. 3. Currently, there are 25 tanks on the FGWL; 19 of these tanks are single-shell tanks. These tanks are:

| A-101 | S-102 | SX-101 | U-103 |
|--------|-------|--------|--------------|
| | S-111 | SX-102 | U-105 |
| AX-101 | S-112 | SX-103 | U-107 |
| AX-103 | | SX-104 | U-108 |
| | T-110 | SX-105 | U-109 |
| | | SX-106 | |
| | | SX-109 | |

The hydrogen SSTs were placed on the watch list mainly because waste level increases were observed without liquid addition. That is, slurry growth acted as the main criterion for watch list designation. There was concern that the growth of slurry could indicate a situation similar to that experienced with Tank 101-SY, even though the available data indicate total growth values as opposed to episodic-type behavior.

Alternate criteria for tanks included having a surface crust, having a total organic compound (TOC) level >3g/L, or containing B-Plant waste. The B-Plant wastes are organic-bearing wastes generated from the B-Plant fractionation process, primarily during strontium recovery. The compounds making up the organic fractions are complexing/chelating agents or their degradation products.

Recently, a new methodology was developed to identify tanks that may be candidates for inclusion on the FGWL. The new method involves evaluating the waste surface level changes in response to changes in barometric pressure.⁴ By applying this criterion, a number of additional tanks were identified as candidates for potential storage of flammable gases. These tanks are

| A-103 | BY-101 | C-104 | S-101 | TX-102 | U-102 |
|--------------|--------|---------------|-------|--------|--------------|
| | BY-102 | C-107 | S-103 | TX-111 | U-106 |
| B-111 | BY-105 | | S-104 | TX-112 | U-110 |
| B-201 | BY-106 | T-2 01 | S-105 | TX-113 | U-111 |
| B-202 | BY-109 | T-102 | S-106 | TX-115 | |
| | | T-203 | S-107 | TX-116 | |
| BX-107 | | T-204 | S-109 | TX-117 | |
| | | | | | |

Hereafter, these 34 tanks along with the original 19 FGWL single-shell tanks are referred to as the flammable-gas tanks (FGTs). Presently, the total number of single-shell FGTs is 53.

2.1.4. Safety Criteria for Tanks on the Organic Watch List

Levels of safety for tanks on the Organic Watch List are addressed in Ref. 5. The safety criteria are based on a set of tests in which dry sodium acetate nitrate/nitrite mixtures exhibited propagating behavior at about 300°C (572°F) with a TOC value greater than 6 wt%. Appendix G evaluates this criteria in detail for each SST.

2.2. PHENOMENOLOGY

Gas phenomena include considerations of gas storage and release mechanisms, gas composition, waste characteristics, and flammability and ignition.

2.2.1. Gas Storage and Release Mechanisms in SSTs

The model and data available that describe gas storage and release mechanisms are discussed in Appendix L. The conclusion is that large and prompt releases are not likely in single shell tanks.

The 19 single-shell tanks on the FGWL were separated by Los Alamos Nation: Laboratory (LANL) in 1994⁶ into the following four gas-release categories:

- 1. Tanks that do not experience episodic behavior nor exhibit long-term growth in the waste level,
- 2. Tanks for which not enough data are available to evaluate the behavior,
- 3. Tanks that potentially exhibit episodic gas-release behavior*, and
- 4. Tanks that exhibit long-term waste growth but do not exhibit episodic gasrelease behavior.

*There is only one SST in this category (A-101), and there is no data to suggest that SST exhibit episodic release. However, episodic behavior occurrences are addressed in Appendix L.

TABLE 2-2 provides selected data pertaining to the SSTs on the FGWL, including the designated gas-release category. A similar analysis for all the FGTs currently is not available. TABLE 2-2 also notes which tanks are on the Organic Watch List.

2.2.2. Gas Composition

The gas-concentration measurements in the SST dome space are very scarce compared to some of the double-shell tanks (DSTs) (e.g., Tank 101-SY). The available data obtained from the vapor grab samples are analyzed in Appendix C of this SA. As shown in Appendix C, hydrogen, nitrous oxide, ammonia and methane are detected at varying concentrations in the dome space of the FGTs.

| | | Organic | Vapor Space | Gas-release |
|--------|------|------------|-------------|-------------|
| Tank | FGWL | Watch List | Volume, m³ | Category |
| A-101 | ٠ | • | 1,454 | 3 |
| AX-101 | • | | 2,481 | 2 |
| AX-103 | • | | 4,892 | 1 🔪 |
| S-102 | • | • | 1,955 | 4 |
| S-111 | • | • | 916 | 1 |
| S-112 | • | | 760 | 1 |
| SX-101 | ٠ | | 3,283 | 2 |
| SX-102 | • | | 2,953 | . 2 |
| SX-103 | • | • | 2,540 | 1 |
| SX-104 | • | | 2,593 | 2 |
| SX-105 | • | | 2,422 | 2 |
| SX-106 | • | • | 2,972 | 1 |
| SX-109 | • | | 4,064 | 1 |
| T-110 | • | | 1,738 | 4 |
| U-103 | • | • | 1,433 | 4 |
| U-105 | • | • | 1,590 | 4 |
| U-107 | • | • | 1,636 | 4 |
| U-108 | • | | 1,401 | 4 |
| U-109 | • | | 1,420 | 4 |

TABLE 2-2 GAS DATA FOR SSTS ON THE FGWL

2.2.3. Waste Characteristics

Sludge and salt cake are generally the two forms of waste in SSTs of concern, although supernatant liquid also exists in some tanks. Sludge results from the precipitates formed during the neutralization of chemical separation wastes and is composed principally of hydrous metal oxides. Salt cake results from actual dewatering by pumping and from thermal evaporation of aged chemical and miscellaneous wastes. For SSTs that contain both types of solids, the salt cake layer is typically on top of the sludge layer. Liquid is present in SSTs as supernatant and/or as interstitial liquid existing in the void spaces of the solid wastes. Data on waste type and volume⁷ are provided in TABLE 2-3 for information purposes only.

The waste types found in the SSTs on the FGWL are of four types: concentrated complexant; dilute complexed (DC) waste; double-shell slurry feed; and noncomplexed waste. Concentrated complexant (CC) is a concentrated product from the evaporation of dilute complexed waste. DC waste is characterized by a high content of organic carbon, including organic complexants. Ethylenediaminetetra-

acetic acid (EDTA), citric acid, and hydroxyethyl-ethylenediaminetriacetic acid (HEDTA) are the major complexants used. The main sources of DC waste are the salt well liquid inventory from the SSTs. Double-shell slurry feed (DSSF) is waste concentrated just before reaching the sodium aluminate saturation boundary (of 6.5 molar hydroxide) in the evaporator without exceeding teceiver tank composition limits. Noncomplexed waste (NCPLX) is a general waste term applied to all Hanford Site liquors not identified as complexed.

| Tank | Waste Type | Total Waste Vol, Kgal | Super- natant Vol, Kgal | Sludge Vol, Kgal | Salt cake Vol, Kgal | Waste Temp, °F | Waste Depth, In. |
|----------------|---------------|--------------------------------|----------------------------------|------------------------|---------------------------|----------------------|------------------------|
| A-101 | DSSF | 953 | 0 | 3 | 950 | 154 | 354 |
| AX -101 | DSSF | 748 | 0 | 3 | 745 | 136 | 279 |
| AX -103 | CC | 112 | 0 | 2 | 110 | 111 | 48 |
| S-102 | DSSF | 549 | 0 | 4 | 545 | 107 | 207 |
| S-111 | NCPLX | 596 | 10 | 139 | 447 | 92 | 224 |
| S-112 | NCPLX | 523 | 0 | 5 | 518 | 8 3 | 239 |
| SX-101 | DC | 456 | 1 | 112 | 343 | 138 | 173 |
| SX-102 | DSSF | 543 | 0 | 117 | 426 | 151 | 206 |
| SX-103 | NCPLX | 652 | 1 | 115 | 536 | 174 | 245 |
| SX-104 | DSSF | 614 | 0 | 136 | 478 | 167 | 231 |
| SX-105 | DSSF | 683 | 0 | 73 | 610 | 180 | 256 |
| SX-106 | NCPLX | 538 | 61 | 12 | 465 | 111 | 203 |
| SX-109 | NCPLX | 250 | 0 | 0 | 25 0 | 148 | 9 8 |
| T-110 | NCPLX | 379 | 3 | 376 | 376 | 63 | 145 |
| U-103 | NCPLX | 468 | 13 | 32 | 423 | 87 | 178 |
| U-105 | NCPLX | 418 | 37 | 32 | 349 | 89 | 159 |
| U-107 | DSSF | 406 | 31 | 5 | 360 | 78 | 155 |
| U-108 | NCPLX | 468 | 24 | 29 | 415 | 8 8 | 178 |
| U-109 | NCPLX | 463 | 19 | 48 | 396 | 86 | 176 |

TABLE 2-3 WASTE INFORMATION FOR FGWL SSTs

2.2.4. Flammability and Ignition

I ammability issues are highlighted by two aspects in tanks with a flammable environment, the broad flammability range of hydrogen in air (4% to 75%) and the low energy required for ignition (0.01 mJ). The ignition hazard is increased because nitrous oxide is a strong oxidant and is cogenerated with the hydrogen in amounts that place the gas mixture well within the flammability range before mixing with air. Burning hydrogen releases a relatively large amount of energy. The heat of combustion is 57.8 and 77.3 kcal/g-mol of hydrogen for oxygen and nitrous oxide reactions, respectively. Ammonia has a flammability range in air of 15% to 30%, and the heat of combustion with oxygen is 75.8 kcal/g-mol NH_3 (25°C).

Secondary exothermic reactions in the waste surface crust can also be induced by a hydrogen burn. The surface crust usually contains an oxidant, such as a mixture of sodium nitrate and sodium nitrite, in which various amounts of organic carbon are well mixed.

The presence of flammable gases and the release of chemical energy are compounded by the presence of radioactive waste, thereby increasing the potential consequences of a release. Toxic gases, especially ammonia, are known to be associated with the waste and may have the greatest potential for release during an episodic event.

2.3. FLAMMABLE-GAS TANKS

The 53 FGTs are spread over 11 of the 12 single-shell tank farms: A, AX, B, BX, BY, C, S, SX, T, TX and U. The cylindrical, dome-roofed tanks⁸ are constructed of reinforced concrete with a structurally independent mild carbon-steel liner covering the bottom and sidewalls. Each tank is buried with a minimum of 6.5 ft of earth for shielding and heat dissipation from radioactive decay. The tanks were designed to hold approximately 15 to 30 ft of liquid, with a nominal capacity of 530,000, 758,0000, and 1,0000,000 gallons. Of the three possible tank configurations, the 1,000,000-gal.-capacity tank used in Farms A, AX, and SX is schematically shown in Fig. 2-1. The other tank configurations are similar to those given in Fig. 2-1. The BY-, TX- and S-Tank Farm has tanks with 758,000-gal. capacity, but tanks in Farms B, BX, C, T and U have a capacity of 530,000 gallons.

A typical single-shell tank has numerous vertical pipes called risers that penetrate the tank dome and extend to various depths of the tank. The dome risers, which vary in diameter from 4, 12, or 42 in., provide access to the tank interior for a variety of operating and monitoring equipment, such as the breather inlet, a camera observation point, the center pump pit, a dome elevation bench mark, a solids level detector, a liquid observation well, a surface level probe, the temperature thermocouple assembly, and a leak detection drywell.

Most SSTs use a passive form of ventilation² that allows airflow through the tanks to be dictated by atmospheric conditions such as temperature and atmospheric pressure. The system, called the breather inlet, minimizes the pressure changes that could damage the tank structure if the tanks were completely sealed. Each breather filter is mounted on a tank riser, and consists of a housing containing a HEPA (high efficiency particulate air) filter, an outlet screen, and a small seal loop that acts as a pressure relief should the filter become plugged. An isolation valve, which is normally open, allows flow between the tank vapor space and the environment through the filter. The flow moves horizontally through the 12 in. x 12 in. filter, and then vertically through the downward-facing exit weather hood. During FC = MCS operations, the breather inlet will be fitted with a portable, sealable, 15-ft to up diameter stack to control the direction of gases exiting the tank

ta. -in.-diameter stack to control the direction of gases exiting the tank.



Fig. 2-1. Typical A-Farm SST cross section.

2.4. RMCS SYSTEM AND ASSOCIATED EQUIPMENT

The system required for retrieving waste samples in the rotary-mode consists of the sampling truck, the portable exhauster, the N_2 supply system, the generator, and associated equipment. Primary components associated with the sampling truck include the grapple hoist assembly, the shielded receiver (SR) with remote latch unit (RLU), the drill string (DS), the nitrogen purge system, and the change-out assembly.^{9, 11, 12} Associated equipment includes the X-ray machine, cask stand and truck, power distribution trailer, and support vehicles.

Functional criteria for the FG/RMCS are found in Ref. 13. Several critical elements of the former document include the following:

- Equipment with energized circuits that can come in contact with waste degradation gases before dilution by the tank vapor space or other gases shall be protected in accordance with National Fire Protection Association (NFPA) for use in Class I, Division 1, Group B for a flammable hydrogen atmosphere. Protection may be in the form of intrinsically safe electrical components, purging in accordance with NFPA Article 496, or other acceptable method as defined by the requirements of the National Electric Code (NEC) Article 501, for use in flammable hydrogen atmospheres.
- Equipment where energized circuits have the potential, under abnormal conditions, to come in contact with waste degradation gases before dilution by the tank vapor space or other gases shall be protected in accordance with NFPA for use in Class I, Division 2, Group B for a flammable hydrogen atmosphere. Protection may be in the form of intrinsically safe electrical components, purging in accordance with NFPA Article 496, or other acceptable method as defined by the requirements of NEC Article 501, for use in flammable hydrogen atmospheres.

2.4.1. RMCS Trucks

Three RMCS trucks provide mobility to position and move the core sampling equipment from tank to tank. Fig. 2-2, derived from Ref. 15 shows the general arrangement of equipment on the rotary platform that is mounted on the rear of each of the three sampling trucks. The rotating platform supports and positions core sampling equipment, including the platform hoist, the grapple hoist assembly, the drill rig and drill string, and the shielded receiver over the tank riser to be sampled. Five stationary hydraulic jacks act as outriggers to level the truck for drilling operations.



Fig. 2-2. Sampling truck configuration.

Two of the three trucks have diesel engines, while the third truck has a gasoline engine. All drill engines are gasoline fueled.

2.4.1.1. Platform Hoist

The electric platform hoist is located on the rotating platform between the grapple hoist assembly and the shielded receiver assembly. With a capacity of 500 lb, it provides an on-site method to handle riser adapter equipment, insert and remove drill rods, and position the cask stand.

2.4.1.2. Grapple Hoist Assembly

The grapple hoist assembly, shown in Fig. 2-3 consists of an electric motor-driven hoist contained in a pressurized box, the electric motor (external to the pressure vessel), and a grapple connected to the hoist cable. The grapple hoist assembly controls the sampler piston movement.

Grapple Hoist and Box. The grapple hoist box, Ref. 16, houses the grapple cable, cable reel, and a load cell. The grapple box and the pneumatic piping connecting it to the purge gas enclosure provide containment of drill purge gases.

The hoist is used to lower the grapple into the drill string after a sampler has been installed. The 3/4 hp grapple motor lowers the cable at a maximum speed of 58 ft/min, and employs a cable with a breaking strength of 2400 lb. A load cell attached to the cable tension assembly is designed to shut off the motor if the load equals or exceeds 250 lb. Roll pins on the hoist shaft are designed to shear before the hoist motor can exceed the structural capability of the hoist shaft.





Inner and Outer Bellows Assembly. The inner and outer bellows assembly,¹⁷ also shown in Fig. 2-3, provides a collapsible pressure boundary between the grapple box and the quill rod for containment of purge gas.

Grapple. The grapple (sample actuator), Ref. 18, is a spring-loaded device, schematically represented in Fig. 2-4 that is lowered to connect with the pintle rod of the sampler and holds the sampler piston in position while a sample is being taken.



Fig. 2-4. Grapple Schematic.

2.4.1.3. Drill Unit and Quill Rod

The drill unit is composed of the drill rig, the drill head, and a hydraulic chuck that clamps to the quill rod.

Drill Rig. The drill rig engine provides energy through the drill head for drill rotation and through the hydraulic rams for down force to the quill rod, which in turn transfers these motions to the drill string. The drill head raises and lowers the quill rod, and a hydraulically operated chuck clamps the quill rod in place. Manual hydraulic controls are provided for quill rod and shielded receiver positioning.

Quill Rod. The quill rod, Ref. 19, which transfers power to the drill string, is the topmost rod of the drill string. This unit remains in the drill head and transmits power through the hydraulic chuck.

Quill Rod Adapter. The stainless-steel quill rod adapter, Ref. 19, is attached at the bottom of the quill rod and is a 6-in. section with the same diameter as the quill rod. The adapter has a quick connection feature that allows for the addition of water to wash the drill bit. The adapter connection feature is also used for flammable-gas sniffing.

2.4.1.4. Drill String Assembly

The drill string assembly, Ref. 20, is comprised of the drill bit and core barrel assembly, mated to multiple interconnected drill rods.

Drill Bit. The drill bit rotationally bores into the waste to produce a nominal 1-in.diameter core sample, and acts as the leading tip of the drill string. The bit has a hollow-cored center section surrounded by cutting teeth and holes on the drilling surface for nitrogen purge flow. The commercially available unit (nominally 2.5-in. o.d.) is made of copper-based (sintered bronze) material with teeth designed to "smear" when they come into contact with the bottom of the tank to prevent penetration. Appendices T and F address the safety concerns in regard to material properties.

Core Barrel. The core barrel, when screwed onto the drill bit, forms an assembly that houses the universal sampler. The fluted core barrel with drill bit is 40 in. long with a 2.25-in. o.d (an effective 2.5 o.d. with 1/8-in. flutes). It also is a commercially available unit made of nickel-plated carbon steel. The serrated edges, or grooves, are machined into the inside of the core barrel so that the quadralatch fingers can slide over them easily in one direction (toward the drill bit) but cannot normally slide past in the opposite direction (away from the drill bit). The section of the core barrel containing the serrated edges is made of 304 stainless steel.

Drill Rods. The drill string is comprised of drill rods that are sections of thin-walled pipe that when mated together transmit power between the quill rod and the drill bit/core barrel assembly. The commercially available drill rods used in the waste have a 2.25-in. o.d and 1.91-in. i.d., with a spirally-wound, fluted ribbing (approximately 1/4 pitch) on the nickel-coated exterior surface to remove drill

debris; drill rods above the waste surface may be unfluted. Each drill rod has an internally threaded (female) end and an externally threaded (male) end, and is available in 19-in., 60-in., 24-in., 12-in. and 6-in. lengths. The drill string is constructed of rods until the drill bit is just above the waste surface. At that point, only 19-in. rods are attached to the top of the drill string, consistent with drilling 19-in. core samples of waste.



Fig. 2-5. Universal sampler.

Pintle Rod. The pintle rod attaches to the piston in the sampler and holds the piston in place

during sampling when the grapple is attached. A pin on the pintle rod trips the trigger mechanism to close the rotary valve. The grapple removes the pintle rod by releasing a spring clip connecting the rod to the piston.
Seal. A chevron seal is used to prevent the flow of waste into the drill string when hydrostatic pressure is not present.

Rotary Valve. The rotary valve is located at the bottom of the sampler and is rotated closed after completing the sampling of a 19-in. segment of waste. The sample is sealed inside the sampler when the valve is closed by actuating a spring-loaded trigger mechanism as the pintle rod is separated from the piston during the sampling operation.

2.4.1.6. Shielded Receiver Assembly

The SR assembly, Ref. 22, is schematically represented in Fig. 2-6, and consists of the weatherproof cover, sampler hoist box with an enclosed winch system, the shielded receiver tube, the RLU that is attached to the sampler hoist cable, and an isolation ball valve attached to the bottom end of the SR tube. The shielded receiver design is independent of the core sampling mode and provides interim sampler shielding. A power winch internal to the weatherproof covering, a cable, and a reel internal to the sampler hoist box are used to retrieve the sampler from the drill string, and to deposit the sampler in the transfer cask. The SR assembly is also used to remove a clean sampler from the transfer cask and transfer it to the drill string for the next sampling operation. The receiver valves, receiver tube, pressure vessel, and pneumatic piping connecting the shielded receiver to the purge gas enclosure provide containment for hydrostatic head gases. The receiver has a load cell to detect cable tension and slack cable, and has a decontamination spray wash. A mechanical counter and digital encoder are used to determine the depth of the RLU and are attached to the cable reel shaft inside the weather cover.

Sampler Hoist. The 1.5 hp sampler hoist motor, Ref. 23, raises and lowers the cable at a maximum speed of 23 ft/min, and employs a cable with a breaking strength of 3000 lb. A load cell shuts off the motor at a maximum load of 300 lb.

Shielded Receiver Tube. The SR tube, Ref. 24, provides shielding for personnel, thereby reducing radiation exposure, and aids in transferring and depositing samplers into the transfer casks.

Remote Latch Unit. The RLU, Ref. 25, is a mechanical latching device that provides a mechanism for latching onto and releasing the sampler. The configuration shown in Fig. 2-7 schematically represents the most recent design, as provided by WHC personnel in December 1995. The RLU is raised and lowered by the sampler hoist assembly.

Ball Valve. The 3-in. ball valve at the bottom of the shielded receiver tube, Ref. 26, isolates the shielded receiver from the surrounding environment and has a male Kamlok® interface.

Kamlok® adapter assemblies. The Kamlok® adapter assembly, Ref. 27, is a commercially available, two-part, male/female assembly that provides rapid, manually actuated connect and disconnect capabilities. In general, the male

Kamlok® adapter is connected to the female Kamlok® adapter. The design convention was established so that movable components, like the shielded receiver, have a male Kamlok® adapter, and the stationary components that are connected to the shielded receiver, such as the change-out assembly, X-ray system, and cask system, have female Kamlok® adapters.



Fig. 2-6. Shielded receiver assembly.

string, and prevent gas accumulation.

2.4.1.7. Purge Gas Enclosure Assembly

The purge gas enclosure assembly, Ref. 28, is located on the truck's rotary platform, and houses, protects, and includes the pneumatic components used to monitor and distribute the hydrostatic head and gas purge nitrogen supplies (including solenoid regulators, valves, analog gauges, control valves, piping, wiring, and instrument transducers.)

Nitrogen is supplied for five different functions during FG/RMCS operations: the DS purge gas system used during FG/RMCS drilling; the purge through the riser sleeve annulus, the hydrostatic head in the drill string and in the shielded receiver, and the Zpurge (NFPA 496) in the SR weather cover. The systems provide drill bit cooling and cleaning during rotary drilling, help prevent waste flooding in the drill



Fig. 2-7. Mechanical remote latch unit configuration.

2.4.1.8. Instrumentation and Control System This section is primarily focused on the systems for drill engine shutdown, and the instrumentation used by the operator for operational controls.

2.4.1.8.1. Programmable Logic Controller (PLC). The PLC processes out-of-tolerance alarm signals to activate alarm strobe, horn, and indicator lights, followed by engine shut-down signals sent to the shut-down relay. Located in the instrument cabinet, the PLC controls the alarm sequencing and interlock logic for the RMCS Truck System. Alarm contacts from the truck instrumentation and external sources (exhauster) are monitored. When a valid alarm is received by the PLC, it initiates both visual and audible annunciation as appropriate. Additionally, if the received alarm requires a drill rig engine shutdown, the PLC deenergizes the shut-down relay, shutting down the drill rig engine.

2.4.1.8.2. Engine shutdown. The following mechanisms are elements of a safety system referred to sampling truck the engine as shutdown, defined as in Ref. 9. The drill rig engine will automatically shut down for out-of-tolerance drilling parameters, exhauster shutdown, or detection of a GRE.

Shutdown Interlock (K5 relay). The shut-down interlock relay is controlled by the PLC and shuts down the drill rig engine by interrupting electrical power to the drill rig ignition.

RPM (Revolutions per Minute) Measurement. Two drill rotation sensors measure drill rotational speed and send signals to two digital units that display drill rpm. If out of tolerance, an alarm signal is sent to the programmable logic controller, and the drill rig engine is shut down. Exhauster operation and nitrogen purge flow are not terminated under this shut-down condition.

Down Force Measurement System. The down-force measurement system electronically measures and calculates the down force of the drill string, provides a signal to the digital display unit, and digitally displays the measurement. If the down force is above the designated down force limit, an alarm signal is sent to the PLC which shuts down the drill rig engine. Exhauster operation and nitrogen purge flow are not terminated under this shut-down condition.

Riser sleeve Purge. Two differential pressure switches measure the pressure drop across a flow controller, and provide a shutdown signal to the PLC on low differential pressure. Exhauster operation and nitrogen purge flow are not terminated under this shut-down condition.

DS Purge Gas Measurement. The purge gas measurements include three turbines to measure flow and transducers to measure pressure and temperature. Signals are sent from the purge gas enclosure to the three digital flow indicators that display compensated flow in scfm (standard cubic feet per minute) units. Any of the three indicators can detect an out-of-tolerance condition and send an alarm signal to the PLC; the PLC then executes two-out-of-three voting logic to activate drill rig engine shutdown.

Penetration Rate Shutdown. The penetration rate measurement system electronically measures the penetration rate of the DS and provides a signal to the digital display unit, and digitally displays the measurement. If the penetration rate is below the designated limit, an alarm signal is sent to the PLC which shuts down the drill rig engine. Exhauster operation and nitrogen purge flow are not terminated under this shut-down condition.

Exhauster-Induced Shutdown. The exhauster can induce drill rig engine shutdown based on signals from the flammable-gas detection system or based on exhauster operational parameters. The operational parameters that provide a shut-down signal to the PLC to shut down the drill rig engine are discussed in Section 2.4.2 A keylock override switch allows operation of the truck when the exhauster is not needed.

2.4.1.8.3. Instrumentation Cabinet. The instrumentation that the operator has available on a directly accessible panel for control of the sampling operations is discussed in this section.

Enclosure Temperature Instrument/Display. The instrument enclosure is temperature-controlled with separate air conditioning and heating systems. The temperature instrument/display measures and digitally displays the cabinet temperatures, and an alarm sounds for out-of-bounds $50^{\circ}F < T > 90^{\circ}F$.

Purge Gas Temperature Display. This instrument displays purge gas temperature, and alarms for out-of-bounds conditions that are $< 10^{\circ}$ F and $> 140^{\circ}$ F.

Purge Gas Pressure Display. This instrument converts a transducer signal to a digital display of purge gas pressure, and if greater than the currently set value of 0.3 psig, sends a signal to the PLC. This display is for information only during drilling modes.

Shielded Receiver and Drill String Pressure Displays. These instruments convert transducer signals to a digital display of pressure in the shielded receiver and drill string, respectively, and if greater than the currently used value of 0.2 psig, send a

signal to the PLC. The displays are used by the operators to ensure sufficient presse for sample change-out operations and to verify that the DS and SR are depressively unized when breaking containment. If hydrostatic head pressure is not maintained at the required level, then waste intrusion into the drill string could result.

Other Informational Displays. The Lower Ram Pressure Display converts transducer signals to digital displays for the walkdown or hydraulic bottom function (HBD) setpoint pressure with a selector switch. The Enclosure Indicator Lights provide visual status of various limits and logic controller functions. The Purge Gas Flow Display selects and digitally displays the output from one of three purge gas flow meters.

2.4.1.8.4. Hydraulic bottom detector. When obtaining the final sample, the hydraulic bottom detector detects loss of lower ram pressure, and energizes a solenoid value to automatically reverse the ram direction to raise the drill head.

2.4.2. Exhauster Assembly

The exhauster train, Refs. 29 and 30, as depicted in Fig. 2-8 is composed of a flexible, conductive duct connecting the exhauster to the riser, a heater to dehumidify exhaust gases, a filter housing containing a prefilter with two high-efficiency particulate air (HEPA) filters in series, and a stack assembly. The exhauster system, designed to operate continuously, is required during all FG/RMCS activities to maintain a negative tank pressure with respect to atmospheric pressure and to prevent uncontrolled particulate emissions.

Exhauster. The exhauster filter train is composed of a prefilter immediately upstream of two HEPA filters in series mounted on a single skid (15 by 7 ft). To limit filter loading, the allowed dose rate on contact with the HEPA filter housing is 100 mrem/h.²⁹ Flow into the exhauster from nitrogen purge and tank in-leakage is designed for $9.4 \times 10^{-2} \text{ m}^3/\text{s}$ (200 ft³/min), resulting in a tank pressure of about -250 Pa (-1 in. w.g.).

The flexible exhauster duct connects the tank riser to the exhauster and is held in place with stand assemblies. The electrically conductive duct is 1/32 in. thick, neoprene over a polyester base. A seal pot assembly is positioned between the riser and the exhauster, and the drain lines to the seal pot are 1/2-in. stainless steel.

Some tanks have high humidity levels. Therefore, a hot-water heat exchanger meeting Class-1, Div.-I, Group-B electrical requirements is supplied upstream of the HEPA filters to lower the relative humidity of the tank gases being exhausted. The heater meets the constraints of the Washington State Operating permit that limits the humidity of the air stream passing through the HEPA filters to be no greater than 80%.





Fan and Stack. The fan, driven by a 7.8 hp electric motor, is qualified for operation in a flammable-gas environment, provides the motive force for drawing tank gases from the riser through the filters and prevents tank overpressurization during core sampling activities. The fan/motor combination are capable of drawing 1900 scfm, so the motor has a speed controller to reduce the rpm to obtain the nominal 200 scfm flow rate. The controller uses output from the stack velocity transmitter to automatically adjust motor rpm to maintain constant flow as loading increases and flow decreases.

Exhauster Control and Monitoring. The exhauster is designed to automatically shut down when flow through the stack is greater than 250 scfm for greater than 5 min., or shutdown immediately when flow is less than 150 scfm. Two pressure conditions can also induce automatic exhauster shutdown: when the tank pressure falls to less than -3 in. w.g.; and when the differential pressure across the prefilter/HEPA filter bank is greater than 5.9 in. w.g. Vendor information indicates that the HEPA filter performance is undetermined at a differential pressure of 10 in. w.g. after 15 minutes, but there are no relief valves or vacuum breakers installed to protect the HEPA from excessive delta pressure. In order to protect against filter collapse, the blower is limited to 9 inches of water static pressure.

The tank pressure, HEPA differential pressures, and flow through the exhauster are continuously monitored using intrinsically safe systems in the exhauster. Even though totalized flow is continuously data-logged in the exhauster instrumentation cabinet, only pressure alarms are recorded.

Exhaust gases can be monitored for radionuclides, ammonia, and total vapor space organic compounds at the stack outlet. The exhauster has no organic or toxic vapor control technology.

2.4.3. Flammable Gas Detection System

The flammable gas detection system consists of four primary components; a spool piece with gas sensors to obtain gas samples from the exhaust stream, two identical, separate, electronic packages and a power distribution skid with redundant shut-off contactors. The system is powered by the same source as the exhauster. The flexible duct from the waste tank is attached to the spool piece which is bolted directly to the exhauster heater. The ventilation stream passes through the spool piece and into the exhauster. Attached to the spool piece are two separate flammable gas sensors; a Whittaker hydrogen detector cell and a Sierra Monitor Corporation (SMC) combustible gas detector.

The purpose of the gas sensors on the spool piece is to provide safety shutdown signals for both flammability and toxic hazards during core sampling operations. Out-of-tolerance conditions include concentrations of hydrogen equivalent flammable gas greater than 5000 ppm, or concentration rate increases greater than 100 ppm/s for 10 s. Upon detection of out-of-tolerance conditions, the interlock will initiate drill rig shutdown and alert personnel to evacuate the tank farm.

The Wittaker Cell, an electrochemical cell with a membrane placed between the sample gas and the active element, is very selective for hydrogen and responds directly to the partial pressure of hydrogen on the other side of the membrane. Significant experience with Wittaker Cells has shown them to be stable and reliable in the tank farm environment.

The SMC combustible gas sensor uses a catalyst to "burn" the gas and detects the resulting heat release. To increase sensitivity and decrease drift, the heat detection is done by comparing the temperature of a reference (uncatalyzed bead) to that of a signal (catalyzed) bead. The beads are imbedded in a sintered metal housing which prevents the combustion energy from igniting a flammable mixture. It has the advantage of responding to both ammonia and hydrogen. Appendix U presents functional design requirements of SMC combustible gas sensors as well as Wittaker cells.

Sample flow to each instrument is provided by a pressure differential within the spool piece--no sample pumps are used. Signals from the flammable gas instruments are processed by redundant programmable logic controllers. If flammable gas concentrations exceed 5000 ppm, the rate of change in flammable gas concentrations better 100 ppm/s for 10 seconds, or the tank pressure increases more than two inches water gage in five minutes, the exhauster will remain operational

and the truck will be shut down. In addition, the exhauster will shutdown on internal alarms (low and high flow, and HEPA filter differential pressure) when the interlock is used.

If interlock power is lost, or tank pressure falls to less than -3 inches w.g. electric power to the exhauster is terminated. Exhauster shutdown will automatically result in core sampling drill truck shutdown via the existing connection.

2.4.4. Riser and Adapter Equipment

This section discusses equipment attached to the riser as illustrated in Fig. 2-9.

Riser and sleeve. The riser to be sampled will have an internal sleeve of sparkresistant stainless steel with a nominal length of 15 ft. The annulus between the sleeve and the DS will be purged with nitrogen during FG/RMCS operations to prevent the accumulation of flammable gas. The riser purge gas system will have two differential pressure detectors which are interlocked to the PLC and can cause an automatic trip of the drill rig. Each detector's set point will be approximately 40 psid across a flow controller that is sized for 5 scfm. The sleeve has a separate spray wash assembly with operational parameters like the DS spray wash system.

Riser Adapters. The riser adapter, Ref. 31, is basically a flanged plate, located on top of the riser, with an offset orifice to allow for the connection of riser equipment, regardless of the size of the riser.

Drill String Spray Washer/Frisbee Wiper Assembly. The drill string is washed to reduce contamination with a hot-water spray wash of the exterior surfaces as the drill string is being extracted through the drill string spray washer/frisbee wiper assembly.³² Water is supplied to the spray washer at a temperature less than or equal to 140°F and a flow rate less than or equal to 3 gal./min from the water heater and 55-gal. water supply on the support truck.

The frisbee seal around the drill rod provide a wiping action during drill rod recovery operations and serves to stabilize the drill string during rotation. The frisbee also effectively provides a seal between the tank and the environment by sealing around the drill string outer diameter, and between the spray washer and the foot clamp.

Pneumatic Foot Clamp. The commercially-available pneumatic foot clamp³³ holds the drill string when it is disconnected from the platform hoist, the quill rod, or the shielded receiver. The three-legged, spider-like clamp must be pneumatically opened to release the drill string. If the pneumatic pressure is lost, the clamp fails in the closed position and the spider-like legs rotate for a three-point positioning around the drill string, locking the drill string in place. The foot clamp does not prevent upward motion of the drill string.

Locking wrench. A commercially-available, carbon steel locking wrench³⁴ is used in conjunction with the pneumatic foot clamp as a redundant mechanism to support the drill string when drill rods are installed or removed, or when the drill string is disconnected from the drill rig. Positioned just above the foot clamp, the fingers of the wrench completely surround the drill string, employing a toothless, ratchet-action grip to grasp the drill string.

Lifting Bale. An electrically-bonded lifting bale is attached to the hoist to support the drill string during installation and removal of drill rods.



Fig. 2-9. Riser adapter, washer/frisbee wiper assembly and foot clamp.

2.4.5. Cable Spray Washer Assembly and Changeout Assembly

Drill String Spray Washer The change-out assembly,³⁵ Fig. 2-10, is placed on top of the cable spray washer when the drill string is disconnected from the quill rod. It provides a means to isolate and maintain hydrostatic head pressure within the drill string while samplers аге exchanged. It also provides containment of hydrostatic head and DS gases. It is comprised of male Kamlok® adapter assembly, a 3-in. ball valve, and a female Kamlok® adapter. Once attached to the DS, the

change-out assembly provides quick connect and disconnect capabilities to the shielded receiver.

Cable Spray Washer Assembly. The cable spray washer assembly³⁶ connects to the drill string before installation of the change-out assembly in order to wash the internal cables of the shielded receiver and grapple cables, and to internally wash the drill rods if required.



2.4.6. Primary Support Equipment

The primary support equipment for rotary sampling operations include the N, supply system, an X-ray imaging capability, power and distribution, generation washer supply systems, and casks and the cask truck. Operational support equipment include the power distribution trailer, the breathing air compressor, the service and support truck and trailers, lights, and the crane used during equipment setup and teardown.

Fig. 2-10. Change-out assembly.

2.4.6.1. Liquid Nitrogen Supply Trailer and Heater

The liquid nitrogen support trailer and vaporizer supply the nitrogen used for the purge and hydrostatic head systems during FG/RMCS operations.

Nitrogen Trailer. In the nitrogen trailer, liquid nitrogen is vaporized by a propanefired, forced-convection, water bath vaporizer. The 1500-gal. nitrogen tank and liquid supply piping stores liquid nitrogen and includes a passive closed-loop evaporator (to supply tank pressure), valves, piping, regulators, and gauges to accommodate tank filling and liquid nitrogen supply to the vaporizer.

The normal nitrogen system pressure in the N_2 trailer is 100 to 250 psig while the system is in operation, with a tank relief valve set to relieve pressure at 250 psi. Nitrogen provided to the sampling truck is 100 to 150 psi. The nitrogen trailer remains outside of the tank farm at all times.

Vaporizer. The vaporizer vaporizes both the liquid propane to supply the water heaters and the liquid nitrogen to supply the nitrogen gas regulator and supply hose. The vaporizer includes self-igniting, thermostatically-controlled water heaters, a water circulation pump, and closed-loop water piping and expansion tanks. The control panel and instruments automatically regulate water flow and gas exit temperatures. A nitrogen gas regulator with a shutoff valve regulates the pressure of the gas at the exit of the vaporizer. A supply hose with a quick disconnect fitting supplies gas to the core sample truck. The vaporizer is electrically-powered by a propane-fueled engine generator, or alternate 240 v power source. The vaporizer's propane tank stores pressurized propane and includes piping to supply the water heaters with propane gas, the generator engine with propane gas, and the vaporizer with propane liquid.

2.4.6.2. Mobile X-ray Imaging System

The X-ray imaging system, Ref. 37 is used for a preliminary assessment of the core sample to verify how complete the sample was and the characteristics of the waste form. This assessment is intended to help the operator more accurately set operational controls for the next sample. The system is equipped with a female Kamlok® adapter for connection to the SR.

2.4.6.3. Casks and Sample Transfer Truck

The sample casks and transfer truck are discussed in this section.

Transfer Casks. The transfer casks are held at the sample site in a five-cask holder, or cask stand, in an upright position.³⁸ The transfer casks are lead-lined chambers that provide shielding and containment for the core samples during shipment to the analytical laboratory. Each cask is 40 in. long, about 6 in. in diameter, and weighs 480 lbs. Casks are equipped with a female Kamlok® adapter for connection to the SR.

Sample Transfer Truck. The sample transfer truck, or cask truck, transports the sampler/cask assemblies to and from the laboratory and moves samples in the field. The truck is capable of carrying three casks at a time, and field positioning is facilitated through an overhead rail chain hoist crane on the truck.

2.4.6.4. **Portable Generator Set**

Two types of generators are available to support FG/RMCS activities—150 kVA and 200 kVA. The portable generator set described in Ref. 39 provides standalone power for the core sample truck and auxiliary equipment. The grounded generator is powered by a turbocharged-diesel engine to produce power with a rating of 150 kW, 480 Vac, 60 Hz, $3\emptyset$, 4-wire and 120/240 Vac, 60 Hz, $1\emptyset$. The diesel generator remains outside of the tank farm at all times.

2.4.6.5. **Power Distribution Trailer**

The power distribution trailer distributes power from the generator to the sampling equipment such as the sampling truck, the exhauster, the water heater on the support truck, the X-ray imaging system, the truck's air compressor, and 120V outlets.

2.4.6.6. Breathing Air Trailer/Compressor

The compressor is a two-stage, oil-free design powered by a 30 hp, 480 Vac, 60 Hz, three-phase electric motor. A 30-gal. receiver tank and a 30-gal. surge tank allow the compressor to cycle, and collected moisture is manually drained in the receiver tank. The breathing-air compressor remains outside the fenced area and away from sources of contamination and toxic fumes.

2.4.6.7. Support Truck and Trailer

The support truck transports personnel and miscellaneous equipment, and can be parked on the tank. The support truck acts as a lock-up rack for drill rods, and also carries a drum heater and pump for supplying water. The support trailer, located outside the tank farm, provides equipment storage and shelter for personnel.

2.4.6.8. Crane

A standard crane is used in the setting up and taking down of the sampling activities. It is used on an as-needed basis only and is not normally retained at the job site except when in use.

2.4.6.9. Light Units

Diesel-powered portable light units are used on and off the tank farm during drilling operations. Each light is capable of producing 2500 W of light (5 halogen lights at 500 W per light). The diesel generator units are refueled on an as-needed basis.

2.4.6.10. Tent

A large tent can be installed on top of the tank over the sampling truck and some of the auxiliary equipment. It's purpose is to provide protection against and reduce the impacts of atmospheric weather conditions such as sun, rain, snow, cold weather, wind, etc. The tent weighs 7000 lb and is made stationary with 33,000 lb of weights located on the tent periphery.

2.4.6.11. Video Vehicle

A vehicle weighing 5,000 lb can be used for video documenting the sampling activities. Even though the installation/operation/removal of the video is not within the scope of this SA, the vehicle is mentioned because of its contribution to tank loading.

2.4.7. RMCS System Weights

RMCS operations for single-shell FGWL tanks increase the live weight on the tank dome. TABLE 2-4 lists the calculated weights of various components that could be placed on the tank dome surface.⁴⁰ However, all of the listed components are not simultaneously placed on the tank because of tank load limits. The dome loading for SSTs is controlled by limits specified in the approved procedure,⁴¹ and the additional tank dome loading is considered to be a live load in the WHC evaluation of the tank structural integrity.

Tank structural integrity can be at risk if the FG/RMCS drill string falls and impacts the tank bottom. For this reason, the total weight of the drill string suspended over the tank bottom is an important factor. The total weight will be the sum of the core barrel, sampler, and drill string, but will vary as a function of the drill string length. The drill rod nominally weighs 4 lb/ft, and the universal sampler, which includes the quadralatch and pintle rod, weighs 10.3 lb.⁴⁰ The combined suspended weight for an FG/RMCS operation will peak as sample operations approach the tank bottom

(e.g., 50-ft drill string length effectively would weigh more than 210 lb), but the impa nergy will peak at an intermediate sampling depth because it is a product of the s ended weight and drop height.

| Component | Weight (lb) |
|--|-------------|
| Core sample truck (includes grapple hoist assembly and shielded receiver assembly) | 30,000 |
| Truck platform | 6,000 |
| Universal sampler (11 @ 10.3 lb) | 113 |
| Drill String (50 ft @ 4 lb/ft) | 200 |
| Change-out assembly | 45 |
| Riser adapter and drill rod washer | 280 |
| Riser sleeve | 200 |
| Inlet breather filter stack | 2,000 |
| Support truck | 7,000 |
| Cask truck | 8,000 |
| Cask stand | 300 |
| Casks (5 @ 480 lb) | 2,400 |
| Mobile X-ray system | 5,000 |
| Exhauster and flammable gas detection systems | 12,200 |
| Light plants (2 @ 1000) | 2,000 |
| Video vehicle | 5,000 |
| Tent | 7,000 |
| Tent weights | 33,000 |
| People (10 total) | 2,000 |
| Total Potential Weight | 122,738 |

TABLE 2-4 RMCS COMPONENT WEIGHT BREAKDOWN

2.5. RMCS OPERATIONS

For the purposes of this safety assessment, FG/RMCS operations are divided into four phases as depicted in Fig. 2-11: (1) preinstallation activities, (2) installation, (3) drilling operations, and (4) removal. Key steps and limits are then provided within each phase. The following section describes the operations associated with rotarymode core sampling, and is a summarization of input from safety analyses in Refs 11, and 12, and verbal discussions with WHC personnel.

The fundamental premise of FG/RMCS operations is to minimize the source or ignition and to ensure the capability to enact safe shutdown upon detection of

unacceptable levels of flammable gas and shut down sampling operations until it becomes safe to continue operations.



Fig. 2-11. FG/RMCS process.

2.5. Preinstallation

Preinstallation activities are assumed to include site preparations, equipment setup and ...nnection, electrical bonding, and verifying critical alarms and trips. Fig. 2-12 schematically shows the relative positions of the pieces of FG/RMCS equipment on and surrounding any given tank, along with the anticipated power requirements.

Setup and operation of all equipment is assumed to be in compliance with appropriate procedures. For information only, several commonly used procedures are listed in TABLE 2-5.

Preinstallation activities include:

- Collect all appropriate procedures.
- Obtain sign off on all necessary conditions, concurrence, forms, and permits.
- Comply with all contractor safety, radiation and contamination, environmental protection, permitting and quality assurance controls, procedural limits and precautions, and records maintenance.
- Investigate and identify farm, tank, and riser locations. Prepare the site. Acquire and stage all supplies and equipment needed to perform operations. Calibrate measurement devices as procedurally required.
- Verify tank ventilation method and operability. Verify spark resistance of tools and lanyard as necessary to prevent tool entry into tank.
- Set up auxiliary support equipment, including the generator, the compressor, power distribution trailer, the support truck with the drum H_2O heater, nitrogen trailer, and service trailer. Position and set up primary systems, including the sampling truck, exhauster, the cask stand with casks, and the mobile X-ray image system.
- Perform grounding and bonding activities. Call the weather service to verify that there are no lightning storms within a 50-mile radius of the sample site.
- Measure the quill-rod-to-riser distance, and determine the number and size of drill rods needed in accordance with the procedure data sheet. Obtain drill rods. Place drill rods in the lockup rack on the support truck.
- Place the quill rod in a full down position, hydraulically level the sampling truck, and verify stability. Verify alarms and annunciators.



Fig. 2-12. Relative placement of equipment required for core sampling in the rotary mode.

2.5.2. Installation

Installation activities are assumed to include removing the riser blank flange and gasket, installing the riser sleeve and riser equipment (riser adapter, spray washer, frisbee wiper, and foot clamp), and core barrel/drill bit/sampler unit through the riser equipment. It should be noted that during the opening of a riser, either toxic gases (i.e., ammonia, organic vapors, and nitrogen dioxide), and/or combustible gases (i.e., methane, hydrogen) could be released and are monitored in a way consistent with appropriate procedures. Operations may proceed only if the combustible-gas meter, calibrated according to appropriate procedures, reads $\leq 25\%$ LFL.

- Crack open the blank flange or pipe cap to off-gas the tank for 5 minutes. Start air sampling. Perform a breathing zone survey and sniff the riser and surrounding area.
- Install the riser assembly, using the mobile crane or the platform hoist. The riser assembly is assumed to include the riser adapter, conductive sleeve, spray washer/frisbee wiper assembly, and foot clamp.

- Screw the drill bit onto the core barrel and gently insert a universal sampler. Attach and install the drill rods in the order specified by the procedural worksheet. Screw the electrically-bonded lifting bail onto each newly attached drill rod, open the foot clamp, and lower the DS with the platform hoist. Close the foot clamp each time before releasing the DS to attach another drill rod.
- Install drill rods until the string is just above the waste surface or until only 19-in. drill rods remain. The predrilling configuration should resemble Fig. 2-13.

TABLE 2-5 EQUIPMENT PROCEDURES USED IN SA

| • | Perform Rotary Core Sar Organic, Organic/Ferrocyar Storage Tanks ⁴² | TO-080-056 | |
|---|--|------------------------------|------------|
| • | Liquid Nitrogen Trailer, Nitr Nitrogen Heater Operations | TO-060-345 | |
| • | ONAN 150DGFA Generator S | TO-020-900 | |
| • | AEROFLOW Model 2AN137 Operation | TO-020-056 | |
| • | Transfer the On-site Transfer Cask | | TO-080-090 |
| • | Sample Transfer Truck Operation | | TO-080-075 |
| • | Pick Up/Transport Radioactive Material and Waste Packages | | TO-100-010 |
| • | Katolight Model D200FRJ4 Standby Power System TO-020-825 Operation | | |
| • | Perform Waste Generation, Segregation and TO-100-052 Accumulation | | |
| • | X-ray Procedure To be specified following SA approval | | A approval |
| • | Exhauster Procedure | To be specified following SA | A approval |

2.5.3. Sampling Operations

Sampling operations are assumed to include drilling operations, removal of the universal sampler from the drill string, X-ray imaging, placing the sampler into the receiving cask, obtaining a new sampler, and placing the new sampler into the drill string.



Fig. 2-13. Pre-drilling configuration.

2.5.3.1. Drilling

- Monitor and record flammable-gas concentrations and other appropriate parameters, according to the operational safety document (OSD) requirements.
- Rotate the truck platform to position the grapple hoist assembly/drill unit over the drill string. Attach a 19-in. drill rod to the drill string. Open the drill chuck, connect the quill rod, and close the chuck. Open the foot clamp. Raise the drill rams if necessary.
- Establish nitrogen purge flow through the riser/sleeve annulus.
- Ensure that the exhauster is operating according to the appropriate procedure, verify that the exhauster interlock is activated, and initiate nitrogen purge gas flow through the DS.
- Lower the grapple through the quill rod with grapple hoist until it latches onto the pintle rod of the universal sampler. Record the mechanical grapple counter value. Raise the grapple only enough to remove the slack in the grapple hoist cable.
- If obtaining a final sample, activate the hydraulic bottom detector, with independent verification.
- Establish the nitrogen purge gas flow at about 40 scfm, or as necessary. Engage the clutch, adjust the DS rpm, and proceed with rotary drilling by adjusting

the ram control value to obtain the desired penetration rate, i.e., down force, as appropriate, within the operating parameter envelope. Operational limits on FG/RMCS parameters are provided in Section 2.5.5.1.

• If the drill bit becomes plugged, refer to Section 2.5.5.2.4.

- If the drill string becomes stuck, refer to Section 2.5.5.2.5.
- When a full sample stroke is completed, close the ram control valves, disengage the clutch, and close the nitrogen purge flow control valve.
- Raise the grapple to pull the pintle rod, release the spring clip that separates the pintle rod from the sampler piston, and close the rotary valve at the bottom of the universal sampler. Pull the pintle rod up through the drill string and into the quill rod.
- Raise the drill string about 1 in. to ensure trouble-free installation of the next sampler. Close the foot clamp. Depressurize the grapple hoist box, and verify depressurization.
- Open the chuck, disconnect the quill rod adapter from the DS, and close the chuck. Screw the cap with the male Kamlok® adapter onto the quill rod adapter. Rotate the platform to place the grapple hoist assembly aside from the drill string.
- Screw the cable wash assembly onto the DS. Connect the male Kamlok® of the change-out assembly to the female Kamlok® of the spray washer assembly. Close the change-out isolation valve. Pressurize the DS to maintain hydrostatic head.
- Kamlok®-connect the pintle rod overpack to the bottom of the quill rod adapter, and mechanically release the pintle rod from the grapple into the overpack. Disconnect the overpack from the Kamlok® cap on the quill rod adapter.

2.5.3.2. Removing the Universal Sampler

- Rotate the truck platform to position the SR over the change-out assembly connected to the DS. Connect the Kamlok® on the end of the SR tube to the Kamlok® of the change-out assembly.
- Open the SR ball valve. Establish hydrostatic pressure in the SR. Open the change-out isolation valve. Lower the RLU at full speed to impact on and engage with the quadralatch of the universal sampler.
- If the DS pressure is greater than 0.5 psi times the sample number, vent the excess SR pressure.
- Slowly increase the hoist upward speed to unseat the sampler. The load cell value should read 60 to 70 lbf, but if the value is >150 lbf, then waste could be in the core barrel.

- Raise the sampler through the DS and into the SR tube. Inspect the sampler in the sight glass for cleanliness and record abnormal conditions.
- If the sampler exhibits visible waste material, wash the sampler by connecting the hot water line to the cable spray washer and raising the sampler slowly through the washer. Record water usage on proper data sheets and chain-of-custody documents.
- Raise the sampler into the SR tube. Close the change-out isolation ball valve. Depressurize the SR. Close the SR ball valve. Disconnect the SR Kamlok® from the change-out Kamlok®.

2.5.3.3. Mobile X-ray Image System Operations

- Rotate the truck platform to position the SR over the mobile X-ray image system. Connect the SR Kamlok® to the Kamlok® of the X-ray system.
- Open the SR ball valve. Lower the sampler into the mobile X-ray image system. Complete the imaging and raise the sampler into the SR tube. Close the SR ball valve.
- Disconnect the SR Kamlok® from the Kamlok® of the X-ray system.

2.5.3.4. Sampler Into Receiving Cask

- Rotate the truck platform to position the SR over the receiving cask, and remove the cap from the cask adapter. Connect the SR Kamlok® to the Kamlok® of the cask.
- Open the SR ball valve. Lower the sampler into the cask until the cable is slack. Disengage the RLU from the sampler quadralatch mechanism. Raise the RLU back into the SR tube. Close the SR ball valve.
- Disconnect the SR Kamlok® from the Kamlok® of the receiving cask.
- Prepare the cask for shipping. Remove the PVC sleeve from the cask. Remove the Kamlok® adapter from the cask; install the inner cask container plug, flange, and a new gasket; and install flange bolts. Complete the appropriate data sheets and chain-of-custody documents. Place a Waste Tank Sample Seal on the cask so that the seal must be broken to open the cask.

2.5.3.5. New Sampler Preparation

• If another sample is required, place a new universal sampler into a cask liner, and gently insert the sampler into a new cask.

- Rotate the truck platform to position the SR over the new cask. Connect the SR Kamlok® to the Kamlok® of the new cask.
- Open the SR ball valve. Lower the RLU at full speed to impact on and engage with the quadralatch of the new universal sampler. Visually verify that the sampler is attached while raising the sampler into the SR tube. Close the SR ball valve.
- Disconnect the SR Kamlok® from the Kamlok® of the new cask.

2.5.3.6. New Sampler Insertion into Drill String

- Rotate the truck platform to position the SR over the change-out assembly connected to the DS. Connect the SR Kamlok® to the Kamlok® of the change-out assembly. Open the SR ball valve.
- If directed by the PIC (person in charge), wash the drill bit by adding 0.1 to 0.3 gal. of hot water through the cable wash assembly. Criteria for this direction are circumstances in which the drill string has been idle for >4 hours or purge flow has not been established. Record water usage on the appropriate data sheets and chain-of-custody documents.
- Pressurize the SR. Open the change-out isolation valve. Lower the RLU and new sampler into the DS until the cable is slack (NOTE: there is no indication that the new sampler is fully latched into the core barrel/drill bit assembly.) Disengage the RLU from the sampler quadralatch mechanism. Raise the RLU back into the SR tube. Close the change-out isolation valve.
- Depressurize the SR. Close the SR ball valve.
- Disconnect the SR Kamlok® from the change-out Kamlok®.
- Rotate the truck platform to position the grapple hoist assembly/drill unit next to the drill string.
- Depressurize the drill string. Remove the change-out assembly.
- Return to Section 2.5.3.1 until the last core segment is obtained.

2.5.4. DS Removal

Removal operations are assumed to occur when the DS equipment is washed and removed. The hoist or drill head is connected to the drill string, and the DS is retrieved and externally washed. The hoist or drill head is disconnected, and the

drill sections are discarded. Auxiliary equipment is disassembled, and the site is restored.

The drill string is washed as follows. Retain the last sampler in the core barrel, or install a new sampler. Depressurize the drill string, connect the water line to the drill rod washer, disconnect the change-out assembly and cable spray washer, and use the drill head to remove all 19-in. drill rods, washing while removing. Retrieve the final sampler when above the waste surface. Rotate the platform, and use the platform hoist to retrieve the remaining segments of drill rod and the core barrel, washing while removing.

2.5.5. Operational Conditions and Characteristics

RMCS activities include both normal and abnormal operating conditions during drilling and sample retrieval.

2.5.5.1. Normal Operations

Normal operations include normal rotation within the established parameter envelope, a walkdown mode, and a bottom-detection mode. Truck stabilization can also be described in certain cases as a normal condition.

2.5.5.1.1. Normal Drilling and Sample Retrieval. TABLE 2-6 lists pertinent operational characteristics associated with the normal drilling and sample retrieval activities described in Section 2. Nominal values are provided, along with minimum and maximum range values. Alarm and trip points, if appropriate, are specified.

TABLE 2-6 OPERATIONAL PARAMETERS FOR ROTARY-MODE CORE SAMPLING METHOD

| | Normal | High | Low | | |
|---|-------------------------------|-------|-------|---------------------|---------|
| Parameter | Range | Value | Value | Alarm | Trip |
| Down force* (lb) | 0 to 750 | 750 | 0 | 750 | >750 |
| DS rotational speed* (rpm) | 2 to 55 | 110 | 2 | 55 | >55 |
| RMCS enable system (rpm) | 2 | 2 | 2 | NA | 2 |
| DS purge gas flow (scfm) | 30 to 50 | 100 | 0.1 | 30 | <30 |
| DS purge gas pressure (psig) | 30 to 50 | 90 | 0.3 | NA | NA |
| Riser purge gas flow | >40 psid (2-5 scfm) | NA | NA | 40 psid | 40 psid |
| Purge gas temperature (°F) | 60 to 80 (atmosphe ric) | 140 | 10 | <10 >140 | NA |
| Instrument enclosure temperature (°F) | 70 to 80 | 90 | 50 | <50 > 9 0 | NA |
| Penetration rate (in./min) | 3 to 10 | 25 | 0 | 0.75 | <0.75 |
| Lower ram pressure (psi) (walkdown mode) | 50 to 250 | 250 | 20 | NA | NA |
| Hydrostatic DS pressure (psi) | 0.5 to 30 | 35 | 0.2 | NA | NA |
| Hydrostatic DS flow (scfm) | 0.5 to 2 | 7.8 | 0.2 | NA | NA |
| Hydrostatic SR pressure (psi) | 0.5 to 30 | 35 | 0.2 | NA | NA |
| Hydrostatic SR flow (scfm) | 0.5 to 2 | 7.8 | 0.2 | NA | NA |

* Limits are given for automatic shut-down features. Appendix N discusses administrativelycontrolled structural limits. For drill strings shorter than or equal to 45 ft, the down force of 750 lbf and rotational speed of 55 rpm are valid. For drill strings longer than 45 ft, the down force limit is reduced to 650 lbf and the rotational speed is reduced to 40 rpm for structural considerations.

2.5.5.1.2. Walkdown mode. The walkdown mode establishes a setpoint to allow the drill to "walk" through the drill stroke. The mode utilizes a solenoid-operated valve (SOV) to automatically start and stop ram motion by stopping hydraulic fluid flow through the drill rams when the specified pressure is reached.

2.5.5.1.3. Bottom detection. To prevent penetration of the tank bottom, a hydraulic bottom detector (HBD) is activated with the last sample. The four-way valve controls whether hydraulic fluid flows into the top or bottom side of the drill rams, thus controlling the direction of drill ram movement. During normal operation,

the flow control values control the amount of fluid that flows from the downstream side of the drill ram, which controls the drill penetration rate and the amount of force applied to the drill string. When activated, the HBD monitors the pressure on the lower or downstream side of the drill ram. When the hydraulic pressure sensors detect a loss of lower ram pressure, the drill direction is automatically reversed. When the stroke is complete, the HBD alarms may have triggered; the operator will silence the siren, stop the stroke, and disable the HBD.

2.5.5.1.4. Stabilized mode. A stabilized mode for the tank and sampling truck can be defined to include the following:

- If stabilization is required when the tank is open, then the open riser is covered.
- If the sampling truck is connected to the drill string, then it remains connected unless lightning is approaching, in which case, the drill string is disconnected and capped.
- If the sampling truck is not connected to the drill string, then stabilization assumes that the truck is placed in stabilized mode: the sampler is in the drill string; the drill string, the shielded receiver and quill rod are sealed; the shielded receiver and quill rod are above the rotary platform; the skid is traverse centered; the quill rod is to the back of the truck; all control panel breakers on the truck are off, unless otherwise directed by the PIC; the PG, SR and DS Gas switches are off; PG mode switch is positioned to DRILL; PG, SR and DS flow control valves are closed; the foot clamp is closed; the four-way valve is in FLOAT position; the Up and Down ram-control valves (RCVs) are closed; and the hydraulic bypass valve is closed.
- The PIC should record the status of sampling in the log book for recovery from the stabilized mode.

2.5.5.2. Abnormal Drilling Conditions

2.5.5.2.1. Reduced nitrogen flow. If nitrogen purge flow is less than the total of 30 scfm, then the drill rig will be automatically tripped, and drilling will be immediately terminated. The exhauster shall remain operational. The operator will correct the condition that caused the trip before drilling operations are reinitiated.

2.5.5.2.2. Excess rpm or down force on drill string. If the rotational speed exceeds 55 rpm or the down force is greater than 750 lbf, then the drill rig will be automatically tripped and drilling will be immediately terminated. The nitrogen purge and exhauster shall remain operational. The operator will correct the condition that caused the trip before drilling operations are reinitiated.

2.5.5.2.3. Penetration Rate. If the penetration rate falls below 0.75 in./min, then an alarm is triggered, and a 60-second period is provided for operator intervention to increase the penetration rate. If the rate is not increased, then the drill rig will be automatically tripped and drilling will be immediately terminated. The exhauster shall remain operational. The operator will correct the condition that caused the trip before drilling operations are reinitiated.

2.5.5.2.4. Drill bit plugging. If 30 scfm nitrogen purge flow cannot be maintained in normal operations, then the drill bit could be plugged with waste. In this case, the grapple hoist box is depressurized, and 0.1 to 0.3 gallons of hot water are added to the drill string through the quill rod adapter. The water usage is noted on the chain-of-custody for that sample.

Purge flow is then reestablished at 40 to 70 scfm. If the bit is cleared, then operations can resume. If not, then the cognizant engineer is consulted for alternate methods to unplug the bit.

2.5.5.2.5. Stuck drill bit. If the drill bit becomes stuck in the waste, then the grapple hoist box is depressurized if necessary and about 1 gal. of hot water is added to the drill string through the quill rod adapter. The water usage is noted on the chain-of-custody for that sample.

Purge flow is then reestablished at 40 to 70 scfm. If the bit becomes unstuck, then operations can resume. If not, then the cognizant engineer is consulted for alternate methods to loosen the bit.

2.5.5.2.6. DS flooding or structural failure. The procedures for handling DS flooding or unplanned maintenance activities are not covered in this safety analysis. The actions needed to handle DS structural failure or extreme jamming of the DS are not delineated in this document.

2.5.5.3. Loss of exhauster

Exhauster operation can be automatically terminated as a result of exhauster operational issues. Operationally tripping the exhauster will automatically trip the drill rig through the PLC on the truck. Similarly, the PLC also deenergizes the SOV in the nitrogen purge enclosure, which stops the purge flow to the DS.

Operationally, the exhauster automatically shuts down under the following conditions:

- Excessive negative tank pressure (-750 Pa or -3 in. w.g.).
- High differential pressure of 10 in. of water across the optional in-riser prefilter.
- High differential pressure of 5.9 in. of water across the HEPA filter bank.

• Exhaust stack flow greater than 250 scfm or less than 150 scfm.

A seismic event will not invoke automatic shutdown of the exhauster.

2.5.5.4. Gas-release Event

A gas-release event can be measured by the flammable gas detection system, discussed in Appendix U, that is connected upstream of the exhauster, or by the tank pressure system in the tank dome. The flammable gas detection system is setpoint-limited at the equivalent of 5000 ppm hydrogen concentration, or a rate of rise of 100 ppm/second. The flammable-gas detection system is also required to provide a cut off and alarm on out-of-tolerance conditions at 12,000 ppm ammonia for toxic considerations. Likewise, an increase in the tank pressure of 2 in. w.g. above back ground will trip the drill rig.

With out-of-tolerance conditions, the exhauster remains operational, but the drill rig engine operation is terminated. Personnel evacuate the site, don protective clothing, and can return to the tank for further equipment stabilization.

2.5.5.5. Emergency Response

All emergency conditions that could result in personnel injury or equipment damage are handled by the PIC in the following manner.

- Direct personnel to attend to any injured personnel, and evacuate as appropriate. Notify the fire department and the occurrence notification center.
- Depending on the nature of the emergency, and at the discretion of the PIC, stabilize the drill site as much as feasible commensurate with Section 2.5.5.1.4.
- Monitoring should be continued in support of all emergency activities.
- Evacuate personnel, and ensure the prevention of uncontrolled access to the drill site area.

Notify the Sampling Operations and Tank Farms Shift Management of the emergency.

2.5.6. **Restart**

Restart could be required for numerous reasons, including a power outage, exhauster shutdown, loss of nitrogen purge, exceeding drilling setpoints, or even starting a new work day. In general, the following conditions would be verified before sampling is reinitiated. Restart following off-normal incidents should be performed in a way consistent with the requirements of the Interim Safety Basis.

- Complete the Daily Core Sample/Inspection Data Sheet as required by the procedure.
- Turn all breakers on the truck's Core Sampler Power panel to ON. Acknowledge all alarms, and reset all immediate alarms. Resume exhauster operation. Resume sampling operations.

2.6. **REFERENCES**

- 1. "Tank Farm Health and Safety Plan," Westinghouse Hanford Company report WHC-SD-WM-HSP-002, Rev. 0A (September 28, 1993).
- 2. "Hanford Site Tank Farm Facilities Interim Safety Basis," Westinghouse Hanford Company report WHC-SD-WM-ISB-001, Rev. 0 (August 1, 1993).
- 3. D. Hopkins, "Criteria for Flammable Gas Watch List Tanks," Westinghouse Hanford Company report WHC-EP-0702 (August 1994).
- 4. P. Whitney, "Screening the Hanford Tanks for Trapped Gas," Pacific Northwest Laboratory report PNL-10821 (October 1995).
- 5. H. Babad and Turner, "Interim Report for Organic Watch List Tanks at the Hanford Site, Westinghouse Hanford Company report WHC-EPP-0681 (1993).
- 6. "Bounding Gas Release Calculations of Flammable Gas Watch List Single-Shell Tanks," Los Alamos National Laboratory report LA-UR-94-1323, Rev. 0 (1994).
- 7. B. M. Hanlon, "Waste Tank Summary Report for Month Ending October 31, 1995," Westinghouse Hanford Company report WHC-EP-0182-91 (January 1991).
- 8. Westinghouse Hanford Company, "Waste Storage Tanks Composite Section," Drawing No. H-2-55911.
- 9. "Safety Equipment List," Westinghouse Hanford Company report WHC-SD-WM-SEL-032, Rev. 1. (February 15, 1994).
- 10. Reference deleted.
- 11. "Safety Assessment for Push- and Rotary-Mode Core Sampling in Ferrocyanide Tanks," Westinghouse Hanford Company report WHC-SD-WM-SAD-013, Rev 3-A (November 1993).
- "Safety Analysis for the Push Mode and Rotary Mode Core Sampling," Westinghouse Hanford Company report WHC-SD-WM-SARR-031, Rev. 2 (July 1995).
- 13. "Safety Requirements for Safety Basis for Rotary-Mode Core Sampling Flammable Gas Safety Issue Tanks," Westinghouse Hanford Company report WHC-SD-WM-FDC-048 (September 1995).
- 14. R. J. Blanchard, "Justification on Functional Criteria for Rotary Mode Core Sample Truck," Westinghouse Hanford Company memo (January 19, 1995).

August 8, 1996

- 15. Westinghouse Hanford Company, "Truck Assembly (RMCST)," Drawing H-2-690000 (June 27, 1995).
- 16. Westinghouse Hanford Company, "Grapple Hoist Assembly (RMCST)," Drawing No. H-2-690057 (March 11, 1994).
- 17. Westinghouse Hanford Company, "Inner/Outer Bellows > Assembly (RMCST)," Drawing No.H-2- 690059 (March 21, 1994).
- 18. Westinghouse Hanford Company, "Grapple Hoist Rotary Valve Actuator (RMCST)," Drawing No. H-2-690052 (March 7, 1994).
- 19. Westinghouse Hanford Company, "Quill Rod Assembly (RMCST)," Drawing No. H-2-690051 (February 17, 1994).
- 20. Westinghouse Hanford Company, "Drill String Arrangements," Drawing No. H-2-821457 (March 29, 1994).
- 21. Westinghouse Hanford Company, Core Sampler Assembly," Drawing No. H-2-690140 (December 1, 1995).
- 22. Westinghouse Hanford Company, "Shielded Receiver Assembly (RMCST)," Drawing No. H-2-690020 (March 11, 1994).
- 23. Westinghouse Hanford Company, "Sampler Housing Assembly (RMCST)," Drawing No. H-2-690030 (March 14, 1994).
- 24. Westinghouse Hanford Company, "Shielded Receiver Tube Section (RMCST)," Drawing No. H-2-690022 (February 19, 1994).
- 25. Westinghouse Hanford Company, "Mechanical Remote Latch Unit," Drawing No. H-2-690142 (January 15, 1996).
- 26. Westinghouse Hanford Company, "3" Ball Valve Modification (RMCST)," Drawing NO. H-2-690023 (February 19, 1994).
- 27. Westinghouse Hanford Company, "Kamlok Adapter Assemblies," Drawing No. H-2-81845 (August 5, 1993).
- 28. Westinghouse Hanford Company, "Purge Gas Assembly," Drawing No. H-2-690060 (January 25, 1994).
- 29. "System Design Description for the Rotary Mode Core Sampling Exhauster," Westinghouse Hanford Company report WHC-SD-WM-SDD-035, Rev. 2 (September 9, 1995).

August 8, 1996

2-43

- 30. Westinghouse Hanford Company, "Exhauster Assembly Rotary Mode Core Sampling," Drawing No. H-2-818186 (March 3, 1994).
- 31. Westinghouse Hanford Company, "12 in. to 4 in. Offset Riser Adapter," Drawing No. H-2-99335 (May 1995).
- 32. Westinghouse Hanford Company, "Drill Rod Washer Housing Assembly and Details," Drawing No. H-2-91670 (September 27, 1994).
- 33. Guiberson Dresser Company, "Tubing Spider, Type T-40."
- 34. Parmelee Company, "Pipe Wrench."
- 35. Westinghouse Hanford Company, "Change-out Assembly," Drawing No. H-2-690133 (December 1995).
- 36. Westinghouse Hanford Company, "Cable Spray Washer Assembly," Drawing No. H-2-81842 (May 1993).
- 37. Westinghouse Hanford Company, "Mobile Core Sample X-ray System," Drawing No. H-2-79076 (June 22, 1995).
- Westinghouse Hanford Company, "Transfer Cask Assembly," Drawing No. H-2-38079 (August 1994).
- 39. "Tank Farm Operating Procedure, ONAN 150DGFA Generator Set Operation," Westinghouse Hanford Company procedure TO-020-900, Rev. B-3 (July 1995).
- 40. D. Hamilton, J. Smalley, and J. Criddle, Westinghouse Hanford Company personal communication (November 1995).
- 41. Boyles, "Operating Specifications for Single-Shell Waste Storage Tanks," Westinghouse Hanford Company report OSD-T-151-00013, Rev. 0 (1994).
- 42. "Tank Farm Operating Procedure, Perform Rotary Core Sampling of Ferrocyanide, Organic, Organic/Ferrocyanide Watch-List Waste Storage Tanks," Westinghouse Hanford Company procedure TO-080-056, Rev. A-1 (October 10, 1995).

 $\sum_{i=1}^{n}$

3.0. IDENTIFICATION OF HAZARDS

This section presents the methodology and results of a hazards identification study used to formally identify all hazards associated with the proposed action of the installation, operation, and removal of rotary-mode core sampling (RMCS) equipment in single-shell tanks (SSTs) on the Flammable Gas Watch List (FGWL) or those tanks recommended by the contractor to be included on the FGWL, hence referred to as FG/RMCS operations. This process, called hazard identification, is equivalent to the hazard analysis (HA) or hazard evaluation process in a safety analysis report. The final product of this hazard identification process is a list of design-basis accidents (DBAs) that will be examined in more detail in the accident analysis section of the safety analysis (SA).

According to Ref. 1, hazard is defined as "A source of danger (i.e., material, energy source or operation) with the potential to cause illness, injury or death to personnel or damage to an operation or to the environment." A hazard is not an accident initiator, cause or deviation. Rather, it is a property, typically radiological and toxicological, inherent to the danger. Based on this definition, the hazards associated with the FG/RMCS operations in flammable-gas tanks are summarized in Table 3-1.

The major consequence of accidents in the flammable-gas SSTs is the release of radioactive and toxic materials that might expose the public sector and/or on-site personnel to unacceptable doses. Airborne, underground, and surface release pathways are considered in the study. In addition, any structural damage of the tank that would cause major damage in the dome area or leaks in the liquid region are evaluated. Radiological and toxicological consequences resulting from dome collapse are also evaluated in this safety assessment. Leaks below the waste level, however, are identified as potential environmental hazards and the long-term consequences are not evaluated.

3.1. METHODOLOGY

The guidance used to perform the hazard identification (or HA) is DOE-STD-3009 (Ref. 1). DOE-STD-3009 provides the following guidance on the requirements for a complete hazards analysis. A complete hazards analysis should (STD-3009, p. 31)

- 1. Consider the complete spectrum of accidents that may occur as a result of facility operations;
- 2. Analyze potential consequences to the public and worker;
- 3. Estimate the likelihood of occurrence;
- 4. Identify and assess associated preventive and mitigative features;

TABLE 3-1SUMMARY OF POTENTIAL HAZARDS

| HAZARDOUS MATERIALS | | | |
|---------------------------------------|--|--|--|
| MATERIAL | LOCATION | FORM and QUANTITY | |
| Flammable Gases | Stored in the waste with potential release to - the dome space, - the drill string, drill unit and shielded receiver - the environment (above | Major species are hydrogen, ammonia, methane, nitrous oxide (oxidizer), and oxygen. Composition and quantities vary. See discussions in Appendixes C and L. | |
| | ground) | · | |
| Flammable Solids and Liquids | In the waste. | Organic compounds with oxidizers. Composition and quantities vary. | |
| | | See discussion in Appendix G. | |
| Radioactive Solids and Liquids | In the waste | Bounding dose is discussed in Appendix R. | |
| Fissile Materials | In the waste | Bounding quantity discussed in Appendix R. | |
| Toxic Gases | Stored in the waste with potential release to the | Major species are ammonia and nitrous oxide. Composition and quantities vary. | |
| | environment. | See discussion in Appendix C. | |
| Toxic Solids and Liquids | In the waste | Bounding dose is discussed in Section 5 of this SA. | |
| | ENERGY SO | URCES | |
| ENERGY | LOCATION | FORM and QUANTITY | |
| Electrical (sp ark sources) | - Dome space - Ventilation system | Various electrical equipment Material with potential electrostatic | |
| | - Truck - Above ground near risers. | charge built-up. | |
| Mechanical (spark sources) | Dome space Ventilation system Above ground (near risers) Truck Inside the drill string In the waste | Drill bit and drill string with kinetic (rotational and linear) and potential energy Truck and other vehicles with kinetic energy Moving parts with kinetic energy (pumps, motors, etc.) Heavy equipment with potential energy Tools with spark potential Ventilation fan with kinetic energy Compressed gases | |

| ENERGY SOURCES | | | |
|-----------------|-----------------------|---|--|
| ENERGY | LOCATION | FORM and QUANTITY | |
| Chemical | - Tank | - Propane tank | |
| | - Truck (on the tank) | - Diesel and gasoline | |
| | - Tank Farm | - Lubricants | |
| | | - Water and nitrogen added to the waste | |
| Radiant | Truck (on the tank) | - X-Ray machine | |
| | | - Waste samples | |
| External Events | Tank Farm | - Lightning/Tornadoes/Heavy Rains | |
| | | - High Winds/Dust Devils | |
| | | - Earthquake/Volcanoes | |
| | | - Range fires | |

TABLE 3-2 (cont) SUMMARY OF POTENTIAL HAZARDS

- 5. Identify safety-significant structures, systems, and components; and
- 6. Identify a select subset of accidents to be formally defined in accident analysis.

In the hazards identification performed for rotary-core mode drilling, four of the six requirements listed above were met in full. Namely, (1) a complete consideration of the spectrum of accidents, (2) analysis of the potential consequences, (3) estimation of likelihood, and (6) identification of a select subset of accidents for accident analysis (the end product for this hazard identification). However, two of the requirements, (4) the identification of preventive and mitigative features; and (5) identification of safety-significant structures, systems, and components, were only partially met. For the rotary core mode drilling activity, hardware design and procedures were being developed during the hazard identification (hazard analysis) process. Therefore, at that time, preventive and mitigative features, the identification of safety-significant structures, systems, and components could also not be performed. However, identification of preventive and mitigative systems is performed in the design change/control implementation phase of this SA and documented in Section 6.

Hazard identification is the first step in the safety analysis process. The goals of hazard identification for this SA are a subset of the hazard analysis requirements presented earlier, namely;

• Consideration of the complete spectrum of potential accidents,

- Qualitatively assess the consequences to the worker and the public,
- Qualitatively estimate the frequency (or likelihood) of occurrence,
- Identification of a select set of representative and unique accidents (DBAs) for further evaluation.

Because of the relative complexity of the rotary-core drilling system and its unique intrinsic hazards, a detailed hazards identification study was performed considering all phases and aspects of the rotary-core drilling operation. The intent was to meet the requirements of Department of Energy (DOE) Order 5480.23 (Ref. 2) to identify all the hazards and accidents scenarios. The rotary-core drilling hazard identification process was performed by a multidisciplinary team consisting of Los Alamos and Westinghouse personnel, using a combination of two standard techniques, the hazard and operability (HAZOP) technique and "what if" checklist techniques. The operations examined were the installation of the equipment, the individual steps of the rotary-mode sampling, and the removal of the equipment. The operation was tracked in this way to ensure completeness of the HA.

At the beginning of the HA, the Westinghouse Hanford Company (WHC) Characterization Project Engineering and Operations personnel presented and described the operations. Subsequently a process-flow diagram was prepared to describe all phases of the installation, operation, and removal of the rotary-mode sampling equipment on a typical tank. Previous safety analyses, the available design documentation, and the operating procedures were reviewed before another meeting with FG/RMCS engineering and operations personnel. The hazards analysis was developed based on the process-flow sheets and the questions that resulted from the documentation review. Hazards identified in previous safety assessments^{3,4,,6} were reviewed and included in this study. The results of the hazards identification study are documented in Appendix A.

The HA includes estimates of the frequencies and consequences of the hazards that have been combined to provide a risk ranking. The risk ranking is one factor used to select the accidents. The accident database was examined and nine accident classes were selected for further analysis. Section 4 of this SA evaluates these nine accident classes. Alternative groupings are possible, and inevitably the grouping in some cases is not very clear. However, although the boundary between the groups may be subjective, the grouping process was complete, and all of the identified hazards are captured in these groupings.

This SA discusses all of the hazards identified and how they are managed to acceptable levels of risk. Some of this will be a discussion of the design features and controls. In some cases, analyses are used to show that the accident cannot happen physically. Also, analyses are presented to quantify the bounding consequences in the event that preventative and mitigation features are ineffective.

3.2. RESULTS OF THE ROTARY-CORE DRILLING SYSTEM HAZARDS IDENTIFICATION

The hazard identification conducted for the proposed action examined three processes; installation, operation, and removal of the rotary-core drilling unit. The hazards associated with transportation, of a contaminated rotary-core drilling unit or its auxiliary equipment from the tank farm or its ultimate decontamination and disposal are not considered. Transportation of the cask where core samples are stored also is not considered. These activities are included in the safety analysis reports for site transportation waste storage, and handling (see Ref. 7), and are subject to the applicable controls listed there. Operations evaluated included X-ray examination and storing the core sampler in the cask.

The results of the hazards identification indicate that the potential contributors to the release of radioactive and toxic materials and structural damage to the tank can be categorized in nine general categories of accidents. In Appendix A, details of the hazards identification and the general accident categories are presented. A total of 180 scenarios resulting in waste and toxic-gas releases were identified. The individual accidents are evaluated based on their qualitative accident frequency and resulting consequences. In Appendix A, a frequency and a dose class are assigned to each accident in order to rank them. The dose rates indicated in Appendix A are qualitative values. Likewise, frequency determination did not include a detailed failure-rate evaluation, but qualitative frequency estimates are provided. Selection of representative and unique accidents consider frequency and consequence in order to rank individual hazards. These representative unique accidents are categorized in nine groups based on their release characteristics.

The results of the hazards identification process are summarized in Tables 3-2 to 3-10. For each accident category, a separate table is given. The accident, the applicable scenario, principal causes, and design safety features are given for each case.

Specific design-related features, primarily those provided to manage identified hazards, are included because their failure may cause an accident.

The relationship between hazards and accident analyses is determined for each accident in a given category. The accident analysis is cross-referenced to the section in Chapter 4 where the potential accident is evaluated. In some cases, the same accident analysis covers more than one hazard or initiator. In other cases, several accident analyses will be required to assess the various manifestations of the hazard.

The following is the summary of the tables in which the different accident hazard groups are summarized. Industrial hazards such as installations in the wrong tanks or risers, operation of the liquid nitrogen tank, traffic accidents, slips, falls, etc., are beyond the scope of this safety assessment.

SUMMARY OF TABLES

| Group | Table |
|------------------------------------|------------|
| Aboveground fire | Table 3-2 |
| Dome fire | Table 3-3 |
| Drill string fire | Table 3-4 |
| Waste fire | Table 3-5 |
| Chemical reactions and criticality | Table 3-6 |
| Containment breach | Table 3-7 |
| Gas releases | Table 3-8 |
| Spills and radiation exposure | Table 3-9 |
| External events | Table 3-10 |
TABLE 3-2ABOVEGROUND FIRE HAZARD ASSESSMENT RESULTS

| Accident | Scenario | Design Safety Features | Analysis |
|---|--|--|---------------|
| Flammable- gas clease to exhauster and bum | Flammable-gas release from waste into exhauster and ignition source. Ignition in the exhauster and burn back to the tank dome (see Dome | All electrical equipment in the exhauster flow stream must be rated to Class-1, DivI, Group-B environment. Do continuous exhausting with spark- resistant fan. Use heavy-duty, tear-resistant, conductive hose. | Sec. 4.1.1. |
| | Fires) | • Use redundancy/diversity in gas-release detection system (flammable gas and pressure). | |
| Flammable gas release through torn duct and burn | Gas release through torn hose | Use heavy-duty, tear-resistant, conductive hose. Nonqualified equipment behind deflectors or in enclosures. | Sec. 4.1.2. |
| Flammable- gas release and burn outside an open riser | Flammable-gas release from waste through open risers and ignition source Ignition above the tank and burn back to the tank dome (see Dome Fire) | Inlet and exhaust have a stack height of 5 m (15 ft). All electrical equipment near open risers must be rated to Class-1, DivI, Group-B environment or Class-1, DivII, Group-B environment with automatic shutdown. N2 purge of riser liner | Sec. 4.1.2.1. |
| Flammable- gas release and bum in shielded receiver. | • Flammable-gas release from shielded receiver, drill string, and ignition source. | Use hydrostatic head purge in the shielded receiver. Electrical equipment is designed for Class I, Division 1, Group B in the shielded receiver. Spark-resistant mechanical RLU/sampler. Cable hoist structural strength prevents RLU drop. | Sec. 4.1.2.2. |
| Flammable- gas release and bum in the X-ray or cask | • Flammable-gas release into X-ray machine and into storage casks. | X-ray sample liner is made of plastic. No unqualified equipment in Class I, Division 1 or Class I, Division 2 space in x- ray machine. Liner is painted with conductive graphite paint, and grounded and bonded. | Sec. 4.1.3. |
| Flammable- gas release and burn | • Flammable-gas accumulation or release from the loss of electrical power and ignition source | • No unqualified equipment in Class I, Division 1 or Class I, Division 2 space | Sec. 4.1.4. |

.

TABLE 3-2 (cont.)ABOVEGROUND FIRE HAZARD ASSESSMENT RESULTS

| Accident | Scenario | Design Safety Features | Analysis |
|---------------------------------------|--|-------------------------------|-------------|
| Flammable- gas release and burn | • Release from propane tank on nitrogen trailer | • No credited design feature. | Sec. 4.1.5. |
| Flammable material and bum | Flammable diesel and gasoline fuel | • No credited design feature. | Sec. 4.1.6. |
| Equipment fire | • Collision caused by trucks and other equipment | • No credited design feature. | Sec. 4.1.7. |

TABLE 3-3GAS RELEASE AND DOME FIRE HAZARD ASSESSMENT RESULTS

| Accident | Scenario | Design Safety Features | Analysis |
|---------------------------|--|---|-------------|
| Flammable gas and burn | • Drill string break | Automatic shutdown on high flammable gas. | Sec. 4.2.1. |
| Flammable gas and burn | • Equipment, tools, or drill string/ component drop on riser induces mechanical spark. | • Use spark-resistant tools. | Sec. 4.2.2. |
| Flammable gas and burn | • Equipment, tools, or drill- string/component drops on crust. | Pneumatic clamp is designed to fail closed. Use of spark resistant tools. Use of locking collar | Sec. 4.2.3. |
| Flammable gas and burn | Frictional spark in the riser. | Use stainless-steel sleeve in the riser. Inject nitrogen into riser sleeve to prevent hydrogen penetration. Automatic shutdown on loss of sleeve purge. Automatic shutdown on high flammable gas. Unique connectors for sleeve purge. | Sec. 4.2.4. |
| Flammable gas and burn | • Electrostatic spark in the riser | Equipment grounded and bonded. | Sec. 4.2.5. |

777 2 6

TABLE 3-3 (CONT)GAS RELEASE AND DOME FIRE HAZARD ASSESSMENT RESULTS

| Accident | Scenario | Design Safety Features | Analysis |
|---------------------------|--|---|-------------|
| Flammable gas and burn | • Frictional spark caused by drill bit on crust | • Drill bits do not have spark-inducing carbide teeth. Cutting teeth are copper-based soft material. | Sec. 4.2.6. |
| | | • Drill bit design must be qualified by testing to non sparking. | ` |
| | | Automatic shutdown on high rpm and down force. | |
| | | • Use of walkdown function and HBD. | |
| Flammable gas and burn | Spark sources in the dome in the ventilation system, in connected tanks | • All electrical equipment in the dome and ventilation system is rated for operations in Class-1, DivI, Group-B environment or Class-1, DivII, Group-B environment with automatic shutdown. | Sec. 4.2.7. |
| Dust explosions | Aerosol accumulation | Use of qualified exhauster. | Sec. 4.2.8. |

TABLE 3-4DRILL STRING FIRE HAZARD ASSESSMENT RESULTS

| Accident | Scenario | Design Safety Features | Analysis |
|---|--|---|-------------|
| Flammable gas in the drill string | Failure of sampler chevron seal Hydrogen diffusion Waste in the drill string Depressurization of waste Loss of N2 Incompatible material | Sampler chevron seal. Drill string purge gas. Hydrostatic head purge. Shut down on low nitrogen flow. Use of compatible material. | Sec. 4.3.1. |
| Flammable gas and burn in drill string | • Drop impact on drill bit by sampler, remote latch unit, or grapple | Components within the drill string must be qualified to the requirements of Appendix T, or prevented from dropping. Sampler chevron seal. | Sec. 4.3.2. |
| Flammable gas and burn in drill string | Ignition caused by assembly/ disassembly of drill strings Ignition by drill- rod/quill-rod adapter impact | • Components must be qualified to the requirements of Appendix T | Sec. 4.3.3. |
| Flammable gas and burn in the drill string | • Unqualified in the drill head or SR | • Electrical equipment meets Class I, Div. 1, Group B requirements | Sec. 4.3.4 |
| Flammable gas and burn in drill string | • Drill string failure | Use N₂ purge of drill. Sampler chevron seal | Sec. 4.3.5. |
| Ignition source and flammable gas in the drill string | Friction bearings RLU Grapple | Use N₂ purge. RLU/grapple insertion rate limited to 1 ft/s. | Sec. 4.3.6. |
| Flammable gas in the drill string | • Shear pin break | Use N₂ purge.Shear pin is replaced by a clip. | Sec. 4.3.7. |

TABLE 3-4 (cont.)DRILL STRING FIRE HAZARD ASSESSMENT RESULTS

| Accident | Scenario | Design Safety Features | Analysis |
|---|---|--|--------------|
| Detonation and ejection | • Flammable gas ignition in the drill string | Sampler chevron seal. Electrical equipment meets Class-1, DivI, Group-B requirements. | Sec. 4.3.8. |
| | | Shut down on low nitrogen flow. Components within the drill string must be qualified to the requirements of Appendix T, or prevented from dropping. Use of compatible materials. | 1 |
| Ignition in the drill string caused by lightning strike | Lightning strike | • No credited design feature. | Sec. 4.3.9. |
| Ignition in drill string caused by static electricity | Static electricity between O-rings Static electricity on the Frisbee | Maintain continuous contact with metal and is bonded. Sampler design maintains contact with drill string. | Sec. 4.3.10. |

TABLE 3-5 WASTE FIRE-HAZARD ASSESSMENT RESULTS

| Accident | Scenario | Design Safety Features | Analysis |
|-----------------------------------|--|---|-------------|
| Drill bit over- temperature | Loss of N2 Down force Rotational speed Low penetration rate | Automatic shutdown on loss of N2 purge. Automatic shutdown on high RPM. Automatic shutdown on high downforce. Use bottom detector and walkdown function. Automatic shutdown on low penetration rate | Sec. 4.4.1. |
| Exothermic reactions | Incompatible materials | • Use of compatible material. | Sec. 4.4.2. |
| Impact on crust of waste | • Drop of drill string or tool | • No credited deesign feature | Sec. 4.4.3. |
| Gas fire under surface | Spark induced with drill bit impact Inadvertent increase in force and rpm | Components in contact with the waste must be qualified to the requirements of Appendix T. Automatic shutdown on high rpm, down force. Automatic shutdown on low nitrogen purge flow and penetration rate. | Sec. 4.4.4. |

TABLE 3-6 CHEMICAL REACTIONS AND CRITICALITY HAZARD RESULTS

| Accident | Scenario | Design Safety Features | Analysis |
|--|---|--|-------------|
| Criticality | Drilling operation | No credited design feature | Sec. 4.5.1. |
| Gas release caused by water addition | • Water addition | Limited supply of water. Temperature control on water heater. | Sec. 4.5.2. |
| Exothermic runaway reactions Waste melting | Drill bit frictional energy Water addition Loss of N₂ purge Plugged purge holes | Carry out N₂ purge during rotation. Automatic shutdown on high rpm, and down force. Automatic shutdown on low penetration rate and nitrogen purge flow. | Sec. 4.5.3. |
| Energy transfer to/from the waste | Frictional heating Loss of N₂ purge | Automatic shutdown on high rpm, and down force. Automatic shutdown on low penetration rate and nitrogen purge flow. | Sec. 4.5.4. |
| Impact sensitive compounds | • Drilling • Pushing | • No credited design feature. | Sec. 4.5.5. |

TABLE 3-7 CONTAINMENT BREACH HAZARD ASSESSMENT RESULTS

| Accident | Scenario | Design Safety Features | Analysis |
|----------------------------|--|--|-------------|
| Dome Loading | High static loading Dynamic loads Truck falls off platform Crane drop | • No credited design feature. | Sec. 4.6.1. |
| Vacuum | Exhauster failure Inadvertent closure of inlet riser | Automatic shutdown on exhauster with high vacuum. Seal loop on breather filter. | Sec. 4.6.1. |
| Tank bottom penetration | Drill into bottom Drill string drop and penetration | Use hydraulic bottom detector. Use soft drill bit material. Automatic shutdown on high down force. Use of pneumatic foot clamp. | Sec. 4.6.2. |
| Drill-string break | • Excessive down force • Excited frequency | • No credited design feature. | Sec. 4.6.3. |
| Riser damage | Equipment (conductive sleeve and drill string) and tool drops | • No credited design feature. | Sec. 4.6.4. |
| Side penetration | • Drill string failure | • No credited design feature. | Sec. 4.6.5. |

TABLE 3-8GAS RELEASE WITHOUT BURN HAZARD ASSESSMENT RESULTS

| Accident | Scenario | Design Safety Features | Analysis |
|-------------------------|---|--|-------------|
| Toxic-gas release | Drilling operations Multiple release modes Additional N₂ purge | Automatic shutdown on high gas concentration. Use exhauster stack for worker protection. Use of inlet breather filter stack. | Sec. 4.7.1. |
| Unfiltered release | Ventilation failureTank pressurization | Use of qualified exhauster HEPA and breather HEPA filters | Sec. 4.7.2. |
| Steam release | • Drill temperature induces steam | No credited design feature. | Sec. 4.7.3. |
| N ₂ addition | Accumulation in waste causing a gas- release event Ammonia scrubbing | Use of qualified exhauster. Automatic shutdown on tank pressure and gas concentration. | Sec. 4.7.5. |

TABLE 3-9 SPILLS AND RADIATION EXPOSURE HAZARD ASSESSMENT RESULTS

| Accident | Scenario | Design Safety Features | Analysis |
|--|--|---|-------------|
| Exhauster releases | • HEPA failure | Use ΔP limits. Use high- and low-flow shut down exhauster. Loss of exhauster flow shuts down N2 and drill. | Sec. 4.8.1. |
| Exhauster continuous release after filter failure | • HEPA failure | Use ΔP limits. Use high- and low-flow shut down exhauster. Loss of exhauster flow shuts down N₂ and drill. | Sec. 4.8.2. |
| Inlet duct failure | • Breather HEPA filters fail | • No credited design feature. | Sec. 4.8.3. |
| Aerosol | Loss of ventilation flow and failure to shutdown purge | Use of qualified exhauster. Use of breather inlet HEPA filter stack. Use of exhauster stack. | Sec. 4.7.2. |
| Sprays | • No initiators | No credited design feature. | NA |
| Drop sampler | Operational hazard | No credited design feature. | Sec. 4.8.4. |

TABLE 3-9 (cont.)SPILLS AND RADIATION EXPOSURE HAZARD ASSESSMENT RESULTS

| Acadent | Scenario | Design Safety Features | Analysis |
|---|--|--|--------------|
| Open sampler | Stuck ball valve | • Use viewing window. | Sec. 4.8.5. |
| | | Use transfer to shielded cask through closed system. | |
| Spill in core barrel | Waste accumulation in core barrel | • Cable spray washer. | `Sec. 4.8.6. |
| Drop of con- taminated drill string | Ineffective decontamination and drop | • Spray wash system. | Sec. 4.8.7. |
| Radiation exposure | High loading in HEPA | • No credited design feature. | Sec. 4.8.8. |
| | • Spills | | |
| | • Open riser | | |
| | Failure of decontamination | | |

TABLE 3-10 RESULTS OF HAZARD ASSESSMENT FOR EXTERNAL EVENTS

| Accident | Accident Scenario | Design Safety Features | Analysis |
|-------------|---|---|-------------|
| Lightning | • Ignition of flammable gases | • No credited design feature. | Sec. 4.9.1. |
| Winds | Flow induced vibration | Conductive duct. | Sec. 4.9.2. |
| | • Static electricity buildup | • Heavy skid. | |
| | Spread of contamination Operator errors/equipment drop | • Exhauster stack and inlet breather HEPA stack is designed for high winds. | |
| Range fires | • Ignition of flammable gases | • No credited design feature. | Sec. 4.9.3. |
| Seismic | • Tank failure | • No credited design feature. | Sec. 4.9.4. |
| | • Gas-release event | | |
| Tornadoes | • See lightning/flooding/high winds | • No credited design feature. | Sec. 4.9.5. |
| Flooding/ | • Tank overfill | • No credited design feature. | Sec. 4.9.6. |
| heavy rains | Equipment malfunction | | |
| | Operator errors | | |
| Volcanoes | • Flooding, gas ignition, dome loading | No credited design feature. | Sec. 4.9.7. |
| Dust devils | Spread of contamination | • No credited design feature. | Sec. 4.9.8 |
| | Operator error/equipment drop | | |

3.3. **REFERENCES**

- 1. "DOE Standard Preparation Guide for U. S. Department of Energy Nonreactor Nuclear Facility Safety Analysis Reports," DOE-STD-3009-94 (July 1994).
- 2. "Safety Analysis Report," US Department of Energy Order 5480.23 (April 30, 1992).
- 3. N. J. Milliken and G. R. Geschke, "Safety Analysis for Push-Mode and Rotary-Mode Core Sampling," Westinghouse Hanford Company report WHC-SD-WM-SARR-031, Rev. 2 (July 1995).
- 4. N. J. Milliken, "Safety Assessment for Push-Mode and Rotary-Mode Core Sampling in Ferrocyanide Tanks," Westinghouse Hanford Company report WHC-SD-WM-SAD-013 (June, 1993).
- 5. Reference deleted.
- R. L. Koontz, "Hazards Identification and Evaluation for Waste Tank Core-Sampling Equipment," Westinghouse Hanford Company report WHC-SD-WM-SAR-007 (June 1985).
- H. E. Adkins, Jr., "Safety Analysis Report for Packaging on Site, on Site Transfer Cask," Westinghouse Hanford Company report WHC-SD-TP-SARP-002, Rev. 0-0 (November 11, 1995).

•

4.0. HAZARD ANALYSIS

In Section 3, the hazards associated with the installation, operation, and removal of rotary-mode core sampling (RMCS) equipment in single-shell tanks (SSTs) on the Flammable Gas Watch List (FGWL) or those tanks recommended by the contractor to be included on the FGWL, hence referred to as FG/RMCS operations, were identified, and the resulting accidents were grouped into nine categories. In Section 4, each accident category is quantitatively discussed in the following subsections:

- Section 4.1: Aboveground Fire Accidents
- Section 4.2: Dome Fire Accidents
- Section 4.3: Drill String Fire Accidents
- Section 4.4: Waste Fire Accidents
- Section 4.5: Chemical Reactions and Criticality Accidents
- Section 4:6: Containment Breach Accidents
- Section 4.7: Gas Releases without Burn
- Section 4.8: Spills, Releases, and Hazardous Material and Radiation Exposure
- Section 4.9: External Events

Radiological and toxicological consequences of the above accidents are discussed in Section 5 of this safety assessment (SA).

In this section, the frequencies are estimated and discussed for each accident in the corresponding subsection. Both the unmitigated accident frequency (UAF) and the mitigated accident frequency (MAF) for each accident are provided in which credit is taken for the administrative controls established in this safety assessment (SA) and in other safety basis documents. In Appendix D, the equipment reliabilities are computed for each FG/RMCS activity. An FG/RMCS activity is defined to include preinstallation equipment setup, installation of the FG/RMCS equipment, the collection of a complete set of core samples representing an entire tank depth, and removal of the FG/RMCS equipment. One hundred forty-four hours are required to complete an entire activity. Within this 144-h period, 40 h (approximately 2 samples collected per shift) are required to retrieve the 11 samples (based on an average SST waste depth). Total drilling time is approximately 4 h (20 min. per sample). Exhauster and truck shutdown instrumentation systems and calibration

are checked and calibrated every six months. Assuming there are two FG/RMCS activities per year per tank, the accident frequency is calculated per year per tank.

There are several key assumptions that underlie assigning a probability to a given event (Appendix E). The probability of a human error was assumed to be 0.003 based on assumptions defined in NUREG-CR-4772¹ and listed in Appendix E. FG/RMCS operations are required to comply with these assumptions. Therefore, they are listed in Section 6 as administrative controls.

Also considered in determining the accident frequencies is the probability of phenomenological events such as gas-release events (GREs), fraction of the waste causing propagating exothermic reaction, spark generation, and propagation of fire into the dome or into the waste.

The major hazard associated with FG/RMCS operations in flammable-gas tanks is the existence of flammable material. Consequently, four of the nine accident categories identified in the SA are burn scenarios, which generally result in the highest radiological and toxicological consequences because a dome space deflagration in an SST is likely to result in a catastrophic dome failure (see Section 5 of this SA).

To eliminate a fire hazard, either the fuel, the oxidizer, or the ignition source must be eliminated or controlled, and for the safety of FG/RMCS operations, one or more of these factors is controlled under different conditions. Each identified ignition source was analyzed and is discussed in terms of how each is managed by either controls or design safety features. In analyzing the fire risk associated with the proposed FG/RMCS activities, the following multistep approach was used in the following specified order:

- 1. The most important issue was to develop and implement a sparkmanagement strategy that is appropriate for a hazardous flammable-gas environment. In summary, the spark-management strategy provides a minimum of two protective system barriers against spark sources (including mechanical sparks) so that no single failure leads to a sparking condition. The details of the spark management strategy are summarized in Appendix B.
- 2. After implementing the spark-management strategy, the reliability of the equipment used to protect against fire accidents was quantified (Appendix D). Considering the type of operation, failure probabilities on the order of 10⁻⁴ to 10⁻⁵ per activity for the protection systems were used to provide reasonable assurance that all practicable preventive measures would be taken against burn accidents.
- 3. After completing the first two steps, the probability of a GRE was introduced to assess realistic accident frequencies. In this SA, GRE is

defined as any gas release that exceeds the steady-state releases in the SSTs by either volume or rate. The probabilistic model of GREs is discussed in Appendix L.

It is recognized that the order of steps differ from the order one would use to quantify the event trees. Typically, the GRE would be the initiating event and the probability of a burn accident would be the product of the following sequence probabilities:

Burn Probability = (Probability of a GRE causing flammable conditions locally)

- x (Failure probability for the protective system)
- x (Probability of a spark that can ignite the gas mixture)
- x (Probability of flame propagation).

However, in the beginning of the SA process, it was recognized that there is much uncertainty in the magnitudes and probabilities of a GRE for the SSTs as defined in the first term of the above equation, and that design decisions and design controls could not be based on a poorly quantified event probability. Therefore, our initial assumption was that the probability of a large GRE is high, and that flammable gases exist continuously in the areas where potential spark sources are located. Thus, design-changes and design-controls were conservatively developed without taking credit for the GRE probabilities.

System reliabilities are estimated in Appendix D. Table 4-1 gives a summary of the system reliability quantification (see Appendix D for details). Frequency estimates for the initiating event are given for the activity. These frequency estimates are used in event trees for the postulated accidents discussed in this section to determine the accident frequencies. The final accident frequency estimates are also listed in Appendix E.

After completing the assessment of equipment reliabilities, the GRE probability was evaluated. Based on the analysis provided in Appendix L, it is assumed that GREs would occur during the FG/RMCS operations. However, the frequency of having a resulting flammable-gas concentration exceeding the lower flammability limit (LFL) in the dome space with the dome pressure being positive is estimated as 7.0E-5 per activity. The bounding period of the dome concentration being greater than the LFL limit during the FG/RMCS operations was computed to be <0.12 min./activity, assuming that the ventilation system is continuously operational during a GRE. The time period during which the LFL is exceeded (time-at-risk) divided by the 144-hour mission time of FG/RMCS operations per activity is 1.4 x 10⁻⁵, which is the probability of a GRE during FG/RMCS activities for accidents involving a random spark. For certain accidents, the GRE probability based on the exposure period to flammable gas is not adequate. These are accidents in which the spark sources are not random in time. In those accidents, the GRE probability of causing the dome

pressure to be positive is more appropriate. Thus, depending upon the accident scenario, a GRE probability of 7.0E-5 or 1.4E-5 is used.

TABLE 4-1SUMMARY OF SYSTEM RELIABILITY QUANTIFICATION

| Initiating Event Frequency Probability of Subsequent Failures | | Frequency |
|---|---|----------------------|
| (1/activity) | | {1/activity) |
| Excessive force used while taking any sample except last sample, 0.05 per sample, 0.55 total for 11 samples | Failure to detect excessive force and stop drill, 7.9E-5. | 4.3 E-5 |
| Excessive force used while taking last sample .0.05. | Failure to detect excessive force and stop drill, 7.9E-5. | 4.0E-6 |
| Total for excessive force with failure to | stop drill. | 4.7E-5 |
| High H ₂ level in exhauster, 1.0. | Failure to detect hydrogen, 7.6E-4. Failure to detect hydrogen and trip the drill string 1.6E-3 | 7.6E-4 1.6E-3 |
| High H ₂ rate of rise in exhauster, 1.0. | Failure to detect hydrogen, 7.6E-4. Failure to detect hydrogen and trip the drill string 1.6E-3 | 7.6E-4 1.6E-3 |
| High dome pressure caused by H ₂ release, 1.0. | Failure to detect hydrogen, 7.7E-4. Failure to detect hydrogen and trip the drill string, 1.7E-3 | 7.7E-4 1.7E-3 |
| H2 from waste, 1.0. | Failure of drill string N ₂ hydrostatic system, 6.4E-3. | 6.4E-3 |
| H ₂ from waste, 1.0. | Failure of shielded receiver N ₂ hydrostatic system, 6.4E-3. | 6.4E-3 |
| H2 from waste, 1.0. | Failure of both drill string N ₂ hydrostatic system and shielded receiver N ₂ hydrostatic systems 69E-4 | 6.9E-4 |
| Drill string held above waste with foot clamp, 1.0. | Foot clamp drops drill string onto waste surface, 3.0E-5. | 3.0E-5 |
| Excessive rpm during drilling , 0.011. | Failure to trip drill string on excessive rpm, 6.2E-4. | 6.8E-6 |
| Perform drilling operation, 1.0. | Total loss of N_2 cooling and failure to stop drill, 3.3E-6. | 3.3E-6 |
| Perform drilling operation, 1.0. | Loss of drill bit N ₂ cooling from N ₂ bypass leakage, 1.6E-5. | 1.6E-5 |
| Total for overheating drill bit caused by | inadequate N ₂ cooling. | 1.9E-5 |
| Take a sample, 11.0. | Failure of rotary valve in sampler to completely close. 0.1. | 1.1 |
| Perform drilling operation, 1.0. | Failure of sampler chevron seal. | 3.3E-2 |
| H ₂ from Waste, 1.0. | Failure of H_2 sniff in drill string, 4.6E-3 (3.0E-3 operator, 1.6E-3 hardware). | 4.6E-3 |
| Excessive filter DP , 1.0. | Fail to trip exhauster on filter DP, 2.6E-3. | 2.6E-3 |
| Contact rock in waste, 0.1. | Failure of penetration rate and failure of N_2 cooling to drill bit systems, 3.3E-6. | 3.3E-7 |

August 8, 1996

97. 97. The third term in the burn probability equation is the probability of a spark, which is also not easy to evaluate for a general case. It was assumed that the probability of a spark is high, given that flammable conditions exist near nonqualified electrical equipment. Sparks caused by materials with potential electrostatic discharges are also assumed to occur with a probability of 1.0 when the environment is flammable. Ignition as a result of sparks caused by mechanical impacts and drops are, in general, evaluated by performing mechanical ignition tests.

Propagation of a fire from the tank top into the tank dome is assumed to be likely for certain accidents (especially for gas releases through openings at the top of the tank). The velocity of the released gas could be small enough to allow fire propagation to occur.

In the following subsections, the accidents of concern are quantitatively discussed. In each subsection, the accident, its causes (dominant failures), and mitigation or credited assumptions or controls are defined in evaluating accident frequencies. The accident frequency based on credited assumptions or controls is defined as the MAF. The UAF does not consider the effects of credited assumptions or controls listed in Tables given in the rest of this section. Appendices D and E provide the details of how the UAFs and MAFs are obtained. Appendix E provides two sets of event trees; one for mitigated accidents and one for unmitigated accidents. The mitigated and unmitigated accident frequencies are given in the summary tables and used primarily to identify the level of the controls.

4.1. ABOVEGROUND FIRE ACCIDENTS

Accidents in this category include all fire initiators on the top of the tank for which FG/RMCS activities are performed. For this accident scenario, it is postulated that waste-intrusive FG/RMCS operations may cause a large GRE resulting in a flammable-gas environment above the tank through possible leak paths. If a fire occurs at the top of the tank, it could propagate into the tank dome. The fire propagation into the tank may result in structural damage to the tank and in significant material releases.

4.1.1. Flammable-Gas Release to Exhauster and Burn (Ignition in the Exhauster, Electrostatic and Electrical/Mechanical, and Other Spark Sources, Operation, and Removal)

Preinstallation and installation phases do not include waste-intrusive activities. Therefore, a gas-release event is not expected in single-shell tanks during these phases. Westinghouse Hanford Company (WHC) standard controls require verification that the tank dome does not have flammable gas before starting the installation phase. The preinstallation phase also includes testing of critical alarms. The exhauster and its safety system, as briefly discussed in Section 2, are required to be operational during FG/RMCS operations, including drilling and sample retrieval activities. Drilling starts by turning the nitrogen purge flow to the drill bit at a rate > 30 scfm. Any time the nitrogen purge flow is terminated, a 16-hour, post-drilling exhauster operation is required. During this period, sample retrieval, or other FG/RMCS-related activity could occur. If drilling is resumed during this period, the 16-h requirement must be reestablished after termination of the nitrogen purge flow. The 16-h requirement corresponds to the time in which the equivalent of 4 dome-volumes is circulated at a minimum nominal flow rate of 200 scfm, primarily to remove aerosols and flammable-gas accumulations created during the drilling. To completely circulate the equivalent of one dome volume, the exhauster needs to operate approximately 4 hours. It is expected that the 16-h post-drilling exhauster operation will remove more than 95% of the accumulated aerosol and flammable gases.

The exhauster must be turned on at least one hour before drilling begins in order to

- Mix any potential flammable-gas pockets,
- Reduce the flammable and toxic-gas concentrations,
- Obtain flammable-gas concentration and pressure data, and
- Perform flammable-gas meter verifications as necessary.

In the current system design, when the flammable-gas detection system detects concentrations higher than those specified in this SA, exhauster operation is continued without interruption while the drilling operations are automatically terminated through a separate interlock. The exhauster will be operational during and after a GRE until the flammable concentration falls below acceptable levels.

There are numerous ways to initiate a fire in the exhauster. The possible ignition sources in the exhauster are as follows:

- 1. Static electricity from the flexible duct or other parts of the exhauster,
- 2. A lightning strike to the exhauster,
- 3. An electrical spark from nonqualified electrical equipment,
- 4. Mechanical frictional sparks from crane load or other heavy equipment /tool drops on the exhauster (note that in order to damage the duct it is not necessary to have drop accidents involving cranes; any other equipment drops can cause damage to the duct), and
- 5. Mechanical frictional sparks from the fan and housing contact.

The first four causes are independent of whether the ventilation system is active or not.

4.1.1.1. Static Electricity

Static electricity in the flexible exhauster duct is managed by grounding the conductive duct and exhauster to the tank using WHC standard controls for grounding and bonding. The flexible duct is 1/32 in. thick and made of neoprene over a polyester base, and is conductive from the inner to the outer layer.² Vendor information indicates the conductivity of the duct material is 100,000 ohms per square inch, which is within the standards issued by the Institute of Electrical Electronics Engineers (IEEE) 142-1991,³ paragraph 3.2.6.2, which states that a resistance of 1.0E6 ohm is adequate for static grounding. Robinson⁴ measured the resistivity of the conductive duct (across the 10-in. diameter of the duct). In all samples the resistivity was measured between 500 and 900 ohms. These values are much lower than 1.0E6 ohm/in² required to prevent static build up. Therefore, the conductive duct meets the requirements of IEEE standards and is not expected to cause a static electricity discharge.

Failure to bond/ground the exhauster duct is unlikely because the duct is connected at both ends to the metal flanges of grounded and bonded components. Mechanical failure of grounding also is considered unlikely. The procedural steps requiring a physical inspection of the ground system before the operation help prevent this failure. Procedural steps requiring the physical inspection of the resistance between the exhauster and the tank are performed by WHC as a standard requirement.

Materials such as plastic bags must be carefully controlled because of the potential for static sparking.

Based on these considerations, static electricity concerns associated with the exhauster flexible duct are considered to be properly managed and will not cause a burn accident.

4.1.1.2. Lightning

Details of a lightning strike and a burn accident initiated by lightning strikes are provided in Section 4.9.1.

4.1.1.3. Electrical Spark from Nonqualified Electrical Equipment

Appendix B-concludes that the tank dome space must be treated as a Class-I, Div.-1, Group-B environment during active waste-intrusive FG/RMCS operations. The FG/RMCS exhauster flow path has a direct path to the tank and therefore is also treated as a Class-I, Div.-1, Group-B environment.

Emission control requirements for the exhauster are not discussed in this SA, but based on the information provided by WHC, Washington State permits exhauster operation without stack monitoring.⁵

All of the electrical equipment in the exhauster air stream is either intrinsically safe or deenergized. The fan motor is outside the exhauster duct. The shaft between the motor and exhauster fan is purged with nitrogen. The measurement devices for exhauster flow and the differential pressure across the high-efficiency particulate air (HEPA) filter unit are electrical components that have been made intrinsically safe. The exhauster has a heater to prevent condensation in the HEPA filter. The heater was originally an electric heater but hat been replaced with a heat exchanger driven with hot water supplied by a heater locted in an unclassified area.

Based on these design features, it is believed that an electrical spark in the exhauster is not considered as a credible initiator of a burn accident.

4.1.1.4. Mechanical Sparks

One credible mechanical, frictional spark source can result when heavy equipment or tools are dropped on the exhauster or flexible duct. Generally, WHC has a standard practice of not transferring heavy equipment and tools over the exhauster or other equipment critical to safety. Administrative controls prohibiting the transport of equipment over the exhauster are established in the SA to manage this source of frictional sparking.

The frequency of a fire accident as a result of dropping equipment on the exhauster during operation is determined as 9.5E-12/yr in Appendix E, and that value is very low. The accident is caused by failure to terminate the lifting operation, given a high tank dome pressure, hydrogen concentration, or a high rate of hydrogen concentration increase, failure to observe lift-over-tank control, and a crane load drop and exhauster impact.

Another mechanical spark source is the exhauster fan that is built to the Air Movement Division of the Air Movement and Control Association (AMCA) Standard 99-0401-86.⁶ The standard requires the following:

> All parts of the fan in contact with the air or gas being handled should be made of nonferrous material. The hole where the shaft passes through must be made so that ferrous materials could not rub. Steps must be taken to ensure that the impeller, bearings, and shaft are adequately attached and/or restrained to prevent lateral or axial shift in these components. The fan must be so constructed that a shift of the shaft or impeller must not allow two ferrous parts to rub. No bearings, drive components, or electrical devices must be placed in the air or gas stream unless they are constructed or enclosed in such a manner that failure of that component cannot ignite the gas. However, the customer must accept both the type and design with full recognition of the potential hazard and the degree of protection required.

The WHC fire protection engineer has determined that the construction requirements for AMCA Type A adequately address the spark-resistant criteria in National Fire Protection Association (NFPA) 91. However, there is no requirement to demonstrate that if the fan and housing come into contact because of bearing failure, the contact is not capable of igniting a potential flammable atmosphere. For the frequency of this event, it is assumed that if a bearing fails 10% of the time, it

results in contact between the fan and the housing, with a conditional frequency of 6.4E-10/yr.

However, this event is not considered a credible source of ignition because WHC performed an assessment study, including a literature survey, on the possibility of generating sparks as a result of aluminum fan-to-aluminum housing impact.⁷ Based on the findings of this study, the following recommendations are made:

- Fan material should be made of aluminum alloys containing less than 0.8% Mg,
- The housing or any place where impact of the fan is possible should not include iron alloys,
- All surfaces should be aluminum, and
- Any lubricant that may cause sparking should not be used on surfaces where impact is possible.

The exhauster fan meets all the conditions recommended above. In addition, Appendix P investigates the possibility of a hot spot for rubbing aluminum surfaces and found no credible evidence to conclude that if the fan mechanically fails and impacts the housing, sparks capable of igniting hydrogen-air and hydrogen-nitrous oxide mixtures could be created. Although the vendor did not have data to confirm this conclusion, they confirmed that there have been no accident reports indicating this type of failure. Consequently, it was concluded that the failure of the fan is not a credible initiator for a burn accident, and the frequency is designated as $<< 10^{-6}/yr$.

4.1.1.5. Summary of Exhauster Fires

Table 4-2 summarizes the above discussions. The bounding accident frequency for exhauster fires is 1.1E-10/yr caused by lightning strikes. For bounding consequences, it was conservatively assumed that a fire initiated in the exhauster would propagate into the dome, resulting in a dome collapse accident, as discussed in Section 5 of this SA.

4.1.2. Flammable-Gas Release Through Torn Duct and Burn (Operation)

This event postulates a GRE occurring at a high rate, creating a positive tank pressure, which results in an ignition of flammable gases that have escaped through the torn exhauster duct. The ignition source for this scenario is potentially nonqualified electrical equipment at the exhauster skid and around the exhauster duct.

If the exhauster duct has torn and a GRE occurs, some region around the duct must be assumed as a Class-I, Div.-2 environment, based on the diameter of the leak. The undetected leak diameter is assumed to be ≤ 1 in. Based on the criteria given in

Appendix B, nonqualified equipment with no automatic shut-down features should be 18 in. away from the leak. Therefore, an exclusion zone should be established 18 in. from each side of the exhauster duct, and no equipment with sparking potential should be placed within this exclusion zone. This control protects the tank from fire initiated through a torn duct. All electrical equipment currently on the skid not meeting Class-I, Div.-1 requirements is protected by enclosures, and most is 18 in. away from the point where the exhauster duct is attached to the metal flange near the heater.

| Accident | Frequency of Failure 1/yr | Dominant Failures | Controls Credited in Calculating UAF and MAF Values |
|--|---------------------------------|--|---|
| Lightning strike causes dome fire. | MAF and UAF << 1.0E-06 | Random lighting strike hits risers or other equipment on top of tank that connects to tank interior. | Do not drill if a lightning strike is observed within a 50-mile radius. |
| Drop of equipment from crane onto exhauster leads to dome fire. | MAF and UAF << 1.0E-06 | Equipment is dropped from crane onto exhauster. | No equipment lifts over exhauster. |
| Ignition of flammable gas in the exhauster is caused by fan failure. | MAF and UAF << 1.0E-06 | Failure of fan bearings results in fan housing mechanical sparking. | No controls are credited. |

TABLE 4-2 SUMMARY OF FIRE ACCIDENTS IN EXHAUSTER

Even though the flexible duct is heavy and difficult to tear,² the following requirements are established to prevent duct damage. The flexible duct must be made stationary before operations start to prevent any motion that may result in damage to the duct when there is a strong wind. A control requires the termination of FG/RMCS operations when the sustained wind velocity is greater than 25 mph because of concerns about possible structural failure of the flexible duct. A control to inspect the flexible duct for possible leaks before operations begin also is established, as well as the control to preventing the transport of any equipment or tools over the exhauster or duct.

The MAF is calculated as 1.4E-9/yr, and the UAF is 5.6E-8/yr, as discussed in Appendix E. Note that no credit was taken for the probability of the propagation of fire into the dome. Dominant failures are summarized in Table 4-3. The consequence of this accident is conservatively considered to be the dome collapse discussed in Section 5 of this SA.

| TABLE 4-3 |
|---|
| ABOVE-TANK FIRE ACCIDENTS CAUSED BY TORN EXHAUSTER DUCT |

| Accident | Frequency of Failure 1/yr | Dominant Failures | Controls Credited in Calculating UAF and MAF Values |
|--|---------------------------------|--|---|
| Tear in the exhauster duct exposes flammable gas to nonqualified electrical equipment leading to dome fire. | MAF = 1.4E-9 UAF = 5.6E-8 | Big tear occurs in the duct during operation. Tear in the duct is not detected. | Inspect the duct for tears before operation. Locate unqualified equipment at a distance >18 in. from exhauster duct, or provide deflectors/enclosures for equipment located within 18 in. |

4.1.2.1. Flammable-Gas Release through Open Riser (or Possible Leak Paths) Driven by Gas Release Event and Burn (Operation and Removal)

This event postulates that a large GRE occurs, releasing flammable gas from openings in the tank, exposing the flammable gas to equipment with possible spark sources, resulting in a fire on top of the tank. This event is of concern during operation and removal phases of FG/RMCS. None of the auxiliary support equipment on top of the tank is qualified to operate in a flammable-gas environment.

Preventing flammable-gas exposure to this equipment is managed in several ways. Open paths from the tank dome to the tank top include the exhauster stack, the breather inlet riser, open risers, drill string riser, and other possible tank leak sources (unsealed risers, pits, etc.). It is assumed that inspection of tank top penetrations for potential leak paths will find leak paths with a 1-in. effective leak diameter. Therefore, it is assumed that undetected leak paths with a 1-in. effective leak diameter could exist. Therefore, the top of the tank must be examined to identify leaks other than risers used for passive/active ventilation, and when identified, leaks greater than or equal to 1-in. effective leak diameter must be sealed.

A portable stack over the breather inlet HEPA system will be used during wasteintrusive FG/RMCS operations. The portable stack is at least 15-feet tall, has an upper 4-in. diameter, is sealed at the ground level, and is grounded. The purpose of using a portable stack over the breather inlet is two-fold:

• The gas release would be released through the stack, resulting in increased atmospheric dilution and reducing the toxicological consequences of a GRE.

• Any nonqualified equipment on the top of the tank around the breather inlet HEPA system would be protected from flammable-gas exposures.

The positioning of the drill truck, the X-ray machine, the light plant or other auxiliary equipment that could be a spark source is based on the criteria given in Appendix B. It is required that any nonqualified electrical equipment must be placed at least 36 diameters away from an open riser during waste-intrusive operations. If a GRE occurs, immediate personnel evacuation is required.

Another control also requires that when the drill truck needs to be parked over an unused, closed riser or pit, the riser or pit must be sealed. Note that the riser or pit considered here is not the riser being sampled but the one that the front part of the truck is parked on. If the truck is parked over an unused riser or pit, the potential spark location is not considered random, and no credit can be taken for the probability of a random placement. However, the leak size from pit or riser is assumed to be no bigger than 1 in. There are at least 3 feet between the top of the pit/riser and any potential ignition sources on the truck. A control was established to make sure the distance between potential ignition sources on the truck and the top of the pit/riser is \geq 36 in. Combining the failure probability to seal the riser/pit and violate the 36 in. distance criteria and the GRE probability that makes the dome pressure positive, the accident frequency is determined as 2.1E-8/yr. This frequency is low. However, the unmitigated frequency becomes 1.4E-4/yr if the control to seal the riser/pit and 36 in. distance criteria are not implemented.

The flammable-gas release could occur from other unused risers if they have undetected leak paths. The control requiring the examination of risers before operations reduces the probability of having an unknown open path. It is assumed that leaks from threaded junctions, flanges, and cover plates could be identified with an effective leak diameter greater than 1 in. If a GRE occurs and nonqualified equipment is located close to these unknown openings, the accident frequency of an above-tank fire becomes 1.8E-7/yr. This frequency includes the probability of a GRE based on exposure time (Appendix L) and the probability of a temporal random spark. It also assumes that 50% of all risers on the top the tank leak after the initial inspection is performed. The existence of a spark on the equipment located around risers is also assumed. The unmitigated accident frequency is 2.8E-5/yr for this accident scenario; therefore, the control requiring that leaks be limited before the FG/RMCS operation is important.

The last accident scenario includes the ignition of a flammable-gas release from an intentionally opened riser during FG/RMCS activities. This may be needed for other daily activities. Appendix E gives the event tree for a fire accident. A control is required not to place any nonqualified equipment within 36 diameters of the riser being opened during waste-intrusive operations. Considering this control and a GRE probability based on exposure time including a temporal random spark, the mitigated accident frequency becomes 4.2E-8/yr. This number is conservative

because it was not considered that the riser may be open for only a fraction of the mission time. The unmitigated accident frequency is 1.4E-5/yr.

The combined mitigated frequency of these three events is 2.4E-7/yr. Note that the fire propagation into the dome conservatively is assumed to be 1.0. This number is conservative because it is assumed that nonqualified equipment does include a spark source and the probability of a random spark in time is based on a conservative dome concentration.

Positioning the drilling truck needs special attention. The closest release path to the truck is the drilling riser that can momentarily be open to the environment during installation and removal of FG/RMCS equipment, even though the riser is sealed with a rubber frisbee during operation. However, because some drill rods are fluted, it is likely that the frisbee can be damaged, and a leak can result. A nitrogen purge is provided in the annulus between the drill string and the conductive riser sleeve. This purge flow is designed to provide 5 standard cubic feet per minute (scfm) flow for all postulated tank dome conditions.⁸ The purge system has redundant differential-pressure alarms across the flow controller. Necessary controls for this system will be discussed in the dome fires accident category (ignition in the riser).

For a gas release to occur from the riser path, the nitrogen purge would have to fail, and a GRE would have to occur. However, even if a 1-in. effective leak diameter path exists through the frisbee, potential ignition sources on the FG/RMCS truck are at least 3 ft away. There is a direct path from the frisbee to the top of the platform when the shielded receiver or the drill rig are connected to the drill string. However, the spark-generating electrical motors are located at the top of these components and meet the 36-equivalent-leak-diameter distance requirements of Appendix B. This distance is acceptable because the rotating platform also acts as a jet deflector. The nitrogen instrumentation and piping enclosure have solenoid valves, but they are in an enclosure (not leakproof but reasonably sealed) and protected from direct flammable-gas jet impingement. The major spark contributor is the drill engine and it is at least 3 ft away from the frisbee. In summary, if a leak occurs from the frisbee, the gas would not reach the spark sources on the truck with a flammable concentration.

To further reduce the likelihood of a gas release from the drill string during removal, the drill string must not be removed from the waste without a sampler or a dummy sampler in the string. The drill string and frisbee hole must comply with the 36 equivalent-leak-diameter distance requirements during waste-intrusive activities.

The summary of this accident is given in Table 4-4. With the above controls in place, it is believed that the ignition from the truck or any other nonqualified equipment is adequately controlled as demonstrated by the low magnitude of the MAF.

4.1.2.2. Flammable-Gas Release through Shielded Receiver (SR) and Ignition in the Shielded Receiver (Operation)

There are two cases for this accident, depending on the operating condition:

- When the shielded receiver (SR) is connected to the drill string during sampler recovery, and
- When the SR is isolated from the drill string while the sampler is being transferred to the X-ray machine or the shielded cask.

This section treats the cases where the SR is isolated from the drill string. The other case is examined in Section 4.3.2 along with drill string fires. Flammable gas may accumulate in the SR either by a gas transfer from the drill string or flammable gas may be released from the sampler into the SR.

TABLE 4-4

ABOVE-TANK FIRE ACCIDENTS CAUSED BY GAS RELEASES FROM OPEN RISERS

| Accident | Frequency of Failure 1/yr | Dominant Failures | Controls Credited in Calculating UAF and MAF Values |
|---|---------------------------------|--|--|
| Release of flammable gas from riser/pit that the drilling truck is parked on leading to dome fire. | MAF = 2.1E-8 UAF = 1.4E-4 | Fail to seal to riser or pump pit. Fail to locate potential ignition sources on the FG/RMCS truck within 36 in. of the riser/pit. | Seal the riser or pump pit being parked on. Distance between the potential ignition sources on the FG/RMCS truck and the riser/pit is greater than or equal to 36 in. or provide enclosures/deflectors for equipment loceted within 36 in. |
| Releases of flammable gas to unqualified electrical equipment from unknown leaks leading to dome fires | MAF = 1.8E-7 UAF = 2.8E-5 | Randomly located unqualified equipment is located too close to riser/pit not in use. | Limit leakage from all unused risers/pits to less than 1 ineffective leak diameter. |
| Releases of flammable gas to unqualified electrical equipment from an intentionally open riser during FG/RMCS waste- intrusive activities leading to dome fire | MAF = 4.2E-8 UAF = 1.4E-5 | Equipment located too close to open riser/pit. Randomly located unqualified equipment is located too close to open riser/pit. | Restrict location of equipment to greater than 36 diameters from open risers/pits <u>or provide</u> <u>enclosures/deflectors for</u> <u>equipment located within</u> 36 in. Inlet stack 15 feet tall is installed on HEPA inlet riser. |

For gases to be released into the SR from the drill string, the hydrostatic head must fail. Calculations provided in Appendix J show that as long as the hydrostatic head purge of 0.3 sfcm exists, the flammable gas in the drill string does not diffuse back upstream. There are two sources of hydrostatic head purge while the SR is connected to the drill string. One purge is connected to the SR, and another is connected to the change-out assembly. Considering the operating procedures, the analysis shows that failure of both purges and not detecting this failure combined with a frequency of dropping the core sampler, which is the only credible spark source in the SR, results in an ignition frequency of 1.9E-6/yr.

The consequences are small when the SR is isolated from the tank, and the quantity of gas in the SR also is small.

Besides being transported from the drill string, the flammable gas can accumulate in the shielded receiver in two other ways as follows:

- Waste accumulation in the SR resulting in a gas accumulation; and
- A gas release from the sampler.

The maximum waste release into the SR is the equivalent of one full sampler (0.39 kg). Gas that could be retained in this amount of waste is small. If the sampler is full of gas at approximately 2 atm, the maximum gas volume at atmospheric pressure would be 611 cm^3 . The volume of the receiver is $120,000 \text{ cm}^3$. Thus, the resulting flammable-gas concentration would be less than the lower flammability limit (LFL), and ignition would not be possible. There is no incompatible material inside the SR, so that additional gas generation caused by chemical reaction is not a concern.

The SR has a load cell to measure the weight on the cable. This load cell is protected by an intrinsic safety barrier. Therefore, an electrical spark inside the SR is not a concern.

The RLU is based on a mechanical design. A frictional spark caused by RLU motion may be considered as a spark source. The electrical motors controlling RLU use are direct current (dc) pulse width modulated power. The hoist motor for the RLU has a 150:1 worm-gear reduction, and because of the gearbox, the RLU does not freewheel if the motor fails. The normal hoist velocity is less than 1 ft/s. Studies have been done to examine the characteristics of mechanically-produced sparks that lead to the ignition of tank-like gases. Krok and Shepherd⁹ carried out frictional spark ignition studies of H₂/air and H₂/N₂/N₂O/air mixtures in which they used rusted steel and aluminum plates impacted by steel or aluminum bars. A pneumatically-actuated piston made of steel or aluminum was stroked on an inclined, rusted-steel plate. The impact velocity varied from 0.5 to 2.5 m/s (1.64 to 8.2 ft/s). These experiments showed that the frictional ignition in a mixture of 10% 日本語が見

or 15% H_2 in air at 1 atm was very unlikely. Based on this data, a frictional spark caused by the normal RLU motion is not expected.

Raction exposures were calculated in Appendix R for possible doses if a full sam_{E}^{2} were to be released in the SR. Based on these calculations, measurable quantities of waste would be recorded by manual radiation readings. WHC has radiation controls performed by the health and safety personnel, but an additional radiation control limits the SR dose rate to 100 mrem/h on contact. This radiation monitoring must be performed once per shift during waste-intrusive activities to prevent waste and flammable-gas accumulation in the SR.

One other possibility of frictional spark occurrence is an equipment drop in the shielded receiver. The RLU may be dropped because of a mechanical failure, or the sampler may drop from the RLU. In order for the RLU to drop, the core sampler should be stuck and the operator should fail to respond. The hoist cable is designed to fail at loads higher than 2000 lb. The hoist motor fails at about 800 to 1300 lb. Thus, the cable cannot fail before the electrical motor fails. Because of the gearbox, the RLU does not freewheel if the motor fails. It is concluded that an RLU drop is not a credible event. In addition, dropping a 5.2-lb stainless-steel piece from a 44-in. height to simulate the quill-rod-to-carbon-steel impact showed no ignition in a bounding flammable gas environment (see Appendix T and Witwer¹⁰). Note that the weight of the RLU is higher than 5.2 lb but the drop height in the shielded receiver is much smaller than 44 in. The core sampler drop can be bounded by the ignition tests performed on quill-rod adapter to drill string impact because its weight is comparable and the drop height is small. Because BOM tests showed no ignition (Witwer¹⁰), all accidents discussed in this section are considered to be not credible. The drill string is sniffed before it is connected to the shielded receiver. The sampler rotary valve needs to fail in order to cause a gas release in the shielded receiver. Considering these failures, this accident is assumed to be incredible, as shown in Table 4-5.

The core sampler may drop from the RLU. The conditional MAF is estimated as 2.8E-5/yr in Appendix E and given in Table 4.5. Ignition caused by dropping the core sampler in the shielded receiver was investigated by tests conducted by the Bureau of Mines (BOM). Appendix T describes the test requirements and Witwer (Ref. 10) summarizes the results. Tests are performed by dropping the core sampler in the drill string from a 60-ft height. No ignition is observed in a stoichiometric hydrogen-air mixture. The drop height in the shielded receiver is much smaller than 60 ft. Therefore, this accident is considered to be not credible.

If ignition occurs in the SR with low concentrations, the burn pressure, 1.2 atm, does not exceed failure pressure.¹¹ The SR maximum design pressure is 52 psia (Ref. 12). The ball valve isolating the SR fails at very high pressures (1100 psia). Therefore, no failure and release is expected if ignition occurs in the SR.

| TABLE 4-5 |
|--|
| ABOVE-TANK FIRE ACCIDENTS IN SHIELDED RECEIVER |

| | · · · · · · · · · · · · · · · · · · · | Y ¹¹⁻¹ | |
|---|--|---|---|
| Accident | Frequency of Failure 1/yr | Dominant Failures | Controls Credited in Calculating UAF and MAF Values |
| Flammable gas in drill string (DS)/SR during sampler handling leading to fire in SR (core sampler drop). (Fire does not propagate back to the dome.) | Without BOM test results: MAF = 1.9E-6 UAF = 2.8E-3 With BOM test results: MAF and UAF <<1.0E-6 | Nitrogen purge to both drill string and to SR fails. RLU drops sampler. | Leak test of N ₂ : hydrostatic systems is done for both drill-string and for shielded receiver. Unique connections for N ₂ : purge systems for both drill string and SR are used (design feature). Verification of N ₂ purge: supply to both drill string and SR during activation of hydrostatic mode of N ₂ supply is carried out. Controls over operation of RLU are used. |
| Flammable gas in SR with SR isolated from drill-string results in aboveground fire (RLU or core sampler drop). (Fire does not propagate back to the dome.) | Without BOM test results: MAF = 2.8E-5 UAF = 3.0E-4 With BOM test results: MAF and UAF <<1.0E-6 | Waste prevents closure of sampler rotary valve leading to gas release as sample is retrieved and depressurized. Flammable gas is not sniffed or detected. RLU drops sampler. | Controls over operation of RLU are used. |

A control has also been established for a visual inspection of the sampler through the sight glass as it is withdrawn from the drill string. This visual inspection does not directly indicate the failure of the sampler rotary valve, but it may indicate leaking from either the rotary valve or rotary valve seal. An administrative control shall be developed for handling a leaking sampler. An additional control prohibits putting a known leaking sampler into the x-ray machine.

One other concern is the release of gas from the SR. For this accident to occur, the manually operated SR ball valve would have to be opened before it is connected to the x-ray machine, which would be a violation of the procedures. The engine has several potential ignition sources. The likelihood of this event is low because several failures have to occur; inadvertent opening of the ball valve, failure of the rotary ball valve, and dropping the sampler at the same time. In addition, the operator must fail to identify the leak by visual inspection. Besides, the flammable

August 8, 1996

gas may not come from the bottom of the pipe but may accumulate at the top of the horizontal section of the shielded receiver. Even if flammable gas comes from the valve, the maximum amount of flammable gas that can be accumulated in the receiver is limited to 611 cm³. It is expected that the consequence of the accident is insignificant because of the limited amount of flammable gas present and because there is no entrained waste. The fire cannot propagate back to the tank because the shielded receiver is not connected to the drill string. The only concern in terms of material release would be dropping the waste on the ground when the valve is opened. This accident is discussed along with the spill accidents in Section 4.8.

4.1.3. Flammable Gas and Ignition in X-ray or Cask

If the flammable gas were accumulated in the SR, it could be discharged into the x-ray machine or into the shielded cask when the SR is connected to these components. The concern with this scenario is a fire in the cask or x-ray machine with a local release of radioactive material and possible propagation of fire into the tank dome.

If the flammable gas in the SR is discharged into the cask or the x-ray machine, ignition is not expected. No unqualified electrical equipment in a Class-I, Div.-1 or Class-I, Div.-2 space exists in the x-ray machine.¹³ The cask does not contain a spark source. The only ignition source identified is dropping of the sampler into the plastic x-ray container or the cask. The x-ray container may not produce frictional sparks but rather static electric charges. However, the plastic container is grounded and bonded by a graphite paint as a coating surrounding the interior of the plastic container.

Appendix E examines the probability of ignition in these components. The accident frequencies are found as 1.6E-9/yr and 2.8E-5/yr for the x-ray and the cask, respectively, as indicated in Table 4-6. The x-ray and cask are grounded and bonded through the SR, and the sampler is inspected for external contamination. Even if the ignition occurs, the available flammable gas that can come from the SR or sampler is limited so that the flame would not propagate to the tank. For propagation to the tank to occur, there would have to be hydrogen in the dome and a flammable pathway, which does not exist. Under these conditions, operations would not be conducted.

4.1.4. Flammable Gas Accumulation or Release Caused by the Loss of Electrical Power and Ignition Source (Installation, Operation, and Removal)

In case of loss of electrical power, the FG/RMCS truck is shut down. When the electrical power is lost, the nitrogen shutoff valve closes. If the sampler is being removed or installed, the hydrostatic head purge that keeps the waste from entering the drill string is lost, allowing the drill string to flood. Flammable-gas accumulation in the dome, drill string, drill unit, and shielded receiver is of concern in this scenario. Waste can flood the drill string and release flammable gas into the

drill string and the SR. Appendix J shows that the diffusion of hydrogen is relatively fast. Flammable gases also could be released as a result of depressurization of the waste.

The management of ignition sources in these components has been discussed previously. No attempt was made to evaluate the failure probabilities of the loss-ofpower accident. An administrative control must be developed for the startup after loss of power considering the possibility of the presence of flammable gas. A control requiring the purging of the shielded receiver, drill string, and drill unit long enough to evacuate the possible flammable gas in the drill string is established. If the drill string is flooded, the SA requires that the drill string be washed before the operation is restarted.

4.1.5. Propane Release from Refueling of Nitrogen Trailer Propane Tank and Fire (Installation, Operation, and Removal)

This accident scenario involves a propane spill during the refueling of the propane tank. The main concern is the propagation of fire to the tank being sampled or to another tank in the farm. The propane tank is kept outside of the tank farm. Drilling on the tank being sampled is prohibited during refueling. This is a typical industrial accident and is not discussed in detail. WHC standard controls already exist to prevent this type of accident.

4.1.6. Flammable Diesel and Gasoline Fuel Fire (Installation, Operation, and Removal)

This accident scenario involves a spill of diesel, gasoline, or hydraulic fluid during the refilling of the FG/RMCS equipment. The main concern is the propagation of fire to the tank because the refueling occurs when the truck is already over the tank dome.

The SA requires that all engines or motors on affected equipment be shut down during refueling. Exhaust pipes on affected equipment must be cool. No drilling operations, open risers, or open drill string or non-RMCS activities on affected equipment are allowed on the tank being sampled during refueling. A restricted smoking area is required, and all possible ignition sources need to be kept outside of the refueling area. These controls are established to prevent fire propagation to the tank. In addition, proper fire extinguisher equipment in the vicinity of the drilling truck must be available in case of a local fire.

4.1.7. Collision Caused by Trucks and Other Equipment (Installation, Operation, and Removal)

This accident involves frictional sparks created by the collision of trucks or other equipment with the riser. The drill truck and other equipment must be operated safely on top of the tank. In the unlikely event that the truck impacts an open riser, all operations must be stopped, and possible gasoline or diesel leakage into the tank

must be evaluated. The introduction of flammable material into a flammable-gas tank may have serious consequences. Thus, the Tank Farm Operations management must be notified, and approval must be granted before operations may resume.

| Accident | Frequency of Failure 1/yr | Dominant Failures | Controls Credited in Calculating UAF and MAF Values |
|--|---|---|--|
| Flammable gas in x- ray machine leads to aboveground fire. (Propagation into dome not possible.) | No fire propagation into the dome: MAF = 1.6E-9 UAF = 1.6E-8 Fire propagation into the dome: MAF and UAF << 1.0E-6 | Waste prevented previous closure of the sampler rotary valve leading to gas release as sample is retrieved and depressurized. Previous sniff in shielded receiver failed to detect flammable gas. RLU drops sampler. Plastic receiver breaks. Isolation barrier volume fails. | Use sealed plastic sampler receiver surrounded by isolation barriers. Use controls over operation of RLU. |
| Flammable gas in cask leads to aboveground fire. (Propagation into dome not possible.) | No fire propagation into the dome: MAF = 2.8E-5 UAF = 2.8E-4 Fire propagation into the dome: MAF and UAF << 1.0E-6 | Waste prevented previous closure of sampler rotary valve leading to gas release as sample is retrieved and depressurized. Operator fails to perform sniff for flammable gas. RLU drops sampler. | Use controls over operation of RLU. |

TABLE 4-6 ABOVE-TANK FIRE ACCIDENTS IN X-RAY MACHINE AND CASK

Before installation, standard WHC controls are implemented to prevent an unexpected flammable-gas release from the tank dome. Frequent and large naturalgas release events are not expected in single-shell flammable-gas tanks. The tank vapor space must be sampled before installation, and the flammable gas must be less than 25% of the LFL. If a GRE were to occur, operations would be stopped.

A possible scenario is the flammable-gas accumulation in the dome and failure to detect it before FG/RMCS tank-intrusive activities. If there is flammable gas in the riser, a fire can be initiated by a frictional spark from a collision. During the

removal phase, the tank dome and drill string must be sniffed before drill string removal and other riser activities. If flammable gas is detected above limits, all operations must be stopped, including drilling. For this accident to occur, the detector systems would have to fail simultaneously with a collision. This event is not considered credible.

Without fire, the consequence of collision is damage to the tank riser. The tank liner is not connected to the riser or to the dome. Thus, the damage would be to the riser, which can be sealed if this accident occurs. Standard WHC safe practice requirements are required to prevent the collision. The likelihood of serious consequences of this type of accident are not considered credible.

4.1.8. Summary of Aboveground Fire Accident Category

The total conditional accident frequency for the aboveground fires that lead to a dome fire is 2.4E-7/yr. The dominant contributor to this frequency is gas release to nonqualified equipment. Randomly located unqualified equipment too close to risers that have an open path results in the highest accident frequency of 1.8E-7/yr. The consequence of an above-tank fire is treated as a dome collapse. Dome collapse consequences are discussed in Appendix I. The total frequency for aboveground fire accidents, 2.4E-7/yr, is added to the other three frequencies considered in the next section to determine the frequency of dome collapse.

4.2. GAS RELEASE AND DOME FIRE ACCIDENTS

The postulated accidents in this category consider fires initiated in the dome space of the tank. Most of the cases discussed in this section can occur during the operation phase of the drilling. The flammable gas in the dome is the first necessary condition for a fire scenario to occur. The gas can be released when the drill bit penetrates the crust or waste sludge.

Below, each identified spark source in the dome is examined closely, and the associated control systems are discussed.

4.2.1. Ignition of Flammable Gas as a Result of a Drill String Break (Operation)

This accident is caused by a mechanical spark created by the drill string breakage during the drilling. This failure is assumed to occur at a portion of the drill string that is in the dome. The flammable-gas detection system is the primary protection against this accident. This system prevents drilling operations when the flammable-gas concentration is above 25% of the LFL in the dome.

The force applied to the drill string is measured. The force limit, based on drill bit heating (Appendix F), is set to 750 lbf. There are other down force and rotational speed limits caused by structural concerns as discussed in Appendix N. The drill truck has a torque capacity that is more than that required to break the drill string. Several drill-string breakages have occurred in the past. A control is established not

to exceed the buckling limit (down force) and not to operate at a rotational speed that could excite the drill string at its natural frequency (Appendix N). The MAF listed in Table 4-7 for this accident is determined as 4.8E-10/yr (see Appendix E). The UAF is 2.8E-7/yr. Controls to prevent drill string buckling and resonance are given in Section 4.6.3.

| Accident | Frequency of failure 1/year | Dominant Failures | Controls Credited in Calculating UAF and MAF Values |
|--|-----------------------------------|---|--|
| Drill-string break in dome leads to dome | MAF = 4.8E-10 | Fail to shut down drill string on detection of H ₂ . | Drill string automatic trip when high H ₂ level, |
| fire. | UAF = 2.8E-7 | Drill string breaks by jamming in waste causing overtorque or buckling. | high rate of change of H2 level, or high dome pressure occurs. |

TABLE 4-7

DOME FIRE ACCIDENTS CAUSED BY DRILL STRING BREAKAGE

4.2.2. Ignition of Flammable Gas in the Riser Caused by the Drill-String, Equipment or Tool Drop (Installation and Removal)

This accident is postulated to occur when there is flammable gas in the dome, and equipment, drill string, or tools are dropped into or onto the riser during the installation or removal phase. As discussed previously, the riser condition must be determined before installation to make sure there is no flammable gas in the riser or dome. During the removal, gas may exist in the dome only if the flammable-gas release detectors fail. This accident analysis covers the friction in the riser caused by these drops. The more limiting case is the dropping of the drill string with an impact on the crust; this accident is examined in the next section. The friction in the riser as a piece of equipment falls through the riser is not considered to be a significant contributor and is bounded by the tool drop on the crust in the next section. Nevertheless, Appendix E determines the accident frequency of this scenario as 1.4E-9/yr as given in Table 4-8. The unmitigated frequency is also small, and 2.8E-9/yr. BOM testing performed by Witwer (Ref. 10) also included frictional spark tests for drill-string-to-drill-string impact and quill-rod adapter to drill string impact. These tests showed no ignition and bound the tool drop into the riser.

4.2.3. Flammable-Gas Release and Burn Caused by the Drill String or Tool Drop on Crust (Installation, Operation, and Removal)

This accident involves dropping the drill string or other tools that may create frictional sparks on the crust and thus results in ignition of the flammable gas in the vicinity of the crust.

General practice at Hanford site is to use spark-resistant tools during activities around the riser. Riser covers are also grounded and bonded. Standard procedure requires sampling of the dome space when opening a riser. The crust reaction/burn from impact heat is considered separately in the waste burn accident category.

There are several simultaneous conditions required to cause a dome collapse if the drill string is dropped. First of all, there must be a drop, and flammable gas must be present in the dome or in the vicinity of the impact point. Debris is found in the tanks from contaminated or unwanted material being disposed of in the tanks. There must be debris at the impact point, and the impact energy must be large enough to cause a frictional spark. A gas pocket large enough to sustain propagation must exist immediately beneath the surface where impact occurs.

The drill string is held by a pneumatic foot clamp. The drill string is manually inserted into the rubber frisbee by an operator. A lubricant is used to insert the drill string because the inside diameter of the frisbee is smaller than the outside diameter of the drill string. The rubber frisbee creates a frictional force of 200 lbf. However, this value may go down to 40 lbf when the lubricant is used. As the drill rods are added to the drill string, weight increases and exceeds the frictional force of the rubber frisbee. The maximum drop weight may be as high as 210 lb (Appendix G). The pneumatic foot clamp supports the drill string when it is not supported by the hoist or connected to the drill. The pneumatic foot clamp is designed to fail-close. To open the pneumatic foot clamp, the operator must activate the foot clamp pedal. The drill string is then raised about 1/2 in. before the foot clamp can be actuated, although this is not necessary by design but is caused by the seal used in the system (no credit is taken for this feature). The drill string is held by the foot clamp for a short period, about 4 minutes, during the collection of one sample. Considering the short lifting period, the drop accident frequency is estimated at 3.0E-3 /activity (see Appendix D).

In order to reduce the drop frequency further without taking into account additional events necessary to cause a dome collapse, the use of a locking collar when the drill string is held by the hydraulic foot clamp is required for all modes. The use of a locking collar reduces the drop frequency to 6.0E-5/yr. The collar needs to be installed before the drill string is disconnected from the drill unit. This requirement prevents flammable-gas exposure from the drill string during the installation of the collar.

The rotary drill bit is made of nonsparking copper-based material. Appendix T describes ignition tests performed for core sampler drops at BOM. In these tests, the core sampler is dropped in the drill string from a height of 60 ft in a flammable hydrogen-oxygen mixture. The impact includes a contact of a stainless-steel core sampler and copper-based drill bit. Witwer¹⁰ performed these tests and found no ignition. Drill string impact on metal objects is not bounded with these tests. However, test results indicate that the probability of a spark from the drill string dropping on the crust or metals in the crust may not be high.

As mentioned above, additional events may be required to cause a local hydrogen burn that could result in a dome collapse. These events must occur simultaneously and are as follows: having metals in a dry environment at the impact point that might cause sparks, having hydrogen in sufficient amounts to cause propagation in the waste in order to cause a dome collapse, or the occurrence of sparks from impacts with nonsparking materials on metals. Considering these additional probabilities and a low drop frequency, this accident is concluded not to be a cause for a dome fire. However, the possibility of a waste burn is considered in Section 4.4.3. The conditional frequency of tool drops on/into a riser is estimated as 1.4E-9/yr and is already small.

4.2.4. Ignition of Flammable Gas in the Riser Caused by a Frictional Spark (Installation, Operation, and Removal)

In this subsection, the main concern is the ignition of flammable gas in the riser caused by frictional sparks. There are several ways a frictional spark can be initiated in the riser. These possibilities and their management methods are discussed below.

A flanged, 2.9-in. i.d. (3-in., Schedule 80), and 15-feet-long conductive stainless-steel riser sleeve must be installed in the riser. The installation requires that the piece is bonded. The insertion and removal velocity of the conductive sleeve should be no more than 1 ft/s to prevent frictional sparks in the riser. The riser must be sniffed before insertion of the sleeve as a part of WHC standard control before opening a riser.

The other concern is frictional heating from the drill string rotating in contact with the conductive sleeve, leading to ignition in the riser. This situation is analyzed in Appendix P, and the conclusions are summarized below.

The expected temperature is less than the autoignition temperature for hydrogen in air for the period of the calculation (Appendix P). Given the conservative nature of the problem assumptions, frictional ignition in the tank dome from drill-string-tosleeve contact is considered unlikely. This determination is confirmed by BOM ignition test results obtained by Witwer (Ref. 10). App. T describes the experiments, and Witwer's results showed no ignition of stoichiometric hydrogen-air mixtures.

A nitrogen purge system is designed to provide flow into the annulus between the riser sleeve and the drill string. This system gives a minimum flow of 5 scfm at any tank pressure. There are two differential pressure detectors for riser sleeve purge gas. Either detector could cause an automatic trip of the drill. The set point is 40 psid across a flow controller sized for 5 scfm. The probability of a GRE pressurizing the dome by 3 in. w.g. is very low. Furthermore, a control is established to stop drilling when the dome pressure increases by 2 in. w.g. in any 5-min period. The differential pressure across the flow controller is monitored and set to sound an alarm to detect a loss of flow. The purge flow also further prevents the hot spot that could occur at

the drill string and sleeve contact point. This system provides the necessary protection to prevent ignition in the riser sleeve.

TABLE 4-8 DOME FIRE ACCIDENTS CAUSED BY EQUIPMENT AND TOOL DROPS INTO RISER

| | Frequency of | | Controls Credited in |
|---|--|----------------------------------|--|
| | Failure | Dominant | Calculating UAF and |
| Accident | 1/yr | Failures | MAF Values |
| Drop of raised drill | Drop frequency: | Foot clamp spuriously | Pneumatic foot clamp |
| string onto waste | | fails open. | fails, closes upon loss of |
| ignites H ₂ below the | MAF = 6E-5 | | air. |
| crust leading to a dome | | Locking collar is not used | |
| fire. | UAF = 2 | or fails. | A second locking collar is used. Collar needs to be |
| | Burn Frequency: | | installed before drill string is disconnected from |
| | MAF and UAF <<1.0E-6 | | the drill unit. |
| | from BOM test results for core sampler drops and other necessary probabilities. | | |
| Drop of tool into open riser leads to dome bum. | Drop frequency: MAF = 1.4E-9 | Tool is dropped into open riser. | Operator uses spark- resistant tools within 36D of open riser. |
| | UAF = 2.8E-9 | | |
| | Burn frequency with credit from BOM tests: | | |
| | MAF and UAF << 1.0E-6 | | |

If the nitrogen purge system fails, the ignition of flammable gases is possible. In order to demonstrate that sparks cannot be generated from drill string and sleeve contact, ignition testing experiments were performed by the BOM. The test conditions' requirements and acceptance criteria are discussed in Appendix T. BOM tests showed no ignition in stoichiometric hydrogen-oxygen mixtures (see Ref. 10).

Another case considered is the accumulation of hydrogen in the riser, but not in the dome. In this case, flammable-gas detection and the shut-down system could not
interrupt drilling. Protection is provided by the riser purge with a failure probability of 6.5E-3 (Appendix E). The accident frequency is calculated in Appendix E and given in Table 4-9. Based on the sparking tests of the drill against the riser sleeve at the BOM and the calculation that was done for frictional heating, ignition is not expected to occur in this potential pocket of gas. In addition, if it did, the volume would be small and not lead to a dome collapse.

TABLE 4-9

DOME FIRE ACCIDENTS CAUSED BY FRICTIONAL SPARKS IN THE RISER

| | Frequency of | | Controls Credited in |
|--------------------------------|----------------|---------------------------------------|---------------------------------------|
| | Failure | Dominant | Calculating UAF and |
| Accident | 1/yr | Failures | MAF Values |
| H ₂ in riser during | Without BOM | Fail to shut down drill | Drill-string |
| drilling leads to dome | test results | string upon detection of | automatically trips when |
| fire. | | H _{2.} | high H ₂ level, high rate |
| | MAF = 3.1E-10 | | of change of H ₂ level, or |
| | | N ₂ supply to riser sleeve | high dome pressure. |
| | UAF = 2.8E-5 | fails. | |
| | (assumes drill | | Leak check of N2 supply |
| | contact may | | to riser sleeve is done. |
| | spark) | | |
| | , | | DS trips on loss of N ₂ to |
| | With BOM test | | riser sleeve. |
| | results: | | |
| | | | Unique connector for N ₂ |
| | MAF and UAF | | supply to riser sleeve is |
| | <<1.0E-6 | | required. |
| | | | No supply to sleeve |
| | | | during actuation of sustem |
| | | | is verified. |
| | | | |

Even though the MAF is small, BOM experiments were needed (to verify that the ignition of a bounding hydrogen-oygen mixture is not credible, Appendix T) because of the uncertainties associated with the GRE probabilities. This test showed no ignition (Witwer¹⁰).

4.2.5. Ignition of Flammable Gas in the Riser Caused by Electrostatic Spark (Installation, Operation, and Removal) (Rubber Frisbee and Drill String)

For this event, the concern is with the electrostatic discharge from the rubber frisbee as the drill string rotates on this piece. The rubber frisbee diameter is slightly smaller than the drill-string diameter so that there is always a force exerted by the frisbee on the DS. A lubricant is applied to the DS in order to ease the rotation and insertion. The use of this lubricant can reduce the probability of the static electricity discharge during drilling. The frisbee is in contact with the grounded DS and washer, and the DS is connected to the drill unit that is grounded through the drill truck. The riser sleeve is purged with a nitrogen flow. The probability of failure of the nitrogen purge to the riser sleeve is very low; thus, a static discharge is not an issue. However, a control requires the use of the lubricant as a part of the procedure because it also helps to reduce the likelihood of damaging the rubber.

The resistivity of the frisbee is measured, and the findings show that the frisbee is not a good conductor. However, although the frisbee is composed of a nonconductive material, and if the drill string is always grounded by either the foot clamp or the hoist/truck, no credible spark source has been identified. Therefore, the SA does not consider a dome fire accident caused by this initiator and does not require the replacement of the frisbee as a mandatory control.

4.2.6. Ignition of Flammable-Gas Release Caused by Crust Penetration and Frictional Sparks Caused by the Drill Bit (Operation)

This accident analysis deals with the ignition of flammable gas that could exist in the crust or under the crust during the drill-bit penetration through the crust or waste sludge. The possibility of metal objects at the waste surface or in the crust always exists because it is known that manual tapes, wires, and metal pieces have been dropped through the risers. A frictional spark caused by the drill bit can ignite the hydrogen. The ability of the drill bit to cause a frictional spark when rotating on a steel plate has been observed. In testing discussed by Keller,¹⁴ the drill bit had carbide teeth and was operated on a steel piece in the dark. Sparks were observed. All carbide teeth are eliminated in later designs. New drill-bit cutting teeth are made of a proprietary sintered bronze with small tungsten chips in the bronze matrix. This material can wear easily when the drill bit is operated on metal objects The core sample drill bits used by FG/RMCS are Longyear or hard materials. (trademark of Longyear Incorporated) Part Number 100IVD/8 (currently used). It is known that the probability of sparks from copper-based materials is not high. Nevertheless, the inability of the drill bit to cause a frictional spark in a bounding flammable-gas mixture needed to be demonstrated by testing.

A series of ignition tests were conducted to demonstrate that the present drill bit design does not ignite a sensitive flammable-gas mixture. This testing addressed the frictional spark when the drill bit encounters a metal object that could be in the waste. Tests were designed to simulate the action of a drill bit striking a hard object inside a waste tank, such as a piece of structural steel or a rock. Tests were conducted in a bounding stoichiometric hydrogen oxygen and ammonia nitrous oxide mixture. The detail of the testing procedures are provided in Appendix T.

Controls and automatic shut-down features are used to keep the down force and rotational speed within the limits (750 lbf and 55 rpm) at any time during operations. As long as these limits are not exceeded, the accident frequency is very much less than 1.0E-6. In addition, the BOM tests demonstrated no ignition, as

discussed in Appendix T and by Witwer (Ref. 10). If the controls are exceeded, it is assumed that an ignition could occur if there is enough flammable gas. The frequency of the failures of down force and rotational speed, assuming that the controls are exceeded, is determined in Appendix E and given in Table 4-10. In order to initiate a dome burn when the drill bit is penetrating the crust, other necessary conditions must exist: (a) rocks (or other hard material) must exist at the surface or in the crust, (b) flammable gas with sufficient concentrations must exist, and (c) flammable-gas volume must be significantly high to cause propagation into the waste or the dome (propagation to the dome requires flammable gas be in the dome). Because failure frequencies are already small, the consideration of additional probabilities to cause a dome burn makes this scenario incredible. However, the same initiator when the drill bit is in the waste will be considered in Section 4.4 to address the flammable-gas ignition in the waste.

| Accident Excessive drilling rpm leads to ignition of flammable gas resulting in dome collapse. | Frequency of Failure 1/yr MAF = 1.4E-5 UAF = 2.2E-2 With qualification of drill bit, per test specified in Appendix T, UAF<<1.0E-6 | Dominant Failures The rpm setting is too high. Drill string fails to trip on high rpm. | Controls Credited in Calculating UAF and MAF Values Control over speed setting is used. Drill string auto trips on excessive rpm. |
|---|--|--|--|
| Excessive down force on drill leads to ignition of flammable gas resulting in dome collapse. | MAF = 9.4E-5 UAF = 1.2 With qualification of drill bits, per test specified in Appendix T, UAF <<1.0E-6 | Excessive down force is used. Drill string is on excessive force with either force detector or walkdown detector. | Control over down force is used. Auto trip drill string on excessive force is needed: force detector and walkdown detector function for all samples except last; force detector and bottom detector for last sample are needed. |

 TABLE 4-10

 DOME FIRE ACCIDENTS CAUSED BY DRILL BIT FRICTIONAL SPARKS

4.2.7. Ignition of Flammable Gas in the Dome (Operation)

This event considers the possibility of an ignition caused by the existence of energized equipment in the dome or domes of connected tanks or connecting ventilation systems. Any activity in the connecting tanks may initiate a fire that may propagate into the tank being sampled. Appendix B requires that all equipment in the dome be rated for operations in a Class-I, Div.-1, Group-B environment or a Class-I, Div.-2, Group-B environment with automatic shutdown. Any equipment that does not meet this requirement must be deenergized during FG/RMCS wasteintrusive operations. No other activities in the connecting tanks or on the same tank are allowed during FG/RMCS waste-intrusive operations. These controls reduce the likelihood of ignition caused by existing equipment in the tank dome and the domes of connecting tanks. Violation of this control may result in an unanalyzed initiator.

4.2.8. Ignition Of Dust and Flammable Gas in the Dome (Operation and Removal) (Static Electrical Charges and Other Causes)

In this accident scenario, the ignition of aerosols generated in the dome is analyzed. In Appendix H, a bounding analysis is performed to show that the energy contribution of combustible dust is negligible and that the addition of dust in a hydrogen-air mixture would not result in explosion.

4.2.9. Summary of the Dome Fire Accident

The drill string break resulting in a dome fire is the dominant fire initiator in the dome space. The principal contributor to the accident sequence is exceeding the operating limits. The resulting dome fire MAF is 4.8E-10/yr. The UAF is 2.8E-7 and is low. The consequence of this accident is the dome collapse analyzed in Section 5 of this SA.

4.3. DRILL STRING FIRE ACCIDENTS

Drill string fire accidents consider all possible fire initiators inside the drill string. Each ignition initiator determined for this category is discussed separately. The ignition source may exist as a result of normal as well as abnormal operations.

The consequences of a drill string fire vary and may end up with small amounts of waste release as well as a dome collapse. Examples of how a drill string fire may propagate to other accidents are discussed in Appendix A. It is assumed that the fire in the drill string (if it occurs) propagates to the tank and results in dome collapse. As explained in the introduction, this assumption is conservative, and there is only one consequence, dome collapse, if a drill-string fire occurs. The conditional frequency of each accident scenario, however, is discussed and the adequacy of preventive features is demonstrated.

4.3.1. Flammable-Gas Penetration into the Drill String (Operation)

Section 3 identified the fact that there may be several reasons for hydrogen penetration into the drill string:

- Failure of the chevron sampler seal between the drill bit and the drill string,
- Hydrogen diffusion,
- Waste in the drill string,
- Failure of the core sample ball valve on the core sampler,
- Depressurization of waste in the drill string,
- Loss of hydrostatic pressure,
- Incompatible material, and
- Failure to latch the sampler.

The hydrogen can be generated in the drill string if there is waste penetration into the drill string. A leaky chevron seal may allow waste penetration into the drill string. If the sampler encounters large gas pockets in the waste (although credible evidence for this mechanism is not provided), the sampler could be filled with flammable gas. Failure of the core sampler valve in the drill string could release hydrogen into the drill string. Failure to latch the sampler to the core barrel can cause flooding and gas generation in the drill string. However, as long as hydrostatic pressure during sampler retrieval and nitrogen purge during drilling are available, the hydrogen release into the drill string from the sampler or waste is not of concern because there is enough flow rate to purge or prevent diffusion up the drill string. Appendix J concludes that the minimum flow rate at which hydrogen could diffuse against the flow is 5E-4 scfm, which is lower than a minimum flow rate of 0.3 scfm provided by the purge system.

One other mechanism that can produce flammable gas is the use of incompatible material in the drill string design. The drill string is made of steel, and steel is known to be compatible with the waste in terms of violent chemical reactions that would result in gas releases or other undesirable consequences. WHC recently reviewed the sampler to ensure that there were no incompatible materials used in the construction; in particular, they examined the design for the presence of aluminum. Also, the drill bit and seals used in the design have been used in actual waste and found to be compatible (Ref. 15).

.

WHC-SD-WM-SAD-035, Rev. 0-a

There are three major causes for hydrogen accumulation in the drill string:

- 1. Failure of nitrogen flow during drilling,
- 2. Failure of hydrostatic flow during retrieval, and
- 3. Failure of the chevron seal before the change-out assembly is installed.

The sampler is designed to prevent waste and flammable-gas penetration into the drill string through a chevron seal located at the end of the core sampler. It is a one-way seal and allows nitrogen flow from the drill string into the tank waste. When the drill string is rotated, the core barrel rotates around the core sampler while the seal is slightly compressed to allow nitrogen flow. When the sampling is finished and the purge flow stopped, the seal provides a barrier between the tank waste and the drill string. Flammable gas may penetrate into the drill string if the seal does not provide an adequate barrier. It is reasonable to assume that the seal may fail partially if not completely before the retrieval of the sampler (as change-out assembly is installed). Calculations (Appendix J) show that hydrogen can diffuse in nitrogen relatively fast.

It is required that the drill string be purged to evacuate flammable-gas accumulation following a procedure developed by WHC for the following conditions:

- The sampler is left in the DS without nitrogen flow or hydrostatic pressure for ≥ 60 minutes, or
- DS hydrostatic head is lost with no sampler in place.

If these conditions occur, the grapple or remote latch unit should not be operated unless the nitrogen purge criteria are met. The hydrostatic head pressures for each sampler need to be calculated before operation.

4.3.2. Ignition of Flammable Gas in the Drill String Caused by Drops (Operation)

The remote latch unit and the grapple are lowered and raised during the insertion and removal of the sampler. Each unit is driven by an electrical motor through a gear box. Each is mechanically attached to a driving system through a cable. The insertion and removal rates of the remote latch unit and grapple are specified as less than 1 ft/s.

In this section, accidents resulting from frictional sparks in the drill string as a result of dropping the grapple, remote latch unit, and core sampler are considered. In the case that the hoist systems for the grapple and remote latch unit or electrical motors used to raise or lower these units fail, the grapple and remote latch unit may be dropped on the core sampler. The keys or pins connecting the shaft to the drum can be broken as a result of a stuck core sampler, grapple, or remote latch unit. The following are assumed to cause frictional sparks:

- Multiple drive train failures,
- Failure of keys or pins connecting the shaft to the drum, and
- RLU failure.

Studies have been performed to examine the characteristics of mechanically produced sparks that could lead to ignition of hydrogen-nitrous oxide mixtures. Drop velocity could be as high as 19.8 m/s and is well above the range given by Krok and Shephard's work.⁹ Therefore, if the remote latch unit or sampler is dropped in the drill string, it is assumed that an ignition source can be created.

The remote latch unit and grapple are driven with DC pulse-width modulated electric motors. Their principal operating parameters are given in Section 2. ecause of the worm-gear reduction boxes, failure of these motors would not result in a drop. However, a drop accident may occur as a result of the failure of the metal key or pins that are used to attach the shaft to the cable drum. These pins are the weakest points of the system and may fail if the grapple or the RLU is stuck or overweighted because of some unknown reason.

The operation of the RLU is entirely mechanical and is discussed in Section 2. When the RLU is lowered and the sampler comes to rest at the bottom of the drill string, the cable stops when the load cell detects the cable going slack. As the tungsten weights cause the dashpot piston in the RLU to descend, the load cell again engages the downward motion of the hoist.

When the sampler is being raised or lowered in the drill string, the hydrostatic head must be in operation. There are two independent sources of hydrostatic head, one through the shielded receiver and one through the change-out assembly.

Drops caused by failure of the remote latch unit were concluded in section 4.1.2.2 to be not credible. However, the grapple can be dropped in the drill string if the shear pin on the drum fails. The load on the grapple is measured. The load cell automatically trips the electrical motor when the reading is out of tolerance. A control is established to not exceed a maximum load of 250 lb. For an ignition to occur, flammable gas must be present, which means that the hydrostatic head purges and chevron seal must have failed. Before the pintle rod is removed, the hydrostatic head must be established. This requirement is established as a control. The probability of a drop is estimated as 1.4E-7, based on the controls established for load measurements. The MAF of this accident is calculated as 6.0E-11/yr in Appendix E and is given in Table 4-11. The UAF is 1.3E-3/yr. Because the MAF frequency is so small, no special ignition testing is required for the grapple drop caused by ignition accidents. However, the established control becomes important.

The frequency for dropping the core sampler has been estimated as 1.9E-6/yr in Section 4.1.4. The estimated frequency is not <<1.0E-6/yr. Therefore, drop tests

simulating the drop of the core sampler on the drill bit were performed by the BOM (test description and requirements are given in Appendix T). A prototypical core sampler was dropped from a height of 60 feet through the drill string with the drill bit attached to the lower end of the drill string. The test chamber and the drill string were filled with a stoichiometric hydrogen-air mixture. The tests were repeated ten times. Ignition did not occur as reported by Witwer (Ref. 10). Therefore, this accident is not considered further.

4.3.3. Ignition of Flammable Gas in the Drill String Caused by Assembly/Disassembly of Drill Rods or Drill Rod-Quill Rod Adapter Impact

Ignition of the flammable gas in the drill may be caused by impacts that could be created during drill-string or drill-string-to-quill-rod assembly or disassembly. The postulated accident can occur when the drill string is in the waste or dome. Accident frequency is different for these two cases because of the assumption regarding to the flammable gas. Flammable gas exists in the dome only if a GRE occurs while it is assumed to exist in the waste.

A spark during the disconnecting of the quill rod from the drill string has been observed.¹⁶ The quill rod adapter and drill string were made of carbon steel. Any misalignment between the drill string and quill rod caused by undesired platform movement or operator errors could create a relatively fast impact between the quill rod and drill string when the drill string is disconnected. There is no instrument to detect the misalignment or any stress level on the drill string or quill rod adapter. Therefore, it is very difficult to evaluate the condition of the drill string to quill rod adapter before disconnecting the drill string. Because a spark is observed in the operation, one must assume that the likelihood of this event is high.¹⁶

The drill string is assembled by adding drill rods. The operator picks the drill rod from the storage location and lifts and screws it onto the drill string by hand. Then the lifting bail is attached to the drill string.

During insertion, drill-rod-to-drill-string impact is very possible. Impact can be caused by dropping the drill string or operator error. Drop height is limited with the drill rod length. However, the operator can also cause a lateral impact. It is determined that a realistic impact velocity is around 10 ft/s. It is not clear that the impact with this velocity would not cause a spark that is capable of igniting the flammable-gas mixtures of concern. A spark is possible because the drill rods are made of carbon steel and assembling/disassembling is performed for each sample.

The cable spray washer is installed after the drill string is disconnected from the quill rod adapter. Dropping the cable spray washer may produce a spark. Next, the change-out assembly is installed on the cable spray washer. Dropping the change-out assembly on the cable spray washer may produce a spark. In order to have an

ignition in the drill string, the flammable gas must exist in the drill string. The drill string must be sniffed before it is disconnected from the quill rod adapter.

A drill string fire initiated by a spare caused by the drill-rod-drill-string or the drillstring-quill-rod adapter impact is discussed in this section. Event and fault trees for this scenario are given in Appendix E. Sniffing the drill string before the drill string is disconnected from the quill rod adapter through a port on the quill rod is required. The reliability of the inspection is limited with the reliability of the sniffing equipment, which is on the order of 1.0E-3. If the change-out assembly is installed immediately (within half an hour), no other sniffing is required. If, however, the drill string is capped and the change-out assembly is installed later, there is a need for a second sniffing from a port on the cap sealing the drill string while the drill string is closed. This is because of the possibility of hydrogen diffusion through the chevron seal (Appendix J).

TABLE 4-11

DRILL STRING FIRE ACCIDENTS–DRILL STRING IMPACT, GRAPPLE, AND CORE SAMPLER DROP

| | Frequency of | | Controls Credited in |
|------------------------|--------------------------|-------------------------|--------------------------------------|
| | Failure | Dominant | Calculating UAF and |
| Accident | 1/yr | Failures | MAF Values |
| Drop of grapple in the | | Nitrogen hydrostatic | Controls are applicable to |
| drill string | MAF = 6.0E-11 | head system fails. | nitrogen hydrostatic head system. |
| | UAF = 1.3E-3 | Load cell fails to trip | |
| | | electrical motors. | Electrical motor |
| | | | automatically trips when |
| | | Cable inspection fails. | the load is \geq or equal 250 lb. |
| | · | Chevron seal fails. | |
| | | | Inspect the cable for |
| | | | possible damage every 6 months. |
| | | | |
| | | | |
| | | | |
| Drop of core samplers | Without BOM | Nitrogen hydrostatic | Controls are applicable to |
| in the drill string | test results | head system fails. | nitrogen hydrostatic head system. |
| | MAF = 1.9E-6 | RLU drops the sampler. | |
| | UAF =2.8E-3 | | |
| | With BOM test results | | |
| | MAF and UAF << 1.0E-6 | | |

TABLE 4-11 (cont) DRILL STRING FIRE ACCIDENTS–DRILL STRING IMPACT, GRAPPLE, AND CORE SAMPLER DROP

| | Frequency of | Dominant | Controls Credited in |
|---|---|---|---|
| Accident | 1/vr | Failures | MAF Values |
| Spark from assembly/disassembly of drill string sections or drill string to quill rod impact when the drill string is in the waste (does not contribute to the total frequency of dome collapse) | Without BOM test results MAF = 7.6E-5 UAF = 1.6E-2 With BOM test results MAF and UAF << 1.0E-6 | Sniff for flammable gas is not performed in the drill string. Sampler chevron seal fails. | Sniff enclosed volume for hydrogen in drill string. Presence of sampler with chevron seal in drill string prevents hydrogen movement into drill string. |
| Spark from assembly/disassembly of drill string sections or drill string-quill rod impact when the drill string is in the dome (does not contribute to the total frequency of dome collapse) | Without BOM test results MAF = 1.1E-9 UAF = 2.3E-7 With BOM test results MAF and UAF << 1.0E-6 | Sniff for flammable gas is not performed in the drill string Sampler chevron seal fails. | Sniff enclosed volume for hydrogen in drill string. Presence of sampler with chevron seal in drill string prevents hydrogen movement into drill string. |

The chevron seal failure probability is estimated as 3.3E-2. The frequency of operator error causing drill-rod-to-drill-string impact is 0.5/activity. Considering the reliability of sniffing, an accident frequency of 7.6E-5 /activity is calculated if the drill string is left in the waste (note that the probability of hydrogen in the waste is assumed to be 1.0). This frequency is valid for a fire accident caused by drill-string-quill-rod impact. The drill string could be in the dome. In this case, hydrogen must exist in the dome in order to penetrate into the drill string. Thus, a GRE is required.

Considering the GRE probability, the accident frequency becomes low, on the order of 1.1E-9/yr. These frequencies are summarized in Table 4-11. The accident frequency when the drill string is in the waste is high; therefore, laboratory ignition testing is designed to demonstrate that ignition is not possible.

The following design changes were made to manage the above-mentioned mechanical sparks. First, the quill rod adapter material was changed from carbon steel to stainless steel. Second, all pipe compounds and lubricants used in FG/RMCS operations were reevaluated.¹⁶ It was found that one of the pipe joint compounds, Bostik Never-Seez Anti-Sieze and Lubricating Compound, showed evidence of enhanced sparking. Necessary procedural changes were made to use only acceptable joint thread compounds listed in Ref. 16. In addition, a series of ignition tests were performed as described in Appendix T. These tests address the drill-rod-to-drill-string impact, drill-string-to-quill-rod-adapter impact, the change-out assembly-to-cable spray washer impact, and cable-spray-washer-to-drill-string impact.

The first series of tests simulated the carbon-steel drill strings impact by dropping a prototy e 19-in. drill string on another vertically oriented prototype drill string from a height of 3 feet. The height of 3 feet corresponds to the impact velocity of 14 ft/s, and impact masses are conserved. Tests are performed 30 times. These tests also address the ignition caused by the drop of the change-out assembly. Witwer (Ref. 10) reported that no ignition was observed in the above-mentioned ignition tests. Therefore, they are considered to be incredible as indicated in Table 4-11.

The quill rod adapter impact test (see Appendix T) used a section of the same type stainless-steel pipe dropped on its end onto the end of a section of carbon drill string. A maximum kinetic energy during drill string/quill rod misalignment is estimated as 115 in. \times lb. A safety factor of two was employed, raising the value to 230 in. \times lb. A 5.22-lb piece was dropped 44 in. onto the carbon-steel pipe. The gas mixture used both in the drop tube and the test chamber was stoichiometric hydrogen and oxygen. The test chamber and drop pipe were both sufficiently purged before the drop. Tests were repeated ten times. The tests results reported by Witwer (Ref. 10) and showed no ignition. Therefore, the accident is considered to be incredible as shown in Table 4-11.

4.3.4. Ignition of Flammable Gas in the Drill String Caused by Unqualified Electrical Equipment in the Drill Head or Shielded Receiver (Operation)

This accident is concerned with the ignition of the flammable gas by electrical sparks. The SR and grapple box have load cells. These load cells are used to measure the tension on the cable that is attached to the remote latch unit or grapple. The load cells are protected by intrinsic safety barriers. WHC quality assurance requirements ensure that these barriers are certified. Thus, the load cells are not electrical spark sources. The electrical motors are located outside of the grapple box and the SR, which are sealed containers. There is no other equipment in the SR and the drill unit that can cause electrical sparks.

August 8, 1996

4-36

4.3.5. Ignition of Flammable Gas in the Drill String Caused by Drill-String Failure (Operation)

This scenario is very similar to the event discussed in the dome fire (drill string break). The difference is this scenario assumes that the fire starts in the drill string (flammable gas is in the drill string, not in the dome). The initiating event is the drill string failure, causing a hot spot in the vicinity of the failure location. This hot spot ignites the hydrogen in the drill string. The propagation of the fire in the drill string into the waste is considered to be possible. Therefore, this postulated accident may result in dome collapse.

Two failures must occur for this scenario to occur; hydrogen in the drill string and failure of the drill string. Hydrogen is not expected to be in the drill string during drilling because it requires a failure of the nitrogen flow and the chevron seal. The conditional mitigated frequency of this accident (1.3E-8/yr) is the frequency of the undetected nitrogen purge failure (1.9E-05), the failure of the chevron seal (3.3E-2), and the frequency of the drill-string breakage (1.0E-2).

4.3.6. Ignition of Flammable Gas in the Drill String Caused by Frictional Sparks (Operation)

This accident scenario considers an ignition of hydrogen in the drill string caused by frictional sparks. Hydrogen can exist in the drill string because of the failure to latch a sampler, a leaking chevron seal, etc. The drop cases were treated in Section 4.4.2. The possible sources for mechanical sparks are

- Failure of bearings,
- Frictional sparks caused by operation of the remote latch unit, and
- Frictional sparks resulting from the operation of the grapple,

Acceptance testing has verified the operability of bearings. The sampler and these bearings are used only once. The only time the bearing sees relative motion is during rotary sampling at which the time nitrogen purge is present. Waste penetration into the bearings is minimized because they are out of the waste path entering the sampler. Therefore, the failure of bearings resulting in sparks is not credible.

Metal-to-metal sparks are managed by limiting the velocity of movement of the remote latch unit and grapple to no more than ~0.3 m/s (1 ft/s). This velocity is adopted from previous studies¹⁷ concerning a similar hazard of lowering or raising a metal pipe through a riser. It has been shown that mechanically produced sparks that could lead to ignition of tank-like gases is not likely when the frictional impact velocity is less than 2.5 m/s (8.2 ft/s) (Ref. 9).

4-37

A gear reducer and electric motor are used to limit the grapple and remote latch unit motion to 1 ft/s and 0.4 ft/s, respectively. Therefore, frictional sparks are not likely. The frictional sparks caused by a drop accident are discussed in Section 4.3.2.

A credited design feature controls the insertion or removal speed of the grapple and the remote latch unit to no more than 0.3 m/s (1 ft/s). In addition, a purge pressure equal to hydrostatic head pressure before insertion or removal of the grapple or remote latch unit is established. Sniffing the drill string each time the drill string is disconnected from the quill rod adapter reduces the possibility of hydrogen accumulation.

4.3.7. Ignition of the Flammable Gas in the Drill String Caused by Frictional Sparks from the Shear Pin Break (Operation)

This accident has been identified during the hazard identification process that considered a design involving a copper pin used to attach the piston to the pintle rod. The pintle rod release mechanism was redesigned to rely on a mechanical release instead of the copper wire shearing, and it is a metal clip. The use of a metal clip is not expected to produce a spark. Thus, this event is not considered to be credible. During the pintle rod removal operation, the hydrostatic head purge is established, and this provides additional protection.

4.3.8. **Detonation** in the Drill String and Drill-String Ejection (Operation)

Appendix J discusses the possibility of detonation in the drill string. It assumes that there is hydrogen in the drill string and that it can be ignited. Theoretical detonation pressure for hydrogen-air mixtures in a pipe can be calculated by using a numerical method to solve the differential equation for isentropic compression in the burn gas. From the analysis given in Appendix J, the overpressure and the rate of pressure rise during a burn in the drill string are 630 kPa and 2279.5 bar/s, respectively. During a detonation in the drill string, the overpressure and the rate of pressure rise are 15 bars $(1.5 \times 10^6 \text{ Pa})$ and $3 \times 10^6 \text{ bar/s}$, respectively. As explained in Appendix J, the induction distance for the H₂-N₂O mixture is short relative to the maximum drill string length. Therefore, a deflagration-to-detonation transition (DDT) may occur in the drill string. The rate of pressure rise is so high $(3 \times 10^6 \text{ bar/s})$ during the detonation, that the structure of the drill string is expected to fail.

This conclusion is important from the consequence point of view. If the detonation in the drill string causes ignition in the waste with propagation, then a dome collapse would be expected. However, the consequence of dome collapse is not increased by this initiator because the dome collapse accident considered in Appendix I is already conservative and includes bounding waste release amounts. However, detonation may only fail the drill string and result in ejection of the core sampler. In this case, the consequences of this accident may be a release of the waste in the sampler. In Section 5, a waste release of 0.39 kg is considered.

4.3.9. Ignition in the Drill String Caused by Lightning

The lightning strike is discussed in Section 4.1.1. The controls developed to prevent ignition caused by lightning in Section 4.1.1 are applicable to this section as well.

4.3.10. Ignition in the Drill String Caused by Static Electricity Between Seals and Rotating Parts

In this scenario, the concern is the static electric discharge between the seals and the rotating parts in the sampler. The core sampler motion is in the closed space. The rotational or penetration speed is slow, and there is always direct contact with metals.

4.3.11. Summary of Drill-String Fires

All of the initiators previously identified in Section 3 were found not to cause fire in the drill string because of safety design features and experimental results obtained from ignition testing. The accident frequencies of hydrogen penetrations into the drill string are developed in Appendix E and summarized for this category in Table 4-12. The drill string fire accident frequency is 1.3E-8/yr.

4.4. WASTE FIRE ACCIDENTS

Accident sequences considered in the waste fire category include waste ignition as well as flammable-gas ignition in the waste. The waste in single-shell flammablegas tanks includes organics. Below, the initiators that could result in waste fire accidents are discussed.

4.4.1. Waste Fire Caused by Drill Bit Over-Temperature (Operation)

Wastes including mixtures of sodium nitrate and sodium nitrite with organic compounds can produce violent exothermic reactions (Appendix G). Increasing the temperature of the waste in the vicinity of the drill bit can cause a thermal runaway. There are several hazards that are associated with a local thermal runaway, and they are discussed in Appendix G. Two major important hazards are the ignition of the flammable gas and the initiation of a self-propagating exothermic reaction in the waste. Reactions in mixtures containing relatively small amounts of organic compounds can result in temperatures greater than the autoignition temperature of hydrogen mixtures, so the ignition of flammable gases is the more limiting condition. However, a self-propagating reaction would produce very high temperatures, which would cause structural damage to the tank. The consequences of a self-propagating reaction could be severe.

Because the possibility of a flammable-gas mixture cannot be eliminated, the approach used in this Safety Assessment is to take all practical measures to eliminate ignition sources. A local runaway reaction is a potential ignition source, so the requirement that there be no local runaway reaction is consistent with the

philosophy used in this SA. Preventing a local thermal runaway is also protection against a propagating exothermic reaction, and it eliminates the possibility of generating additional flammable gas as a result of elevated temperatures. Appendix G discusses runaway reactions and waste ignition in great detail.

| | Frequency of | Dominant | Controls Credited in |
|--|---|--|--|
| Accident | 1/vr | Failures | Values |
| Drill string break leads to a fire in the drill string. | MAF = 1.3E-8 UAF = 6.6E-4 | Drill String fails to shut down on loss of nitrogen purge. Drill string breaks by jamming in the waste causing overtorque or buckling. | Drill string automatically trips when the nitrogen purge to the DS is lost. |
| H2 in drill string/shielded receiver during sampler handling leading to waste fire. (does not contribute to the total frequency of dome collapse) | Without BOM tests results MAF = 1.9E-6 UAF = 2.8E-3 With BOM test results MAF and UAF <<1.0E-6 | N2 hydrostatic systems to both drill string and to shielded receiver fails. RLU drops sampler. | Leak test of N2 hydrostatic systems for both drill string and for shielded receiver are done. Unique connections for N2 hydrostatic systems for both drill-string and shielded receiver are used. Verification of N2 hydrostatic supply to both drill-string and shielded receiver during activation of hydrostatic mode of N2 supply is made. Controls over operation of RLU are used. |

TABLE 4-12 DRILL STRING FIRE ACCIDENTS-CONDITIONAL FREQUENCIES OF HYDROGEN PENETRATION INTO THE DRILL STRING

Basic conclusions of Appendix G are that local runaway reactions can be prevented by establishing waste temperature limits. The following temperature limits are established:

• The temperature of small waste fragments produced at the drill tip must not exceed 180°C.

• The temperature of the drill bit and the average temperature of the waste affected by drilling must not exceed 160°C for more than 10 minutes.

Because the consequences of a propagating exothermic reaction are severe, FG/RMCS should not be performed in tanks in which a propagating exothermic reaction may occur.

New envelope testing has been performed by WHC to determine the operating parameters, rotational speed, down force, and nitrogen purge flow to comply with the safety criteria given above. Witwer¹⁸ discusses the details of testing and the results obtained. Tests and results are also summurized in Appendix F.

As a summary, results of envelope testing (Witwer, Ref. 18) and their analysis (Appendix F) showed that the drill bit surface temperature correspondingly the waste substrate temperature can be kept below 160°C, including an uncertainty of 10° C, if the following limits are applied: a down force \leq 750 lbf, a rotational speed \leq 55 rpm, a minimum nitrogen flow \geq 30 scfm, and a penetration rate \geq 0.75 in./min. The chip temperatures under these conditions are also limited to 180° C as required. If a trip is initiated when one of the set points for these four parameters is exceeded, drilling must be stopped. After a shutdown there must be a waiting period of 10 minutes before drilling can continue. The waiting period of 10 min is based on the experimentally determined cooling time. The testing and the analysis included plugged holes on the drill bit.

The minimum purge flow must be greater than 30 scfm; however, it is possible that necessary cooling to the drill bit would not be provided if there were a leak from the nitrogen purge system between the flow measurement location and the drill bit. As indicated in Table 4.13, the leak rate from the nitrogen system must be checked once every 6 months. This control requires that the leak rate must be within the uncertainty range of instrumentation or less than 2% of the nominal flow.

Drill rods are threaded to each other. An O ring is used to provide a seal. The leaks are possible if the O rings are left out. WHC¹⁹ determined the possible leak rates could not be higher than 0.3 scfm when the O rings are not used. This is less than 1% of a nominal flow of 30 scfm and negligible. With the use of O rings, the leak rate also was measured and was shown to be negligible. Therefore, O rings on the drill rod are not required, and the nitrogen purge flow for drill bit cooling is sufficient when set to a minimum of 30 scfm.

There is one event that would include an unknown leak path as a result of failure of the drill string during drilling. If the drill bit or string becomes embedded in the waste momentarily because of debris in the waste, torque could continue to be applied to the drill string at a constant rate. If such a condition occurs, there is a possibility that the drill string could partially fail. Continuing to operate with a partially failed drill string could result in a nitrogen flow bypass through the failed area. This concern is assessed below. Appendix N examines the possibility of over-torquing the drill string. The drill string is considered as having torque applied from the upper end, but the rotation of the lower end is not allowed. Appendix N presents two methods to determine the time necessary to break the drill string. Linear elastic methods are applied as a first approximation to obtain the lower bound failure estimate. Second, strain-energy methods are used to determine an upper bound by assuming that the ultimate shear strain in the drill rod is proportional to the shear modules. It is estimated that failure would occur in less than 15 seconds for all rotational speeds. Note that Appendix N did not take any credit for the threaded drill rods. Experience shows that the drill string always fails at the threaded sections. The real failure time is expected to be in a few seconds because of the stress concentration factor for threaded sections. Therefore, it is concluded that a drill string tear without a break is very unlikely.

Envelope testing measured the rate of increase in the drill-bit surface temperature when nitrogen flow is terminated at steady-state operating conditions. The test results are summarized in Appendix F and by Witwer (Ref. 18). Results showed that an average heat-up rate of 2°C/s is observed in the time period of 0 to 20 seconds after the nitrogen flow is shut down. This rate corresponds to a temperature increase of 30°C in 15 seconds in which the drill string would be broken when overtorqued. Envelope testing established the operating parameters so that the drill bit and waste temperature is less than 150°C. Considering a 30°C heat-up of the drill bit for this accident, the drill bit/waste temperature would be 180°C. Appendix G shows that the waste temperature would be allowed to be at the minimum exothermic-reaction temperature of 180°C for a short period of time because the induction time of reaction is expected to be much larger than 10 to 20 seconds. Therefore, it is concluded that if the drill fails because it is over-torqued it would fail in a time period in which the waste in the vicinity of the drill bit would not experience runaway reactions.

4.4.1.1. Summary of Controls for Drilling Operation.

Envelope testing shows that a maximum drill-bit temperature corresponding to the maximum waste substrate temperature increase is limited to $\Delta T = 60^{\circ}$ C when the rotational speed, down force, and nitrogen purge flow are 750 lbf, 55 rpm, and 30 scfm, and there is good penetration (penetration rate is higher than 0.75 in./min). These parameters consider a partially plugged drill bit and ensure that waste chip temperatures are bounded by the safety limit of 180°C. In order to meet the safety criteria established in this SA, the down force and the rotational speed must not exceed 750 lbf and 55 rpm. The minimum nitrogen flow rate must not be lower than 30 scfm. The penetration rate must not be less than 0.75 in./min. The probability of having a hard layer in the waste is 0.1. This is obtained with the assumption that 20% chance for a need for rotary drilling in an activity and 50% chance of hard layer in rotary mode drilling.

When a trip value is reached in one of the three parameters, down force, rotational speed, and nitrogen purge flow, the drilling must be stopped within the time period that data acquisition activates the shut-down signal. The alarm set point may be chosen lower than the trip values. If the alarm set points are set to trip values, 750 lbf, 55 rpm, and 30 scfm, the programmable logic controller (PLC) sends a shutdown signal to the drill engine upon receipt of a valid shutdown signal with no additional programmed-delay. It is understood that the determination of a valid alarm signal requires approximately 2 seconds. The penetration rate is alarmed at 0.75 in./min and shut down occurs with a 60-s cumulative time in any 3-min period. This gives a chance for the operator to penetrate a thin, hard layer (if encountered), provided that the force, rotational speed, and nitrogen purge controls are not violated. There must be at least a 10-min waiting period following the trip before the continuation of drilling.

The nitrogen inlet temperature to the drill string must be maintained between 10°F to 140°F.

4.4.1.2. Accident Frequency of Waste Ignition

Failure to shut down the drilling when limiting operating parameters are exceeded is assumed to cause runaway reactions in SSTs. The reliability of the nitrogen purge system and control system for the rotational speed and down force are examined in Appendix D. In Table 4-13, conditional accident frequencies are given. The accident frequencies caused by loss of cooling, excessive rotational speed, down force, and low penetration rate are estimated as 3.8E-5, 1.4E-5, 9.4E-5, and 7.4E-5/yr. These numbers consider the use of a bottom detector or walkdown option to shutdown. These frequencies are added to obtain the conditional frequency of a waste fire accident caused by drilling operations. The combined frequency is 2.3E-4/yr.

4.4.2. Waste Fire as a Result of Exothermic Reaction Caused by Incompatible Materials (Installation, Operation, and Removal)

Aluminum is known to be incompatible with tank waste. The existence of an exothermic chemical reaction between aluminum material and the waste has been observed. The reactions can produce large quantities of flammable and toxic gas, and heat. This material must not be used inside the waste tank. Inadvertent placement of this material in the waste tank environment presents itself as one of the more logical sources of an accident initiator. Use of this material in the drill-string column could lead to a rapid chemical reaction.

All that is required for this accident to occur during the core-sampling phase is for the sampler seal to leak waste into the drill string. Personnel exposure to this hazard occurs when the drill string is opened to add a section. At this point, the pressure balance across the seal provided by the nitrogen gas has been removed, and the string is at atmospheric pressure. This allows the waste to rise up into the

column. When the waste encounters incompatible aluminum material, an exothermic reaction occurs.

| | Frequency of Failure | Dominant | Controls Credited in Calculating UAF and |
|------------------------|---------------------------------|--|---|
| Accident | 1/yr | Failures | MAF Values |
| Loss of drill-bit | Without | N ₂ cooling holes in the | Auto trip of the drill |
| cooling leads to waste | probability of | drill bit are partially | string occurs on low N ₂ |
| fire. | reactive waste | blocked, there is no trip of | purge gas flow. |
| | MAE - 38E-5 | the drill string on low N ₂ | America strate of high No. |
| | 10111 - 0.01 0 | flow, and there is | Annunciation of high N ₂ |
| | UAF = 2.0 | sections of drill bit. | required. |
| | With | | No purge system is tested |
| | consideration of probability of | | for bypass leakage. |
| | reactive waste | | Test results show that N2 |
| | for selected | | leakage from drill string |
| | G) | | with section O rings not |
| | -, | | installed is acceptable. |
| | MAF <1.0E-6 | | A |
| | · · · | | drill string tear without a |
| | | | break is very unlikely. |
| | | | |
| Excessive drilling rpm | Without | The rpm setting is too | Control over speed setting |
| leads to waste fire. | probability of | high. | is used. |
| | reactive waste | Drill string fails to trip on | Drill string auto trips on |
| | MAF = 1.4E-5 | high rpm. | excessive rpm. |
| | UAF = 2.2E-2 | | |
| | With | | |
| | consideration of | | |
| | probability of | | () () () () () () () () () () |
| | reactive waste | | |
| | tanks (Appendix | | |
| | G) | | |
| | | | |
| | MAF <1.0E-6 | | |

TABLE 4-13WASTE FIRE CAUSED BY FRICTIONAL HEATING FROM THE DRILL BIT

TABLE 4-13 (cont)WASTE FIRE CAUSED BY FRICTIONAL HEATING FROM THE DRILL BIT

| | Frequency of Failure | Dominant | Controls Credited in Calculating UAF and |
|--|--|---|---|
| Accident | 1/yr | Failures | MAF Values |
| Excessive down force on drill leads to waste fire. | Without probability of reactive waste MAF = 9.4E-5 UAF = 1.2 With consideration of probability of reactive waste for selected tanks (Appendix G) MAF <1.0E-6 | Excessive down force is used. Drill string on excessive force with either force detector or walkdown detector. | Control over down force is used. Auto trip drill string on excessive down force: force detector and walkdown detector function for all samples except last; force detector and bottom detector for last sample. The drill bit type tested by BOM is used. |
| Slow penetration rate results in waste fire. | Without probability of reactive waste MAF = 7.4E-5 UAF = 0.2 With consideration of probability of reactive waste for selected tanks (Appendix G) MAF <1.0E-6 | Operator fails to act. | Provide a 1 of 1 system for detection of penetration rate. Auto trip on slow penetration rate. One minute is available for operator to recognize slow penetration conditions and stop drilling. |

To mitigate this hazard, the materials used in the FG/RMCS and drill string were chosen to be compatible with the contents of the tank so that neither chemical action nor materials failure is expected as a result of expected or accidental contact with the waste. WHC performed a compatibility study on seal components as well as for the drill bit using an ammonia-saturated environment. Additionally, these materials have been used in actual waste. No significant degradation was observed. Study results summarized in Ref. 15 confirm these observations. Using aluminum tools around the open risers could have a potential for a drop rident. Administrative controls and work plans must be enforced to ensure that sonnel near an open riser do not use waste incompatible tools.

o minimize the potential for exothermic chemical reactions, aluminum pans and ntainers must not be used in contact with significant quantities of tank waste.

The drilling truck has some aluminum parts around the rotating platform. However, the tank waste is not normally handled above these parts.

A review of the compatibility of aluminum with the tank gases has been conducted and determined to be without detrimental effect and is therefore not considered further.²⁰

Based on References 15 and 16, pipe compounds, lubricants, seals and tapes in contact with the DS or tank waste that contain spark-inducing or waste-incompatible materials must not be used.

4.4.3. Crust Burn Caused by Drop Impact (Installation, Operation, and Removal)

In this section, the drop of the drill string or tools on the crust or waste surface is analyzed. The frictional spark aspect of this accident has already been discussed. In this section, crust reaction and the possibility of waste ignition are discussed. The drill string impact is a bounding impact accident.

Appendix G assesses the possibility of crust ignition and propagation following a drop of the drill string. Based on the analysis summarized in Appendix G, the maximum crust temperature produced by impact is 126°C, which is much less than the limit of 160°C given in Section 4.4.1.1 for preventing local exothermic reactions. Therefore, no local exothermic reactions are expected.

It is concluded that with the limited energy input possible and with the crustburning characteristics, propagation is impossible.

4.4.4. Flammable-Gas Ignition under the Waste Surface Caused by Friction From the Rotation or Penetration of the Drill Bit (Operation)

One of the safety concerns of sampling with the rotary-mode drill in flammable gas tanks is the ignition of the flammable gas in the waste by the frictional sparks created by the drill bit. The condition of waste in terms of hardness is not known before the operation. A possibility of penetrating a very hard waste layer in a tank exists. In addition, there may be some metal debris lost or dropped from the riser in the past. Hard materials such as rocks or metals can also exist in the waste. Thus, it is likely that the drill bit may strike against metal and other hard objects during drilling. Ignition caused by frictional sparks is evaluated by performing ignition testing in bounding conditions. Appendix T discusses the ignition testing requirements and acceptance criteria. The objective of the tests is to demonstrate that the operation of rotary-core drilling in a bounding frictional environment with bounding gas composition does not cause an ignition. Testing was performed by the Witwer¹⁰ and BOM personnel and showed no ignition.

The conditional frequencies of a fire accident resulting from exceeding the rotational speed and down force are estimated as 1.4E-5 and 9.4E-5/yr as indicated in Table 4.13. Controls are established to trip the drilling operation when the rotational speed and down force exceed 55 rpm and 750 lbf. There is no delay time for the trip except the delay time from the data acquisition system. If needed, the alarm points will be set at lower values. However, drilling must stop when the trip value is reached. The FG/RMCS operations must use only the drill bit type meeting the requirements listed in Appendix T.

Some of the early tests performed at the BOM showed some interesting results. The first series of tests were conducted at a down force of 1360 lbf and rotational speed of 65 rpm. Frictional tests performed with metal objects did not ignite the hydrogenoxygen mixture. Ignition was observed while testing an experimental bit that had carbon-steel pins embedded in the tooth region against rocks. No ignition was observed using a bit where the carbon-steel pins were not in the tooth region. The tooth region is defined as the material beneath the carbon steel blank. Therefore, this SA requires that no spark-inducing materials be located in the tooth region. Because the drill bit base also is made of carbon steel, an additional control requires that the drill bit must be replaced if drilling is shut down four times consecutively as a result of a low penetration rate, and if the cumulative penetration is <0.3 in. for the last three attempts. This conclusion was confirmed by Witwer (Ref. 10).

During frictional ignition tests using the current bit with assorted rocks and with a test period of 3 min, no ignition was observed. The last test was run to failure. Ignition occurred in the hydrogen-oxygen mixture within the sixth minute. It was postulated that the ignition occurred because the autoignition temperature at the teeth surface was reached. The test was repeated and ignition was observed at almost the same time. The bit teeth were not worn significantly to cause the carbon steel blank to be exposed to the rock. Additional tests without flammable gas were performed to determine the interface temperature. In one of the tests, a thermocouple was placed 1/8 in. beneath the assorted rock. The rock was not worn significantly; therefore, the temperature just beneath the rock could be measured. The rock temperature 1/8 in. beneath the surface was 236°C 6 minutes after testing has started. An infrared temperature probe was also used to determine the teeth surface temperatures up to 400°C were observed. These tests indicate that the autoignition temperature would be reached between 4 and 6 minutes.

The analytical and experimental program described in Appendix T provided the data to verify that the ignition observed in these tests was indeed caused by the interface temperature reaching the autoignition temperature (see Witwer¹⁰).

A penetration control is established in order not to operate in the frictional mode of cutting that results from worn teeth. However, this control is established to set the substrate temperature within the limits. In order to cause a hot spot on a substrate, the penetration rate control must be violated. Appendix F shows that with reasonable penetration, the substrate temperature is less than 160°C. Appendix F also concludes that as long as a nitrogen purge flow of 30 scfm is provided, down force is less than 750 lbf, or rotational speed is less than 55 rpm, the substrate temperature of hydrogen nitrous oxide ammonia mixtures, a triple failure needs to be obtained; penetration rate and purge flow; penetration rate and force; or penetration rate and rotational speed (see Table 4-14). These results were verified by using rocks in the envelope testing discussed in Appendix F (Witwer^{18,10}).

TABLE 4-14 FLAMMABLE-GAS IGNITION UNDER THE WASTE SURFACE CAUSED BY FRICTION FROM THE ROTATION OF THE DRILL BIT

| Accident | Frequency of Failure 1/yr | Dominant Failures | Controls Credited in Calculating UAF and MAF Values |
|--|---------------------------------|-----------------------------------|---|
| Drill bit hits a rock and spark leads to ignition of flammable | Without BOM data: | Rocks are contacted in the waste. | Auto trip occurs on slow penetration rate and a loss of or low N2 drill |
| gas and consequently waste fire. | UAF=0.2 | fails. | string purge gas flow results. |
| | With BOM Data | | |
| | MAF and UAF <1.0E-6 | | · · · · · · |

4.4.5. Summary of Waste Fire Accidents

The waste fire accidents considered various initiators. The overheating of waste or the drill bit and the ignition of flammable gas by friction that could be created by drilling are the dominant initiators. Experiments have been conducted to define safe operating parameters at which these two initiators could not cause ignition. The critical operating parameters are down force, rotational speed, nitrogen purge flow, and penetration rate. Failure to control these parameters could result in dome collapse. The total accident frequency is found to be 2.3E-4/yr.

The analysis of the waste ignition did not account for the probability of encountering a waste layer where the burn propagation is expected. The proportion of the reactive waste in each SST is considered in Appendix G. As shown in Appendix G, only some tanks satisfy the low fraction of reactive waste criterion. The accident frequency for these tanks becomes less than 1.0E-6/yr.

4.5. CHEMICAL REACTIONS AND CRITICALITY ACCIDENTS

4.5.1. Criticality Caused by Drilling (Operation and Removal)

The criticality issues associated with the FG/RMCS operations are addressed in Appendix R. The analysis given in Appendix R concludes that nuclear criticality within the single-shell flammable-gas tanks could not occur as a result of FG/RMCS.

4.5.2. Water Addition and Temperature Limits (Operation and Removal)

These accidents address the possibility of excessive water additions to the tank that would cause the level to rise above what is allowed by safety controls. It also addresses the effect of the water addition on the waste temperature and gas releases. In the limit, flooding could result from the release of tank materials into the environment caused by hydrostatic failure of the tank. Flooding from natural events is considered in Section 4.9.

Also, water addition can have several effects: (1) changes of the waste temperature, (2) release of gases, (3) changes in pH level, and (4) an increase in radiolysis activity. Appendix O discusses the effect of water additions. It is concluded that, if done within the specified limits, water addition does not cause a safety concern.

4.5.3. Waste Ignition and Melting (Operation)

Waste ignition as a result of exothermic chemical reactions is discussed in the waste fire accident category. In this section, the waste melting hazard is addressed. The estimation of the melting temperature of waste depends upon the waste composition. A mixture of sodium nitrate, sodium nitrite, and sodium acetate melts at 175°C (Ref. 21). In addition to these components, the waste contains sodium hydroxide, sodium aluminate, sodium carbonate, sodium sulfate, sodium phosphate, and a host of minor components. Adding additional salts lowers the melting point. Therefore, the melting point of waste is expected to be much lower than the 175°C observed for mixtures of sodium nitrate, sodium nitrite, and sodium acetate. The maximum allowable drill-bit temperature is 160°C, which is probably above the melting point of the waste. Thus, waste melting may occur during drilling.

One consequence of melting could be a gas release, but it is not expected that the gas release will be significant. At worst, the FG/RMCS perturbs the temperature in the neighborhood of the drill bit. Thus, the affected volume is small. Much larger gas releases are considered in the consequence analysis.

4-49

Melting has been proposed as a necessary precursor to runaway exothermic reactions. Melting would cause different chemical components to be brought into contact with each other. Although melting may be necessary for a runaway reaction, it is not the only consideration. Appendix G discusses the temperature limit to prevent runaway exothermic reactions based on experiments conducted on simulants, real wastes, and waste surrogates. These experiments already include the melting phenomenon. Therefore, melting as an initiator of runaway reactions is not considered as a separate hazard.

The drill bit holes can become totally or partially plugged during drilling. Such a scenario could cause overheating of the drill bit, resulting in waste ignition. Envelope testing discussed in Appendix F includes testing with plugged holes. It is demonstrated that the plugging would not cause overheating within the bounds of the operating parameters controlled in Section 6 of this SA.

4.5.4. Energy Addition to the Tank (Operation)

As discussed in Appendix O, energy additions or nitrogen cooling during FG/RMCS operations do not cause a safety concern, provided the operations are performed within the specified limits.

4.5.5. Impact-Sensitive Compounds

The presence of impact-sensitive explosive materials was evaluated by Martin²² and Beitel.²³ It was concluded that impact-sensitive compounds are not considered to be credible initiators for chemical reactions.

4.6. CONTAINMENT BREACH ACCIDENTS

A loss of confinement of the toxic and radioactive waste from a structural failure of the tank liner is an important safety issue. Installation, removal, and/or decontamination operations potentially constitute hazards to the structural integrity of the tank. The following subsections discuss the assessment made of these situations.

4.6.1. Excessive Static and Dynamic Tank Dome Loading (Installation, Operation and Removal)

Appendix N examines static and dynamic dome loading and its consequences. The static load capacity of the tank dome is monitored carefully, and an overload state that could precipitate a structural failure must be avoided. The equipment required on the surface of the tank to support FG/RMCS sampling operations qualifies as a live load (see Appendix N). The tank loads study permits a 50-ton live load on the tank dome. Even though the weight of all FG/RMCS equipment exceeds 50 tons, administrative procedures will limit the tank loading to less than 50 tons. Therefore, the dome loading must be controlled by the dome loading limits for SSTs as specified in OSD-T-151-00013 (Ref. 24).

The dome would be subjected to dynamic dome loads if a truck were to fall from the hydraulic jack or from a platform. Appendix N considers this scenario and analyzes the consequences of the dynamic loading caused by dropping the truck. It is concluded that the dome could withstand the impact force of the 30,000-lb truck dropping on it from the 3-ft-high platform.

Excessive vacuum in the dome is another accident identified as a result of exhauster operations. The exhauster design prevents the occurrence of excessive vacuum because the shutoff head is 14 in. w.g. A dome collapse would not occur until -15 in. w.g. Also, the inlet HEPA filter has a vacuum breaker to prevent excessive negative tank pressure. The vacuum breaker is set at about -4 in. w.g.

4.6.2. Penetration of Tank Bottom, Drill String Drop, and Drilling against the Liner (Installation, Operation and Removal)

The drill string is restrained from falling and impacting the tank bottom by the pneumatic foot clamp. After numerous sections of the drill string have been added, the suspended weight could cause the drill string to fall if the clamp is released because the force of gravity exceeds the frictional forces of the frisbee. Initially, the frictional force developed at the riser seal interface exceeds the string weight. The frictional force is produced by the rubber frisbee that girths the outside diameter of the drill string. This constant force eventually is overcome by the column weight as the drill rods are added. The drill string extending nominally halfway into the tank poses the largest hazard to the integrity of the tank bottom from an impact. The impact force that would occur if the drill string were released was evaluated.²⁵

Appendix N includes a ballistic impact analysis for different drop heights. The resulting stress level is expected to challenge the ultimate strength of the liner material despite the inherent ability of the carbon-steel liner to withstand higher stresses under high strain rate conditions. Because this accident can lead to a breach of the waste confinement, it is a major concern and must be avoided through the application of administrative controls. The very low expected accident frequency associated with this drop significantly ameliorates the risk. This mitigating consideration is discussed next.

The pneumatic foot clamp provides a positive grip around the string due to gravity and interference fit. Even when deenergized, the clamp provides positive restraint to prevent the string from falling. Although gravity and the interference fit prevent the string from falling under gravity, it does not preclude upward motion. Several events must occur for the drill string to fall and impact the tank bottom. Appendix E determines that the frequency of a DS drop impacting the tank bottom is 3.0E-5/yr. The consequence of this drop of the drill string is damage to the bottom liner resulting in a radioactive liquid release into the ground.

Sampling near the bottom of the tank introduces the possibility of contacting the tank bottom. Precautions are taken in setting up the drill string simply by recording

the string depth during installation. Nonetheless, in the rotary-mode sampling method, the drill string can produce an axial force of 750 lb. The force produced on the drill string by hydraulic action neither causes the drill string to penetrate the tank liner nor threatens the tank's structural integrity in any way. The drill string is a very long, slender column when fully extended to the tank bottom. This column buckles at a load less than the ram capacity force of 5370 lb. As an added protective feature, the FG/RMCS incorporates a bottom-contact sensor that reverses the hydraulic ram pressure used in the drill string operation. The pressure reversal causes the sampling operation to cease and pulls the drill bit away from the tank bottom.

Because the axial forces that can be produced by the FG/RMCS during sampling are significant, contacting the tank liner may pose an unacceptable risk to the tank's structural integrity. Therefore, a control is established not to drill closer than 3 in. from the tank bottom (see Appendix N for the basis of this control). Note that the failure probability of the walkdown and force plus the hydraulic bottom detector protection systems to detect the bottom are low and are estimated as 3.0E-5/yr.

4.6.3. Failure of the Drill String (Installation, Operation, and Removal)

The failure of the drill string during drilling is evaluated as a fire hazard. A torque and axial load are applied to the drill string during drilling. Buckling of the drill string from the application of axial force and torque as well as excitation of the drill string with natural frequency could yield drill string failure. The down force is limited to prevent drill bit overheating. The other limit on the down force should be based on the buckling limit. Appendix N discusses the structural buckling limit under various boundary conditions. The buckling limits as a function of the drill string length are given in Fig. N-1 of Appendix N.

Appendix N concludes that the drill string could not be excited with natural frequency as a result of torque only. Figure N-4 of Appendix N plots the first and second mode-resonance rotational speed as a function of the drill string length. The suggested range of rotational speeds is tabulated in Table N-2.

When the drill string length is less than 45 ft, down force must be limited to 750 lbf, and rotational speed is limited to 55 rpm. When the drill strength length is > 45 ft, administrative controls must be established to limit the speed to 40 rpm and down force to a maximum of 650 lbf.

Buckling loads and rotational speed to prevent resonance are established as required controls in Section 6. The frequency of the exceeding force limits is 9.4E-5/yr (Appendix E). The consequences of drill string breakage are several. The major concern is to cause a hot spot resulting in a fire in the dome (assumes that a failure occurs in the dome space and coincides with the existence of flammable gases). In Section 4.2.1, the fire accident caused by the drill string breakage was discussed. The

other concern is the operational difficulty of removing the broken drill string. This SA does not address the removal of damaged drill strings.

4.6.4. Riser Damage (Installation, Operation, and Removal)

The risers in the SSTs, although differing in diameter and overall length, share common support features. Each riser is anchored into the concrete tank dome with horizontal studs. The risers are not welded to the primary tank liner because the primary liner does not cover the tank dome. Failure of these studs conceivably would result in the riser being driven into the tank space. This prospect is an unlikely event for small-diameter risers.

Riser loads have been investigated by Miller.²⁶ The stress produced in the riser from this impact is directly proportional to the square root of the falling body's kinetic energy and the material modules of elasticity, and is inversely proportional to the riser volume. Thus, stress buildup in the riser pipe is not only influenced by the riser cross-sectional area but also by pipe length.

The drill string is to be placed on and removed from the riser during installation, operation, and removal activities. The new drill rods are lifted manually and threaded to the drill string. Even if they are dropped, their weight is only about 7 lb. An analysis of the drop height performed by Miller shows that a maximum height exists that depends upon the mass of the object to be dropped. The analysis further indicates that the scaling of the mass is a linear function of the ratio of the masses. Miller calculated a 1.7-in drop height for the studs to fail if an object with 363 kg (800 lb) is dropped. Using this information and the weight of a drill string, the scaling ratio is 800/7, or 114.3. The maximum lift height associated with a drill rod is 5.0 m (16.4 ft). This height is larger than the distance between the rotary platform and the riser. Considering all of the arguments above, riser damage from a drill-rod drop is not expected.

During operation, the drill string rotates in the conductive sleeve installed in the riser. Because the riser would be subjected to lateral loading, excessive stresses could be introduced on the riser's lower lip, upper flange, the drill string, or to the conductive sleeve, causing either or both to fail. Lateral loads can result from many factors, including misalignment or excessive lateral movement of the drill string, wasteberg impact on the submerged portion of the drill string, or attempts to remove a bent drill string. The failure of the riser lower lip has no consequence other than a possible sparking source as it plastically deforms from the load application; however, failure of the upper flange results in a breach of confinement from an inability to properly close the riser. The failure of the drill string has other consequences, including portions of the unit falling into the tank or deformation of the drill string is not in the scope of this SA. Dropping of a broken drill string could only occur when the drill string is submerged into the waste. There are controls on the downward load requiring it to be less than the buckling limit. If the

control is violated and this results in a failure, the drill string would not sink because the high load would not cause penetration. Therefore, the only consequence from this type of event is that the riser could become unusable and need to be sealed.

During operation, the hydraulic capacity of the drill head and shielded receiver can also apply a vertical load to the foot clamp and the riser when the foot clamp is closed. This sequence can result in riser damage, so that the riser may become unusable and need to be sealed.

Riser damage is a more likely event in the preinstallation phase. In this phase, the vehicles are positioned on the tank top and could inadvertently run into the riser, resulting in damage to the riser. This type of accident is considered an industrial accident, and standard WHC controls prevent this from occurring.

Riser damage resulting in the ignition of flammable gas is prevented by cessation of all activities upon receipt of a flammable-gas alarm.

Heavy equipment, such as casks and exhauster skids, are positioned on the tank top using cranes. Failure of the cranes during the positioning would cause riser damage. WHC safe practice standards provide the necessary protection for the proper handling of components.

4.6.5. Tank Wall Penetration (Operation)

A tank-wall penetration accident addresses the cases in which the drill string could damage the vertical section of the tank liner. This accident is considered to be incredible. Appendix N presents a calculation to estimate the bounding horizontal load to break the drill string at the riser. It was found that the drill string could be broken if a 610 lbf is applied. The deflection of the drill string is estimated as 5.7 in. Thus, before the drill string reaches the tank wall it should be broken already. When the drill string is broken, the debris would not cause a threat to the side wall because the waste in single-shell, flammable-gas tanks is very viscous. Both the consequences and the frequency of this accident are bounded by the bottom penetration accidents.

4.7. GAS RELEASES WITHOUT BURN

4.7.1. Toxic-Gas Releases (Installation, Operation, and Removal)

The toxic gases of interest are ammonia and nitrous oxide. The consequences of toxic-gas releases are divided into anticipated, unlikely, and very unlikely bins and are discussed in Section 5 of this SA.

4.7.2. Unfiltered Releases

The consequences of unfiltered waste releases associated with the gas-release events also are discussed in Section 5 of this SA. The consequences are analyzed in anticipated, unlikely, and very unlikely accident bins, consistent with the toxic-gas release accidents.

4.7.3. Steam Release and Moisture Removal from Tank (Operation)

Mass removal by evaporation caused by drill bit heating and by ventilation operations are discussed in Appendix O. Water can also be evaporated as a result of an energy release by local exothermic reactions. However, operating parameters prevent the drill bit temperature being close to the critical reaction temperature. The analysis shows that water removal during FG/RMCS operations is not a safety concern.

4.7.4. Undetected Gas Release (Operation)

In this accident, the concern is an undetected release path at the top of the tank. The consequences of this accident may be a fire as considered in the above tank fire accident category. Toxic-gas releases are bounded by the releases from the riser.

4.7.5. Nitrogen Additions (Installation, Operation, and Removal)

Nitrogen gas is added to the tank during the normal FG/RMCS operation. Nitrogen is added to waste at the rate of 0.3 to 100 scfm. Nitrogen addition to the waste is discussed in Appendix M. Although very unlikely, a nitrogen addition may cause a gas-release event. The exhauster must be operable during drilling.

4.8. SPILLS, RELEASES, AND HAZARDOUS MATERIAL AND RADIATION EXPOSURES

Because the tank waste contains toxic, radioactive, and hazardous materials, it is possible that workers or the environment could be exposed to harmful levels of hazardous materials.

It is possible that personnel could be exposed to toxic or hazardous materials in activities that require handling potentially contaminated components. This accident scenario does not consider exposures associated with a GRE or with fires, both of which have been discussed. The subject of this section is the potential for exposure associated with contaminated components and equipment and aerosol releases from spills.

4.8.1. HEPA Releases as a Result of Excessive Loading and HEPA Failures (Operation and Removal)

Appendix R evaluates the HEPA particulate loading based on results reported by Francis.²⁷ The total mass from out of the riser and to the HEPA filter is reported as 901 $_{\text{E}}$, a value based on continuous operation of the exhauster, a total drilling depth of 266 in., and a schedule of 40 min of drilling (19 in.) followed by a 60-min delay before starting the next drilling period. The results take into account settling of the aerosol created.

If the HEPA filters fail, a 1% release is assumed as given by Voice²⁸ who gave some experimental values for the release fractions of blowout incidents. The experimental estimated values are smaller than 1%. The release from HEPA would be less than 9 gr.

Differential pressures across the HEPA filters are measured. Failure to detect high HEPA loading and failure to shut down the exhauster may result in HEPA failure. The frequency of this accident is estimated as 1.6E-5/yr. (Appendix E, Table 4-15).

4.8.2. Exhauster Continuous Release after Filter Failure (Operatio and Removal)

This accident is considered in the unfiltered release accidents in Section 4.7.2. addresses the failure of HEPA filters following a continuous release with a flow rate of 250 scfm.

4.8.3. Inlet Duct Failure (Operation and Removal)

This accident addresses the release of aerosol from the riser where the breathing HEPA filter is installed. It is assumed that a HEPA filter fails, and a GRE occurs. The consequence of this accident is bounded by the gas releases from an open riser as discussed in Section 4.7.1.

4.8.4. Releases from Core Sampler Drops (Operation)

This accident addresses the releases to the atmosphere if the failed sampler (ball valve is open) is dropped from the shielded receiver. The maximum amount of waste in one sampler is 0.39 kg. A conservative airborne release fraction of 2E-4 is used (Appendix S). The aerosol release is 0.08 g for this accident. A control is established to inspect the sampler during sampler retrieval. The ball valve cannot be closed if the sampler is not inserted completely in the shielded receiver when held by the RLU.

The frequency of an accident depends upon the failure rate of the rotary ball valve and the probability of a drop or inadvertent opening of the ball valve. The frequency of this accident is determined as 8.6E-9/yr (Appendix E) as shown in Table 4.15. There are two administrative controls that are credited in obtaining this frequency.

TABLE 4-15SPILL ACCIDENTS CONTROLS

| | Frequency of | Dominant | Controls Credited in |
|--|---|---|--|
| Accident | 1/vr | Failures | MAF Values |
| Radioactive aerosols are released from the HEPA filter blowout. | MAF = 1.6E-5 (not a dome fire) UAF = 1.6E-5 | Excessive differential pressure occurs. | No controls are credited. |
| Spill results from dropped sampler in shielded receiver. (Similar accidents are spills from drops of sampler in x-ray machine or in cask but they are of lower frequency.) | MAF = 8.6E-9 (not a dome fire) UAF = 0.02 | RLU drops sampler. Sampler rotary valve fails, given drop of sampler. Ball valve in shielded receiver is inadvertently left open. | Controls over operation of RLU are used. Control to close ball valve when shielded receiver is used. |
| Waste from contaminated drill string interior is released above ground. | MAF = 6.2E-7 (not a dome fire) UAF = 2 | N2 hydrostatic systems fail to both drill string and shielded receiver. Contamination is not detected during removal of drill string sections. | Leak test N2 hydrostatic systems for both drill string and for shielded receiver are used. Unique connections for N2 hydrostatic systems for both drill string and shielded receiver are used. Verification is made of the N2 hydrostatic supply to both drill string and shielded receiver during activation of hydrostatic mode of N2 supply. Monitoring for radioactive contamination during removal of drill string sections is done. |

.

WHC-SD-WM-SAD-035, Rev. 0-a

| | Frequency of Failure | Dominant | Controls Credited in Calculating UAF and |
|--|--|---|---|
| Accident | 1/yr | Failures | MAF Values |
| Waste from contaminated drill string exterior is released above ground. | MAF = 4.4E-6 (not a dome fire) UAF = 2.0 | Wash of drill string exterior surface is ineffective before removal of drill string. Contamination is not detected during removal of drill string sections. | A water wash of the drill string exterior surface is done before removal of drill string. Monitoring is done for radioactive contamination during removal of drill string sections. |

TABLE 4-15 (cont)SPILL ACCIDENTS CONTROLS

4.8.5. **Open Sampler Spills (Operation)**

The concern is the spill of 0.39 kg of waste from the core sampler. For this to occur, the rotary ball valve needs to fail with the inadvertent opening of the shielded receiver valve. The release fraction is estimated as 2.0E-4 (Appendix S), and the release to the atmosphere would be 0.08 gr. The frequency of this accident is the same as for the previous accident and is estimated as 8.6E-9/yr (Appendix E).

4.8.6. Spill from Accumulated Waste in Core Barrel

If waste collects inside the core barrel or drill rods, handling could cause a release. Even though accumulated waste may be discovered by radiological monitoring, this would only aid workers in preparing to catch the spill that could follow removal. This release would be a spill. Considering that the length of the core barrel is 1.1 m (40 in.), and the diameter is 5.08 cm (2 in. i.d.), there could be 3.3 kg (7.3 lbm) of waste. The release fraction is recommended as 2.0E-4 (Appendix S). Thus, the material release would be 0.66 gr.

The frequency of this accident is dominated by failure of the nitrogen hydrostatic system and failure to detect contamination as indicated in Table 4.15. The frequency of spills is estimated as 6.2E-7/yr.

4.8.7. Ineffective Decontamination and Drop of Drill String

If decontamination is not effective and cannot be made effective, some spilling of contamination from the drill rod could be expected. The quantity of material on the exterior of the drill string can be no more than 0.37 kg assuming a 3-mm waste film on a drill rod. The frequency of spill is calculated by considering a failure to detect the contamination and an ineffective wash as indicated in Table 4.15. A spill of the

maximum quantity of material assumes no provisions to catch and contain the spill once the failed decontamination system is discovered. For conservatism, the spill quantity is assumed to be the maximum possible, however. The frequency of the ineffective decontamination and drop is estimated as 4.4E-6/yr. A release fraction of 2.0E-4 (Appendix S) gives 0.75 gm of waste to be released.

4.8.8. Radiation Exposures (Installation, Operation, and Removal)

It is possible for personnel to be exposed to excessive levels of radiation in association with proposed operations for the FG/RMCS. Several accidents are posed and discussed below. It is important to remember, however, that the exposure calculations are performed using the bounding nuclide inventory for all the SSTs (see Appendix R). Thus, such high exposure levels are not expected in any given SST.

4.8.8.1. Exposures from HEPA filters

Appendix R represents a calculation for possible radiation doses from HEPA filters. Using the code MicroShield (Appendix R), a filter geometry of 24 in. x 24 in. x 11.5 in. containing SiO₂ at a density of 0.1 g/cm^3 , it is found that the exposure at 1 cm from the surface is 317 mrem/h at the end of 100 minutes of operation. This exposure may be scaled with units of 100 minutes of time because it is based on an average mass flow rate. Controls are established to survey the radiation level from HEPA filter housing each shift during waste-intrusive operations.

4.8.8.2. Radiation Doses from the Shielded Receiver

Appendix R represents the radiation calculation from a full sample when it is in the SR. The bounding value of exposure calculated is 211 mrem/h at 1 cm from the surface of the receiver. It would be expected that there would be no samples that would actually produce this value because the samples would contain liquid that has a lower source strength. In addition, the source strength used here is the maximum found in any waste sample.

Administrative controls are established to survey radiation levels from the SR per WHC procedures.

4.8.8.3. Radiation from the X-ray Machine.

The requirements for operating the x-ray machine are explained in the document entitled "Requirements for Mobile Core Sample X-Ray Systems Number 2, 3, and 4" that the WHC provided.²⁹ Some of the safety requirements for shielding are listed below (taken from the reference). Radiation Shielding and Safety Requirements satisfy all the safety requirements dictated in American National Standards Institute (ANSI) N43.3, 21, CFR Section 1020, Chapter 246-243 WAC.³⁰

The maximum allowable dose rate at any accessible area 5 cm from the outside surface of the enclosure must not exceed 0.5 mrem/h. This is demonstrated or measured using calibrated equipment before final acceptance of the system.

Particular attention is placed on the dose rate at the Kamlock® adapter coupling. The rest of the safety requirements are available in the above-mentioned document.

4.8.8.4. Excessive Radiation Exposure as a Result of Personnel Being in a Direct Line of Sight to the Tank Waste (Installation and Removal)

Ion ing radiation from inside the tank caused by radioactive waste produces a nearly constant radiation field when viewed from an open riser. The maximum radiation field at the throat of an open 4-in. riser has been determined to be in excess of WHC limits for areas not considered to be high in radiation. Controls are required to ensure that workers do not receive excessive exposure during installation and removal, during which a direct line of sight to the tank wastes is possible. Because the FG/RMCS essentially fills the riser, there is little chance for workers to be exposed to a direct line of sight during operation. Only through the failure of workers to follow procedures limiting the exposure or through improper development of work plans can workers be exposed to excessive levels of waste on the FG/RMCS if the decontamination system is not effective. This issue is discussed in the next section.

4.8.8.5.Excessive Personnel Radiation Exposure during FG/RMCS Decontamination The FG/RMCS may be contaminated by tank waste during removal or operation phases. The amount of waste on the FG/RMCS can increase the radiation caused by tank shine. Workers are required to aid in the assessment of decontamination effectiveness during removal actions. Therefore, workers could be exposed to the combined fields of tank waste on the drill string and tank shine.

A hand-held radiation monitor is used by radiological health technicians to ensure that radiation levels are acceptable for unrestricted work. Protective equipment and other work limitations (such as work duration) are specified by tank radiological and industrial health authorities, according to established procedures. All open-riser work requires respiratory protection, as indicated in Section 6 of this SA.

The likelihood that workers could be exposed to this high level of radiation depends on whether the decontamination system and handling procedures are effective. WHC has experience with other similar activities,^{31, 32} and what can be expected for the FG/RMCS should not result in increased personnel risk. In general, the water decontamination method is successful in removing hazardous levels of waste from the exterior surfaces of components, particularly if the components are designed to facilitate decontamination, as in the FG/RMCS.

In summary, workers could be exposed to high levels of radiation; however, this is a very unlikely occurrence because the controls are strictly enforced and monitored, and workers are highly trained in both radiation protection techniques and the types of exposure possible at the tank farm.

4.9. EXTERNAL EVENTS

In this section, the accidents associated with external events are evaluated. It is concluded that the risk associated with lightning, wind, fire, earthquake, tornado, flood, volcanoes, and dust devils are acceptable as long as the activities are performed in accordance with the available administrative controls. Before and periodically during FG/RMCS operations, at the discretion of the person in charge (PIC), external event status must be verified with meteorological stations or appropriate authorities. WHC is required to identify an acceptable meteorological station and acceptable verification authority.

4.9.1. Lightning

The Hanford Site is not a major thunderstorm area. On average, only ~10 thunderstorm days per year are recorded at the Hanford Site, although this number has varied from a low of 3 to a high of 23 thunderstorm days per year. Thunderstorms theoretically can occur during any month of the year; however, they occur most frequently from April through September. The largest number of thunderstorm days recorded in a single month is eight, a number of days that has occurred in both June and August. Large differences in electric potential can occur during thunderstorms, which in turn can lead to lightning strikes. In general, ~20% of lightning strikes are cloud-to-ground/ground-to-cloud discharges. Lightning strikes in the summer occasionally have ignited range fires in the Hanford Site region.

A lightning strike could initiate a hydrogen burn during FG/RMCS operations. Lightning is a generic source for all the fire accident categories analyzed in this section. Lightning strikes are discussed in this section only, and this section will be referred to when discussing the same issue in other parts of Section 4.

Lightning hazards have been extensively treated by Cowley.³³ For background and detailed treatment, the reader is referred to this reference. Relevant features of that analysis that are pertinent to the FG/RMCS operations are discussed below.

- Lightning strikes at Hanford have a frequency of 1 per km²/yr.
- A rough rule of thumb is that a grounded vertical rod attracts any lightning that would have struck the ground in a circle around the rod with a radius equal to the height of the rod.

Conclusions from the analysis are

• Where the object struck is not directly attached to a tank riser, the lightning could not be an ignition source for flammable-gas mixtures in the tank.
• Strikes to control panels/enclosures, power lines, backup supplies, structures, and ventilation systems result in accident situations that are insignificant, covered by interim operational safety requirements, or analyzed in the interim safety basis (ISB).

Umman says that the structure of an SST, the large quantity of rebar in the concrete, and the fact that the tank is buried gives an SST some of the properties of a Faraday cage. However, the lack of a complete steel inner liner makes the SST less effective than a double-shell tank (DST) as a Faraday cage. Construction drawings do not indicate that any effort was made during construction to make electrical connections between the risers on the SSTs and the rebar in the concrete. Therefore, there are electrically noncontinuous paths through the tank that can result in arcing. The paths are as follows:

- Arcing between inadequately bonded equipment extending through risers and the risers.
- Arcing between the riser or equipment and the rebar in the concrete dome.
- Arcing between an equipment item and the waste surface.
- Arcing at bolted flanges.
- Ohmic heating of the waste by conduction of a current into the waste by an equipment item such as a thermocouple (TC) tree.

In his report, Cowley assumes that any lightning strike on a riser or riser-mounted equipment will have enough energy to ignite a flammable mixture. To minimize the probability of a lightning strike, all FG/RMCS equipment (including the drill string) must be bonded to the riser or inserted into the tank, and must be grounded, using existing or an alternate grounding methods consistent with the principals outlined in Cowley's report (Ref. 33). WHC has developed and implemented appropriate grounding/bonding procedures. Equipment not attached to the riser or inserted into the tank must be grounded following adequate WHC grounding and bonding controls consistent with the NFPA requirements.

Lightning could strike the exhauster or any other auxiliary system or drill truck during a thunderstorm. This cannot be prevented. Therefore, no FG/RMCS wasteintrusive activities can proceed during thunderstorms, or when thunderstorm activity or lightning strikes are reported or predicted within a 50-mile radius. This reduces the likelihood that drilling would cause flammable-gas conditions in the dome when there is a high probability of a lightning strike. The drill string could be left in the riser for different reasons. When it needs to be left in the riser, and the truck is not manned, the drill string needs to comply with the grounding requirements developed by WHC.

Grounding systems and bonding need to be verified as being adequate before initiation of operations. If the strike occurs, a fire is assumed to result. Assumptions and event trees in estimating the frequency of fire accidents caused by lightning are given in Appendix E.

The low probability of a lightning strike during a GRE and during FG/RMCS operations combine to make a lightning-induced burn frequency beyond extremely unlikely.

4.9.2. Wind

At the Hanford Site, the severe-weather phenomenon that occurs most frequently and has the greatest impact is dust storms (Ref. 34). The maximum recorded peak gust at 15.24 m (50 ft) above the ground was 35.8 m/s (80 mi/h), which occurred in January 1972. A 100-yr return period peak gust of 38.4 m/s (86 mi/h) has been calculated at the 15.24-m (50-ft) elevation.

Procedures governing FG/RMCS operations constrain operations to be performed only when the sustained wind velocity is less than 11.1 m/s (25 mi/h). This limit reduces the risk of wind-caused accidents.

4.9.3. Fire

The tank farm area has been subject to range fires in the past. Procedures governing FG/RMCS operations involving installation, operation, and removal prevents operations if a range fire is within 8.0 km (5 mi) of the tank being sampled. The FG/RMCS operation may be terminated very quickly once a fire is detected within an 8-km (5-mi) radius. In addition, procedures require limitations on flammable material in the vicinity of the tank, and fire extinguishers need be on hand before beginning installation or removal operations.

4.9.4. Earthquake

An earthquake occurring during FG/RMCS installation and removal operations does not change the risk assessed in this SA significantly because the operation period of FG/RMCS is short. The low probability of a large earthquake occurring during FG/RMCS installation, operation, and removal operations is acceptable risk, and accidents that could result as a consequence of this initiator are not considered in this report.

The design-basis earthquake has a return period of 7500 yr (frequency > 1×10^{-4} /yr). SSTs are designed to withstand a 0.2-g zero-period acceleration (ZPA), but there is a high probability that a stronger earthquake with a magnitude exceeding 0.2 g may produce structural failure of the dome.

4-63

4.9.5. Tornado

Tornadoes are very rare in the vicinity of the Pasco Basin and, on average, the state of Washington experiences just over one tornado each year. Reference 35, "Design and Evaluation Guidelines for Department of Energy Facilities Subjected to Natural Phenomena Hazards"³⁶ are used for natural phenomena loadings for nonreactor facilities. This document states that tornadoes are not considered a viable threat or hazard at the Hanford Site and are eliminated as an external initiating event. Nevertheless, in Section 6 of the SA, a control restricts FG/RMCS activities when tornado activity has been predicted within the next 8 hours within a 50 mile radius.

4.9.6. Flood and High Rainfall

Reference 36 describes probable maximum flooding of the 200-West Area—streams, rivers, surge and seiche flooding, flooding from ice dams, tsunamis, and flooding from dam failures. The worst-case flood was found to be caused by a hypothetical catastrophic failure of the Grand Coulee Dam. In this case, it was concluded that the floodwaters (elevation ~140.2 m (460 ft) would be well below the elevation of the 200-West Area [elevation ~213.4 m (700 ft)]. Thus, flooding was eliminated as an external initiating event. In the case of excessive rainfall, the operations are required to be stopped. Necessary controls are established in Section 6.

4.9.7. Volcanism

Volcanic hazards of the 200-West Area were examined in Ref. 36. In this report, it is stated that there is no evidence of lava flows, ash flows, or mudflows from Cascade Range volcanoes having reached the Pasco Basin during the Quaternary period. The nearest Cascade Range volcano is more than 96.5 km (60 mi) from the Hanford Site. With the exception of mudflows and airborne ejecta, most eruption products remain within 48.3 km (30 mi) of Cascade volcanoes. At increasing distances from the eruptive vent, flows of lava, debris, and mud tend to become more confined to existing drainage channels. Because no streams flow directly from the Cascade Range volcanoes to the Hanford Site, this type of volcanic product is not considered likely at Hanford.

A volcanic ashfall event is considered to be a potential natural phenomenon occurrence at the Hanford Site. The design criterion for ashfall loading on Hanford Safety Class I (SC I) structures, systems, and components is an uncompacted ashfall of 11.4 cm (4.5 in.). This ashfall should cause no undue additional loadings on critical structures.

The low probability of a volcanic event, combined with the minor consequences of an ashfall, are not considered to pose a significant hazard to the proposed operations. The installation and removal procedures contains a requirement to suspend operations in the event of volcanic activity that could lead to an ashfall.

4.9.8. Dust Devils

A dust devil is a localized wind pattern that moves in a circular motion which spawns and decays quickly and travels at relatively low velocities. Dust devils occur frequently in the Hanford Site areas during the daytime in the summer months. Wind speeds are believed to be in the low tens of miles per hour. The major concerns related to dust devils entering the site are (1) the effects on lifting the equipment, (2) the control of contaminants during installation and removal, and (3) the effect on personnel around an open riser from contaminant dispersal. It is difficult to predict the effect of a dust devil moving over the open riser because the dust devil wind speeds are unknown and because of the variety of operations being performed. To prevent adverse effects from dust devils, FG/RMCS activities must be suspended at the discretion of the PIC when there is observable dust devil activity.

4.10. **REFERENCES**

- A. D. Swain, "Accident Sequence Evaluation Program Human Reliability Analysis Procedure," US Nuclear Regulatory Commission report NUREG-CR-4772 (February 1987).
- 2. J. Robinson, "Description of Conductive Flexible Hose for Exhauster," Westinghouse Hanford Company letter to J. Darby (January 31, 1996).
- 3. "Standard Reliability Data," IEEE Standard 500, Published by the Institute of Electrical and Electronics Engineering, Inc. (1984).
- 4. J. Robinson, "Exhauster Duct Resistivity Measurements," Westinghouse Hanford Company letter 75230-96-012 to Cetin Unal (June 4, 1996).
- 5. J. E. Rasmussen, "Approval Requested for New Potential-to-Emit for the Rotary Mode Core-Sampling System Exhauster," Westinghouse Hanford Company letter (enclosure WHC Environmental Compliance Document, WHC-CM-7-5, Rev. 1) to Mr. A. W. Conklin and Mr. Jerry Leitch (December 1995)
- 6. Air Movement and Control Association, Standard 99-0401-86 (1986).
- 7. R. N. Johnson "Sparking Potential of Aluminum vs. Aluminum," Pacific Northwest National Laboratory letter to J. Robinson (February 10, 1996).
- 8. J. Smalley, Westinghouse Hanford Company, "Design Review Meeting for Purge Gas System," Purge Gas Pneumatic Diagram, H-2-690009 (December 1995).

- J. C. Krok and J. E. Shepherd, "Electrical and Frictional Spark Ignition of N₂O-H₂-N₂-Air Mixture," Los Alamos National Laboratory report LANL-WT-800 (June 1993).
- 10 K. S. Witwer, "Test Report for Ignitability Testing," Westinghouse Hanford Company report WHC-SD-WM-TRP-257 (June 1996).
- 11. C. Unal, "Burn in the Shielded Receiver," Los Alamos National Laboratory Calc-Note TSA10-CN-WT-SA-TH-099 (January 1996).
- 12. J. Smalley, Westinghouse Hanford Company personal communication, (November 1995).
- 13. R. A. Huckfeldt, "NFPA Barriers," Westinghouse Hanford Company internal memorandum 75230-96-006, to D. Hamilton, Characterization Equipment Design Project (February 27, 1996).
- 14. C. M. Keller, "Push Mode Core Sample Test," Westinghouse Hanford Company Report WHC-SD-WM-TRP-048 (January 23 1991).
- 15. A. J. Kostelnik, "Rotary Mode Core Sample Truck Material and Component Compatibility with Tank Waste," letter (attachment Westinghouse Hanford Company internal memo, 75230-96-001 R2) to Dennis Hamilton (February 5, 1996).
- L. F. Ermold, "Actions in Response to Spark Reported During Work on Tank 241-U-109," Westinghouse Hanford Company letter WHC-9650352, (enclosure WHC Occurrence Report, RL-WHC-TANKFARM-1996-0009) to S. Burnam (January 23, 1996).
- 17. H. Sullivan, et al., "A Safety Assessment for Proposed Pump Mixing Operations to Mitigate Episodic Gas Releases in Tank 242-SY-101," Los Alamos National Laboratory report LA-UR-92-3196, Rev. 14 (1995).
- 18. K.S. Witwer, "Envelope Testing for Rotary-mode Core Sampling," Westinghouse Hanford Company reprot WHC-SD-WM-TRP-252 (June 1996).
- 19. J. F. McCormick, "Test Report For Drill String Seal Pressure Test," WHC-SD-WM-TRP-DRAFT, Rev. 0 (January 1996).
- 20. C. Shepard and L. Pederson, "Reaction of Aluminum with Gases Released by Wastes in Tank 101-SY," Los Alamos National Laboratory memorandum LANL-WT-1152, to Troy Stokes (April 11, 1994).

- 21. D. Scheele, R. Shell, J. Sobolik, and L. Burger, "Organic Tank Safety Project: Preliminary Results of Energetics and Thermal Behavior Studies of Model Organic Nitrate and/or Nitrite Mixtures and a Simulated Organic Waste," Pacific Northwest National Laboratory report PNL-10213 (August 1995).
- 22. E. C. Martin, "Complexant Stability Investigation Task 2-Organic Complexants," Pacific Northwest National Laboratory report PNL-5453 (1985).
- 23. G. A. Beitel, "Exothermic Potential of Sodium Nitrate Salt Cake," Atlantic Richfield Hanford Company report ARH-LD-163 (1977).
- 24. V. C. Boyles, "Operating Specifications for Single-Shell Waste Storage Tanks," Westinghouse Hanford Company report OSD-T-151-00013, Rev. D-4 (1994).
- 25. W. O. Miller, "Retained Gas Sampler Drill String Drop Analysis," Los Alamos National Laboratory Calc Note TSA-6-CN-WT-SA-075 (November 30, 1994).
- 26. W. O. Miller, "Riser Impact Loads," Los Alamos National Laboratory Calc-Note TSA6-CN-WT-SA-ST-063 (March 1994).
- 27. P. M. Francis, "Aerosol Study for the Rotary Mode Exhauster," Westinghouse Hanford Company report WHC-SD-WM-ES-225, Rev. 0 (1993).
- J. D. Voice, "Tank Farm Accelerated Safety Analysis; Tank Ventilation System," Westinghouse Hanford Company report WHC-SD-WM-SARR-018, Rev. 0 (June 29, 1994).
- 29. "General Safety Standard for (ANSI NBS 114) Installations Using Non-Medical X-ray and Sealed Gamma-Ray Sources, Energies up to 10 MeV," ANSI N43.3 (American National Standards Institute, Inc., New York, 1993).
- 30. "Section Performance Standard for 1020-1991 Ionizing Radiation Emitting Products," Code of Federal Regulations (CFR), 21 CFR
- 31. Westinghouse Hanford Company report WHC-SD-WM-HSP-022, Rev. 0A (April 12, 1993).
- 32. "Hanford Site Tank Farm Facilities Interim Safety Basis," Westinghouse Hanford Company report WHC-SD-WM-ISB-001, Rev. 0 (August 1, 1993).
- 33. W. L. Cowley, "Evaluation of Hazards From Lightning Strikes to Tank Farm Facilities," Westinghouse Hanford Company report WHC-SD-WM-SARR-027, Rev. 0 (June 29, 1994).

- 34. D. I. Herborn, et al., "Hanford Waste Vitrification Plant Preliminary Safety Analysis Report," Westinghouse Hanford Company report WHC-EP-0250, Rev. 0 (May 1991).
- 35. "General Design Criteria," US Department of Energy Order 6430.1A (1989).
- 36. Kennedy, et al., "Design and Evaluation Guidelines for Department of Energy Facilities Subjected to Natural Phenomena Hazards," Department of Energy report UCRL-15910 (1990).

remain within 48.3 km (30 mi) of Cascade volcanoes. At increasing distances from the eruptive vent, flows of lava, debris, and mud tend to become more confined to existing drainage channels. Because no streams flow directly from the Cascade Range volcanoes to the Hanford Site, this type of volcanic product is not considered likely at Hanford.

A volcanic ashfall event is considered to be a potential natural phenomenon occurrence at the Hanford Site. The design criterion for ashfall loading on Hanford Safety Class I (SC I) structures, systems, and components is an uncompacted ashfall of 11.4 cm (4.5 in.). This ashfall should cause no undue additional loadings on critical structures.

The low probability of a volcanic event, combined with the minor consequences of an ashfall, are not considered to pose a significant hazard to the proposed operations. The installation and removal procedures contains a requirement to suspend operations in the event of volcanic activity that could lead to an ashfall.

4.9.8. **Dust Devils**

A dust devil is a localized wind pattern that moves in a circular motion which spawns and decays quickly and travels at relatively low velocities. Dust devils occur frequently in the Hanford Site areas during the daytime in the summer months. Wind speeds are believed to be in the low tens of miles per hour. The major concerns related to dust devils entering the site are (1) the effects on lifting the equipment, (2) the control of contaminants during installation and removal, and (3) the effect on personnel around an open riser from contaminant dispersal. It is difficult to predict the effect of a dust devil moving over the open riser because the dust devil wind speeds are unknown and because of the variety of operations being performed. To prevent adverse effects from dust devils, FG/RMCS activities must be suspended at the discretion of the PIC when there is observable dust devil activity.

4.10. **REFERENCES**

- 1. A. D. Swain, "Accident Sequence Evaluation Program Human Reliability Analysis Procedure," US Nuclear Regulatory Commission report NUREG-CR-4772 (February 1987).
- 2. J. Robinson, "Description of Conductive Flexible Hose for Exhauster," Westinghouse Hanford Company letter to J. Darby (January 31, 1996).
- 3. "Standard Reliability Data," IEEE Standard 500, Published by the Institute of Electrical and Electronics Engineering, Inc. (1984).

- 4. J. Robinson, "Exhauster Duct Resistivity Measurements," Westinghouse Hanford Company letter 75230-96-012 to Cetin Unal (June 4, 1996).
- 5. J. E. Rasmussen, "Approval Requested for New Potential-to-Emit for the Rotary Mode Core-Sampling System Exhauster," Westinghouse Hanford Company letter (enclosure WHC Environmental Compliance Document, WHC-CM-7-5, Rev. 1) to Mr. A. W. Conklin and Mr. Jerry Leitch (December 1995)
- 6. Air Movement and Control Association, Standard 99-0401-86 (1986).
- 7. R. N. Johnson "Sparking Potential of Aluminum vs. Aluminum," Pacific Northwest National Laboratory letter to J. Robinson (February 10, 1996).
- 8. J. Smalley, Westinghouse Hanford Company, "Design Review Meeting for Purge Gas System," Purge Gas Pneumatic Diagram, H-2-690009 (December 1995).
- J. C. Krok and J. E. Shepherd, "Electrical and Frictional Spark Ignition of N₂O-H₂-N₂-Air Mixture," Los Alamos National Laboratory report LANL-WT-800 (June 1993).
- 10 K. S. Witwer, "Test Report for Ignitability Testing," Westinghouse Hanford Company report WHC-SD-WM-TRP-257 (June 1996).
- 11. C. Unal, "Burn in the Shielded Receiver," Los Alamos National Laboratory Calc-Note TSA10-CN-WT-SA-TH-099 (January 1996).
- 12. J. Smalley, Westinghouse Hanford Company personal communication, (November 1995).
- 13. R. A. Huckfeldt, "NFPA Barriers," Westinghouse Hanford Company internal memorandum 75230-96-006, to D. Hamilton, Characterization Equipment Design Project (February 27, 1996).
- 14. C. M. Keller, "Push Mode Core Sample Test," Westinghouse Hanford Company Report WHC-SD-WM-TRP-048 (January 23 1991).
- 15. A. J. Kostelnik, "Rotary Mode Core Sample Truck Material and Component Compatibility with Tank Waste," letter (attachment Westinghouse Hanford Company internal memo, 75230-96-001 R2) to Dennis Hamilton (February 5, 1996).
- 16. L. F. Ermold, "Actions in Response to Spark Reported During Work on Tank 241-U-109," Westinghouse Hanford Company letter WHC-9650352, (enclosure

2

WHC Occurrence Report, RL-WHC-TANKFARM-1996-0009) to S. Burnam (January 23, 1996).

- 17. H. Sullivan, et al., "A Safety Assessment for Proposed Pump Mixing Operations to Mitigate Episodic Gas Releases in Tank 242-SY-101," Los Alamos National Laboratory report LA-UR-92-3196, Rev. 14 (1995).
- 18. K.S. Witwer, "Envelope Testing for Rotary-mode Core Sampling," Westinghouse Hanford Company reprot WHC-SD-WM-TRP-252 (June 1996).
- 19. J. F. McCormick, "Test Report For Drill String Seal Pressure Test," WHC-SD-WM-TRP-DRAFT, Rev. 0 (January 1996).
- 20. C. Shepard and L. Pederson, "Reaction of Aluminum with Gases Released by Wastes in Tank 101-SY," Los Alamos National Laboratory memorandum LANL-WT-1152, to Troy Stokes (April 11, 1994).
- 21. D. Scheele, R. Shell, J. Sobolik, and L. Burger, "Organic Tank Safety Project: Preliminary Results of Energetics and Thermal Behavior Studies of Model Organic Nitrate and/or Nitrite Mixtures and a Simulated Organic Waste," Pacific Northwest National Laboratory report PNL-10213 (August 1995).
- 22. E. C. Martin, "Complexant Stability Investigation Task 2-Organic Complexants," Pacific Northwest National Laboratory report PNL-5453 (1985).
- 23. G. A. Beitel, "Exothermic Potential of Sodium Nitrate Salt Cake," Atlantic Richfield Hanford Company report ARH-LD-163 (1977).
- 24. V. C. Boyles, "Operating Specifications for Single-Shell Waste Storage Tanks," Westinghouse Hanford Company report OSD-T-151-00013, Rev. D-4 (1994).
- 25. W. O. Miller, "Retained Gas Sampler Drill String Drop Analysis," Los Alamos National Laboratory Calc Note TSA-6-CN-WT-SA-075 (November 30, 1994).
- 26. W. O. Miller, "Riser Impact Loads," Los Alamos National Laboratory Calc-Note TSA6-CN-WT-SA-ST-063 (March 1994).
- 27. P. M. Francis, "Aerosol Study for the Rotary Mode Exhauster," Westinghouse Hanford Company report WHC-SD-WM-ES-225, Rev. 0 (1993).
- J. D. Voice, "Tank Farm Accelerated Safety Analysis; Tank Ventilation System," Westinghouse Hanford Company report WHC-SD-WM-SARR-018, Rev. 0 (June 29, 1994).

- 29. "General Safety Standard for (ANSI NBS 114) Installations Using Non-Medical X-ray and Sealed Gamma-Ray Sources, Energies up to 10 MeV," ANSI N43.3 (American National Standards Institute, Inc., New York, 1993).
- 30. "Section Performance Standard for 1020-1991 Ionizing Radiation Emitting Products," Code of Federal Regulations (CFR), 21 CFR
- 31. Westinghouse Hanford Company report WHC-SD-WM-HSP-022, Rev. 0A (April 12, 1993).
- 32. "Hanford Site Tank Farm Facilities Interim Safety Basis," Westinghouse Hanford Company report WHC-SD-WM-ISB-001, Rev. 0 (August 1, 1993).
- 33. W. L. Cowley, "Evaluation of Hazards From Lightning Strikes to Tank Farm Facilities," Westinghouse Hanford Company report WHC-SD-WM-SARR-027, Rev. 0 (June 29, 1994).
- 34. D. I. Herborn, et al., "Hanford Waste Vitrification Plant Preliminary Safety Analysis Report," Westinghouse Hanford Company report WHC-EP-0250, Rev. 0 (May 1991).
- 35. "General Design Criteria," US Department of Energy Order 6430.1A (1989).
- Kennedy, et al., "Design and Evaluation Guidelines for Department of Energy Facilities Subjected to Natural Phenomena Hazards," Department of Energy report UCRL-15910 (1990).

5.0. CONSEQUENCES OF ACCIDENTS

In this section, the consequences of accidents are discussed for the installation, operation, and removal of rotary-mode core sampling (RMCS) equipment in single-shell tanks (SSTs) on the Flammable Gas Tanks (FG tanks) or those tanks recommended by the contractor to be included on the FG tanks, hence referred to as FG/RMCS operations. Section 5.1 presents the Risk Guidelines (RGs). Section 5.2 contains a summary of bounding accidents identified in Section 4. In Sections 5.3 through 5.5 the bounding consequences of different accidents are quantified. The consequences are then compared with the RGs in Section 5.6.

Long-term radiological and toxicological consequences of material releases into the soil are not quantified in this safety assessment (SA). Accidents that may lead to a containment breach and to a release of material into the soil are identified and discussed in Section 4. The mitigative features of such accidents also are included in the discussion provided in Section 4. Environmental risk associated with the rotary mode core sampling operations in flammable tanks (FG/RMCS) is not further discussed in this section.

5.1. RISK GUIDELINES

This section defines the RGs used for comparing the consequences of the accidents analyzed in this SA. These criteria are divided into two major areas: radiological committed effective dose equivalent (CEDE) and toxic material exposure.

5.1.1. Radiological Risk Criteria

The RGs used in this SA for radiological consequences are obtained from Ref. 1. Reference 1 requires that the radiological RGs contained in Revision 3 of Ref. 2 be used. Revision 4 of Ref. 2 recommends a less conservative set of radiological RGs. The more conservative set proposed in Ref. 1 and in Revision 3 of Ref. 2 were used.

To use the RGs, a frequency and consequence for the particular accident must be analyzed. The best estimates of the frequencies are given in Section 4, and the basis of the frequencies is discussed in Appendixes D, E and L. The present analysis was performed using conservative modeling assumptions, and the conservative consequences have been judged using best estimates for the frequencies. Ref. 1 provides the frequency-dependent, radiological dose limits shown in Table 5-1a. Also, Ref. 1 specifies that if a specific single-point frequency is used, the guidelines are to be applied as curves. However, if a qualitative frequency ranking is used, the corresponding consequence limit (in rem) must be used equal to the lowest value for that frequency range.

The less restrictive alternative guidelines contained in Ref. 2 are shown in Table 5-1b for comparison purposes. Note that the guidelines in Ref. 2 are to be applied as a step-function.

5.1.2. Toxicological Risk Guidelines

Toxicological acceptance criteria have been developed from the guidelines presented in Ref. 2. In the referenced material, the onsite and offsite concentration limits are given in terms of Emergency Response Planning Guidelines (ERPGs) and Permissible Exposure Limit—Time-Weighted Average (PEL-TWA). The toxicological RGs are summarized in Table 5-2.

ERPG-1 is the maximum airborne concentration below which it is believed that nearly all individuals could be exposed for up to 1 h without experiencing anything other than mild transient adverse health effects or perceiving a clearly defined objectionable odor.

ERPG-2 is the maximum airborne concentration below which it is believed that nearly all individuals could be exposed for up to 1 h without experiencing or developing irreversible or other serious health effects or symptoms that could impair an individual's ability to take protective action.

TABLE 5-1A RADIOLOGICAL RISK GUIDELINES (REF. 1, WHC-CM-4-46, REV. 3)

| Frequency | Frequency Range | On-site CEDE | Off-site CEDE | |
|--------------------|--------------------------------------|--------------|---------------|--|
| Category | (yr¹) | (rems) | (rems) | |
| Anticipated | 10 ⁻¹ to 10 ⁻² | 1 to 5 | NA | |
| Anticipated | 1 to 10-2 | NA | 0.01 to 0.5 | |
| Unlikely | 10 ⁻² to 10 ⁻⁴ | 5 to 25 | 0.5 to 4 | |
| Extremely Unlikely | 10 ⁻⁴ to 10 ⁻⁶ | 25 to 100 | 4 to 25 | |

TABLE 5-1BRADIOLOGICAL RISK GUIDELINES (REF. 2, WHC-CM-4-46, REV. 4)

| Frequency | Frequency Range | On-site EDE * | Off-site EDE | |
|--------------------|--------------------------------------|---------------|--------------|--|
| Category | (yr1) | (rems) | (rems) | |
| Anticipateda | 1.0 to 10 ⁻² | 5 | 0.5 | |
| Unlikely | 10 ⁻² to 10 ⁻⁴ | 25 | 5 | |
| Extremely Unlikely | 10-4 to 10-6 | 100 | 25 | |

* EDE: effective dose equivalent

TABLE 5-2 TOXICOLOGICAL RISK GUIDELINES

| Frequency | Frequency Range | On-site | Off-site |
|--------------------|--------------------------------------|----------|----------|
| Category | (yr1) | Limit | Limit |
| Anticipated | 1 to 10-2 | ≤ ERPG-1 | PEL-TWA |
| Unlikely | 10 ⁻² to 10 ⁻⁴ | ≤ ERPG-2 | ≤ ERPG-1 |
| Extremely Unlikely | 10 ⁻⁴ to 10 ⁻⁶ | ≤ ERPG-3 | ≤ ERPG-2 |

ERPG-3 is the maximum airborne concentration below which it is believed that nearly all individuals could be exposed for up to 1 h without experiencing or developing life-threatening effects.

When specific values for PEL-TWA and ERPGs are not available, alternative concentration guidelines also are included in Ref. 2.

Based on Ref. 2, these guidelines are applied using a step-function within the specified frequency range. In this SA, the following RGs are used:

- Ammonia: Ammonia (NH₃) is a corrosive or irritant and it is known to exist in large quantities in the waste gas.
- Nitrous Oxide: Nitrous oxide (N₂O) is a central nervous system depressant, and it is known to exist in large quantities in the waste gas.

August 8, 1996

Waste Material: The human reception of solid/liquid waste material results in systemic toxicity and corrosive effects. Various accidents analyzed in this SA (unfiltered releases, burns, spills, etc.) result in a wide range of waste please quantities. Note that all the releases that result in radiological consequences also yield toxicological consequences as a result of an exposure to the same material.

The results of the vapor space sampling program were reviewed.³ Major toxic gas a that are found in the dome space of the presently defined flammable-gas tanks a ammonia and nitrous oxide. Other gases are found in trace quantities and do not pose a concern. However, it was recognized that the data contained in Ref. 3 are limited, and all tanks of interest are not covered. Consequently, the toxic gas evaluation was made a checklist item (Section 7). Thus, it is required that the potential for toxic gases other than ammonia and nitrous oxide must be Is-evaluated before FG/RMCS operations may be initiated in a specific tank.

Also, combustion products were not considered in this SA. Major combustion products that must be considered for their toxic effects are nitrogen oxides (NO_x) . As discussed in Ref. 4, large quantities of NO_x may be produced if a large volume of waste gases is burned. However, such a large burn in the single-shell tanks (SSTs) also result in releases of large quantities of waste material (see the dome collapse accidents). Under these circumstances, the contribution of the NO_x releases to the total toxic consequences of a burn is very small and well within the uncertainties of the toxicological consequence calculations.

Table 5-3 provides the PEL and ERPG values for the toxic gases considered in this SA. To determine the acceptance limits for liquid/solid waste material, Westinghouse Hanford Company (WHC) uses different types of composite waste for different tank grouping.⁵ The risk associated with waste releases is divided into three categories: particulate, toxic effects and corrosive effects. The chemical species in the composite waste is divided into toxic and corrosive bins. Within each bin, the allowable releases are computed using the "sum of the fractions" methodology. The minimum among the three categories (particulate, toxic, corrosive) is chosen in each frequency range. For further details of the methodology used in determining the maximum acceptable waste releases, the readers are referred to Ref. 5. The

A REAL PROPERTY AND A REAL

WHC-SD-WM-SAD-035, Rev. 0-a

TABLE 5-3 ESTABLISHED RISK GUIDELINES FOR SPECIFIC GASES

| Species | PEL-TWA | ERPG-1 | ERPG-2 | ERPG-3 |
|---------|----------------------|-----------------------|--------------------------|--------------------------|
| NH3 | 17 mg/m ³ | 17 mg/m ³ | 140 mg/m ³ | 680 mg/m ³ |
| | (25 ppm) | (25 ppm) | (200 ppm) | (1000 ppm) |
| N20 | 90 mg/m ³ | 270 mg/m ³ | 18,000 mg/m ³ | 36,000 mg/m ³ |
| | (50 ppm) | (150 ppm) | (10,000 ppm) | (20,000 ppm) |

TABLE 5-4 MAXIMUM ACCEPTABLE WASTE MATERIAL RELEASES⁵

| | SST LIQUID RELEASES | | | | |
|-------------------------------------|-------------------------|-------------------------|-------------------------|-------------------------|--|
| | ON | SITE | OFI | SITE | |
| Freq. | Puff | Continuous | Puff | Continuous (L/s) | |
| (yr-1) | (L) | | (L) | (2, 3) | |
| 10 ⁻² - 1 | 3.57 x 10 ⁻⁴ | 1.04 x 10 ⁻⁴ | 7.69 x 10 ⁺¹ | 1.85 x 10 ⁻¹ | |
| 10 ⁻² - 10 ⁻⁴ | 4.55 x 10 ⁻³ | 1.33 x 10 ⁻³ | 7.69 x 10 ⁺¹ | 1.85 x 10 ⁻¹ | |
| 10-4 - 10-6 | 1.75 x 10 ⁻² | 5.00 x 10 ⁻³ | 1.01 x 10 ⁺³ | 2.38 x 10 ⁺⁰ | |
| | SST SOLID RELEASES | | | | |
| 10 ⁻² - 1 | 9.09 x 10 ⁻⁵ | 2.56 x 10 ⁻⁵ | 6.67 x 10 ⁺⁰ | 1.59 x 10 ⁻² | |
| 10 ⁻² - 10 ⁻⁴ | 1.67 x 10 ⁻⁴ | 5.00 x 10 ⁻⁵ | 1.92 x 10 ⁺¹ | 4.55 x 10 ⁻² | |
| 10-4 - 10-6 | 3.45 x 10 ⁻³ | 1.02 x 10 ⁻³ | 3.70 x 10 ⁺¹ | 9.09 x 10 ⁻² | |

resulting maximum acceptable releases for SST liquid and solid releases are shown in Table 5-4. Note that the values reported in Table 5-4 represent the release quantities at the source because the atmospheric dispersion coefficients already are accounted for in deriving these magnitudes.

5.2. ACCIDENT CONSEQUENCES

The accidents for the FG/RMCS operations identified in Section 3 are analyzed in Section 4. The radiological and toxicological consequences of these accidents are analyzed in this section. Not all the accidents have unique consequences; the bounding consequences for all the burn accidents are analyzed as a dome collapse. Section 4.4 discusses that the waste burn resulting in a dome collapse is highly unlikely given the controls introduced by this SA. Consequences of a local waste burn are not currently analyzed and assumed to be bounded by a dome collapse. Table 5-5 summarizes how the accidents analyzed in Section 4 are mapped into the consequence analysis contained in this section.

Accidents resulting in environmental contamination are identified in Section 4. However, long-term radiological and toxicological consequences of environmental contamination are not analyzed in this SA.

In Section 5.6. all the consequences are summarized and graphically compared with the RGs.

5.3. CONSEQUENCES OF DOME COLLAPSE ACCIDENTS

In SSTs, burn accidents result in unacceptable consequences because even a small pressurization in the dome may result in dome failure potentially followed by a catastrophic collapse. Because of large radiological and toxicological consequences (especially onsite) of a dome collapse accident, one must demonstrate that such accidents have a frequency of less than 10^{-6} /yr. However, burn accidents are not the only accidents that may result in dome failure.

The design basis earthquake (DBE) has a return period of 7500 yr (frequency > 1×10^{-4} /yr). SSTs are designed to withstand a 0.20-g zero-period acceleration (ZPA). Because they are independent of the RMCS activities, structural consequences of seismic events are not analyzed in this SA.

The frequency of the dome collapse as a result of dome space deflagration is obtained by summing the frequency of the first 3 accident categories shown in Table 5-5. Thus, the unmitigated accident frequency for the dome collapse during FG/RMCS operations is > 1.0E-4/yr/tank.

Taking credit of the controls discussed in Section 4 and listed in Section 6, the frequency of the dome collapse accident (denoted as mitigated frequency) is lowered to less than 2.5 E-7/ yr/tank.

TABLE 5-5SUMMARY OF CONSEQUENCE ANALYSIS

| Accident | Bounding | [| Radiological | Toxicological |
|-----------------------|--------------------------------|-----------------------|--------------|---------------|
| Category | Consequence | Frequency | Consequences | Consequences |
| Above Ground | Dome Collapse | $MAF \approx 2.4E-07$ | Sec. 5.3.1 | Sec. 5.3.2 |
| Fires (Sec. 4.1) | (Sec. 5.3) | UAF = 1.8E-04 | | |
| Dome Fires | Dome Collapse | MAF = 4.8E-10 | Sec. 5.3.1 | Sec. 5.3.2 |
| (Sec. 4.2) | (Sec. 5.3) | UAF = 2.8E-07 | · · · | |
| Drill String Fires | Dome Collapse | MAF = 1.3E-8 | Sec. 5.3.1 | Sec. 5.3.2 |
| (Sec. 4.3) | (Sec. 5.3) | UAF = 6.6E-4 | | |
| Waste Fires | Dome Collapse | MAF < 1.0E-06 | Sec. 5.3.1 | Sec. 5.3.2 |
| (Sec. 4.4) | (Sec. 5.3) | UAF ~ 3.7 | | |
| Chemical | Environmental | NA | NA | NA |
| Reactions | Contamination | | | |
| (Sec. 4.5) | | | | |
| Containment Breach | Environmental Contamination | NA | NA | NA |
| (Sec. 4.6) | | | | · · |
| GRE (no burn) | Toxic Gas Release | Qualitative | Sec. 5.4.1.2 | Sec. 5.4.1.1 |
| (Sec. 4.7) | (Sec. 5.4.1) | | | |
| | Unfiltered Material Release | Qualitative | Sec. 5.4.2.1 | Sec. 5.4.3 |
| | (Sec. 5.4.2) | | | |
| Spills, etc | HEPA | MAF = 1.6E-5 | Sec. 5.5.1 | Sec. 5.5.2 |
| (Sec. 4.8). | Blowout (Sec. 5.5) | UAF = 1.6E-5 | | |
| | Sampler | MAF = 8.6E-9 | Sec. 5.5.1 | Sec. 5.5.2 |
| | Drop (Sec. 5.5) | UAF = 2.9E-6 | | |

| | Core barrel drop (Sec. 5.5) | MAF = 6.2E-7 UAF = 2.0 | Sec. 5.5.1 | Sec. 5.5.2 |
|----------------------------|--|---------------------------|------------|------------|
| | Drill String Drop (Sec. 5.5) | MAF = 4.4E-6 UAF = 2.0 | Sec. 5.5.1 | Sec. 5.5.2 |
| External Events (sec. 4.9) | Seismic Induced Dome Collapse (Sec. 5.3) | 1.0E-04 | Sec. 5.3.1 | Sec. 5.3.2 |

TABLE 5-5 (cont) SUMMARY OF CONSEQUENCE ANALYSIS

MAF = Mitigated accident frequency

UAF = Unmitigated accident frequency

HEPA= High-efficiency particulate air (filter)

The material releases as a result of dome collapse are discussed in Appendix I. Based on the calculations provided in Appendix I, the conservative amount of prompt respirable material release during a dome collapse in an SST is obtained as 62.5 L. The long-term respirable release from the crater left after the dome collapse is obtained as 12.5 L/wk.

5.3.1. Radiological Consequences

Appendix R provides the on-site and off-site receptor doses for a 1 L release for solid and liquid waste. Because the surface of the SST waste may be quite dry and the solid waste may result in larger doses than the liquid waste, it is assumed that all the short-term and long-term releases correspond to solid waste. Using the unit doses provided in Appendix R, the on-site and off-site radiological doses are obtained as shown in Table 5-6. As shown in Table 5-6, the long term doses are much lower than the prompt doses and are well within the uncertainty of the prompt doses. Also shown in Table 5-6 are the doses corresponding to RGs. Note that assuming that a fraction of the release is liquid lowers the doses.

5.3.2. Toxicological Consequences

The toxicological consequences of waste and toxic gas releases are discussed below.

5.3.2.1. Waste Material

For waste material, the release volume of 62.5 L must be compared with the puff release limits given in Table 5-4. The long term release of 12.5 L/wk (2×10^{-5} L/s) may be compared to continuous release limits given in Table 5-4. The prompt release volume (if assumed to be solid waste) exceeds the offsite RGs even for a very unlikely accident.

TABLE 5-6 BOUNDING RADIOLOGICAL CONSEQUENCES OF DOME COLLAPSE ACCIDENT

| | | Receptor Dose (rem) |
|---------|----------------|---------------------|
| ONSITE | Prompt | 17250 |
| | Long-Term 1 wk | 193 🔍 |
| OFFSITE | Prompt | 9.3 |
| | Long-Term 1 wk | 0.1 |

5.3.2.2. Toxic Gases

The toxic gases of interest (ammonia and nitrous oxide) are flammable. Thus, they will be totally or partially consumed during the burn event and the consequences of toxic gas releases during a burn are bounded by the consequences of a gas-release event (GRE) without a burn (Section 5.4.1.). As discussed previously, the combustion products are not considered.

5.4. GRE WITHOUT A BURN

A large gas release event that pressurizes the dome space is of concern because of the potential toxic gas and unfiltered material releases.

5.4.1. Toxic-Gas Releases.

This section considers the release of toxic gases (ammonia and nitrous oxide) through the open risers, frisbee, and the exhaust and inlet stacks.

As bounding magnitudes, the maximum ammonia fraction in the waste gas is set to 60%, and the maximum nitrous oxide fraction in the waste gas is set to 75% (Appendix C). Only the toxicological consequences of ammonia are discussed in this section because if ammonia releases meet the RGs, the nitrous oxide releases also will meet the RGs. While the maximum nitrous oxide fraction is 125% of the ammonia fraction, the guidelines for nitrous oxide are always more than twice the ammonia guidelines for all the accident frequencies (see Table 5-3).

The peak ammonia concentrations in the dome are provided in frequency bins in Appendix L. In this section, the peak concentration as well as the release rate are needed. For the purpose of this consequence assessment, the total release volume of waste gas is obtained by multiplying the peak concentration by the dome volume at a given frequency bin. The results are shown in Table 5-7. The toxicological guidelines are given in terms of 1-h average values (Section 5.1.2.). For added conservatism, the total release is postulated to occur in less than 15 minutes, and the

average release rate is obtained by dividing the volumes shown in Table 5-7 by 15 minutes. The release to the atmosphere is assumed to occur at this release rate with the peak ammonia concentration. This simplification is conservative because the expected gas-release rates are much lower than the magnitudes obtained by this simple approach (see Appendix L). If the actual release is faster than 15 min. (which is very unlikely based on the discussion in Appendix L), the proposed approach still yields the correct time-averaged release rate over a 15-min. period.

The dome peak ammonia and nitrous oxide concentrations then become 1.2%, 6%, and 12% for the anticipated, very unlikely, and extremely unlikely gas-release events defined based on frequency ranges and expected gas release amounts and marked as GRE-1, GRE-2 and GRE-3 in Table 5-7. These three GREs categories will be used in tables given in the rest of this section. The releases are not expected during installation because the installation is not a waste-intrusive activity. However, gas releases could be induced during the operation or removal phase. Postulated gas releases are listed in Table 5-8. Note that release rates given in Table 5-8 is based on 15-minute average releases.

The first three accidents may occur during removal when the riser is open for a period of much less than 8 hours. The conservative 8-hour open riser period relative to mission time is considered in the frequency determination, resulting in a gas-release probability of 8/144 = 5.6E-2. The third accident will not be considered further because the frequency is low. Accidents 4-6 consider gas releases from the frisbee. The riser sleeve is protected with nitrogen purge. This system prevents any gas penetration between the riser sleeve and the DS. Thus, the gas release from the frisbee could occur only if the nitrogen purge fails. Accidents 5 and 6 are beyond extremely unlikely. Accident scenarios 7 to 9 are gas releases through the drill string. This accident can occur when the drill string is disconnected from the drill unit if the seal between the sampler and drill bit fails or if the sampler fails to latch before drilling. Accidents 8 and 9 are beyond extremely unlikely. The last 6 accidents considered only the GRE from the exhauster and inlet stack during operations.

| Event Probability | Event Category | Q-Prompt Release, (ft ³) | Peak NH ₃ Conc. (%) |
|-------------------|----------------|---|--------------------------------|
| ≥1.0 E-2 | GRE-1 | ≤1000 | 1.2 |
| ≥1.0 E-4 | GRE-2 | ≤ 5000 | 6 |
| ≥1.0 E-6 | GRE-3 | ≤ 10,000 | 12 |

TABLE 5-7 ANTICIPATED AND UNLIKELY GAS-RELEASE EVENTS

TABLE 5-8SUMMARY OF GAS-RELEASE ACCIDENTS

| Accident Condition | Frequency (yr) | Gas Concentration & Release Rate |
|--|------------------------------------|-------------------------------------|
| 1-GRE-1 and open riser (removal) | 1.1E-3 | 1.2% NH3 in the dome, |
| | (0.01x5.6E-2x2) | 66 scfm* |
| 2-GRE-2 and open riser (removal) | 1.1E-5 | 6% NH ₃ in the dome, |
| | (1E-4x5.6E-2x2) | 333 scfm* |
| 3-GRE-3 and open riser (removal) | 1.1E-7 | 12% NH ₃ in the dome, |
| | (1E-6x5.6E-2x2) | 666 scfm* |
| | Beyond extremely unlikely | |
| 4-GRE-1 and open frisbee (operation) (nitrogen | 3.2E-6 | 1.2% NH ₃ in the dome, |
| purge to riser sleeve fails) | (2x0.01x1.6E-4) | 66 scfm* |
| 5-GRE-2 and open frisbee (operation) (nitrogen | 3.2E-8 | 6% NH3 in the dome, |
| purge to riser sleeve fails) | (2x1.0E-4x1.6E-4) | 333 scfm* |
| | beyond extremely unlikely event | |
| 6-GRE-3 and open frisbee (operation) (nitrogen | 3.2E-10 | 12% NH ₃ in the dome, |
| purge to riser sleeve fails) | (2x1.0E-6x1.6E-4) | 666 scfm* |
| | beyond extremely unlikely event | |
| 7-GRE-1 and drill string open at the top with | 2.6E-5 | 1.2% NH ₃ in the dome, |
| sampler in the drill string | (2x0.01x1.3E-3) | 66 scfm* |
| 8-GRE-2 and drill string open at the top with | 2.0E-7 | 6% NH_3 in the dome, |
| sampler in the drill string | (2x1.0E-4x1.3E-3) | 333 scfm* |
| | beyond extremely unlikely event | |
| 9-GRE-2 and drill string open at the top with | 2.0E-9 | 12% NH ₃ in the dome, |
| sampler in the drill string | (2x1.0E-6x1.3E-3) | 666 scfm* |
| | beyond extremely unlikely event | |
| 10-GRE-1 from exhaust and inlet stack | 0.02 | 1.2% NH ₃ in the dome, |
| (operation) | (2x0.01) | 66 scfm* |

| Accident Condition | Frequency (yr) | Gas Concentration & Release Rate |
|--|----------------------|-------------------------------------|
| 11-GRE-2 from exhaust and inlet stack (operation) | 2.0E-4 (2x1.0E-4) | 6% NH₃ in the dome, 333 scfm* |
| 12-GRE-3 from exhaust and inlet stack (operation) | 2.0E-6 (2x1.0E-6) | 12% NH3 in the dome, 666 scfm* |
| 13-Continuous releases from exhauster after an GRE-1 | 0.02 (2x0.01) | 1.2% NH3 in the dome, 250 scfm** |
| 14-Continuous releases from exhauster after an GRE-2 | 2.0E-4 (2x1.0E-4) | 6% NH3 in the dome, 250 scfm** |
| 15-Continuous releases from exhauster after an GRE-3 | 2.0E-6 (2x1.0E-6) | 12% NH3 in the dome, 250 scfm** |

TABLE 5-8 (cont) GAS-RELEASE ACCIDENTS

* Averaged over 15-minute period

** Maximum exhauster flow rate

5.4.1.1. Toxicological Consequences

The toxic-gas concentrations at the on-site and off-site receptor locations are calculated using Eq. (K-2) given in Appendix K. Atmospheric dispersion factors for ground and stack releases are summarized in Table K-3 of Appendix K. Table 5-9 gives the calculated on-site and off-site ammonia concentrations for ground and stack releases. Numbers given in parentheses are acceptable concentrations for ammonia.

In the first nine accidents given in Table 5.8, the GREs are treated as ground releases because they involve an opening at the top of the tank. Note that there are still inlet and exhaust stacks where approximately 2/3 of the release occurs. It is clear from Table 5-9 that the on-site RGs are not exceeded (accidents 1, 2, 4, 7 in Table 5.9) for accidents involving ground releases.

Accidents 10 to 15 do not involve ground releases. For these accidents, the results of both ground or stack releases are given in Table 5.9. Accidents 11, 12, 13, and 14 result in exceeding RGs if they were considered as ground releases. However, the exhauster has a 15-ft height stack with a 4-in. internal diameter. This SA requires the use of an inlet stack with the same exit diameter and height as the exhaust stack. For gas releases of 66, 330, and 250 scfm, the velocity of released gas in these stacks become 3.8 m/s, 19.2 m/s, and 7.3 m/s, respectively. With these velocities and the consideration of the height of the stacks, the plume effective emission height increases as a result of momentum effects. Dispersion coefficients for these conditions are smaller than those given for ground releases, as shown in Appendix K. Consideration of the stack height reduces the on-site concentrations 6 to 7 times.

TABLE 5-9

AMMONIA CONCENTRATIONS AT ON-SITE AND OFF-SITE RECEPTORS FOR GROUND AND STACK RELEASES FOR ALL GAS RELEASE ACCIDENTS

| | | Ground Release | | Stack Release | |
|---|--------|----------------|---------|---------------|---------|
| Accident | Freq. | Onsite | Offsite | Onsite | Offsite |
| | (1/yr) | (ppm) | (ppm) | (ppm) | √(ppm) |
| 1-GRE-1 and open riser (removal), 1.2% | 1.1E-3 | 13 | 0.01 | NA | NA |
| NH₃ , 66 scfm | | (200) | (25) | | |
| 2-GRE-2 and open riser (removal), 6% | 1.1E-5 | 324 | 0.18 | NA | NA |
| NH ₃ , 633 scfm | | (1000) | (200) | | |
| 4-GRE-1 and open frisbee (operation) | 3.2E-6 | 13 | 0.01 | NA | NA |
| (nitrogen purge to riser sleeve fails), 1.2% NH ₃ , 66 scfm | | (1000) | (200) | | |
| 7-GRE-1 and drill string open at the top | 2.6E-5 | 13 | 0.01 | NA | NA |
| with sampler in the drill string, 1.2% NH ₃ , 66 scfm | | (1000) | (200) | | |
| 10-GRE-1 from exhaust and inlet stack | 0.02 | 13 | 0.01 | 2 | 0.01 |
| (operation),1.2% NH ₃ , 66 scfm | | (25) | (25) | (25) | (25) |
| 11-GRE-2 from exhaust and inlet stack | 2.0E-4 | 324 | 0.18 | 48 | 0.18 |
| (operation), 6% NH ₃ , 333 scfm | | (200) | (25) | (200) | (25) |
| 12-GRE-3 from exhaust and inlet stack | 2.0E-6 | 1297 | 0.72 | 193 | 0.72 |
| (operation), 12% NH ₃ , 666 scfm | | (1000) | (200) | (1000) | (200) |
| 13-Continuous releases from exhauster | 0.02 | 49 | 0.03 | 7 | 0.03 |
| after a GRE-1, 1.2% NH ₃ , 250 scfm | | (25) | (25) | (25) | (25) |
| 14-Continuous releases from exhauster | 2.0E-4 | 244 | 0.13 | 36 | 0.13 |
| after an GRE-2, 6% NH ₃ , 250 scfm | | (200) | (25) | (200) | (25) |
| 15-Continuous releases from exhauster | 2.0E-6 | 487 | 0.27 | 72 | 0.27 |
| after an GRE-3, 12% NH ₃ , 250 scfm | | (1000) | (200) | (1000) | (200) |

To be conservative, credit is taken only for the height of the stack. A point source at a 15-ft height is considered. With this assumption, the ammonia concentrations at the on-site and off-site receptor locations meet the RGs as shown in Table 5-9. The consideration of the velocity and diameter of the stack further decreases ammonia concentrations. Furthermore, the flammable-gas monitors will alarm at concentrations much lower than those postulated in these conservative estimates, providing adequate time for workers and nearby individuals to take protective measures. Response procedures for a GRE must be developed, assuming that the release contains a large fraction of ammonia (flammable-gas monitors cannot differentiate between hydrogen and ammonia).

5.4.1.2. Radiological Consequences

The radiological consequences of gas release is a concern when the riser is open. If the release is through HEPA filters, there is no particulate release. Open riser particulate releases are discussed in the next section.

5.4.2. Unfiltered Material Releases

This section discusses the unfiltered waste releases associated with the GREs. Releases from spill accidents are considered in the next section. Unfiltered releases can occur during operation and removal. During installation, a GRE is not expected; thus, only the operation and removal phases involve unfiltered material releases. Accidents for unfiltered releases are listed in Table 5-10. Release amounts are determined in Appendix M.

Appendix M gives the release amounts for unfiltered releases through an open riser. The waste releases from the open riser become 12 g, 60 g, and 120 g for gas release amounts of 1000, 5000, and 10,000 ft³, respectively. Accidents 1 and 2 in Table 5-8 are considered for unfiltered releases through the open riser. The third accident given in Table 5.8 is beyond extremely unlikely. In the fourth scenario, it is conservatively assumed that waste material is released from an opening between the frisbee and the drill string if the nitrogen purge fails, the frisbee fails and a GRE occurs. Credit was not taken for the failure of the frisbee. The frequency includes the failure probability of a nitrogen purge while drilling. Accidents 5 and 6 are not considered for unfiltered releases because they are beyond extremely unlikely events. The seventh scenario addresses a release through the drill string with a failed chevron seal. It is assumed that the internal surfaces of the drill string are contaminated. If there is a 0.5-mm layer inside the surface of the drill string with a maximum length of 395 in., the total waste could be estimated as 1280 g. Assuming a 10% release fraction (because velocities could be high in the drill string), a release amount of 128 g for both anticipated and unlikely GREs can be estimated. Accidents 8 and 9 (Table 5.8) are not considered for unfiltered releases because they are beyond extremely unlikely events. Accidents 10 to 12 given in Table 5.8 do not involve unfiltered releases because they are through HEPA filters. The last scenario involves releases through the exhauster if the HEPA filters fail. The frequency of this accident is 1.6E-5/yr and it is same for anticipated or unlikely GREs. The

| Accident | Freq. (1/yr) | Release Type |
|---|-----------------|------------------------|
| GRE-1 and open riser (removal) (1)* | 1.1E-3 | 12 g material release |
| GRE-2 and open riser (removal) (2) | 1.1E-5 | 60 g material release |
| GRE-1 and open frisbee (operation) (nitrogen purge to riser sleeve fails) (4) | 3.2E-6 | 12 g material release |
| GRE-1 and drill string open at the top with sampler in the drill string (7) | 2.6E-5 | 128 g material release |
| Continuous releases from exhauster after a HEPA filter fails (NA) | 1.6E-5 | 0.08 g/s |

TABLE 5-10UNFILTERED WASTE RELEASE ACCIDENTS

* Numbers in parentheses correspond to accident numbers in Table 5-8.

maximum flow rate is 250 scfm. Based on a dome loading of 950 g, the solid flow rate is 0.08 g/s. These accidents cover the cases 13 to 15 of Table 5.8 with the assumption that the HEPA filter fails.

There is another case in which a GRE during removal with a contaminated drill string occurs. Failure of the decontamination procedures and the seal or the sampler latching is assumed. The frequency of this accident is very small because a contaminated drill string must occur during removal and coincide with a GRE.

A non-GRE unfiltered release occurs if the riser is open and the tank-dome pressure becomes positive. Ventilation system failure combined with open-tank conditions has been identified as one of these conditions. Controls are established to help ensure that this kind of release is minimized. No waste-intrusive activities can be started if the ventilation system is not working properly. However, the ventilation system can fail during the periods when the riser is open. In comparison to other releases driven by GREs, this release is small, and the frequency of this accident is also small.

The controls derived from these analyses are as follows:

- Existence of a decontamination system operation and the decontamination of the sleeve and drill string are required when the radiation level exceeds the allowable limits.
- RMCS operations must be shut down if the dome flammable gas concentration is above 5000 ppm hydrogen equivalent. Based on currently available calibration data, this set point gives protection for ammonia concentrations of ~10,000 ppm if the gas is purely ammonia.

- Risers shall not be opened within four hours after drilling.
- The riser shall be open no more than eight hours.
- Evacuation is necessary if a gas release occurs.

5.4.2.1. Radiological Consequences

Radiological doses at the onsite and offsite receptor locations are summarized in Table 5-11. All the accidents listed in Table 5.11 meet the radiological on-site and offsite RGs.

5.4.3. Toxicological Consequences

The off-site and on-site consequences are measured at the release source. For a GRE and open riser or a GRE and open frisbee the volume release rates are 8.3E-6 L/s for an anticipated GRE and 4.2E-5 L/s for a very unlikely GRE. For an unfiltered release and failed chevron seal, the volume released is 9.0E-5 L/s. HEPA filter failure can cause a release of 0.08 g/s (5E-5 L/s). The waste-release rates are compared with the RGs in Table 5-12.

Toxic waste release rates are based on 15 minute continuous releases. Acceptable release rates at the source are obtained from Table 5-1 and are shown in parentheses in Table 5-12. The consequences meet the on-site and off-site RGs.

Onsite Offsite Frequency Accident Condition (yr) (rem) (rem) GRE-1 and open riser (removal), 1.1E-3 2.04 0.001 (10)(2) 12 g. (1)* GRE-2 and open riser (removal), 60 g. (2) 1.1E-5 10.2 0.006 (49) (10) 3.2E-6 2.04 0.001 GRE-1 and open frisbee (operation) (nitrogen purge to riser sleeve fails), 12 g. (4) (70) (15) 21.8 GRE-1 and drill string open at the top with 2.6E-5 0.012 sampler in the drill string, 128 g. (7) (38)(10) 12.4 Continuous releases from exhauster after a 1.6E-5 0.007 HEPA filter fails, 0.08 g/s. (NA) (10) (40)

TABLE 5-11 RADIATION DOSES AT ON-SITE AND OFF-SITE RECEPTORS FOR GRE ACCIDENTS

* Numbers in parentheses correspond to accident numbers in Table 5-8.

5.5. SPILL ACCIDENTS

Spill accidents are summarized in Table 5-13. The frequencies of the second and third accidents in Table 5-13 are small, and these accidents are beyond extremely unlikely events.

5.5.1. Radiological Consequences

The calculated on-site and off-site radiation doses for HEPA filter blowout and waste spill from drill string accidents are given in Table 5-14 with the acceptable dose amounts given in parentheses. HEPA filter blowout does not result in exceeding onsite and off-site RGs. On-site and off-site radiological consequences of spill meet the radiological RGs.

| Accident | Freq. | Onsite | Offsite | |
|--|--------|---------------|--------------------|--|
| | (1/yr) | | | |
| GRE-1 and open riser (removal), 12 g. (1)* | 1.1E-3 | 8.3E-6 L/s | 8.3E-6 L/s | |
| | | (5E-4 L/s) | (4.55E-2 L/s) | |
| GRE-2 and open riser (removal), 60 g. (2) | 1.1E-5 | 4.2E-5 L/s | 4.2E-5 L/s | |
| | | (1.02E-3 L/s) | (9.09E-2 L/s) | |
| GRE-1 and open frisbee (operation) (nitrogen' purge to | 3.2E-6 | 8.3E-6 L/s | 8 .3E-6 L/s | |
| riser sleeve fails), 12 g. (4) | | (1.02E-3 L/s) | (9.09E-2 L/s) | |
| GRE-1 and drill string open at the top with sampler in the | 2.6E-5 | 9E-5 L/s | 9E-5 L/s | |
| drill string, 128 g. (7) | | (1.02E-3 L/s) | (9.09E-2 L/s) | |
| Continuous releases from exhauster after a HEPA filter | 1.6E-5 | 5E-5 L/s | 5E-5 L/s | |
| fails, 0.08 g/s. (NA) | | (1.02E-3 L/s) | (9.09E-2 L/s) | |

TABLE 5-12 ONSITE AND OFFSITE TOXIC SOLID WASTE RELEASES

* Numbers in parentheses correspond to accident numbers in Table 5-8.

| Accident | Frequency (1/yr) | Release (g) |
|---|-------------------------------------|----------------|
| Radioactive aerosols are released from HEPA filter blowout. | 1.6E-5 | 9 |
| Spill results from dropped sampler in shielded receiver. | 8.6E-9 beyond extremely unlikely | 0.08 |
| Waste from contaminated drill string interior is released above ground. | 6.2E-7 beyond extremely unlikely | 0.66 |
| Waste from contaminated drill string exterior is released above ground. | 4.4E-6 | 0.75 |

TABLE 5-13 SUMMARY OF SPILL ACCIDENTS

5.5.2. Toxicological Consequences

The off-site and on-site consequences are measured at the release source. For the short term release for HEPA failure the volume released is 5.63E-3 L. A spill from drill string yields 4.7E-4 L of material. Release amounts and acceptable release amounts to meet the ERPG values are given in Tables 5.14a. Spill releases meet the off-site RGs. The on-site RGs are exceeded for HEPA filter blowout accidents. However, in deriving allowable limits, the atmospheric dispersion coefficients for ground releases were used.⁵

TABLE 5-14 ONSITE AND OFFSITE RADIOLOGICAL DOSES FOR SPILLS

| | Freq. | On-site Dose | Off-site Dose |
|--|--------|--------------|---------------|
| Accident | (1/yr) | (rem) | (rem) |
| Radioactive aerosols are released from HEPA filter | 1.6E-5 | 1.53 | 8.3E-4 |
| blowout. | | (43.4) | (2) |
| Waste from a contaminated | 4.4E-6 | 0.13 | 7.0E-5 |
| released aboveground. | | (64) | (4) |

TABLE 5.14A ONSITE AND OFFSITE TOXIC SOLID WASTE RELEASES FOR SPILLS

| | Freq. | On-site | Off-site |
|--|--------|-------------|-----------|
| Accident | (1/yr) | | |
| Radioactive aerosols are released from HEPA filter blowout. | 1.6E-5 | 5.63E-3 L | 5.63E-3 L |
| | | (3.45E-3 L) | (37 L) |
| Waste from contaminated drill string exterior is released aboveground. | 4.4E-6 | 4.7E-4 L | 4.7E-4 L |
| | | (3.45E-3 L) | (37 L) |

HEPA filter blowout would be an elevated release with a factor of 6 reduction in the dispersion coefficient. Thus, the receptor doses will be 1×10^{-3} L and within the allowable limits.

5.6. COMPARISON OF THE ACCIDENT CONSEQUENCES WITH THE RGS

Section 5.6 presents the comparison of the consequence calculated in Section 5.5 to the RGs presented in Section 5.1. The comparison is broken into the following sections,

- Section 5.6.1 Comparison to on-site and off-site radiological dose.
- Section 5.6.2 Comparison to on-site and off-site toxic gas releases.
- Section 5.6.3 Comparison to on-site and off-site toxic waste release.

In order to perform the comparison, the frequency of the accidents on a per-year basis must be computed. It is assumed in Section 4 that there will be two FG/RMCS activities per year per tank. The accident frequencies are calculated based on this assumption. The risk computed is on a per-tank basis and is not a site-wide risk, where the site-wide risk is defined as the cumulative risk of FG/RMCS operations over multiple tanks in a given year. The information contained in this SA cannot be easily converted to a site-wide risk because of the following:

- The SA uses bounding tank parameters for a hypothetical worse-case tank;
- The SA uses 95% meteorology data to compute the consequences;

Appendix Q discusses this issue in detail. The discussion in Appendix Q demonstrates that the use of the bounding tank consequences computed in this SA is not adequate to obtain site-wide risk. This approach is expected to grossly overestimate the risk.

On the other hand, this SA should not be interpreted as only being applicable to a single tank in a given year. This SA does not place any restrictions on the number of tanks that can be sampled in a given year, and multiple tanks can be sampled under the present safety basis.

This interpretation is in agreement with the general concept of RGs and the Safety Assessment approach and does not contradict the guidance provided in Ref. 2. However, the guidance provided by Ref. 2 on how to combine the risk from different activities and from different tanks is not very explicit and is subject to interpretation.

5.6.1. Accidents with Radiological Consequences

All accidents (except the dome collapse accidents and other accidents which are beyond extremely unlikely events) resulting in radiological consequences are summarized in Table 5-15. Comparison with the on-site and off-site RGs is shown in Figs. 5-1 and 5-2. The numbers next to the data points correspond to the accident numbers given in Table 5.15. As shown in Fig. 5-1 and 5-2, all the conservative doses fall below the RG using best-estimate frequencies.

5.6.2. Toxic-Gas Releases

All gas-release accidents (except the accidents which are beyond extremely unlikely events) with toxicological consequences are summarized in Table 5.16. The computed ammonia concentrations shown in Table 5.16 are plotted in Figs. 5.3 and 5.4 in comparison with the RGs. As shown in these figures, all conservatively calculated ammonia concentrations at the on-site and off-site receptor locations fall below the RGs using best-estimate frequencies. This conclusion is valid also for nitrous oxide releases because the calculated bounding nitrous oxide concentrations are slightly higher than the ammonia concentrations shown in these figures, but the RGs for nitrous oxide are much larger.

5.6.3. Toxic Liquid/Solid Releases

All of the accidents (except the dome collapse accidents and other accidents which are beyond extremely unlikely events) resulting in material release are listed in Table 5-17. Release rates are given over a 15-minute release period. The first five accidents are continuous releases, and the last two are puff releases. Table 5-17 lists the frequency each of accident and the release rates at the release point.

TABLE 5-15SUMMARY OF ALL ACCIDENTS RESULTING IN RADIOLOGICAL RELEASES

| | Frequency | Onsite | Offsite |
|--|--------------------|--------|---------|
| Accident Condition | (yr) | (rem) | (rem) |
| 1-GRE-1 and open riser (removal), 12 g. | 1.1E-3 | 2.04 | 0.001 |
| 2-GRE-2 and open riser (removal), 60 g. | 1.1E-5 | 10.2 | 0.006 |
| 3-GRE-1 and open frisbee (operation) (nitrogen purge to riser sleeve fails), 12 g. | 3.2E- 6 | 2.04 | 0.001 |
| 4-GRE-1 and drill string open at the top with sampler in the drill string, 128 g. | 2.6E-5 | 21.8 | 0.012 |
| 5-Continuous releases from exhauster after a HEPA filter fails, 0.08 g/s. | 1.6E-5 | 12.4 | 0.007 |
| 6-Radioactive aerosols are released from HEPA filter blowout. | 1.6E-5 | 1.53 | 8.3E-4 |
| 7-Waste from the contaminated drill string exterior is released aboveground. | 4.4E-6 | 0.13 | 7.0E-5 |

- 1



Fig. 5-1. Comparison of calculated radiological doses with the on-site RGs.



Fig. 5-2. Comparison of calculated radiological doses with the off-site RGs.

TABLE 5-16SUMMARY OF GAS-RELEASE ACCIDENTS

| | Freq. | Onsite | Offsite |
|--|--------------------|--------|---------|
| Accident | (1/yr) | (ppm) | (ppm) |
| 1-GRE-1 and open riser (removal), 1.2% NH ₃ , 66 scfm | 1.1E-3 | 13 | 0.01 |
| 2-GRE-2 and open riser (removal), 6% $\rm NH_3$, 633 scfm | 1.1E-5 | 324 | 0.18 |
| 3-GRE-1 and open frisbee (operation) (nitrogen purge to riser sleeve fails), 1.2% NH ₃ , 66 scfm | 3.2E-6 | 13 | 0.01 |
| 4-GRE-1 and drill string open at the top with sampler in the drill string, 1.2% NH_3 , 66 scfm | 2.6E-5 | 13 | 0.01 |
| 5-GRE-1 from exhaust and inlet stack (operation), 1.2% NH ₃ , 66 scfm | 0.02 | 2 | 0.01 |
| 6-GRE-2 from exhaust and inlet stack (operation), 6% NH ₃ , 333 scfm | 2.0E-4 | 48 | 0.18 |
| 7-GRE-3 from exhaust and inlet stack (operation), 12% NH ₃ , 666 scfm | 2.0E-6 | 193 | 0.72 |
| 8-Continuous releases from exhauster after a GRE-1, 1.2% NH_3 , 250 scfm | 0.02 | 7 | 0.03 |
| 9-Continuous releases from exhauster after an GRE-2, 6% NH ₃ , 250 scfm | 2.0E-4 | 36 | 0.13 |
| 10-Continuous releases from exhauster after an GRE-3, 12% NH ₃ , 250 scfm | 2.0E- 6 | 72 | 0.27 |



Fig. 5-3. On-site ammonia concentrations for gas release accidents.



Fig. 5-4. Off-site ammonia concentrations for the GRE accidents

5-24

TABLE 5-17SUMMARY OF TOXIC SOLID WASTE RELEASES

| | Freq. | | |
|--|--------|--------------------|------------|
| Accident | (1/yr) | On-site | Off-site |
| 1-GRE-1 and open riser (removal), 12 g. | 1.1E-3 | 8.3E-6 L/s | 8.3È-6 L/s |
| 2-GRE-2 and open riser (removal), 60 g. | 1.1E-5 | 4.2E-5 L/s | 4.2E-5 L/s |
| 3-GRE-1 and open frisbee (operation) (nitrogen purge to riser sleeve fails), 12 g. | 3.2E-6 | 8 .3E-6 L/s | 8.3E-6 L/s |
| 4-GRE-1 and drill string open at the top with sampler in the drill string, 128 g. | 2.6E-5 | 9E-5 L/s | 9E-5 L/s |
| 5-Continuous releases from exhauster after a HEPA filter fails, 0.08 g/s. | 1.6E-5 | 5E-5 L/s | 5E-5 L/s |
| 6-Radioactive aerosols are released from HEPA filter blowout. | 1.6E-5 | 5.63E-3 L | 5.63E-3 L |
| 7-Waste from contaminated drill string exterior is released aboveground. | 4.4E-6 | 4.7E-4 L | 4.7E-4 L |
The computed waste-release rates at the source of releases shown in Table 5-17 are plotted in Figs. 5-5 and 5-6 in comparison with the RGs. The RGs in these figures are given in terms of the release amount or release rate at the release point. As shown in these figures, all the conservative waste release rates at the on-site and off-site receptor locations fall below RGs using bestestimate frequencies, except for Accident 6. Accident 6 is the HEPA failure accident. Total release is estimated as 9 g. The onsite acceptable release amount is 5.5 g. To minimize a significant particulate accumulation at the HEPA filters, a radiation survey control is established. The radiation rate is limited with 100 mrem/h. This rate corresponds to waste amounts much less than 900 g that is assumed to accumulate in the HEPA filters. This control was not credited in obtaining the frequency of 1.6E-5. Furthermore, in deriving allowable limits, the atmospheric dispersion coefficients for ground releases were used.⁵ A HEPA filter blowout would be an elevated release with a factor of 6 reduction in the dispersion coefficient. Thus, the receptor doses will be within the allowable limits.



Fig. 5-5. Consequence of toxic waste releases at the on-site receptor location.



Fig. 5-6. Consequence of toxic waste release at the off-site receptor location.

5.7. **REFERENCES**

- 1. A. B. Sidpara, "Risk Acceptance Criteria for Tank Farm Operation," Department of Energy/Richland Operations, letter, 95-TOP-063 to the President of Westinghouse Hanford Company (June 14, 1995).
- 2. Westinghouse Hanford Company, "Safety Analysis Manual," Westinghouse Hanford Company report WHC-CM-4-46 Rev. 4 (September 30, 1995).
- J. L. Huckaby, H. Babad, and D. R. Bratzel, "Headspace Gas and Vapor Characterization Summary for the 43 Vapor Program Suspect Tanks," Westinghouse Hanford Company report WHC-SD-WM-ER-514, Rev. 1A (September 1995).
- 4. W. L. Kubic, "Production of Nitrogen Oxides During Burn Accidents in Tank 241-SY-101," Los Alamos National Laboratory Calc-Note TSA10-CN-WT-SA-GR-040 (September 1995).
- 5. Westinghouse Hanford Company, "Toxic Chemical Considerations for Tank Farm Releases," Westinghouse Hanford Company report WHC-SD-WM-SARR-011 Rev. 1 (October 1995).

6.0. CONTROLS

In this section, the comprehensive list of all credited design features, and procedural and administrative controls imposed on the installation, operation, and removal of rotary-mode core sampling (RMCS) equipment in single-shell tanks (SSTs) on the Flammable Gas Watch List (FGWL) or those tanks recommended by the contractor to be included on the FGWL, hence referred to as FG/RMCS operations, are described.

6.1. ADMINISTRATIVE CONTROLS

This section provides the controls to be used for all phases of FG/RMCS activities in: (1) pre-installation and installation activities; (2) drilling and sample retrieval; and (3) removal and decontamination. The controls have been grouped for each phase of the FG/RMCS activities for clearer and easier procedures development, and have been developed based on the results, assumptions, and initial conditions of this SA, in conjunction with existing WHC controls. Those WHC standard controls important to the activities have been repeated in this SA for clarity; however, the set of controls listed in this SA is intended only to supplement the WHC standard controls, not replace them.

Westinghouse Hanford Company (WHC) standard controls include a series of WHC documents that define the safety envelope for the tank farm, waste transfer activities, and waste storage activities. The primary document is the WHC Health and Safety Plan¹, although other documents include the Safety Assessment for Push- and Rotary-Mode Core Sampling in Ferrocyanide Tank,² Safety Analysis for the Push Mode and Rotary Mode Core Sampling,³ and the Interim Operational Safety Requirements for Rotary Mode Core Sampling in Flammable Gas Single Shell Tanks.⁴ During the development of the procedures for each of the activities, the current operational safety requirements (OSRs), interim operational Safety requirements (IOSRs), and operational safety documents (OSDs) must be considered (refer IOSR⁴ for restart requirement). The safety envelope established by the analyses shall not be changed unless approved by the Department of Energy (DOE). The controls provided in this section can be modified if the appropriate organization grants approval.

Most of the controls presented in this section are based on the analyses conducted for this safety assessment (SA). These controls have been designed to ensure that the analysis assumptions and initial conditions are maintained throughout each phase of the activities. In several cases, the controls have been developed to add an additional safety margin, consistent with a philosophy of defense-in-depth. Therefore, the controls should be an integral part of the procedure development process to maintain the level of safety demonstrated in this SA.

Administrative controls are the requirements that shall be followed to ensure that the activity stays within the bounds of the SA. As such, this set of administrative controls shall be used during the development of the procedures for each activity.

Sections 6.5 through 6.9 provide the specific controls in tabular format to be used for the activities covered by this SA. Of these activities, FG/RMCS operations are the most complex. Therefore, more controls have been instituted for the operational phase than for other processes. Because of the complexity of the FG/RMCS operation, these controls have been divided into three levels. Each control level requires a different level of approval for modification and for restarting operations after abnormal shut downs.

FG/RMCS operation controls are designated as either Level 1, Level 2, or Level 3. This graded approach reflects the importance of a particular control, the level of approval required for modification, and the level of approval required to restart operation if a particular limit is exceeded.

Level 1 controls are the most important, they are under the most stringent management supervision and are equivalent to OSRs (see Section 6.2.). Level 1 controls ensure that the most important bounds, established in the SA, are maintained at all times, as demonstrated by the appropriate analyses for both potential prompt effects and postoperation effects. Changes to the Level 1 controls, if required, will be developed by Characterization/RMCS support personnel, and approved by the Plant Review Committee (PRC), WHC management, and DOE/RL. The WHC Design Authority is responsible for approval of all aspects of equipment design.

Level 2 controls are the next level in importance. The PRC must approve modifications to Level 2 control parameters and notify WHC Management and DOE/RL of the modifications and the technical bases for the modifications. Changes to those Level 2 controls that are included in the IOSR's require DOE approval.

Level 3 controls are the lowest level in importance. Changes to the Level 3 controls will be approved by the PRC. WHC Management and DOE/RL will be notified of the changes and their technical bases.

The PRC may charter a separate technical review group to perform the review and approval responsibilities of the PRC.

All changes to any of the controls or equipment credited in this safety assessment, and any special tests, must be screened for unreviewed safety questions, in accordance with DOE Order 5480.21, and in accordance with WHC policies and procedures.

An automatic shutdown as defined in the following administrative control tables is not considered a violation of the control. Restart of any FG/RMCS activity shall be commensurate with its designated level, and with DOE and contractor procedures.

6.2. LEVEL DESIGNATION PROCESS

The designation of the control levels was performed using the following methodology Mitigated and unmitigated accident frequencies and consequences were estimated for each individual accident scenario discussed in Section 4 by considering the application of the controls (mitigated) or without the controls (unmitigated). In general, controls were designated Level 1 when the control prevented the mitigated consequence of an accident scenario from exceeding the offsite Risk Guidelines.

Level 2 was assigned to controls when the accident was considered to result in consequences that exceed the onsite Risk Guidelines, based on unmitigated accident frequencies.

A control was designated Level 3 if the unmitigated accident frequency was lower than 1.0E-6, or if it enhanced the defense-in-depth arguments even when the accident consequences are within the Risk Guidelines.

Certain controls appeared in multiple accident scenarios. In these cases, the most conservative level was designated for that control.

6.3 FG/RMCS SAFETY DESIGN FEATURES

Principal safety criteria have been established to ensure safe operations during rotary-mode core sampling activities. Design features within the FG/RMCS equipment and procedures have been established to comply with these safety criteria.

The safety equipment list with design criteria classifications, as approved by the WHC Design Authority and plant management, is provided in Reference 5. Reference 6 lists the qualitative safety design features to prevent the identified hazards. Table 6-1, which identifies those credited design features used in this SA, is based on new design features added to the FG/RMCS trucks, and Ref. 5 and 6.

Table 6-1 identifies design features with significance to safety credited for mitigation of "offsite" or "onsite" consequences. Combinations of design features with significance to safety and administrative controls were used to meet "offsite" and "onsite" Risk Guidelines (RG). Appendix E describes how the design features and administrative controls are combined to meet the guidelines. No modifications to the design features identified in Section 6 are required to meet the RGs. Therefore, classification of a design feature as "offsite" would not require the modification of a design feature to increase its reliability beyond that credited in Appendix E. For example, redundant sensors, beyond that specified in Section 6, would not be necessary for a detector system identified as "offsite".

6.3.1 Worker Health and Safety Requirements

This safety assessment addresses the risks to onsite individuals at 100 m and to the offsite public, and provides for safety equipment and administrative controls to reduce such risks to within risk guidelines. Included are the risks from radioactive materials, toxic gases, and toxic chemicals, as described in Section 5. The RGs are defined for individuals to 100 m from the source of releases. This SA does not specify any new controls necessary to

August 8, 1996

protect workers who are located closer than 100 m the RMCS activity. The current controls, that protect workers within 100 m, are defined in the contractors Hanford Radiological Control Manual and the Tank Farm Health and Safety Plan (HASP). Those current controls which were credited in this SA are listed in Section 6.

There is a potential for worker exposure to high concentrations of toxic gases during tank intrusive work around open tank risers or from emissions from the ventilation exhauster, including ammonia, nitrous oxide, and various organic species. Concentrations of these gases may be in excess of OSHA-allowed values in the tank vapor space during the rotary sampling work. Because the tank vapor space will be exhausted into the ambient air above the tank a pathway for worker exposure is recognized to exist. Assessment of the hazards from these materials, and protection of the workers is provided by the contractors HASP (Reference 1). Significant elements of this plan include monitoring of the work area for organic vapors, ammonia and other chemical species whenever there is a potential for elevated employee exposure levels. In addition, personnel monitoring is performed on those tasks which are judged to have the highest potential for exposure. Finally administrative barricades have been erected around areas with known vapor releases, and monitoring or protective equipment is required whenever employees work within these areas.

Tank Farms are considered to be a RCRA Treatment, Store and Disposal (TSD) facility, and all work in Tank Farms must comply with the HASP. The contractors Industrial Hygiene staff has responsibility for implementation of the HASP, including monitoring and personnel sampling. Trained industrial hygiene technicians perform monitoring of each task. If any tank vapor levels are detected that present a possibility of employee over exposure, the IH technicians will take appropriate actions, including the use of personal protective equipment or suspending the activity, to ensure that no employee overexposures occur.

August 8, 1996

TABLE 6-1FG/RMCS CREDITED DESIGN SAFETY FEATURES

| General Features | | | | | |
|---|---|--|--|--|--|
| Material compatibility (Onsite) Significant to Safety | Materials used in FG/RMCS activities and the drill string components are compatible with the waste stored in single-shell tanks to prevent chemical action or material failures resulting from expected or accidental contact with the wastes. ^{7,8} | | | | |
| Spark-resistant tools (NA) | Spark-resistant tools are used at all times around open risers or the sampled riser to mitigate mechanical sparking. | | | | |
| Grounding and bonding (NA) | The FG/RMCS equipment is grounded and bonded to mitigate electrical sparking and mitigate the buildup of static electricity. | | | | |
| Radiological controls (NA) | The area around the riser is radiologically monitored, and the exhauster HEPA filter housing is monitored to prevent exposure of personnel to hazardous radiological conditions as set by WHC ALARA controls. | | | | |
| | Riser Adapter Features | | | | |
| Riser sleeve (Offsite) Significant to Safety | A conductive sleeve is inserted into the riser to be sampled to mitigate frictional sparking. In addition, the sleeve is provided with a manually controlled nitrogen purge system to prevent flammable gas accumulation in the riser sleeve. | | | | |
| Drill string spray washer (NA) | A hot-water spray washer is provided on the riser adapter to reduce the contamination of drill rods as they are removed from the tank. Check valves and a positive displacement pump are provided on the system to restrict back flow. | | | | |
| Frisbee/DS interface lubricant (NA) | The frisbee/DS interface is lubricated with a non-spark inducing, waste- compatible material to decrease the friction between the rubber seal and the drill string, and thereby mitigate damage to the seal. | | | | |
| Pneumatic foot clamp (Offsite) Significant to Safety | The pneumatic foot clamp is designed to fail closed to prevent the drill string from falling into the tank when the drill truck or platform hoist are not connected. | | | | |
| Locking wrench (Offsite) Significant to Safety | The locking wrench is a mechanism used concurrently with the foot clamp to ensure the support of the drill string during installation, sampling operations, and removal. | | | | |
| | Drill String Features | | | | |
| Drill bit configuration and material (Offsite) Significant to Safety | The drill bit is made of a waste-compatible material, and must be qualified commensurate with the requirements provided in Appendices F, G and T of this SA. | | | | |
| Drill centering spike (NA) | The first rotary-core sampler shall be equipped with a centering spike on initial use to prevent random bit motion when first entering the waste. | | | | |

| Chevron seal between drill bit and sampler (NA) | The chevron seal between the drill bit and sampler provides a barrier between tank waste and the drill string. This seal allows one-way flow of nitrogen purge from the drill string into the tank, but does not allow waste or waste gas to flow back into the drill string. |
|---|---|
| Core sampler and drill string components (Offsite) Significant to Safety | The core sampler and drill string components must meet the requirements specified in Appendix T to prevent sparking. The section of the core barrel having the serrated edges (grooves) and the quadralatch fingers and body must be made of stainless steel to reduce the likelihood of spark (see Appendix T). |
| Sniffing ports (NA) | Sniffing ports are provided to allow measurement of flammable gas from a contained environment. |
| Change-out assembly (NA) | The change-out assembly is provided with a ball valve to isolate and maintain pressure within the drill string during sampler exchange. |
| Cable spray washer (NA) | The hot-water cable spray washer connects to the drill string to wash cables and samplers that are contaminated. |
| Purge flow limitation (Offsite) Significant to Safety | Drilling Features In FG/RMCS operations, the drill string is purged with a minimum of 30 scfm of nitrogen to prevent drill bit overheating and to remove cutting- products from the drill bit. Purge flow is initiated just before drilling begins. With a loss of nitrogen purge, drilling is automatically terminated with no delay to maintain drill bit temperature within the defined limits. |
| Rotational speed limitation (Offsite) Significant to Safety | In FG/RMCS operations, the maximum rotational speed of the drill string is 55 rpm to maintain the drill bit temperature within the defined range. Drilling is automatically terminated with no delay if the speed exceeds 55 rpm. |
| Down force limitation (Offsite) Significant to Safety | In FG/RMCS operations, a maximum down force of 750 lbf shall be imposed on the drill string to maintain drill bit temperature within the defined range. When the rotational speed is greater than 2 rpm, the set point is selected at 750 lbf upon which shutdown will be initiated. Drilling is automatically terminated with no delay if the down force exceeds 750 lb. |
| Drill string penetration rate (Offsite) Significant to Safety | In FG/RMCS operations, a minimum penetration rate of 0.75 in./min is limited to < 60 seconds to maintain drill bit temperature within the defined range. Drilling automatically terminates when the penetration rate is <0.75 in./min. for a cumulative time of 60 seconds in any 3 min. period. |
| Hydraulic bottom detector (Offsite) Significant to Safety | The hydraulic bottom detector detects the increased resistance during the last sample to prevent increasing the downward force and penetrating the tank bottom by reversing ram motion upward. Two operators are needed to enable the hydraulic interlock system to ensure activation redundancy. The hydraulic interlock system has a pressure-relief valve to control overpressure and check valves to prevent inappropriate flow reversal. |

| Walk Down Function (Offsite) | The walk down function automatically limits down force by controlling lower ram pressure. | | | |
|---------------------------------|--|--|--|--|
| Significant to Safety | | | | |
| Hydrostatic head | Hydrostatic head pressure is maintained and monitored during sample | | | |
| (NA) | retrieval to prevent drill string flooding. | | | |
| | Sampling Truck Features | | | |
| Truck position | All truck ignition sources are greater than 3-ft. from the frisbee or | | | |
| (Offsite) | protected with a barrier to separate potential sources and flammable gases leaking from the frishee | | | |
| Significant to Safety | | | | |
| Stabilizing jacks | The sampling truck is provided with stabilizing jack locking collars to | | | |
| (NA) | prevent truck movement in case of hydraulic failure. | | | |
| Quill rod adapter | The quill rod adapter shall be fabricated of a waste-compatible, spark- | | | |
| (Offsite) | resistant material. | | | |
| Significant to Safety | | | | |
| | Grapple Hoist Assembly Features | | | |
| Grapple hoist assembly | The grapple hoist assembly is designed to meet NEC requirements as | | | |
| (Offsite) | defined in this SA to allow operation in a flammable gas environment. | | | |
| Significant to Safety | | | | |
| Grapple (sample | The mechanical design of the grapple (sample actuator) mitigates | | | |
| actuator) | eleculcal and mechanical sparking. | | | |
| Similicant to Safata | | | | |
| Grannla incertion | The design of the grannle beist assembly limits are a limited as the second sec | | | |
| Grapple Insertion | removal rates to < 1 ft/s to minimize frictional sparking. | | | |
| (INA) | | | | |
| tension (Offsite) | prevent grapple hoist damage. The electrical motor automatically shuts | | | |
| Significant to Safety | down when the load is >250 lb. | | | |
| <u> </u> | Shielded Receiver Assembly Features | | | |
| Shielded receiver | The shielded receiver assembly is designed to meet NEC as defined in this | | | |
| assembly (Offsite) | SA to allow operation in a flammable gas environment. | | | |
| Significant to Safety | | | | |
| SR tube (NA) | Shielding materials are used in the SR to protect against sampler radiation. | | | |
| SR view port (NA) | The SR view port provides a means to inspect the sampler during retrieval to control the spread of contamination. | | | |
| SR hoist cable tension | The SR hoist assembly measures cable tension during sample retrieval to | | | |
| (Ottsite) | prevent damage to the SK hoist assembly. The cable has a structural capacity of 2000 lbf in tension. | | | |
| Significant to Safety | | | | |
| SR hoist motor | The SK hoist motor | | | |
| (NA) | | | | |

| Remote latch unit (Offsite) | The mechanical design of the remote latch unit mitigates electrical and mechanical sparking. |
|--|--|
| Significant to Safety | |
| RLU insertion (NA) | The design of the SR assembly limits RLU insertion and removal rates to \leq 1 ft/s to minimize frictional sparking. |
| RLU position indicator (NA) | The position of the RLU is monitored to ascertain whether the core sampler is completely raised into the SR, or inserted into the drill string. |
| | Exhauster Features |
| Exhauster Operation (Offsite) Significant to Safety | The exhauster continues to operate, as described in Appendix L, but terminates drilling upon detection of flammable gases or dome pressure rise exceeding the specified limits to protect personnel and equipment during a potential gas-release event. |
| Exhauster intrinsic safety (Offsite) Significant to Safety | All electrical equipment within the exhauster flow stream, including the exhauster fan and instrumentation, is qualified to operate in a NEC Class- 1, Division-I, Group B environment to allow operation in a flammable gas environment. |
| Exhauster PLC (Offsite) Significant to Safety | An exhauster programmable logic controller processes out-of-tolerance alarm signals to activate the alarm strobe and indicator lights, and sends a signal to the shut-down logic to initiate drill engine shutdown for all operating parameters listed in the administrative control table. |
| Exhauster duct (Offsite) Significant to Safety | The exhauster duct is made of a conductive material and is grounded to mitigate static sparking. In addition, the duct is attached to the exhauster to prevent damage from wind. |
| Exhauster heater (Offsite) Significant to Safety | The heat exchanger is designed for operation in a flammable gas environment. |
| Exhauster fan and motor assembly (Offsite) Significant to Safety | The exhauster fan and motor assembly as defined in this SA is qualified for operation in a flammable gas environment. |
| Inlet breather stack (Offsite) Significant to Safety | The tank inlet breather stack shall be at least 15-ft tall with a nominal 4-in. i.d. to reduce toxicological consequences and to protect non-qualified equipment on the tank dome during GRE. |
| Tank pressure detection (Offsite) Significant to Safety | The exhauster is equipped with a tank pressure detection system to prevent low tank pressure. |
| - • | Flammable Gas Detection |
| Flammable gas detector (Offsite) Significant to Safety | The flammable gas detection system, with redundant sensing systems, is provided to meet the functional requirements and performance acceptance criteria as defined in Appendices C and U to terminate FG/RMCS operations in the event of a GRE. |
| | Tank pressure is also measured to provide GRE detection and diversity. |
| | MODILE A-ray System reature |

| X-ray containment | The mobile X-ray system has a sealed, conductive-tube enclosure and redundant sleeve to separate waste and waste-generated flammable gases | | | | | |
|---|---|--|--|--|--|--|
| | from the electrical components of the system. | | | | | |
| Nitrogen System Safety Features | | | | | | |
| DS nitrogen purge supply (Offsite) | The DS nitrogen purge supply is provided to prevent drill bit overheating. Automatic termination of drilling operations on low purge flow is provided to prevent drill bit overheating. | | | | | |
| Significant to Safety | | | | | | |
| Nitrogen hydrostatic head supply (NA) | Maintaining the hydrostatic equilibrium in the drill column with hydrostatic head nitrogen is controlled by a flow controller to ensure that excessive amounts of nitrogen are not injected into the waste. | | | | | |
| Riser sleeve nitrogen purge supply (NA) | The riser sleeve nitrogen purge flow is provided to prevent flammable gas accumulation in the annulus between the riser sleeve and drill string. Automatic termination of drilling operations occurs on loss of flow. | | | | | |
| Unique connections (NA) | Unique connections are provided for both the DS and riser/sleeve annulus nitrogen purge systems, as well as the hydrostatic head systems for the DS and SR, to prevent incorrect connections. | | | | | |
| Inst | rumentation, Control, and Interlock Features | | | | | |
| Truck PLC (Offsite) | The truck programmable logic controller processes out-of-tolerance alarm signals to activate the alarm strobe, horn, and indicator lights and to send a signal to the shut-down logic to initiate drill engine shutdown for all | | | | | |
| Significant to Safety | operating parameters listed in the administrative control table. | | | | | |
| Audible and visual annunciation (NA) | Audible and visual annunciation is provided to alert the operator to out- of-tolerance alarm conditions as listed in the administrative control table. | | | | | |
| Shut-down interlock | The shut-down interlock accepts signals from the PLC to shut down the | | | | | |
| (Offsite) | drill rig engine ignition. | | | | | |
| Significant to Safety | | | | | | |

6.4. **FG/RMCS CONTROLS**

The control tables in this section are discussed in terms of the applicable phases, the descriptions of the columns and several definitions that are pertinent to accurately interpreting the controls.

6.4.1. FG/RMCS Controls Phases

- All of the controls developed for FG/RMCS activities have been categorized into one of the following 5 phases:
 - All Phases—these controls will apply to all phases of FG/RMCS operations. See Section 6.6, Table 2.

- Pre-installation Phase—controls in this phase are focused at the activities necessary to prepare the site, acquire and set up equipment, and perform grounding and bonding activities. See Section 6.7, Table 3.
- Installation Phase—these controls are directly related to all activities associated with opening and preparing the riser, and inserting the drill string. See Section 6.8, Table 4.
- Operations Phase—this phase contains all of the controls related to sampling operations and sampler handling. See Section 6.9, Table 5.
- Removal Phase—this final phase includes the decontamination of the drill string, breaking down the equipment and restoring the site. See Section 6.10, Table 6.

6.4.2. Control Table Columns

Each of the tables has the following columns:

- System or condition—defines the situation or equipment for which the control is being developed.
- Safe operating condition-defines the levels or conditions under which safe operation occurs.
- Surveillance or monitoring method—defines the surveillance or monitoring method used to check the operation condition.
- Administrative controls—defines the controls for the given system or equipment in terms of operation or performance.
- Level—defines the Level 1, 2 or 3, as described in Sections 6.1 and 6.2.
- Basis for Control—defines the reason behind or necessity for the control.
- Safe Shutdown Definition—defines the conditions and actions necessary if the control requirements are not met.

| System or Condition | SCO | Surveillance/ | Administrative Controls | Level | Basis for Control | Safe Shut Down |
|---|---|--------------------------|--|-------|--|--|
| | | Monitoring | | | | Condition |
| 6.5.ALL PHASES / | | | | | | · |
| 6.5.1.Use of FG/RMCS C | Controls | ********* | I | • | *************************************** | |
| 6.5.1.1. Control Applicability | Approved checklist | Checklist evaluation | This Safety Assessment and its controls shall be applicable to single-shell tanks (SSTs) on the Flammable Gas Watch List (FGWL) or those tanks recommended by the contractor to be included on the FGWL. This SA does not cover the RMCS operations in tanks involving floating organic layers or organic remnant layers and ferrocyanide tanks as specified in Appendix G. Each tank shall be evaluated against the checklist in Section 7, and the evaluation results shall be reviewed and approved by the PRC before initiating any FG/RMCS operations. | 1 | SA assumptions in regard to tank specific parameters such as gas and waste composition, gas release probability etc. may become non- conservative by a new analysis or data for a specific tank considered to be sampled by RMCS prior operations. See section 7 and Appendix G. | No FG/RMCS operations can proceed without checklist approval. If FG/RMCS operations are initiated without checklist approval, immediately terminate activities and notify plant management. |
| 6.5.1.2 Existing procedures and controls | Use controls identified in this document for specified phases of FG/RMCS operations. | Procedural evaluation | The PRC shall verify that all FG/RMCS controls specified in this SA are implemented in the operating procedures. Non-RMCS-related activities on the flammable gas tank being sampled shall be performed under existing Contractor controls and procedures, unless superseded by FG/RMCS controls. | 3 | SA analyses requires changes in current procedures. New procedures are needed to maximize safety to personnel, equipment and the environment. Provides defense in depth to all aspects of safety. | NA |
| 6.5.2.Flammable Gas De | tection System | | | | | |
| 6.5.2.1. Flammable gas detection system calibration | Use of calibrated detection system | Periodic calibration | New SMC sensors shall be initially calibrated in a laboratory environment using hydrogen and ammonia. A functional SMC calibration test shall be performed in the field at initial setup and then every day the flammable gas detection | 1 | The use of a redundant and adequate flammable gas detector is a key assumption to provide safe shutdown of FG/RMCS | No FG/RMCS activities shall proceed without the use of a calibrated flammable gas detection system. |

| System or Condition | SCO | Surveillance/ Monitoring | Administrative Controls | Level | Basis for Control | Safe Shut Down Condition |
|---|---|--|---|-------|---|--|
| | / | | A functional SMC calibration test shall be performed in the field at initial setup and then every day the flammable gas detection system is used. The Whittaker sensor shall be calibrated every three months. The functional calibration test shall consist of setting the zero with pure air or nitrogen and calibration with nominal 6000 ppm hydrogen for both sensors. The functional calibration test procedure at a nominal 6000 ppm hydrogen shall test the shutoif electronics as well as the sensor reading for both sensors. | | detector is a key assumption to provide safe shutdown of FG/RMCS operations during a GRE. This reduces the likelihood of spark in flammable gas atmosphere and also provides protection for toxic gas exposures. This mitigates unacceptable offsite consequences resulting from a dome collapse. | calibrated flammable gas detection system. |
| 6522 Flammable gas detection system operation | Limit setpoint of flammable gas detection system at <u><</u> 25% LFL. | Flammable gas detection system connected between riser and exhauster housing with system readout and alarmed setpoints One out of two redundant logic | A redundant, calibrated flammable gas detection system shall be operational before FG/RMCS waste intrusive operations begin, and operate consistent with exhauster operation defined in Section 4.1.3. The flammable gas detection system shall be trip setpoint limited at: • > 5000 ppm hydrogen concentration equivalent; and • > 100 ppm/s rate of equivalent hydrogen concentration increase over a 10-s period. Send a shutdown signal to the truck PLC. One out of two redundant logic. SMC sensors shall be replaced at least once each month with new sensors. | 1 | System detects flammable gas accumulation sufficient for deflagration or detonation in tank. Detects GRE and provides protection for fire and toxic hazards. Reduces the likelihood of spark in flammable gas atmosphere. Provides protection for exposure to toxic gas releases. This mitigates unacceptable offsite consequences resulting from a dome collapse. Section 4 and | Upon detection, the exhauster will continue to operate. If drilling, the drill rig engine will automatically trip. All personnel within a 100-m radius from the edge of the tank will evacuate and don protective equipment before returning for further equipment stabilization. FG/RMCS activities can resume in compliance with appropriate controls in Section 4.1.1. |

| WHC-SD-WM-SAD-035, | Rev. 0-a |
|--------------------|--------------------|
| | WHC-SD-WM-SAD-035, |

| System or Condition | SCO | Surveillance/ Monitoring | Administrative Controls | Level | Basis for Control Appendix B. | Safe Shut Down Condition |
|--|--|---|---|--------|---|---|
| 6.52.3. Tank gas pressure detection system | Detect rate of tank pressure increase. | Gas pressure detection system One out of two redundant logic | A redundant gas pressure detection system shall detect an increase in tank pressure of > 2 in. w.g. in any 5 minute period. Send a shutdown signal to the truck PLC. One out of two redundant logic. | 1 | Prevent burn in and out of dome by detecting gas release rates >1000 ft ⁷ /min. Provides protection for fire and toxic hazards. Reduces the in flammable gas atmosphere. Provides protection for exposure to for exposure to f | Upon detection, the exhauster will continue to operate. If drilling, trip. All personnel within a 100-m ration in a 100-m ration prometive edge of the tank will evacuate and don protective equipment before returning for further equipment stabilization. FG/RMCS activities can proceed in compliance with appropriate controls in Section 6.4. |
| 653Spark-free environ | nent | | | •••••• | | |
| 6.5.1. Spark-free environment | Use spark-resistant tools, materials, and FG/RMCS components at all times within a radius of 36 in. of open riser or sampled riser | Procedural evaluation | An administrative control shall be developed and implemented that imposes an environment that significantly reduces the likelihood of sparking by: • The use of spark-resistant tools, materials, and FG/RMCS components at all times within a radius of 36 in. of the riser being sampled or any open riser. • Nonsmoking environment on top of the tank. | σ | Prevent ignition sources in proximity with flammable gas environments. Prevent fire and its propagation above the tank dome. See Sections 4.3.3, 4.2.2, and 4.2.3. | No FG/RMCS activities can be initiated or proceed until a spark-free environment is established. |
| 65.4 Material compatibil | lty | | | ¢ | | |
| 65.4.1. Material | Use waste compatible | Procedural | An administrative control shall be | 2 | Prevent gas | No FG/RMCS |

August 8, 1996

6-13

| System or Condition | SCO | Surveillance/ | Administrative Controls | Level | Basis for Control | Safe Shut Down |
|--|--|--------------------------|---|-------|---|--|
| | | Monitoring | | | | Condition |
| compatibility | materials and lubricants | evaluation | developed and implemented that requires, when in contact with the DS or tank waste, the use of: • waste-compatible materials, and • pipe compounds, lubricants, seals and tapes that do not contain spark-inducing or waste- incompatible materials. | | generation caused by material incompatibilities. Protects against local exothermic chemical reactions that could produce large quantities of flammable gas and heat and cause a waste fire if propagates. Prevents equipment damage. Sections 4.3.1 and | activities can be proceed until the use of waste compatible materials is implemented. |
| 655 Exhauster emission | | <u>I</u> | Capturate Capturate | ••••• | 4.4.2. | |
| 656Simultaneous activ | ities | ****** | Contractor Controls | ••••• | See Section 4.1.1 | |
| AEA1 EC/DMCC | | | | | | |
| vehicle operations on tank | operations during open riser conditions or waste-intrusive activities | evaluation | An administrative control shall be developed and implemented that evaluates flammable gas concentrations in the FG/RMCS tank during open riser conditions or waste-intrusive activities before any vehicle operations. | 3 | Minimize the potential for vehicle operations igniting flammable gas concentrations. See Section 4.1.7. | Upon detection, vehicle operations will be immediately terminated and not resumed until flammable gas limits are met. |
| 6562 Simultaneous activities on FG/RMCS tank | Limits on simultaneous non- FG/RMCS activities on sampled tank | Procedural evaluation | An administrative control shall be developed and implemented that prevents non-FG/RMCS activities on the tank during waste-intrusive operations. Simultaneous dome-intrusive activities can be performed provided: the equipment is qualified for operation in a Class I, Division 1, Group B environment, and operation is based on its own safety assessment. operation does not physically interact with the drill string, and | 1 | Tank dome is considered to be Class I, Div. 1, Group B environment because RMCS can induce a GRE. Sparks in the dome or volumes having a connecting path to the tank dome must be controlled. Control prevents ignition of flammable gas by non-FG/RMCS- related ignition sources. | No non-FG/RMCS activities can proceed during period defined by control, or until any simultaneous tank activities comply with the control. |

| Ĩ |
|-----------|
| 0 |
| ~ |
| 5 |
| ž |
| |
| ഹ് |
| 3 |
| Ģ |
| Ó |
| |
| 5 |
| ဟု |
| ÷ |
| 4 |
| 2 |
| - |
| \square |
| ភ |
| 1 |
| Q |
| T |
| 5 |
| 5 |
| |

| Safe Shut Down Condition | | No non-FG/RMCS activities can proceed during period defined by control, or until all simultaneous simultaneous activities on interconnected tanks comply with the control. | | All FC/RMCS activities will be terminated until flammable gas concentrations are within limits specified in Section 6.9. |
|-----------------------------|---|--|---------------------------------|---|
| Basis for Control | This mitigates unacceptable offsite consequences resulting from a dome collapse. See Section 4.2.7 and Appendix B. | Tank dome is considered to be Class I Div. 1 Group B environment because RMCS can induce a CRE. Sparks in the dome or volumes having a connecting path to the tank dome must be controlled. Control mitigates gas release or fire from interconnected tank resulting in burn in FG/RMCS tank. See Section 4.2.7 | | Limit flammable gas concentrations in proximity with ignition sources. Operations when the tank dome have flammable gas concentration above 25% LFL is prohibited to prevent dome fires. Control reduces the likelihood of fire accident in the dome. |
| Level | | ς | | m |
| Administrative Controls | is not waste-intrusive. | An administrative control shall be developed and implemented that prevents non-FG/RMCS activities on interconnected tanks when waste-intrusive activities are in progress. If necessary, an action plan shall be prepared for PRC review and approval to proceed with noncompliant activities. | | No refueling operations shall be performed inside the tank farm when FG/RMCS tank flammable gas concentrations are 25% LFL. |
| Surveillance/ Monitoring | | Procedural evaluation | | Procedural evaluation |
| sco | ~ | Limits on simultaneous non- FG/RMCS activities on interconnected tanks | 5 | Limits on refueling with flammable gas concentrations_25% LFL |
| System or Condition | | 6.5.6.3. Inter- connected tanks | 657.Refueling Activition | 6.5.7.1. Flammable gas concentrations |

August 8, 1996

6-15

August 8, 1996

9-1*F*

non chuine Traibh chuine

| System or Condition | SCO | Surveillance/ Monitoring | Administrative Controls | Level | Basis for Control | Safe Shut Down Condition |
|---|--|-----------------------------|--|-------|--|--|
| | / | | | | Prevents equipment damage. See Section 4.9.3. | · · · |
| 6.5.9.3. Thunder-storm & lightning, high winds, dust devils and tornadoes | No thunderstorm activity or lightning strikes within 50 mile radius No sustained wind velocities > 25 mph | Procedural evaluation | At the discretion of the PIC, no FG/RMCS activities shall be performed when: • thunderstorm activity or lightning strikes are reported within a 50 mile radius. • Sustained wind velocities are > 25 mph. • Dust devil activity that can disrupt operations • tornado activity has been predicted within the next 8 hours within a 50 mile radius. | 3 | Prevent personnel injury, equipment damage, and reduce potential for electrostatic sparking. See Section 4.1.2 and 4.9. | As required by General External Event control. |
| 6.5.9.4. High rain fall or flooding, seismic activity, and volcanic activity/ashfall | Reduce potential for personnel injury and equipment damage. | Procedural evaluation | An administrative control shall be developed and implemented that assesses the limits of safe FG/RMCS activities with: • excessive rain fall • extreme temperature conditions • seismic activity • volcanic activity or ashfall accumulation | 3 | Prevent personnel injury and equipment damage. See Section 4.9.4, 4.9.6. and 4.9.7 | As required by General External Event control. |

| System or Condition | 500 | Surveillance/ Monitoring | Administrative Controls | Level | Basis for Control | Safe Shut Down Condition |
|------------------------------|---|---|--|-------|--|--|
| 6.6.PRE-INSTALLA | TION PHASE | | | | | |
| 661.Waste Reactivity E | :valuátion | | | | | |
| 6.6.1.1. Waste reactivity | FG/RMCS operations on tanks specified in Appendix G of this SA | Procedural implementation | FG/RMCS operations shall be performed only in tanks listed in Table G-3 of Appendix G in this SA. | _ | Prevents propagating exothermical evothermical chemical reaction that could produce large quantities of flammable and toxic gas and heat. Maste and flammable gas ignition consequently a possible dome collapse is prevented. This mitigates unacceptable offsite consequences resulting from a dome collapse. See Appendix G | No FG/RMCS proceed in tanks not listed. |
| 662Tank and riser cor | iditions | | | | | |
| 6621. Gas leak paths | Identify and repair leaking risers prior to waste-intrusive activities | Visual inspection Procedural evaluation | An administrative control shall be developed and implemented for a formal tank walkdown prior to waste-intrusive activities that: • assesses the general condition of risers, • identifies observable leaks • as a minimum, documents identified leaks > 1 in. effective diameter, • as a minimum, seals or add deflectors to identified leaks with a leak diameter > 1 in. effective | | Above ground fires that could propagate into the dome are managed by controlling position of the non- qualified equipment. Since sparks from non- qualified a eliminated, a be eliminated, a criteria is developed in App. B. This criteria requires that | No FG/RMCS activities can proceed until procedural controls are implemented. |
| | | | diameter. | | poŝsible leak | |

August 8, 1996

7 400 por 1/20

6-18

| System or Condition | SCO | Surveillance/ | Administrative Controls | Level | Basis for Control | Safe Shut Down |
|--|---|---|---|--------|---|--|
| 66.2.2 Open riser exclusion zone | Exclusion zone with radius of 36-riser- diameters around open risers during waste-intrusive activities | Procedural evaluation and verification | An administrative control shall be developed and implemented for an exclusion zone with a radius of 36-riser-diameters around any open riser during waste-intrusive FG/RMCS activities. | 1 | diameter is to be known and controlled. This mitigates unacceptable offsite consequences resulting from a dome collapse. See Section 4.1.2.1 and Appendix B. Minimizes ignition sources in proximity with flammable gas. Minimizes gas leakage in proximity with ignition sources or unqualified equipment. This mitigates unacceptable offsite consequences resulting from a dome collapse. See Section 4.1.2.1 and Appendix B | No FG/RMCS activities can proceed until procedural controls are implemented. |
| 6.6.3.Grounding and be | onding | . | | ****** | | ************ |
| 6.6.3.1. Equipment attached to tank | Ensure equipment grounding | Resistance measurement Procedural evaluation and verification | An administrative control shall be developed and implemented for grounding and bonding all FG/RMCS equipment attached to the riser or inserted into the tank, including the drill string such that: Electrical power grounding shall comply with the requirements of NFPA 70, Article 250. Lightning protection shall be consistent with the principles | 3 | Prevent static electricity discharge and lightning strike initiated sparks. See Sections 4.1.1.1, 4.2.5, 4.3.10, 4.91. | No FG/RMCS activities can proceed until grounding and bonding activities are completed and verified. |

and a second sec

| System or Condition | SCO | Surveillance/ | Administrative Controls | Level | Basis for Control | Safe Shut Down |
|--|--|---|--|-------|---|--|
| | / | Monitoring | stated in "Evaluation of Hazards From Lightning Strikes to Tank Farm Facilities", W. L. Cowley, WH-SD-WM-SARR-027, Rev 0 (June 29,1994). | | | Condition |
| 6.6.3.2 Equipment not attached to tank | Ensure equipment grounding | Resistance measurement Procedural evaluation and verification | Equipment not attached to the riser or inserted into the tank shall be grounded commensurate with Contractor grounding and bonding controls, using independent verification before FG/RMCS activities. | 3 | Prevent static electricity discharge and lightning strike initiated sparks. See Section 4.1.1. | No FG/RMCS activities can proceed until grounding and bonding activities are completed and verified. |
| 664.Energized equipm | ient | | *************************************** | ***** | *************************************** | ******* |
| 6.6.4.1. Energized equipment in the dome | All energized equipment exposed to the tank dome (as defined in approved contractor safety documentation) or dome of connecting tanks shall be rated for operations in Class I, Division 1, Group B environment or Class I, Division 2, Group B environment with automatic shut down for flammable gas concentrations >25%LFL. | Procedural evaluation of verification and certification | During waste-intrusive operations, all energized equipment exposed to the tank dome vapor space (as defined in approved contractor safety documentation) or dome vapor space of connecting tanks shall be rated for operations in Class I, Division 1, Group B environment or Class I, Division 2, Group B environment with automatic shut down for flammable gas concentrations >25%LFL. All existing energized equipment not meeting the above control shall be de-energized. | 1 | Prevents ignition sources in proximity with flammable gases. This mitigates unacceptable offsite consequences resulting from a dome collapse. See Section 4.0, 4.1.1.3, 4.3.4 and Appendix B. | No FG/RMCS activities can proceed until equipment in the tank dome or dome of connecting tanks complies with control. |
| 6642 Energized equipment in open riser exclusion zones | Qualifications of energized equipment in open riser exclusion zones. | Procedural evaluation of verification and certification | During waste-intrusive operations, all energized equipment in open riser exclusion zones as defined in section 6.6.2.2 shall be rated for operations in Class I, Division 1, Group B environment, or rated for Class I, Division 2, Group B environment with automatic shut down for flammable gas concentrations >25%LFL. All energized equipment in riser exclusion zones not meeting the above control shall be de- energized. | 1 | Prevents ignition sources in proximity with flammable gases. This mitigates unacceptable offsite consequences resulting from a dome collapse. See Section 4.1.2.1. | No FG/RMCS activities can proceed until equipment in the riser exclusion zones complies with control. |

| System or Condition | SCO | Surveillance/ Monitoring | Administrative Controls | Level | Basis for Control | Safe Shut Down Condition |
|---------------------------------------|---|---|---|----------|--|--|
| 665.Equipment positi | oning and Setup | ****** | | [| , | |
| 6651. Tank loading | Preyent loading exceeding tank structural capabilities. | Procedural evaluation Engineering review for equipment positioning on tank top | Loading on each tank shall comply with IOSR requirements for simultaneous static and dynamic loading for each specific FG/RMCS tank. | 1 | Limits on tank loading can prevent collapse. This mitigates unacceptable offsite consequences resulting from a dome collapse. See Section 4.6.1 and Appendix N. | No equipment positioning or set up can proceed until tank load capabilities are verified. |
| 66.5.2. Vehicle operations on tank | Prevent vehicle impacts. | Procedural evaluation | An administrative control shall be developed and implemented for controlling the safe operation of vehicles on top of the tank during waste-intrusive activities. | 3 | Prevent ignition sources caused by vehicle impacts. See Section 4.1.7. | With procedural noncompliance, corrective actions must be implemented before resuming any FG/RMCS activities. |
| 6.6.5.3. Truck ramp and jacks | Safe use of ramps and hydraulic jacks. | Procedural evaluation | An administrative control shall be developed and implemented for the safe use of the ramp supporting the sampling truck and the truck's hydraulic jacks. | 3 | Prevent tank riser damage See Section 4.6.1. | With procedural noncompliance, corrective actions must be implemented before resuming any FG/RMCS activities. |
| 66.5.4 Truck position | No truck-induced ignition sources near potential gas leak source | Procedural evaluation | An administrative control shall be developed and implemented that, prior to waste-intrusive operations: • prevents positioning the sampling truck over an open riser, and • seals any risers under the truck, and • raises the truck a minimum of 36 inches between any potential ignition source on the truck and the top of any riser or pit over which the truck is positioned. | 1 | Truck when parked over an unused riser must be protected from flammable gas releases. This requires to control the distance between the leak source and truck. Reduces above tank fire frequencies by mitigating human error and minimizing ignition sources | No FC/RMCS activities can proceed until procedural controls are implemented. |

WHC-SD-WM-SA. J35, Rev. 0-a

| System or Condition | SCO | Surveillance/ Monitoring | Administrative Controls | Level | Basis for Control | Safe Shul Down Condition |
|------------------------------------|--|-----------------------------|---|-------|---|---|
| | / | | | | within area surrounding leak sources. This mitigates unacceptable offsite consequences resulting from a dome collapse. See Section 4.1.2.1. | |
| 6.6.5.5. Lift operations | Prevent sparks induced from equipment lifting over the tank. | Procedural evaluation | An administrative control shall be developed and implemented that, during waste-intrusive activities: • controls safe lifting operations over the tank, and • restricts lift operations over the exhauster during installation, sampling or restart activities. | 3 | Minimizes potential for above tank fires caused by frictional spark induced by load impacts on tank and exhauster. See Section 4.1.1.4 and 4.1.2. | No lifting of FG/RMCS equipment, during waste-intrusive activities, over the tank can proceed without the procedure. With procedural noncompliance, corrective actions must be implemented before proceeding with any FG/RMCS activities. |
| 6.6.6 Exhauster setup | | .k | | 1 | | ****** |
| 6.6.6.1. Exhauster installation | Prior to waste- intrusive activities, ensure accurate exhauster positioning, set-up and maintenance per procedures | Procedural evaluation | An administrative control shall be developed and implemented that, prior to waste-intrusive activities, controls the positioning, setup and maintenance of the exhauster system. | 3 | Optimum exhauster performance mitigates unfiltered releases. See Section 5.4.2. | No waste-intrusive FG/RMCS activities can proceed without procedure. With procedural noncompliance, corrective actions must be implemented before proceeding with any FG/RMCS activities. |
| 6662 Exhauster | i sevent duct-induced | Procedural | An administrative control shall | 3 | Grounding | No FG/RMCS |

| System or Condition | sco | Surveillance/ | Administrative Controls | Level | Basis for Control | Safe Shut Down |
|---------------------------------|---|--|--|-------|---|---|
| | | Monitoring | | | | Condition |
| duct | static ignition sources | evaluation and verification | be developed and implemented that controls the following: The exhauster duct shall be mounted in compliance with Contractor procedures. The exhauster duct shall be inspected for leaks after the exhauster is installed, with independent verification. | | verification controls sparking caused by static electricity, lightning, etc. Prevents exposure of flammable gas to non-qualified equipment around flexible hose. See Section 4.1.1, 4.1.2. | activities can proceed without procedural compliance. |
| 6663. Duct exclusion zone | No unqualified equipment < 18 in. from exhauster duct | Procedural evaluation and verification | An administrative control shall be developed and implemented that locates unqualified equipment greater than 18 in. away from exhauster duct, or provides deflectors for equipment located within 18 in., using independent venfication. | 3 | Prevent undetected hose leak in contact with unqualified equipment resulting in fire. See Section 4.1.2. | No FG/RMCS activities can proceed until exhauster duct exclusion zone is established. |
| 6664 Exhauster fan | Prevent ignition source from exhauster fan. | Procedural evaluation | The exhauster fan shall meet the requirements of this SA for operation in flammable gas environment commensurate with Std 99-0401-86, Air Movement and Control Association, and NFPA 91. | 3 | Prevent a fan- induced ignition source in a potentially flammable env ironment. See Section 4.1.1.4. | No FG/RMCS activities can proceed until ignition sources are eliminated from the exhauster's potentially flammable environment. |
| 667.Tank Breather Inl | et HEPA Filter | | | | | |
| 66.7.1. Portable Inlet Stack | Controlled releases from stack during GRE | Procedural evaluation and verification | An administrative control shall be developed and implemented that, prior to waste-intrusive activities, ensures that inlet breather filter effluent (in the event of tank pressurization) is directed vertically to a height of at least 15 ft above ground level. | | Directing the breather inlet HEPA effluent 15 ft vertically: • allows atmospheric dispersion coefficients to decrease so that toxicological acceptance guidelines at the onsite boundary can be met. | No waste-intrusive FG/RMCS activities can proceed until inlet HEPA effluent is directed 15 ft vertically. |

| System or Condition | SCO | Surveillance/ Monitoring | Administrative Controls | Level | Basis for Control | Safe Shut Down Condition |
|---------------------|-----|-----------------------------|-------------------------|-------|--|-----------------------------|
| | / | | | | • prevents electrically unqualified equipment in proximity with flammable gas. This mitigates unacceptable offsite consequences resulting from a dome collapse. | |
| | | | | | See Section 4.1.2.1. | |

| System or Condition | \$CO | Surveillance/ | Administrative Controls | Level | Basis for Control | Safe Shut Down |
|---|---|--|---|-------|--|--|
| | | j Monitoring | <u> </u> | | | Condition |
| 6.7.INSTALLATION | PHASE | | | | | |
| 6.7.1.Riser preparation | | | | | | |
| 6.7.1.1. Gas concentrations while breaching tank containment | Detect flammable gas concentrations | Flammable and toxic gas monitoring procedures | An administrative control shall be developed and implemented for monitoring the riser for flammable and toxic gas concentrations every time before opening a riser. | 3 | Detects flammable gas concentrations sufficient for deflagration before installing riser equipment. | No FG/RMCS activities can proceed without procedural compliance. |
| | | | | | and 4.3 involving accidents during installation phase. | |
| 67.1.2 Riser exposure | Limit personnel exposure | Radiation monitoring procedures | Direct exposure to open riser shall be limited to values specified by approved Tank Farm Operating Procedures. | 3 | Minimize personnel direct exposure from open riser conditions. See Section 4.8.8.4. | No FG/RMCS activities can proceed until personnel protection is appropriately ensured. |
| 6.7.1.3. Riser equipment | Prevent tool drops to waste through open riser. | Procedural evaluation | Activities and use of tools around open risers shall be strictly controlled, especially during installation and removal of the riser adapter, frisbee and washer and foot clamp assemblies, in compliance with approved Tank Farm Operating Procedures. | 3 | Limited open riser time reduces the potential for aerosol, particulate and gas releases. Frequency of gas release accidents considers a 8-hour open riser period. See Section 4.2.3 and 5.4.1. | No FG/RMCS activities can proceed until procedural corrective actions are implemented. |
| 6.7.1.4 Sleeve insertion | Minimize ignition sources from sleeve insertion | Procedural evaluation | An administrative control shall be developed and implemented for installing the riser conductive sleeve. • The sleeve insertion and removal rate shall be limited to <u><</u> 1 foot/sec. | 3 | Equipment drops are potentially spark inducing. Control further reduces the likelihood of spark caused by drops. See Sections 4.2.3, 4.2.4, and 4.4.3. | No FG/RMCS activities can proceed until procedural corrective actions are implemented. |

| System or Condition | SCO | Surveillance/ Monitoring | Administrative Controls | Level | Basis for Control | Safe Shut Down Condition |
|---|--|---|---|-------|--|---|
| 67.1.5 Frisbee installation | Minimize flammable gas leakage from frisbee / | Procedural evaluation Visual inspection | An administrative control shall be developed and implemented for inspecting frisbee to verify that: • the frisbee inner diameter is smaller than the drill rod outer diameter, and • the frisbee is in good physical condition before DS insertion. | 3 | Control minimizes frictional sparking potential between riser and sleeve. See Section 4.1.2.1. | No DS insertion or operations can be initiated unless functional frisbee is in place. |
| 6.7.1.6 Frisbee/DS interface lubrication | Improved contact between frisbee and DS Mitigate frisbee damage. | Procedural evaluation and verification | The frisbee/DS interface shall be lubricated with a waste- compatible, spark-resistant lubricant before DS insertion. | 3 | Correct frisbee/DS interface reduces gas leak potential during operations. Lubricant eases drill rod insertion and removal, and mitigates seal damage, therefore gas/particulate releases. See Section 4.2.5. | No DS insertion can be initiated unless frisbee is lubricated. |

| System or Condition | \$CO | Surveillance/ Monitoring | Administrative Controls | Level | Basis for Control | Safe Shut Down Condition |
|---|--|---|--|-------|--|---|
| 6.8.OPERATIONS P | HASE | t | 1 | ŧ | | |
| 6.8.1.Drill Bit | | | I | | | |
| 6.8.1.1. Drill bit qualification | Use of qualified FG/RMCS drill bit | Procedural implementation. | Only drill bits qualified by BOM and envelope testing, as specified in Appendix T, shall be used for FG/RMCS sampling operations. | 1 | FG/RMCS operational parameters, including downforce, RPM, purge flow rate and penetration rate were derived and verified using the drill bit specified in Appendix T. Frictional spark issues were verified for the specified bit based on BOM test results. | FG/RMCS sampling activities can proceed only with a qualified drill bit. |
| 6.8.2.Core Sampler | • | | | | | |
| 6.8.2.1. Core sampler qualification | Use of qualified core sampler | Procedural implementation. | Only core samplers qualified by BOM and envelope testing, as specified in Appendix T, shall be used for FG/RMCS sampling operations. | 1 | DS fire and frictional spark potential for the core sampler specified in Appendix T are negated by BOM test results. | FG/RMCS sampling activities can proceed only with a qualified core sampler. |
| 6.8.2.2. Leakage | Reduce contamination from sampler leakage | Procedural evaluation Visual inspection | An administrative control shall be developed and implemented for: • visually inspecting the sampler for leakage during retrieval, and • properly handling a leaking sampler. | 3 | Leaking samplers result in equipment contamination, and the potential for gas generation and accumulation in uncontrolled spaces. See Section 4.1.2.2 and 4.1.3. | No FC/RMCS activities can proceed until leakage is contained. |
| 6.8.3.Drill string | | | | | | |
| 6.8.3.1. Drill bit, core barrel and drill rods | Baselined performance of rotary drilling equipment | Procedural evaluation QA/QC over | The FG/RMCS drill bit, core barrel and drill rods shall be of the configuration and material | 1 | Results from envelope testing and ignition testing | No waste-intrusive FG/RMCS activities can |

August 8, 1996

WHC-SD-WM-SA_ J35, Rev. 0-a

| System or Condition | i sco | Surveillance/ | Administrative Controls | Level | Basis for Control | Safe Shul Down |
|----------------------------------|------------------------------------|---|--|-------|---|--|
| | <u>.</u> | Monitoring | | | | Condition |
| | | selection and use of materials | tested by the Bureau of Mines and performance evaluated in this 5A. | | are only valid for equipment of the design tested. This mitigates unacceptable offsite consequences resulting from a dome collapse. See Section 4.4.4. | proceed until incorrect drilling equipment is replaced or tested and evaluated for performance. |
| 6.8.3.2 DS open to atmosphere | Detect flammable gas accumulation. | Flammable gas meter Procedural evaluation | An administrative control shall be developed and implemented for: • sniffing the contained DS before the quill rod is disconnected, and • isolating the DS immediately (maximum of 60 minutes) after the drill string is disconnected from the quill rod. • Sniffing the DS before breaking containment if left isolated for > 60 minutes. | 3 | Sniffing detects gas accumulation in the DS and reduces potential for a fire. Immediate change- out assembly installation mitigates flammable gas interactions. See Section 4.1.2.2 and 4.3.3. | No sample retrieval operations can proceed without DS sniffing commensurate with control. |
| 68.3.3. Gas accumulation | Purge drill string | Procedural evaluation | An administrative control shall be developed and implemented for sniffing and purging the DS sufficiently to evacuate flammable gas accumulations before FG/RMCS activities if: • sampler is left in DS without nitrogen flow or hydrostatic pressure for ≥ 60 minutes; or • DS hydrostatic head is lost with no sampler in place. | 3 | Purging eliminates gas accumulation in the DS and reduces potential for a fire. See Section 4.3.1. | No FG/RMCS waste-intrusive activities can proceed until requisite purging activities have been completed. |
| 6.8.3.4 Waste flooding | Minimize flooded waste in DS. | SR visual inspection of sampler or loss of hydrostatic head during change-out activities | The DS shall be flushed and purged with indications of waste flooding in the DS, based on approved Tank Farm Plant Operating Procedures, to the extent practicable. | 3 | Waste flooding in the DS can result in gas generation and accumulation. See Section 4.1.4. | No waste-intrusive FG/RMCS activities can proceed until control is implemented, and the source of flooding is verified. |

| System or Condition | sco | Surveillance/ Monitoring | Administrative Controls | Level | Basis for Control | Safe Shut Down Condition |
|---------------------------------|--|---|--|-------|---|---|
| 684Shielded Receiver | | | | | | |
| 68.4.1. Sampler retrieval | Ensyre sampler completely in SR | Visual inspection of sampler | An administrative control shall be developed and implemented for verifying that the sampler is completely within the SR when held by the RLU before disconnecting the SR from the DS, the X-ray or the cask. | ñ | Prevents spills and waste discharges from samplers, which mitigates unanticipated waste releases to the environment. See Sections 4.1.4 and 4.8.4. | If sampler is not completely in SR, no FG/RMCS activities can proceed until procedural corrective actions are implemented. |
| 68.42 Waste accumulation | Detect waste accumulation >100 mrem/hr. | Radiological monitoring procedures | The external SR contact dose rate shall not exceed 100 mrem/hr, without sampler in SR. | e | Monitoring reduces potential personnel exposure and excessive waste accumulation in the SR which can SR which can concentrations sufficient for ignition. See Section 4.1.2.2 and Appendix R. | No FG/RMCS activities can proceed until waste accumulation levels are decreased below the specified limit. |
| 6.8.4.3. Waste contamination | Minimize waste- generated gas accumulation | Procedural evaluation | An administrative control shall be developed and implemented for decontaminating the SR if it becomes contaminated with waste levels > 100 mrem/hr. | £ | Decontaminating the SR mitigates the effects of waste- generated gas accumulation, and reduces personnel exposure. See Section 4.1.2.2. | No FG/RMCS activities can proceed without procedural compliance for SR decontamination. |
| 685RLU/Grapple | | | | ***** | | |
| 68.5.1. Travel rate | Travel rate< 1 ft/s | Sampling truck instrumentation Procedural evaluation | The travel rate for the grapple and the RLU shall not exceed 1 foot/sec. | π | Controlled travel rate mitigates potential for frictional sparking. See Section 4.3.2 and 4.3.6. | With excessive travel rates, no FG/RMCS activities can proceed until procedural corrective actions are implemented. |

August 8, 1996

| System or Condition | SCO | Surveillance/ Monitoring | Administrative Controls | Level | Basis for Control | Safe Shut Down Condition |
|------------------------------|---|--|--|-------|---|--|
| 68.5.2.Grapple Load | Grapple load ≤ 250 lb | Sampling truck instrumentation Load Cell | The electrical motor driving the grapple hoist shall automatically shutdown when the load reading is >250 lb. The grapple hoist cable shall be inspected once every 6 months. The grapple hoist load cell shall be calibrated once every 6 months. | 1 | Controlled loads prevent grapple drop consequences and potential for fire. See Sections 4.3.2 and 4.3.6. | No FG/RMCS activities can proceed until procedure is implemented. |
| 686X-ray | | | | | | |
| 68.6.1. Sampler insertion | Minimize waste accumulation in X- ray | Visual inspection Procedural evaluation | An administrative control shall be developed and implemented preventing a known leaking sampler, or sampler contaminated in excess of Contractor controls, to be placed into the mobile X-ray system. | 3 | Mitigates waste- generated gas accumulation. See Section 4.1.2.2 | No FG/RMCS activities can proceed until procedural corrective actions are implemented. |
| 68.6.2 X-ray sample liner | Minimize waste or flammable gas entering X-ray | Procedural evaluation | A redundant barrier shall protect the X-ray system from flammable gas penetrating into the system, and from sampler contamination. | 3 | Prevents flammable gas in proximity with X- ray ignition source. See Section 4.1.3. | No FG/RMCS activities can proceed until procedural corrective actions are implemented. |
| 6.8.7.Exhauster | | | | | | |
| 68.7.1. Duct inspections | Prevent duct leaks and static-induced sparking. | Procedural evaluation Visual inspection | The exhauster duct shall be visually inspected for leaks and adequately supported for environmental loads before FG/RMCS operations, and after any situation the operators believe the duct may have been damaged. | 3 | Inspecting verifies hose functional capabilities. Reduces likelihood of above ground fire that could propagate into the dome. See Section 4.1.2. | No FG/RMCS activities can proceed until duct has been inspected. |
| 68.7.2 Operations | Operate exhauster 1 hr before purge established, during rotary drilling, and 16 hours following purge termination. | Procedural evaluation | An administrative control shall be developed and implemented that mandates that the exhauster be fully operational: • 1 hour before the nitrogen purge flow to the DS is established at 30 scfm, and the flammable gas concentration < 1000 ppm | 1 | Flammable gas concentrations are mixed, dispersed and reduced with exhauster preparation. Accumulation of aerosols, particulates and flammable gases | No DS nitrogen purging or waste- intrusive activities can proceed unless the exhauster is operational. With noncompliance to procedures, no FG/RMCS |

| System or Condition | SCO | Surveillance/ Monitoring | Administrative Controls | Level | Basis for Control | Safe Shut Down Condition |
|---|---|--|---|-------|---|---|
| | | | during all rotary drilling operations cumulative 16 hours following termination of nitrogen purge flow to the DS. When rotary drilling operations are resumed within any 16 hour waiting period, a new 16 hour period will be initiated following termination of DS nitrogen purge flow of 30 scfm. The PIC shall be authorized to prematurely terminate exhauster operation for external events that threaten personnel or equipment. | | following waste intrusive activities must be minimized. This mitigates unacceptable offsite consequences resulting from a dome collapse. See Sections 4.1.1 and 5.4.2. | activities can proceed until procedural corrective actions are implemented. |
| 6.8.7.3. Riser aerosols | Mitigate aerosol, particulates and gas releases | Radiation monitoring procedures | An administrative control shall be developed and implemented that: • prevents opening a riser within 4 hour following termination of rotary drilling, and • limits the riser open period to less than 8 hours per activity. | 2 | Limited open riser time reduces the potential for aerosol, particulate and gas releases. See Section 5.4.1. | For periods greater than 8 hours, no FG/RMCS activities can proceed until procedural corrective actions are implemented. |
| 68.7.4 Exhauster- induced tank pressure | Tank pressure • <atmospheric pressure, and • ≥ -3 in. w.g.</atmospheric | Procedural evaluation | Exhauster operation shall not draw a tank pressure less than value specified in approved IOSR. | 1 | Mitigate excessive tank negative pressure, potential for dome collapse. This mitigates unacceptable offsite consequences resulting from a dome collapse. See Section 4.6.1. | No FG/RMCS activities can proceed until tank pressures are in the specified limits. |
| 68.7.5. HEPA radiological monitoring | HEPA filter housing loading < 100 mrem/hr contact dose rate on filter housing | Radiological monitoring procedures | An administrative control shall be developed and implemented for manually monitoring the HEPA filter housing loading daily during operations for radioactivity. | 3 | Control prevents accumulation of radioactive waste particulate on HEPA filters reduces the likelihood of | If the radiological levels are exceeded, then no waste- intrusive FG/RMCS activities can proceed until the |

| System or Condition | SCO | Surveillance/ | Administrative Controls | Level | Basis for Control | Safe Shut Down |
|---|---|--|---|-------|--|--|
| ***** | | Monitoring | | | | Condition |
| | 1 | | • HEPA filter housing radiological loading shall be limited to < 100 mrem/hr contact dose rate. | | unfiltered releases. See Section 4.8.8.1 and Appendix R. | HEPA filters are changed. |
| 68.7.6 Operational- induced exhauster shut down | Exhauster flow during operation • <250 scfm, and • >150 scfm | Exhauster detection system | The exhauster shall automatically shutdown and send a shutdown signal to the truck PLC: • immediately with > 5.9 inches of water differential pressure across the HEPA filter bank, or • after 5 minutes for ≥250 scfm high flow, or • immediately with ≤150 scfm low flow. An administrative control shall be developed and implemented for timely exhauster restart: following an operational shut down. | 3 | Exhauster operation in out-of- tolerance condition can impair overall exhauster performance and could result in unfiltered radioactive waste release, exposure of radioactive waste to workers. To prevent tank overpressurization due to purge gas. To prevent unacceptable vacuum in the dome To provide acceptable emission of radioactive waste particulates. | Upon detection, the exhauster will shut down within 5 min. when flow exceeds 250 scfm. If drilling, the drill rig engine will automatically trip. No FG/RMCS activities can proceed until the cause of shut down is identified and corrected. |
| 688.05 Nitrogen purg | e sysiem | | | | | |
| 6881. Bypass leakage verification | Ensure sufficient nitrogen purge flow during rotary drilling | Procedural evaluation and verification | The nitrogen purge system for the DS shall be tested for bypass leakage every 6 months, with independent verification and indication of failures. System leak rates must be within the uncertainty range of the instrumentation system, or less than 2% of the required flow. | 1 | Bypass leakage testing ensures that the purge flow is adequately provided to drill bit for cooling during drilling. This mitigates unacceptable offsite consequences resulting from a dome collapse. See Section 4.4.1 | No nitrogen purge flow activities can proceed unless testing has been completed, and flow levels increased as appropriate. |

#2dis

| System or Condition | SCO | Surveillance/ Monitoring | Administrative Controls | Leyel | Basis for Control | Safe Shut Down Condition |
|---|---|--|---|-------|--|---|
| 689.Riset/sleeve Purge | Flow | | | | | |
| 68.9.1. Bypass leakage verification | Ensure adequate purge in sleeve to DS annulus | Procedural evaluation | The nitrogen purge system for the sleeve to DS annulus shall be tested for bypass leakage every 6 months, with independent verification and indication of failures. | 3 | Bypass leakage testing ascertains nitrogen flow in annulus. See Section 4.2.4. | No nitrogen purge flow activities can proceed unless testing has been completed, and flow levels increased as appropriate. |
| 6.8.10.Hydrostatic Head | | F | | | | ********* |
| 68.10.1. Bypass leakage verification | Mitigate waste flooding | Procedural evaluation and verification | The hydrostatic head systems for the DS and the SR will be tested for bypass leakage every 6 months, with independent verification and indication of failures. System leak rates must be within the uncertainty range of the instrumentation system, or less than 2% of the required flow. | 3 | Ensure hydrostatic head capabilities before sample retrieval. Prevents inadvertent waste and flammable gas accumulation in the drill string during retrieval of core samplers. See Section 4.3.1. | No hydrostatic head activities can proceed unless testing has been completed. |
| 6.8.11.Instrumentation | and controls | | | | | |
| 68.11.1.Truck I&C | Ensure accurate l&C measurements | Procedural evaluation and verification | All truck shutdown indication system elements shall be calibrated every 6 months with independent verification and tested with indication of all failures. This control does not include the sensors in the flammable gas detection system. | 1 | Periodic calibration ensures accuracy. Control provides assurances for critical measurements, proving protections for waste, dome ,drill string fires, to be adequate during RMCS operations. | With noncompliance, no FG/RMCS activities can proceed until procedural corrective actions are implemented. |
| | | | | | This mitigates unacceptable offsite consequences resulting from a dome collapse. See Section 4.0. | |
| 6.8.11.2.Exhauster 1&C | Ensure accurate I&C measurements | Procedural evaluation and | All exhauster shutdown indication system elements shall | 3 | Periodic calibration ensures accuracy. | With noncompliance, no |
| System or Condition | SCO | Surveillance/ | Administrative Controls | Level | Basis for Control | Safe Shut Down |
|--|--|--|---|----------|---|--|
| | | Monitoring | | | | Condition |
| | / | verification | be calibrated every 6 months with independent verification and tested with indication of all failures. This control does not include the sensors in the flammable gas detection system. | | Control provides assurances for critical measurements, proving protections for waste, dome ,drill string fires, to be adequate during RMCS operations. | FG/RMCS activities can proceed until procedural corrective actions are implemented. |
| | | | | | This mitigates unacceptable offsite consequences resulting from a dome collapse. | |
| | | | | | See Section 4.0. | |
| 6812 Rotary Drilling C |)peration\$ | | | | * | |
| 68.12.1. Installation and removal of drill rod, spray washer and change-out assembly | Restrict ignition sources around open risers | Procedural evaluation | Activities and use of tools around open risers shall be strictly controlled during installation and removal of the drill rod, spray washer and change-out assembly. | 3 | Limits potential for inducing spark in flammable environment. See Section 4.3.3. | With noncompliance, no FG/RMCS activities can proceed until procedural corrective actions are implemented. |
| 6.8.12.2_Last sampler | Terminate drill 3 in. From bottom | Procedural evaluation | Based on evaluation of the depth to the tank bottom, drill string length shall be calculated to terminate drilling > 3 inches above the tank bottom. | 3 | Reduces the likelihood of the bottom liner penetration. See Section 4.6.2. Appendix N | With noncompliance, no FG/RMCS activities can proceed until procedural corrective actions are implemented. |
| 68.13.Rotary Drilling (| Operational Parameters | | | . | *** | ************************************** |
| 68.13.1. Down force | DS down force <u><</u> 750 lbf | Force detector and walkdown function used for all samples except the last The force detector and the hydraulic bottom detector used for the last sample | The drill string shall not be operated with a down force > 750 lbf. • The drill rig engine shall automatically shut down with down force > 750 lbf. • Automatic shutdown signal is assumed to occur within 2 seconds for out-of-tolerance | 1 | Drill bit overheating and waste ignition is prevented, while factoring the potential effects of DS failure. Control prevents a local exothermic chemical reactions | With noncompliance, the DS will shut down. Nitrogen purge and exhauster systems will remain operational within specified limits. No FG/RMCS |

| System or Condition | SCO | Surveillance/ | Administrative Controls | Level | Basis for Control | Safe Shut Down Condition |
|---|---------------------------------|--|--|-------|---|--|
| | / | Procedural evaluation and verification | conditions. • The force detector shall be used for all samples, automatically armed. The walkdown function used for all samples except the last, shall be enabled using independent verification. The hydraulic bottom detector used for the last sample, shall be enabled using independent verification. | | as well as an ignition of flammable gas in the waste, consequently a possible dome collapse due to frictional sparks. This mitigates unacceptable offsite consequences resulting from a dome collapse. See Sections 4.4.1, 4.4.4, and 4.6.3. | activities can proceed until procedural corrective actions are implemented. |
| 68.13.2. Down Force with DS length > 45 ft. | DS down force ≤ 650 lbf | Walkdown function used for all samples except the last The hydraulic bottom detector used for the last sample Procedural evaluation and verification | An administrative control shall be developed and implemented that manually maintains the down force < 650 lbf with DS length >45 ft. | 3 | Prevent drill string failure. See Section 4.2.1. | With noncompliance, the DS will be manually shut down. Nitrogen purge and exhauster systems will remain operational within specified limits. No FG/RMCS activities can proceed until procedural corrective actions are implemented. |
| 68.13.3. RPM | DS rotational speed ≤ 55 rpm | RPM detection system Procedural evaluation One out of two redundant logic | The drill string shall not be operated with rotational speed > 55 rpm. • The drill rig engine shall automatically shut down with DS rotational speed > 55 rpm. • Automatic shutdown signal is assumed to occur within 2 seconds for out-of-tolerance conditions. | 1 | Drill bit overheating and waste ignition is prevented, while factoring the potential effects of DS failure. Control prevents local exothermic chemical reactions as well as an ignition of flammable gas in | With noncompliance, the DS will shut down. Nitrogen purge and exhauster systems will remain operational within specified limits. No FG/RMCS activities can proceed until procedural |

WHC-SD-WM-SAL J35, Rev. 0-a

A Construction of the second s

47

| System or Condition | SCO | Surveillance/ | Administrative Controls | Level | Basis for Control | Safe Shut Down |
|---|---------------------------------|--|--|-------|---|--|
| | | Monitoring | | I, | | Condition |
| | | | | | the waste, consequently a possible dome collapse due to frictional sparks. This mitigates unacceptable offsite consequences resulting from a dome collapse. See Sections 4.4.1., 4.4.4. and 4.6.3 | corrective actions are implemented. |
| 6.8.13.4. RPM with DS length > 45 ft | DS rotational speed ≤ 40 rpm | RPM detection system Procedural evaluation | An administrative control shall be developed and implemented that manually maintains the rotational speed < 40 rpm with DS length >45 ft. | 3 | Prevent drill string failure. See Section 4.2.1. | With noncompliance, the DS will be manually shut down. Nitrogen purge and exhauster systems will remain operational within specified limits. No FG/RMCS activities can proceed until procedural corrective actions are implemented. |
| 6.8.13.5. Nitrogen purge flow | DS nitrogen purge ≥ 30 scfm | Purge flow detection system Procedural evaluation | The DS shall be purged with nitrogen at ≥ 30 scfm. The drill rig engine shall automatically shut down with nitrogen flow less than 30 scfm, using two out of three redundant logic. Automatic shutdown signal is assumed to occur within 2 seconds for out-of-tolerance conditions. Nitrogen purge gas temperature entering the tank shall be > 10°F and < 140°F. | 1 | Drill bit overheating and waste ignition is prevented. Control prevents a local exothermic chemical reactions as well as an ignition of flammable gas in the waste, consequently a possible dome collapse due to frictional sparks. This mitigates | With noncompliance, the DS will shut down. Nitrogen purge and exhauster systems will remain operational within specified limits. No FG/RMCS activities can proceed until procedural corrective actions are implemented. |

| System or Condition | SCO | Surveillance/ Monitoring | Administrative Controls | Level | Basis for Control | Safe Shut Down Condition |
|--|--|---|--|-------|--|---|
| | / | | | 1 | unacceptable offsite consequences resulting from a dome collapse. See Sections 4.4.1 and 4.4.4. | |
| 6.8.13.6. Riser/sleeve purge operations | Sleeve to DS annulus purge ≥ 2 scfm | Designated measurement system Procedural evaluation and verification | The sleeve to DS annulus shall be purged with nitrogen at ≥ 2 scfm during rotary drilling operations. Nitrogen flow in the sleeve to DS annulus shall be verified before rotary drilling operations, using independent verification. Nitrogen purge ≤ 40 psid shall be alarmed and will automatically shut down the drill rig with one out of two redundant logic. | 3 | Prevents accumulation of flammable gas in the riser and consequently a dome fire caused by the frictional spark between the rotating drill string and sleeve. See Section 4.2.4. | No FG/RMCS activities can proceed until nitrogen purge flows are adequately established. |
| 68.13.7. Slow penetration rate | Prevent extended waste heatup and ignition | Penetration detection system Procedural evaluation 1 out of 1 control logic with operator backup. | The DS shall not be operated with a penetration rate < 0.75 in./min for \geq 60 seconds. • The DS shall automatically trip when the penetration rate is <0.75 in./min. for a cumulative time of 60 seconds in any 3 min. period. The drill bit shall be replaced if drilling is shutdown four times consecutively as a result of low penetration rate, and if the cumulative penetration is <0.3 in. for the last three attempts. | 1 | Waste burn induced by slow penetration is negated. Control prevents a local exothermic chemical reactions as well as an ignition of flammable gas in the waste, consequently a possible dome collapse due to frictional sparks. Control also prevents drilling against materials not covered by envelope testing. Prevents the ignition of flammable gas due to hot spots | With noncompliance, no FG/RMCS activities can proceed until procedural corrective actions are implemented. |

NUMBER OF

| System or Condition | SCO | Surveillance/ Monitoring | Administrative Controls | Level | Basis for Control | Safe Shut Down Condition |
|---------------------------------|----------------------------------|-----------------------------|---|-------|--|--|
| | | | | | (exceeding autoignition temperature). | |
| | | | | | Provides protection against severe drill bit tooth wear and protects the Carbon Steel Blank from exposure to metals. | |
| | | | | | This mitigates unacceptable offsite consequences resulting from a dome collapse. | |
| 4014Ch.4 J. | | L | | ····· | See Sections 4.4.1 and 4.4.4. | |
| aa 14.5 nut down and H | Cestart Conditions | | | | | |
| 68.14.1. Drill Rig shut down | Prevent drill bit overheating | Alarmed setpoints | The drill rig shall automatically shut down with out-of-tolerance values for down force, RI ^A M, nitrogen purge or penetration rate, operational exhauster shutdown, or upon detection of a GRE. A minimum 10-minute wait period shall be imposed following automatic drill rig shut down before any waste-intrusive activities can be resumed. | 1 | Wait period allows cool down of drill bit and localized waste temperatures due to out-of- tolerance down force, RPM or purge flow values. Control ensures that the initial local waste temperature is in thermal equilibrium with the tank waste and provides assurances for drill bit/waste temperature not to be exceed above allowable limits. | No FG/RMCS activities can proceed until wait period is satisfied. |
| | | | | | This mitigates unacceptable offsite consequences resulting from a dome collapse. | |

00 0 m.

| System or Condition | SCO | Surveillance/ Monitoring | Administrative Controls | Level | Basis for Control | Safe Shut Down Condition |
|---------------------------------|---|--|---|-------|--|---|
| | / | ¥************************************* | | ····· | See Section 4.4.1. | *************************************** |
| 6.8.14.2. GRE shut down | Protect personnel | Alarmed setpoints | A GRE is defined to occur when: • > 5000 ppm hydrogen concentration equivalent, or • > 100 ppm/s rate of equivalent hydrogen concentration increase over a 10-s period, or • > 2 in. w.g. increase in a 5 minute period above the tank background pressure. Upon detection of a GRE, the exhauster will continue to operate, and drill rig operation will be terminated | 1 | Automatic shut down during GRE prevents exposure to flammable gas concentrations sufficient for ignition. Reduces the possible spark sources in the dome, waste and around drill string. Provides protection for toxic gas exposures. This mitigates unacceptable offsite consequences resulting from a dome collapse. | All personnel within a 100-m radius from the edge of the tank will evacuate and don protective equipment before returning for further equipment stabilization. |
| 6.8.14.3. GRE restart | Minimize unanticipated ignition sources following GRE. | Procedural evaluation and verification | An administrative control shall be developed and implemented for restart following GRE. | 3 | Flammable gas removal following a GRE is enhanced by exhauster operation and the use of pressure measurement systems qualified to operate in a Class I, Division 1, Group B environment. Appendix B. | FG/RMCS activities cannot resume until restart requirements are implemented and flammable gas concentration limits are met. |
| 6.8.14.4. Power loss restart | Prevent unanticipated ignition sources following power loss. | Procedural evaluation | An administrative control shall be developed and implemented for restart following power loss to include: • The DS shall be sniffed and purged following shut down caused by power loss, | 3 | Loss of hydrostatic flow from a power loss can allow waste flooding and gas release into DS and SR. See Section 4.1.4. | FG/RMCS activities cannot proceed until restart requirements are implemented. |

WHC-SD-WM-SA_ J35, Rev. 0-a

| System or Condition | SCO | Surveillance/ Monitoring | Administrative Controls | Level | Basis for Control | Safe Shut Down Condition |
|-------------------------|---|-----------------------------|--|-------|--|---|
| | | | sufficiently to evacuate flammable gas accumulations. | | | |
| | 1 | | Hydrostatic head shall be reestablished. | | | |
| | | | The exhauster shall be restarted following a power loss event. | | | |
| 6.8.15.Water addition | | | | | | |
| 68.15.1. Quantity | Minimize water addition | Procedural evaluation | Per activity, water addition shall be limited to < 250 gallons in compliance with approved Tank Farm Operating Procedures. Increased water addition quantities shall require PRC approval. | 3 | Contractor control See Section 4.5.2 and Appendix O. | No FG/RMCS activities can proceed until procedural corrective actions are implemented. |
| 68.15.2. Temperature | Provent unanticipated temperature-induced, waste-related chemical reactions | Temperature measurement | The temperature of added water shall be limited to < 200°F in compliance with approved Tank Farm Operating Procedures. | 3 | Contractor control See Section 4.5.2 and Appendix O. | No FG/RMCS activities can proceed until water temperature is within specified limits. |

| 0-a |
|-------|
| Rev. |
| -035, |
| SAD |
| -WN |
| -CD-I |
| VHC- |
| S |

| System or Condition | 800 | Surveillance/ Monitoring | Administrative Controls | Level | Basis for Control | Safe Shut Down Condition |
|-------------------------------------|--|--|---|-------|--|---|
| 69.REMOVAL PHA | SE | × | | | | |
| 69.1.DS Removal | | | | | | |
| 69.1.1. DS removal | Reduce gas production and ALARA personnel exposures | Procedural evaluation | DS shall be removed to above the waste surface level only with the last sampler, a dummy sampler or water in place, in compliance with approved Tank Farm Plant Operating Procedures | C | Minimize gas release potential. See Section 4.1.2.1. | Removal and decontamination activities cannot proceed until procedural control is implemented. |
| 69.2 Decontamination | | | | | | |
| 69.21. Decontamination system | Complete DS decontamination | Procedural evaluation | Decontamination systems shall be verified operational before DS removal in compliance with approved Contractor procedures. | e | Verification ensure decontamination capabilities. See Section 5.4.2. | No decontamination activities can proceed until system operability is verified. |
| 69.22 Riser exposure | Minimize radioactive releases | Radiation monitoring procedures | Personnel direct exposure from open riser shall be limited to values specified by approved Tank Farm Operating Procedures. Riser open period shall not exceed 8 cumulative hours per activity. | m | Minimize personnel direct exposure from open riser conditions. See Section 4.8.8.4, 4.8.8.5 and 5.41. | No FG/RMCS activities can proceed until personnel protection is appropriately ensured. |
| 6923. Drill rod removal | Minimize radioactive releases | Radiological monitoring procedures Procedural evaluation verification | Drill rods shall be monitored and controlled as they are removed commensurate with approved Tank Farm Plant Operating Procedures. | ñ | Limit unfiltered, uncontained releases to environment, as required by Contractor controls. See Section 4.8.5, 4.8.6, 4.8.7 and 5.4.2. | With noncompliance, no decontamination activities can proceed until procedural corrective measures are implemented. |

August 8, 1996

ϵ **REFERENCES**

- 1. "Tank Farm Health and Safety Plan," Westinghouse Hanford Company report WHC-SD-WM-HSP-002, Rev. 0A (September 28, 1993).
- 2. "Safety Assessment for Push- and Rotary-Mode Core Sampling in Ferrocyanide Tanks," Westinghouse Hanford Company report WHC-SD-WM-SAD-013, Rev 3-A (November, 1993).
- 3. "Safety Analysis for the Push Mode and Rotary Mode Core Sampling," Westinghouse Hanford Company report WHC-SD-WM-SARR-031, Rev. 2 (July, 1995).
- 4. J. E. Corbett, "Rotary-mode Core Sampling Safety Equipment List, WHC-SD-WM-SEL-032, Rev. 0 (February 16, 1994).
- 5. J. E. Corbett, "Rotary-mode Core Sampling Safety Equipment List, WHC-SD-WM-SEL-032, Rev. 0 (February 16, 1994).
- R. L. Koontz, "Hazards Identification and Evaluation for Waste Tank Core-Sampling Equipment," Westinghouse Hanford Company Report, SD-WM-SAR-007, (June 1985).
- 7. "Rotary Mode Core Sample Truck Material and Component Compatibility with Tank Waste," Westinghouse Hanford Company Internal Memo 75230-96-001 R2 (February 5, 1996).
- 8. L. H. Sullivan et al., "A Safety Assessment for Proposed Pump Mixing Operations to Mitigate Episodic Gas Releases in Tank 241-101-SY: Hanford Site, Richland, Washington," Los Alamos National Laboratory report LA-UR-92-3196 (October, 1992).

6-42

7.0. CHECKLIST ITEMS

This safety assessment (SA) assessed safety aspects of the installation, operation, and removal of rotary-mode core sampling (RMCS) equipment in single-shell tanks (SSTs) on the Flammable Gas Watch List (FGWL) or those tanks recommended by the contractor to be included on the FGWL, hence referred to as FG/RMCS operations.

This SA was developed using bounding assumptions. However, these assumptions are not verified against each one of the existing or potential flammable gas tanks (see Section 1 of this SA). The following assumptions must be verified before this SA can be used for RMCS operations in any FGT.

CHECKLIST ITEMS

TANK SPECIFIC HAZARDS/OTHER WATCHLISTS

If a given tank has a specific hazard or accident initiator that is not analyzed in Sections 3 and 4 of this SA, the analysis must be supplemented to cover the tank specific conditions. For instance, this SA does not address ferrocyniade issues even though some of the FGTs may also be on the ferrocyianite watchlist. This checklist item is especially important for tanks that are on multiple watchlists (in addition to flammable gas watchlist).

FLAMMABLE GAS COMPOSITION

This SA assumes that 25% of the LFL is greater than 5000 ppm hydrogen based on the analysis provided in Appendix C. The only flammable gas species considered are hydrogen and ammonia with small amounts of methane. If new information (information that is not cited in Appendix C) reveals that, for a given tank, there are other flammable gas species and/or the assumed value of the LFL is not conservative, the analysis in Appendix C must be revised to incorporate the new data.

CHECKLIST ITEMS (cont)

TOXIC GAS COMPOSITION

For toxic effects, the gas composition in a given GRE is assumed to be 60% ammonia or 75% nitrous oxide. If any evidence before the FG/RMCS operation exists to indicate that these values (especially the ammonia fraction) may be exceeded in one of the SSTs as a result of new analysis or data, or if they are not conservative, the consequence analysis must be re-evaluated. This can be achieved by simply scaling the new data with the 60% fraction used in the consequence analysis. The consequence analysis is the linear function of gas composition; thus, simple multiplication would be used to consider different gas concentrations.

Also, the results of vapor space sampling program were reviewed. Major toxic gases that are found in the dome space of the presently defined flammable-gas tanks are ammonia and nitrous oxide. Other gases are found in trace quantities and do not pose a concern. However, it was recognized that the existing data are limited and all tanks of interest are not covered. Thus, if new data reveal that toxic gases in excess of the hazardous limits are detected in a given tank, the consequence analysis must be reevaluated. The reevaluation may be done by simply scaling the toxic gas fraction and the guidelines against the ammonia fraction and the associated RGs.

WASTE TEMPERATURE

The best available tank temperature data must show that the peak waste temperature (considering uncertainties) must be less than 90°C. If the peak waste temperature is \geq 90° C, the envelope testing results discussed in Appendix F must be re-evaluated.

WASTE ENERGETICS

Table G-3 of Appendix G lists the frequency of waste fire accidents for the 100-series SSTs. The analysis given in Appendix G is based on waste composition data through December 1995. New data taken after December 1995 or revisions to the old database may alter the results of the analysis given in Appendix G. Therefore, in using new or revised data, it first must be verified that the conclusions of the analysis in Appendix G are not changed.

CHECKLIST ITEMS (cont)

LIKELIHOOD OF GAS RELEASE EVENTS

The GRE probability includes statistical distributions for gas-release amounts and rates that are based on limited data and expert judgment. If additional data or analyses exist for a specific tank to indicate that the GRE probabilities used in this SA are not conservative, the accident frequencies need to be re-evaluated for that tank.

In general, before the FG/RMCS operation starts on a given tank, the best available tank specific data for gas inventory and gas release evidence must be evaluated to confirm that the statistical model for gas-release amounts and rates used in the SA are still conservative. In general, if one or more of the following conditions are observed for a given tank, the GRE probability model given in Appendix L must be re-evaluated.

- Periodic level drops and level swells in excess of ± 3 in.
- Level drop \geq 3 in. during or after an intrusive event
- Dome concentration measurements ≥ 25% of the LFL before, during or after a waste intrusive event.
- A well defined nonconvective layer (parabolic temperature profile) below a supernate or convective layer (flat temperature profile) that would be indicative of potential rollovers.
- Retained gas inventory estimates (via level swell, fill history, etc) is greater than 20% of the available dome space volume.

If the reevaluation indicates that the existing GRE model given in Appendix L is not conservative for a given tank, a revision to the SA will be necessary.

• . **.**

APPENDIX A

HAZARD IDENTIFICATION

This appendix describes the process used to identify and qualitatively assess the potential operational events, external events, and natural phenomena that can cause the release of hazardous material during the installation, operation, and removal of rotary-mode core sampling (RMCS) equipment in single-shell tanks (SSTs) on the Flammable Gas Watch List (FGWL) or those tanks recommended by the contractor to be included on the FGWL, hence referred to as FG/RMCS operations. This process, called hazard identification, is equivalent to the hazard analysis (HA) or hazard evaluation process in a safety analysis (SA) report. The final product of this hazard identification process is a list of Design Basis Accidents that will be examined in more detail in the Accident Analysis section of the SA.

The guidance used to perform the hazard identification (or HA) is DOE-STD-3009).¹ DOE-STD-3009 provides the following guidance on the requirements for a complete hazards analysis.

A complete hazards analysis should include the following requirements (STD-3009, pg. 31);

- 1. Consider the complete spectrum of accidents that may occur as a result of facility operations;
- 2. Analyze potential consequences to the public and worker;
- 3. Estimate the likelihood of occurrence;
- 4. Identify and assess associated preventive and mitigative features;
- 5. Identify safety-significant structures, systems, and components; and
- 6. Identify a select subset of accidents to be formally defined in accident analysis.

In the hazard identification (hazard analysis) performed for FG/RMCS drilling, four of the six requirements above were met in full, namely, (1) a complete consideration of the spectrum of accidents; (2) analysis of the potential consequences; (3) estimation of likelihood; and (6) identification of a select subset of accidents for accident analysis (the end product for this hazard identification).

However, two of the requirements; (4) the identification of preventive and mitigative features; and (5) identification of safety-significant structures, systems and components, were only partially met. For the FG/RMCS activity, hardware design and procedures were being developed during the hazard identification (hazard analysis) process. Therefore, at that time, preventive and mitigative procedures

were not fully identified. By not identifying the preventive and mitigative features, the identification of safety-significant structures, systems, and components could not be performed. However, identification of preventive and mitigative systems is performed in the design change/control implementation section of the SA (Section 6).

A safety equipment list will be developed prior to the beginning of FG/RMCS operations.

The hazard identification performed for rotary-core mode drilling is presented in Sections A.1 through A.4. Section A.1 describes the role of hazard identification in the SA process and the methods chosen to perform hazard identification. Section A.2 briefly describes the FG/RMCS activity and presents the organization of the process for analysis. Section A.3 presents the results of the hazard identification and the list of design basis accidents (DBAs). Finally, Section A.4 presents accident progression results that were derived as an extension of the hazard identification process.

A.1. ROLE OF HAZARD IDENTIFICATION AND METHODS

The hazard identification process is one step in the overall safety analysis for rotarycore drilling. The role that hazard identification plays in safety analysis is discussed in Section A.1.1. The methods chosen to perform hazard identification are briefly described in Section A.1.2.

A.1.1. Role Of Hazard Identification In Safety Analysis Process

Hazard identification is the first step in the SA process. The goals of hazard identification for this SA are a subset of the HA requirements presented earlier, namely:

- Consideration of the complete spectrum of potential accidents;
- Qualitatively assess the consequences to the worker and the public;
- Qualitatively estimate the frequency (or likelihood) of occurrence;
- Identification of a select set of representative and unique accidents (DBAs) for further evaluation.

The main focus of the hazard identification process is on the completeness of the spectrum of accidents, and assigning a qualitative measure of consequence and frequency. An attempt is made to be exhaustive in identifying all possible initiators that can lead to adverse consequences in the process. By attempting to assess all initiators, a basis for examining some select accidents in more detail can be made

(i.e., justification is provided for why some accidents are looked at and others are not).

The representative and unique accidents (or DBAs) identified in the hazard identification process will be examined in more detail in the Accident Analysis portion of the SA. A part of this analysis is a more in-depth examination of the preventive and mitigative features. The final product of the SA will be a validation of the controls/design used in rotary-core drilling (presented in Chapter 6).

The hazard identification process is not an attempt to define what is acceptable risk or provide a quantified assessment of consequence or frequency. For this reason, the measure of risk used in the hazard identification process is a "relative" measure of risk used only for accident selection.

A.1.2. The Hazard-Identification Methodology

The FG/RMCS hazard identification process was performed by a multi-disciplinary team consisting of Los Alamos and Westinghouse personnel, using a combination of two standard techniques, the hazards and operability (HAZOP) technique and "what if"/checklist techniques. We will not elaborate on details of these techniques; they are described in detail in <u>Hazard Evaluation Procedures</u>, published by the American Institute of Chemical Engineers (AIChE).² These techniques produce tabulated data that identify the following for each accident.

- Hazard "A source of danger (i.e., material, energy source, or operation) with the potential to cause illness, injury, or death to personnel or damage to an operation or to the environment." (Ref. 1) A hazard is not the accident initiator, cause or deviation. Rather, it is a property, typically radiological or toxicological, inherent to the danger.
- Cause/Deviation-A description of the initiating event and accident progression.
- Preventive/Mitigative Features-A description of the controls or components that could either prevent the accident or mitigate its consequences.
- Frequency Estimate-Accident frequency assignment to a bin (frequency range). Other than for seismic accidents, each frequency assignment is for a single tank assuming 2 drilling activities per year for a complete core sample.
- Consequence Estimate-An assignment to an accident consequence bin.
- Measure of relative risk-A risk index used for accident selection.

• Recommendations-Generic information that indicates what further analysis is needed.

Stand occupational hazards covered by Occupational Safety and Health Administration (OSHA) regulations are not included in this HA.

The MCS system is an operation utilizing mobile equipment. The guidelines and cristic defined in DOE Std. 3009 were used in this hazards assessment, but are considered conservative because FG/RMCS operations are not considered a "facility" as defined in the standard, and do not have the operational lifetime of a facility.

Following accident frequency and consequence assignment, each value is assigned to bins as defined in Tables A-1 and A-2, respectively.

| Classification | Relative Likelihood | Description |
|---|---------------------------------|--|
| I (> 10 ⁻² /yr) | Anticipated | Incidents that may occur several times during the lifetime of the facility, e.g., incidents that commonly occur. |
| II (10 ⁻² to 10 ⁻⁴ /yr) | Unlikely | Accidents that are not anticipated to occur during the lifetime of the facility, frequency between once in 100 years and once in 10,000 operating years. |
| III (10 ⁻⁴ to 10 ⁻⁶ /yr) | Extremely Unlikely | Accidents that will probably not occur during the life cycle of the facility. Frequency between once in 10,000 years and once in a million years. |
| IV (< 10 ⁶ /yr) | Beyond Extremely Unlikely | All other accidents. Frequency of less than once in a million years. |

TABLE A-1 LIKELIHOOD CLASSIFICATION TABLE

| TABLE A-2 | |
|----------------------|-------|
| CONSEQUENCE SEVERITY | TABLE |

| | Public | On-Site Worker | Facility Worker | Environment |
|---------------------|--|--|---|---------------------------------------|
| High (A) | Off-site EGs ^a exceeded • > 25 rem off-site • > ERPG-2/h | > 100 rem > ERPG^b- 3/h | Prompt fatality • 250 rem • > ERPG- 3/h | Significant off-site contamination |
| Mod (B) | Challenge to Off- site EGs • 1- 25 rem • > ERPG- 1/h | • 10-100 rem • > ERPG- 2/1h | Serious injury • 100-250 rem • >ERPG- 2/h | Significant on-site contamination |
| Low (C) | Minor challenge to off-site EGs • 0.1- 1.0 rem • < ERPG-1/h | 1-10 rem > ERPG-1/h | Minor injury • 1-100 rem • > ERPG- 1/h | Significant facility contamination |
| No Impact (D) | Less than low | Less than low | Less than low | Less than low |

NOTE: ERPG values are defined over a one hour period

^a EG ⁼ Evaluation Guideline

^b ERPG = Emergency Response Planning Guidelines

After frequency and consequence bins are assigned, a relative measure of risk is determined for each accident scenario. This measure of risk is based on the risk ranking table presented in TABLE A-3. This table, used for accident selection only, is not used to identify high or unacceptable levels of risk.

TABLE A-3 RISK-RANKING MATRIX

| | Likelihood | | | | | | | | | | |
|-------------|------------|----|-----|----|--|--|--|--|--|--|--|
| Consequence | I | II | III | IV | | | | | | | |
| Α | 1 | 1 | 2 | 3 | | | | | | | |
| В | 1 | 2 | 3 | 4 | | | | | | | |
| С | 2 | 3 | 4 | 4 | | | | | | | |
| D | 3 | 4 | 4 | 4 | | | | | | | |

| Rank | Accident Disposition |
|------|----------------------------------|
| 1 | Major candidate for selection |
| 2 | Serious candidate for selection |
| 3 | Marginal candidate for selection |
| 4 | Not a candidate for selection |

.....

WHC-SD-WM-SAD-035, Rev. 0-a

A.2. PROCESS FLOW SHEETS

The GF/RMCS is described in detail in Section 2 of the main report. During the hazard identification process the following activities were examined

- Startup of operations,
- Rotary-core drilling mode,
- Bump-drilling mode,
- Change of mode between push and rotary mode, and
- Shutdown of operations.

A simplified flow sheet showing an understanding of the tasks involved in the FG/RMCS activities at the time the hazards identification was initiated is presented in Fig. A-1. Subsequent changes in the FG/RMCS process were evaluated and included in the accident analysis, but are not reflected in Fig A-1. For example, evaluation of hazards determined that the consequences of bump drilling mode were unacceptable. Therefore, bump drilling mode was eliminated from the operation. All changes were evaluated in the accident analyses, consequently, it was unnecessary to redo the hazards identification.

Each activity was broken into numbered study nodes that represent the smallest individual task examined in the hazard identification process. In the results section, accidents can be traced to the study node from which they were initiated by means of the number scheme assigned to each study node.



August 8, 1996



A-8



A-9

August 8, 1996



an gaing



activity 37

activity 37

A-11

Figure A-1 (Page 4)

 $\begin{array}{c} {}^{0}{}^{0}{}^{0}{}^{1}\left({1,00,u} \right) \\ {}^{0}{}^{0}{}^{1}{}^{1}{}^{1}\left({1,01,u} \right) \\ {}^{0}{}^{1}{$

A.3. RESULTS

The hazard identification process produced a database of approximately 180 accident scenarios. These accidents were sorted to obtain design-basis accidents that are examined in more detail in the Accident Analysis section of the SA (see Section 4.0 of the SA). A summary of the DBAs is presented in Section A.3.1. The complete accident database is presented in Section A.3.2.

A.3.1. Summary of Results (Identification of DBAs)

The types of accidents selected for additional analysis can be grouped into two broad categories: representative accidents and unique accidents. Representative accidents are those of a similar type, often having the same initiators or similar accident progression. Seismic events, explosions, or tank dome fires are all examples of representative accidents. Unique accidents are typically low-frequency, highconsequence accidents or those not defined as representative accidents. An example of a unique accident is an explosion resulting from the mixing of incompatible materials.

The process of selecting FG/RMCS DBAs involved sorting the database of accidents by accident keyword (this was done to examine groups of similar accidents by consequence, frequency, and relative measure of risk). These sorted results were reviewed, and engineering judgment was used to select representative accidents and to identify any unique accidents.

For the FG/RMCS activity a list of ten representative accidents were identified. No unique accidents were identified that were not already covered in the list of representative accidents. The list of DBAs is presented in TABLE A-4.

A.3.2. Accident Database

The accident database was produced during team meetings held at the Hanford Site. The team consisted of safety personnel, FG/RMCS design engineers, and drilling operators. The team approach and methodology are presented in more detail in *Hazard Evaluation Procedures*.² A condensed version of the accident database is presented in TABLE A-5. Columns only include data relevant to the accident analysis.

TABLE A-4 DESIGN-BASIS ACCIDENTS

| # | Name | Description |
|------------|--|---|
| Ι. | Aboveground Fire | Aboveground fires include all fires initiated by a flammable gas mixture above the tank, ignited by either (a) vehicle impact, (b) undetected ignition source (e.g., electrical, static), (c) truck fire, or (d) random spark. |
| 11. | Dome-space fire | Dome fires include all fires in which the dome gas space is ignited directly by either (a) vehicle impact, (b) drill-string breakage exposing electrical equipment, or (c) mechanical spark generation from objects (e.g., drill string) impacting metal. |
| III. | Drill-string fire | Drill-string fires include all fires started inside the drill string. This accident typically involves either a human error or mechanical failure that allows waste inside the drill string, creating flammable-gas concentrations above the lower flammability limit (LFL). The fire is then started by an electrical equipment or a mechanically generated spark. |
| IV. | Waste fire | Waste fires include all fires that start in the portions of the tank waste that are combustible, including fires caused by a tank-dome collapse. These fires are primarily caused by drill bit over-temperature. The most likely initiators include excessive drill string rpm, wrong drill bit, loss of nitrogen purge, excessive downward drilling force, and intentional or unanticipated drilling outside the bit parameter envelope. |
| V . | Gas release | This accident type includes all gas releases, either filtered or unfiltered, in which the containment is breached or gas is vented through the exhauster. These gas releases are not accompanied by fire. |
| VI. | Chemical reactions in the waste | This accident category includes chemical reactions caused by incompatible material additions, energy addition/subtraction or water additions to the tank waste. |
| VII. | Containment breach | This group of accidents includes all mechanisms (other than fire) that result in containment breaches such as dome collapse or breach, tank-bottom penetration, or riser penetration. This group includes (a) seismic events, (b) static overload of the dome, and (c) impact to the dome (e.g., dropping a heavy object). |
| VIII | Spills, releases and radiation exposure | Spills aboveground are caused mainly by human error and have low impact to the public and on-site workers. These spills are an as low as reasonably achievable (ALARA) concern for the workers in the vicinity. Spills as a result of ejection or detonation are also considered in this section. High-efficiency particulate air (HEPA) filter failure releases and aerosol releases are included bounded and by this accident. |
| IX. | External event | Accidents due to range fire, seismic, flooding, tornadoes and high winds, lightening, and volcanoes are included in this category. |

| ł | #61%-3 | , | r | 4 | 7 | • | - | n | - | 4 | 19 | 4 |
|---|--|--|--|--|---|--|--|--|---|--|---|---|
| ļ | XHIN-W | • | 1 47 | • | - | 2 | en | n | ien | 4 | ~ | • |
| ļ | MIR-BRANO | 4 | • | - | - | 2 | 4 | n | | 4 | 2 | - |
| ļ | Methodole | * | 4 | * | 2 | en | * | • | | | • | 4 |
| ł | Finduncy | | ÷ | | | E | = | 2 | - | 2 | ± | |
| ļ | Them non I voil and | 0 | 8 | | 8 | 0 | 0 | < | | - 0 | 100 | 0 |
| ł | Worker | 0 | | | < | < | 0 | < | | | < | |
| ł | SUPLC) | | | <u> </u> | < | < | 0 | < | | | < | |
| ł | Public | <u>م</u> | | | 8 | | | < | 8 | 4 | 0 | |
| | Recommendations | A study on the list of incompatible instantist should be assembled (a g. a list location a stuminum. A minimum amount of incompatible material needed for a significant reaction. | 1) A study on the first of incompatible instantists should be assemilated (e.g. a list including aluminany 2, A minimum amount of incompatible meterial needed for a significant reaction. | Walk down tank farm and de energite equipment as necessary. 2) Verity condition of new seals. Document farm equipment and provide inter map | review dome konding analyses | review dome loading analyses | Ispiew dome boeding analyses | review dome toading enaityses | edd administrative control on vehicle spee | add administrative control on vehicle spas | edd edinihistethe control on vehicle spee | Terler dame toding mahras |
| | Consequences | Representative concequence is an estantiad low concentration taxic pas (no solid) personnel seposure. See Taas release event' event taxa for complete spectrum of consequences | Representative consequence is a weste- fite-induced filter blowcut release. See "veste fite" event tree for complete spectrum of consequences | Réprésentative consequence la an extendeut low concentration taxic ges (no estes) personnel exposure. Ses "ges release ever, tres lor complete spectrum of convergences | Representative consequence is a dome collepes. See "but leading" event tree for complete spectrum of consequences | Representative connequence is a dome collepee. See Tank hosting" event tree for complete spectrum of connequence | Representative consequence is an edented low concentration taxic gas (no addit) personal exposure. See Tamit beding" event tree lor complete spectrum of consequences. | Assumed consequence is a dome collapse followed by a want fin. See "Bink loading" event has fin complets spectrum of consequences | Representative consequence is an attitude the concentration take ge (no edit) personnel appaure. See "gas release event" event tre to complete spectrum of consequences | Representative consequences is a HEPA filler holdup release. See "shore ground fille" event tree for complete spectrum of consequence | Assumed consequence is a fine-induced dense colleges. See Yome spaces fire seekt tea for complete spectrum of consequence | Representative consequence is an estanded low concentration tonic gas (no wild) personnel exposure. See "tank tiding" event has for complete spectrum of consequences |
| | Proventive Antitigentive Features | Design revew process | Design review process | 1) High winds in area are prominant. 2) WHC-SD-WM- HSP-002 controls gas detection | Admin. control OSD-1-51-0013 adis limitis on tank loading | Admin. control gives procedures for critical crave lifts | Admin. control gives procedures for critical crans Mis | Admin. control gives procedures for critical crane Wa | | | | jack collars hold jack positions |
| | Accident Description (Cause or Deviation) | Internitional or untritentional introduction of lecompatible meteralist this tent because of human error or inadequate design control. | Internitional or universitional introduction of incompatible metantida into the because of human error or hadequate design control. | Toxic gaser released due to random bound gas release with non-delaction of built gas above lank | Dome static load enceaded, dome oollepee With aerosol release of veste | Load dropped on top of bink during ML dame collapse fotowe. | Load dropped on by of lank during M. Bink containment breach | Load dropped on top of lank during MR, lank dome collepse followed by weets fre | Vehicle Impacts Riser of Oter part of Tent containment, containment breached | Vehicle Impacts Ruar of Other part of Tank containment, sparit ignites above ground file | Vehicle impacts and invectore Riser or Vehicle per of tenk containment, vehicle fuel tenk treacting, epect lipskes dome space fre. | 1 al 3 dill buck jacks fait is support buck, dome containment breached |
| | Hazand | Chemical reaction with tank weeks | Chemical rection with tank water with run-every excitentic reaction | Tanic Gas above tank | Tenk Wash | Farnk Waste | Tank Waste | Tark Weste | Tank Wade | Tank Waste | Tank Waste | Tark Wash |
| | Presidery | Ges Release | Worde Fire | Gas Relate | ł | College | Gas Relation | | Cas Read | Abone Ground Fire | Dome File | Gas Release |
| | Scenarie D | Global 3 | Globel 4 | Globel 5 | - | 51 | 2 | - | si. | • | 2 | |

 TABLE A-5. Rotary Mode Core Drilling HA Accident Database

Table A.5. Robery Core Mode Drilling HA Accident Detabase

A-14

August 8, 1996

Table A-S. Rolary Core Mode Drilling HA Accident Database

| | | | A | Bennen unblum mitter unberen Bennen von | A | Recommendations. | T-1 | | | | — | · T | - T | 7-7 |
|----------------|--------------------------------|------------|--|---|--|--|--------|--------|--------|-------------|-----------|------------|-----|-------|
| Scenario 10 | Keyword | Hazard | Accident Description (Ceuse or Devision) | Plevendveningstive Festures | Consequences | Recommendations | Public | Omarte | Worker | Environment | Frequency | Public Alt | | ERisk |
| 1.9 | Gas Release | Tenk Waste | 1 of 3 drill truck jacks mis-positioned due to human error, truck falls and dome containment breached | | Representative consequence is an extended low concentration taxic gas (no solid) personnel exposure. See "tank loading" event tree for complete spectrum of consequences | review dome loading analyses | D | D | С | D | Ħ | 4 | 1 | |
| 1.11 | Gas Release | Tenk Weste | 1 of 3 drill truck jacks struck by vehicle, truck falls and dome conteinment breached | | Representative consequence is an extended low concentration toxic gas (no solid) personnel exposure. See "tank loading" event tree for complete spectrum of consequences | review domé loading analyses | D | D | C | D | Ħ | 4 | 4 | Ĩ |
| 2.1 | Gas Release | Tenk Waste | Drill truck impacts Riser or Other part of Tank containment, containment breached | | NA - Included in scenario 1.5 | add administrative control on vehicle speed | D | D | c | D | t | 4 | 4 | 11 |
| 2.3 | Dome Fire | Tank Waste | Vehicle Impacts and breaches Riser or Other part of Tank containment, vehicle fuel tank breached, spark ignites dome space fire | | NA - Included in scenario 1.7 | add administrative control on vehicle speed | 8 | • | ^ | 8 | झ | 3 | 2 | 3 |
| 29 | Above Ground Fire | Tank Waste | Vehicle Impects Riser or Other part of Tank containment, vehicle fuel bank branched, spark ignities above ground fire | | Representative consequences is a HEPA filter holdup release. See "above ground fire" event tree for complete spectrum of consequence | add administrative control on vehicle speed | D | C | c | c | 111 | 4 | 4 | Ĩ |
| 24 | Dome Collapse | Tank Waste | Core drilling truck driven off remps into open pit, dome collepse follows | ramps have edges and stop at ramp and | Representative consequence is a dome collepse. See "tank loading" event tree for complete spectrum of consequences | review dome loading ensityses | 8 | ^ | ۸ | Ø | Ħ | 2 | 1 | 2 |
| 2.5 | Gas Release | Tenk Weste | Core drilling truck driven off ramps into open pR, dome containment breached | ramps have edges and stop at ramp and | Representative consequence is an extended low concentration toxic gas (no solid) personnel axposure. See "tank loading" event tree for complete spectrum of consequences | review dome loading analyses | D | 8 | 8 | 8 | ŧ | 4 | 2 | 2 |
| 20 | Above Ground Fire | Tank Waste | Core drilling truck driven off remps into open pit followed by truck fuel fire | ramps have edges and stop at ramp and | Representative consequences is a HEPA filter holdup release. See "above ground fire" event tree for complete spectrum of consequence | | D | С | Ċ | c | U | 4 | 3 | 1 3 |
| 2.7 | Dome collapse waste fire | Tank Waste | Core drilling truck driven off ramps into open pit, dome collepse followed by a weste fire | ramps have edges and stop at ramp and | Assumed consequence is a dome collepse followed by a weste fire. See "tank loading" event tree for complete spectrum of consequences | review dome loading analyses | ^ | ^ | ۸ | ۸ | MI | 2 | 2 | 1 2 |
| 28 | Above Ground Fire | Tenk Waste | Core drifting truck driven on burm, truck overturne, burns and fire propagates back into tank ferm | | Representative consequences is a HEPA filter holdup release. See "above ground fire" event tree for complete spectrum of consequence | Determine whether procedural controls prevent trucks from being driven on burms | D | Ċ | Ċ | С | • | 4 | 3 | 1 3 |
| 5.1 | Above Ground Fire | Tank Weste | Non-detection of flemmable gas above tank with ignition | 1) High winds in area are prominent, 2) WHC-SD-WH- HSP-002 controls gas detection | Representative consequences is a HEPA filter holdup release. See "above ground fire" event tree for complete spectrum of consequence | 1) Walk down tank farm and de-energize equipment as necessary prior to connecting to sudilary equipment 2) Verity condition of riser seets. 3) Review gas detection calibration procedures. | D | C | C | C | 111 | 4 | 4 | |
| 41 | Above Ground Fire | Tank Waste | Non-detection of flammable gas above over open riser followed by ignition | 1) High winds in area are prominent, 2) WHC-SD-WM- HSP-002 controls gas detaction | Representative consequences is a HEPA filter holdup release. See "above ground filter event toe for complete spectrum of consequence | 1) review gas detector calibration procedures 2) require 30 minimum radius detance to energized equipment if riser open or 3) require class 1 Div. 1, Group B equipment within 20' radius when riser open | D | C | c | Ċ | Ħ | 4 | 4 | 4 |

A-15

 $\begin{array}{c} (1,0) \\$

August 8, 1996

| Accident Detabase |
|-------------------|
| Aode Dritting HA |
| Rotary Core N |
| Teble A.G. |

| | | | 10 | | | 1- | 100 | im | 10 | | |
|--|---|--|---|--|---|---|---|--|---|---|---|
| E-1614 | | · • | | | | | 6 | | | | |
| | | | 2 | - | 10 | - | e. | 0 | 10 | ~ | |
| | | | n | | 0 | • | + | - | 0 | 0 | - |
| Anunchas | | | | = | I E - | = | = | | - | | |
| | 0 | - 0 | | <u>с</u> | | 0 | | | 0 | | 0 |
| | <u>.</u> | 0 | | 10 | - | <u> </u> | | | 0 | | |
| | 0 | ο <i>γ</i> | • | | <u>د</u> | - | | | 0 | | 0 |
| | 0 | 0 | 80 | 0 | 80 | - | 0 | 0 | 0 | | <u> </u> |
| | | | <u> </u> | | - | - | | F | | ~ | |
| Recommendations | determine gas detector calibration details | verify all koose equipment near open riser. Including tools, connected to lanyards | verity all koose equipment maar open rinar, including sools, connected to lanyards | perform general wrate material competibility atach, including minimum mara required for uncontrolided reaction 2) do not allow reactive materials to enter tank ferm | perform general vesta material competibility study, including minimuru anas required for uncombridied reaction 2) do not allow reactive materials to enter tests farm | perform general vests material competibility study, including minimum mana required for uncontrolled reaction 2 do not sillow reactive materials to enter bank farm | perform general waste makerial competibility study, including minimum mess required for uncontrolled reaction 2) to bot allow reaction materizia to enter back form | | | Recommend using hoist for set number of rods attached. Add to procedure. | trapect drill acting looking for O-ting. Operator currently does not use an O-ting during pash mode operation. |
| Consequences | Representative contexquence is an estended low concentration buck gas solid) personnel exposure. See "gas rateaae avery event wee for complete spectrum of consequences | Representative consequences is a HEPA Riter holdup release. See "above ground firs" event tree for complete spectrum of consequence | Assumed consequence is a fire-induced dome collapse. See "dome specifie" event tree for complete spectrum of consequence | Representative consequences is a HEPA filter Indutus release. Sea "shore ground file" event tree for complete spectrum of consequence | Assumed consequence is a fire-induced doma collapse. Sea "doma spece fire" event tree for complete spectrum of consequence | Representative consequence is an administry concentration taxic ges (no adding personnel exposure. Ses "ges release avert" revis thes for complete spectrum of contacquences. | Representative consequence is a waste fite-induced fitter blowout release. See "veste fite" event tres for complete spectrum of consequences | Representative connecturing is a weets- fire-induced filter blowout release. See "wests fire" event tree for complete spectrum of consequences | Representative consequence is an extended low concentration toxic gas (no exist) personnel exposure. See "gas release aren" event tree for complete spectrum of contecturiors | Assumed consequence is a fire-induced dome collapse. See "dome spect fire" event tree for complete spectrum of consequence | Representative consequencia is a minimal representative material relevan. Sae "this string film" event tree for complete spectrum of consequence. |
| Preventive/hitigative Features | 1) High winds in area are prominent, 2) WHCSCWMA. NSP-002 controls pas detection | lanyards connected to (some) hardware | lanyaida comeched to (eome) hardware | | | | | Seal is lubricated to provent demage. Seal is inspected before installation | Seel is lubricatied to prevent damaga. Seel la Inspectad before Installation | Clamp faiture mode is chosed with hoss of pressure, but operatur action required to chose clamp | |
| Accident Description (Clause or Devision) | Nion-detection of taxic gas above over open Near | Hardwere dropped into open riser stitues riser side and cueses a spect and ignition of ges above riser | Hardware dropped into open riter causes a sperit, and ignition of dome space gas | Reactive material dropped into open riser, uncomfoltable chemical reaction generates Rammable gases followed by ignition of above-tank gas | Reactive material dropped into open inser, uncontratable chemical reaction generates Nemmable geses followed by ignition of dome space gas | Reactive meterial dropped into open riser, uncontroltable chemicul reaction generates tado gases | Reactive metentel dropped into open meer, uncontrelleble chemical reaction reaute in veste lignition | Friebes seul damage combined with operator error causes drift string drop and skille of lank waste surface, enchemic reaction results in veste term | Fritben seul demaged - kosa of containment | Operator and float not attached) or foot clamp father country and shing frop and child works surface, spark generated and doma spince gas fanties. | Weath in duit acting results in generation or therease of formacion games into diff acting ignition follows. Several instanton:) No trabilition of O-drag. 2) O-drag and failune Disea of program form. 9) blass of Phytocraticity freed gam pressure |
| Hazard | Tank Weste | Tank Waste | Tenk Weste | Tank Wade | Tank Waste | Tank Waste | Tank Waste | Tank Waste | Tank Wade | Tank Wade | Water in Arti |
| Kayumerd | Gas Roles | Above Ground Fire | Dome Fire | Above Ground Fire | Come Fe | Ges Refeete | Weeks Fire | Waste Fire | Gas Rohan | Come Fire | |
| Scenaria D | 42 | 43 | 4 | 47 | 4.4 | \$ | 9 | 6.0 | 52 | 6.1 | 6 .2 |

Table A.S. Rotary Core Mode Drilling HA Accident Detabase

| | | | | | - | <u>.</u> | 1 | | | | | |
|---|--|---|--|--|---|--|--|--|--|--|---|---|
| 제비원-3 | ~ | | | 2 | | | | 1 | - | - | | 1 4 |
| Hellin-W | m | | ·•· | | - | | | 1 | - | | <u> </u> | 1 |
| ALL - Chench | | | - | | | | | 1 | | - | | |
| Americana - | | | | - | | | | = | - | > | | = |
| Frances | · | - | - | - | - | - | - | - | - | | | = |
| memorive3 | 0 | | 0 | 0 | 0 | 0 | 0 | | 0 | | | 0 |
| MORILINE | 0 | < | 0 | | 0 | 0 | 0 | - | 10 1 | | | |
| | | < | | | 0 | | | ≪ | 0 | K | | |
| SHOLD | 0 | a | 0 | | | | | 100 | <u> </u> | | 0 | |
| Recommendations | verity ectual mee helgitta match drawing velues. Determine whether or how lank bottom buckleng condition determined | Ensure rear riser tank dome volume viewed with a camera prior to drill sping installation | Ensure naar riser tank dome volume viewed with a camera prior to drill string installetion | verify rotation operations are monitored | verify hold and inting cable are inspected prior to use | 1) Jakesminia gas delactar calitarion destil 2) require 20 minimum radua distance to respited equipment if riser open or 3) require dass 1 Dix. 1, Croup B equipment Weihin 20 radius when deep open | determine gas detector calibration details | welfy all loose equipment near open riser, including tools, connected to lanvants | verify all loose equipment near open near, including tools, connected to tanyarda | Perform general weak material competibility study. Including minimum mass regulated for uncontrolled needlon 2) most regulate to materials to anter bank form. | parform general weeks malacial competibility study, including minimum mass regulated for uncomfold neuclion 2) in the inclusion reactive methods to enter them them | perform general weste meterial competibility study, including minimum mass required for uncontrolled reaction 2) includes reactive meteries are study form. |
| Comsequences | hridation of a increased rate of tank contants leakage | Assumed consequencies a fine-induced dome collapse. Sea "dome space fire" event trea for complete spectra of consequence | Representative consequence is en entended low concentration taxic gas (no exits) personnel exposure. See "Tee sectors of consequences spectrum of consequences | Representative consequence is an exercised law concentration trail: gue (no solid) personnel exposure. See "gue solid ment ben har bu complete seectum of consequences | Representative contequence is a minimul on-the radioactive material release. See "still string fire" event ines for complete spectrum of consequence | Representative consequences is a HEPA film holdup release. See "above ground firs" event the for complete spectrum of consequence | Representative consequence is an extended low concentration toxic gas (no activity personnel exposure. See "gas release avent," event tree for complete spectrum of consequences | NA - Included in scenario 4.7 | NA - Included in econaria 4.5 | NA - Included in scenario 4.4 and 4.8 | NA - Included in scenario 4.5 | NA - included in econario 4.6 |
| Preventivefikitigative Features |)) Call: Fonds to determine the number of red sections meeting Chyfr the required number an Lakan. 2) num persaura interiodd provent than to obtion pervebration. 3) Mi makeuid can't pervebration. 3) Mi makeuid can't pervebration. 4) drift pervebration. | | | | String is inerted with nitrogen String should not confisin Rammable gas | i) itigh winds in sea are prominent 2 WHC-SD-MH. HSP-002 controls gas detection | 1) High winds in see are prominent 2) WHC-SD-WH- HSP-002 controls ges desection | laryards connected to (some) hardware | lanyarda connected to (some) hardware | | | |
| Accident Description (Cause or Devinden) | Addition of antra divid radi and sections of concessive indeption added because of lank, depth measurement uncertaintides results in tank bottom breach | Drift string stiffee metal object during Installation causing doma spece ignition | Drift string settuae metal object during indehiseton cuuring drift string to buckto and to cause thebee east failure | Crill platform schilles drill ating during rotation. Containment is breached | Hoist failure or failure to connect results in grappis drop and impact of samples. Ignition in diff ethng follower spark searration | Kon detection of filammable gas shows over open riser followed by tgnWon | Non-detection of toxic gas above over open fear | Hardware dropped into open riser causes a spert and ignition of dome spece gas | Hardware dropped into open rises strikes rises side and causes a spart and ignition of gas above rises | Reactive neteral dropped into open riser, uncontrollable channical neutron generates flemmede gases followed by ignition of above tenk or dome space gas | Reactive meterial dropped hilo open riser, uncontrollable chamical reaction generates todo genes | Reactive trainerial dropped into open riser, uncontrollative chemical neuction results in weeks lynition |
| Haish | Tank Waste | Tank Warda | | Tank Weste | Weste in delle string | Tank Weetb | Tank Week | Tank Weste | Tark Waste | Tank Watte | Tank Weeks | Tank Wade |
| P | Subbraneen apili | Come Fire | Gas Retere | Gas Release | | Above Ground Fire | Gas Roisses | Dome Fire | Above Ground Fire | Come Fia | | Weeks Fire |
| Scmerte 10 | 5 | - | • | i. | | | 6 11 | 511 | 1.11 | | 5 11 | |

| dent Detabase |
|------------------|
| Drilling HA Acci |
| teny Core Mode |
| Table A.S. Ro |

| E-KINK | ~ | in | - | In | ····· | | | | - | - | - |
|---|---|--|---|---|---|--|---|---|--|---|--|
| *** | ~ | ~ | - | ¦∾ ┿ | | | | | - | * | 1 |
| ARIA-GINNO | ~ ~ | 1 | • | 10 | n | ļ | 1 | | - | - | - |
| ABIT-SHOUR | n | n | | n . | * | | | | - | - | • |
| Frequency | ž. | - | 2 | 2 | | | # | /= + | 2 | 2 | E |
| memogrivn3 | ¢0 | 0 | 0 | 60 | 0 | | 0 | 10 | U | 0 | U |
| MOURE | < | U | υ | < | 0 | | | 0 | 0 | U | υ |
| 00000 | < | ٥ | 0 | < | <u>ں</u> | | <u> </u> | U | U | 0 | U |
| Public | æ | 0 | 0 | at. | 0 | | | 0 | 0 | 0 | 0 |
| Recommendations | review dome koeding analyzes | verify how leakage petits determined. Specify separation to energized aquipment | defariting whether for exhauster flow results in purge gas shut off | Lees science 14.1 | Replace autout pressure protohelics, mundity and emperature prode, IEPA define P photohelic prode, IEA, and CAMS with Class 1, Div. 2, Group B equiment an alternative approach may be to re- mengue these devices if high H2 concentration measured or | installation of a flame arreator in acrise with the actuaut. I required to cample for Other flammable grasse, recommend Chastly af equipment to Class 1, Dv. 2, Other B. | Require tanto ges semeons and personnel enscution if tange ges release believed eminent or if one occurs | Replace metring exhauster hose with conductive hose or with hose having a spirally wound grounded conductor | | | The pump and heet trace on condensate fines need to be checked for quelification |
| Consequences | Representative conséquence la a dome calinges. Ses Tank bading" event bre for complées spectrum d' consequences | Representative consequence is an mainded low concentration bails ges (no acticit) personna exposure. See "ges release and non-acquiters for complete spectrum of consequences. | Representative connecturence is an advinded lev concentration hale, gas (no bolid) prenome expount. See "ges release neut" event tree for complete spectrum of consequences | NA - inclused in scenario 14.1 | Representative consequences is a HEPA Mar hadde eleves. See "above ground Mar event tres to complete spectrum of consequence | | Representative connectaence is an manufact low concentration baic gas (no united personnel appound. See "gas united personnel apportants perchant of consequences | Representative connequences is a HEPA Mile holdup release. See "showe ground fire" event tree for complete spectrum of connequence | Representative consequences is a HEPA Ner holdup release. See "shore ground ke" event the for complete spectrum of consequence | | Representative connectances is a HEPA New holdup referees. See Tebore ground Ins" event tree for complete spectrum of connectance |
| Proventivatilityative Fasture | Photoheiko pressure monitora on eshaudar: Limita est on ha speed. Probability of tank sesting to the point of access vectore considered unitualy. | Photoheiko preseure monitora on exheuuler. Limite est on han apeed. Probebliky of tank seating to the provid of encess pressure comiddened unitiety. | Photohekic pressure monitors on advaustas: Lintis est on tan speed. Prodebility of tank seeing to the point of access pressure considered unitaly. | Photahelic pressure monitors on exhauster. Limits est on fan speed. Die point of envis sealing is the point of enves pressure considened unificely. |)) Extransfer heretor cleanañed Cleana 1, Div. 2, Oroup B | ndia, wa assume flammedra gas release from snother tank in farm dispenses prior to reaching operational adhaustar | | | Hoee is menually inspected during installation. | Seal pot is incised except to open drain line and manual pump down. | |
| Accident Description (Cause or Davisition) | Tank collapse cavead by excessive vecuum on bark. Low presure caused by plugged helds and balve of CP photohelic monitors | Unifilmend reveau through leakage poths caused by excessive lank, pressure as a result of high arhauder flow | low etheweter flow causes normal data sub-grupp gas to one-presentite abing and result in untilitered releases | Tami collepse caused by successive vectum on tank. Low pressure caused by high flow and failure of DP photohelic monitors | Enhauster component initates enhaust gas the | | Random or activity-initialad gas release from weate generates high levels of tado genese | Electrostelic charge on exheuder hose Initiates above ground fire | Fileditio enfauturitor hose test, tear not disconseruit, surge gas finer lancas Remmedite gas out test, fine results on tep of tests. | Seal put hikine neute in fins in condensate into | Heat thice type for condensate line or condensate pot causas lignifican |
| Hazerd | Tank waste | Terk week | Tank waate | Tank weeks | Turk West | | Tank Weste | Tonk Weste | Tark Weeks | Tank Wards | |
| Keyword | | Gas Release | Qas Robase | Colleges Colleges | Abore Ground Fire | | Gas Robers | Above Ground Fire | Abore Ground Fire | Above Ground Fire | Above Ground Fine |
| Stenart B | 14.1 | 14.2 | 14.5 | 11 | 5 1 | 14.5 (cent.) | 87 | 111 | | 14.0 | |

Table A.5 Retery Care Mode Diffing HA Accident Database

| THE SHIP | en l | 4 | io. | in . | 7 | en | - | ~ | 2 | 101 | 2 | - | 10 |
|--|--|---|---|---|--|---|--|---|--|---|--|--|--|
| - | • | 4 | e) | e . | ~ | n | n | 2 | 2 | ~ | - | - | 1 |
| Mel 2-estand | n | 4 | en . | n | 2 | n | 4 | 2 | 2 | ~ | - | - | 1 |
| Public-Risk | ٦ | 4 | 4 | • | T | 4 | 4 | • | 4 | 4 | 2 | en | - |
| Frequency | æ | | ¥ | 2 | = | # | * | # | | = | Ŧ | - | |
| Memory val | 4 0 | ٥ | æ | 6 | 6 | 6 | ٥ | 6 | 8 | a | • | œ | 60 |
| MOLION | ø | ۵ | œ | Ð | 8 | æ | с U | 8 | 6 0 | œ | .≪ | 6 | 100 |
| | ¢ | 0 | 60 | Ø | 8 | .au | ٥ | 6 | 8 | æ | < | æ | 100 |
| | 0 | 0 | ٥ | 0 | 0 | 0 | ۵ | 0 | 0 | a | 8 | 0 | 0 |
| Recommendations | o require 10 drill string volume purges each s'inna purge la interrupted Determine retebility of purge system | evou | e investigate refability of temperature I controller | a determine number of barriers between a propare and nitrogen | a review vendor QA procedures | a determine whether a purge line back to the a lank is a possible gas bypase peth | | e determine suighity of high RPM detection system to shul down drifting | edd lock out testure to 2 remeining trucks | | replace mater Arven RLU with mechanically operated ane | i thrift bit targes to sufficiently low value so I that themme energy input to bit four enough Ib be conducted every from bit by warks | o determine releability of purge loss detection a end dell etudown systeme. |
| Contequences | Representative consequencie a waste-fin Induced Riter blowcur neeses. See "waste fine" event tree for complete epectrum of consequences | qualitatively assigned - speculative | Representative consequencia a waste-fit induced Mar Isborcul release. See "weak fite" event tree for complete spectrum of consequences | Representative consequencia a varata fre induced fiber Mowcul release. See "wash fre" event trea for complete spectrum of consequences | Representative consequencia a weste-fin Induced filter blowout release. Ses "weste fire" event trea for complete spectrum of consequences | Representative consequenci is a vesta file reduced file bivorud release. Sao "vesta file" sent bea for complete spectrum of consequences | ALARA Issue | Representative consequencia a veste fin Induced Mar biowcut release. See "vest Are" event tree for complete spectrum of consequences | Representative consequence is a waste- fire-induced filter blowcut release. See Veste fire" exect tree for complete opectrum of consequences. | Representative consequencia a waste-fin induced filter Isbourd release. See "west fire" event tree for complete spectrum of consequences | Assumed consequences is a fire-induced dome colleges. See "Vome space fire" event here for complete spectar of consequence | Representative connectures in a waste fire Induced filter blowsuk reference. See "weete fire" event tree for complete spectrum of connectuences | Representative consequencia a vesta fra Induced Ray Iduncul release. San "vesta fire" event the for complete spectrum of consequences |
| Preventivefittigative Factures | Two of three redundant flow methos must agree for unit to operate | Temperature controller stops operation of nitrogen truck et temp below 40 F | Temperature controller stope operation of nitrogen truck at temp above 100 F | Heat exchanger is an ASME pressure vessel | Vendor OA program to ensure no other gas in substituted. | Chrygen monitor on truck may series fou onrgen and sound sherm | Orygen monitor on fruck may sense fow crygen and sound sterm | Redundant magnetic speed monitor with alarm followed by eviormatic rotation shutdown | Revense direction is locked of on 2 of 4 drift trucks | bits are visually different and must and bit when rotary mode operation expected | | | none - bit temperature mersulamment not possible given seieling technology |
| Accident Ceacription (Cause ar Davistion) | Low nitrogen jurge gas frow causes bi over temperature which results in a weste fre | Low nitrogen purge gas temperature because of failed controller or failed heater | High mitrogen temperature because of faced controller or runnwy heater results in high bit temperature and subsequent wates fre | nitragian hadi auchangar fažura resulta In propene flowing into drift atmig | Vendor supplies gas ofter från nårogen in purge gas truck bank, gas in flammadet, gas contact with bit during drifting hillisies a webe fre | Buolen V2 line downisteern of flow meters cauling th over temperature and subsequent watte fire | diseal fume build under core driffing truch tent | High diff bit RPM causes high bit Ismperature and subsequent weath burn | Reverse drill bit rotation caves accessive bit temperature and vests burn | Wrong bit or no bit installing (push mode setup) . normal robary care drifting cases high bit temperature and autosequarit watte fire | Chill bit catches fast in waste and is unable to rotate, drill sting breakage neulla, gattere winnig or non-qualified qeligiment gattere sources, ignition in dome apoos bolignes. | Combination of accessive briggs bading and dominised force curves by to oretheat, resulting in wests fire | Loos of jurge gas flow causes bit to contrast resulting in weap fla |
| Marah | Tank Wate | Tank Wate | Tent West | Tank Wedle | Tank Watte | Tank Waste | | Tenk Wash | Tank Warts | fank weee | | Tank Wade | Tork Warb |
| Kayward | Wath Fire | Ach | Warts Fire | Waste Fire | Wath Fire | Weth Fire | Gas Reisess | Weste Fire | Wate Fre | Weste Fire | Dame Fire | Veite Fin | Warte Fire |
| Scenarie 10 | 15.1 | 15.2 | 153 | 15.4 | 15.4 | 12 25 | Globe | 18.1 | 16.2 | 16.4 | 10.5 | 10.0 |) e |

| | | | | | | | · | | | | |
|--|--|---|--|--|--|--|--|---|--|---|---|
| 1142-3 | - | ~ | 2 | ~ | ~ | ۲. | | | • | 4 | n |
| Marking | - | ~ | N | - | <u> n</u> | | ien | 1 | i - | n | N |
| ARIA-ChENO | - | 2 | ~ | - | N1 | 4 | ie. | | * | 4 | m |
| Public-Risk | n | * | • | ~ | . | - | - | · · · · · · · · · · · · · · · · · · · | 14 | | n |
| Frequency | - | | - | | - | | 2 | = | | * | - |
| memnenivn3 | a. | 8 | | | (B) | | | | | 0 | |
| MOURE | | 8 | | < | (D) | | × | 0 | 0 | 0 | 0 |
| | 8 | 8 | 8 | < | æ | | < | | 0 | | |
| 2000 | <u>ບ</u> | <u>ل</u> | 0 9 | | . | | | | ں | 10 | 0 |
| Recommendations | review and verify present hast 4 modeling enalyses | consider adding drift bit torque limit, imenack | review bit operational modeling and besting parameters including sensed generation and ignition in different geneous atmospheres | replace motor 4then RLU with mechanically operated one | | determine driffing served generation rates | review bit operational modeling and testing perameters including served generation and ignition in different geneous atmospheres | require toxic gas annaurs and personnal everuation if isinge gas release believed eminent or if one occurs | require bado ges sensors and personnel execution I large ges release believed eminent or if one occurs | | Imestigets high purge gas frow detection and controls |
| Contequences | Representative consequence is a verta-fire induced fitter blowout release. See "verta fire" event trae for complete spectrum of consequences | Representative consequence is a weste-file induced filter blowout release. See Tweste first event tees for complete spectrum of consequences | Representative consequence is a weste-fire induced filter blowout indexxe. See "veste fire" event tee for complete spectrum of consequences | Assumed consequenc is a fire induced dome collapse. See "dome space fire" event tree for complete spectre of consequence | Representative consequence is a veste fire induced film: blowout release. See "veste fire" event tree for complete spectrum of consequences | Representative consequence is an entimoded low concentration taxle ges (no exist) personnel expounts. See "ges infesse event tree hu complete specture of consequences | Assumed consequence is a fire-induced dome collapa. See "dome space fire" event tree for complete spectra of consequence | Representative consequence is an entrolled by concentration but ges fro soft) percensel expose release event tes for complete spectrum of consequences | Representative consequence is an anothed hav conventation have get ho cold) perconnel seportes. Se sea revease event event thes for complete spectrum of consequences | Representative consequence is an extended law concentration tarks gas (no subjectment argument. San "gas Iniverse event" event text complete spectrum of consequences | Representative connectuence is an entended low concentration tradic gas (no solid) personnel expanses. See "ges release event" event tree for complete |
| Preventive Feature | | | ediministrative procedures, visual differences between qualified and non-qualified bits | | | | unitably, propagation most titely to occur for situations where initial prices strations are initial prices in a spread of additional flammable dust or flammable dust or | 55 galton timit on winger augely. ISSD stimmable timit in 100 galtons for basing tanks, 230 19 bs 1500 galtons can be added 19 bs 1500 galtons can be added to after with Nuclear Safety be after with Nuclear Safety | 56 gaton limit on water auppir, 1550 allowable limit in 100 gatore for leading banks, 250 gatore for non-leading tenks, 10-b to 1500 gatore can be added to other with Aucteer Safety to other with Aucteer Safety | | |
| Accident Description (Cause or Deviation) | haddquift hatbig envelope of improper heat up modeling reutle in hit conheating huming normal (or unamicipated) stiffing outsing bit to overhead neutifing in weaks fite | (High dril toque bad cerees dril bi breat which bilitetes a sperit of over temperature resulting in a weate fre | current operational emelope victabled by substructure results from normal entitie. Watch file results | Drifl bit impacts hern in wester, drifl sting breekunge nesutta, esposed wiring ar non- qualities deguisment ignition sources. Jarition in donne systers follows. | Drift bit impacts from in weeks, sport generating which ignities weeks generue phane, fre propegates to weeks. | incorrect operational emelope results in drifting in a mode which generates encrease durg, antwurke filter plugs, alget dome space over-presultation needs in unfiltered interest involut leaturge paths | incorrect operational envelop results in drifting in a mode which penerates excessive dust, possible dust ignition and dome cultapes. | eddition of an excessive amount of weter fouring stampts to clean ar locaen and by results in ammoula release from wete | drill string contaminated, addition of an excession amount of weak during stimutis to chean or broas drill bit teurlis in emmonia generation and release | Normal diviling meke sufficient weste to result in a statem velocoso | encessive purge gas from initiates ges referee frough leakinge petha |
| Hazard | Tenk W uto | tank Waste | Tark Wette | Tank Weste | Tank Week | | Tank Weste | | | | |
| Kayword | Weeth Flig | Wash Fire | Weath Fire | Dome Fire | Waste Fire | Ges Role | Dome Fire | | | Gas Rofesso | Cas Retent |
| Cenerte D | 158 | 4 Q | 10 10 | 11 01 | 9.9 | 10.17 | 10.12 | 18.7 | 88 | 18.0 | 10.15 |

Table A-5. Robery Core Mode Dritting HA Accident Database

Table A-5 Robry Core Mode Drilling HA Accident Detabase

| ****** | 7 | 2 | • | | n | 4 | 5 | 2 | 4 | 4 | - | | |
|--|--|--|---|-------------|--|---|--|--|---|---|--|--|--|
| | - | n | 4 | • | | • | 2 | n | • | - | - | | |
| Ael7-etienO | + | • | - | | 4 | * | 2 | • | - | 4 | 4 | | |
| Public-Risk | + | • | 4 | | • | 4 | en . | ~ | 4 | 4 | 4 | | |
| Frequency . | 2 | | E |] • | = | 2 | 莱 | * . |) dar | 1. | | | |
| Sector and the sector | ٥ | iu I | ٥ | •···· | ပ | ٥ | æ | ٤. | 0 | 0 | 0 | | |
| MOLEN | υ | 0 | U | L | 0 | U | < | 8 | U | 0 | ٥ | | |
| | ٥ | 0 | 0 | | ٥ | 0 | < | 8 | 0 | ٥ | ٥ | | |
| Plate | ٥ | 10 | 0 | <u> </u> | ٥ | ٩ | æ | < | ٥ | 0 | 9 | | |
| Recommendations | i Use purge ges on 1st sample to next sting | determine reliability of bottom force detection and drift direction revenant systeme | | | | effection and drift direction reversal systems require bade gas semicors and personnel enclution if large gas releases believed entreen of Core occurs | | | | | | | |
| Consequences | Representative consequence is a minuma carella notacive material release. See Cetti sting fice acter thesi to complete spectrum di connequences | h Nieton of or Increaced rais of tenk contents leakage | Representative consequence is a minima on the reductive material release. See Cellif sing the "even thes for complete spectrum of consequences | | h Nation of or increased mee of tank contents leatage | Representative consequence is an estanded proconcertaintance gas (no colid) permanent apposure. See "pas refease event" event tree for complete spectrum of connequences | Assumed consequence is a fina-induced dome collapse. See "dome spece fina" event tree for complete spectrum of consequences | dome collapse without a fire | Representative consequence is an adminided four concentration butic ges (no relief) personnel exposure. See "ges release avent" event thes for complete spectfung of connectances | Representative consequence is an estanded four concentration butc gas (no solid) personnel argosur. See "gas misese event" event thes for complete spectrum of consequences | ALARA lesus | | |
| Fraventivenkingetve Festures | | 1) bit nurs cooler when drilling metal (more hest removed) | | | United down force 2 through the it operator. through the it operator. through the it operator observation 3 drill string may buck to prior to bruit permetador 9 only cathole hit a expected to pervictible bottom. | 55 galon limit on week supply. 2010 advention limit is 100 galons for insuing tanks. 200 galons for non-leaking tanks. Up to 1900 galons can be added to eller with Nucleus Safety approved | Clemp faikure mode is closed with loss of pressure, but operator action required to close clamp | Administrative requirements call for vaing the defi-motor in reverse to lift dell string until sting bit is above veste | | | Requirements to anif string and use a ned. detector on outside of diff string | | |
| Accident Description (Cause of Deviation) | Lisis di sezi during burup mode (no N2 periodi - amina antare dali string. Flammable concentration encended in duli string. Ignition eccurs | Bit hits and breeches tank bottom | Failure be more drift string above vesta strations on its amysie print, to removing semplar - results is drift string full of vesta Flammable concentration escaeded in drift string, ignition occurs | | High down force punctures tank bottom | Addition of large volumes of water to tank Indiature gas volueses | Operator error (hoist not sitteched) or boot champ fakure counting drift stirring drop and shits of weath surfaces, agent generated and dame spece ges lignities. | Humain ence - jib holdt used to kin drill sting - truch pulled off jocks, falls, impacts dome. Dome collspee results | Human error - jib hoist used to kit drill sting - buck pulled off jacks, Ingects dorne. Dome containment breached | Sweb of bore by drill sting during removel causes a two pressure pull release of gas to bunk dome | Error causing wants in shill stitug (i. a. seel taliant) bolowed by physiong of laak - nasuit ts 5 it of westa spilled to ground | | |
| Hazari | Wash Tank | Wette Tank | Waste Tach | | Wester Test | Weathe Tank | Tank Wade | Tank Warte | Tark Wash | Tark Weste | tank Wasta | | |
| Keyward | Critica Series | Subharranean apili | Örill Skring Fing | | Subtraction of the second seco | Gas Reises | Dume Fire | | Gan Release | Ges R | Abore ground | | |
| Bcentrie D | 10.3 | 505 | 30 F | 30.4 (cont) | 8 | 32.1 | 32.2 | 575 | 74 | ** | 1 3 | | |

| Deter |
|------------|
| Accident |
| Drilling H |
| Con Mat |
| Retury (|
| Table AS |

| | | | | | | 1 | | 1 | | | | | | | · |
|--|---|---|---|--|--|---|---|--|--|---|---|---|--|---|---|
| 1012-3 | - | 1 PP | - | | • | - | n | - | - | <u> </u> | | | 14 | n | 17 |
| XBIN-AA | ~ | n | | n | | | ~ | | | | | <u> </u> | N | | 14 |
| ANT-WIND | 2 | (n) | * | - | - | | N | - | | N | | n | 2 | 10 | 4 |
| Public-Risk | | - | - | - | * | 14 | | | - | • | | | ~ ~ | * | * |
| Louinbert | | # | | 2 | | 12 | E | | | * | | | = | * | 1 2 |
| memory 13 | < | æ | <u> </u> | < | 0 | 0 | | 0 | 0 | | <u> </u> | 6 | < | υ | U |
| MOLIOM | 60 | (6) | 0 | < | 0 | 0 | < | 0 | 0 | • | 0 | U | < | U | U |
| | 8 | a c | 0 | < | 0 | 0 | < | | 0 | æ | | U U | < | U | U |
| Public | < | <u> </u> | 0 | < | 2 | <u> </u> | | 0 | 0 1 | 0 | | <u>ہ</u> | < | 0 | 0 |
| Recontinendations | review dome beding analyses | review dome loading snalyses | teriew dome toeding analyses | retex dome loading evelyzes | add administrative control on vehicle speed | add administrative control on vehicle speed | add administrative control on vehicle space | add administrative control on vehicle speed | add administrative control on vehicle speed | review dome loading analyses | teview dome bading analyses | | review dome loading analyzes | Determine whether proceedural controls prevent trucks from being driven on burnes | 1) iverview ges detector calibration (intercondense 3) requires 50 (or other) minimum radius distance to enderal equipment 8 rese open or 3) require class other) regims when they coon other) regims when they coon |
| Consequences | Representative consequence is a dume colleges. See "and buding" event tree fo complete spectrum of consequences | Representative consequence is a dome collapse. See Tank loading' event tree fo complete spectnam of consequence | Representative consequence is an estandard low concentration tude gas (no solif) personnel esposue. See "and, bading" event the for complete spectrum of consequences | Assumed consequence is a doma collapea followed by a waste fine. Sae Tank loading" event tree for complete spectrum of consequences | NA - included in scenario 1.5 | NA - Included in scientria 1.6 | NA - Included in economia 1.7 | NA - Included in ecenario 1.5 | NA - Included In scenario 2.9 | Representative consequence is a dome collapse. See Tank bacing" event bee fo complete spectrum of consequence | Representative consequence is an estimated low concentration tradoges (no solid) personnel exposite. See "trad, totaling" sent trae for complete spectrum of consequences | Representative consequences is a HEPA Mer holdup release. See "above ground fre" event the for complete spectrum of consequences | Assumed consequence is a dome contepes followed by a weath flag. See "Early toading" event thes for complete spectrum of consequences | Representative consequences is a HEPA Bits holdup release. See "Bone ground An" event tee by complete spectrum of consequences | Representative consistuances is a HEPA Regressentative summar. See "above ground Re" send too any advertant of consequences |
| Pravonti vafilitigativo Featuros | Admin. control OSD-1-51-0013 ads limits on lant loading | Admin. control gives procedures for critical crane illis | Admin, control ghes procedures by critical creme lifts | Admin control gives procedures tor critical creve title | | | | | | nampa hare adges and stop st namp and | دهدیده مروحه میش داده در دهدی میش | nampa hare edges and ship at ramp end | rainge have edges and stop at rainp end | | 1) High winds in ana ara prominant. 2) WHC-3D-WH. HSP-002 controls gas describen |
| Accident Drechption (Causes or Deviation) | Domme static trad encreaded, dome colleges with served referees of weste | Lond dropped on top of tank during M. dome collapse toflows. | Load dropped on top of bank during W. Next, containment branch | Load dropped on top of tank during Mr. tank dome coltapes followed by wards fre | Vehicle Impacts Riser or Other part of Tank containment, containment Inseched | Vahida Impada Riser or Other part of Tank containment, sperk Ignites above ground Rea | Vahicle impacts and breaches River or Other peri of Tank containment, which built brit breached, spach lightes dome search the | Drift truck impacts Riser or Other part of Tank containment, containment breached | Vahicle hryacts Riser in Other part of Tank containment, which had lank breached, spart lankes shore ground fite | Corre drilling truck driven all ramps into open pil, dome collegee follows | Core defines buck driven of ramps into open pit, dome containment breached | Core drifting truck driven of nampa into open pit, truck hud fre reads | Core drifting buck einen of range into even på, éome odlapse holowed by varie i fine | Core shifting truck show on hum, Fruck continue, huma and fine propagates book this and hum. | Men detection of Recommands gain above cose open free followed by gallon |
| Hazard | fank Wede | Tank Weste | Tank West | Tank Weeks | Tenk Waste | Tank Watte | Tank Waste | Tank Weste | Tank Weste | Tank Wedte | Tank Watt | Tark Waste | Tank Weeks | Tank Wade | |
| Press (e) | C Dome | | Gas Robers | | Gen Release | Above Ground Fire | Dome Fire | Gas Reisese | Alone Ground Fire | | Gan Release | Abore Ground Fine | | Abore Ground Fire | Abore Grunn |
| R B C B C B C B C B C B C B C B C B C B | V GE | 37.2 | FAS | r.c. | 375 | 37.6 | 97.6 | 37.10 | 37.12 | 37.13 | 37.14 | 37.15 | 57.10 1 | 37.15 | 37.16 |

Table A-5. Robry Core Mode Drilling HA Accident Detabase

| Resperte I | Kerepre | Hatard | Accident Description | Preventive/Mitigative Features | Consequences | Recommendations | | | | ¥ | | _ | - | $\neg \neg$ |
|--|----------------------|------------|---|--|--|--|---|--------|--------|-----------|-----------|-------------|------------|---------------------|
| iD . | | | (Cause or Devision) | | | | Ĩ | Onerte | Warker | Environme | Frequency | Puttic-Rhal | Onstanding | W.F.INE E.F.Inel |
| 37.20 | Gas Release | Tank Waste | Non-detection of timic gas above over open riser | 1) High winds in ansa ara prominent. 2) WHC-SD-WMA- HSP-002 controls gas detection | Representative consequence is an extended low concentration trade gas (no solid) personnel exposure. See Tank loading" event tree for complete spectrum of consequences. | determine gas detector calibration details | Ö | D | C | Ď | N | 4 | • | 3 4 |
| 37.21 | Above Ground Fire | Tank Weste | Hardware dropped into open riser causes a spark and ignition of gas above riser | lanyards connected to (some) hardware | NA - Included in scenario 4.3 | verify all toose equipment near open riser, including toots, connected to lanyards | D | С | C | Ē | Ī | 4 | 4 | 4 4 |
| 37.22 | Dome Fire | Tank Waste | Hardware dropped into open riser causes a aperic and ignition of dome space gas | lenyards connected to (some) hardware | NA - included in scenario 4 7 | verify all loose equipment near open riser, including bols, connected to lanyards | B | Ä | ^ | 8 | Ħ | Ĵ | 2 | 2 3 |
| 37.23 | Above Ground Fire | Tank Waste | Reactive material dropped into open riser, uncontrollable chemical reaction generates flemmable gases followed by ignition of above-tank gas | | NA - included in scenario 4.6 | perform general weste material compatibility study, including minimum mass required for uncontrolled reaction 2) do not allow reactive materials to enter tank farm | Ď | Ċ | C | C | Ħ | 4 | 4 | 11 |
| 37.24 | Dome Fire | Tenk Waste | Reactive material dropped into open riser, uncontrolistie chemical reaction generates flammable gases followed by ignition of dome space gas | | NA - included in scenario 4.4 | 1) perform general weste material compatibility study, including minimum mass required for uncontrolled reaction 2) do not allow reactive materials to enter bank farm | 8 | ^ | ^ | 8 | Ħ | 3 | 2 | 2 3 |
| 37.25 | Gas Release | Tank Wasto | Reactive material dropped into open riser, uncontrollable chemical reaction generales ando gases | | NA - included in scenario 4.5 | 1) perform general weste material compatibility study, including minimum mass required for uncontrolled reaction 2) do not allow reactive materials to enter tank farm | D | D | Ĉ | Đ | Ħ | 4 | 4 | 34 |
| 37.26 | Waste Fire | Tank Waste | Reactive material dropped into open risar, uncontrollable chemical reaction results in weste ignition | | NA - included in scenerio 4.6 | perform general waste material compatibility study, including minimum mass required for uncontrolled reaction 2) do not allow recrute materials to enter tenk farm | D | 8 | 8 | 8 | N | 4 | 3 | 3 3 |
| external events and natural phenomena | Above Ground Fire | Tank Waste | Fuel spill during operation resulting in fire | Fuel epill limited to 5 gallons | Representative consequences is a HEPA filter holdup release. See "above ground firs" event the for complete spectrum of consequences | Put spill pan beneath truck | D | C | C | C | Ħ | 4 | 3 | 3 3 |
| lametra bna stneve latutan phenometra | Wante Fire | Tank Waste | Fuel spill near pump pit with pump pit drain open results in exothermic reaction in weste | 2. | Representative consequence is a waste- fire-induced filter blowout retease. See "waste fire" event tree for complete spectrum of consequences | Put spill pen benesth truct. Do not allow fuel tank filling while riser open | D | 8 | 8 | 8 | | 4 | 3 | 3 3 |
| external events and natural phenomena | Above Ground Fire | Tank Waste | Ignifion within equipment within 30 ft of riser causes fire | | Representative consequences is a HEPA filter holdup release. See "solve ground filter" event tree for complete spectrum of consequences | Prevent other operations while rotary core drilling is being performed or prevent equipment operation within 30 R of near (assuming riser dia = 1 R and adding 50% safety factor) Verify 20 R separation distance correct | D | c | C | Ċ | | 4 | 3 | 3 3 |
| external events and natural phenomena | Waste Fire | Tank Waste | Drilling is performed in the wrong tank. Tank could be high heat or here non- queffied equipment, weste fire results | QC verifies tank location | Representative consequence is a waste- fire-induced filter blowout release. See "weste fire" event tree for complete spectrum of consequences | | D | 8 | 8 | B | 1 | 4 | 2 | 2 2 |
| external events and natural phenomena | Waste Fire | Tank Waste | The wrong truck is used in the drilling process (i.e. crew uses rotary mode on a push mode truck operation outside operational gerameters and waste burn result | | Representative consequence is a veste- fire-induced filter blowcut release. See "weste fire" event tree for complete spectrum of consequences | | D | B | 8 | 8 | | 4 | 2 | 2 2 |
| external events and natural phenomens | Above Ground Fire | Tank Wasto | The exhauster is connected to the wrong dear - tank is not exhausted. Tank could twoe a gas release above ground and subsequent fire | | Representative consequences is a HEPA Mar holdup release. See "above ground fire" event tree for complete spectrum of consequences | | D | C | C | Ĉ | | 4 | 3 | 3 3 |

A-23

٢

avir S

WHC-SD-WM-SAD-035, Rev. 0-a

•

.
| | | · · · · · · · · · · · · · · · · · · · | | | | | | | | | | | |
|---|--|--|---|--|--|--|---|--|--|--|---|---|--|
| Anin-3 | <u>م</u> | - | |)#1 | n | 14 | 147 | P | 10 | 10 | n | 10 | 1 m |
| THE M | - | m | - | n | • | | - | - | - | 10 | 1- | [n | - |
| Nel%-shenO | - | n | 4 | n . | 1 | | - | - | - | 10 | - | 10 | - |
| Public-Rink | 2 | 4 | 4 | - | n | - | n | (m) | e7 | 2 | n | N | 1 |
| Frequency | - | 2 | E | * | - | æ | ≥ | 2 | 2 | E | 2 | is . | ≥ |
| mennenives | ø | 8 | ٥ | U | ٥ | 0 | < | < | < | < | < | < | < |
| MUCHA | < | 6 0 | υ | U . | ٥ | U | 8 | 8 | 8 | 0 | 100 | 60 | 6 |
| OHNE | < | Ð | 0 | υ | D | ٥ | 8 | 60 | œ | ¢ | æ | 100 | 6 |
| bublic | æ | ۵ | 0 | 0 | D | D | < | < | < | < | 1 | 1< | < |
| Recommendedone | Prevent other operations while rotary core driffing is being performed | | | | | | review dome hading analyses | | | | | | |
| Consequences | Assumed consequence is a fire induced dome collapse. See Yame spece fire event tree for complete spectrum of consequences | Representative consequence is a wase- fine-induced film: blowuld release. See Vesta fine" event bes for complete spectrum of consequences | Representative consequence is a minimal redit radiocide mitalant areas. See "All diving fina" event tree for complete specifium of consequences | Representative consequences is a HEPA Mile Induly release. See "shore ground fire" event the for complete spectrum of consequences | extended has of also power and Interruption of tank farm activities | Representative consequence is a minimal on-site redioactive material release. See "defit string fire" event tree for complete spectrum of consequences. | dome coflapee without a fire | dome collegee(s) without fire due to structural failure | NA - seurre bounded by releases from structural faitures | dome collepee without fire of a few (4) tenhs due to structural failure | NA - assume bounded by releases from souching failures | dome ochtpes without fina a' a hew (12) berlis due to structural failure | NA - maxima bounded by releases from Studunsi fallures |
| Praventive Mittigetive Feetures | | | | | | | | | | | | | |
| Accident Description [Cause ar Devision] | Activity in interconnected birth. In term courses fire that propagates to term bain sempted | Wrong bit wed when going from push mode to rotary mode - bit over-temperature results in weste fre | Whong sampler insented when gang from they much to prack much with any filed with watch, increased rate of fammable gas generation results, formerable generations enceeded in anit ahing, gendam secured of anit ahing, | deed hui released han buck during deed generator buil, ratiking achites, hui flows into text ferm and ignites | volcants explicin results in buildup of each and dust | valezanic eruption results in buildup of esh and duct, gue builds up in drill ething prior to recovery, fre initialme at restart of activities | volganic anythion results in buildup of each and duat, built dorne loading limit exceeded | .255 (middle range) ania mia ment | .25g (middle range) selamic event neuka In fire(s) in tank dome spece | .45g (middle fange) aelemic exert | .45g (middle range) existmic event results In time in tank dome speces of a few (4) tanks | .55g (middle range) seismis event | . 55g (midde renge) seismic event naute In firms in texts dome specces of a few (12) betts |
| hicehi | Tark Waste | Tank Weeks | Tank Wate | | | | | | | | | | |
| Kayward | Dome Fine | Weethe Fire | Crit Setre Fire | Allow Ground Fire | volcane | Com State File | 2 3 2 3 | | En Fie | | Denia Fia | | Dome Fire |
| Cenarto D | And a state of the | Changing operations modes | Changing operational modes | Increase and a second s | arbunde search and natural | erternel Netre end | external events and natural | and a stand | And the second s | | | | |

Table A-5. Robery Core Mode Ording HA Accident Detabase

WHC-SD-WM-SAD-035, Rev. 0-a

A-24

| | | 1 | | | 1 | | | - | | | | | · · · · · · · · · · · · · · · · · · · |
|--|---|--|--|---|---|--|---|--|---|--|--|---|---|
| 1483 | <u>^</u> | - | [m | | | | - | · • | | | | n - | T |
| MARA-W | ~ | - | 1 | | - | | - | | | n | n | n | |
| Arin-etrano | ~ | | n | - | | n | | - | | - | n | • | 1 4 |
| MHA-SHdug | ~ | - | 147 | 10 | 10 | | | - | - | | | - | 4 |
| Frequency | | 2 | 2 | 2 | 2 | 2 | £ | 2 | 2 | 2 | 2 | • | 2 |
| meenoritrn3 | | « | < | 4 | | • | | 0 | | E | | 0 | 0 |
| MOLIER | < | | < | 160 | | | 0 | 0 | < | < | < | 0 | 0 |
| OMMAN | < | 0 | < | 100 | 100 | | 0 | 0 | < | K | < | 0 | 0 |
| Public | < | < | < | | • | < | • | | a _ | | · | 0 | 0 |
| Recommendations | | | | | new dome points and read | determine how bruck poetformed to secure condition during periods of possible high winds | imestigate electrostatic charging polanitat of lant windows | | | | | | review rectart procedures to ensure F2 concentration measured prior to re- energizing electrical equipment |
| Contexpremotes | domé collapie without fire d'a mariy (45) tents due to stuctural feiture | NA - assume bounded by releases from structural failures | dome collapse without fire of the majority (122) of the tanks due to structural failure | NA - essume bounded by releases from structural failures | Representative consequence is a dome colleges. See "tank bading" event leve for completes spectrum of consequence | Rapresentative consequence is a dome collepse. See "tank boding" event thes for complete spectrum of consequence | Representative consequences is a HEPA Rite holdup release. See "book ground Rife "event tae for complete spectrum of consequences | Representative connecturences is a HEPA filter holdup release. See "above ground filte" event tree for complete spectrum of connectuences | Assumed consequence is a fire-induced dome collapse. See "dome space fire" event bee for complete spectrum of consequences | Assumed consequence is a fire-induced dome collapse. See "dome space fire" event thes for complete spectrum of consequences | Assumed consequence is a fire-induced dome collapse. See "dome space firs" event tree for complete spectrum of consequences | Representative consequences is a HEPA film holdup release. See "boove pround film" event the for complete spectrum of consequences | Representative consequences is a HEPA Mex holdup reases. See "shore ground firs" event the for complete spectrum of consequences |
| Freventingative Farance | | | | | اع سام وسيد ما مهمدووسه | 15 mph wind limit on operations | main tant body la camea, only whidows are pleatic. 15mph wind finit on operations | | | | | graund cover removed or controlled | |
| Accident Description (Cause or Daviation) | .70g (middle range) saismis event | . 70g (middle range) externic event results in fee in tenit dome speces of a many (45) tents | 1.0 g (middle nange) selamic event | 1.0 § (middle mage) seismic event results In fite in tent doma spaces of the majority (122) of the tentus | esterne winds during equipment hubbleton or removal activities results in drop of equipment anto dome | estreme winds during duling activities results in drilling tuck uppet due to juck or remp tokue | moderate or extreme winds cause truck text to become electronizative charged, flammable gene present when activities commence, text brushes truck igniting genese | kehtning strike results in above ground fre | fightning strike on en inedequelehy grounded tank results in tank dome space fire | alipitene cresh into tunk results in dome collepse | reaction of non-conductive material into dome space rewite in static discharge and subsequent dome space fire | propers has released from truck during strongen truck besh milling activities lignless range fre which spreads into tank farm | formable ges builds up during isas af okszince power portag, formmable ges builds up during estemated abadown, fro tightes for athewater equipment upon re- tightes for athewater equipment upon re- dart. |
| Asia Distant | | | | | | | | | | | | | |
| Party | | Dome File | Co the second | Dome Fire | Colleges Selected | | Above Ground Fire | Above Ground Fire | Dome Fire | Dome Fire | Dome Fire | Abore Ground Fire | Alone Ground Fine |
| Acenaria 10 | | | externel events and natural | adamat averta and brantan | | at the second se | external events and network phenomene | automatic and and animatic anomana | external events and return | | external events and natural | enternat anternation maternation anternation | |

A-25

Table A-5. Rotary Core Mode Drilling HA Accident Detabase

| Accident Detabese |
|-------------------|
| ₹ |
| Critica |
| Mode |
| 8 |
| R of the |
| Ŷ |
| Tate |

| Keyword | Prezert | Accident Description (Caree or Deviation) | Preventive/Mittigstive Features | Consequences | Recommendations | alian0 | MOLICIC | ADWINES . | 25 | Help-M | 3417-3 |
|----------------|---------|---|---------------------------------|--|---|--------|---------|-----------|----|--------|--------|
| 50.2 | | Remmeble gas builds up during loss of electrical power period, fitammsble gas builds up during adamded shubdown, fine ignitise in drift buck equipment upon re-start | | tepresentative consequences is a HEPA fibri holdup inteses. See "above ground ire" event tree for complete apectrum of ponsequences | review restart procedures to ensure H2 concentration messured plots to re- energisting electrical equipment | ပ ဝ | 10 0 | ≥ | | - | - |
| e Cours | 7 | adhmal brush fino causes sparts that carry Is tank farm, equipment firs results | | lepresentative consequences is a HEPA Mer holdup release. See "above ground te" event tree for complete spectrum of consequences | | ບ 6 | 0 | | | | n |
| | | contaminated equipment stored in non- compatible material (atuminum) containers, chemical reaction generates | | issumed consequence is a short, low oncentration toxic ges (no solid) erronnel exposure. | do not store contaminated equipment in shuminum containers | 0 | 0 | - | n | 10 | in |
| Ground Fire | 70 | contaminated aquipment abreed in non- compatible malerical (aluminum) container, chemistal processes generate sufficient heat to cause weste seri-gention | | LARA Issue | do not store contaminated equipment in aluminum containers | 0 0 | 0 | = | 4 | | 14 |

A.4. ACCIDENT PROGRESSION OF REPRESENTATIVE ACCIDENTS

Even though each accident postulated during the hazard identification process was described as having a unique progression and therefore assigned a single consequence, the actual progression could take several paths and result in different consequences. For example, the HA team postulated that if a fire were to start above a tank at ground level, several accident end-states could be possible. The fire could be confined to the aboveground area and do no damage, could be confined to aboveground but damage HEPA filters containing hazardous material (the HA database consequence assignment), could spread to the tank waste, or could spread to the tank dome space causing a dome fire that may have several possible consequences. In order to show actual accident progression for a limited number of DBAs, event trees were constructed. These trees are presented in Section 4.1.

Even though each event-tree accident end-state can be assigned to a consequence bin, each tree (and therefore each representative accident) was assigned to a representative accident consequence bin. The consequence bin assignment justification is included in Section 4.2. As mentioned earlier, when the HA database was constructed, it was not feasible to include all possible consequences for each accident initiator. Therefore, the team assigned each accident (or initiator) a "representative" consequence corresponding to the outcome considered most likely.

A.4.1. Event Trees

Six of the nine DBAs selected for accident analysis required the construction of event trees. These six are

- 1. Aboveground fire
- 2. Dome-space fire
- 3. Drill-string fire
- 4. Waste fire
- 5. Gas release
- 6. Containment breach (tank loading)

The event trees for each DBA are presented below, along with descriptions of the top events and end-states.

A.4.2. Aboveground Fire

The event tree for an above ground fire is shown in Fig. A-2. A description of the top events and end-states is presented in TABLE A - 6.

| Top Event | Description |
|--|---|
| Aboveground fire occurs. | Fire is either initiated at or spreads to an above-tank area. |
| Flammable-gas concentrates in dome space. | Fuel, oxidizers, and inerting agents exist in sufficient proportions to allow deflagration in the tank dome space. |
| Fire propagates to tank dome space. | The fire spreads from the above-tank area into the tank dome space volume. |
| Tank dome space fire occurs. | A fire (deflagration reaction) is initiated in the dome space volume. |
| Fire propagates to waste. | The (dome space) fire results in waste combustion. |
| Deflagration to detonation occurs in dome space. | Proper confinement conditions and ratios of fuel, oxidizers, and inerting agents exist to allow a deflagration-to-detonation transition in the dome space. |
| HEPAs are burned. | Exhauster HEPA filters are ignited by the aboveground fire. |
| Dome collapses. | A significant portion of the tank dome fails structurally. |
| End-State | Description |
| No release | (Aboveground) Fire does not cause any radioactive or toxic material release. |
| HEPA release | Burning results in release of the exhauster HEPA material burden. |
| Pressure-induced filter blowout | Exhauster HEPA hold-up material is released. |
| Fire-induced dome collapse | Tank dome deflagration-induced collapse results in an aerosol and toxic gas release. |
| Detonation-induced dome collapse | Tank dome detonation-induced collapse results in a pressurized aerosol and toxic gas release. |
| Waste-fire-induced filter blowout | Initial exhauster HEPA burden release followed by release of waste radioactive and toxic combustion products. |
| | 4 |
| Fire-induced dome collapse with waste fire | Initial tank-dome, deflagration-induced-collapse aerosol release is followed by waste radioactive and toxic combustion products release. |

TABLE A - 6ABOVEGROUND FIRE EVENT TREE DESCRIPTIONS

A.4.3. Dome Space Fire

The event tree for a dome-space fire is shown in Fig. A-3. A description of the top events and end-states is presented in TABLE A - 7.

| Top Event | Description | | |
|--|---|--|--|
| Tank dome-space fire occurs. | A fire (deflagration reaction) is initiated in the dome- space volume. | | |
| Fire propagates to waste. | The (dome-space) fire results in waste combustion. | | |
| Deflagration to detonation occurs in dome space. | Proper confinement conditions and ratios of fuel, oxidizers, and inerting agents exist to allow a deflagration-to-detonation transition in the dome space. | | |
| Dome collapses. | A significant portion of the tank dome fails structurally. | | |
| End-State | Description | | |
| Pressure-induced filter blowout | Exhauster HEPA hold-up material released. | | |
| Fire-induced dome collapse | Tank-dome, deflagration-induced collapse results in an aerosol and toxic-gas release. | | |
| Detonation-induced dome collapse | Tank-dome, detonation-induced collapse results in a pressurized aerosol and toxic-gas release. | | |
| Waste-fire-induced filter blowout | Initial exhauster HEPA burden release followed by release of waste combustion products. | | |
| Fire-induced dome collapse with waste fire | Initial tank-dome, deflagration-induced collapse aerosol release followed by waste radioactive and toxic combustion products release. | | |
| Detonation-induced dome collapse with waste fire | Initial tank-dome detonation-induced collapse pressurized-aerosol release is followed by waste radioactive and toxic combustion products release. | | |

TABLE A - 7DOME-FIRE EVENT TREE DESCRIPTIONS

A.4.4. Drill String Fire

The event tree for a drill string fire is shown in Fig. A-4. A description of the top events and end-states is presented in TABLE A - 8.

| Top Event | Description |
|--|---|
| Drill string fire occurs. | A fire (deflagration reaction) is initiated in the drill string volume. |
| Detonation in drill string occurs. | Proper confinement conditions and ratios of fuel, oxidizers, and inerting agents exist to allow a deflagration-to-detonation transition in the drill string. |
| Fire propagates to waste-gas phase. | The drill string fire propagates to a gaseous waste volume. |
| Waste fire occurs. | The waste-gas phase fire spreads to result in waste combustion. |
| Combustibles in dome space occur. | Materials (fuel and oxidizers) that may become involved in a deflagration or detonation reaction materials are present in the dome space. |
| Deflagration to detonation in dome space occurs. | Proper confinement conditions and ratios of fuel, oxidizers, and inerting agents exist to allow a deflagration-to-detonation transition in the dome space. |
| Tank dome space fire occurs. | A fire (deflagration reaction) is initiated in the dome- space volume. |
| Dome collapse occurs. | A significant portion of the tank dome fails structurally. |
| End-State | Description |
| Minimal radioactive release (small containment breach) | Release of radioactive material is present in sampler and drill-string volume. |
| Gas release | Waste toxic gas is released. |
| Waste-fire-induced filter blowout | Initial exhauster HEPA burden is released followed by release of waste combustion products. |
| Pressure-induced filter blowout | Exhauster HEPA hold-up material is released. |
| Fire-induced dome collapse | Tank-dome, deflagration-induced collapse results in an aerosol, and toxic gas is released. |
| Detonation-induced dome collapse | Tank-dome, detonation-induced collapse results in a pressurized aerosol and toxic gas release. |
| Waste-fire-induced filter blowout | Initial exhauster HEPA burden release is followed by release of waste combustion products. |

TABLE A - 8DRILL-STRING FIRE EVENT TREE DESCRIPTIONS

A.4.5. Waste Fire

The event tree for a tank waste fire is shown in Fig. A-5. A description of the top events and end-states is presented in TABLE A - 9.

| Top Event | Description | | | |
|--|---|--|--|--|
| Waste fire occurs. | Fire is initiated in the tank waste, portions of which may be combustible. | | | |
| Combustibles in dome space occur. | Materials (fuel and oxidizers) that may become involved in a deflagration or detonation reaction are present in the dome space. | | | |
| Deflagration to detonation occurs in dome space. | Proper confinement conditions and ratios of fuel, oxidizers, and inerting agents exist to allow a deflagration-to-detonation transition in the dome space. | | | |
| Tank-dome space fire occurs. | A fire (deflagration reaction) is initiated in the dome space volume. | | | |
| Dome collapse occurs. | A significant portion of the tank dome fails structurally. | | | |
| End-State | Description | | | |
| Waste-fire-induced filter blowout | Initial exhauster HEPA burden release is followed by release of waste combustion products. | | | |
| Pressure-induced filter blowout | Exhauster HEPA hold-up material is released. | | | |
| Fire-induced dome collapse | Tank-dome deflagration-induced collapse results in an aerosol and toxic-gas release. | | | |
| Detonation-induced dome collapse | Tank-dome, detonation-induced collapse results in a pressurized aerosol and toxic-gas release. | | | |

TABLE A - 9WASTE FIRE EVENT TREE DESCRIPTIONS

A.4.6. Gas Release

The event tree for a gas release is shown in Fig. A-6. A description of the top events and end-states is presented in TABLE A - 10.

| Top Event | Description |
|---|---|
| Gas release from waste occurs. | Random release of gases normally bound in the waste volume |
| Significant dome pressure occurs. | Dome pressurization is sufficient to result in large volume bypass of HEPA-filtered release path. |
| Significant unfiltered paths occur. | Large unfiltered leakage paths exist. |
| Personnel are in tank vicinity. | Personnel (workers) at risk from gas exposure. |
| Personnel are in breathing suits. | Personnel (workers) in self-contained breathing apparatuses. |
| Rapid area evacuation is carried out. | Personnel (workers) leave quickly and therefore limit their exposure. |
| End-State | Description |
| No exposure | Personnel (workers) do not receive toxic gas exposure. |
| Short, small-volume, low-concentration toxic gas release | Personnel are exposed to short-term, low-concentration toxic gases. |
| Extended, small-volume, low-concentration toxic-gas release | Personnel are exposed to long-term, low-concentration toxic gases. |
| Short, large-volume, low-concentration toxic- gas release | Personnel are exposed to short-term, low-concentration toxic gases. |
| Extended, large-volume, low-concentration toxic-gas release | Personnel are exposed to long-term, low-concentration toxic gases. |
| Short, small-volume, low-concentration toxic- gas and solids release | Personnel are exposed to short-term, low-concentration toxic gas and solids. |
| Extended, small-volume, low-concentration toxic-gas and solids release | Personnel are exposed to long-term, low-concentration toxic gas and solids. |
| Short, large-volume, low-concentration toxic- gas and solids release | Personnel are exposed to short-term, low-concentration toxic gas and solids. |
| Extended, large-volume, low-concentration toxic-gas and solids release | Personnel are exposed to long-term, low-concentration toxic gas and solids. |

TABLE A - 10GAS RELEASE EVENT TREE DESCRIPTIONS

A.4.7. Containment Breach (Tank Loading)

The event tree for tank loading (containment breach) is shown in Fig. A-7. A description of the top events and end-states is presented in TABLE A - 11.

| Top Event | Description |
|---|---|
| External load on tank dome | Equipment load placed over tank dome. |
| Load < design load limit | Load placed atop tank dome exceeds tank-dome structural design limits. |
| General dome collapse | A significant portion of the tank dome fails structurally. |
| Localized dome collapse | The portion of the tank dome below the load fails structurally. |
| Tank-dome space vented | General or localized tank-dome structural failure results in a venting of the tank-dome gases. |
| Flammable-gas concentration in dome space | Fuel, oxidizers, and inerting agents exist in sufficient proportions to allow deflagration in the tank-dome space. |
| Tank-dome-space fire occurs | A fire (deflagration reaction) is initiated in the dome- space volume. |
| Deflagration to detonation in dome space | Proper confinement conditions and ratios of fuel, oxidizers, and inerting agents exist to allow a deflagration-to-detonation transition in the dome space. |
| Waste fire | The tank load or a dome space fire initiates waste combustion. |
| Dome collapse | A significant portion of the tank dome fails structurally. |

 TABLE A - 11

 TANK LOADING (CONTAINMENT BREACH) EVENT TREE DESCRIPTIONS

TABLE A - 11 (cont.)TANK LOADING (CONTAINMENT BREACH) EVENT TREE DESCRIPTIONS

| End-State | Description |
|---|---|
| No release | No radioactive or toxic material exposure. |
| Dome breach, extended, large-volume, low- concentration, toxic-gas release | Personnel are exposed to long-term, low-concentration toxic gases. |
| Dome breach, extended, small-volume, low- concentration, toxic-gas release | Personnel are exposed to long-term, low-concentration toxic gases. |
| Dome breach with waste fire | A waste fire is initiated following a dome breach; waste combustion products are released. |
| Dome breach with dome fire | A dome-space fire is initiated following a dome breach. |
| Fire-induced dome collapse | Tank-dome, deflagration-induced collapse results in an aerosol and toxic gas release. |
| Dome breach with dome and waste fire | A dome-space fire and a waste fire are initiated following a dome breach. |
| Fire-induced dome collapse with waste fire | Tank-dome, deflagration-induced collapse results in aerosol, toxic-gas, and waste-combustion product releases. |
| Detonation-induced dome collapse | Tank-dome, detonation-induced collapse results in a pressurized-aerosol and toxic-gas release. |
| Detonation-induced dome collapse with waste fire | Tank-dome, detonation-induced collapse results in a pressurized-aerosol and toxic-gas as well as a waste- combustion products release. |
| Dome collapse | Tank-dome collapse results in an aerosol and toxic-gas release. |
| Dome collapse with waste fire | A waste fire is initiated following a dome collapse; waste-combustion products are released. |
| Dome collapse with dome fire | A dome-space fire is initiated following a dome collapse. |
| Dome collapse with dome and waste fire | A dome-space fire and a waste fire are initiated following a dome breach. |
| Dome collapse with detonation | Pressurized aerosol and toxic gas are released. |
| Dome collapse with detonation and waste fire | Pressurized aerosol, toxic gas, and waste-combustion products are released. |

A.4.8. Justification of Consequence Assignments

The consequence assignments in the hazard identification process were made using the following steps:

- 1. For each accident in the database, the end-state was either postulated or assigned to the appropriate accident-progression event tree.
- 2. As was noted earlier for each accident-progression event tree, a representative accident end-state was selected for each event tree. Engineering judgment was used to pick the appropriate end-state, and it is noted that for each event tree, the range of end-state consequences varies greatly. However, an attempt was made to find the most likely end-state, given that no calculation of frequency was attempted.
- 3. A few members of the Hazards Identification team met and examined previous consequence analysis for the Hanford waste tanks. The team used engineering judgment to assign consequence bin assignments to each accident end-state.
- 4. A table of accident end-states (which is also the same as the keywords assigned to each accident) was constructed. The table relates the accident end-state to the appropriate letter identifier for each accident-consequence bin as applied to each accident receptor.

A.4.9. Assignment of Accident End-States to Consequence Bins

The accident end-states (which also correspond to accident keywords) are assigned to consequence bins using the previously discussed data and assumptions. TABLE A - 12 shows the accident consequence assignments used in the HA database; TABLE A-2 provides the quantitative doses associated with the consequence bin letter identifiers.

TABLE A - 12 CONSEQUENCE ASS: GNMENTS FOR HAZARD IDENTIFICATION DATABASE

| | | On-site | Facility | |
|--|--------|---------|----------|-------------|
| Accident End-State | Public | Worker | Worker | Environment |
| Dome collapse | В | А | А | В |
| Dome collapse w/ fire | В | A | A | В |
| Dome collapse w/ waste fire | A | А | A | A |
| Filter blowout (SST passive ventilation) | D | С | С | С |
| Aboveground fire | D | С | C | C |
| Drill-string fire | D | D | С | D |
| Waste fi r e | D | С | С | С |
| Subterranean spill | D | D | D | С |
| Gas release (filtered) | D | D | С | D |
| Aboveground spill | D | D | D | D |
| Auxiliary equipment fire | D | D | D | D |

August 8, 1996

Figure A.2 Above Ground Fire Event Tree

| Above ground fire | Flammable gas concentration in dome space | Fire propogates to tank dome space | Tank dome space fire occura | Fire propogales to waste | Deflagration to detonation in dome space | HEPAs burned | Dome collapses | Path | Outcome | Description |
|-------------------|---|------------------------------------|--------------------------------|-----------------------------|--|---------------|------------------|------|----------|---|
| GNDFIRE | FLAME | PAPDOME | DOMEFRE | DOMEWAS | DOT | HEPAS | FRECOLLP | 1 | | |
| | | | | no waste fire | not possible | no HEPAS burn | not possible | ۱ | | no release |
| | gas < LFL | NA | not possible | | | HEPAs burn | not possible | 2 | | HEPA release |
| | | | l | waste fire | | | | 3 | transfer | to waste fire event tree |
| | | | | | | no HEPAS bum | not possible | 4 | | no release |
| | | no propogation | not possible | | not possione | HEPAs burn | nat possible | 5 | | HEPA release |
| | | | Į | waste fire | | | | | transfer | to waste fire event tree |
| | | | | | | na HEPAS bum | not possible | 7 | | no release |
| | gas > LFL | | ne dome fire | no waste fire | not possible | HEPAs burn | not possible | | | HEPA release |
| | | | | waste fire | ···· | | | 9 | transfer | to waste fire event tree |
| | | | | | | | no dome collapse | 10 | | pressure induced filter blow |
| | | fire propogates | | no waste fire | | <u>NA</u> | dome collapse | 11 | | fire induced dome collpase |
| | | | | | 100_ | NA | bemuees | 12 | | detonation induced dome collapse |
| | | | dome fire | | | | na dome collapse | 13 | | waste fire induced filter blow |
| | | | | wanto fire | | <u>NA</u> | dome collapse | 14 | | ire induced dome collapse v waste fire |
| | | | | | DOT | NA | assumed | 15 | transfer | detonation induced dome collapse with waste fire |

Constant Social Sealer Providence Me Constant Sealer Social Second Sealer Social Second Secon

Figure A.3 Dome Space Fire Event Tree

| Tank dome space fire | nk dome space Fire propoagates to fire waste de | | Defiagration to Dome collapses detonation in dome space | | Outcome | Description |
|-------------------------|---|--------|---|---|---------|---|
| TANKFRE | DOMEWAS | DDT | COLLAPSE | | | |
| L | - | no DDT | no dome collapse | 1 | | pressure induced filter blowout |
| | no waste fire waste fire | | dome collapse | 2 | | fire induced dome collapse |
| | | DDT | assumed | 3 | | detonation induced dome collapse |
| tank dome fire | | | no dome collapse | 4 | | waste fire Induced filter blowout |
| | | | dome collapse | 5 | | fire induced dome collapse with waste fire |
| | | DDT | assumed | 6 | | detonation induced dome collapse with waste fire |

di seria ang

| Tree |
|--------|
| Event |
| Fire |
| String |
| Drill |
| A.4 |
| Figure |



Figure A.5 Waste Fire Event Tree

| Waste fire | Combustibles in dome space | deflagration to detonation in dome space | Tank dome space fire occurs | Dome collapses | Path | Outcome | Description |
|--|----------------------------|--|--------------------------------|------------------|------|---------|--------------------------------------|
| WASFIRE | LOADS | DDT | DOMEFRE | FRECOLLP | | | |
| | none released | not possible | not possible | not possible | 1 | | waste fire induced filter blowout |
| | | | | not possible | 2 | | waste fire induced filter blowout |
| | not detonatable | not possible | dome fire | no dome collapse | 3 | | pressure induced filter blowout |
| ······································ | 4 | | | dome collapse | 4 | | fire Induced dome collapse |
| | | | no dome fire | not possible | 5 | | waste fire induced filter blowout |
| | | no DDT | dome fire | no dome cóllapse | 6 | | pressure induced filter blowout |
| | detonatable | | | dome collapse | 7 | | fire induced dome collapse |
| | | DDT | NA | assumed | 8 | | detonation induced dome collapse |

Figure A.6 Gas Release Event Tree

| Gas Release from waste | Significant dome pressure increase | Significant unfiltered paths to atmosphere | Personnel in tank vicinity | Personnel in breathing apparatus | Rapid area evacuation | Path | Outcome | Description |
|---------------------------|---------------------------------------|--|-------------------------------|--|--------------------------|------|---------|---|
| GASREL | DOMEPRESS | PATHS | PERSON | SUITS | LEAVE | | | |
| L | | · · | ۲۲ | yes | NA | - 1 | | no exposure |
| | | no paths | assumed | 00 | yes | - 2 | | short, small volume, low concentration toxic gas release |
| | no increase | | L | | no | - 3 | | extended, small volume, low concentration toxic gas release |
| | | | г | yes | NA | - 4 | | no exposure |
| | | paths exist | assumed | 20 | yes | - 5 | | short, large volume, low concentration toxic gas release |
| | | | L | | no | 6 | | extended, large volume, low concentration toxic gas releas |
| | | | Г | yes | NA | 7 | | no exposure |
| | | no paths | assumed | no | yes | - 8 | | short, small volume, low concentration toxic gas and solids exposure |
| | increase | | Ľ | | no | 9 | | extended, small volume, low concentration toxic gas and solids exposure |
| | | | Г | yes | NA | - 10 | | no exposure |
| | | paths exist | assumed | - | yes | - 11 | | short, large volume, low concentration toxic gas and solids exposure |
| | | | L | | no | - 12 | | extended, large volume, low concentration toxic gas and solids exposure |

Figure A.7 Tank Loading Event Tree



WHC-SD-WM-SAD-035, Rev. 0-a

August 8, 1996

A-42

LINE CONTRACTOR

Figure A.7 Tank Loading Event Tree



A-43

0 21 1 0 0 100 0 211 (0,1)

A.5. REFERENCES

- 1. DOE Standarc Preparation Guide for U. S. Department of Energy Nonreactor Nuclear Facility Safety Analysis reports DOE-STD-3009-94 (July 1994).
- 2. Center for Process Safety of the American Institute-of Chemical Engineers, Guidelines for Hazard Evaluation Procedures, Second Edition with Worked Examples (New York, New York, 1992).

APPENDIX B

EQUIPMENT RATING AND AUTOMATIC SHUT-DOWN REQUIREMENTS

B.1. OBJECTIVE

This appendix provides the basis for the requirements imposed on equipment capable of generating sparks in a flammable-gas (FG) environment. Also included in this appendix are the trip set points necessary to shut down the equipment that is not designed to operate in an FG environment.

B.2. BACKGROUND

The gas-retention and release mechanisms in the waste tanks are not fully understood. The understanding is even poorer for the single-shell tanks (SSTs). Although semi-quantitative estimates are available for SST gas-release mechanisms,¹ in the absence of detailed data, it is difficult to argue that such analyses provide bounding estimates. The gas composition and maximum allowable gas-release estimates provided in Appendix C show that, in some of the SSTs, small releases may result in a flammability hazard. Our current understanding of the gas-release phenomena in the SSTs is limited and does not allow us to develop reliable administrative controls to limit or control the gas releases during the proposed intrusive activities. Consequently, the major emphasis in this safety assessment (SA) is on managing the spark sources when flammable gases (FGs) are present. Furthermore, a burn even with a small amount of FGs in the tank dome space is likely to result in dome failure (Appendix C), which may result in unacceptable radiological and toxicological consequences (see Section 5 of this SA) if the failure is catastrophic.

As with gas releases, gas-storage mechanisms are poorly understood. Waste gases may be stored in the waste in the form of elongated/dentritic bubbles that may be connected through the pores. Demonstrating that an ignition in the waste will not propagate also is difficult without a detailed knowledge of pore sizes, heat dissipation mechanisms, and the moisture content in the pores. The pressure resulting from a burn in the waste, the subsequent damage to the tank, and the amount of material release is difficult to bound. Consequently, this SA is aimed at preventing ignition inside the waste as opposed to demonstrating that propagation is impossible.

Under these conditions, as a general rule, design restrictions are imposed so that no single failure should lead to sparking conditions. A minimum of double-engineering protection against a sparking condition is required. These requirements are developed assuming that the likelihood of FG presence during the activity is high.

B.3. SAFETY RATING REQUIREMENTS

During the proposed intrusive activities, the following regions must be carefully protected from possible spark sources:

- Tank dome space,
- Regions below the waste surface (waste volume),
- Confined regions with possible flow paths to waste or waste gas in-leakage and accumulation,
- Pump pits (if not sealed from the dome),
- A specified region outside an open riser or a leak path,
- Tanks adjacent and connected to the subject tank, and
- Other equipment connected to the dome space.

B.3.1. Electrical Equipment

Before beginning the discussion of safety requirements for the electric equipment, it is important to note that the intent of this section is not to develop a National Fire Protection Association (NFPA) classification for the tank farms. NFPA classification requirements are used merely as guidance in achieving the desired very low frequencies for the burn accidents.

During FG/RMCS operations, all electrical equipment located in these regions must be rated to operate in a Class-I, Division-1 (Div.-1) Group-B environment according to NFPA classification. Recognizing the fact that the FGs will not continuously exist in these regions, Class-I, Div.-2, Group-B-rated equipment also may be operated in these regions, provided that they are equipped with a reliable automatic shut-down system. It requires an unlikely sequence of multiple failures and malfunctions for Class-I, Div.-1, Group-B-rated equipment to cause a spark.

Class-I, Div.-2, Group-B equipment is capable of sparking upon single failure. Consequently, such equipment must be protected by a reliable shut-down circuit. Furthermore, the background concentrations of FGs must be measured and shown to be less than 25% of the lower flammability limit (LFL) before energizing this equipment. In order to cause a spark, a sequence of double independent failures are necessary (the shut-down circuit must fail, and the Class-I, Div.-2, Group-B equipment must also fail simultaneously). If Class-I, Div.-2, Group-B equipment is used in the above regions, the reliability of the shut-down circuit must be evaluated and must be shown that it will perform with a desired reliability. All equipment that must continue to operate upon detection of FGs must be designed to Class-I, Div.-1, Group-B requirements. Examples of such equipment are the ventilation fan needed to discharge the FGs and FG monitoring equipment.

Regions immediately adjacent to the Class-I, Div.-1, Group-B regions discussed above are automatically classified as Class-I, Div.-2, Group-B spaces, and the equipment must meet the requirements of this classification.

B.3.2. Mechanical Systems

All moving mechanical systems capable of generating frictional forces must be assumed to be sparking, unless bounding analyses or experiments show otherwise. Without specific experiments, it is very difficult to demonstrate that such mechanical systems will not generate sparks capable of igniting the flammable mixtures of interest. Consequently, such mechanical systems also must have double protection against either sparking or being exposed to a flammable atmosphere. The necessary protective measures could be a combination of the following systems:

- Spark-resistant and electrically conductive material(s) (to protect against static electricity buildup) must be used in the regions specified above.
- Mechanical devices must be protected by a seal that prevents waste or FG in-leakage.
- The region around the mechanical device must be protected by an adequate purge. The adequacy of the purge system must be demonstrated by analysis.
- Upon loss of purge flow, the system must be quickly deenergized.
- Upon detection of FGs, the system must be quickly deenergized before the gas mixture reaches the LFL.

As a minimum, two of the above methods must be used to protect the mechanical systems capable of sparking. Depending on the reliability required by accident analysis, more than two systems may be necessary.

B.3.3. Equipment over the Tank Dome

Using an algebraic momentum jet model,² FG concentration as a fraction of the source concentration may be obtained from the following equation:^{2,3}

$$\frac{C_{c}}{C_{o}} = \frac{(\lambda^{2} + 1)}{\lambda^{2}} \frac{0.25}{\left[1.414\alpha \frac{(s - s_{o})}{D} + 0.5\right]} \quad \text{for } s \ge s_{o},$$
(B-1)

where C_c is the centerline concentration, C_o is the source concentration, s_o is the jet length in the establishment zone ($s_o = 6.2 \times D$), s is the distance to the source, D is the source diameter, λ is the spreading parameter ($\lambda = 1.16$ taken from Ref. 2) and α is the ϵ -npirical entrainment parameter.

Setting the source concentration to 8% and using an entrainment parameter $\alpha = 0.057$, the jet centerline concentration drops to 2.4% at a distance of 18 diameters. Using a Gaussian concentration profile,² the concentration at the axial distance of 18 D drops to <0.6% at a radial distance of 2.2 x D from the jet centerline.³ The centerline concentration drops to 1.2% at a distance of 36 x D away from the source.³

As discussed in Appendix C, 2.4% H₂ corresponds to the LFL of gas mixtures that are very rich (70%) in ammonia and <30% in hydrogen. For these gas mixtures, prompt gas releases >283 m³ (10,000 ft³) are required to exceed the 8% H₂ concentration at the source. If we postulate pure hydrogen releases, only 113 m³ (4000 ft³) of prompt release may result in 8% H₂ in the source. However, the LFL of hydrogen is 4%. A hydrogen concentration of 4% at 18 diameters away from the source corresponds to 13.3% H₂ at the source, which requires a prompt release >190 m³ (6700 ft³). These calculations are based on a minimum dome volume of 1400 m³ (50,000 ft³). Therefore, it is believed that an 8% H₂ concentration at the source combined with 2.4% H₂ because the LFL provides a conservative bound. Based on the model discussed in Appendix L, the frequency of exceeding 8% H₂in the dome would be ~10⁻³ per intrusion.

Entrainment Parameter. The use of $\alpha = 0.057$ is conservative because this value is the lowest magnitude given in Ref. 2. Using the linear mixing law model and the momentum jet solution provided in Schlichting,⁴ α is obtained as 0.087. As shown in Eq. (B-1), larger α results in enhanced dilution within a given distance. Furthermore, this analysis does not account for atmospheric turbulence that would further enhance the entrainment.

Probability of Exceeding 2.4% H₂. If we assume a random spark source in the hemispherical region around an open riser, the probability of exceeding 2.4% H₂at the spark location is computed in Ref. 3 and shown in Fig. B-1. As shown in this figure, the probability of exceeding 2.4% H₂ drops below 10^{-3} at distances $\geq 16 \times D$.



Fig. B-1. Probability of exceeding 2.4% H₂.

The jet direction is determined by the direction of the open area, initial momentum, buoyancy, and wind direction. In general, the open riser will result in a vertical jet unless the direction is affected by the wind. Thus, the above probabilities may be reduced further by not placing the equipment directly above an open riser and in the prevailing wind direction. The jet coming out of an opening typically is buoyant because

- The dome space is warmer than the ambient, and
- The waste gases are lighter than air if they are predominantly flammable (hydrogen, ammonia, and methane).

Thus, to reduce the probability of the equipment being exposed to a flammable atmosphere, the equipment should be placed near ground level and in the opposite direction to the prevailing wind direction. Furthermore, any opening directly aimed at the equipment location must be sealed, or the equipment must be protected from direct jet impingement.

Based on these conservative calculations, up to a distance of 18 diameters away from the source, the equipment must be rated for operation in a Class-I, Div.-1, Group-B environment. Equipment rated for Class-I, Div.-2, Group-B may also be used, provided it is automatically deenergized upon detection of FGs in the riser.

At distances from $18 \times D$ to $36 \times D$, the equipment must be rated for operations in Class-I, Div.-2, Group-B environments and must be deenergized manually upon detection of FGs in the dome space.

If the above classification conditions cannot be met, the open riser must be equipped with a flame arrester to prevent flame propagation into the dome, provided that the consequences of an external fire are shown to be accoptable.

B.4. AUTOMATIC SHUT-DOWN REQUIREMENTS

If an automatic shutdown of equipment upon detection of FGs is relied upon, the shut-down system

- must be reliable,
- must respond to rapid transient surges, and
- must be located near the equipment that will be shut down.

The reliability of the protective systems (shut-down systems) is discussed in Appendix D of this SA.

To protect against spatial maldistribution of gases in the dome space, the gas samples must be taken from the region where the target equipment is located. During operations with an active ventilation system, the sampling in the vent lines is adequate, provided that the trip points are conservatively set.

The FG concentrations in the tank dome space are typically very low during the steady-state periods. Only during a gas-release event (GRE) does the FG concentration start to increase. If the gas release rates are high, the FG concentrations may increase very rapidly. Slow detectors may not be able to perform their shut-down function adequately during fast GREs. In this SA, two detection mechanisms are considered,

- Trip on high hydrogen concentration or on a high rate of hydrogen increase, and
- Trip on high dome pressure.

B.4.1. Hydrogen Trip

The adequacy of a hydrogen detector time constant is assessed using the Whittaker cell analysis. The standard hydrogen monitoring systems (SHMS) that are used in the tank farms contain Whittaker hydrogen detection cells. These cells measure 90% of a step change in <2 min.⁵ If we assume a first-order relaxation model that is adequate for diffusion-based devices, the time constant of the cell is obtained as 52 s (Ref. 6). Also using a first-order relaxation model and a constant rate of increase in hydrogen concentration expressed as

$$X_{a}(t) = \frac{(X_{s} - X_{o})}{T}t + X_{o}$$
, (B-2)

where X_a is the actual hydrogen concentration, X_o is the background (initial) concentration, X_s is the target concentration, T is the time necessary for the actual concentration to reach the target concentration, and t is the time. For a linear increase in the actual hydrogen concentration, the <u>measured</u> hydrogen concentration (X) may be obtained as⁵

$$X(t) = \begin{cases} (X_{s} - X_{o})\frac{\tau}{T} \left(e^{-(t - t_{d})/\tau} + \frac{(t - t_{d})}{\tau} - 1 \right) + X_{o} & t \ge t_{d} \\ X_{o} & t < t_{d} \end{cases}$$
(B-3)

where t_d is the delay time (time that it takes to transport the dome gases to the instrument). Figure B-2 shows the necessary trip set points to deenergize the equipment before the dome concentration reaches 2.4% H₂. The following assumptions are used in obtaining the results shown in Fig. B-2:

- Initial background concentration (X₀) is 1000 ppm,
- After the trip is initiated, it takes 10 s to deenergize the circuits of interest totally,
- The monitor provides one data point per ~5 s, and a minimum of 10 s worth of data is needed for a rate trip,
- X_s is set to 2.4%,
- Dome volume is equal to 1416 m^3 (50,000 ft³), and
- 90% of the waste gas is hydrogen.

Based on Fig. B-2, by setting the trip point at 5000 ppm H₂, the equipment can be deenergized on time if the release rate into the dome is $\leq 0.47 \text{ m}^3/\text{s}$ (1000 ft³/min).

If a rate trip set at 100 ppm/s is used, such a trip will provide protection for release rates between 0.14 and 1 m³/s (300 and 2200 ft³/min). Thus, in the range between 0.14 and 0.71 m³/s (300 and 1500 ft³/min), both the concentration trip and the rate trip will deenergize the equipment before reaching the target concentrations. A rate trip of 100/s corresponds to a concentration increase of 1000 ppm over the 10 s averaging time. Setting the rate trip to lower values may result in premature and unnecessary shutdown when there are small fluctuations in the instrument readings.



Fig. B-2. Necessary trip points to shut down the equipment before the concentration reaches 2.4% H₂ during constant rate of concentration increase.

A trip based on hydrogen concentration is not adequate in protecting against gasrelease rates $\geq 1 \text{ m}^3/\text{s}$ (2200 ft³/min). Note that $1 \text{ m}^3/\text{s}$ (2200 ft³/min) is the conservative gas-release rate obtained for 101-SY releases during the first 200 s of a rollover. Based on the analysis provided in Ref. 1, the best-estimate release rates for the most likely fast-release scenario in SSTs is 0.006 m³/s (12 ft³/min).

B.4.2. Pressure Trip

No specific analysis is performed for the pressure traces during a large gas release. The experience in 101-SY demonstrated that for release rates as low as $0.19 \text{ m}^3/\text{s}(400 \text{ ft}^3/\text{min})$, the dome space experiences pressure pulse of 50.8 mm (2 in.) w.g. or greater. The pressure pulse is a function of the dome volume, relief area, and the pressure drop through the relief area. The relief area for Tank 101-SY is 0.145 m² (226 ft²).

A pressure pulse of 50.8 mm (2 in.) w.g. would correspond to a 0.5% increase in the dome pressure. If we use a dome volume of 1416 m³ (50,000 ft³), which is the ideal gas law and adiabatic compression, a 5.1-m^3 (180-ft³) sudden release into the dome is sufficient to generate a 2-in. w.g. pressure pulse. Thus, large release rates $\geq 0.47 \text{ m}^3/\text{s}$ (1000 ft³/min), a pressure switch set at 50.8 mm (2 in. w.g.) above the background pressure will detect the gas release and deenergize the equipment.

B.4.3. Automatic Shut-Down Trips

For deenergizing the equipment that does not meet Class-I, Div.-1, Group-B requirements, the automatic shut-down trip may be set by using the hydrogen detectors and a dome pressure gauge. The hydrogen detector switch must be set to ≤ 5000 ppm, and the pressure gauge trip must be set to 50.8 mm (2 in. w.g.) w.g. above the background pressure. The trip for the rate of increase must be set to 100 ppm/s (averaged over 10 s).

If FGMs are used instead of a hydrogen detector, upon availability of the calibration curves, the FGM should be set to provide an equivalent trip to 5000 ppm H_2 and a 100 ppm/s rate of increase in the hydrogen. Also, it must be confirmed that the response time of the FGM is bounded by the assumptions listed above.

B.5. REFERENCES

- 1. R. T. Allemann, "A Discussion of Some Mechanisms for Sudden Gas Release from Single-Shell Waste Tanks," Pacific Northwest Laboratory letter report PNL-WTS-101095 (October 1995).
- 2. B. Gebhart, D. S. Hilder, and M. Kelleher, "The Diffusion of Turbulent Buoyant Jets," Advances in Heat Transfer 16 (Academic Press, Inc., 1984).
- K. O. Pasamehmetoglu, "Flammable Gas Concentration Outside an Open Riser," Los Alamos National Laboratory Calc-Note TSA10-CN-WT-SA-GR-044 (December 1995).
- 4. H. Schlichting, Boundary Layer Theory, 7th Edition (McGraw-Hill, New York, 1979).
- 5. T. C. Schneider, "Wittaker Electro-Chemical Cell H2 Monitoring System Accuracy," Westinghouse Hanford Company memorandum to G. F. Vargo (January 8, 1993).
- 6. K.O. Pasamehmetoglu, "Automatic Shutdown Requirements," Los Alamos National Laboratory Calc-Note TSA10-CN-WT-SA-GR-045 (November 1995).

APPENDIX C

BOUNDING GAS COMPOSITION, LEAN FLAMMABILITY LIMIT, AND ALLOWABLE GAS RELEASE AND BURN VOLUMES

C.1. OBJECTIVE

This appendix provides a bounding estimate for the gas composition of the waste gas that might be released during intrusive operations in single-shell tanks (SSTs) that are on the Flammable Gas Watch List (FGWL). This appendix also evaluates the bounding value for the lower flammability limit (LFL) for the estimated gas compositions. A brief summary of the maximum allowable gas-release volumes into the dome space and the maximum allowable burn volume in the waste are discussed in this appendix. Finally, the current estimates for the gas inventory in the tanks are summarized and compared to the maximum allowable releases.

C.2. SST DOME VOLUMES

The SST types typically are characterized by their waste storage capacity. The three types of interest and the associated tanks are summarized in Table C-1.

Figure C-1 shows a schematic of the different types of SSTs in which the characteristic dimensions taken from Ref. 1 are given in Table C-1. The dome vapor space volumes are obtained assuming an ellipsoidal dome, as shown in Fig. C-1, where H is the dome apex height, W is the height of the cylindrical tank, R is the tank radius, and L is the waste level. Using the approximate waste levels obtained from Refs. 2 and 3, the dome volumes (V) are calculated⁴ and reported in Table C-2. Also, the maximum drop height (Z) used during a dome collapse accident and the waste volume (V_w) are shown in Table C-2

C.3. GAS COMPOSITION

There is no direct measurement for the waste gas composition for any of the tanks. The waste gas composition must be inferred from dome space concentration data. Unfortunately, there are very limited data regarding concentration in the dome space of the SSTs. The SSTs that are on the FGWL currently are being instrumented with a standard hydrogen monitoring system (SHMS) for continuous hydrogen measurements. As a means of baselining the SHMS, vapor grab samples from the FGWL SSTs were taken during the summer of 1995, and the samples were analyzed using mass spectroscopy. The resulting data are reported in Ref. 5. Only the noncondensable gases are identified in the grab samples.

TABLE C-1

CAPACITY AND CHARACTERISTIC DIMENSIONS OF THE SSTS OF INTEREST

| TYPE | CAPACITY (gal.) | TANKS | R (ft) | W (ft) | H (ft) |
|------|-----------------|---|--------|--------|--------|
| 1 | 1,000,000 | A-101, AX-101, AX-103, SX- 101, SX-102, SX-103, SX-104, SX-105, SX-106, and SX-109 | 37.5 | 32.5_ | 12.0 |
| 2 | 758,000 | S-102, S-111, and S-112 | 37.5 | 22.5 | 14.0 |
| 3 | 530,000 | T-110, U-103, U-105, U-107, U-108, and U-109 | 37.5 | 19.2 | 9.5 |





| TANK | L (in.) | V (ft ³) | V _w (ft ³) | Z (ft) |
|--------|---------|----------------------|-----------------------------------|--------|
| A-101 | 345 | 51,910 | 127,010 | 15.8 |
| AX-101 | 278 | 76,570 | 102,350 | 21.3 |
| AX-103 | 42 | 163,460 | 15,640 | 41.0 |
| SX-101 | 172 | 115,600 | 63,320 | 30.2 |
| SX-102 | 205 | 103,450 | 75,470 | 27.4 |
| SX-103 | 245 | 88,720 | 90,200 | 24.1 |
| SX-104 | 237 | 91,570 | 87,250 | 24.8 |
| SX-105 | 257 | 84,310 | 94,620 | 23.1 |
| SX-106 | 207 | 102,710 | 76,210 | 27.3 |
| SX-109 | 95 | 143,950 | 34,980 | 36.6 |
| S-102 | 205 | 65,160 | 75,470 | 19.4 |
| S-111 | 205 | 65,160 | 75,470 | 19.4 |
| S-112 | 197 | 68,110 | 72,530 | 20.1 |
| T-110 | 148 | 57,600 | 54,490 | 16.3 |
| U-103 | 167 | 50,600 | 61,480 | 14.8 |
| U-105 | 150 | 56,860 | 55,220 | 16.2 |
| U-107 | 168 | 50,240 | 61,850 | 14.7 |
| U-108 | 165 | 51,340 | 60,750 | 14.9 |
| U-109 | 145 | 58,700 | 53,380 | 16.6 |

TABLE C-2DOME VOLUMES AND IMPORTANT DIMENSIONS

C.3.1. Noncondensable Gas Composition

The sample analysis provided dome concentrations for hydrogen, methane, and nitrous oxide. The available data are converted to waste-gas composition, assuming that these three species make up the noncondensable portion of the waste gas. The following model is used in converting the measured concentrations into a waste-gas composition estimate:⁶

$$\frac{X_{nc,hydrogen} + X_{nc,methane} + X_{nc,nitrous} = 1}{X_{nc,hydrogen}} = \text{constant} \text{ and } \frac{X_{nc,methane}}{X_{nc,hydrogen}} = \text{constant} \text{ or } \frac{X_{nc,methane}}{X_{nc,nitrous}} = \text{constant}$$

(C-2)

August 8, 1996

C-3

This model is based on the assumption that the species ratios obtained from the dome space measurements are the same for the gas bubbles that exist in the waste. Thus, it is assumed that

- The mass transfer (including the molecular diffusion out of the waste) from the waste surface is negligible, and
- The species ratios are established by equilibrium in the waste prior to initiation of GRE.

The noncondensable species of interest are not very soluble in the waste. Consequently, the mass-transfer effects are expected to be negligible. Likewise, provided that all the species are generated at constant proportions independent of the location, the surface diffusion effects also are expected to be small. As discussed later the amount of methane in all the tanks analyzed is small. Thus, the uncertainties associated with the above assumptions for methane releases are not expected to influence the final conclusions. Continuous monitoring data for nitrous oxide and hydrogen concentrations during and between large gas-release events are available for Tank 101-SY. The data show that during steady-state releases, the nitrous oxide concentration is slightly above the hydrogen concentration (nitrous-oxide-to-hydrogen ratio is >1). During gas-release events (GREs), the hydrogen-to-nitrous-oxide ratio is slightly less than 1 (see Ref. 7). Moderate solubility of nitrous oxide, the response time, the accuracy of different instruments used for monitoring hydrogen and nitrous oxide, and variations in the waste temperature near the surface may be responsible for small differences. As long as the results of the present model are interpreted conservatively, such small differences are not expected to influence the conclusions.

In determining the bounding gas composition in which the fuel is maximized, $X_{nc,hydrogen}/X_{nc,nitrous}$ and $X_{nc,methane}/X_{nc,hydrogen}$ ratios are used. In cases where the nitrous oxide is maximized for toxicological release analysis, $X_{nc,hydrogen}/X_{nc,nitrous}$ and $X_{nc,methane}/X_{nc,nitrous}$ ratios are used.

In summary, for each grab sample obtained from a given tank, the ratios shown in Eq. (C-2) are computed. The mean values and the standard deviations for these ratios are obtained using all the grab samples' data from a given tank. To obtain the bounding gas composition with the maximum fuel content, $X_{nc,hydrogen}/X_{nc,nitrous}$ and $X_{nc,methane}/X_{nc,hydrogen}$ ratios are obtained as the mean plus 2 standard deviations (for tanks where there are greater than two samples). For tanks with only two samples, the maximum computed ratio is used.

The same methodology is followed in obtaining the gas composition with maximum nitrous oxide content. For this case, mean plus 2 standard deviations are used for $X_{nc,nutrous}/X_{nc,hydrogen}$ and $X_{nc,nitrous}/X_{nc,methane}$ ratios for tanks with more than two grab samples. The maximum of the two samples is used for tanks with only two grab samples.

In the grab samples, the minimum detection limit appears to be 1 to 2 ppm. Thus, analyzing samples with very low waste gas concentrations results in a bias toward the minor species. Consequently, only those samples that contain hydrogen concentrations that are \geq 30 ppm are analyzed. The results are shown in Table C-3.

As shown in Table C-3, in general, the tanks in the A, AX, S and SX Farms appear to be rich in fuel (hydrogen and methane), with the exceptions of AX=103 and S-102, which show ~40% N₂O in the release gas. The maximum hydrogen concentration is estimated as 89.2% for Tank SX-106. The maximum methane concentration is estimated for Tank AX-103 as 7.2%. However, the grab sample data for this tank yielded low concentrations in the dome space and are believed to be biased toward minor species (methane). The maximum methane concentration measured in the dome of AX-103 is 3%. Furthermore, the waste level in this tank is only 1.07 m, and the flammable-gas (FG) release in this tank with a large dome volume is not expected to be a major problem.

The tanks in the U Farm appear to be richer in nitrous oxide. The upper-bound estimate for the nitrous oxide is obtained as 75.4% in Tank U-105. This value will be used to compute the toxicological consequences of nitrous oxide releases.

Justifying the use of tank-specific gas compositions based on these analyses is difficult. The difficulty is primarily in extrapolating steady-state gas concentration measurements to predict the gas concentration during a prompt release event and stems from the modeling assumption discussed above. Conservatively, the gas ratios used in obtaining the values in Table C-3 are evaluated at 2 standard deviations in which either fuel or nitrous oxide is maximized and the existence of nitrogen (that is known to be generated in the waste) is ignored. However, the concentrations shown in Table C-3 are obtained using a very limited database. Furthermore, the SSTs within a given farm are connected through overflow lines, and most SSTs are passively ventilated. Consequently, it is possible that the dome gases in connected tanks intermix, and the grab sample data may not necessarily represent the waste gas that is released from a specific waste tank. Thus, we must be cautious in using the computed gas compositions on a tank-by-tank basis.

To determine the bounding case, the concept of equivalent fuel⁶ is used for the compositions shown in Table C-3. Table C-3 lists the equivalent fuel in terms of hydrogen concentration in the noncondensable waste gas. The equivalent fuel⁶ is obtained as

$$EF = R_m \times X_{methane} + X_{hydrogen}$$
,

(C-3)

where R_m is the ratio of combustion energy of methane (798 kJ/mole) to the combustion energy of hydrogen (241 kJ/mole). Thus, R_m is obtained as 3.32.
TABLE C-3 HYDROGEN, NITROUS OXIDE, AND METHANE FRACTION IN THE NONCONDENSABLE RELEASE GAS

| | MAXIMUM FUEL | | | | MAXIMUM N2O |
|--------|--|----------------------|----------|--------------------|----------------------|
| | X _{hydrogen} | X _{methane} | Xnitrous | EF | X _{nitrous} |
| TANK | (%) | (%) | (%) | (%H ₂) | (%) |
| A-101 | 86.7 | 1.3 | 12.0 | 90.9 | 15.4 |
| AX-101 | 88.1 | 4.5 | 7.4 | 103.0 | 11.3 |
| AX-103 | 55.1 | 7.2 | 37.7 | 79.1 | 49.6 |
| SX-101 | | | | | |
| SX-102 | 81.4 | 4.6 | 14.0 | 96.8 | 15.4 |
| SX-103 | 79.7 | 3.8 | 16.4 | 92.4 | 18.1 |
| SX-104 | | | | | |
| SX-105 | 81.8 | 3.0 | 15.2 | 91.9 | 15.2 |
| SX-106 | 89.2 | 3.7 | 7.1 | 101.7 | 33.3 |
| SX-109 | | | | | |
| S-102 | 59.1 | 0.9 | 40.0 | 62.2 | 42.5 |
| S-111 | 82.8 | 2.2 | 15.0 | 90.0 | 17.8 |
| S-112 | | | | | |
| T-110 | | | | | |
| U-103 | 51.9 | 1.5 | 46.6 | 57.0 | 61.2 |
| U-105 | 40.9 | 1.6 | 57.5 | 46.2 | 75.4 |
| U-107 | 53.3 | 2.1 | 44.6 | 60.4 | 62.6 |
| U-108 | 59.2 | 2.1 | 38.7 | 66.2 | 57.2 |
| U-109 | 55.0 | 1.4 | 43.6 | 59.7 | 58.7 |
| | The data from these samples are not used because the concentrations are either too | | | | |

The data from these samples are not used because the concentrations are either too low or are reported as less-than values.

Based on this analysis, the composition obtained for Tank AX-101 results in the maximum equivalent fuel being 103% H_2 and appears to be the limiting composition for the flammability analysis at the conditions prior to GRE.

C.3.2. Condensable Gases

The gas monitoring done in the tank farms in general show that almost all tanks contain and release ammonia. Unfortunately, there are limited ammonia data available to quantify the amount of ammonia in the tanks of interest. Because the waste in SSTs is typically older than the waste in the double-shell tanks (DSTs) for which ammonia content estimates are available, the amount of dissolved ammonia in the SST waste is expected to be higher than the amount of ammonia in the DST waste. There are two DSTs for which there are sufficient data to estimate the ammonia content in the waste gas during a release event. For Tank 101-SY, the maximum ammonia fraction in the release gas, including the mass transfer from a freshly exposed surface during a rollover, is obtained as 15%.⁷ For Tank 103-SY, the bounding ammonia fraction in the release gas is given as 17%.⁸

For Tank A-101, the amount of dissolved ammonia is estimated by comparing the total organic carbon, radiolytic loads, sodium nitrite inventory, and the age of the waste to Tank 101-SY waste using bounding assumptions.⁹ Based on these calculations, the mean plus 2 standard deviations ammonia fraction in the release gas is estimated as ~40%.⁹ Currently, similar analyses are not available for other tanks.

There are limited amounts of ammonia data available in Ref. 10. Typically, one value for a few of the tanks on the FGWL is reported, along with the corresponding value of nitrous oxide and hydrogen. Nitrous oxide and hydrogen values are within the range reported in Ref. 1. The gas concentration measurements reported in Ref. 10 are given in Table C-4 for those tanks that are on the FGWL. Data for each tank on Table C-3 are not available.

| TANK | NH3 (ppm) | H ₂ (ppm) | N ₂ O (ppm) | X _{ammonia} (%) |
|-------|-----------|----------------------|------------------------|--------------------------|
| A-101 | 754 | 758 | 218 | 46.5 |
| S-102 | 412 | 669 | 509 | 26.8 |
| S-111 | 122 | 39 1 | 48 | 20.5 |

TABLE C-4GAS CONCENTRATION DATA REPORTED IN REF. 10

TABLE C-5GAS CONCENTRATION DATA REPORTED IN REF. 10

| TANK | NH3 (ppm) | H ₂ (ppm) | N ₂ O (ppm) | X _{ammonia} (%) |
|--------|-----------|----------------------|------------------------|--------------------------|
| SX-103 | 77 | <23 | <23 | 72.8 |
| SX-106 | 179 | 98 | 14 | 62.0 |
| U-103 | 730 | 555 | 878 | 40.7 |
| U-105 | 325 | <49 | 154 | 61.4 |
| U-107 | 453 | 500 | 701 | 32.5 |

One can obtain a conservative estimate of ammonia fraction in the release gas through

$$\frac{X_{\text{ammonia}}}{1 - X_{\text{ammonia}}} = X_{\text{nc},i} \frac{X_{\text{ammonia}}}{X_{i}} , \qquad (C-4)$$

where $X_{nc,i}$ is the hydrogen or nitrous oxide fraction in the noncondensable gas that may be obtained from Table C-3 and $X_{ammonia}/X_i$ is the ratio obtained from data shown in Table C-4. The resulting ammonia fraction ($X_{ammonia}$) is shown in the last column in Table C-4.

In obtaining the results for $X_{ammonia}$, either the hydrogen or the nitrous oxide data from Tables C-3 and C-4 are used. In general, the hydrogen data are used, except for Tank U-105, for which the nitrous oxide data are used because the hydrogen data are given as "less-than" value. For Tank S-103, both the hydrogen and nitrous oxide numbers are given as less-than value, and the hydrogen data are used for this tank neglecting the < sign. The results for Tank S-103 have the highest uncertainty. First, the analysis reported in Table C-3 is based on two data points only. Secondly, the low concentrations in ammonia and hydrogen result in a large uncertainty in the ratio. Especially, the low ammonia concentration measured suggests that the mass-transfer contribution to the total ammonia release is probably very large (see discussion for Tank 101-SY below).

It is evident that there is much uncertainty in the ammonia results reported in Table C-4, primarily because the analysis is based on a single data point. However, the application of these results in consequence analysis for prompt and large gas releases is conservative because of the following:

- Nitrogen in the gas composition is neglected. Nitrogen is known to exist in the waste gas. Especially, nitrogen fraction is expected to be high in waste gas that is rich in ammonia and nitrous oxide.
- Undoubtedly, some water vapor also will exist in the form of condensable gas. The water vapor content of the waste gas is conservatively set at zero.
- Extensive data from Tank 101-SY indicate that a large fraction of ammonia during the steady-state periods is released by mass transfer of the surface. During steady-state periods, dome space ammonia-to-hydrogen ratio typically ranges between 2 and 3 in Tank 101-SY.⁷ During the steady-state period, the background hydrogen concentration typically would be ~20 ppm, but the ammonia concentration would be between 50 and 60 ppm. In this respect, the data for Tank S-103 appear similar to the Tank 101-SY data. However, during episodic large GREs, the ammonia-to-hydrogen ratio drops to 0.3 to 0.5 (Ref. 7). This behavior is explained by the high solubility of ammonia in the waste. During steady-state releases, the mass transfer of the surface significantly contributes to ammonia releases.

Also, slowly released bubbles are expected to come to thermodynamic equilibrium at the waste surface, where the pressure is atmospheric. During large episodic releases, mass-transfer contribution is small, and the bubbles are rapidly released from the lower layers where the ammonia fraction is small because of high hydrostatic pressure. Thus, applying the ammonia-to-hydrogen ratios obtained from background data is believed to be very conservative in predicting the large prompt releases that may be triggered from the deep layers during an intrusion.

Considering all these conservative assumptions, it is expected that the ammonia fraction during a prompt release to be much less than the 60% computed for Tanks SX-106 and U-105. For instance, Ref. 9 uses a simple model based on the energy of formation of nitrogen and other species to obtain the best-estimate ammonia fraction in the release gas as 20% for Tank A-101, but the above methodology yields \sim 47%. However, given the other uncertainties, the ammonia fraction is set conservatively to 60% the for bounding analysis. Once the ammonia concentration is known, the overall waste-gas composition is estimated as

$$X_i = (1 - X_{ammonia})X_{nc,i}$$

(C-5)

where X_i (i is hydrogen, nitrous oxide, or methane) is the species fraction in the release gas and $X_{i,nc}$ is the species fraction in the noncondensable portion of the release obtained from Table C-3.

The ammonia fraction in the release gas is important in determining the safety envelope for the proposed activities. The presence and amount of ammonia affects the safety analyses in the following major areas:

- Ammonia is a more energetic fuel than hydrogen. Thus, the existence of ammonia increases the fuel energy of the waste gas. The effect of ammonia on the equivalent fuel is discussed in Section 3.2.1 of this appendix.
- Ammonia affects the LFL of the mixture. When expressed in terms of hydrogen concentration, the LFL decreases with increasing ammonia content. The effect of ammonia on the LFL is discussed in Section 3.3 of this appendix.
- Ammonia appears to combust very energetically in a pure nitrous oxide environment.^{11,12}
- Ammonia is corrosive, and the consequences of the ammonia releases must be kept below the acceptance guidelines (see Section 5 of this SA).

C.3.2.1. Effect of Ammonia on the Equivalent Fuel. Including ammonia, the equivalent fuel is obtained as

$$EF(X_{hydrogen}) = R_m \times X_{methane} + R_a \times X_{ammonia} + X_{hydrogen} , \qquad (C-6)$$

where R_m is 3.32 (discussed above) and R_a is the ratio of combustion energy of ammonia (317 kJ/mole) to the combustion energy of hydrogen (241_kJ/mole). Thus, R_a is obtained as 1.32.

Using 60% NH₃ and the noncondensable gas composition given in Table C-3 for Tank AX-101, the equivalent fuel becomes equal to 120% $H_{2,c}$ compared to 103% H_{2} obtained without any ammonia.

C.3.3. Lower Flammability Limits

The LFL of hydrogen, ammonia, methane, and nitrous oxide mixtures is discussed in Refs. 13 and 14. Based on Refs. 13 and 14, Le Chatelier's linear mixing law¹⁵ appears to be adequate, and the effect of nitrous oxide on the LFL is minimal for the mixtures of interest. The linear mixing law can be expressed in terms of hydrogen concentration as follows:

$$LFL\left(\frac{1}{LFL_{o,hydrogen}} + \frac{X_{ammonia} / X_{hydrogen}}{LFL_{o,ammonia}} + \frac{X_{methane} / X_{hydrogen}}{LFL_{o,methane}}\right) = 1 \quad (C-7)$$

where the LFL of the hydrogen, methane, and ammonia $(LFL_{o,i})$ are bounded by 4, 5, and 15%, respectively^{-13,14} The LFL for each gas composition shown in Table C-3 is computed as a function of the ammonia fraction in Ref. 6. If we add 60% NH₃ to the AX-101 composition shown in Table C-3, the LFL is obtained as 2.7% H₂.

C.4. MAXIMUM ALLOWABLE RELEASES INTO THE DOME

In this section, the maximum allowable gas releases into the dome space considering the flammability issues and toxicological consequences are discussed first. Also, the maximum allowable burn volume in the waste is quantified. Finally, current estimates for the retained-gas inventory in the tanks of interest are summarized at the end of this section.

Using the LFL values computed for each tank, the maximum allowable release based on the flammability hazard is defined as the gas-release volume necessary to reach 25% of the LFL when the waste gas is homogeneously mixed in the dome space.

For the bounding tank with a 1415.8-m³ (50,000-ft³) dome volume, the maximum allowable release corresponding to 6000 ppm H_2 in the dome space would be 8.5 m³

(300 ft³) H₂. Currently, all intrusive activities are stopped when 6000 ppm H₂ is detected in the dome space (see Appendix B). Using the gas composition obtained for 101-AX (with 60% NH₃), the release volume corresponding to 6000 ppm H₂becomes 24 m³ (850 ft³).

Using an adiabatic burn model, we can show that, if burned, a gas-release volume of 96 m³ (3400 ft³) corresponding to 2.4% H₂ in the dome space would result in a peak pressure >240 kPa (35 psia). The maximum peak pressure that could be experienced by SSTs without dome failure is <82 kPa (<12 psig).¹⁶ Thus, even small volumes of gas burned in the dome space may result in catastrophic dome failure. Note that the gas release volume of 96 m³ is not 100% hydrogen. It was assumed that there is 60% ammonia in the released gas. Remaining 40% includes 88.1% hydrogen (Tank AX-101) as indicated in Table C-3. Thus, the hydrogen concentration becomes 35% in the released gas. This results in 2.4% hydrogen concentration in the dome.

The above values for the maximum allowable limit are obtained considering flammability issues only. Ammonia and nitrous oxide also must be considered for toxicological consequences. The Emergency Response Planning Guideline-1 (ERPG-1) values for ammonia and nitrous oxide are 25 and 150 ppm, respectively. ERPG-1 values are used to determine the dose limit for an on-site receptor located 100 m (328 ft) from the tank for likely releases (see Section 5 of this SA for the Risk Acceptance Guidelines). The maximum allowable release then is determined as the maximum waste-gas volume that results in \leq ERPG-1 when discharged out of the dome and received 100 m (328 ft) from the source.

The atmospheric dispersion factor using 95% weather conditions at the Hanford Site for a 100-m receptor is ~3.44 x 10^{-2} s/m³, which is obtained from Ref. 17. For a 0.12-m³/s (250-ft³/min) discharge rate, the maximum source concentration may be obtained as 6000 ppm NH₃ and 36,000 ppm N₂O. If we assume 60% NH₃ in the waste gas and a dome volume of 1415.8 m³ (50,000 ft³), the maximum allowable release may be obtained as 14.2 m³ (500 ft³) to meet the ammonia guidelines. Waste gas releases of up to 68 m³ (2400 ft³) will meet the nitrous oxide guidelines, assuming a maximum of 75% N₂O in the waste gas and a 0.12-m³/s (250-ft³/min) discharge rate. These numbers are directly proportional to the discharge rates.

C.5. MAXIMUM ALLOWABLE BURN VOLUME IN THE WASTE

In this safety assessment (SA), it is assumed that if the gases are ignited in the waste, the combustion will propagate through the gas phase. If a large enough gas volume is burned, the resulting pressure may be high enough to cause structural damage to the tank and/or dome.

In the previous sections, the concern was the release of flammable and toxic gases into the dome space. For dome burn accidents, the gas composition with the

maximum fuel is used as the bounding composition because the dome contains sufficient oxidizer. Likewise, for bounding toxicological analysis, the maximum toxic gas fraction is used. For a burn inside the waste, the worst-case composition would be a stoichiometric mixture of fuel and nitrous oxide. For gases to burn inside the waste, there must be sufficient oxidizer. The only known oxidizer in the waste gas is nitrous oxide. Although the results in Table C-3 have uncertainties associated with them, they show that most of the tanks may be oxidizer limited. Among the tanks for which the gas composition data are analyzed, the nitrous oxide concentration is <20% except for the tanks in the U Farm, Tanks AX-103, SX-106, and S-102. Within the uncertainty of the data, the tanks in the U Farm could certainly have a stoichiometric composition of fuel and oxidizer. In the subsequent analysis, a stoichiometric mixture of hydrogen and nitrous oxide is used. Adding methane and ammonia into the mixture would increase the energy per mole. However, theoretically, complete combustion of 1 mole CH_4 requires 4 moles N_2O_1 , and 1 mole NH₃ requires 1.5 moles N₂O. Thus, methane and ammonia, while more energetic, require more oxidizer than the hydrogen. Thus, for a constant gas volume, adding ammonia and methane to the stoichiometric mixture of hydrogen and nitrou oxide has a small impact on the total combustion energy.

Starting at atmospheric pressure, the adiabatic constant-volume combustion of 1 mole H_2 with 1 mole N_2O results in a peak pressure of 11.2 atm. If the combustion products are allowed to expand isentropically while compressing the dome gases, the equilibrium pressure may be obtained by solving the following simultaneous equations:

$$V_{f} + V_{cf} = V + V_{c} , \qquad (C-8)$$

$$\frac{P_{a}}{P_{f}} = \left(\frac{V_{f}}{V}\right)^{k_{a}} , \qquad (C-9)$$

and

$$\frac{P_c}{P_f} = \left(\frac{V_{cf}}{V_c}\right)^{k_c} , \qquad (C-10)$$

where P_a is the initial dome pressure (1 atm), P_c is the waste gas burn pressure (11.2 atm), P_f is the final equilibrium pressure, V is the initial dome volume, V_c is the volume of the waste gas (combustion products), V_f is the final dome volume after compression, V_{cf} is the final waste gas (combustion products) volume after expansion, and k is the specific heat ratio. For both the dome gases and the combustion products, $k_a = k_c = 1.4$ is used in these calculations. If we set the structural limit to 1.8 atm (Ref. 16), the maximum allowable gas volume may be obtained as 184 m³ (6500 ft³) for V = 1416 m³ (50,000 ft³).

August 8, 1996

This approach is conservative because it assumes that there is no pressure relief (no outflow from the dome) during the pressurization phase. Furthermore, it is very unlikely that the burn will easily propagate in the waste. The total energy generated by the combustion of 283 m³ (10,000 ft³) stoichiometric waste gas is 4.1 x 10⁶ kJ (4.1×10^6 Btu). The latent heat of 1.8 m^3 (64 ft^3) of water will be sufficient to absorb all of this energy. Thus, if the water volume in the waste is only 0.3% of the gas volume and if the gas and water are homogeneously distributed, the combustion cannot propagate. If a large gas volume exists, a 184-m^3 (6500-ft^3) gas volume would correspond to a spherical bubble 7.1 m (23 ft) in diameter. Alternatively, a dendritic bubble 2.54 cm (1 in.) in average diameter would have to be 3.6×10^5 m (1.2×10^6 ft) long.

Based on these arguments, a gas burn in the waste that could result in dome failure is very unlikely. However, it should not be dismissed completely because of the various unknowns in the gas-retention mechanisms and the waste properties in the SSTs.

C.6. Estimates of Tank Gas Inventory

As part of the screening program for potential FG retaining tanks, Westinghouse Hanford Company (WHC) has a continuing effort to estimate the tank gas inventory using the barometric pressure and level variation data.¹⁸ The results for all the tanks of interest are not currently available. The gas inventories of some of the tanks that are not currently on the FGWL are given in Ref. 18.

C.7. CONCLUSIONS

In this appendix, the available gas-composition data obtained from the grab samples are analyzed. However, the grab samples only provide limited data. Conservatively assuming that the waste gas does not contain any nitrogen, the grab sample data are converted to noncondensable waste gas composition. The upper-bound values for the composition with maximum fuel and maximum nitrous oxide are obtained. In terms of fuel, the bounding tank is shown to be AX-101. Tank U-105 appears to show the highest nitrous oxide content. Tanks A-101, AX-101, SX-102, SX-105, SX-106, and S-111 indicate >80% H₂ in the waste gas. The highest hydrogen content is estimated for SX-106 as nearly 90%.

Even though the available data are analyzed conservatively, we must exercise caution in using the current findings. The concentrations are obtained using a very limited database. Furthermore, the SSTs within a given farm are connected through overflow lines, and most SSTs are passively ventilated. Consequently, it is possible that the dome gases in connected tanks intermix, and the grab sample data may not necessarily represent the waste gas that is released from a specific waste tank. Furthermore, data for all the tanks of interest are not available. Using a conservative interpretation of the limited ammonia data provided in Ref. 10, the LFL is set equal to 2.4% H_{2} , corresponding to an ammonia-to-hydrogen ratio of 2.5. For toxicological consequence analysis, the bounding ammonia fraction in the release gas is set conservatively equal to 60%.

Based on these findings, it is concluded that

- The potential gas releases during the proposed intrusive activities are poorly understood and cannot be controlled to meet these limits.
- Continuous hydrogen or FG monitoring is necessary as part of the spark management strategy.
- Continuous or frequent ammonia monitoring also is necessary, both for flammability and toxicological consequence considerations.
- A deflagration in the dome space is a more likely dome collapse mechanism than a deflagration of the gas trapped in the waste.
- Ammonia concentrations must be monitored, and operations must be stopped when the ammonia concentration in the dome exceeds 5000 ppm.

C.8. REFERENCES

- 1. Westinghouse Hanford Company, "Hanford Site Tank Farm Facilities Interim Safety Basis, Volume 2: Design Description," Westinghouse Hanford Company report WHC-SD-WM-ISB-001, Rev. 0, Vol. 2 (August 1993).
- 2. H. R. Brager, "Summary of Information on Flammable Gas Watch List Tanks," Westinghouse Hanford Company report WHC-EP-0711, (January 1994).
- 3. P. Whitney, "Screening of the Hanford Tanks for Trapped Gas," Pacific Northwest Laboratory report PNL-10821 (October 1995).
- 4. K. O. Pasamehmetoglu, "Single-Shell Tank Dome Volume Calculations," Los Alamos National Laboratory Calc-Note TSA10-CN-WT-SA-TH-086 (November 1995).
- 5. R. G. Brown, "Status of Single-Shell Tank SHMS," Westinghouse Hanford Company internal memorandum 74A10-95-068 to G. D. Johnson (October 2, 1995).
- K. O. Pasamehmetoglu, "Gas Composition Estimates for Single-Shell Tanks," Los Alamos National Laboratory Calc-Note TSA10-CN-WT-SA-GR-048 (December 1995).
- 7. L. H. Sullivan et al., "A Safety Assessment for the Proposed Pump Mixing Operations to Mitigate Episodic Gas Releases in Tank 241-SY-101: Hanford Site, Richland, Washington," Los Alamos National Laboratory report LA-UR-92-3196, Rev. 14 (March 31, 1995).
- 8. R. J. Van Vleet, "Basis for Closure of USQ for Tank 103-SY," presented to Tank Waste Chemical reactions Sub-Panel (October 11, 1995).
- 9. W. L. Kubic, "Summary, Review, and Analysis of Data for Tank 241-A-101," Los Alamos National Laboratory Calc-Note TSA10-CN-WT-SA-DA-001 (December 1995).
- 10. J. L. Huckaby, H. Babad, and D. R. Bratzel, "Headspace Gas and Vapor Characterization Summary for the 43 Vapor Program Suspect Tanks," Westinghouse Hanford Company report WHC-SD-WM-ER-514, Rev. 1A (September 1995).
- 11. M. Hertzberg and I. A. Zlochower, "Explosibility of Nitrous Oxide: the Effect of H-Atom Bearing Impurities," Twenty-Fifth International Symposium on Combustion, University of California, Irvine, July 31-August 5, 1994.

- 12. E. Jones and J. C. Kerr, "Inflammability Limits of Ammonia, Nitrous Oxide and Air," J.S.C.I 68, 31-34 (January 1949).
- 13. K. O. Pasamehmetoglu, "Composition and Lean Flammability Limit for the Release Gases in Tank 241-SY-101," Los Alamos National Laboratory report LA-UR-94-4229 (December 1994).
- 14. K. O. Pasamehmetoglu, "LFL for 101-SY Release Gas," Los Alamos National Laboratory Calc-Note N6-CN-WT-SA-GR-010 (March 1994).
- 15. W. Jost, Explosion and Combustion Processes in Gases (McGraw Hill, New York, New York, 1946), p. 134.
- 16. L. J. Julyk, "Static Internal Pressure Capacity of Hanford Single-Shell Waste Tanks," Westinghouse Hanford Company report WHC-SD-WM-TI-623 (July 1995).
- 17. Westinghouse Hanford Company, "Tank Waste Compositions and Atmospheric Dispersion Coefficients for Use in Accelerated Safety Analysis Consequence Assessment," Westinghouse Hanford Company report WHC-SD-WM-SARR, Rev. 1 (October 1995).
- K. M. Hodgson et al., "Evaluation of Hanford Tanks for Trapped Gas," Westinghouse Hanford Company report WHC-SD-WM-ER-526 Rev. 0 (December 1995).

APPENDIX D

RELIABILITY OF SYSTEMS

D.1. INTRODUCTION

This appendix documents the results of an assessment of the reliability of various systems that support the installation, operation, and removal of rotary-mode core sampling (RMCS) equipment in single-shell tanks (SSTs) on the Flammable Gas Watch List (FGWL) or those tanks recommended by the contractor to be included on the FGWL, hence referred to as FG/RMCS operations. The systems that were evaluated are those considered in the accident sequences discussed in Section 4 and documented in Appendix E of this safety assessment.

The reliability of the systems were estimated using fault tree analysis, and event tree analysis in a few cases. The system fault trees are not detailed fault trees; however, they do provide a reasonable model for estimating system reliabilities. The level of detail included in the fault trees was limited by the availability of detailed up-to-date system design and operating information.

The system models were developed using the following sources of information:

- 1. Existing Westinghouse Hanford Company (WHC) safety studies,¹⁴
- 2. System schematics and operating procedures provided by WHC,
- 3. Verbal information from WHC cognizant engineers and technicians,
- 4. Published sources of data for component reliabilities, and
- 5. Engineering judgment.

It should be noted that automatic trip of drill string is defined as automatic trip of the drill rig engine. The following systems were modeled:

- Nitrogen purge gas cooling for drill bit and associated automatic trip of drill string on low nitrogen flow;
- Nitrogen hydrostatic supply to drill string;
- Nitrogen hydrostatic supply to shielded receiver;
- Automatic trip of drill string on excessive force;
- Automatic trip of walkdown function;
- Automatic trip of hydraulic bottom detector;

- Automatic trip of drill string on high rpm;
- Pneumatic footclamp for holding drill string;
- Automatic trip of drill string on:

Flammable-gas concentration,

Flammable-gas concentration rate of rise, and

Dome pressure;

- Sampler rotary valve operation;
- Sampler chevron seal function; and
- Manual sniff of drill string for flammable gas.

To estimate component failure exposure times, the following time periods were used:

- Two drilling activities on a tank per year;
- 144 hours for an activity from arrival at tank for core sampling until leaving tank after core sampling complete;
- 40 hours to take all samples; 2 samples taken per 8-hour shift;
- 20 minutes drilling time required for each of 11 samples; and
- 6 months between calibration check of instrumentation.

D.2. SUMMARY AND RESULTS

The details of the systems reliability analysis are provided in Refs. 5 and 6. The results are summarized in Table D-1.

Based on these results, human error is concluded to be an important contributor to system failure.

During the development of the system models it was noted that the following qualitative aspects of hardware faidlures. The reliability of the PLC is important because it is a single failure element for both

- automatically tripping the drill, and
- providing annunciation to the operators for manual response to stop drilling or to trip the drill.

The reliability of the rpm-detection system is important because it must

- Trip the drill on high RPM,
- Detect drill string rotation above 2 rpm to enable the automatic trip on low nitrogen purge gas flow, high down force, low penetration rate, and high RPM.

Both the PLC and the rpm detection systems are sufficiently reliable so that failures in these systems do not excessively contribute to the quantified system failure values.

| Initiating Event Frequency (1/activity) | Probability of Subsequent Failures | Overall Frequency (1/activity) | Contributing Failures (Failure of Following Events Dominates Probability of Subsequent Failures) |
|---|---|--|--|
| Excessive force used while taking any sample except last sample, 0.05 per sample, 0.55 total for 11 samples | Failure to detect excessive force and stop drill 7.9E-5 (Event: force) | 4.3E-5 | Common mode calibration error, pressure transmitters, operator arm walkdown function, solenoid-operated valve (SOV) 12 |
| Excessive force used while taking last sample 0.05 | Failure to detect excessive force and stop drill 7.9E-5 | 4.0E-6 | Common mode calibration error, Pressure transmitters, operator arm, bottom detector, SOV 11 |
| | | 4.7E-5 total for excessive force with failure to stop drill (sum of two previous values) | |

TABLE D-1 RESULTS OF SYSTEM RELIABILITY QUANTIFICATION

TABLE D-1 (cont)RESULTS OF SYSTEM RELIABILITY QUANTIFICATION

| Initiating Event Frequency (1/activity) High H ₂ level in exhauster | Probability of Subsequent Failures Failure to detect hydrogen | Overall Frequency (1/activity) 7.6E-4 | Contributing Failures (Failure of Following Events Dominates Probability of Subsequent Failures) 2 exhauster programmable logic controllers (PLCs) |
|--|---|---|--|
| 1.0 | 7.6E-4 Failure to detect hydrogen and trip the drill string 1.3E-3 + 3.0E-4* | 1.6E-3 | (common mode), instrumentation and control (I&C) calibration 2 exhauster PLCs (common mode), truck PLC, I&C calibration |
| High H ₂ rate of rise in exhauster 1.0 | Failure to detect hydrogen 7.6E-4 Failure to detect hydrogen and trip the drill string 1.3E-3 + 3.0E-4* | 7.6E-4 1.6E-3 | 2 exhauster PLCs (common mode), I&C calibration 2 exhauster PLCs (common mode), truck PLC, I&C calibration |
| High dome pressure caused by H ₂ release 1.0 | Failure to detect hydrogen. 7.7E-4 Failure to detect hydrogen and trip the drill string. 1.4E-4 + 3.0E-4* (Event: hydrl) | 7.7E-4 1.7E-3 | 2 exhauster PLCs (common mode), I&C calibration 2 exhauster PLCs (common mode), truck PLC, I&C calibration |

TABLE D-1 (cont) RESULTS OF SYSTEM RELIABILITY QUANTIFICATION

| Initiating Event Frequency (1/activity) H ₂ from waste | Probability of Subsequent Failures Failure of drill string No hydrostatic | Overall Frequency (1/activity) 6.4E-3 | Contributing Failures (Failure of Following Events Dominates Probability of Subsequent Failures) Operator connect drill string hydro, operator arm drill |
|--|--|---|---|
| 1.0 | system. 6.4E-3 | | string hydrostatic system |
| H ₂ from waste | Failure of shielded receiver N ₂ hydrostatic system. 6.4E-3 | 6.4E-3 | Operator connect shielded receiver hydrostatic system, operator arm shielded receiver hydrostatic system (NOTE: The contributing failure associated with connecting the SR hydrostatic system is overly conservative because the system is hard-piped.) |
| 1.0 | Failure of both drill string N ₂ hydrostatic system and shielded receiver N ₂ hydrostatic systems. 6.9E-4 (Event: hydro) | 6.9E-4 | Operator connect drill string hydrostatic system and connect shielded receiver hydrostatic system (common mode), operator arm drill string hydrostatic system and shielded receiver hydrostatic system(common mode) (NOTE: The contributing failure associated with connecting the SR hydrostatic system is overly conservative because the system is hard-piped.) |
| Hold drill string above waste with footclamp 1.0 | Footclamp drops drill string onto waste surface. 3.0E-5 (Event: foot) | 3.0E-5 | Operator opens foot clamp. Lock wrench left out. |

TABLE D-1 (cont) RESULTS OF SYSTEM RELIABILITY QUANTIFICATION

| Initiating Event Frequency (1/activity) | Probability of Subsequent Failures | Overall Frequency (1/activity) | Contributing Failures (Failure of Following Events Dominates Probability of Subsequent Failures) |
|---|---|--|---|
| Excessive rpm during drilling 0.011 | Failure to trip drill string on excessive rpm. 6.2E-4 (Event: rpm) | 6.8E-6 | truck PLC, I&C calibration |
| Perform drilling operation 1.0 | Total loss of N ₂ cooling and failure to stop drill. 3.3E-6 | 3.3E-6 | N ₂ cooling holes in drill bit plug, truck PLC |
| Perform drilling operation 1.0 | Partial blockage of N_2 cooling holes in drill bit does not lead to total low N_2 flow 5E-4 original calculation; operating envelope will ensure no overheating can occur unless N_2 low-flow trip also challenged and fails | Operating envelope controls will prevent this. | No blockage of some, but not all, of the N ₂ cooling holes in the drill bit, no low N ₂ flow as a result of partial blockage, no local overheating of portions of drill bit |
| Perform drilling operation 1.0 | Loss of drill bit N ₂ cooling from N ₂ bypass leakage 1.6E-5 | 1.6E-5 | No N ₂ bypass flow, no high N ₂ flow as result of bypass, operator detects N ₂ bypass flow |

TABLE D-1 (cont) RESULTS OF SYSTEM RELIABILITY QUANTIFICATION

| Initiating Event Frequency (1/activity) Perform drilling operation 1.0 | Probability of Subsequent Failures High N ₂ temperature results in loss of drill bit cooling, Low likelihood based on WHC information that N ₂ reaches ambient waste temperature | Overall Frequency (1/activity) very small | Contributing Failures (Failure of Following Events Dominates Probability of <u>Subsequent Failures</u>) No high N ₂ temperature, no overheating of drill bit from high N ₂ temperature, operator trips drill on annunciation of high N ₂ temperature, temperature transducer |
|---|---|---|--|
| | | 1.9E-5 total for overheating drill bit caused by inadequate N ₂ cooling (sum of four previous values) (Event: cool) | |
| Take a sample 11.0 | Failure of rotary valve in sampler to completely close.0.1(Event: rot) | 1.1 | No waste blockage during valve closure |
| Perform drilling operation 1.0 | Failure of sampler chevron seal. (Event: seal) | 3.3E-2 | Installation of chevron seal |
| H2 from Waste 1.0 | Failure of H₂ sniff in drill string. 4.6E-3 (3.0E-3 operator, 1.6E-3 hardware) (Event: sniff) | 4.6E-3 | Operator fails to perform sniff, sniff sensor fails low |

TABLE D-1 (cont) RESULTS OF SYSTEM RELIABILITY QUANTIFICATION

| Initiating Event Frequency (1/activity) | Probability of Subsequent Failures | Overall Frequency (1/activity) | Contributing Failures (Failure of Following Events Dominates Probability of Subsequent Failures) |
|---|--|-----------------------------------|---|
| Excessive filter ΔP 1.0 | Fail to trip exhauster on filter ΔP. 2.6E-3 (Event: gate 32) | 2.6E-3 | Failure to sense high ΔP across filter, calibration I&C error |
| Contact rock in waste 0.1 | Failure of penetration rate and failure of N₂ cooling to drill bit systems. 3.3E-6 (Event: prrate) | 3.3E-7 | N2 cooling holes to drill bit plug, PLC |

* 3.0E-4 accounts for requirement to enable the exhauster interlock.

D.3. REFERENCES

- 1. W. L. Cowley and D. D. Stepnewski, "Evaluation of Hazards to Tank Farm Facilities from Lightning Strikes," Westinghouse Hanford Company report WHC-SD-WM-SARR-027 Rev 0 (June 29, 1994).
- 2. D. J. Braun, "Risk Assessment for Nitrogen Purge System for Rotary Mode Sampling Truck," Westinghouse Hanford Company report WHC-SD-WM-RA-066 Rev. 0 (September 1992).
- 3. Westinghouse Hanford Company, "Fault Tree Systems Analysis," Westinghouse Hanford Company report WHC-IP-0690 6.4, Rev 0 (June 30, 1992).
- 4. J. E. Kelly, "High Downward Force on Rotary Drill," Westinghouse Hanford Company Internal Memorandum 29220-93-030 (July 7, 1993).
- 5. J. Darby and M. Butner, "Reliability of Systems for Rotary Mode Core Sampling," Los Alamos National Laboratory Calc-Note TSA10-CN-WT-SA-HAZ-004, Rev. 1 (February 1996).
- 6. J. Darby, "Additional Calculations for Push Mode Event Trees," Los Alamos National Laboratory Calc-Note TSA10-CN-WT-HAZ-009, Rev.0 (June 4, 1996).



APPENDIX E

ACCIDENT SEQUENCES

E.1. INTRODUCTION

Quantification results from event trees are discussed in this appendix for the installation, operation, and removal of rotary-mode core sampling (RMCS) equipment in single-shell tanks (SSTs) on the Flammable Gas Watch List (FGWL) or those tanks recommended by the contractor to be included on the FGWL, hence referred to as FG/RMCS operations. These event trees, developed using the SANET (Ref. 1) code, document the accident sequences discussed in Section 4 of this report. Reference 2 includes the event trees, the detailed accident discussions, and quantification details.

E.2. BACKGROUND INFORMATION

This section discusses the bases for the human-error rate assignments used throughout the accident quantification as well as mission time values used to quantify failure rates and frequencies.

E.2.1. Assumptions Underlying .003 Human Error Probability as Given in NUREG-CR-4772 (Ref. 3)

- 1. Task failure is not immediately annunciated either if a task is not performed or if an error is made during the task.
- 2. A post-task test is either not required or it does not immediately annunciate error.
- 3. Neither a shift nor a daily check is made to verify the status of the system associated with a task.
- 4. Independent task verification is ensured by requiring that either a second person verify whether a task is performed correctly after a task is completed or whether the original task performer verifies a task correctly performed at a different time and location.
- 5. Verification procedures are written correctly and verification procedure is followed correctly.
- 6. Verification fails to discover a task error with a probability of 0.1.

In those activities to which we have assigned a .003 human error failure rate, we implicitly assume that the associated activity is controlled by a correctly

written procedure and that the activity has independent verification using a check-off sheet.

If the second-person verification is not performed, this probability increases to .03. Assignment of this human failure probability still requires the associated task to be controlled by procedure.

E.2.2. Assignment of the .05 Failure Probability for Judgment-Based Tasks

The Systematic Human Reliability Procedure (SHARP) gives a failure probability range of .5 to 5E-03 for judgment-based actions (Ref. 4). For actions considered to be judgment-based (dependent on a mental representation of a system), we assigned a probability of .05, the mid-point of the SHARP range.

E.2.3. Mission Time Assumptions

An activity is defined as the process of collecting a set of core samples corresponding to an entire tank depth. One hundred forty-four hours are required to complete an entire activity. Within this 144-h period, 40 h (approximately 2 samples collected per shift) are required to retrieve the 11 samples [based on an average single-shell tank (SST) waste depth]. Total drilling time is approximately 4 h (20 min. per sample). Instrumentation and controls (I&C) are checked, and if necessary, calibrated every six months.

E.2.4. Dependence of Event-tree Events.

Because of the lack of complexity inherent in the rotary-core mode drilling activities, it was possible for us to construct event trees having independent top events. Therefore, accident frequencies can be determined by multiplying path frequencies and probabilities. Important dependencies were handled by broadening the event definition so that the associated fault tree considers them.

E.3. ACCIDENT SEQUENCES

Table E.1 summarizes the quantification of the accident sequences based on assigning a high value to all phenomena except for the likelihood of having flammable gas at a flammable concentration contacting equipment randomly located near risers. This open-riser probability is documented in Ref. 5. The Table E.1 values in parentheses are the accident frequencies that result if only passive (non-control) system hardware is available to respond to accident initiators. These frequencies are called "unmitigated accident frequencies." Both active control hardware and administrative controls are credited in the other listed accident frequencies, which are called "mitigated frequencies".

TABLE E-1 SUMMARY OF ACCIDENT QUANTIFICATION RESULTS

| Accident | Frequency (1/Activity) | Dominant Failures | Controls Credited for Mitigative Frequency Quantification |
|--|---|--|--|
| Lightning strike causes dome fire. (Note 1) | 4.1E-6 (8.2E-6) | Random lightning strike hits risers or other equipment on top of tank that connects to tank interior | Do not drill during stormy weather. |
| Hot spots caused (by drill bit contacting rocks in waste. (Note 2) | 3.3E-7 (0.1) | Hot drill bit contacts rock in waste resulting in flammable gas ignition due to a hot spot exceeding autoignition temperature of the mixture. Penetration rate and nitrogen purge systems fail. | Trip drill string on slow penetration rate (no time for operator trip if penetration rate trip fails). Auto trip of drill string on low N ₂ purge gas flow. |
| Spark from failure of exhauster heater leads to dome fire. (Note 1) | 0 (0) | None | Replace electric heater with hot-water heat exchanger in exhauster to provide moisture removal for high-efficiency particulate air (HEPA) filters |
| Spark from Assembly/disassembly of drill string sections: drill string in dome (Notes 1 and 2) drill string in waste (Note s 2 and 4) | 7.6E-5 total (1.65E-2) 3.8E-5 (8.25E-3) 3.8E-5 (8.25E-3) | Fails to perform sniff for flammable gas in drill string. Sampler chevron seal fails. | Sniff enclosed volume for flammable gas in drill string. Presence of sampler with chevron seal in drill string prevents flammable gas movement into drill string. |
| Drop of equipment from crane onto exhauster leads to dome fire (Note 1). | 3.4E-7 (6.8E-6) | Drops equipment from crane onto exhauster. | Control of equipment lifts over exhauster. |
| Failure of exhauster fan causes spark leading to dome fire. (Note s 1 and 5) | 2.3E-5 (2.3E-5) | Fan bearings seize, shaft fails, and impeller impacts fan housing causing spark. | Fan impeller and housing are Al, which has low potential for sparking. |
| Leaking riser penetration releases flammable gas to unqualified electrical equipment leading to dome fire (Note 1) | 6.5E-3 (1.0) | Randomly located, unqualified equipment is located too close to riser/pit that is not sealed. | Limit leakage from all unused risers/pits to less than 1 in. equivalent diameter. |
| (Note: For this accident we calculated the probability that flammable gas in excess of 2.4% reaches equipment randomly placed on the tank top.) | | | Platform on truck is present to mitigate release from drill-string rubber seal. Following tank walkdown, 50% of risers are assumed to leak. |

.

1. S. ,

E-3

TABLE E-1 (cont) SUMMARY OF ACCIDENT QUANTIFICATION RESULTS

| Accident | Frequency (1/Activity) | Dominant Failures | Controls Credited for Mitigative Frequency Quantification |
|--|---|---|--|
| Truck placed over an unused penetration results in an above ground fire leading a dome fire | 1.5E-4 (1.0) | Operators fail to seal unused penetration Failure to locate truck ignition sources a minimum of 36" away from risers or pit over which the truck is positioned | Limit leakage from penetration so flammable gas will not reach unqualified equipment on the drill truck Truck is located greater than 36" away from risers or pit over which the truck is positioned. |
| Intentional opening of penetration during drilling results in an above ground fire | 1.5E-3 (0.5) | Equipment located too close to open penetration | Locate unqualified equipment > 36 D from open penetration Inlet stack 15 ft tall installed on HEPA inlet riser. |
| Flammable gas is in drill string/shielded receiver during sampler handling leading to dome fire. (Notes 2 and 4) | 9.7E-7 (1.4E-3) | N2 hydrostatic system to both drill string and to shielded receiver fail. Remote latch unit (RLU) drops sampler. | Leak test N2 hydrostatic systems for both drill string and for shielded receiver. Unique connections for N2 hydrostatic systems for both drill string and shielded receiver. Operators verify N2 hydrostatic supply to both drill string and shielded receiver during activation of hydrostatic mode of N2 supply. Controls over operation of RLU. |
| Flanmable gas in shielded receiver with shielded receiver isolated from drill string results in aboveground fire. (Notes 2 and 6) | 1.4E-5 (not a dome fire accident) (1.5E-4) | Waste prevented previous closure of sampler rotary valve leading to gas release as sample is retrieved and depressurized. Sniff for flammable gas fails to detect flammable gas. RLU drops sampler. | Controls over the operation of RLU are provided. |

| Accident | Frequency (1/Activity) | Dominant Failures | Controls Credited for Mitigative Frequency Quantification |
|---|--|---|---|
| Flammable gas in X-ray machine leads to aboveground fire. (Notes 2 and 6) | 8.3E-10 (not a dome fire accident) (8.3E-9) | Waste prevented previous closure of sampler rotary valve leading to gas release as sample is retrieved and depressurized. | Design includes sealed plastic sampler receiver surrounded by isolation barrier. Controls over operation of RLU are |
| | | Previous sniff for flammable gas in shielded receiver failed to detect flammable gas. | used. |
| | | RLU drops sampler. | |
| | | Plastic receiver breaks. | |
| · | | Isolation barrier volume fails. | |
| Flammable gas in cask leads to above ground fire. (Notes 2 and 6) | 1.4E-5 (not a dome fire accident) (1.4E-4) | Waste prevented previous closure of sampler rotary valve leading to gas release as sample is retrieved and depressurized. | Controls over operation of RLU are used. |
| | | Operator fails to perform sniff for flammable gas or sniff not successful | |
| · · · | | RLU drops sampler. | |
| Flammable gas in riser/sleeve during drilling leads to dome fire. (Notes 1 and 2) | 1.1E-5 (1.0) | N_2 to riser sleeve fails and with subsequent failure to shut down drill on loss of N_2 (slow release not seen in dome). Drill fails to shut down on detection of flammable gas (rapid release seen in | Design includes drill string automatic trip when high flammable gas level, high rate of change of flammable gas level, or high dome pressure. Provide for drill string trip on loss of N ₂ to riser sleeve. |
| | | dome), and failure of riser sleeve N ₂ supply and shut down of drill on loss | Provide for leak check of N ₂ supply to riser sleeve. |
| | | of N <u>2</u> . | Provide for unique connector for N ₂ supply to riser sleeve. |
| | | | Provide for verification of N2 supply to sleeve during actuation of system. |
| Drop of tool into open riser leads to dome burn. (Note 1) | 5E-5 (1.0E-4) | Drop of tool into open riser. | Use of spark-resistant tools within 36 D of open risers. |

TABLE E-1 (cont) SUMMARY OF ACCIDENT QUANTIFICATION RESULTS

WHC-SD-WM کر___-035, Rev. 0-a

| · · · · · · · · · · · · · · · · · · · | | | |
|--|------------------------|--|--|
| Accident | Frequency (1/Activity) | Dominant Failures | Controls Credited for Mitigative Frequency Quantification |
| Drill string break in dome leads to dome fire. (Notes 1 and 2) | 1.7E-5 (1.0E-2) | Drill string fails to shut down on detection of flammable gas | Design includes drill string automatic trip, given high flammable gas; auto |
| | | Drill string break caused by buckling or overtorque from jamming in waste on obstructions. | trip exhauster on: nigh fiammable gas level, high rate of change of flammable gas level, or high dome pressure. |
| Loss of drill bit cooling leads to waste fire. (Note 3) | 1.9E-5 (1.0) | N ₂ bypass flow occurs w/subsequent failure to stop drilling. | Design includes auto-trip of drill string on low N2 purge gas flow. |
| | | Drill string N ₂ cooling holes plug w/subsequent failure to stop drilling. | Annunciation of high N2 purge gas temperature is provided. |
| | | | Test N ₂ purge system for bypass leakage is done. |
| | | | Test results show that N2 leakage from drill string with section O rings not installed is acceptable. |
| | | | Test results show that partial plugging of some of the drill bit nitrogen cooling holes leading to localized overheating without calling for trip on low nitrogen flow is not of concern. |
| | | | Analysis shows that drill string tear without break is very unlikely. |
| | | | WHC analysis shows that overheating of nitrogen source gas is not of concern because in passage through hoses and into drill string, the nitrogen will not exceed waste ambient temperature. |
| Excessive drilling rpm leads to waste | 6.8E-6 | The rpm setting is too high. | Control over speed setting is provided. |
| hre. (Note 3) | (1.1E-2) | Trip of drill string on high rpm fails. | Design includes auto-trip of drill string on excessive rpm. |

TABLE E-1 (cont) SUMMARY OF ACCIDENT QUANTIFICATION RESULTS

ora a

TABLE E-1 (cont) SUMMARY OF ACCIDENT QUANTIFICATION RESULTS

| Accident | Frequency (1/Activity) | Dominant Failures | Controls Credited for Mitigative Frequency Quantification |
|--|--------------------------------------|---|---|
| Excessive downward force on drill leads to waste fire. (Note 3) | 4.7E-5 (0.6) 3.7E-5 | Excessive downward force is used. Drill string fails to trip on excessive force. Walkdown detector fails to function. | Provide control over downward force used. Design includes: auto-trip drill string on excessive force; force detector and walkdown detector for all samples except last; force detector and bottom detector for last sample. Provide 1 of 1 system for detection of |
| waste fire. (Note 3) | (0.1) | operator fails to manually stop drill. | drilling penetration rate. Five minutes are available for operator to recognize slow penetration and stop drilling. |
| Collision of vehicle with riser leads to dome fire. (Note 1) | 2.5E-7 (0.01) | Vehicle hits riser. Riser ruptures fuel tank on vehicle. | Provide controls over operation of vehicles on top of tank. |
| Blowout of HEPA exhaust filter. | 7.8E-6 (not a dome fire) (7.8E-6) | Excessive differential pressure occurs across HEPA filter. | The exhauster HEPA is verified to be sufficiently free from loading at the start of drilling so that subsequent drilling will not result in plugging. The exhauster is shut down given high HEPA differential pressure. |
| Drop of drill string | 3.0E-5 | Failure to install lockwrench. | Lock wrench is used. |
| Drop of drill string results in breach of | (1.0) 15E 5 (not a doma fire) | Operator opens toot clamp. | Operator training is given. |
| tank bottom liner. | (0.5) | Operator opens foot clamp. | Operator training is given. |
| Spill from dropped sampler in shielded receiver. (Similar accidents are spills from drops of sampler in X-ray machine or in cask, but they are of lower frequency.) | 4.3E-9 (not a dome fire) (0.11) | RLU drops sampler. Sampler rotary valve fails given drop of sampler. Ball valve in shielded receiver inadvertently left open. | Controls over operation of RLU are used. Use control to close ball valve in shielded receiver. |

NE VISION

| f | | | |
|---|-----------------------------------|--|---|
| Accident | Frequency (1/Activity) | Dominant Failures | Controls Credited for Mitigative Frequency Quantification |
| Waste from contamin åted drill string interior is released aboveground. | 3.1E-7 (not a dome fire) (1.0) | N2 hydrostatic systems fail to both drill string and shielded receiver. Operators fail to detect contamination during removal of drill string sections. | Use leak test N2 hydrostatic systems for both drill string and for shielded receiver. Use unique connections for N2 hydrostatic systems for both drill string and shielded receiver. Verify N2 hydrostatic supply to both drill string and shielded receiver during activation of hydrostatic mode of N2 supply. |
| | | | Monitor for radioactive contamination during removal of drill- string sections. |
| Waste from contaminated drill string exterior is released aboveground. | 2.2E-6 (not a dome fire) (1.0) | Ineffective wash of drill-string exterior surface before removal of drill string is ineffective. | Provide water wash of drill-string exterior surface before removal of drill string. |
| | | Operators fail to detect contamination during removal of drill string sections. | Monitor for radioactive contamination during removal of drill string sections. |
| Drill string break causes drill string fire (Note 4) | 6.3E-9 (3.3E-4) | N ₂ bypass flow occurs with sub. failure to stop drilling. | Test N ₂ purge system for bypass leakage is done. |
| | (0.02 2) | Sampler chevron seal fails. | Test results show that N_1 leakage from drill string with section O-rings not installed is acceptable. |
| | | | Presence of sampler with chevron seal in drill string to prevent flammable gas movement into drill string. |
| Drop Grapple in Drill String | 3.0E-11 | Sampler seal fails. | Sampler seal is used. |
| | (6.6E-4) | N ₂ hydro to DS fails. | DS N ₂ hydro is used. |
| | | Grapple is dropped. | Load cell and current control for hoist motor. |

TABLE E-1 (cont) SUMMARY OF ACCIDENT QUANTIFICATION RESULTS

Notes:

1. Probability of GRE to dome space will lower the frequency of this accident sequence.

1875 (2771 1875 (2771 278 (2771

- 2. BOM spark tests may lower the frequency of this accident sequence.
- 3. Probability of waste burn will lower the frequency of this accident sequence.
- 4. Probability of gas source to drill string with drill string in waste is estimated as 1.0.
- 5. WHC documentation on aluminum-on-aluminum spark potential will lower the frequency of this accident sequence.
- 6. Source of gas is from depressurization of sample.

Some accident frequencies will be lowered as a result of consideration of the gas-release-event (GRE) frequency, the waste-burn probability, and of the Bureau of Mi (BOM) testing results. Section E.4.0 discusses the quantitative effect of thes equency-reduction phenomena.

E.4. ACCIDENT SEQUENCE QUANTIFICATION CONSIDERING PHENOMENOLOGICAL FACTORS

This section considers the impacts of the phenomena frequencies or probabilities upon the accident-sequence quantification considered previously by this appendix.

As shown in Appendix L of this report the probability that tank dome space lower flammability limit (LFL) is exceeded as a result of a GRE when a random spark-initiating event occurs is 1.4E-5 based on the mean time that LFL is exceeded. This value is incorporated into the Table E.2 frequencies for all accidents in which the spark source is essentially random in time. This situation is true for all except one accident. For this accident, the "truck over an unused penetration" accident, we assume the spark source is always present and use 7.0E-5 as the probability that a burn occurs. As shown in Appendix L, 7.0E-5 is the probability that LFL is exceeded concurrent with positive tank pressure.

The waste-burn probability quantification and BOM testing are not completed, so we are not yet able to use them to lower our previous accident frequencies.

The frequencies presented in Table E-2 are frequency/yr for rotary-mode drilling into a single tank, assuming 2 drilling activities per yr. They are not accident frequencies for an individual tank farm or for the entire Hanford site. Mitigated frequencies precede the unmitigated frequencies listed in parentheses.

| Accident | Frequency/yr |
|---|-------------------|
| Lightning strike causes dome fire. (Note 1) | 1.1E-10 (2.3E-10) |
| Spark from drill bit contacting rock in waste. (Note 2) | 6.6E-7 (0.2) |
| Spark from failure of exhauster heater leads to dome fire. (Note 1) | 0 (0) |
| Spark from assembly/disassembly of drill string sections: | 7.6E-5 total |
| Drill string in dome (Notes 1 and 2) | (1.65E-2 total) |
| Drill string in waste (Notes 2 and 4) | 1.1E-9 (2.3E-7) |
| | 7.6E-5 (1.65E-2) |

TABLE E-2

SUMMARY OF ACCIDENT QUANTIFICATION RESULTS INCLUDING PHENOMENOLOGICAL FACTORS

TABLE E-2 (cont) SUMMARY OF ACCIDENT QUANTIFICATION RESULTS INCLUDING PHENOMENOLOGICAL FACTORS

| Accident | Frequency/yr |
|--|--------------------------------------|
| Drop of equipment from crane onto exhauster leads to dome fire. (Note 1) | 9.5E-12 (1.9E-10) |
| Failure of exhauster fan causes spark, leads to dome fire. (Notes 1 and 5) | 6.4E-10 (6.4E-10) |
| Tear in exhauster duct releases flammable gas to unqualified electrical equipment leading to dome fire. (Note 1) | 1.4E-9 (5.6E-8) |
| Leaking riser penetration releases flammable gas to unqualified electrical equipment leading to dome fire. | 1.8E-7 (2.8E-5) |
| (Note 1) | |
| (Note: For this accident we calculated the probability that flammable gas in excess of LFL reaches equipment randomly placed on the tank top). | |
| Flammable gas is in drill string/shielded receiver during sampler handling leading to dome fire. (Notes 2 and 4) | 1.9E-6 (2.8E-3) |
| Flammable gas in shielded receiver with shielded receiver isolated from drill string results in aboveground fire. (Notes 2 and 6) | 2.8E-5 (not a dome fire) (3.0E-4) |
| Flammable gas in X-ray machine leads to above ground fire. (Notes 2 and 6) | 1.6E-9 (not a dome fire) (1.6E-8) |
| Flammable gas in cask leads to above ground fire. (Notes 2 and 6) | 2.8E-5 (not a dome fire) (2.8E-4) |
| Flammable gas in riser during drilling leads to dome fire. (Notes 1 and 2) | 3.1E-10 (2.8E-5) |
| Drop of tool into open riser leads to dome burn. (Note 1) | 1.4E-9 (2.8E-9) |
| Drill string break in dome leads to dome fire. (Note 1) | 4.8E-10 (2.8E-7) |
| Drill string break in dome leads to DS fire. (Note 4) | 1.3E-8 (6.6E-4) |
| Loss of drill bit cooling leads to waste fire. (Note 3) | 3.8E-5 (2.0) |
| Excessive drilling rpm leads to waste fire. (Note 3) | 1.4E-5 (2.2E-2) |
| Excessive downward force on drill leads to waste fire. (Note 3) | 9.4E-5 (1.2) |
| Slow drilling penetration rate causes waste fire. (Note 3) | 7.4E-5 (0.2) |
| Collision of vehicle with riser leads to dome fire. (Note 1) | 7.0E-12 (7.0E-12) |
| Blowout of HEPA exhaust filter. | 1.6E-5 (not a dome fire) (1.6E-5) |
| Drop of drill string results in breach of tank bottom liner. | 3.0E-5 (not a dome fire) (1.0) |
| Spill from dropped sampler in shielded receiver. | 8.6E-9 (not a dome fire) |
| (Similar accidents are spills from drops of sampler in X-ray machine or in cask, but they are of lower frequency.) | (0.22) |

TABLE E-2 (cont) SUMMARY OF ACCIDENT QUANTIFICATION RESULTS INCLUDING PHENOMENOLOGICAL FACTORS

| Accident | Frequency/yr |
|--|--------------------------|
| Waste from contaminated drill string interior is released aboveground. | 6.2E-7 (not a dome fire) |
| | (2.0) |
| Waste from contaminated drill string exterior is released | 4.4E-6 (not a dome fire) |
| aboveground. | (2.0) |
| Truck over unused riser | 2.1E-8 (1.4E-4) |
| Penetration intentionally open | 4.2E-8 (1.4E-5) |
| Drop of drill string | 6.0E-5 (2.0) |
| Drop of Grapple in drill string | 6.0E-11 (1.3E-3) |

Notes:

- 1. Frequency lowered using GRE LFL probability.
- 2. BOM spark tests may lower the frequency of this accident sequence.
- 3. Probability of waste burn will lower the frequency of this accident sequence.
- 4. Probability of gas source to drill string with drill string in waste is estimated as 1.0.
- 5. WHC documentation on aluminum-on-aluminum spark potential will lower the frequency of this accident sequence.
- 6. Source of gas is from depressurization of sample.

August 8, 1996

Lightning Strikes Tank (Section 4.1.1.2)

| | _ | _ | _ | | _ | _ | _ | |
|--|-----------|---------|------------|---------|------------|----------|-------------|----------|
| Outcome | | | | | | | Sequence of | Concern |
| Probability | 1.0000E00 | | 5.0000E-04 | | 4.1000E-06 | | 4.1000E-06 | |
| No Dritting in Stormy Weather | | | | | | 5.00E-01 | | 5.00F-01 |
| Driffing Underway (hns per acthrs per year) | | | | 1.00E00 | | note) | 1.64E-02 | |
| Lightning Stilkes Tank (1/yr) | | 1.00E00 | | | | 5.00E-04 | | |
| H2 Present | | | | 1.00E00 | | | | |

August 8, 1996

E-13

Lightening Strikes Tank (Section 4.1.1.2) Unmitigated Event Tree

| H2 Present | Lightning Strikes Tank (1/yr) | Dritting Underway (tws per actities per year) | Probability | Outcome |
|------------|----------------------------------|--|-------------|-------------|
| | | | 1.0000E00 | |
| | 1.00E00 | | | |
| | | | 5.0000E-04 | |
| 1.00600 | note I | 1.00E00 | | |
| | 5.00E-04 | note) | 8.2000E-06 | Sequence of |
| | | 1.64E-02 | | Concern |

Drop onto Exhauster (Section 4.1.1.4)

| H2 In Exhauster | Control Lifts with Crare | Crane Drops Equipment onto Exhauster | Probability | Outcome |
|-----------------|--------------------------|---|-------------|-------------|
| | | | 1.0000E00 | |
| | 1.00E00 | | | |
| | | | 5.0000E-02 | |
| 1.00E00 | | 1.00E00 | | |
| | 5.00E-02 | note a | 3.4000E-07 | Sequence of |
| | • | 6.BOE-06 | | Concern |

August 8, 1996
| · . | | | | • | | | |
|---|-----------|------------------------|--|---|--|---|--|
| | | | | | | _ | |
| Outcome | | Sequence of Concern | | | | | |
| Probability | 1.0000E00 | 6.8000E-06 | | | | | |
| Crane Drops Equipment onto Echauster | 1 MFm | 6.80E-06 | | | | | |
| H2 In Exhauster | | 1.00€00 | | | | | |
| | | | | | | | |
| | | | | | | | |
| | | | | | | | |

E-16

•

August 8, 1996

.

Spark from Extrauster Fan (Section 4.1.1.4)

| H2 In Exhauster | Spark from Fan Faiture | Probability | Outcome |
|-----------------|------------------------|-------------|-------------|
| | | | |
| | | 1.0000E00 | |
| | 1.00E00 | | |
| 1.00E00 | note k | 2.3000E-05 | Sequence of |
| | 2.30E-05 | | Concern |



Sequence of Concern Outcome 4.8500E-05 9.90006-04 0.0000E00 0.0000£00 1.0000E-02 5.0000E-04 0.9000E-01 1.0000€00 0.0000E00 Probability Unquelified Equipment Bum Back to Too Chose to Hose Eshauster and Dome 1.00E00 1.00E00 1.00E00 Exhauter Falls to Operate Tear Reisman H2 1.DOEOD Operator Detects Tear \$.00E-02 1.00£00 1.00E00 Big Tear during Operation 1.005-03 1.00E00 Small Tear during Installation 10-308.8 H2 In Exheuster Norr Operation 1.00600

Exhauster Hose Tear (Section 4.1.2)





4.5000E-04 0.0000E00

1.00E00

9.005-01

Solf-at

Sequence of Concern

1.5000E-07

\$.0000E-05

1.00E00

1.00500

1.00E-01

8880

3.00E-85

E-19

1.006-02

Exhauster Hose Tear (Section 4.1.2) Urmitigated Event Tree

Sequence of Concern Sequence of Concern Outcome 0.0000E00 9.9000E-04 0.0000E00 9.0000E-03 1.0000E-03 0.0000E00 0.0000E00 9.9000E-01 Probability Burn Back to Extrauster and Dome 1.00E00 1.00E00 Exhasutar Falls to Operate 9.00E-01 nofe v 1.00E-01 Tear Releases H2 1.00E00 1.00E00 Big Teer during Operation 1.00E-03 1.00E00 Smell Tear during Installation 1.00E-02 9.90E-01 H2 in Exhauster Normal Operation 1.00E00

Leaking Penetration (Section 4.1.3

| H2 h Riser | Equip. Too Close to Riser with 1 Inch Leak | Spark and Ignition | Probability | Outcome |
|------------|---|--------------------|-------------|-------------|
| | | | 1.0000E00 | |
| | 1.00E00 | | | |
| | | | 0.0000E00 | |
| 1.00E00 | note c | | | |
| | 6.50E-03 | | 6.5000E-03 | Sequence of |
| | • | 1.00E00 | | Concern |

WHC-SD-WM-SAD-035, Rev. 0-a

| | Leaking Penetration (Section 4.1.3) Unmitigated Event Tree | Spark and ignition Probability Outcome | 0.000E00 | 1.00E00 Sequence of Concern |
|--|--|--|----------|-----------------------------|
|--|--|--|----------|-----------------------------|

WHC-SD-WM-SAD-035, Rev. 0-a

Operators Fail to Seal Riser/Pit All Truck Ignition Sources Not > 36 Inches From Riser/Pit Path Truck Parked Over Unused Spark and Ignition Probability Outcome Penetration 9.9700E-01 1 9.97E-01 2.8500E-03 2 1.00E00 9.50E-01 0.0000E00 3.00E-03 3 5.00E-02 1.5000E-04 Sequence of 4 1.00E00 Concern

Drift String Truck Parked Over Unused Tank Penetration (Section 4.1.3)

1023

Drill String Truck Parked Over Unused Tank Penetration (Section 4.1.3) Unmitigated Event Tree

| Truck Parked Over Unused Penetration | Operators Fail to Seal Riser/Pit | All Truck Ignition Sources Not > 36 Inches From Riser/Pit | Spark and Ignition | Path | Probability | Outcome |
|---|-------------------------------------|--|--------------------|------|-------------|------------------------|
| | <u> </u> | | | 1 | 0.0000E00 | |
| 1.00E00 | 1.00E00 | 1.00E00 | 1 00600 | 2 | 1.0000E00 | Sequence of Concern |

١

WHC-SD-WM-SAD-035, Rev. 0-a

Sequence of Concern Outcome 1.5000E-03 4.0850E-01 0.0000E00 5.0000E-01 Probability Path ~ + 67 -Spark and Ignition 1.00E00 Operators Fall To Locate Equipement Beyond 36D 3.00E-03 9.97E-01 Penetration Intentionally Opened During Dritting 5.00E-01 5.00E-01 Perform Driffing Operation 1.00E00

Intentional Penetration Opening During During (Section 4.1.3)

August 8, 1996

WHC-SD-WM-SAD-035, Rev. 0-a

E-25

Intentional Penetration Opening During Dritting (Section 4.1.3) Unmitigated Event Tree

| Perform Driffing Operation | Penetration Intentionelly Opened During Driffing | Operators Fall To Locate Equipement Beyond 36D | Spark and Ignition | Path | Probability | Buccho |
|-------------------------------|---|---|--------------------|------|-------------|-------------|
| | | | | | | |
| | | | | • | 5.0000E-01 | |
| | 5.00E-01 | | | - | | |
| | | | | • | 0.0000E00 | |
| 1,00600 | | | | 2 | | |
| | 5.00E-01 | 1.00500 | | • | 5.0000E-01 | Sequence of |
| | | r | 1.00500 | 2 | | Concern |

H2 Sample Handling In Drill String (Section 4.1.4)

| H2 in Waste | N2 Hydro to Drill and Shield Receiver | RLU Drops Sampler in Drill String | Probability | Outcome |
|-------------|--|--------------------------------------|-------------|-------------|
| | | | 1.0000E00 | |
| | 1.00E00 | | | |
| | | | 6.9000E-04 | |
| 1.00E00 | event Trycho" App. D | 1.00E00 | | |
| | 6.90E-04 | note e | 9.6600E-07 | Sequence of |
| | - | 1.40E-03 | | Concern |

| f String (Section 4.1.4) Unmitigated Event Tree | | | |
|---|--------------------------------------|-----------|------------------------|
| te Handling in Dri | Outcome | | Sequence of Concern |
| H2 Samp | Probability | 1.0000E00 | 1.4000E-03 |
| | RLU Drops Sampler in Drill String | 1.00E00 | note 6 1.40E-03 |
| | H2 In Waste | | 1.00E00 |

H2 Sample Handling in Isolated Shielded Receiver (Section 4.1.4)

| Perform Sample Drilling | N2 Hydro DS and SR Failed Prior | Sampler Rotary Valve Failed Prior | RLU Drops Sampler in SR | Probability | Outcome |
|----------------------------|------------------------------------|--------------------------------------|----------------------------|-------------|------------------------|
| | | | | 9.9000E00 | |
| | 1.00E00 | 9.00E-01 | 1.00500 | 1.1000E00 | |
| | | 1.00E-01 | note 1/11 | 1.4300E-05 | Sequence of Concern |
| 1.10E01 | event 'hvdri' Ann D/11 | | 1.00E00 | 6.9300E-04 | |
| | 6.30E-05 | | 1.30E-05 | 9.0090E-09 | Sequence of Concern |

E-29

29402.74 5570.754 2011.1

ġ.

L. Mariana Regenera



이 전 전 가 내

H2 In X- Ray Machine (Section 4.1.5)



E-31

H2 in X- Ray Machine (Section 4.1.5) Unmitigated Event Tree



E-32

H2 in Cask (Section 4.1.5)

| Place Sampler in Cask | N2 Hydro DS and SR Failed Prior | Sampler Rotary Valve Failed Prior | RLU Drops Sampler In Cask | Spark and Ignition | Probability | Outcome |
|--------------------------|------------------------------------|--------------------------------------|------------------------------|--------------------|-------------|------------------------|
| L | L | 9.00E-01 | ······ | | 9.9000E00 | |
| | 1.00E00 | event 'mt' Ann. D | 1.00E00 | | 1.1000E00 | |
| <u></u> | | 1.00E-01 | note f/11 | | 1.0000E00 | |
| 1.10E01 | | | 1.30E-05 | 1.00E00 | 1.4300E-05 | Sequence of Concern |
| | event 'hydro'/11 App. 6.30E-05 | D. Rest of this branch | ls Negligible | | 6.9300E-04 | |

H2 in Cask (Section 4.1.5) Unmitigated Event Tree



E-34

Narkovin Ozracia States Saka A Totara

Vehicle Collides with Tank Riser (Section 4.1.9)

| Outcome | | | 2 | | 5 | | 7 Sequence of | Concern |
|-------------------------------|-----------|---------|------------|---------|------------|----------|---------------|----------|
| Probability | 1.0000E00 | | 2.5000E-06 | | 2.5000E-06 | | 2.5000E-07 | |
| Spark and Ignition of Fuel | | | | | | 1.00E00 | note n | 1.00E-01 |
| Riser Impects Fuel Tank | | | | 1.00E00 | | estimate | 1.00E-01 | |
| Vehicle Impacts Riser | | 1.00E00 | | | | 2.50E-00 | | |
| H2 Present | | | | 1.00E00 | | | | |

Vehicle Collides with Tank Riser (Section 4.1.9) Unmitigated Event Tree

| H2 Present | Riser Impacts Fuel Tank | Spark and Ignition of Fuel | Probability | Outcome |
|------------|----------------------------|-------------------------------|-------------|-------------|
| | | | 1.0000E00 | |
| | 1.00E00 | | | |
| | | | 1.0000E-01 | |
| 1.00500 | estimate | 1.00E00 | | |
| | 1.00E-01 | note n | 1.0000E-02 | Sequence of |
| | | 1.00E-01 | | Concern |

Drill String Break in Dome (Section 4.2.1)

| H2 in Dome | Detect H2 and Shut Down Drill | Drill String Break: Jam/torque or buckle | Spark and Ignition | Probability | Outcome |
|------------|----------------------------------|---|--------------------|-------------|------------------------|
| | | | | 1.0000E00 | |
| 1.00E00 | | 1.00E00 | | 1.7000E-03 | |
| | event hydr App. D 1.70E-03 | Estimated | | 0.0000E00 | |
| | | 1.00E-02 | 1.00E00 | 1.7000E-05 | Sequence of Concern |

E-37

WHC-SD-WM-SAD-035, Rev. 0-a

Drill String Break in Dome (Section 4.2.1) Unmitigated Event Tree

| robability Outcome | 0000E00 | | 0000E00 | | 0000E-02 Sequence of |
|---|---------|---------|---------|-----------|----------------------|
| Spark and Ignition | • | | 0 | | • |
| Drift String Break: Jam/torque or buckle | | 1.00E00 | 1 | Estimated | 1.00E-02 |
| H2 In Dome | | | | 1.00500 | |

August 8, 1996

E,

E-38

Tool Drop into Open Riser (Section 4.2.3)

| | | | | | | | | |
|-------------------------------------|-----------|---------|------------|-------------|-----------|---|-------------|---------|
| Outcome | | | | | | | Sequence of | Concern |
| Probability | 1.0000E00 | - | 5.0000E-05 | | 0.0000600 | | 5.0000E-05 | |
| Path | • | - | C | × | E | • | • | • |
| Spark and Ignition | | | | | | | | 1.00E00 |
| Non-SpatkingTool Control Failure | | | | 5.00E-01 | - | | 5.00E-01 | |
| Tool Drop | | 1.00E00 | | P. alian de | | | | |
| H2 Present | | | | 1.00E00 | | | | |

Tool Drop Into Open Riser (Section 4.2.3) Unmitigated Event Tree

| Concern | | 7) | 1.00E00 | | |
|-------------|-------------|------|--------------------|-----------|------------|
| Sequence of | 1.0000E-04 | • | | 1.00E-04 | |
| | 0.0000E00 | 2 | | Estimate | 1.00E00 |
| | 0 mmEnn | | | | |
| | | - | | 1.00E00 | |
| | 1.0000E00 | • | | | |
| 1 | | | | | |
| | | | | | |
| Outcome | Probability | Path | Spark and Ignition | Tool Drop | H2 Present |

Ignition in Drill String Riser Sleeve (Section 4.2.4)

WHC-SD-WM-SAD-035, Rev. 0-a

| H2 Present | Detect H2 and Shutdown Drill | N2 Flow to Sleeve Fails | Spark (BOM Tests) or Hot Spot (analysis) | Probability | Outcome |
|------------|-------------------------------------|----------------------------|---|-------------|------------------------|
| | | | | 1.0000E00 | |
| 1.00E00 | - | 1.00E00 | | 1.7000E-03 | |
| | event 'hydri' in App. D 1.70E-03 | | | 0.0000E00 | |
| | | 6.50E-03 | 1.00E00 | 1.1050E-05 | Sequence of Concern |

E-41

| Outcome | | Sequence of Concern |
|---|-----------|------------------------|
| Probability | 0.0000E00 | 1.0000E00 |
| Spark (BOM Tests) or Hot Spot (analysis) | | 1.00E00 |
| N2 Flow to Sleeve Fails | | 1.00E00 |
| Detect H2 and Shuldown Driff | | 1.00E00 |
| H2 Present | | 1.00€00 |

Ignition in Driff String Riser Sleeve (Section 4.2.4) Unmitigated Event Tree

Overheat Waste: Excessive RPM (Sections 4.2.6 and 4.4)

| Perform Drilling | Excessive RPM | Fail to Detect High RPM and Trip Drill | Waste Burn | Probability | Outcome |
|------------------|---------------|---|------------|-------------|-------------|
| | - | | | 1.0000E00 | |
| | 1.00E00 | | | | - |
| | | | | 1.1000E-02 | |
| 1.00E00 | | 1.00E00 | | | |
| | | | | 0.0000E00 | |
| | 1.10E-02 | event tpm' App. D | | | |
| | | 6.20E-04 | | 6.8200E-06 | Sequence of |
| | | | 1.00E00 | | Concern |

WHC-SD-WM-SAD-035, Rev. 0-a

Overheat Waste: Excessive RPM (Sections 4.2.6 and 4.4) Unmitigated Event Tree

| Pertorn Drilling | Excessive RPM | Waste Burn | Probability | Outcome |
|------------------|---------------|------------|-------------|-------------|
| | | | 1.0000E00 | |
| | 1.00E00 | | | |
| 1 OPEAN | | | 0.0000E00 | |
| | note h | | | |
| | 1.106-02 | | 1.1000E-02 | Sequence of |
| | | 1.00F00 | | Concern |

Overheat Waste: Excessive Force All but Last Sample (Sections 4.2.6 and 4.4)

| Perform Sample Drilling | Excessive Force on All but Last Sample | Force and Walkdown Detectors or Trip Fails | Waste Burn | Probability | Outcome |
|-------------------------|---|---|------------|-------------|------------------------|
| <u></u> | 0.505.01 | I | | 1.0450E01 | |
| 1 10501 | 8.0024/1 | 1.00500 | | 5.5000E-01 | |
| | 5.00E-02 | | | 0.0000E00 | |
| | | 7.90E-05 | 00E00 | 4.3450E-05 | Sequence of Concern |

E-45

| Dict-came | | | | | pence of | Concern |
|---|-----------|----------|-----------|---------|---------------|---------|
| Probability 0 | 1.0450E01 | | 0.0000E00 | | 5.5000E-01 Se | |
| Weste Burn | | | L | | 1 | 00Em |
| Excessive Force on All but Last Sample | | 0.50E-01 | 1 | | 5.00E-02 | - |
| Perform Sample Driffing | | | | 1.10501 | , | |

Overheet Waste: Excessive Force All but Last Sample (Sections 4.2.6 and 4.4) Unmitigated Event Tree

Overheat Waste: Excessive Force Last Sample (Sections 4.2.6 and 4.4)

| Pertorm Last Sample Drilling | Force and Bottom Detectors or Trip Falls | Excessive Force On Last Sample | Waste Burn | Path | Probability | Outcome |
|---------------------------------|---|-----------------------------------|------------|------|-------------|-------------|
| | | | | • | 1.0000E00 | |
| | 1.00E00 | | | ~ | | |
| | | | | , | 7.5050E-05 | |
| 1.00E00 | | 9.50E-01 | | v | | |
| | TOPE OF | | | c | 0.0000E00 | |
| | /.806-05 | | | n | | |
| | - | 5.00E-02 | | | 3.9500E-06 | Sequence of |
| | | | 1.00E00 | • | | Concern |

WHC-SD-WM-SAD-035, Rev. 0-a

Overheat Waste: Excessive Force Last Sample (Sections 4.2.6 and 4.4) Unmitigated Event Tree

| Perform Last Sample Drilling | Excessive Force On Last Sample | Waste Burn | Path | Probability | Outcome |
|---------------------------------|-----------------------------------|------------|------|-------------|-------------|
| | | | • | 9.5000E-01 | |
| | 9.50E-01 | | - | | |
| | | | ¢ | 0.0000E00 | |
| 1.00E00 | | | V | | |
| - | 5.00E-02 | | ſ | 5.0000E-02 | Sequence of |
| | | 1.00E00 | 2 | | Concern |

Spark from DS Sections (Section 4.3.3)



August 8, 1996

E-49

No. 2010

| Perform Drilling Operations | DS In Waste or Dome | GRE In Dome or H2 In Waste | Sampler Chevron Seal Fails | DS Sections Bang and Spark | Probability | С жтө |
|--------------------------------|---------------------|-------------------------------|-------------------------------|-------------------------------|-------------|------------------------|
| L | J | ſ | l | 1 | 0.0000E00 | |
| | 5.00E-01 | | 1.00E00 | | 5.0000E-01 | |
| | | 1.00E00 | event 'seal' App. D | ſ | 0.0000E00 | |
| 1.00E00 | | | 3.30E-02 | note v 5.00E-01 | 8.2500E-03 | Sequence of Concern |
| | | | | | 0.0000E00 | |
| | 5.00E-01 | 4 . | ſ | | 0.0000E00 | |
| | | 1.00E00 | | ſ | 1.0000E00 | · · · |
| | | | 3.30E-02 | note v 5.00E-01 | 8.2500E-03 | Sequence of Concern |

Spark from DS Sections (Section 4.3.3) Unmittigated Event Tree

 $\begin{array}{c} \frac{1}{2} & \frac{1}{2} \\ \frac{1}{2} & \frac{1}{2} \\ \frac{1}{2} & \frac{1}{2} \\ \frac{1}{2} & \frac{1}{2} \\ \frac{1}{2} \\$

Drill String Break Drill String Fire (Section 4.3.5)

| Drill String Break: Jarr/Acrque or buckle | N2 Purge to DS: Falls or Bypass Flow | Chevran Seal in Sampler Fails | Spark and Ignition | Probability | Outcome |
|--|---|----------------------------------|--------------------|-------------|-------------|
| | | | | 5.0000E-03 | |
| | 5.00E-01 | | | | |
| Estmated | | | | 1.9000E-07 | |
| 1.00E-02 | | 1.00E00 | | | |
| | | | | 0.0000E00 | |
| | 1.80E-05 | event teef App. D | | | |
| | | 3.30E-02 | | 6.2700E-09 | Sequence of |
| | | | 1.00E00 | | Concern |
Drill String Break Drill String Fire (Section 4.3.5) Unmitigated Event Tree

| Drill String Break: Jam/torque or buckle | Chevron Seq in Sampler Falls | Spark and Ignition | Probability | Outcome |
|---|---------------------------------|--------------------|-------------|-------------|
| | | | 1.0000E-02 | |
| | 1.00E00 | | | |
| CSUMARO • • • • • | | | 0.0000E00 | |
| 1.005-02 | event bear App. D | | | |
| | 3.30E-02 | | 3.3000E-04 | Sequence of |
| | | 1.00E00 | | Concern |

August 8, 1996

E-52

Slow Penetration Rate (Section 4.4)

| Pertorn Dritting Operation | Contact Hard Object In Waste | Penetration Rate Detector Fails and No Operator Trip | Waste Burn | Probability | Outcome |
|-------------------------------|---------------------------------|--|------------|-------------|-------------|
| | | | | 1.0000E00 | |
| | 1.00E00 | | | | 1 |
| | | | | 1.0000E-01 | |
| 1.00E00 | | 1.00E00 | | | |
| | | | | 0.0000E00 | |
| | 1.005-01 | note u | | | |
| | | 3.70E-04 | | 3.7000E-05 | Sequence of |
| | | | 1.00E00 | | Concern |

WHC-SD-WM-SAD-035, Rev. 0-a

Stow Penetration Rate (Section 4.4) Unmitigated Event Tree

| Perform Dritting Operation | Comtact Hand Object In Waste | Waste Burn | Probability | Outcome |
|-------------------------------|---------------------------------|------------|-------------|-------------|
| | | | 1.0000E00 | |
| | 1.00E00 | | | |
| | | | 0.0000E00 | |
| 1.00500 | note tv | | | |
| | 1.00E-01 | | 1.0000E-01 | Sequence of |
| | • | 1.00E00 | | Concern |

WHC-SD-WM-SAD-035, Rev. 0-a

Overheat Waste: Loss of N2 Cooling to Drift Bit (Section 4.4)

| Falls, Bypess, Partial 1.000E00 Block, N2 Temp High 1.000E00 1.00E00 0.000E00 1.00E00 0.000E00 1.00E00 1.900E.05 | Concern | | 1.00500 | | |
|--|-------------|------------|---------|---|---------|
| Falls, Bypess, Partial Falls, Bypess, Partial Block, N2 Temp High 1.0000000 1.00000 1.0000000 1.000000 0.0000000 | Sequence of | 1.9000E-05 | | 1.90E-05 | |
| Failer, Biypease, Partial 1.0000E00 Block, N2 Temp High 1.0000E00 1.00E00 0.0000E00 | | | | event toot App. D | 1.00500 |
| Faller, Bypeese, Partial Block, N2 Temp High 1.00E00 | | 0.0000E00 | | | 1 22722 |
| Fake, Bypees, Partial Block, N2 Temp High 1.0000E00 | | | | 1.00500 | |
| Faults, Bypess, Partial Block, N2 Temp High | | 1.0000E00 | | | |
| | | | | Falls, Bypess, Partial Block, N2 Temp High | |

| Outcome | | | : | Sequence of | Concern |
|----------------------|---|-----------|---|-------------|---------|
| Probability | | 0.0000E00 | | 1.0000E00 | |
| Waste Burn | | | | | 1.00E00 |
| N2 Purge Flow to DS: | Faits, Bypess, Partial Block, N2 Temp High | | | 1.00E00 | I |
| Perform Driffing | | | | 1.00E00 | |

Overheat Waste: Loss of N2 Cooling to Drift Bt (Section 4.4) Unmitter we Event Tree

E-56

Hit Rock in Waste (Section 4.4)

| Pertorn Dr illing Operation | Contact Rock In Waste | Penetration Rate Detector and Nitrogen Systems Fall | Spark and Burn | Probability | Outcome |
|---------------------------------------|-----------------------|---|----------------|-------------|-------------|
| | | | | 1.0000E00 | |
| | 1.00E00 | | | | i |
| | | | | 1.0000E-01 | |
| 1.00E00 | | 1.00E00 | | | |
| | A BOUL | | | 0.0000E00 | |
| | 1.00E-01 | event prrate' App. D | | | |
| | • | 3.30E-06 | | 3.3000E-07 | Sequence of |
| | | | 1.00E00 | | Concern |

August 8, 1996

Hit Rock In Waste (Section 4.4) Unmitigated Event Tree

| Perform Drifting Operation | Contact Rock In Waste | Spark and Burn | Probability | Outcome |
|-------------------------------|-----------------------|----------------|-------------|-------------|
| | | | 1.0000E00 | |
| | 1.00E00 | | | |
| | | | 0.0000E00 | |
| 1.00500 | note w | - | | |
| | 1.00E-01 | | 1.0000E-01 | Sequence of |
| | | 1.00E00 | | Concern |

WHC-SD-WM-SAD-035, Rev. 0-a

Dropped Drill String Penetrates Tank Bottom (Section 4.6.2)



E-59

August 8, 1996

August 8, 1996

E-60

Wt. Exceeds Frisbee Friction Drill String Impacts Tank Bottom Tank Bottom Penetrated **Drop Drill String** Probability Outcome 0.0000E00 5.0000E-01 1.00E00 5.00E-01 0.0000E00 1.00E00 5.00E-01 5.0000E-01 Sequence of Concern 1.00E00

Dropped Drill String Penetrates Tank Bottom (Section 4.6.2) Unmitigated Event Tree

Sample Split from Shielded Receiver (Section 4.8)

| Left Probability Outcome | 1.1000E01 | | 1.4300E-04 | | 1.4300E-06 | | 4.2900E-09 Sequence of | Connarn |
|--|-----------|---------|------------|-----------|------------|----------|------------------------|----------|
| Ball Valve in SR (Open | | | | | | 1.00E00 | | 9 ONE 03 |
| Drop Causes Sample Rotary Valve to Open | | | | 1.00E00 | | estimate | 1.00E-02 | - |
| RLU Drops Sample | | 1.00E00 | | 4 JE 2144 | | 1.305-05 | | |
| Lft Sample Into Shielded Receiver | | | | 1.f0E01 | | | | |

| Jit Sample Into Vielded Receiver | Drop Causes Sample Rotary Valve to Open | Bait Valve in SR Left Open | Probability | Outcome |
|-------------------------------------|--|-------------------------------|-------------|---------|
| | | | 1.0890E01 | |
| | 9.90E-01 | | | |
| 01 | | | 1.1000E-01 | |
| | 1.00E-02 | - | | |

Sample Split from Shielded Receiver (Section 4.8) Unmitigated Event Tree

WHC-SD-WM-SAD-035, Rev. 0-a

August 8, 1996

E-62

Spill Waste from Contaminated Drill String Interior (Section 4.8)



August 8, 1996

Splil Waste from Contaminated Dritt String Interior (Section 4.8) Unmitigated Event Tree

| Disassemble Drill String | Waste Sufficiently Liquid to Flood DS | Waste Sticks in DS During DS Removal | Probability | Outcome |
|--------------------------|--|---|-------------|-------------|
| | | | 0.0000E00 | |
| 1 00500 | _ | | 0.0000E00 | |
| | 00500 | | 1.0000E00 | Sequence of |
| | | 1.00E00 | | Concern |

Split Waste from Contaminated Drift String Exterior (Section 4.8)

١

| Disassemble Drill String | Wash removes Waste from DS Exterior | Visual Detection of Waste on DS Exterior | Detect Contaminated DS Sections | Probability | Outcome |
|--------------------------|--|---|------------------------------------|-------------|------------------------|
| L | 4.00500 | L | | 1.0000E00 | |
| 1.00E00 | 1.00E00 | | | 1.0000E-01 | |
| | estimate 1.00E-01 | 1.00200 | 1.00500 | 5.0000E-03 | |
| | | 5.00E-02 | note s 4.40E-04 | 2.2000E-06 | Sequence of Concern |

Senar Sectors Senar Sectors March Sectors March Sectors March Sectors August 8, 1996

.

| Outcome | Sequence of | Concern |
|---|-------------|---------|
| Probability | 1.0000E00 | |
| Visual Delection of Waste on DS Exterior | | |
| Wash removes Waste from DS Extertor | | |
| Disassemble Drill String | | 1 MEM |

Split Waste from Contaminated Drift String Extertor (Section 4.8) Unmitigated Event Tree

E-66

| Outcome | | | | | Sequence of | Concern |
|---|-----------|---------|------------|---------|-------------------------|----------|
| Probability | 1.0000E00 | | 3.0000E-03 | | 7.8000E-06 | |
| Fail to trip Exhauster on Filter Delta P | | | | 1.00E00 | event 'pate32' h App. D | 2.60E-03 |
| Old Fitters Plug | | 1.00E00 | | - | 3.00E-03 | |
| Perform Drilling | | | | 1.00E00 | | |

HEPA Blowout (Section 4.8.1)

WHC-SD-WM-SAD-035, Rev. 0-a

August 8, 1996

HEPA Blowout (Section 4.8.1) Unmitigated Event Tree

| Perform Driffing | Old Fittens Plug | Fail to the Exhauster on Filter Delta P | Probability | Outcome |
|------------------|------------------|--|-------------|-------------|
| | | | 1.0000E00 | |
| | 1.00E00 | | | _ |
| | | | 3.0000E-03 | |
| 1.00500 | | 1.00E00 | | |
| _ | 3.00E-03 | event pate32' App. D | 7.8000E-06 | Sequence of |
| | | 2.60E-03 | | Concern |

WHC-SD-WM-SAD-035, Rev. 0-a

Drop Grapple in Drill String Unmitigated (Section 4.3.2.1)



Drop Grapple in Drill String Mitigated (Section 4.3.2.1)



| Notes for Acc | cident Event Trees |
|---------------|---|
| Note a. | 6.3E-5/lift [NUREG -0612] x 25 sq m/465 sq m [area exhauster to tank] x 2 lifts/activity [estimate] = 6.8E-6 |
| Note b. | Big Tear Discovered after Shutdown prior to Restart |
| Note c. | Assumes 25 Risers and 10 Pieces of Equipment x value from Mathematica Calculation $[2.6E-5]$ for total value of 6.5E-3 |
| Note d. | Assumes 1 Riser and 10 Pieces of Equipment x value from Mathematica Calculation [5.6E-4] |
| | (superceded by addition of inlet stack to inlet HEPA path: removed from analysis) |
| Note e. | 0.02 prob. Hangup per activity [WHC info] x (0.05 + 0.02) [failure detect hangup in 2 min + fail to recover by change motor direction] = 1.4E-3 |
| Note f. | 0.02 prob. Hangup per activity [WHC info] x (0.005 + 0.002) [failure detect hangup watching through window + fail to recover by change motor direction] =1.4E-4 |
| Note g. | Rubber gasket failure 2.9E-7/hr [blue book] x 2196 hours = 6.4E-4 |
| Note h. | 0.001 first time (commission) + 0.01 second time = 0.011 |
| Note i. | freq. Lightning strike 5E-4/yr [WHC-SD-SARR-027 Rev 0] |
| Note j. | 144 hrs per act/8766 hrs per yr = 1.64E-2 yr/act |
| Note k. | spark modeled as bearing seizure leading to fan impeller break and collision with fan housing |
| | 1.6E-6/hr [NPRD-91 frictional bearings] x 0.10 shaft breaks [estimate] x 144 hrs operation = $2.3E-5$ |
| Note 1. | removed from analysis |
| Note m. | 1.5E-3/tank/yr [LANL 241SY101 PRA Page D-5] x 144 hr per act/8766 hr per year = 2.5E-5/act |
| Note n. | 0.1 for fuel fire in non-qualified area [CA] |
| Note o. | 6.3E-5 drops/lift [NUREG 0612] x 20 lifts/act [estimate] = 1.2E-6 (includes operator error) |
| Note p. | prob 1/3 jacks fail + prob mis-set 1 of 3 jacks + prob drive off ramp= 3 (72 hrs*9E-6/hr [NPRD-91 shaft locking collar]) + 3(0.001) + 0.001 = 5.9E-3 |
| Note q. | ash blockage, LANL 241SY101 PRA Page C-46: 3.8E-4/yr/2 act/yr = 1.9E-4 (removed from analysis) |
| Note r. | LANL 241SY101 PRA Page D-4: 5.2E-3/tank yr/2 act per yr = 2.6E-3 |
| Note s. | 0.0001 (commission) + 5E-6/hr*1hr [WSRC-TR-93 rad monitor Table 6] +0.0003 [I&C] + 3E-5/hr*1hr = 4.4E-4 |
| Note t. | Assuming new NH ₃ detection system is one-of-two for success with reliability equivalent to H_2 detection system |
| Note u. | Assume 1/1 penetration trip design and use value calculated for failure to trip with the single force detector (7.4E-3) and use 0.05 for operator backup trip for total of 3.7E-4 |
| Note v. | Engineering Judgment |
| Note w. | 0.2 probability for needing rotary core drilling while sampling *0.5 probability of hitting hard layer = 0.1 total value |

E.5. REFERENCES

- 1. L. Abeyta and B. Lopez, "Computer Code SANET, Version 2.0," Science and Engineering Associates, Inc., Albuquerque, NM (October 1994).
- 2. J. M. Butner and J. L. Darby, "Quantification of Accident Sequences for Rotary Mode Core Drilling," Los Alamos National Laboratory report TSA10-CN-SA-TH-089 (January 11, 1996).
- 3. Alan D. Swain, "Accident Sequence Evaluation Program Human Reliability Analysis Procedure," US Nuclear Regulatory Commission report NUREG/CR-4772 (February 1897).
- 4. G. W. Hannaman and A. J. Spurgin, "The Systematic Human Action Reliability Procedure (SHARP)," Electrical Power Research Institute report EPRI NP3583, Project 2170-3, Interim Report (June, 1994).
- 5. J. L. Darby and J. M. Butner, "Reliability of Systems for Rotary Mode Core Sampling," Los Alamos National Laboratory report TSA10-CN-SA-HAZ-004 (February 15, 1996).



F.1. INTRODUCTION

Figure F-1 illustrates the phenomena occurring at a drill bit tooth for_normal cutting and for the case with a dull drill bit that is not cutting. Also, the figure shows some of the nomenclature used in this appendix.



Fig. F-1. Sketch illustrating the cutting process of the drill bit.

When the bit is cutting well [Fig. F-1(a)] there is a zone in front of the bit tooth where most of the work is taking place. The material being drilled is compressed and fractured. The zone in front of the teeth where the compression and fracturing takes place is at a high pressure compared to the region where the purge gas flows. This pressure difference causes the chips to be squeezed out along the cutting face of the bit. The purge gas cools the fracture zone and the chips and then carries the chips radially outward. The energy that goes into compressing and fracturing the material is mostly converted to thermal energy. If the drilling is proceeding efficiently, most of the energy goes into the chips. Frictional energy dissipation occurs on the wear-flat as it slides over the substrate and on the cutting face where the chips are squeezed past. Conceptually, the maximum temperature can occur in the chips because of the mechanical-to-thermal energy conversion as the material is compressed and fractured, or along one of the bit surfaces where frictional heating occurs.

In inefficient cutting with dull drill bit teeth [Fig. F-1(b)], penetration may be difficult to achieve. In this scenario, the drill bit and waste in the waste friction zone attain similar temperatures in a short time. In this case, the maximum waste temperature would occur at the interface between the bit and the waste.

A hazard associated with high waste temperatures is the exothermic propagating chemical reaction. Waste ignition could result with unacceptable consequences. In

Appendix G safe waste temperature limits are established to prevent propagating exothermic chemical reactions in the tank during rotary core sampling. A limit of 150°C has been set for bulk material in the vicinity of the drill bit. A limit of 180°C has been set for temperatures that may occur over small regions (a few chips in diameter) and for short times. These temperature limits are intended to prevent exothermic chemical reactions from being initiated. Laboratory experiments are conducted with the prototype drill bit by Westinghouse Hanford Company (WHC).¹ These tests were designed to provide the data necessary to set operational limits and to demonstrate that the waste temperature limits will not be exceeded.

The purpose of this appendix is to evaluate the test results and determine an adequate basis for setting limits for the drilling operation to ensure that temperature limits will not be exceeded. Other objectives are to discuss how laboratory test results are applicable to real conditions in tanks.

F.2. BACKGROUND

Some drill bit thermal modeling and experimental work has been done for oil and geothermal well drilling. In these cases, the interest in thermal modeling was driven by the need to maximize the life of synthetic diamond cutters. The rate of degradation of these cutters is a function of temperature. Prakash and Appl² developed a finite element model of an oil well drilling bit and the rock being cut. Glowka and Stone³ developed a finite element model for a bit for geothermal applications that examined only the friction component of energy dissipation in which the wear-flat slides over the substrate. In both of these papers, the primary interest was in the temperature of the drill bit. The temperature of the material being cut was only considered as it related to the bit temperature.

In addition to the papers on thermal modeling of drill bits, there are papers that discuss the mechanical energy input per unit volume of material removed. In the rock drilling literature this concept goes by the term "specific energy." Teale⁴ discusses this concept and provides numerical values for several materials. Units are expressed as in. lbf/in.³ or MJ/m³. Teale shows that the minimum specific energy for a rock-drilling operation is equal to the compressive strength (σ) of the material. This is shown by the following equation, where "SE" is the specific energy, "F" is force, "A" is area, "x" is distance, and "length" and "force" are generalized dimensions:

$$SE\left(\frac{\text{length} \cdot \text{force}}{\text{length}^3}\right) = \frac{\int F \, dx}{\int A \, dx} = \frac{\int \sigma A \, dx}{\int A \, dx} = \sigma\left(\frac{\text{force}}{\text{length}^2}\right)$$
(F-1)

Applying a pressure equal to the compressive strength through some distance gives a specific energy that is numerically equal to the compressive strength if an appropriate system of units is used. Teale⁴ shows examples of specific energy for laboratory tests of rotary drilling in several different sandstones. Specific energies ranged from near the compressive strength of about 6,000 in. lbf/in.³ to about 20,000 in. lbf/in.³. Rabia⁵ showed that the specific energy in a field drilling operation was often higher than laboratory drilling operations by a factor of about 2. He shows values of specific energy for field drilling of different rock types that range from 16,000 in. lbf/in.³ to 66,000 in. lbf/in.³. These values represent the total power input to the drill bit and include frictional losses as well as the energy required to fracture the material being cut.

A quantity similar to specific energy has been defined for machining.^{6,7} This quantity is known as the "unit horsepower," and is defined as the horsepower needed to remove one cubic inch of material per minute (hp/in.³/min). Operating at less than optimal conditions increases the energy needed to perform the machining operation. Handbook values range from 0.25 hp/in.³/min for soft nonferrous metals and alloys, including free-machining brass, cast aluminum, and zinc alloy to 1.3 hp/in.³/min for hard ferrous alloys. The difference between machining a ductile material and a brittle material is in the nature of the fracturing. To machine a ductile material, shearing takes place along a plane. To drill a brittle material, the material is compressed to the compressive strength, causing a number of fractures to propagate into the material.

The transformation of mechanical to thermal energy takes place along the fracture surfaces of the material being drilled or machined. The temperature that occurs along these fracture surfaces may be significantly higher than the average temperature reached when the thermal energy is distributed over the entire volume of the material removed. The analyses in this appendix consider only these average temperatures. Some of this thermal energy may be transferred to the drill bit, some to the substrate, and some to the coolant, although most of the thermal energy probably heats the chip. If all the mechanical energy needed to remove a chip is converted to thermal energy in the chip, the average temperature rise can be calculated. Table F-1 gives the values for specific energy and unit horsepower discussed above. The original unit is given in bold face. The temperature rise is calculated using values for specific heat and density given by WHC for Tank BY-104 waste simulant.⁸

F.3. SAFETY ENVELOPE TESTS

The operating safety envelope has been defined by Keller.⁸ The safety envelope was obtained by performing experiments with a prototype drill bit and waste simulants. Thermal properties of these simulants were expected to represent typical properties of salt cake. The rotational speed, downward force, and nitrogen purge flow rate were varied. The number of data for a given set of parameters was limited. At maximum operating parameters (55 rpm, 1170 lbf, 20 scfm) the maximum drill bit temperature rise was observed to be 35°C. The maximum waste temperature in single-shell tanks (SSTs) was considered to be 93°C. The critical drill bit temperature

TABLE F-1 'TERATURE VALUES FOR SPECIFIC ENERGY AND UNIT HORSEPOWER

| | Specific | Specific | Unit | Average Temp. |
|--|-----------------------------|------------|----------------------------|---------------|
| | Energy | Energy | norsepower | Kise |
| Muterial | (in. lbf/in. ³) | (MJ/m^3) | (hp/in. ³ /min) | (°C) |
| Marble ⁴ | 1,200,000 | 8,274 | 3.03 | - 6,049 |
| ▶ ble⁴ | 70,000 | 483 | 0.177 | 353 |
| l _nant sandstone⁴ | 13,780 | 95.0 | 0.035 | 69.5 |
| Darley dale sandstone | 7,210 | 49.7 | 0.0182 | 36.3 |
| concrete ^{4*} | | | | |
| Monkato stone ⁵ | 1,581 | 10.90 | 0.00399 | 8.0 |
| Rockville granite ⁵ | 22,826 | 157.38 | 0.0576 | 115 |
| Dresser basalt ⁵ | 54,204 | 373.72 | 0.37 | 273 |
| Soft nonferrous metals ⁶ | 99,000 | 683 | 0.25 | 499 |
| Hard ferrous alloys ⁶ | 514,800 | 3,549 | 1.3 | 2,595 |

was set to 150°C. Thus, the drill bit temperature increase was limited to 57°C. Because the maximum experimentally measured drill bit increase was 35°C, the rotary-mode core sampling (RMCS) operations were considered to be justified (Ref. 8).

Testing by Keller⁸ did not measure the waste chip temperature but the drill bit surface temperature. Keller also performed out-of-limit tests. In these tests, the system is operated with maximum operating conditions and a loss of nitrogen; an increase in downward force or speed is simulated. These tests are used to establish the shut-down period after an alarm on operating parameters is received. Our assessment of the envelope testing indicated that the most important parameter for keeping the drill bit below the critical temperature at which an exothermic reaction can occur is the nitrogen purge flow. Loss of nitrogen flow at maximum operating conditions can heat the drill bit up to 57° C in 5 to 10 seconds. The most important parameter, applied torque, was not measured by Keller.

The data from the tests performed by Keller⁸ were found to be insufficient to determine the safety envelope parameters. Recently, Witwer¹ conducted more bounding tests using dry pumice blocks with new procedures and instrumentation. These new envelope tests included torque measurements and numerous temperatures, including purge gas inlet and outlet temperature, bit surface and side temperatures, drill string temperatures, chip and pumice temperatures, exhaust air temperatures, etc.

The new test series included thermocouple temperature measurements on two drill bit teeth. Additional thermocouples were placed within the pumice in the path of the drilling and in the purge gas. Some tests used a stainless-steel sheath over some of the thermocouples within the pumice. The bit depth, rpm, down force, and torque were also measured. Tests were done with different down forces. Also, in some of the tests, the purge gas was turned off to simulate a loss of purge gas flow condition. The pumice blocks used as a test material were very abrasive, and the drill bits wore quickly. The penetration rate was good at the start of the test and then decayed to zero as the drill bit teeth wore down. These tests had a period with reasonably good cutting followed by a transition from good cutting to no cutting, and finally, a period in which the worn drill bit was spinning without cutting. Test conditions are summarized in Table F-2. The details of these test results are discussed by Witwer.¹

The first five tests were performed without nitrogen flow to assess the rate of increase of the bit temperature. The rest of the tests are aimed to determine the safety limits as well as understand the drilling process. Some tests are performed to address partial plugging in the drill bit. These tests were performed by plugging four of the six purge holes.

F.4. ANALYSES METHODOLOGY

The analysis portion of this report is broken down into two major parts. Figure F-2 shows an energy flow diagram. The mechanical-to-thermal energy conversion discussion makes up the first of these major parts. This part is the key for understanding the drilling thermal processes. The second major part deals with the redistribution of heat. Results in this section tend to clarify and validate the results of the previous section. These pieces also are necessary to understand the experimental results.

TABLE F-2ENVELOPE TEST CONDITIONS

| Test ID | F (lbf) | V (rpm) | Q (scfm) | Drill Bit | Simulant |
|------------|---------|---------|----------|-----------------|----------|
| RETMP-09 | 1170 | 55 | 0 | Sharp | Pumice |
| RETMP-10 | 1170 | 55 | 0 | Sharp | Pumice |
| RETMP-10A | 1170 | 55 | 0 | Dull | Pumice |
| RETMP-10B | 1170 | 55 | 0 | Dull | Pumice |
| RETMP-11 | 1170 | 55 | 0 | Sharp | Pumice |
| TORQTST-5 | 1170 | 55 | 0,20,30 | Sharp | Pumice |
| TORQTST-6 | 1170 | 55 | 20 | Sharp | Pumice |
| TORQTST-7 | 1170 | 55 | 20 | Sharp | Pumice |
| TORQTST-9 | 650 | 30 | 30 | Sharp | Pumice |
| TORQTST-10 | 1170 | 55 | 21 | Sharp | Pumice |
| TORQTST-11 | 900 | 55 | 30 | Sharp | Pumice |
| TORQTST-12 | 900 | 55 | 30 | Sharp | Purnice |
| TORQTST-14 | 750 | 55 | 30 | Slug therm. | |
| TORQTST-16 | 750 | 55 | 30 | | |
| TORQTST-17 | 750 | 55 | 30 | 4 plugged holes | |
| TORQTST-19 | 750 | 30 | 30 | 4 plugged holes | |
| TORQTST-20 | 900 | 55 | 30 | 4 plugged holes | |
| TORQTST-21 | 750 | 55 | 30 | 4 plugged holes | |
| TORQTST-22 | 650 | 55 | 30 | 4 plugged holes | |
| TORQTST-23 | 750 | 55 | 30 | | Steel |

F.4.1. Derivation of Drilling Energy Conversion Equations

Figure F-3 shows a drawing of the drill bit. This bit is a core drilling bit with 18 teeth in 3 rings with 6 teeth per ring. The bit is sintered bronze material. The faces of the

teeth have a negative rake angle of about 30° for the two inner rings. That is, the cutting face of each tooth is inclined so that the top of the tooth leans toward the material being cut rather than away. A bit of this type is also referred to as a drag bit. This bit is a commercially available, proprietary design. The bit material is soft enough to prevent damage to a steel tank liner should one be encountered. Figure F-4 shows the free body diagram used in the analysis of this bit.

Summing forces in the X and Y directions gives the following equations:



Fig. F-2. Drilling energy flow diagram.



Fig. F-3. Drawing of Drill Bit.



Df is the down force.

Nf is the normal force at the cutting surface.

Uf is the upward force of acting on the wear-flat (surface of length B).

 Ff_B is the frictional force acting on the wear-flat surface.

Ff_A is the friction force acting on the cutting face.

Vc is the velocity of the chips relative to the face, assumed equal to V.

V is the cutter velocity.

Ft is the force from torque times radius.

A is the area of the face where the cutting takes place.

B is the area of the wear-flat.

Fig. F-4. Free body diagram of drill bit tooth.

Uf - Df + Nf sin
$$(30^{\circ})$$
 + Ff_A cos (30°) = 0 (F-2)

$$Ff_B + Nf \cos (30^\circ) - Tq/r_{avg} - Ff_A \sin (30^\circ) = 0$$
 (F-3)

where Tq is the torque and r_{avg} is the average radius.

These equations have four unknowns, so an additional two equations are needed to achieve closure. Four possible equations were identified:

$$Ff_B = Cf_B Uf$$
 (F-4)

$$Ff_A = Cf_A Nf$$
 (F-5)

$$Nf = \sigma A$$
 (F-6)

$$Uf = \sigma B \tag{F-7}$$

where Cf_B is the coefficient of friction between the wear-flat and the substrate, Cf_A is the coefficient of friction between the cutting face and the chips, and σ is the compressive strength.

It proved advantageous to use different combinations of Eqs. (F-4) to (F-7) under different circumstances and to determine different quantities. The coefficient of friction where the wear-flat slides over the substrate can be found from Eqs. (F-2) to (F-4) for cases where the bit becomes worn [Fig. F-1 (b)] and where Nf and Ff_A go to zero. Because the coefficient of friction is independent of the contact area, it should be reasonable to assume that this value is a constant, or nearly so, over the range of conditions.

The coefficient of friction on the cutting face (Cf_A) cannot be determined directly as is the case for the coefficient of friction on the wear-flat. However, the coefficient of friction is generally taken as a property of the contacting materials, and the contacting materials are the same at the cutting face and the wear-flat. The material (chips) sliding across the cutting face differs from the substrate in that the chips have been broken up and are no longer a single mass. However, these chips are still under high pressure (about 10,000 psi) and may act like a solid body as they are forced past the cutting face. That is, the chips, which are rough irregular solids, probably cannot move relative to one another when they are under high pressure. Thus, they probably act more like a solid than a fluid in the region between the material being cut and the cutting face. This suggests that Cf_A should at least be approximately equal to Cf_B.

Combining Eqs. (F-2) to (F-5) with the assumption that $Cf = Cf_A = Cf_B$ gives the following equations:

Nf =
$$\frac{-Df Cf + \frac{Tq}{r_{avg}}}{(\cos 30^\circ - 2 Cf \sin 30^\circ - Cf^2 \cos 30^\circ)}$$
 (F-8)

 $Uf = Df - Nf \sin 30^{\circ} - Nf Cf \cos 30^{\circ}$

As the area of contact between the material being cut and the cutting face (area A) decreases, Nf, the normal force on the cutting face, decreases. These equations do not have the correct behavior as the cutting face contact area decreases. Thus, they will break down as the bit wears and the penetration rate decreases. The initial attempts to analyze a full data set on a point-by-point basis with a spread sheet used these equations. As the test progressed and the cutting rate decreased, the lack of closure got larger. Alternative sets of equations were derived in response to this problem. Using Eqs. (F-8) and (F-9) with Eqs. (F-6) or (F-7) allows us to back out a value for σ , the compressive strength. These equations have been used for conditions with good cutting to obtain values for σ . The equation set that was ultimately used for the full test analysis includes σ as a parameter.

Combining Eqs. (F-2), (F-3), (F-4), (F-6), and (F-7), gives an alternative set of equations from which σ may also be found.

$$Ff_{A} = \frac{\frac{Cf Df}{Denom} + \frac{A}{B} Df \frac{\cos 30^{\circ}}{Denom} - \frac{Tq}{r_{avg}}}{\frac{Cf \cos 30^{\circ}}{Denom} + \frac{A}{B} \frac{(\cos 30^{\circ})^{2}}{Denom} + \sin 30^{\circ}}$$

where

Denom =
$$1 + \frac{A}{B} \sin 30^{\circ}$$

Uf = $\frac{Df - Ff_A \cos 30^{\circ}}{Denom}$ (F-11)

The value for the wear-flat area (area B) cannot be determined from the data because it is a function of the wear of the bit teeth. Also, as the bit wears, the assumption that the force on the wear-flat area is sufficient to match the compressive strength of the material may break down. Thus, Eqs. (F-10) and (F-11) have been used to validate the results of Eqs. (F-8) and (F-9) under conditions with good cutting. To

August 8, 1996

(F-9)

(F-10)

use these, a wear-flat dimension was assumed. If results from these two sets of equations are consistent, it is assumed that this supports the validity of the conditions for the use of these equations. Equations (F-10) and (F-11) also do not have the correct behavior as cutting face contact area decreases.

A third set of equations was derived using Eqs. (F-2), (F-3), (F-5), and (F-6). Then

| $Nf = \sigma A$ | | | - | (F-12) |
|------------------|--|--|---|--------|
| $Ff_A = Nf Cf_A$ | | | | (F-13) |

$$Uf = Df - Nf \sin 30^{\circ} - Ff_{A} \cos 30^{\circ}$$
 (F-14)

$$Ff_{B} = \frac{Tq}{r_{avg}} - Nf \cos 30^{\circ} + Ff_{A} \sin 30^{\circ} .$$
 (F-15)

These equations have the correct behavior as the cutting face contact area becomes small but do require that the value of s, the compressive strength, be known. Values for σ were obtained for conditions with good cutting using Eqs. (F-8) to (F-11) in analyses that examined only selected times during the tests considered. The value of σ obtained in this manner was used with Eqs. (F-14) and (F-15) to analyze one complete data set. It is assumed that σ is a constant. It is possible that as the bit wears, the size of the chips broken out will decrease, resulting in an increase in s. That is, more energy per unit volume is required to produce small particles than to produce large particles. In tests with simulants, a distribution of particle sizes was measured.⁹ These are already skewed toward small particle sizes. Over 80% of the particles by mass were smaller than 40 mm, and 50% were smaller than 20 mm. Less than 3% were greater than 100 mm. A 1-in./min cutting rate at 55 rpm gives a tooth penetration depth of 77 mm. If the maximum chip size is limited by the depth of cut as the bit dulls, there will be fewer large particles and more small particles. Because most of the particles were small in the samples measured, this suggests that eliminating the larger particles from the distribution will not make a large change in the size distribution by mass. Thus, the change in the energy required to remove a given volume of chips is not expected to change significantly as the penetration decreases, and the constant σ assumption should be reasonable. To put this argument another way, the measured particle distribution suggests that, for the most part, the material was being crushed to a fine powder rather than broken out as chips. Decreasing the scale of cutting will still result in the material being crushed to a fine powder, though the few larger chips that were being produced may no longer be produced.

The power terms can be derived by multiplying Eq. (F-3) by r_{avg} w. Doing this and rearranging gives

$$Tq \omega - Ff_{B} r_{ave} \omega - Nf r_{ave} \omega \cos 30^{\circ} + Ff_{A} r_{ave} \omega \sin 30^{\circ} = 0$$
 (F-16)

August 8, 1996

This equation assumes that the velocity in the downward direction is negligible compared to the velocity from the rotation. This also assumes that the velocity of the chips is equal to the velocity of the drill bit, which means that the thickness of the layer of chips passing across the drill bit face is equal to the thickness of the layer being cut. If a thinner layer passes across the bit face, the velocity would be higher. If a thicker layer passes over the bit face, the velocity would be lower. The first term is the power input, and the second term is the frictional dissipation between the wear-flat and the substrate. The third term is the power required to compress and fracture the material being cut, and the fourth term is the frictional dissipation at the location where the chips are squeezed across the cutting face. The third term may be readily shown to equal the compressive strength times the volumetric cutting rate. The difference in sign between the fourth term and the second and third terms should be noted. This sign difference means that the energy that goes into the frictional work as the chips are squeezed across the cutter face comes from the energy that was used to compress and fracture the chips. Further, the fourth term is a mechanical work term that acts on the bit and helps pull it along. Thus, it is possible for the third term, which is the energy to compress and fracture the chips, to be greater than the first term, which is the energy input to the drill string. Thus, at any moment, previously stored energy is released, and additional energy is stored as elastic compression.

To calculate the temperature rise of the chips, it has been assumed that the energy given by the third term minus the fourth term goes to heating the chips. The temperature rise from this energy should be practically instantaneous. The heat is generated along the fracture surfaces and is then transferred mostly into the chips, although some may be transferred to the purge gas and also to the bit. Including all of this energy is a conservative assumption. There may be additional elastic compressive energy that is recovered in the velocity of the chips as they decompress as well. These components, though expected to be small, contribute to the conservatism of this temperature rise calculation. That is, the calculated temperature increase will be greater than the actual temperature increase.

To calculate the maximum temperature of the chips, the calculated temperature rise has been added to the measured bit temperature. That is, the temperature of a given layer of material immediately before cutting was assumed to be equal to the bit temperature. This assumes that the temperature measured by the thermocouple on the bit teeth is indicative of the interface temperature, which is in turn indicative of the layer average temperature. This is a conservative assumption, at least for conditions with reasonably good cutting. As a layer of material is removed, a fresh, cooler layer is exposed. The frictional heating generated on the surface being cut raises the surface temperature to the interface temperature almost immediately. It takes a longer time for the rest of the layer to approach the surface temperature. Thus, the surface temperature should provide a conservative (high) estimate for the temperature of a layer just before it is cut. These equations have been applied in two different ways. First, in scoping analyses, portions of tests were analyzed. Average data points for a period of interest were picked off the plots. The mechanical-to-thermal-energy conversion equations derived as shown above were applied. In addition, other calculations that might clarify any the energy distribution terms were made. These included one-dimensional (1-D) slab calculations for cases with no cutting and temperature decay calculations for the chips produced by the cutting. Secondly, for one test, the energy conversion and distribution terms were calculated on a point-by-point basis. This served to validate our understanding of the phenomena involved. The details of the calculations are provided in Ref. 10.

F.4.2. Two-Dimensional (2-D) Transient Heat Transfer Model of Drilling Process

F.4.2.1. Model Description

1

In order to better understand the thermal energy transfer in the drilling process, a 2-D transient heat transfer model was also developed. The primary function of the model is to determine the maximum waste temperature as a function of drilling conditions. The model can then be used to determine the drilling condition limits to avoid reaching the stated temperature limits: 150°C for the substrate and 180°C for the chips. A parametric study can be done using a 2-D model in order to study the effect of extrapolated parameters involved in the drilling process. The model is a conduction/convection solution, where a conduction solution through the solid materials is coupled with a flow solution of the purge gas.

F.4.2.1.1. Conduction Solution. The general heat transfer equation solved is

$$\rho C_{p} \frac{\partial T}{\partial t} = -\nabla \cdot q + \frac{Q}{V} \quad . \tag{F-17}$$

The equation is solved numerically. The coordinate system is fixed to the drill bit; therefore, the drilled material is moving (in effect flowing) with respect to the grid. Therefore, as the bit penetrates, an additional term is added to the drilled material heat transfer solution. In effect, this is a steady state, constant property, incompressible flow solution. The material that "flows" through the upper boundary of the drilled material is added to the gas flow.

F.4.2.1.2. Flow Solution. The flow solution is solved separately from the conduction solution. The time frame of the flow solution is much shorter than for the conduction solution. The gas transit time through the system is much less than one second, and the transit time around the bit is on the order of milliseconds. Therefore, a steady-state solution can be used to approximate the flow at each conduction time step.

The flow solution is obtained by setting up a nodal 1-D flow path based on a paththrough system. Specified at each node are the node length, flow area, hydraulic diameter, the heat transfer area, and the surface temperature of the conduction nodes that surround the flow node. The heat transfer between the flow and conduction nodes is then given by the standard internal flow relations (h is interpolated between Re of 2000 and 3000 to provide a transition between the laminar and the turbulent flow):

$$Q_{w} = hA_{ht}(T_{w} - T_{g}) \text{ where:} \qquad h = h1 \text{ if } Re < 2000 \qquad (F-18)$$

$$h = h2 \text{ if } Re > 3000$$

$$h = h1 + (h2 - h1)(Re - 1000)/2000$$

$$h1 = \frac{4.364k}{D_{H}} \qquad h2 = \frac{.023k}{D_{H}} Pr^{.3} Re^{.8} \qquad (F-19)$$

 Q_w is used both in the conduction solution and to calculate the gas temperature as

$$Tg_{k} = \frac{1}{\dot{m}C_{p_{k}}} \left(\dot{m}_{k-1}C_{p_{k-1}}Tg_{k-1} + \dot{m}_{chip_{k}}C_{p_{chip}}T_{sub} + Q_{w_{k}} + Q_{gas_{k}} \right)$$
(F-20)

where: $\dot{\mathbf{m}}_{\mathbf{k}} = \dot{\mathbf{m}}_{\mathbf{k}-1} + \dot{\mathbf{m}}_{chip_{\mathbf{k}}}$.

 Q_{gas} is the energy deposition directly into the gas from the chips and T_{sub} is the temperature of the substrate at the location where the chips are being generated. Also, as chips are added to the flow caused by drilling, the specific heat and other properties of the flow are adjusted. It is assumed that the gas and chips mix uniformly and come instantaneously to equilibrium.

F.4.2.1.3. Energy Deposition. In the above equations, two important parameters have yet to be determined, Q_{gen} and Q_{gas} . First, it is conservatively assumed that all of the drill power is deposited in the system as thermal energy. Then it is assumed that this energy results from either breaking up the chips (Q_{drill}) or friction (Q_{fric}),

$$Q_{tot} = tw = Q_{drill} + Q_{fric} \quad . \tag{F-21}$$

The fractions are then broken up further into components that go to the gas, bit, or substrate. In the code, the chip energy and frictional components are determined by:

$$Q_{drill} = \frac{E_{sp}}{\dot{z}A_{drill}} \qquad \text{and} \qquad Q_{fric} = tw - Q_{drill} . \qquad (F-22)$$

Here E_{sp} is the specific energy of the drilled material. The fractional splitting of these powers into their gas, bit, and substrate components are specified in the input. The resulting energy deposition fractions are

$$Q_{\text{gen,bit}} = f_{\text{drill,bit}} Q_{\text{drill}} + f_{\text{fric,bit}} Q_{\text{fric}} , \qquad (F-23)$$

 $Q_{gen,sub} = f_{drill,sub}Q_{drill} + f_{fric,sub}Q_{fric}$, and (F-24)

F-15

August 8, 1996
$Q_{gas} = f_{drill,gas}Q_{drill}$.

These fractions are determined from benchmarking the experimental drilling.

F.4.2.1.4. Maximum Theoretical Chip Temperature. It is assumed that all of Q_{drill} goes into heating up a chip before energy is transferred ϵ lsewhere as

$$T_{chip} = T_{sub} + \frac{Q_{chip}\Delta t}{\rho V C_p} = T_{sub} + \frac{E_{sp}}{\rho C_p} \quad . \tag{F-26}$$

This number places a conservative bound on the maximum chip temperature and is meant only as an absolute upper limit. The chip temper ture will be discussed further in the results section.

F.4.2.1.5. Model Uncertainties and Limitations. There are two major model limitations: (1) The model is 2-D, so that the drill bit is smeared in the azimuthal direction. This does not allow for localized heating effects in the "tooth" or the effects of a plugged flow channel, and (2) The flow solution is one-species (chips/gas mix as one fluid). This does not allow for a separate calculation of chip temperature. Other minor limitations are that the model assumes constant properties, and the flow solution is steady state (which would only matter if very small conduction time steps are required).

There are several uncertainties in the model that may significantly affect the results. These include: drilling energy deposition fractions, friction energy deposition fractions, purge gas heat transfer coefficient, bit/pumice interface conductivity, and material properties. These uncertainties have been limited by benchmarking the model with the experimental drilling results.

F.5. RESULTS

The mechanical-to-thermal energy conversion equations show that the energy transfer to the bit can be resolved into three major components, two frictional components and the mechanical energy needed to compress and fracture the material into chips. The analyses show that the frictional components can account for the temperatures measured in the tests. The most significant of the frictional components produces heat between the drill bit and substrate. This heat will be transferred into the bit and into the substrate. The maximum temperature will occur at the interface. The drill bit thermocouples were located on bit teeth where they could measure the interface temperature. Thus, to the extent that the pumice was a limiting material for frictional dissipation, the bit temperature measurements from the test series on pumice blocks should be indicative of the maximum bulk material temperature. A comparison of temperatures between the test series done with pumice blocks as a test material and earlier tests with waste simulants as test materials show that the testing with the pumice blocks produced the highest temperatures. Thus, the coefficient of friction between the bit and the pumice was probably higher than the coefficient of friction between the bit and any of the other materials. Pumice block consists of hard particles in a binder material. This produces a very abrasive material that should have a high coefficient of friction. Thus, it is expected that the testing done on pumice blocks produced bulk temperatures that are limiting, or at least very nearly so.

The mechanical energy needed to fracture the material into chips causes an increase in the temperature of the chips. This temperature increase comes in addition to any temperature increase while the material was part of the substrate and was heated by the frictional dissipation. Thus, a conservative approach to finding the maximum chip temperature is to calculate a temperature increase for the chips and add the interface temperature, as measured on the bit, to this.

Two separate models are used to determine the substrate and chip temperatures. The first model is based on 1-D conduction solutions with adequate boundary conditions and consideration of a force-balance equation. A 1-D model is a first-order approach and is considered in Section 4.1. A 1-D model is used at the earlier phase of study to evaluate expected chip temperatures and to evaluate material properties important to the cutting process. In Section 4.2, a more detailed 2-D model is developed to simulate drilling. The 2-D model is more realistic and considers solution of a moving boundary of the drill bit. The energy equation is solved for the moving drill bit by considering the energy exchange between both solid-solid and fluid-solid. In the remainder of this section, the results of the 1-D model (Section 5.1) and then the 2-D model (Section 5.2) are discussed.

F.5.1. Results Obtained from 1-D Analysis

Two sets of comparisons are performed. The first set involved an evaluation of three tests, Tests TORQTST 6, 12, and 16. All of these tests used a rotational speed of 55 rpm, and 30 scfm purge gas flow (see Table F-2). The nominal down forces were 1170 lbf, 900 lbf, and 750 lbf. These three tests are analyzed to determine the forces involved in cutting, chip, drill bit, and substrate temperatures by solving the energy and force balance equations given in the previous sections. Details of the analysis are available in Reference 10. Parameters involved in analysis are averaged over a period in which cutting was either effective (constant penetration) or not effective (no penetration). In the second series, only Test 15 is analyzed. For Test 15, timeaveraged test parameters are not used but the analysis is performed for each transient recorded data point when the cutting was observed. This test defines the safety limits for RMCS.

F.5.1.1. Effective Waste Cutting

Table F-3 shows some key results obtained from three tests for times near the beginning of the tests when there was good penetration. Temperature increase estimates were obtained from the energy balances. Compressive strength was calculated from the force balance for these tests. The value for compressive strength determined from conditions with good cutting was about 10,000 lbf/in.² with an

| Test | Downforce (lbf) | Chip Temperature Increase from Energy Balance (°C) | Compressive Strength Estimates (lbf/in. ²) |
|-----------|--------------------|--|---|
| TORQTST6 | 1170 | 39.0-45.5 | 9227-11420 |
| TORQTST12 | 900 | 42.2-50.4 | 9482-9513 |
| TORQTST16 | 750 | 30.8-31.9 | 9230-9730 |

TABLE F-3ANALYSIS RESULTS, TESTS 6, 12, AND 16

estimated uncertainty of $\pm 2000 \text{ lbf/in.}^2$ (Ref. 10). Using $\sigma = 10,000 \text{ lbf/in.}^2$ gives a temperature increase of 45° C for compressing and fracturing the chips. These temperature increases are in addition to increases from frictional heating of the substrate before cutting.

The calculation using Eqs. (F-11) to (F-14) with a compressive strength of 10,000 lbf/in.² was performed for Test 15. Figure F-5 shows the measured bit temperature with calculated bit and chip temperatures. The figure shows that, with conservative assumptions, the increase in drill bit/substrate interface temperature approach but do not exceed the $\Delta T = 60^{\circ}$ C safety limit. The calculated chip temperature increase, about 83°C, is also lower than the safety limit of $\Delta T = 90^{\circ}$ C. This calculation presents the first evidence that when the operating parameters are limited with a downward force of 750 lbf, a rotational speed of 55 rpm, and a minimum purge flow of 30 scfm, a local waste ignition is not expected to occur when there is reasonable penetration. However, this conclusion is valid only if the hard waste layers expected in the tanks have similar properties to pumice in terms of both thermal and cutting processes of drilling.

If there is a reactive layer harder than pumice in the tanks, the truck has enough power available to produce higher torque. In this case, it is possible that chip and substrate temperature may exceed the safety limits. However, when the torque increases as a consequence of encountering a harder material, the penetration rate is expected to decrease. As mentioned in the next section, the penetration rate must be above 0.75 in./min. Therefore, higher torque should correspond to lower penetration rates. As a summary, if the waste layers in SST tanks are not harder than pumice, the proposed safety limits adequately protect against waste ignition hazards based on the results obtained from the 1-D model for effective cutting defined by a penetration rate higher than 0.75 in./min.

F.5.1.2. Frictional Heating with No Penetration

In all of the tests on the pumice blocks, the penetration rate slowed and then decreased to zero as the bit became worn. The data for the tests that used the limits of 750 lbf for the down force, 55 rpm for the rotational velocity, and 30 scfm for the purge gas flow, showed acceptable temperatures for conditions with a worn bit spinning on the substrate. The analyses showed that there was significant conduction into the bit for these conditions, and this limited the heat transfer to the slab, and thus the slab temperature. Results of Test 15 show that the drill bit and the corresponding substrate interface temperature is lower than the safety limit of $\Delta T = 60^{\circ}$ C, even when there is not good penetration and as long as nitrogen flow is available. However, this mode of operation in real drilling is not desired because the cause for no penetration will not be known. No penetration could be the result of encountering a hard reactive or nonreactive layer, metal, or other nonmetal object with different properties. The real waste is not expected to have a layer as hard as pumice. Note that every test done by Witwer¹ resulted in worn drill teeth. WHC engineers indicated to us that one sampling activity requires two rotary-mode core drillings out of a total of 10 samples. WHC engineers also expressed the fact that they have never failed to penetrate hard layers in Hanford tanks. Considering this experience as the available data for conditions of waste, it is not expected that the drill bit will encounter any reactive waste layer harder than pumice. However, if a waste layer is as hard as pumice, no penetration would be observed as experienced in laboratory testing.

One of the other reasons why the frictional mode of operation is not desired is that the transition from effective cutting to the frictional mode is not understood clearly. When the downward force was 1170 lbf, in Test 6, a large temperature variation, about 50° C, is observed in the transition region. Although the force is lowered to 750 lbf, the reasons for these higher temperature oscillations are not clearly understood. It is concluded that a penetration rate control is needed. The main reason for this control is to prevent drilling against reactive waste material that is harder than pumice. The lack of penetration rate control also prevents applied torque from being above the values observed in envelope testing.

F.5.1.3. Applicability of Envelope Testing to Real Conditions

Table F-4 compares properties of the pumice material to the properties of the waste simulant materials developed by WHC.⁸ All of the properties except the compressive strength calculated from drilling tests, the specific heats, and thermal diffusivities were measured by WHC.



Figure. F-5. Test 15 - Measured bit and calculated bit and chip temperatures.

F-20

August 8, 1996

WHC-SD-WM-SAD-035, Rev. 0-a

| TA | ٨B | LE | F-4 |
|----|----|----|-----|
|----|----|----|-----|

COMPARISON OF WASTE SIMULANT AND PUMICE TEST MATERIAL

| | Evaporator | BY-104 | Pumice |
|---|------------|--------------------------|--------------------------|
| Property | Bottoms | Simulant | Block |
| Bulk density (kg/m³) | 1560 | 1410 | 1600 |
| Unconfined compressive strength (psi) | 905 | 3,264 | |
| Calculated compressive strength from drilling tests (psi) | | | 10,000 |
| Direct shear strength (psi) | | 459 | |
| Penetrometer resistance (psi) | 23,520 | 34,507 | |
| Thermal conductivity (W/m K) | 2.32 | 1.67 | 1.03 |
| Specific heat (kJ/kg K) | | 0.97* | 0.88** |
| Thermal diffusivity (m ² /s) | | 1.221 x 10 ⁻⁶ | 7.315 x 10 ⁻⁷ |

* Estimate based on handbook value for predominate material.

** Handbook value.

The physical and thermal properties are comparable for pumice and the BY-104 simulant. The density times the specific heat products for the BY-104 simulant and the pumice block are within three percent of each other. Thus, the energy storage terms are very comparable. The lower thermal diffusivity coefficient of the pumice block tends to slow the spread of heat and keep temperatures higher. Thus, the pumice block testing produced temperatures that are conservative compared to the BY-104 simulant, and WHC has identified the BY-104 simulant as the limiting waste simulant.⁸

The pumice blocks consist of hard pumice grains in a binder material. This makes the pumice material very abrasive. The coefficient of friction between the bit and the pumice was evaluated as 0.4. The temperature at the bit/substrate interface where the bit temperatures were measured is determined primarily by frictional heating. The earlier series of tests⁸ with the BY-104 simulant showed lower temperatures. (Torque was not measured for this series of tests, so the effective compressive strength during drilling and the coefficient of friction cannot be calculated from the test data.) The temperatures suggest that the coefficient of friction between the bit and the BY-104 simulant was lower than the value deduced for the bit and pumice. The maximum average chip temperature is a function of two things, (1) the temperature of the waste material layer just before cutting and (2) the energy addition during the cutting process. The effects are additive. The temperature of the material before cutting is primarily a function of the frictional heating, and this was lower for the simulant materials.

The measured compressive strength for the BY-104 simulant is less than one-third the compressive strength deduced from the testing on the pumice blocks. Because the method of determination is very different, these values are not directly comparable. Rabia⁵ showed that specific energy values for a series of drilling tests were often about twice the laboratory measured values for the compressive strength during drilling conditions were three times the laboratory-measured value, the chip temperature rise from the compressing and fracturing operation would be less than the corresponding value for the tests on pumice blocks. Thus, this temperature should be lower for the BY-104 simulant for comparable drilling conditions.

Three reasons were identified for why the maximum temperature during drilling of the BY-104 simulant or comparable material should be lower than the corresponding maximum temperature during drilling in the pumice block. All three of these factors contribute independently of the others. Thus, the pumice block tests should be bounding for maximum temperature if the BY-104 simulant material can be considered as reasonably representative of the hard waste layers in the tanks. Operating limits based on test data from drilling in the pumice blocks should be conservative for materials that are limited by the BY-104 waste simulant properties. This applies to normal cutting and to off-normal conditions such as the bit becoming dull and no longer drilling and a loss of purge gas flow.

Hitting objects harder than pumice in the waste such as a rock, particularly over a small fraction of the bit face, could cause high temperatures if attempts were made to continue drilling. Drilling materials (nonreactive and reactive) harder than pumice may cause the applied torque to be increased, at least initially. Typical torque values measured in envelope tests varied between 35 and 85 ft lbf. There is power available to increase the torque. In an ideal situation, if the ratio of torque to the penetration rate defining the specific energy were controlled, safety limits would be implemented with extremely high confidence because the chip and substrate temperatures primarily depend upon the energy density (or unit horse power) which is linearly proportional to the torque and inversely proportional to the penetration rate. As discussed previously, the penetration rate must be monitored Torque measurements are not easy and reliable although and controlled. application of a torque limiter would be easy to implement. However, without torque measurements, operating safety limits involving downward force, rotational speed, nitrogen force, and penetration rate provide conservative protection for any waste ignition hazard.

The lower limit for the penetration rate of 0.75 in./min was set as protection against drilling in a material with a higher compressive strength than the pumice blocks. In the pumice block tests, there was an initial period with good penetration followed by a period with declining penetration, and finally a period with no penetration. The 0.75 in./min penetration rate represents a value at which the transition from the period of good penetration to the period with rapidly declining penetration occurs. In the pumice blocks, the nominal drilling parameters allow drilling with a bit that is at most slightly worn. If a layer of material with a higher compressive strength (which would give higher chip temperatures) is encountered during drilling, the penetration rate will drop below the lower limit. The operators could continue drilling only after a ten-minute cooling period following a trip initiated by out-of-tolerance down force on force, rotational speed, purge flow, or penetration rate. Additional attempts to restart drilling would begin with a cooled substrate. Then, the only energy component contributing to the temperature increase would be the energy to compress and fracture the chips. If the 0.75 in./min penetration rate could not be attained, drilling would be stopped automatically before the substrate could heat substantially and increase the chip temperatures.

Although this limit significantly reduces the possibility of producing chips at a temperature higher than the 180° C limit should a harder material be encountered, this does not guarantee that the temperature limit cannot be exceeded. If the limit is exceeded, only a very small volume of excessively hot chips will be produced for only a very short period of time. The chips are produced in an environment in which they are immediately surrounded by relatively cold nitrogen purge gas with a high convection heat transfer coefficient. Thus, they are expected to cool to below 180° C in much less than 0.1 s and have no detrimental effect.

F.5.1.4. Cases with Inadequate Nitrogen Flow

In several of the tests, the purge gas flow was shut off after the drill had stopped penetrating. With this condition, the energy balance becomes very simple, with only frictional energy dissipation and then conduction, either into the bit or into the substrate. Test 16 was one of the tests with this condition. Figure F-6 shows that the temperatures were measured and that the purge gas was shut off after the drill stopped penetrating. A 1-D slab analysis was performed for the time 60 s after the purge gas was turned off. To determine the effect of the conduction into the bit, an equivalent convection heat-transfer problem was determined by trial and error. In this case, a convection coefficient of 550 watt/m²K was needed to match the measured temperature. Thus, there was significant conduction heat transfer to the bit. The operating requirements specify that the drilling be stopped for a loss of purge gas. If all the frictional dissipation went into the substrate for even a short period of time before drilling was stopped, an overheat condition could occur. This result shows that there is sufficient heat transfer to the bit to provide some margin for shut-down time should a loss of cooling occur. If all the energy generated were going into the slab, the time for the surface temperature to rise above the limit would be small. The cooling capacity of the bit gives some margin for spin-down

August 8, 1996

time, should a loss-of-purge-gas-flow condition be encountered. The automatic controls on the drill system are to be set to immediately stop the drilling if a loss-of-purge-gas-flow condition occurs.

F.5.1.5. Chip Temperature Decay in Purge Gas

The maximum chip temperature occurs immediately after energy is added to the material to compress and fracture it. The chips are immediately entrained in the purge gas. This cools the chips quickly. This is important in that the chips have little time to heat any material other than the purge gas.

Two estimates were made for the cooling of these particles. WHC⁸ measured a particle size distribution for drilling in simulant materials. The slip velocity was calculated for the top particle size in each range reported by WHC. This was done by equating the weight of the particle and the drag on the particle because of the slip velocity. The drag was calculated from a standard correlation for the drag on a sphere as a function of the Reynolds number. An iterative solution results. Once the velocities had been obtained for each size, a standard correlation for the convective heat transfer was used to estimate the convective heat transfer coefficient (h). References 11 and 12 were checked for appropriate correlations. None were found with a range of applicability that included the small Reynolds numbers being calculated. A correlation with a stated lower limit of applicability of Re = 17 was selected. The extrapolation to lower Reynolds numbers is uncertain. The Reynolds numbers calculated from the slip velocities ranged from 3×10^{-6} to 6.4. The convection coefficient for each size and then a reduced temperature for each particle size as a function of time were calculated. A mass-averaged reduced temperature was also calculated as a function of time. This was used to estimate the chip-to-gas heat-transfer rate.

The second approach was similar, but a natural convection correlation was selected. The Grashof numbers calculated were well below the stated limit of applicability.

Using the forced convection correlation, the mass-weighted average particle goes through 63% of the change from the initial hot condition to the gas temperature in 0.1 s. The 150 micron particle size, which is the largest in the given distribution goes through 44% of its cooling in 0.1 s.

Using the natural convection correlation, the mass-weighted average particle goes through 98.7% of its cooling in 0.1 s, and the 150-micron particle size goes through 65% of its cooling in 0.1 s. Figure F-7 shows the temperature decrease for the mass-weighted average chips as a function of time for the cases discussed above.

Although both of these estimates use correlations outside the stated range of applicability, resulting in a large uncertainty for the above estimates, the conclusion that the chips cool to much less than the temperature limit in the order of 0.1 s should be valid.



Figure F-6. Test 16 – Measured temperature data.

F-25

August 8, 1996

Attempts were made to apply the above calculations to the purge gas conditions at the exit thermocouples to work back to ave ge chip temperatures immediately after cutting. These attempts did not produce any conclusive results. They do support the use of the 180° C limit for the chips because these calculations demonstrate that the chips cool from near this limit to well below this limit in less than 0.1 s.

F.5.2. Results of 2-D Finite Difference Transient Heat Transfer Modeling

F.5.2.1. Experimental Benchmarking

To begin, the model has been benchmarked versus two separate experimental tests: TORQTST16 and TORQTST23. These tests have printouts that contain most of the information necessary to obtain an energy balance (Ref. 13). In addition, TORQTST23 (which involves drilling into steel), contains reliable temperature measurements below the drilling surface. Initially, the energy deposition fractions (and to a small extent the material properties) were varied so that the experimental and computational results matched for each case separately. A fixed set of parameters was then determined combining the results of the two cases.

Next, the model was used to calculate temperatures for three additional tests: TORQTST15, TORQTST17, and TORQTST18. No model parameters were adjusted in assessing the model against these three tests. The calculated results conform fairly well with the experimental results for each test, especially considering the uncertainty in the experimental measurements. The results for some of these calculations are shown in Figure F-8. The calculated results are plotted alongside the experimental bit temperature. The experimental gas flow temperatures are not plotted, but they conform within 1°C of the calculated temperatures. TORQTST23 involved drilling into a steel slug; the experimental and calculated temperatures in the slug are also plotted for this case. The agreement between experimental measurements and calculated data for tests TORQTST15, 17, and 18 indicates that this model can be used for further parametric analysis.

F.5.2.2. Substrate Temperature

The primary result of this heat transfer analysis is the maximum substrate temperature as a function of time and drilling conditions. The substrate temperature limit has been set at 150° C; therefore, the model was used to determine how long it would take to reach this temperature under specific drilling conditions. The time to reach 120° C was also calculated to provide a conservative number. The time to reach each temperature was calculated as a function of torque, penetration rate, and initial waste temperature. The material properties were taken to be the same as for the experimental pumice, which is assumed to be a bounding condition (i.e. the waste is expected to have a higher specific heat and a lower specific energy).



Fig. F-7. Average chip temperature decrease predicted using (a) forced convection correlation and (b) natural convection correlation.



Fig. F-8(a). Comparison of calculated and measured temperatures of Test 16.



Fig. F-8(b). Comparison of calculated and measured temperatures of Test 23.

August 8, 1996



Fig. F-8(c). Comparison of calculated and measured temperatures of Test 15.



Fig. F-8(d). Comparison of calculated and measured temperatures of Test 17.



Fig. F-9. Time to reach 150°C as a function of torque for various penetration rates and an initial temperature of 22°C.



Fig. F-10. Time to reach 150°C as a function of torque for various penetration rates and an initial temperature of 90°C.

All of the results are obtained for a down force of 750 lb, a drill speed of 55 rpm, and a purge gas flow rate of 30 scfm. For this calculation, it was conservatively assumed that all of the energy goes into the bit and substrate (i.e., none goes directly to the gas). These results are plotted for the 150° C limit in Figs. F-9 and F-10. On the plots,

any condition to the lower left of a given line will produce a maximum temperature below 150°C. Some of the results are also listed in tabular form below in Table F-5.

In Table F-5 above "nr" means that the temperature was not reached in 5 minutes. In this case, the system reaches a quasi-steady state, where the energy input equals the energy taken away by the gas plus the energy conducted away to the system

| <u>,</u> | | Initial Temp = 22°C | | Initial Temp = 90°C | |
|-------------------|-------------------------------|---------------------------|---------------------------|---------------------------|---------------------------|
| Torque (ft-lb) | Penetration Rate (in./min) | Time to Reach 120°C | Time to Reach 150°C | Time to Reach 120°C | Time to Reach 150°C |
| | | | | | |
| 50 | 0.00 | nr | nr | 110 | nr |
| 50 | 0.75 | nr | nr | 152 | nr |
| 50 | 1.50 | nr | nr | nr | nr |
| 50 | 3.00 | nr | nr | nr | nr |
| 100 | 0.00 | 28 | 52 | 19 | 36 |
| 100 | 0.75 | 39 | 97 | 20 | 42 |
| 100 | 1.50 | 74 | nr | 22 | 53 |
| 100 | 3.00 | nr | nr | 26 | nr |
| 200 | 0.00 | 7 | 12 | - 4 | 9 |
| 200 | 0.00 | 0 = | 16 | - | |
| 200 | 0.75 | 8.3 | 15 | 5 | 10 |
| 200 | 1.50 | 11 | 20 | 5 | 11 |
| 200 | 3.00 | 20 | 47 | 6 | 15 |

TABLE F-5 TEMPERATURE INCREASE CALCULATIONS



Fig. F-11. Torque at which waste temperature reaches 150°C as a function of penetration rate.

"capacitance." Of course, because the system has a finite heat capacity, a steady state is not truly achieved until all of the energy is removed by the gas, but for a system this size, this will take place over the course of hours. So for this analysis, if the temperature has not reached its limit in 5 minutes, it is safe to assume that the temperature will not rise substantially over the course of several more minutes. In effect, "nr" means that the temperature will not be exceeded over any practical drilling time period.

In summary, the data in Table F-4 and in Figures F-9 and F-10 indicate that if the

torque is limited to 60 ft lbf, there are no conditions in which the 150° C limit will be violated. At 100 ft lbf, the temperature limit is not violated at a penetration rate of more than 3 in./min, and at low penetration rates, the limit is not exceeded for more than 30 seconds. A 200-ft-lbf torque leaves about a 10-second window to halt operations or for the drill to break before temperature limits are violated. Figure F-11 plots that maximum allowable torque as a function of penetration rate that will not result in exceeding the 150°C limit. This data can be used to determine the optimum limits in terms of torque, penetration rate, and time for the drilling process. Also, the torque and penetration rate limits should consider the maximum chip temperature, which is discussed below.

Several off-normal conditions were also considered. One case studied an increased purge gas flow of 55 scfm, with an initial temperature of 90° C and a penetration rate of .75 in./min. For a torque of 100 ft lbf the temperature did not even reach 150° C, as opposed to the 30 scfm case that reached 150° C in 42 seconds. For 200 ft lbf the temperature reached 150° C in 12 s as opposed to 10 s. From this result, it can be seen that more safety margin would be provided if the penetration is alarmed at 1 in./min value and the operator can take an action to increase the purge flow to 55 scfm before tripping the drill rig engine. However, the implementation of this suggestion as a control would be not easy because the time to reach the trip value may be so short.

Another case studied was a situation in which the system had reached a quasi-steady state at 3 in./min, 50 ft lbf, 750 lb, 30 scfm, and 55 rpm, and then the down force was suddenly increased to 4500 lb. Similarly, a case was studied in which the drill speed was increased to 110 rpm (as opposed to the down force). Temperature plots are shown for these two cases on Figures F-12 and F-13. Because the 4500-lb case increases the torque by a factor of 6, the temperature passes through 150°C in only a few seconds. The 110 rpm case doubles the torque, and the temperature reaches 150°C about 80 s after the drill speed is increased. The cases studied above indicate that the period in which the operator can take corrective actions after a trip is initiated can be very short. Therefore, the drilling operation must be shut down immediately (in the time required by the data acquisition system to ensure the



Fig. F-12. Temperature histories during an abnormal operation; speed is increased.



Fig. F-13. Temperature histories during an abnormal operation force is increased.

signal is real), when the downward force exceeds 750 lbf, rotational speed exceeds 55 rpm, or the purge flow becomes less than 30 scfm. Alarm values are not determined in this appendix. However, alarm values may be selected as long as they are lower than trip values by considering a proper waiting period.

F.5.2.2.1. Chip Temperature. As mentioned earlier, this model does not attempt to calculate the chip temperature. The only number calculated is the maximum theoretical chip temperature. This is equal to the substrate temperature at the location of drilling, plus a finite ΔT associated with the energy deposited in the chips caused by drilling of the material. If the very conservative assumption is made that all of the drilling energy goes to the chip, then this ΔT is equal to the specific energy of the material divided by the heat capacity. For the pumice used in the experiments, this ΔT is on the order of 30°C (based on a specific energy of 40 MJ/m³, which can be inferred from TORQTST18). Because the drilling conditions discussed above limit the substrate temperature to 150°C, then this also limits the chip temperature to 180°C (if it is assumed that the pumice is a limiting material on the basis of specific-energy-to-heat-capacity ratio).

However, it is still possible, although unlikely, that a waste material could be encountered that has a high enough specific energy to heat capacity ratio to cause more than a 30° C chip temperature rise. There is no way to absolutely avoid this possibility. The likelihood of exceeding 180° C could be decreased by setting the substrate temperature limit below 150° C, which would provide more margin for

chip ΔT . Also, torque and/or penetration rate limits could be imposed so that the drill is tripped as soon as such a material is encountered (although experiments have shown that the bit wears rapidly as harder materials are encountered, so these limits may not add much safety margin, especially if the detection/trip time is relatively slow). In either case, because the trip or bit wear is not instantaneous, a few chips may be produced that exceed the quoted maximum temperature. Fortunately, it has been determined that the chips cool very rapidly (<< 1 s), so that even if the chip temperature does exceed 180°C for a short time, the risk appears to be negligible.

There are two conclusions that can be drawn from the experimental results that support the contention that the chips are not hot for a substantial period of time; both involve the purge gas temperature. First, there is no abrupt rise in purge gas temperature as drilling begins. If a large fraction of the energy were going to the chips, and then to the gas, there would be an abrupt step increase in the purge temperature in the first few seconds. Because this increase is not seen, then the chips are either not getting very hot, or they are transferring energy quickly to the bit and substrate. Second, the downstream thermocouples do not show any evidence of chip energy deposition as the flow moves downstream. This implies that the chips have lost most of their energy before the flow reaches the thermocouples. The only other possibility is that the chips remain hot as they travel through the entire system. However, the measured gas temperatures are consistent with a total energy balance, and it is unlikely that the thermocouple would read the correct mean (1-species) temperature if the chips were hotter than the gas. Also, simple calculations confirm that the chip cool-down time is << 1 sec.

As mentioned before, ideally the control of the ratio of torque-to-penetration rate would provide reliable protection against waste ignition hazards. These two parameters, however, are not independent. Suppose a hard object/layer is encountered during drilling that is progressing with a good penetration rate. The penetration rate will decrease and the engine would increase the torque to continue drilling. At this point, torque may exceed values that were observed in envelope testing. If material is cut, some penetration can be achieved. Note that drill bit teeth are also starting to be worn out. Based on envelope testing results, teeth became worn down in about 1 minute. If material is much harder than pumice, this period is expected to be on the order of seconds because drill bit teeth material is designed to be worn down if metal objects are drilled. Once teeth get worn down, the penetration rate becomes zero, and the torque will decrease. Current design features only include the monitoring of the penetration rate. There is an automatic shutdown feature when penetration rate decreases below 0.75 in./min. Controlling only the penetration rate will provide protection against exceeding the temperature limits, although until teeth are worn out, there is a short period in which chips and substrate may get hotter than current limits. However, because of nitrogen-cooling capability, chips and substrate will to be cooled and not cause a propagating reaction. In order to further increase safety, a possible design improvement would be the

inclusion of a torque limiter or meter. The established controls provide adequate protection for waste ignition hazards.

F.6. CONCLUSIO!

The analyses in this appendix show that the temperature of the chips generated during the drilling tests may have exceeded the measured-temperatures from the testing. The chip temperatures during the pumice block testing did not exceed the $180^{\circ}C$ ($\Delta T = 90^{\circ}C$) limit [(and exceed the $150^{\circ}C$ ($\Delta T = 60^{\circ}C$)] limit for only a very brief time immediately after cutting. The $180^{\circ}C$ limit seems appropriate for the chips because they are small, the total volume of chips is small, and they are entrained in the relatively cool nitrogen purge gas immediately after being cut. This cools the chips quickly and limits their contact with the bulk material. These analyses also show that the pumice blocks used for a test material are a limiting material in comparison to the waste simulants tested. Thus, the analyses in this appendix support the conclusion that core drilling can be undertaken without exceeding emperature limits if the specified drilling parameter limits are maintained.

The tests analyzed were done on pumice blocks. The temperatures measured on the bits for the drilling tests on pumice blocks were higher than temperatures measured for tests with any of the simulant materials. The analyses suggest that the measured temperatures are dominated by frictional heating of the bulk material. Thus, the drilling tests on these pumice blocks were the most limiting tests for frictional heating. Because of the poor contact between the chips and the areas where the thermocouples were located, it is doubtful that the testing could measure chip temperatures. These are a function of the starting temperature of the material being cut and the compressive strength. The compressive strength calculated from these tests is also about three times higher than the highest value for compressive strength reported by WHC⁸ for waste simulant materials. The conversion of compressive strength to specific energy and to thermal energy is condition dependent, and the margin between compressive strength calculated from these tests and the compressive strength measured for the simulants is notable. One can conclude that the average temperatures resulting from the mechanical-to-thermal conversion process are probably lower with simulant materials than with pumice blocks.

In addition to examining normal drilling conditions, experiments were analyzed to determine the temperature response during off-normal conditions. These include a loss of purge gas flow and the spinning of a worn drill bit with no penetration. In these cases, there is sufficient sensible capacity of the system to prevent overheating if attempts to continue drilling are stopped within a reasonable time period. Operational limits set using the test data should be satisfactory if the material conditions encountered during the drilling are no more severe than seen in the drilling tests with the pumice blocks. Comparisons between the pumice properties and waste simulant materials suggest that the pumice blocks provided more severe drilling conditions than are likely to be encountered in the tanks. Should harder materials be encountered, higher temperatures could be reached.

To summarize, the properties of the pumice block used for drill testing produced more severe temperature conditions than the simulant materials produced for an earlier set of tests. If the simulants are reasonably representative of the materials in the tanks, the drilling tests on the pumice blocks were bounding for temperatures. If harder materials are encountered in the tanks, higher temperatures could be produced. A control is established to limit the penetration rate to 0.75 in./min. This control requires an automatic shutdown when the penetration rate is less than 0.75 in./min. This control provides the necessary protection to prevent a propagating waste reaction, although temperature limits may be exceeded for a very short period. Waste temperature limits established in this SA are already conservative, and no credit is considered for water in the waste. The analyses as well as simulants used in the laboratory experiments are on the conservative side. Therefore, it is concluded that controls established in this SA provide adequate protection against propagating an exothermic waste reaction as a result of rotary drilling if the established limits are not exceeded.

F.7. REFERENCES

- 1. K.S. Witwer, "Envelope Testing for Rotary-mode Core Sampling," Westinghouse Hanford Company reprot WHC-SD-WM-TRP-252 (June 1996).
- 2. V. Prakash and F. C. Appl, "Temperature Distribution- in Synthetic Diamond Cutters During Orthogonal Rock Drilling," SPE Drilling Engineering, 137-143 (June 1989).
- 3. D. A. Glowka and C. M. Stone, "Thermal Response of Polycrystalline Diamond Compact Cutters Under Simulated Downhole Conditions," Soc. Petroleum Engineers Journal, 143-156 (April 1985).
- 4. R. Teale, "The Concept of Specific Energy in Rock Drilling," Int. J. Rock Mech. Mining Sci., Vol. 2, 57-73 (1965).
- 5. H. Rabia, "Specific Energy as a Criterion for Drill Performance Prediction," Int. J. Rock Mech. Min. Sci. & Geomech. Abstr., Vol. 19, 39 to 42 (1982).
- 6. C. Donaldson, G. H. LeCain, and V. C. Goold, *Tool Design*, Third Ed., (McGraw Hill, New York, New York, 1973).
- 7. T. Baumeister, E. A. Avallone, and T. Baumeister III, editors, *Marks' Standard Handbook for Mechanical Engineers*. 8th ed. (McGraw Hill, New York, New York, 1978).
- 8. C. M. Keller, "Core Drill Operating Envelope Test Report," Westinghouse Hanford Company report WHC-SD-WM-TRP-123 (1993).
- 9. P. M. Francis, "Aerosol Study for the Rotary Mode Exhauster," Westinghouse Hanford Company report WHC-SD-WM-ES-225, Rev. 1. (March 1993).
- 10. D. A. Siebe, "Thermal Analysis of Rotary-Mode Core Drilling," Los Alamos National Laboratory Calc-Note TSA10-CN-WT-SA-TH-111 (February 1996).
- 11. J. P. Holman, *Heat Transfer*, 3rd ed. (McGraw Hill, New York, New York, 1972).
- 12. S. Kakac, R. K. Shah, and W. Aung, Eds. Handbook of Single-Phase Convective Heat Transfer (John Wiley & Sons, New York, New York, 1987).
- D. I. Poston, "Two-dimensional Transient Heat Transfer Analysis of Rotary Drilling Process," Los Alamos National Laboratory Calc-Note TSA10-CN-WT-SA-TH-109 (February 1996).

APPENDIX G

WASTE AND CRUST IGNITION

G.1. INTRODUCTION

In this appendix, the waste and crust ignition issues are discussed. Waste ignition is a hazard considered to be the result of drilling. The drill bit cutting includes several modes in terms of heat generation and heat transport that are discussed in Appendix F. In an efficient cutting mode, waste chips created by the drill bit teeth can be hotter than the drill bit or teeth temperature. When there is no efficient penetration, a zone where frictional heating between the drill bit and waste exists. In the first part of the appendix, we discuss the operating safety requirements for drilling. In the second part, we examine another accident in which the drill string is postulated to be dropped on a dry crust. We discuss the possibility of a crust ignition and propagation.

G.2. REACTIVE CHEMICAL HAZARDS

Mixtures of sodium nitrate and sodium nitrite with organic compounds can produce violent exothermic reactions.^{1,2} Reaction rates generally increase with increasing temperature. For exothermic reactions, increasing the reaction rate increases the heat generation rate. Increasing the temperature of the mixture that reacts exothermically can create an imbalance between heat generation and heat transfer out of the system. This imbalance can create a thermal runaway.

Rotary-mode core sampling (RMCS) causes local heating of the waste. These "hot spots" near the drill string may cause a localized thermal runaway. The consequences of such a local thermal runaway are the following:

- A local thermal runaway may result in temperatures in excess of the autoignition temperature of a flammable-gas mixture trapped in the waste.
- A local thermal runaway will increase the gas-generation rate. There are three potential hazards associated with increased gas generation: increased production of aerosols, production of flammable gases, and production of toxic gases.
- The high temperatures produced by a local thermal runaway may vaporize some of the nuclides, which would result in an increased release rate.

• A local thermal runaway may propagate through the entire tank. Propagation of a thermal runaway could produce very high temperatures, which would result in severe structural damage of the tank.

The consequences of these hazards have been evaluated to determine if these hazards are a significant concern.

- The auto-ignition temperature of hydrogen-air mixtures is reported to be 400°C (Ref. 3), and the autoignition temperature of hydrogen nitrous oxide mixtures may be lower.⁴ The maximum temperature produced by waste containing 1.0 wt % total organic carbon (TOC) is estimated to be ~400°C. Therefore, a runaway reaction may produce temperatures exceeding the autoignition temperature, even if it contains only a small amount of organic carbon sustaining an exothermic reaction. Exceeding the autoignition temperature is a real hazard.
- The maximum gas-generation rate is estimated to be 3.6 scfm. The maximum volume that could be generated is 250 scf. The nominal nitrogen flow in the drill string is 30 scfm. The maximum gas-generation rate is small compared to the nitrogen flow, so that gas generation will not have a significant impact on aerosol formation. Tests on dry waste surrogates indicate that nitrous oxide is formed during a runaway reaction.⁵ Hydrogen is generated in the waste at normal waste temperatures. There is no reason to expect that these reactions will stop at elevated temperatures; therefore, hydrogen generation is possible. The ignition of ~1000 scf of a stoichiometric mixture of hydrogen and nitrous oxide under the waste surface will cause structural damage; therefore, generating 250 scf of hydrogen or nitrous oxide could be a significant contributor to a flammability hazard. If all of the gas generated is nitric oxide (NO) or nitrogen dioxide (NO₂), the maximum concentration at 100 m from the tank is <1 ppm; so there is no toxic gas hazard.
- Volatile nuclides can be a problem in a waste vitrification plant. If an uncontrolled reaction produces very high temperatures, cesium and strontium may vaporize. A bounding calculation predicts a maximum dose of <1 mrem Committed Effective Dose Equivalent (CEDE) at 100 m from the tank. Therefore, vaporization of nuclides is not a hazard.
- If an exothermic reaction propagates throughout the tank, the consequences could be severe. The volume of waste involved in a propagating reaction could be a factor 100,000 times greater than a local reaction. Consequences that are negligible for local exothermic reactions could be significant for a propagating reaction. The high temperatures generated would cause a structural failure of the tank. A propagating exothermic reaction is a significant hazard.

A local runaway reaction is a real hazard. The primary concerns are that a runaway exothermic reaction could be an ignition source in a flammable-gas tank, and a local reaction may propagate. The generation of flammable gases or oxidants is a lesser concern because a flammable-gas hazard already exists in the tanks of interest, and there are no consequences of a flammable-gas release unless there is an ignition source.

G.3. LOCAL RUNAWAY REACTIONS

G.3.1. Review of Existing Data

Data on exothermic chemical reaction have been obtained using waste samples and using waste surrogates and simulants.

A review of the chemical analyses performed on waste samples was conducted. Between 1974 and 1995, differential thermal analysis (DTA) and differential scanning calorimetry (DSC) were performed on 169 samples obtained from 70 different tanks. In many cases, the reports do not give the onset temperature for exothermic reactions. They only state that no exothermic reactions were observed below 200°C. Such data are useful in establishing a lower bound on the onset temperature. Of the 169 samples, onset temperatures of <200°C were reported for 17. The lowest onset temperature reported was 180°C for several samples obtained from Tanks 241-SY-101 and 241-SY-103.

Several studies have been performed on waste surrogates and waste simulants. Schelle et al.⁵ studied several waste surrogates and a waste simulant using DSC and accelerated rate calorimetry (ARC). The onset temperatures measured by DSC range from 220°C to 364°C. The lowest values correspond to tests with a waste simulant that includes many of the minor waste components, such as nickel, chromium, and iron. Onset temperatures as measured by the ARC are between 113 and 290°C. The onset temperatures measuring <180°C were obtained using the waste simulant. Although low-onset temperatures were measured between 110 and 120°C, a self-sustaining reaction did not occur until the temperature reached 150°C. A reactive system screening tool (RSST) also was used to measure onset temperatures.^{6,7} The measured onset temperatures were 150°C and 180°C. The lower temperature was obtained using the waste simulant discussed above.

G.4. CRITERIA TO AVOID LOCAL RUNAWAY

Local runaway reactions can be avoided by maintaining the waste temperature below the onset temperature for exothermic reactions. The standard practice is to subtract 100°C from the onset temperature as determined by DSC and to subtract 50°C from the measured onset temperature as determined by ARC.⁸ The reasons for these large safety margins is uncertainty. The scanning rate affects the onset temperature of DSC and RSST. The sensitivity of the instrument is also a consideration. Differences between waste simulants and actual waste is also a source of uncertainty as are variations in waste composition. Other factors such as good heat transfer and short time periods can reduce the need for the safety margin applied to the measured onset temperature.

Two length scales must be considered when evaluating the potential for local runaway reactions. The first scale to consider is the microscopic scale. Strain energy added as the drill bit cuts into the waste will heat fragments to temperatures that are greater than the drill bit temperature. The size of these fragments and chips are estimated to be between 1 µm and 150 µm. In Ref. 9, p. 6, the particle size distribution was determined to be 1 to 100 μ m. The samples were collected from drilling tests in simulants. This range was measured by using an optical microscopy technique used by Pacific Northwest National Laboratory (PNNL). Using the same single-shell waste simulant samples from the same collection, Westinghouse Hanford Company's (WHC's) Plutonium Finishing Plant Laboratory covered a particle size range of 0.1 to 10 µm with a laser diffraction analysis. They are smallsized chips, and they are cooled relatively quickly (Ref. 10). The second scale to consider is macroscopic. The drill bit and the surrounding waste will be heated above the bulk temperature of the waste, but it will be cooler than the fragments formed at the drill tip. The thermal inertia of the drill bit and the surrounding waste is much greater than the microsized fragments. Differences in the time scale and the heat transfer rates justify different temperature limits for these two cases.

First, consider the temperature limit for small fragments. The limit will be based on the available data for actual waste samples. The large amount of data available on actual waste samples addresses the problems of errors in surrogate and simulant tests and variability in waste composition. The large number of waste samples obtained from a large number of different tanks addresses the problem of variability in waste composition. Because of the large number and variety of waste samples analyzed, the large margins often applied to onset temperatures measured by DSC and ARC are not appropriate.

The DTA and DSC results for actual waste samples are used to establish the maximum temperature limit. A comparison of data for waste surrogates and simulants is done with results for actual waste samples so that the surrogates and simulants are not unreasonable models of the waste. Therefore, data for waste surrogates and simulants are used as additional evidence that the proposed limit is reasonable. Because this safety assessment (SA) addresses rotary-mode sampling in all single-shell, flammable-gas tanks, the limit must be bounding for all tanks. Therefore, the limit is based on the lowest measured onset temperature of 180°C. The proposed limit for heating of small waste fragments is 180°C. Because the limit is based on a large number of waste samples and because the fragments cool relatively quickly, no additional safety margin is applied to the lowest measured onset temperature.

This temperature limit is supported by the onset temperature measured for waste surrogates, but it is not supported by the onset temperatures measured for the waste simulant data. Exceeding the measured onset temperature for exothermic reactions does not imply that a runaway reaction will occur. Heat transfer also must be considered. The first-order stability criterion for a runaway reaction in a single fragment is

$$\left(\frac{\mathrm{d}T}{\mathrm{d}t}\right)_{\mathrm{o}} < \frac{6 \cdot \mathrm{U} \cdot (\mathrm{T_{o}} - \mathrm{T_{s}})}{\rho \cdot \mathrm{c_{p}} \cdot \mathrm{D}} \quad ,$$

where

 $\left(\frac{dT}{dt}\right)_{0}$ = self-heating rate at the initial temperature of the fragment,

U = heat transfer coefficient for the fragment,

 $T_o = initial$ temperature of the fragment,

 T_s = temperature surrounding the fragment,

 ρ = density of the fragment,

c_p = heat capacity of the fragment, and

D = diameter of the fragment,

k = thermal conductivity.

Conservative parameters are used to estimate the critical self-heating rate. The heat transfer coefficient is assumed to be 5 watts/m² K, the density of a fragment is ~2.7 g/cm³, and the heat capacity is ~1.8 J/g K. The maximum particle diameter of 150 μ m is used in the calculation, and the temperature of the surrounding waste is assumed to be 160°C. The critical self-heating rate for a fragment is estimated to be ~50°C/min. The heating rate used in the DSC and RSST is 1 to 10°C/min, depending on the experiment; therefore, these devices should be able to detect self-heating rates of <50°C/min. Therefore, the onset temperature measured by these devices should be a conservative estimate of the stability limit for a fragment. To relate this self-heating rate to the self-heating rate observed in the ARC, it must be divided by the Φ factor¹¹

$$\left(\frac{\mathrm{dT}}{\mathrm{dt}}\right)_{\mathrm{ARC}} = \frac{1}{\Phi} \cdot \left(\frac{\mathrm{dT}}{\mathrm{dt}}\right) \quad . \tag{G-2}$$

The Φ factor accounts for the thermal inertia of the calorimeter, and the value is ~4 for the experiments discussed in Ref. 5. The reaction should be stable if the self-heating rates measured for the ARC tests in Reference 5 are <12°C/min. Below the onset temperature, the self-heating rate is <0.025°C/min, so the experiments with waste surrogates support the proposition that a thermal runaway does not occur in a fragment with a temperature of 180°C. The self-heating rate for the waste simulant is ~0.04°C/min. Although this value is greater than the criteria for determining the

(G-1)

onset temperature, it is much less than the stability limit for fragments. The self-heating rate data obtained from the ARC supports the 180°C temperature limit for fragments.

Time is a consideration when evaluating the stability of exothermic reactions if the material is heated for a short period of time; therefore, runaway reactions may not be a problem. In an adiabatic system, there is a period of time before a thermal runaway occurs during which the self-heating rate is relatively small. This period is called the induction time. The adiabatic induction time is a useful gauge for making a qualitative evaluation of whether the duration of the drilling operation is long enough to be a concern. If the drilling duration is less than the adiabatic induction time, runaway reactions should not be a problem. The first-order estimate of the adiabatic induction time is

$$t_{I} = \frac{R \cdot T_{o}^{2}}{E_{a} \cdot \Phi \cdot \left(\frac{dT}{dt}\right)_{o}}$$

where

 t_{I} = induction time,

R = gas constant,

 $E_a = activation energy, and$

 $\Phi = \Phi$ factor for the calorimeter.

Based on the data in Ref. 5, a reasonable, but conservative, estimate of the activation energy is 240 kJ/mole. An upper bound for the self-heating rate at 180°C is the heating rate of 1°C/min used for the DSC and RSST. The lower bound for the adiabatic induction time of a fragment at 180°C is 7 min. The temperature of the fragment as a function of time can be approximated by the following equation:

$$\mathbf{T} = \mathbf{T}_{s} + (\mathbf{T}_{o} - \mathbf{T}_{s}) \cdot \exp\left(-\frac{6 \cdot \mathbf{U}}{\rho \cdot \mathbf{c}_{p} \cdot \mathbf{D}} \cdot \mathbf{t}\right).$$
(G-4)

The typical Biot number (UD/k) for 150 µm waste chips is <0.1; therefore, Eq. (G-4) is adequate. The time constant for cooling is ~24 s, which is much less than the adiabatic induction time. The life of a hot fragment is much less than the induction time, which supports the 180°C temperature limit for fragments.

The estimates of the critical self-heating rate and the adiabatic induction time indicate that there is considerable margin in the 180°C temperature limit for fragments. This margin justifies using the minimum measured onset temperature of waste samples as the limit.

(G-3)

Next, a limit must be set for the drill-bit temperature and the average temperature of the waste surrounding the drill bit. As discussed above, the large margins often applied to the measured onset temperatures are not applicable. However, some margin is warranted because thermal inertia may be an important factor at the macroscopic scale. Therefore, a 20°C margin is applied to the minimum measured onset temperature for actual waste samples. The maximum allowable temperature for the drill bit and the surrounding waste is 160°C.

This temperature limit is supported by the onset temperature measured for waste surrogates, but it is not supported by the waste simulant onset temperature data. As discussed above, the onset temperatures measured for the waste simulants may not be applicable because heat transfer also must be considered. The first-order stability criterion for a runaway reaction for the waste immediately surrounding the drill bit is

$$\left(\frac{\mathrm{d}T}{\mathrm{d}t}\right)_{t} < \sqrt{\frac{\alpha}{t}} \cdot \frac{(T_{t} - T_{s})}{\Delta x} \ ,$$

where

 $\left(\frac{dT}{dt}\right)_{t}$ = self-heating rate at time t,

 α = thermal diffusivity of the waste,

t = duration of the heating caused by drilling,

 T_t = temperature of the waste near the drill string at time t,

 T_s = temperature of the bulk waste, and

 Δx = thickness of the area affected by the drill.

This criterion accounts for the heating of the waste surrounding the drill string. The equation predicts that the stability decreases with time because frictional heating of the waste reduces heat transfer. Significant heating only occurs near the drill bit. The length of the drill bit is ~3 in., and the minimum penetration rate is ~0.5 in./min. Therefore, heating caused by drilling affects the waste at a given location for <10 min. The lower bound on waste thermal diffusivity is 0.0008 cm²/s, a bounding temperature for all flammable-gas tanks is 95°C, and the thickness of material affected by the drill is assumed to be 0.5 cm. The limiting self-heating rate, given the temperature control of 160°C, is 9.0°C/min. The heating rate used in the DSC and RSST is 1 to 10°C/min; therefore, these devices should be able to detect selfheating rates of <9.0°C. Therefore, the onset temperature measured by these devices should be a conservative estimate of the stability limit. Equation (G-2) is used to relate this self-heating rate to the self-heating rate observed in the ARC. As stated above, the Φ factor is ~4 for the experiments discussed in Ref. 5. Therefore, the reaction should be stable if the self-heating rate measured for the ARC tests in Ref. 5 is <2.3°C/min. Below the onset temperature, the self-heating rate is <0.025°C/min; therefore, the experiments with waste surrogates support the proposition that a

(G-5)

thermal runaway does not occur at 160°C. The self-heating rate for the waste simulant at 160°C is ~0.03°C/min. Although this value is greater than the criteria for determining the onset temperature, it is much less than the stability limit. The self-heating rate data obtained from the ARC supports the 160°C temperature limit.

Time is also considered. The adiabatic induction time can be estimated from Eq. (G-3). A lower bound for the adiabatic induction time at 160° C-is 54 min. This estimate is greater than the maximum time allowed for drilling a segment. Because the system is not adiabatic, the actual induction time will be greater than the adiabatic induction time. Thus, the duration of the operation at 160° C is too short for a thermal runaway to be a problem.

The margins for the critical self-heating rate and the adiabatic induction time are smaller on the macroscopic scale than they are for waste fragments. The lesser margin justifies setting the temperature limit below the minimum measured onset temperature. Also, the temperature limit for the macroscopic scale is based on an assumed heating time of 10 min. No experiments have been performed in which actual waste samples, waste simulants, or waste surrogates were maintained at 120° C for >10 min. Additional tests are required to confirm that local heating of the waste for >10 min is safe before the conservatism in the temperature limit may be reduced.

G.5. PROPAGATING EXOTHERMIC REACTIONS

The consequences of propagating an exothermic reaction are serious. The criterion of no local runaway reaction provides protection against propagating exothermic reactions. Because the consequences are very severe and because exceeding the temperature limits is credible, additional measures are required to prevent a propagating exothermic reaction. Rotary-mode sampling should not be performed in tanks in which a propagating exothermic reaction can occur.

G.5.1. Review of Criteria

Criteria have been proposed for evaluating the possibility of a propagating exothermic reactions in the waste.¹¹ According to these criteria, no propagation is possible if one of following conditions is satisfied:

$$\begin{cases} \text{TOC} (wt.\%) < 4.5 + 0.15 (wt.\% H_2 O) \text{ or} \\ 20\% \ge wt.\% H_2 O \end{cases}$$
 (G-6)

These criteria are based on experimental data for waste surrogates,¹² and they are plotted in Fig. G-1.

Chemical energy density is used in the chemical process industries to screen for potential reactive chemical hazards.⁸ This method may also be useful in evaluating reactive chemical hazards in the Hanford waste tanks. In this method, the degree of

hazard as a function of enthalpy of reaction is given in Table G-1. A low degree of hazard means that a deflagration is possible but not likely.

If the chemical energy density criteria are used to evaluate the potential for a propagating exothermic reaction, a low degree of hazard should be sufficient protection considering that controls have been established to prevent local runaway reactions. Therefore, the heat of reaction must exceed -300 cal/g waste in dry waste for a propagating exothermic reaction to occur. Evaporation of water will provide additional protection against propagation. Therefore, basic energy density limit will be modified to account for the heat of vaporization of water. In wet waste, propagating reactions are assumed to be impossible if the heat of reaction does not exceed -300 cal/g waste minus the latent heat of vaporization of water.

In order to compare the chemical energy density approach used as a screening in the Chemical Process Industries to the criteria given by Eq. (G-6), the heat of reaction is required. The largest heat of reaction was determined using linear programming, which is the method used by the American Society for Testing and Materials (ASTM) in the CHETAH program for evaluating reactive chemical hazards. A maximum value of -9.7 kcal/g total organic carbon (TOC) was obtained for the oxidation of sodium hydroxy-ethanol-diamine-triacetic acid (NaHEDTA). NaHEDTA was one of the complexants used at Hanford. Decomposition of NaHEDTA and other complexants into less energetic compounds is expected as a result of radiolysis. Therefore, the heat of reaction of -9.7 cal/g TOC is a bounding value rather than a realistic estimate of the heat of reaction. Based on data for waste surrogates, a heat of reaction of -6.0 kcal/g TOC should be typical of actual waste samples.^{5,7}

By assuming a heat of reaction of -6.0 kcal/g, the maximum TOC content can be estimated as a function of the moisture content. The limit based on a chemical energy density of 300 cal/g waste is plotted in Fig. G-1. This limit approximates the criteria proposed by Webb et al.¹¹ for water content <20 wt %, which indicates that the 300 cal/g waste is a reasonable limit. This comparison demonstrates that the criterion given by Eq. (G-6) is equivalent to the chemical energy density method for the waste surrogates.

G.5.2. Probability of a Propagating Exothermic Reaction

Because the consequences of a large scale propagating reaction are not acceptable, the working criterion for evaluating safety is whether this accident is credible (i.e., has a frequency >10⁻⁶/yr). The frequency of a large-scale propagating reaction is estimated using an event-tree model of waste variability. Although this SA applies only to flammable-gas tanks, all 100-series single-shell tanks (SSTs) were considered in this appendix. There are currently 19 single-shell flammable-gas tanks: A-101, AX-101,



Fig. G-1. Proposed criteria for no propagation of exothermic reactions in the waste.

| TABLE G-1 |
|--|
| DEGREE OF HAZARD AS A FUNCTION OF ENTHALPY OF DECOMPOSITION OR |
| REACTION |
| |

| Degree of Hazard | Maximum Enthalpy of Decomposition/Reaction (kcal/g) | Possible Qualitative Interpretations of the Classifications |
|---------------------|---|---|
| High | < -0.7 | Violently exothermic; detonation likely |
| Medium | -0.3 to -0.7 | Exothermic; detonation possible; deflagration likely |
| Low | -0.1 to -0.3 | Deflagration possible |
| Very Low | > -0.1 | Propagation unlikely |

August 8, 1996

AX-103, S-102, S-111, S-112, SX-101, SX-102, SX-103, SX-104, SX-105, SX-106, SX-109, T-110, U-103, U-105, U-107, U-108, and U-109.¹¹ Twenty-three additional tanks have been listed as possible flammable-gas tanks: A-103, BX-107, BY-101, BY-102, BY-105, BY-106, BY-109, C-104, C-107, S-101, S-103, S-105, S-106, S-107, S-109, TX-102, TX-111, TX-112, TX-113, TX-115, U-102, and U-106. The following tanks have a potential to be added to the list: B-111, B-201, B-202, S-104, T-201, T-202, T-203, T-204, TX-116, TX-117, U-110, and U-111.

G.5.2.1. Event-Tree Analysis

The sequence of events leading to a propagating reaction must be determined in an event-tree analysis to evaluate the frequency of a propagating reaction accident. Three things must happen in order for a propagating exothermic reaction to occur: (1) the drilling controls must be exceeded, which will cause heating of the waste, (2) the drill temperature must increase to the ignition temperature of the waste, and (3) the failure must occur in waste sufficiently rich in organics to support propagation.

The waste surrogate data indicate that both TOC concentration and the distribution of organic species affect the ability of the waste to support a propagating reaction.¹³ The variability of heat of reaction observed in samples taken from Tank 241-SY-101 indicates variability in the distribution of organic species within a tank.¹⁴ At least two variables are required to characterize the organic compounds. Heat of reaction and wt % TOC are used in the study. Because heat of reaction is used to characterize the organic compounds, the chemical energy density is used to evaluate the probability of a propagating exothermic reaction. It is easier to incorporate variability into the organic compounds by using the energy density method than by using the criteria given by Eq. (G-6).

An event tree for evaluating the probability of a propagating reaction is shown in Fig. G-2. The initiating event in the event tree corresponds to exceeding one of the drilling controls. Based on a reliability analysis of the operation, the frequency of violating a drilling control is $2.3 \times 10^{-4}/yr$.

Violating one of the drilling controls will cause the waste to heat. Bounding tests performed using a dry pumice block indicate that failure of a control could result in temperatures in excess of waste ignition temperature. In real waste, the maximum temperature will depend on the hardness of the waste and the moisture content. The ignition temperature also is expected to vary. Because of the factors, the probability of ignition given failure of the controls is <1.0; however, there is insufficient data and understanding to quantify the probability. Therefore, the probability of ignition given a control failure is assumed to be 1.0.



Fig. G-2. Event tree for propagating exothermic reactions.

To simplify the analysis, the variability heat of reaction is incorporated into the event-tree model by using two bins: a low-heat-of-reaction bin and a high-heat-of-reaction bin. The first bin is waste with a heat of reaction <-6.5 kcal/g TOC. The second bin is waste with a heat of reaction between -6.5 kcal/g TOC and the bound value of -9.7 kcal/g TOC. The high heat of reaction compounds are the original organic complexants, so high heat of reaction bin represents waste rich in these organic compounds. Because of degradation of the organic complexants with time, a zone or layer rich in the original complexants is very unlikely. The probability of drilling control failure in a zone containing high heat of reaction organic compounds is assumed to be 0.1. Core sample data from Tank 241-SY-101 are consistent with this assumption.

The final branch in the event tree determines whether the waste exceeds the chemical energy density criterion. The chemical energy density limit is a function of the heat of reaction bin, the TOC concentration, and the moisture content. Because the drill is assumed to fail at a random location in the waste, the probability of exceeding the limit is equal to the fraction of the waste that exceeds the limit. There are two possible methods of evaluating this probability: (1) statistical analysis of data and (2) evaluation of the fill history. These two methods are complementary. Statistical methods quantify the observed variability, which cannot be determined from the current analysis of waste history. Analysis of waste history can be used to identify the possibility of remnant organic layers that cannot be detected by statistical analysis. The statistical estimates of probability are based on Pacific Northwest

National Laboratory's (PNNL's) analysis of variance (ANOVA) model.¹² Estimates based on fill history were obtained from a review of work performed at Los Alamos National Laboratory.¹⁵

As shown in Fig. G-2, only two branches of the event tree corresponding to propagating exothermic reaction. The frequency of a propagating reaction is the sum of the frequencies for these two branches.

$$f_{\text{prop}} = f_{i} \cdot \begin{bmatrix} \Pr(\text{Low } \Delta H_{rx}) \cdot \Pr(\text{Exceeding Limit} | \text{Low } \Delta H_{rx}) + \\ \Pr(\text{High } \Delta H_{rx}) \cdot \Pr(\text{Exceeding Limit} | \text{High } \Delta H_{rx}) \end{bmatrix} , \quad (G-7)$$

where

 f_{prop} = frequency of a propagating reaction, and f_i = frequency of the initiating event.

PNNL provided probability estimates obtained from their ANOVA model for all single-shell tanks.¹⁶ The results for 100-series tanks are given in Table G-2. Both the median value and the 95% upper bound are given. The frequencies of propagating reactions were computed for each 100-series tank. The results are given in Table G-3. The table contains estimates based on the median values predicted by the ANOVA model, as well as estimates based on the 95% upper bound. Safety decisions should be based on the 95% upper bound because it accounts for the uncertainty associated with estimating the probabilities for tanks with no waste sample data.

Two factors were considered in evaluating waste history: the maximum TOC concentration in the organic remnant layer and the volume of the high TOC remnant layer. The TOC concentration is related to the reactivity of the waste in the remnant layer, and the probability of violating a drilling control in an organic remnant layer is related to its volume. Two problem tanks, A-106 and C-107, were identified by this method. Tank C-103 is also a problem because it contains a floating organic layer. This SA does not address exothermic reactions of a floating organic layer.
TABLE G-2 ANOVA MODEL PREDICTIONS OF THE FRACTION OF WASTE THAT IS REACTIVE

| Tank Number | Tank Assay Data Number | | f Reaction High Heat of Reaction ste Waste | | of Reaction iste |
|--|---|---|--|--|--|
| | Available | Median | 95% Bound | Median | 95% Bound |
| A-101 A-102 A-103 A-104 A-105 A-106 AX-101 AX-102 AX-103 AX-103 B-101 B-102 B-103 B-104 | Yes Yes No No Yes No Yes No No No Yes No | $\begin{array}{c} 1.5 \times 10^{-6} \\ 1.2 \times 10^{-5} \\ 1.6 \times 10^{-6} \\ 2.9 \times 10^{-7} \\ 4.4 \times 10^{-8} \\ 1.6 \times 10^{-6} \\ 1.2 \times 10^{-7} \\ 2.5 \times 10^{-3} \\ 1.4 \times 10^{-7} \\ 6.0 \times 10^{-8} \\ 1.6 \times 10^{-7} \\ 1.2 \times 10^{-6} \\ 0.0 \\ 0.0 \end{array}$ | 2.0×10^{-4} 1.3×10^{-3} 1.7×10^{-4} 8.6×10^{-4} 5.8×10^{-4} 4.5×10^{-4} 1.8×10^{-4} 3.5×10^{-2} 4.1×10^{-3} 5.8×10^{-4} 1.3×10^{-3} 3.7×10^{-9} 2.1×10^{-8} | 5.0×10^{-4} 2.5×10^{-3} 6.0×10^{-4} 1.5×10^{-4} 3.8×10^{-5} 6.2×10^{-4} 4.4×10^{-5} 9.2×10^{-2} 7.4×10^{-5} 4.3×10^{-5} 7.1×10^{-5} 2.2×10^{-4} 3.2×10^{-9} 5.0×10^{-9} | 9.1 x 10 ⁻³ 5.0 x 10 ⁻² 1.0 x 10 ⁻² 3.6 x 10 ⁻² 3.1 x 10 ⁻² 1.5 x 10 ⁻² 1.0 x 10 ⁻² 2.7 x 10 ⁻¹ 1.3 x 10 ⁻² 2.9 x 10 ⁻² 1.1 x 10 ⁻² 4.1 x 10 ⁻² 3.4 x 10 ⁻⁶ 2.2 x 10 ⁻⁵ |
| B-104 B-105 B-106 B-107 B-108 B-109 B-110 B-111 B-112 BX-101 BX-102 BX-103 BX-104 | No No No No Yes Yes No No No No Yes | $\begin{array}{c} 0.0\\ 7.6 \times 10^{-8}\\ 1.7 \times 10^{-10}\\ 5.7 \times 10^{-12}\\ 1.4 \times 10^{-10}\\ 2.1 \times 10^{-10}\\ 0.0\\ 3.1 \times 10^{-13}\\ 2.8 \times 10^{-11}\\ 1.6 \times 10^{-9}\\ 2.3 \times 10^{-10}\\ 2.2 \times 10^{-12}\\ 4.3 \times 10^{-8} \end{array}$ | $\begin{array}{c} 2.1 \times 10^{-8} \\ 1.6 \times 10^{-4} \\ 5.1 \times 10^{-6} \\ 1.0 \times 10^{-6} \\ 3.1 \times 10^{-6} \\ 4.9 \times 10^{-6} \\ 4.5 \times 10^{-11} \\ 1.7 \times 10^{-8} \\ 1.6 \times 10^{-6} \\ 1.3 \times 10^{-5} \\ 4.2 \times 10^{-6} \\ 2.9 \times 10^{-7} \\ 5.5 \times 10^{-5} \end{array}$ | 5.0×10^{-9} 5.2×10^{-5} 5.1×10^{-7} 3.9×10^{-8} 4.2×10^{-7} 2.9×10^{-7} 1.1×10^{-11} 2.5×10^{-8} 4.3×10^{-7} 2.0×10^{-6} 3.3×10^{-7} 1.3×10^{-7} 2.8×10^{-5} | 2.3×10^{-5} 7.6×10^{-3} 6.8×10^{-4} 1.6×10^{-4} 8.6×10^{-4} 6.0×10^{-4} 7.4×10^{-8} 9.5×10^{-6} 1.0×10^{-3} 1.5×10^{-3} 6.3×10^{-4} 2.0×10^{-4} 2.8×10^{-3} |

TABLE G-2 (cont) ANOVA MODEL PREDICTIONS OF THE FRACTION OF WASTE THAT IS REACTIVE

| Assay Data | Low Heat Wa | of Reaction iste | High Heat | High Heat of Reaction - Waste | |
|---|--|--|--|---|--|
| Available | Median | 95% Bound | Median | 95% Bound | |
| Yes No Yes No Yes No Yes No Yes No No No No No No No No No No No No No | $\begin{array}{c} 2.6 \times 10^{-8} \\ 5.2 \times 10^{-11} \\ 0.0 \\ 3.7 \times 10^{-10} \\ 9.6 \times 10^{-8} \\ 3.2 \times 10^{-10} \\ 2.7 \times 10^{-10} \\ 3.2 \times 10^{-9} \\ 1.4 \times 10^{-7} \\ 1.8 \times 10^{-7} \\ 1.8 \times 10^{-7} \\ 1.5 \times 10^{-7} \\ 1.2 \times 10^{-7} \\ 1.2 \times 10^{-7} \\ 1.5 \times 10^{-7} \\ 1.5 \times 10^{-7} \\ 1.6 \times 10^{-7} \\ 1.6 \times 10^{-7} \\ 1.3 \times 10^{-7} \\ 1.6 \times 10^{-7} \\ 1.3 \times 10^{-7} \\ 1.6 \times 10^{-7} \\ 1.1 \times 10^{-7} \\ 1.4 \times 10^{-7} \\ 1.5 \times 10^{-7} \\ 3.8 \times 10^{-7} \\ 3.8 \times 10^{-7} \\ 8.2 \times 10^{-8} \end{array}$ | $\begin{array}{c} 4.9 \times 10^{-5} \\ 8.2 \times 10^{-6} \\ 1.7 \times 10^{-9} \\ 1.5 \times 10^{-5} \\ 2.3 \times 10^{-7} \\ 2.3 \times 10^{-7} \\ 2.7 \times 10^{-6} \\ 6.6 \times 10^{-6} \\ 2.6 \times 10^{-4} \\ 3.0 \times 10^{-4} \\ 3.0 \times 10^{-4} \\ 3.1 \times 10^{-4} \\ 3.3 \times 10^{-4} \\ 1.8 \times 10^{-6} \\ 4.0 \times 10^{-4} \\ 6.2 \times 10^{-4} \\ 2.1 \times 10^{-4} \\ 3.9 \times 10^{-4} \\ 2.5 \times 10^{-4} \\ 2.5 \times 10^{-4} \\ 2.4 \times 10^{-4} \\ 1.8 \times 10^{-4} \\ 1.5 \times 10^{-5} \end{array}$ | 6.5×10^{-6} 2.6×10^{-7} 3.8×10^{-10} 3.0×10^{-7} 5.3×10^{-5} 3.1×10^{-7} 4.1×10^{-7} 5.3×10^{-6} 6.1×10^{-5} 6.8×10^{-5} 8.4×10^{-5} 5.7×10^{-5} 3.3×10^{-6} 6.8×10^{-5} 4.6×10^{-5} 4.6×10^{-5} 4.6×10^{-5} 5.6×10^{-5} 4.6×10^{-5} 5.7×10^{-5} 5.6×10^{-5} 4.5×10^{-5} 1.3×10^{-4} 4.5×10^{-5} 2.8×10^{-4} 1.2×10^{-4} 3.9×10^{-5} | $\begin{array}{c} 1.9 \times 10^{-3} \\ 5.0 \times 10^{-4} \\ 7.6 \times 10^{-7} \\ 8.9 \times 10^{-3} \\ 9.8 \times 10^{-3} \\ 1.9 \times 10^{-5} \\ 2.5 \times 10^{-4} \\ 9.4 \times 10^{-3} \\ 1.1 \times 10^{-2} \\ 1.0 \times 10^{-2} \\ 7.6 \times 10^{-3} \\ 7.5 \times 10^{-3} \\ 8.1 \times 10^{-5} \\ 1.1 \times 10^{-2} \\ 7.6 \times 10^{-3} \\ 1.1 \times 10^{-2} \\ 7.6 \times 10^{-3} \\ 1.1 \times 10^{-2} \\ 8.1 \times 10^{-3} \\ 1.9 \times 10^{-2} \\ 8.1 \times 10^{-3} \\ 1.4 \times 10^{-2} \\ 5.9 \times 10^{-3} \\ 2.0 \times 10^{-3} \end{array}$ | |
| No | 6.4 x 10 ⁻⁹ | 8.1 x 10 - 6 | 6.4 x 10 - 6 | 7.6 x 10 ⁻⁴ | |
| | Assay Data Available Yes No Yes No Yes No Yes No Yes No Yes No Yes No No Yes No No No No No No No No No No No No No | Assay DataLow Heat WaAvailableMedianYes 2.6×10^{-8} No 5.2×10^{-11} Yes 0.0 No 3.7×10^{-10} No 3.7×10^{-10} No 3.7×10^{-10} No 3.7×10^{-10} No 2.7×10^{-10} Yes 3.2×10^{-9} No 1.4×10^{-7} No 1.8×10^{-7} No 1.8×10^{-7} No 1.7×10^{-7} Yes 8.5×10^{-9} No 1.6×10^{-7} No 1.6×10^{-7} No 1.6×10^{-7} No 1.1×10^{-7} No 1.1×10^{-7} No 1.1×10^{-7} No 1.1×10^{-7} No 3.8×10^{-7} <td< td=""><td>Low Heat of Reaction WasteAvailableMedian95% BoundYes$2.6 \times 10^{-8}$$4.9 \times 10^{-5}No5.2 \times 10^{-11}$$8.2 \times 10^{-6}Yes0.0$$1.7 \times 10^{-9}No3.7 \times 10^{-10}$$1.5 \times 10^{-5}No9.6 \times 10^{-8}$$2.3 \times 10^{-4}Yes3.2 \times 10^{-10}$$2.3 \times 10^{-7}No2.7 \times 10^{-10}$$2.7 \times 10^{-6}Yes3.2 \times 10^{-9}$$6.6 \times 10^{-6}No1.4 \times 10^{-7}$$2.6 \times 10^{-4}No1.8 \times 10^{-7}$$3.0 \times 10^{-4}No1.8 \times 10^{-7}$$3.0 \times 10^{-4}No1.7 \times 10^{-7}$$3.3 \times 10^{-4}No1.6 \times 10^{-7}$$4.0 \times 10^{-4}No1.6 \times 10^{-7}$$4.0 \times 10^{-4}No1.3 \times 10^{-7}$$2.5 \times 10^{-4}No1.6 \times 10^{-7}$$2.5 \times 10^{-4}No1.6 \times 10^{-7}$$2.4 \times 10^{-4}No1.6 \times 10^{-7}$$2.8 \times 10^{-4}No1.6 \times 10^{-7}$$2.8 \times 10^{-4}No3.8 \times 10^{-7}$$1.6 \times 10^{-4}No3.8 \times 10^{-7}$$1.6$</td><td>Assay DataLow Heat of Reaction WasteHigh Heat WasteAvailableMedian95% BoundMedianYes$2.6 \times 10^{-8}$$4.9 \times 10^{-5}$$6.5 \times 10^{-6}No5.2 \times 10^{-11}$$8.2 \times 10^{-6}$$2.6 \times 10^{-7}Yes0.0$$1.7 \times 10^{-9}$$3.8 \times 10^{-10}No3.7 \times 10^{-10}$$1.5 \times 10^{-5}$$3.0 \times 10^{-7}No9.6 \times 10^{-8}$$2.3 \times 10^{-4}$$5.3 \times 10^{-5}Yes3.2 \times 10^{-10}$$2.7 \times 10^{-6}$$4.1 \times 10^{-7}No2.7 \times 10^{-10}$$2.7 \times 10^{-6}$$4.1 \times 10^{-7}No2.7 \times 10^{-10}$$2.7 \times 10^{-6}$$4.1 \times 10^{-7}Yes3.2 \times 10^{-9}$$6.6 \times 10^{-6}$$5.3 \times 10^{-6}No1.4 \times 10^{-7}$$2.6 \times 10^{-4}$$6.8 \times 10^{-5}No1.4 \times 10^{-7}$$2.6 \times 10^{-4}$$6.8 \times 10^{-5}No1.8 \times 10^{-7}$$3.0 \times 10^{-4}$$8.4 \times 10^{-5}Yes1.2 \times 10^{-7}$$3.1 \times 10^{-4}$$5.3 \times 10^{-5}No1.7 \times 10^{-7}$$3.3 \times 10^{-4}$$5.7 \times 10^{-5}No1.6 \times 10^{-7}$$4.0 \times 10^{-4}$$6.8 \times 10^{-5}No1.6 \times 10^{-7}$$4.0 \times 10^{-4}$$6.8 \times 10^{-5}No1.6 \times 10^{-7}$$3.9 \times 10^{-4}$$4.6 \times 10^{-5}No1.1 \times 10^{-7}$$3.9 \times 10^{-4}$$4.6 \times 10^{-5}No1.1 \times 10^{-7}$$2.8 \times 10^{-4}$$4.6 \times 10^{-5}No1.1 \times 10^{-7}$$2.8 \times 10^{-4}$$4.5 \times 10^{-5}No1.1 \times$</td></td<> | Low Heat of Reaction WasteAvailableMedian95% BoundYes 2.6×10^{-8} 4.9×10^{-5} No 5.2×10^{-11} 8.2×10^{-6} Yes 0.0 1.7×10^{-9} No 3.7×10^{-10} 1.5×10^{-5} No 9.6×10^{-8} 2.3×10^{-4} Yes 3.2×10^{-10} 2.3×10^{-7} No 2.7×10^{-10} 2.7×10^{-6} Yes 3.2×10^{-9} 6.6×10^{-6} No 1.4×10^{-7} 2.6×10^{-4} No 1.8×10^{-7} 3.0×10^{-4} No 1.8×10^{-7} 3.0×10^{-4} No 1.7×10^{-7} 3.3×10^{-4} No 1.6×10^{-7} 4.0×10^{-4} No 1.6×10^{-7} 4.0×10^{-4} No 1.3×10^{-7} 2.5×10^{-4} No 1.6×10^{-7} 2.5×10^{-4} No 1.6×10^{-7} 2.4×10^{-4} No 1.6×10^{-7} 2.8×10^{-4} No 1.6×10^{-7} 2.8×10^{-4} No 3.8×10^{-7} 1.6×10^{-4} No 3.8×10^{-7} 1.6 | Assay DataLow Heat of Reaction WasteHigh Heat WasteAvailableMedian95% BoundMedianYes 2.6×10^{-8} 4.9×10^{-5} 6.5×10^{-6} No 5.2×10^{-11} 8.2×10^{-6} 2.6×10^{-7} Yes 0.0 1.7×10^{-9} 3.8×10^{-10} No 3.7×10^{-10} 1.5×10^{-5} 3.0×10^{-7} No 9.6×10^{-8} 2.3×10^{-4} 5.3×10^{-5} Yes 3.2×10^{-10} 2.7×10^{-6} 4.1×10^{-7} No 2.7×10^{-10} 2.7×10^{-6} 4.1×10^{-7} No 2.7×10^{-10} 2.7×10^{-6} 4.1×10^{-7} Yes 3.2×10^{-9} 6.6×10^{-6} 5.3×10^{-6} No 1.4×10^{-7} 2.6×10^{-4} 6.8×10^{-5} No 1.4×10^{-7} 2.6×10^{-4} 6.8×10^{-5} No 1.8×10^{-7} 3.0×10^{-4} 8.4×10^{-5} Yes 1.2×10^{-7} 3.1×10^{-4} 5.3×10^{-5} No 1.7×10^{-7} 3.3×10^{-4} 5.7×10^{-5} No 1.6×10^{-7} 4.0×10^{-4} 6.8×10^{-5} No 1.6×10^{-7} 4.0×10^{-4} 6.8×10^{-5} No 1.6×10^{-7} 3.9×10^{-4} 4.6×10^{-5} No 1.1×10^{-7} 3.9×10^{-4} 4.6×10^{-5} No 1.1×10^{-7} 2.8×10^{-4} 4.6×10^{-5} No 1.1×10^{-7} 2.8×10^{-4} 4.5×10^{-5} No $1.1 \times $ | |

TABLE G-2 (cont)ANOVA MODEL PREDICTIONS OF THE FRACTION OF WASTE THAT ISREACTIVE

| Tank Number | Tank Assay Data Number | | of Reaction iste | High Heat of Reaction Waste | |
|--|--|---|--|---|--|
| | Available | Median | 95% Bound | Median | 95% Bound |
| C-107 C-108 C-109 C-110 C-111 C-112 S-101 S-102 S-103 S-104 S-105 S-106 S-107 S-108 S-107 S-108 S-107 S-108 S-109 S-110 S-111 S-112 SX-101 SX-102 SX-103 SX-104 SX-105 SX-106 CY 107 | No Yes Yes Yes Yes No No No No No No No No No No No No No | $\begin{array}{c} 1.6 \times 10^{-7} \\ 1.1 \times 10^{-11} \\ 7.0 \times 10^{-9} \\ 0.0 \\ 0.0 \\ 6.0 \times 10^{-9} \\ 3.7 \times 10^{-7} \\ 1.3 \times 10^{-7} \\ 1.3 \times 10^{-7} \\ 1.5 \times 10^{-11} \\ 2.1 \times 10^{-7} \\ 1.5 \times 10^{-11} \\ 2.1 \times 10^{-7} \\ 6.2 \times 10^{-7} \\ 3.8 \times 10^{-8} \\ 1.1 \times 10^{-7} \\ 1.1 \times 10^{-11} \\ 2.4 \times 10^{-7} \\ 9.5 \times 10^{-5} \\ 1.9 \times 10^{-7} \\ 5.1 \times 10^{-7} \\ 5.1 \times 10^{-7} \\ 8.4 \times 10^{-8} \\ 4.1 \times 10^{-7} \\ 2.1 \times 10^{-7} \\ 4.2 \times 10^{-7} \\ 1.0 \times 10^{-10} \\ 1.0 \times 1$ | 2.7 \times 10 ⁻⁴ 2.1 \times 10 ⁻⁸ 4.1 \times 10 ⁻⁷ 2.6 \times 10 ⁻⁹ 1.0 \times 10 ⁻⁸ 5.5 \times 10 ⁻⁷ 5.5 \times 10 ⁻⁴ 2.4 \times 10 ⁻⁴ 4.8 \times 10 ⁻⁴ 6.1 \times 10 ⁻⁸ 2.2 \times 10 ⁻⁴ 6.3 \times 10 ⁻⁴ 8.2 \times 10 ⁻⁵ 2.5 \times 10 ⁻⁴ 1.1 \times 10 ⁻⁷ 2.9 \times 10 ⁻⁴ 1.2 \times 10 ⁻⁴ 1.2 \times 10 ⁻⁴ 1.2 \times 10 ⁻⁴ 3.8 \times 10 ⁻⁴ 3.2 \times 10 ⁻⁴ 3.0 \times 10 ⁻⁴ | 4.3×10^{-5} 9.4×10^{-8} 1.3×10^{-5} 4.2×10^{-9} 1.3×10^{-8} 2.0×10^{-5} 9.0×10^{-5} 5.3×10^{-5} 1.3×10^{-4} 1.0×10^{-7} 5.7×10^{-5} 2.6×10^{-5} 4.7×10^{-5} 2.8×10^{-5} 4.9×10^{-5} 4.0×10^{-3} 4.8×10^{-5} 8.0×10^{-5} 1.0×10^{-4} 4.5×10^{-5} 5.7×10^{-5} 5.7 | 1.1 x 10^{-2} 9.1 x 10^{-6} 1.9 x 10^{-4} 4.9 x 10^{-6} 6.2 x 10^{-6} 3.2 x 10^{-4} 1.1 x 10^{-2} 7.3 x 10^{-3} 1.7 x 10^{-2} 2.2 x 10^{-5} 7.8 x 10^{-3} 1.3 x 10^{-2} 4.1 x 10^{-3} 6.6 x 10^{-3} 1.9 x 10^{-5} 7.9 x 10^{-3} 3.3 x 10^{-2} 8.2 x 10^{-3} 1.1 x 10^{-2} 2.7 x 10^{-3} 8.3 x 10^{-3} 9.3 x 10^{-3} 8.4 x 10^{-3} 9.7 x 10^{-3} |
| SX-107 SX-108 | No | 1.9 x 10 ⁻¹⁰ | 4.5 x 10 ⁻⁶ 7.9 x 10 ⁻⁶ | 3.0 x 10 ⁻⁷ | 5.8 x 10 ⁻⁴ |

١

,

TABLE G-2 (cont) ANOVA MODEL PREDICTIONS OF THE FRACTION OF WASTE THAT IS REACTIVE

| Tank Number | Assay Data | Low Heat Wa | of Reaction aste | High Heat Wa | of Reaction iste |
|--|--|--|---|---|---|
| | Available | Median | 95% Bound | Median | 95% Bound |
| SX-109 SX-110 SX-111 SX-112 SX-113 SX-113 SX-114 SX-115 T-101 T-102 T-103 T-104 T-105 T-106 T-107 T-108 T-107 T-108 T-107 T-108 T-107 T-108 T-107 T-110 T-111 T-112 TX-101 TX-101 TX-102 TX-103 TX-104 TX-105 TX-106 TX-107 | No No No No No No Yes No Yes No Yes No Yes No No Yes No No Yes No No No No No No No No | 2.1 x 10 ⁻⁷ 2.2 x 10 ⁻¹⁰ 1.6 x 10 ⁻¹⁰ 1.4 x 10 ⁻¹⁰ 2.1 x 10 ⁻¹¹ 7.6 x 10 ⁻¹¹ 5.4 x 10 ⁻¹¹ 2.2 x 10 ⁻¹⁰ 0.0 5.7 x 10 ⁻¹¹ 0.0 4.7 x 10 ⁻¹⁰ 9.7 x 10 ⁻¹⁰ 4.0 x 10 ⁻¹² 4.1 x 10 ⁻⁸ 2.4 x 10 ⁻¹⁰ 2.5 x 10 ⁻⁸ 3.8 x 10 ⁻¹¹ 0.0 2.0 x 10 ⁻⁷ 1.4 x 10 ⁻⁹ 1.6 x 10 ⁻⁷ 8.9 x 10 ⁻⁸ 6.8 x 10 ⁻⁸ 3.0 x 10 ⁻⁷ | 4.5×10^{-4} 7.7 × 10 ⁻⁶ 5.2 × 10 ⁻⁶ 5.8 × 10 ⁻⁶ 9.3 × 10 ⁻⁷ 7.7 × 10 ⁻⁶ 1.3 × 10 ⁻⁵ 7.6 × 10 ⁻⁶ 3.4 × 10 ⁻⁷ 6.0 × 10 ⁻⁶ 1.0 × 10 ⁻¹⁰ 2.1 × 10 ⁻⁷ 4.3 × 10 ⁻⁵ 1.5 × 10 ⁻⁹ 3.3 × 10 ⁻⁴ 5.8 × 10 ⁻⁶ 9.0 × 10 ⁻⁵ 2.1 × 10 ⁻⁸ 6.4 × 10 ⁻⁷ 5.9 × 10 ⁻⁴ 1.7 × 10 ⁻⁶ 2.5 × 10 ⁻⁶ 4.6 × 10 ⁻⁴ 1.5 × 10 ⁻⁴ 1.5 × 10 ⁻⁴ 1.5 × 10 ⁻⁴ 1.4 × 10 ⁻⁴ 9.2 × 10 ⁻⁴ | 7.8×10^{-5} 6.1×10^{-7} 4.0×10^{-7} 4.6×10^{-7} 2.3×10^{-7} 3.0×10^{-7} 3.0×10^{-7} 3.3×10^{-7} 1.0×10^{-8} 2.2×10^{-7} 5.6×10^{-10} 1.5×10^{-6} 1.0×10^{-6} 5.6×10^{-9} 5.0×10^{-6} 4.5×10^{-7} 2.7×10^{-5} 9.4×10^{-7} 3.1×10^{-8} 8.5×10^{-5} 2.8×10^{-6} 2.7×10^{-5} 3.9×10^{-5} 3.9×10^{-5} 1.2×10^{-4} | $\begin{array}{c} 1.1 \times 10^{-2} \\ 9.3 \times 10^{-4} \\ 1.5 \times 10^{-3} \\ 5.2 \times 10^{-4} \\ 4.7 \times 10^{-4} \\ 3.5 \times 10^{-4} \\ 1.5 \times 10^{-3} \\ 9.7 \times 10^{-4} \\ 8.2 \times 10^{-5} \\ 7.3 \times 10^{-7} \\ 6.6 \times 10^{-5} \\ 4.4 \times 10^{-3} \\ 3.7 \times 10^{-7} \\ 3.8 \times 10^{-3} \\ 5.7 \times 10^{-4} \\ 7.4 \times 10^{-3} \\ 4.7 \times 10^{-5} \\ 1.5 \times 10^{-4} \\ 1.9 \times 10^{-2} \\ 4.5 \times 10^{-4} \\ 1.9 \times 10^{-2} \\ 4.5 \times 10^{-4} \\ 1.6 \times 10^{-2} \\ 7.6 \times 10^{-3} \\ 7.3 \times 10^{-3} \\ 3.0 \times 10^{-2} \end{array}$ |
| TX-108 | No | 1.3 x 10 ⁻⁷ | 2.2 x 10 ⁻⁴ | 6.7 x 10 ⁻⁵ | 1.2 x 10 ⁻² |

August 8, 1996

TABLE G-2 (conf)ANOVA MODEL PREDICTIONS OF THE FRACTION OF WASTE THAT ISREACTIVE

| Tank | Tank Assay Data Low Heat of Reaction | | High Heat of Reaction | | |
|--|---|---|--|--|---|
| Number | Tank Assay Data Waste | | Waste | | |
| | Available | Median | 95% Bound | Median | 95% Bound |
| TX-109 TX-110 TX-111 TX-112 TX-113 TX-114 TX-115 TX-116 TX-117 TX-116 TX-117 TX-118 TY-101 TY-102 TY-103 TY-104 TY-105 TY-106 U-101 U-102 U-103 U-104 U-105 U-106 U-107 U-108 U-109 U-110 | No No No No No No No No Yes Yes Yes Yes Yes Yes Yes Yes Yes No No Yes No No Yes No No Yes No Yes | $\begin{array}{c} 1.6 \times 10^{-7} \\ 8.2 \times 10^{-8} \\ 1.1 \times 10^{-7} \\ 6.1 \times 10^{-8} \\ 6.5 \times 10^{-8} \\ 5.6 \times 10^{-8} \\ 3.9 \times 10^{-8} \\ 9.5 \times 10^{-8} \\ 4.7 \times 10^{-8} \\ 1.6 \times 10^{-6} \\ 5.1 \times 10^{-12} \\ 0.0 \\ 0.0 \\ 4.3 \times 10^{-11} \\ 5.0 \times 10^{-10} \\ 1.3 \times 10^{-10} \\ 6.6 \times 10^{-11} \\ 3.4 \times 10^{-8} \\ 4.5 \times 10^{-5} \\ 1.2 \times 10^{-7} \\ 7.5 \times 10^{-4} \\ 8.5 \times 10^{-7} \\ 6.4 \times 10^{-7} \\ 5.3 \times 10^{-7} \\ 4.5 \times 10^{-7} \\ 3.3 \times 10^{-11} \end{array}$ | 2.9×10^{-4} 1.7×10^{-4} 1.9×10^{-4} 1.8×10^{-4} 1.5×10^{-4} 1.5×10^{-4} 1.5×10^{-4} 1.0×10^{-4} 4.9×10^{-4} 1.5×10^{-7} 3.9×10^{-8} 7.0×10^{-9} 8.0×10^{-9} 2.1×10^{-6} 1.7×10^{-7} 5.4×10^{-6} 1.7×10^{-7} 5.4×10^{-6} 1.7×10^{-2} 7.4×10^{-4} 1.5×10^{-4} 3.8×10^{-4} 5.5×10^{-4} 3.8×10^{-4} 5.5×10^{-4} 2.9×10^{-9} | 4.5×10^{-5} 3.8×10^{-5} 4.2×10^{-5} 3.4×10^{-5} 3.5×10^{-5} 3.5×10^{-5} 3.6×10^{-5} 3.2×10^{-5} 3.1×10^{-4} 3.0×10^{-8} 7.7×10^{-8} 3.7×10^{-8} 3.7×10^{-8} 5.6×10^{-7} 9.7×10^{-7} 4.6×10^{-7} 2.1×10^{-7} 2.5×10^{-5} 1.6×10^{-3} 7.2×10^{-5} 1.6×10^{-3} 7.2×10^{-5} 1.4×10^{-4} 1.3×10^{-4} 8.1×10^{-5} 1.2×10^{-4} | 7.7 x 10^{-3} 7.1 x 10^{-3} 1.0 x 10^{-2} 4.7 x 10^{-3} 5.1 x 10^{-3} 5.9 x 10^{-3} 9.3 x 10^{-3} 8.2 x 10^{-3} 8.2 x 10^{-3} 5.7 x 10^{-3} 1.3 x 10^{-2} 3.1 x 10^{-5} 5.3 x 10^{-6} 1.7 x 10^{-5} 5.3 x 10^{-6} 1.7 x 10^{-5} 3.4 x 10^{-4} 1.2 x 10^{-4} 9.9 x 10^{-4} 5.1 x 10^{-3} 2.9 x 10^{-2} 1.5 x 10^{-2} 1.9 x 10^{-1} 1.8 x 10^{-2} 1.3 x 10^{-2} 1.0 x 10^{-2} 2.3 x 10^{-7} |
| U-111 | Yes | 1.0 x 10 ⁻⁷ | 8.1 x 10 ⁻⁵ | 6.4 x 10 ⁻⁵ | 4.1 x 10 ⁻³ |
| U-112 | No | 1.1 x 10 ⁻¹⁰ | 7.6 x 10 ⁻⁶ | 4.2 x 10 ⁻⁷ | 8.3 x 10 ⁻⁴ |

August 8, 1996

TABLE G-3 FREQUENCY OF A PROPAGATING EXOTHERMIC REACTION AS A RESULT OF ROTARY MODE CORE SAMPLING

| | | | Assay | Frequency of Propagation (yr-1 | |
|--|---|--|---|--|---|
| Tank Number | Flammable- Gas Tank | Sample Status | Data Available | Best Estimate | 95% Bound |
| A-101 A-102 A-103 A-104 A-105 A-106 AX-101 AX-102 AX-103 AX-104 B-101 B-102 B-103 B-104 B-105 B-106 B-107 B-108 B-107 B-108 B-107 B-108 B-109 B-110 B-111 B-112 BX-101 BX-102 BX-103 | Yes No Possible No No Yes No Yes No No No No No No No No No No No No No | Pushed Pushed Pushed Pushed Pushed | Yes Yes No No Yes No No No No No No No No No No No No No | 1×10^{-8} 6×10^{-8} 1×10^{-8} 4×10^{-9} 9×10^{-10} 2×10^{-8} 1×10^{-9} 3×10^{-6} 2×10^{-9} 3×10^{-9} 5×10^{-9} 7×10^{-14} 1×10^{-13} 1×10^{-13} 1×10^{-11} 9×10^{-13} 1×10^{-11} 7×10^{-12} 3×10^{-16} 6×10^{-13} 1×10^{-11} 5×10^{-11} 5×10^{-11} 8×10^{-12} 3×10^{-12} | 3×10^{-7} 1×10^{-6} 3×10^{-7} 1×10^{-6} 8×10^{-7} 5×10^{-7} 3×10^{-7} 1×10^{-5} 4×10^{-7} 1×10^{-6} 8×10^{-11} 5×10^{-10} 2×10^{-7} 2×10^{-8} 4×10^{-9} 2×10^{-8} 2×10^{-12} 2×10^{-8} 4×10^{-8} 2×10^{-8} 4×10^{-8} 2×10^{-8} 4×10^{-8} 5×10^{-9} |
| BX-104 BX-105 BX-106 | No No | | Yes No | 2 x 10 ⁻¹⁰ 6 x 10 ⁻¹² | 5 x 10 ⁻⁸ 1 x 10 ⁻⁸ |

TABLE G-3 (cont) FREQUENCY OF A PROPAGATING EXOTHERMIC REACTION AS A RESULT OF ROTARY MODE CORE SAMPLING

| | | | Assay | Frequency of Propagation (yr-1) | |
|---|--|--|--|---|---|
| Tank Number | Flammable- Gas Tank | Sample Status | Data Available | Best Estimate | 95% Bound |
| BX-107 BX-108 BX-109 BX-110 BX-111 BX-112 BY-101 BY-102 BY-103 BY-104 BY-105 BY-104 BY-105 BY-106 BY-107 BY-106 BY-107 BY-108 BY-109 BY-109 BY-110 BY-111 BY-112 C-101 C-102 C-103 C-104 C-105 | Possible No No No No Possible Possible No No Possible No Possible No Possible No Possible No Possible No Possible No | Pushed Pushed Pushed Pushed Pushed | Yes No No Yes No No Yes No Yes No No No No No No No No No No No No No | 9×10^{-15} 7×10^{-12} 1×10^{-9} 7×10^{-12} 1×10^{-11} 1×10^{-10} 1×10^{-9} 2×10^{-9} 1×10^{-9} 3×10^{-9} | 2×10^{-11} 2×10^{-8} 3×10^{-7} 5×10^{-10} 6×10^{-9} 1×10^{-8} 3×10^{-7} 3×10^{-7} 2×10^{-7} 2×10^{-7} 2×10^{-7} 2×10^{-7} 3×10^{-7} 3 |
| C-106 C-107 C-108 C-109 | N o Possible N o N o | | No No Yes Yes | 2 x 10 ⁻¹⁰ 1 x 10 ⁻⁹ 2 x 10 ⁻¹² 3 x 10 ⁻¹⁰ | 2 x 10 ⁻⁸ 3 x 10 ⁻⁷ 2 x 10 ⁻¹⁰ 4 x 10 ⁻⁹ |

网络哈哈哈哈哈哈哈哈哈哈哈哈哈

TABLE G-3 (cont) FREQUENCY OF A PROPAGATING EXOTHERMIC REACTION AS A RESULT OF ROTARY MODE CORE SAMPLING

| | | | Assay | Frequency of Propagation (yr-1) | |
|---|---|------------------|--|--|--|
| Tank Number | Flammable- Gas Tank | Sample Status | Data Available | Best Estimate | 95% Bound |
| C-110 C-111 C-112 S-101 S-102 S-103 S-104 S-105 S-106 S-107 S-108 S-107 S-108 S-107 S-108 S-109 S-110 S-111 S-112 SX-101 SX-102 SX-103 SX-104 SX-105 SX-106 SX-107 SX-108 SX-109 SX-110 | No No No Possible Yes Possible Possible Possible Possible No Possible No Yes Yes Yes Yes Yes Yes Yes Yes Yes Yes | | Yes Yes No No No Yes No No Yes No Yes No Yes No No No No No No No No | 1×10^{-13} 3×10^{-13} 5×10^{-10} 2×10^{-9} 1×10^{-9} 3×10^{-9} 2×10^{-12} 1×10^{-9} 2×10^{-10} 1×10^{-9} 7×10^{-13} 1×10^{-9} 1×10^{-9} 1×10^{-9} 2×10^{-9} 3×10^{-9} 1×10^{-9} 2×10^{-9} 3×10^{-9} 1×10^{-9} 2×10^{-9} 3×10^{-9} 1×10^{-9} 2×10^{-9} 3×10^{-9} 2×10^{-9} 3×10^{-12} 7×10^{-12} 2×10^{-9} 1×10^{-11} | $\begin{array}{c} 1 \times 10^{-10} \\ 2 \times 10^{-10} \\ 7 \times 10^{-9} \\ 4 \times 10^{-7} \\ 2 \times 10^{-7} \\ 5 \times 10^{-7} \\ 5 \times 10^{-7} \\ 5 \times 10^{-7} \\ 4 \times 10^{-7} \\ 1 \times 10^{-7} \\ 2 \times 10^{-7} \\ 5 \times 10^{-10} \\ 2 \times 10^{-7} \\ 1 \times 10^{-6} \\ 2 \times 10^{-7} \\ 1 \times 10^{-6} \\ 2 \times 10^{-7} \\ 4 \times 10^{-7} \\ 9 \times 10^{-8} \\ 2 \times 10^{-7} \\ 3 \times 10^{-7} \\ 3 \times 10^{-7} \\ 3 \times 10^{-7} \\ 2 \times 10^{-8} \\ 2 \times 10^{-8} \\ 3 \times 10^{-7} \\ 2 \times 10^{-8} \end{array}$ |
| SX-111 SX-112 | No No | | No No | 9 x 10 ⁻¹² 1 x 10 ⁻¹¹ | 4 x 10 ⁻⁸ 1 x 10 ⁻⁸ |

TABLE G-3 (cont) FREQUENCY OF A PROPAGATING EXOTHERMIC REACTION AS A RESULT OF ROTARY MODE CORE SAMPLING

| | | | Assay | | opagation (yr-1) |
|--|---|------------------|--|---|--|
| Tank Number | Flammable- Gas Tank | Sample Status | Data Available | Best Estimate | 95% Bound |
| SX-113 SX-114 SX-115 T-101 T-102 T-103 T-104 T-105 T-106 T-107 T-108 T-109 T-110 T-110 T-110 T-111 T-112 TX-101 TX-102 TX-103 TX-104 TX-105 TX-106 | No No No No No No No No No No No No No N | | No No No Yes No Yes No Yes No No Yes No No Yes No No Yes No No No No No No | 5 x 10 ⁻¹² 7 x 10 ⁻¹² 7 x 10 ⁻¹² 8 x 10 ⁻¹² 2 x 10 ⁻¹³ 5 x 10 ⁻¹² 1 x 10 ⁻¹³ 3 x 10 ⁻¹¹ 2 x 10 ⁻¹¹ 1 x 10 ⁻¹³ 1 x 10 ⁻¹⁰ 1 x 10 ⁻¹¹ 6 x 10 ⁻¹⁰ 2 x 10 ⁻¹¹ 7 x 10 ⁻¹³ 2 x 10 ⁻⁹ 7 x 10 ⁻¹² 3 x 10 ⁻⁹ 1 x 10 ⁻⁹ 9 x 10 ⁻¹⁰ | 1 x 10 ⁻⁸ 1 x 10 ⁻⁸ 4 x 10 ⁻⁸ 2 x 10 ⁻⁸ 2 x 10 ⁻⁹ 2 x 10 ⁻⁹ 2 x 10 ⁻⁹ 1 x 10 ⁻⁷ 9 x 10 ⁻¹² 2 x 10 ⁻⁷ 1 x 10 ⁻⁷ 1 x 10 ⁻⁷ 1 x 10 ⁻⁷ 1 x 10 ⁻⁹ 4 x 10 ⁻⁹ 6 x 10 ⁻⁷ 1 x 10 ⁻⁸ 1 x 10 ⁻⁸ 5 x 10 ⁻⁷ 1 x 10 ⁻⁸ 1 x 10 ⁻⁸ 5 x 10 ⁻⁷ 2 x 10 ⁻⁷ 2 x 10 ⁻⁷ 2 x 10 ⁻⁷ 2 x 10 ⁻⁷ 1 x 10 ⁻⁸ 1 x 10 ⁻⁷ 2 x 10 ⁻⁷ 1 x 10 ⁻⁷ 1 x 10 ⁻⁸ 1 x 10 ⁻⁷ 1 x 10 ⁻⁸ 1 x 10 ⁻⁷ 2 x 10 ⁻⁷ 1 x 10 ⁻⁷ 2 x 1 |
| TX-107 TX-108 TX-109 TX-110 TX-111 TX-112 | No No No Possible Possible | | No No No No No | 3 x 10-9 2 x 10-9 1 x 10-9 9 x 10-10 1 x 10-9 8 x 10-10 | 9 x 10-7 3 x 10-7 2 x 10-7 2 x 10-7 3 x 10-7 3 x 10-7 |
| | | | | 0 / 10 | - ~ 10 |

TABLE G-3 (cont) FREQUENCY OF A PROPAGATING EXOTHERMIC REACTION AS A RESULT OF ROTARY MODE CORE SAMPLING

| | | Assay | | FrequencyofPr | opagation (yr-1) |
|--|---|--|--|---|---|
| Tank Number | Flammable- Gas Tank | Sample Status | Data Available | Best Estimate | 95% Bound |
| TX-113 TX-114 TX-115 TX-116 TX-117 TX-118 TY-101 TY-102 TY-103 TY-104 TY-105 TY-106 U-101 U-102 U-103 U-103 U-104 U-105 U-106 U-107 U-108 U-109 U-110 U-111 | Possible No Possible Possible Possible No No No No No No No Possible Yes Possible Yes Possible Yes Yes Yes Yes Yes | Pushed Pushed Pushed Pushed Pushed | No No No No Yes Yes Yes Yes Yes Yes Yes No No Yes No Yes No No Yes No No Yes No Yes | 8×10^{-10} 8×10^{-10} 9×10^{-10} 1×10^{-9} 7×10^{-10} 8×10^{-9} 7×10^{-13} 2×10^{-12} 9×10^{-13} 1×10^{-11} 2×10^{-11} 1×10^{-11} 5×10^{-12} 6×10^{-10} 5×10^{-8} 2×10^{-9} 3×10^{-9} 3×10^{-9} 3×10^{-9} 2×10^{-13} 2×10^{-9} | $\begin{array}{c} 1 \times 10^{-7} \\ 2 \times 10^{-7} \\ 3 \times 10^{-7} \\ 2 \times 10^{-7} \\ 2 \times 10^{-7} \\ 4 \times 10^{-7} \\ 8 \times 10^{-10} \\ 5 \times 10^{-10} \\ 1 \times 10^{-10} \\ 4 \times 10^{-10} \\ 8 \times 10^{-9} \\ 3 \times 10^{-9} \\ 2 \times 10^{-8} \\ 2 \times 10^{-7} \\ 1 \times 10^{-6} \\ 4 \times 10^{-7} \\ 8 \times 10^{-6} \\ 6 \times 10^{-7} \\ 4 \times 10^{-7} \\ 1 \times 10^{-7} \\ 4 \times 10^{-7} \\ 4 \times 10^{-7} \\ 1 \times 10^{-7} \\ 1 \times 10^{-7} \end{array}$ |
| U-112 | No | | No | 1 x 10 ⁻¹¹ | 2 x 10 ⁻⁸ |

G.5.2.2. Summary

Safety decisions should be based on the 95% upper-bound frequency estimates given in Table G-3. Fill history considerations indicate that core drilling in Tanks A-106 and C-107 may be a problem because these tanks may contain a significant organic remnant layer. Floating organic layers have not been addressed; therefore, core drilling in Tank C-103 is not covered by this SA. Only those tanks that are designated as flammable-gas tanks and that are listed in Table G-3 are within the scope of this SA. This SA does not address reactivity hazards associated with ferrocyanide compounds because none of the current flammable-gas tanks are on the ferrocyanide watchlist. The tanks currently on the ferrocyanide watchlist are BY-103, BY-104, BY-105, BY-106, BY-107, BY-108, BY-110, BY-111, BY-112, C-108, C-109, C-111, C-112, T-107, TX-118, TY-101, TY-103, and TY-104.

G.6. CRUST IGNITION CAUSED BY IMPACT

This section assesses the possibility of causing a propagating crust ignition as a result of a drill-string drop. The drill string is held by a hydraulic foot clamp during rotarymode sampling operations. If the clamp fails (although this is unlikely), the drill string could be dropped on the crust. The impact kinetic energy of the drill string is identified as a crust-ignition hazard because some of the crust or waste is being heated past its ignition temperature. In this section we examine the crust ignition issue.

G.6.1. Kinetic Energy Determination

A key variable controlling the heat input is the kinetic energy of the drill string. Kinetic energy is maximized at a certain drop height because the weight of the drill string increases and the impact velocity decreases as the drop height decreases. The formulation of Miller (Ref. 17) is used to maximize the kinetic energy for a drop. The kinetic energy is given by

where

m = the mass of the sampler in the drill string, 2.86 kg (6.3 m),

M = the mass of the drill-string column, and

V = the velocity on impact.

 $KE = \frac{1}{2}(m+M) V^2 ,$

The maximum length of the drill string is based chosen conservatively as a 16.5-m (648-in.) distance from the riser flange to the tank bottom. The waste depth is minimized in Tank AX-103 at 1.07 m (42 in.). Consequently, the length of the falling drill string is a function of the fall distance, in other words, the distance from the bottom of the drill string to the surface of the waste.

(G-8)

The impact velocity depends on the fall distance. Substitution of all terms into kinetic energy equation gives KE as a function of fall-distance squared. From this expression, there is a unique drop distance that maximizes the kinetic energy. This distance is found to be 7.7 m (25.3 ft). The associated maximum kinetic energy is estimated as 3640 J.

G.6.2. Frictional Ignition

One of the effects of drill string impact is friction between the waste particles. Friction can ignite gun powder and other pyrotechnic mixtures.^{18,19} However, sensitivity to friction is attributed to elemental sulfur and not the organic compounds.^{18,19} The waste stored at the Hanford Site is basic. Elemental sulfur is not expected to exist under the basic conditions typical of Hanford waste; frictional ignition is not expected.¹⁸

G.6.3. Absorption of Impact Energy

Another possible effect of drill-string impact is conversion of the kinetic energy into heat. Based on the discussion in Section 2 of this appendix, local exothermic reactions will not occur during drilling operations if the waste temperature is below 160°C. This criterion also applies to the impact of the drill string on a hard crust. No local exothermic reaction will occur if the crust is not heated to 160°C.

The strain energy absorbed by the crust as a result of impact will affect a volume that expands at a 45° angle away from the falling object. Therefore, assuming that all of the energy is absorbed by a hemisphere with a radius equal to the radius of the drill string is conservative. The conservative energy balance for the impact is:

$$KE = \frac{2\pi}{3} \cdot r^3 \cdot \rho \cdot c_p \cdot (T - T_o) ,$$

where

KE = kinetic energy of the falling drill string,
 r = radius of the drill bit,
 ρ = density of the crust,
 T = temperature after impact, and

 T_0 = initial temperature of the crust.

The radius of the drill bit is 2.86 cm, the bulk density of dry crust material is assumed to be \sim 1.35 g/cm³, and the heat capacity of the waste solids is \sim 1.8 J/g K. The bound initial temperature is 95°C. The maximum waste temperature as a result of impact is 126°C, so significant local exothermic reactions are not expected to occur.

(G-9)

5.4. Propagation of Exothermic Reactions in a Crust

The criteria for nonpropagating exothermic reactions in the waste developed in Section 4.1 of this appendix also apply to a crust layer. If the crust material satisfies these criteria, exothermic reactions are not expected to propagate.

If heat transfer to the dome is neglected, the stability criterion given by Fauske⁷ applies. The critical radius required for propagation is

$$\mathbf{r}_{\rm crit} = \sqrt{\frac{3.32 \cdot \alpha \cdot \mathbf{R} \cdot \mathbf{T}^2}{\frac{d\mathbf{T}}{dt} \cdot \mathbf{E}_a}} ,$$

where

r_{crit} = critical radius,

 α = thermal diffusivity,

R = gas constant,

T = temperature of the sphere,

 $\frac{dT}{dt}$ = self-heat rate, and

 $E_a = activation energy.$

The temperature in this equation is absolute temperature. Because the crust is assumed to be porous, the thermal diffusivity is assumed to be 0.0004 cm²/s, which is half the value used in Section 3.2. The temperature is determined from Eq. (G-9). Section 3.2 states that a conservative estimate of the activation energy is 240 kJ/mole and that a conservative estimate of the self-heating rate at 180°C is 1°C/min. However, to extrapolate to lower temperatures, a lower activation energy is the conservative value. The lower bound on activation energy based on Reference 5 is 20 kJ/mol. Therefore, a conservative estimate of the self-heating rate at 126°C is 0.008 K/s, which is a very conservative estimate based on the waste surrogate and waste simulant data. The conservative value of the critical radius is 3.3 cm, which is larger than the radius of the drill bit. Therefore, dropping the drill string on a crust will not initiate a propagating exothermic reaction. This result is consistent with the results for local exothermic reactions. If a drop accident does not cause significant local exothermic reactions, there will be no propagating exothermic reactions.

G.7. SUMMARY AND CONCLUSIONS

Local runaway reactions are a hazard in the Hanford Site waste tanks. The high temperature produced by runaway reactions is an ignition source for flammable gas and a local runaway may initiate self-propagating exothermic reactions in the waste. Local runaway reactions can be prevented by establishing waste temperature limits. The following limits are placed on temperatures:

(G-10)

- The temperature of small waste fragments produced at the drill tip must not exceed 180°C.
- The temperature of the drill bit and the average temperature of the waste affected by drilling must not exceed 160°C for more than 10 min.

Because failure of the controls implemented to satisfy these temperature limits can fail, the possibility of a propagating exothermic reaction must be considered. Because the consequences of a large-scale propagating reaction are potentially very severe, FG/RMCS is not permitted unless it can be demonstrated that the frequency of a propagating exothermic reaction is $<10^{-6}$ /yr. The frequency of a propagating exothermic reaction in each tank is given in Table G-3. Safety decisions should be based on the 95% upper bound. In addition, core drilling should not be performed in Tanks A-106 and C-107 because fill history indicates that these tanks may contain a significant organic remnant layer. This SA does not cover Tank C-103 because it contains a floating organic layer.

The possibility of a propagating ignition in tank waste following the drop of a sampler drill string is examined. The maximum kinetic energy was found as 3640 J when the drill string drops 26.0 ft. This energy will not heat the waste above the onset temperature of exothermic reactions, so there will be no local exothermic reaction or self-propagating reaction.

G ^c REFERENCES

- 1. L. Bretherick, Bretherick's Handbook of Reactive Chemical Hazards, 4th Ed. (Butterworths, London, England, 1990).
- 2. O. Kubaschewski, C. Alcock, and P. Spencer, *Materials Thermochemistry*, (Pergamon Press, Oxford, England, 1993).
- 3. F. Lees, Loss Prevention in the Process Industries (Butterworths, London, England, 1980).
- 4. K. Jones, "Nitrogen" in Comprehensive Inorganic Chemistry, J. Bailar, Ed. (Pergamon Press, New York, 1973).
- 5. D. Scheele, R. Shell, J. Sobolik, and L. Burger, "Organic Tank Safety Project: Preliminary Results of Energetics and Thermal Behavior Studies of Model Organic Nitrate and/or Nitrite Mixtures and a Simulated Organic Waste," Pacific Northwest Laboratory report PNL-10213 (August 1995).
- 6. N. Kirch, "Adiabatic Calorimetry (RSST and VSP) Test with Sodium Acetate," Westinghouse Hanford Company report WHC-EP-0683 (September 1993).
- 7. H. Fauske, "Waste Simulant PAS94A and Exothermicity," Fauske & Associates, Inc., letter to Bob Cash, Joe Meacham, and Dave Turner (December 14, 1995).
- 8. Center for Chemical Process Safety, Guidelines for Chemical Reactivity Evaluation and Application to Process Design, American Institute of Chemical Engineers, New York, New York (1995).
- 9. P. M. Francis, "Aerosol Study for the Rotary Mode Exhauster," Westinghouse Hanford Company report WHC-SD-WM-ES-225, Rev. 1 (March 29, 1993).
- 10. D. A. Siebe, "Thermal Analysis of Rotary-Mode Core Drilling," Los Alamos National Laboratory Calc-Note TSA10-CN-WT-SA-TH-111 (February 1996).
- 11. B. M. Hanlon, "Waste Tank Summary Report for Month Ending October 31, 1995," Westinghouse Hanford Company report WHC-EP-0182-91 (January 1996).
- J. Stewart, D. Turner, M. Plys, B. Malinovic, J. Grigsby, D. Camainoni, P. Healser, W. Samuels, and J. Toth, "Preliminary Safety Criteria for Organic Watch List Tanks at the Hanford Site," Westinghouse Hanford Company report WHC-SD-WM-SARR-033 (November 1995).

G-28

- H. Fauske, M. Epstein, D. Dickinson, R. Cash, D. Turner, and J. Meacham, "The Contact-Temperature Ignition (CTI) Criteria for Propagating Chemical Reactions Including the Effect of Moisture and Application to Hanford Waste," Westinghouse Hanford Company report WHC-SD-WM-ER-496, Rev. 0 (September 1995).
- 14. D. Herting, D. Bechtold, B. Crawford, T. Welsh, and L. Jensen, "Laboratory Characterization of Samples Taken in May 1991 from Hanford Waste Tank 241-SY-101," Westinghouse Hanford Company report WHC-SD-WM-DTR-024 (1991).
- 15. S. Agnew, "History of Organic Carbon in Hanford HLW Tanks: HDW Model Rev. 3," Los Alamos National Laboratory report LA-UR-96-989 (March 1996).
- 16. P. G. Heasler, "Reactive Waste Estimates for Hanford Single Shell Tanks", Pacific Northwest National Laboratory letter report, August 1996.
- 17. W. O. Miller, "Retained Gas Sampler Drill String Drop Analysis," Los Alamos National Laboratory Calc-Note TSA-6-CN-WT-SA-ST-075-0 (January 1995).
- G. Beitel, "Chemical Stability of Salt Cake in the Presence of Organic Materials," Atlantic Richfield Hanford Company report ARH-LD-11 (April 1976).
- 19. R. Lancaster, T. Shimizu, R. Butler, and R. Hall, *Fireworks—Principles and Practice* (Chemical Publishing Co., Inc., New York, 1972).



APPENDIX H DUST EXPLOSION IN THE DOME

H.1. INTRODUCTION

The rotary-core drilling generates a dust aerosol. Some of the aerosol produced during the core drilling will disperse into the vapor space of the dome, some of it will settle on the top of the waste, and some of it will flow into the ventilation line of the exhauster. The concentration in the dome of any given tank is discussed in Appendix Q. The objective of this appendix is to evaluate the possibility of dust explosions in the dome space of a flammable-gas single-shell tank (SST).

H.2. EVALUATION OF BUREAU OF MINES EXPERIMENT

The Bureau of Mines (BOM)¹ performed experiments on the dust explosions. The dust particles used in the experiments have the same particle characteristics as those generated in the SSTs during the core-drilling operation. However, the real composition of the salt cake in the SSTs is unknown. An appropriate salt-cake simulant must be used. According to Reference 2, the selection of a simulant was based largely on the following criteria: (1) the average particle density dictates the size distribution of aerosol, (2) the majority of tanks that plan to do rotary drilling are salt-cake tanks and have a high sodium nitrate content, and (3) sodium nitrate would be the most crystalline of all rotary-mode tank solids and therefore would be the waste type that would form the most dust. The simulant used in the BOM experiment covered the sodium nitrate concentration ranging from 0% to 90%. This should cover the sodium nitrate concentration of salt cake in the SSTs.

The dust-particle distribution of simulant is another important factor in studying dust explosion. In Reference 2, a particle distribution analysis was performed on a dust sample generated by the rotary-mode core drilling test operations. The sample was a single-shell tank waste simulant RBM#2, prepared by the Westinghouse Hanford Company (WHC) Chemical Engineering Laboratory (CEL). Two methods were used to analyze the particle distribution, (1) scanning by an optical microscope with automatic image analysis performed by a dapple system and (2) laser diffraction analysis performed on a Brinkman particle-size analyzer. Both sets show a size range between 0 to 150 microns. The actual dust-particle size distribution was not measured in the BOM experiment. However, the fuel and oxidant dusts for these tests were pulverized and sieved through a 200-mesh screen. The test aerosols were apparently very close to those given above.

Another important parameter in studying dust explosions is the minimum explosion concentration. The minimum explosion concentration for various dusts ranges from 0.025 to 2.0 oz/ft³ (0.1 oz/ft³ = 100 g/m³) as given in Reference 3. The dust concentration used in the test ranged from 300 to 2000 g/m³. It covered the

minimum explosion concentration specified in Reference 4. Therefore, the dust concentration was adequately selected.

The minimum cloud ignition energy for various dust concentrations is also given in Reference 3. It ranges from 0.01 to 1.92 joules. The ignition source used in the experiment was a very strong 5,000 J pyrotechnic igniter, which is much higher than the minimum cloud ignition energy given in Reference 3. Therefore, the ignition source should be able to ignite the dust simulant.

The measurement system consists of two optical dust probes measuring the uniformity of dust probe and strain-gauge pressure transducer to measure the pressure generated by the igniter and the pressure from any flame propagation if the dust ignites. The pressure history and degree of mixing in the test vessel were measured as desired.

H.3. EXPERIMENTAL RESULTS FROM BOM TESTS

The results from the BOM test are given in Reference 1. The results are summarized below.

The mixture of 15% Na₄EDTA, 56.7% NaNO₃, and 28.3% NaNO₂ in air produced a pressure ratio of less than one. The mixture not only did not deflagrate but actually reduced the pressure from the igniter.

The 10% NPH, 60% NaNO₃, and 30% NaNO₂ mixture dispersed in air produced pressure ratios greater than one. In these tests there was some burning of the NPH fuel that was within the igniter flame, but they are still not considered to be propagating deflagrations according to the standard propagation criteria. The pressure rises were low. This implies that there was essentially no propagation continuing after the igniter flame ended.

The mixture of 10% polyethylene, which ignites easily, and 90% NaNO₃ dispersed in nitrogen produced less of a pressure rise than would be expected from the igniter alone.

The pure fuels (EDTA and NPH) would be explosion hazards if dispersed in air, but the mixture of 15% EDTA and 85% oxidant or 10% NPH and 90% oxidant would not be considered explosion hazards when dispersed in air or nitrogen.

H.4. BOUNDING ANALYSIS

The ratio of the radionuclide concentration in the dome to the concentration in the liquid during operation of the airlift circulators is given by R. Kimura and S. Johnson.⁴ Airlift circulators were used to simulate the effect of the nitrogen cooling during rotary-core sampling. The ratio of radionuclide concentration in the dome to the concentration in the liquid varies from 1.8×10^{-9} to 3.5×10^{-7} . Assuming the concentration of the aerosol is the same as the liquid concentration and the waste

density is 1.6 g/cm³, the concentration of waste in the dome is calculated as 0.56 g/m³. Adding 350 g to this value, which is caused by rotary-mode core sampling (RMCS) operations, makes a dome concentration of 0.81 g/cm³.

Examination of the available waste composition data indicates that the total organic carbon (TOC) concentration in the waste is bounded by 50 g TOC/L. In terms of weight percentage, it is estimated as 3.1 wt%. The organic carbon concentration in the dome is found as 0.025 g TOC/m³.

The organic compounds in the waste are a complex mixture of degradation products of the chelating agents. It is very difficult to estimate the heat of combustion of this mixture. However, by examining the similar organic constituents, we can roughly estimate the maximum heat of combustion per gram of carbon. We examined 32 compounds including cyanide compounds. The maximum heat combustion of $(CH_3)_2NH$ is -898 kJ/g C. Sodium acetate, which is a likely degeneration product, has a heat of combustion of -310 kJ/g C. We used the upper bound value of -898 kJ/g C to estimate the combustion energy generated by the dust during RMCS activity.

The combustion energy associated with the dust is calculated as -22.5 kJ/m³. This energy can be converted into an equivalent hydrogen concentration. The heat of combustion for hydrogen is -241.8 kJ/mol H₂. The equivalent hydrogen concentration is 0.0931 mol H₂/m³. Using the ideal gas law, the concentration in terms of ppm was estimated as 2292 ppm. The flammability limit for hydrogen in air is 40,000 ppm (4%). This very conservative calculation indicates that combustible dust contains less than 6% of the energy contained in a flammable hydrogen-air mixture. Therefore, we can neglect the contribution of combustible dust. The presence of dust in a flammable hydrogen-air mixture would not result in an explosion.

H.5. CONCLUSIONS

The range of NaNO₃ concentrations in the dust and dust distribution in BOM tests were believed to be very close to those in SSTs. The ignition source was strong enough to generate deflagration in the igniter. The test results showed that the mixture of pure fuel and oxidant (NaNO₃) would not propagate deflagrations and would not result in an explosion when dispersed in air or nitrogen.

The bounding analysis showed that the energy contribution of combustible dust is negligible and that the presence of dust in a hydrogen-air mixture would not result in explosion. In summary, the probability of dust explosion appears to be extremely unlikely.

H.6. **REFERENCES**

- 1. K. L. Cashdollar, "Interim Report Core Drilling into Waste Tanks," Pittsburgh Research Center, Bureau of Mines Interim report WHC-SD-WM-ES-219 (June, 1993).
- 2. P. M. Francis, "Aerosol Study for the Rotary Mode Exhauster," Westinghouse Hanford Company report WHC-SD-WM-ES-225 (March, 1993).
- 3. A. E. Cote and J. L. Linville, *Fire Protection Hand Book*, Sixteenth Edition, (National Fire Protection Association, Quincy, Massachusetts, 1990).
- 4. R. Kimura and S. Johnson, "Tank 101-AZ Vapor Space Radionuclide Concentrations and Partition Function," Rockwell International internal letter 65950-86-607 to D. J. Squires (October 14, 1986).

APPENDIX I

DOME COLLAPSE ACCIDENTS

I.1. OBJECTIVE

The objective of this appendix is to compute the bounding respirable waste released during a dome collapse accident in a single-shell tank (SST). The amount of waste material released is used in this safety assessment (SA) to estimate the radiological and toxicological consequences of a dome collapse accident.

I.2. DOME COLLAPSE AND RELEASE SCENARIOS

In Section 2.1, we discuss the various accident scenarios that may lead to a dome collapse. Section 2.2 discusses the different mechanisms for material release during a dome collapse.

L2.1. Dome Collapse Scenarios

As discussed in Sections 3 and 4 of this SA, there may be multiple initiators that result in a dome collapse. In general, the following accident sequences have a potential of collapsing the dome:

- A deflagration in the dome (either initiated in the dome or initiated elsewhere and propagated into the dome). The consequences of this scenario are discussed in this appendix. A deflagration is believed to lead to the bounding accident sequence because
- 1. Convection associated with the deflagration can entrain additional aerosol.
- 2. The tank blowdown following the deflagration potentially can release aerosol before the actual dome collapse occurs.
- 3. The initial tank blowdown will temporarily lift some soil above the dome, potentially increasing the effective collapse distance.
- 4. Following the deflagration, extensive concrete cracking could result in dome collapse occurring quasi-coherently (where increased coherency would increase the aerosol produced upon impact with the waste).
- 5. The air heated by the deflagration can start a natural convection flow that will assist in the release of aerosol.

August 8, 1996

- 6. With a deflagration, there is perceived to be an increased possibility (relative to most other dome collapse sequences) of an associated waste burn that could provide additional aerosol and energy for dispersal.
- A deflagration in the waste. As discussed in Appendix C of this SA, this is a less likely scenario for dome collapse. For the short-term release, the approach used for fragmentation of waste following a dome deflagration is conservatively bounding regardless of whether a waste burn has occurred. For the long-term release, the deflagration accident sequence includes the possibility of a burn for waste remaining in the tank.
- Dome overloading. This accident is prevented through administratively controlling the allowable loads over the tank dome. The consequences of a dome collapse resulting from overloading are bounded by the consequences of a collapse resulting from deflagration.
- Dome space vacuum. Dome buckling is possible if a large vacuum is created in the dome space. Increased ventilation rates with blocked inlets may result in such an accident. The initiators are administratively controlled, and the consequences are bounded by a dome collapse as a result of deflagration in the dome space.
- Seismic event. The consequences of a dome collapse as a result of a seismic event are bounded by the dome collapse as a result of deflagration in the dome space.

I.2.2. Material Release Mechanisms

In computing the total material release as a result of dome collapse, we analyzed the following mechanisms that contribute to the total release:

- Initial material that is suspended in the dome space before the accident;
- Material entrained from the waste surface during deflagration;
- Material entrained from the failed high-efficiency particulate air (HEPA) filters;
- Solid material fragmented by impact during dome collapse;
- Liquefied material splashed by impact during dome collapse,
- Solid material released during dome collapse, deposited outside the dome, and resuspended.
- Liquefied material entrained by the wind from a open pool of waste after the dome collapse.

These mechanisms are analyzed in Sections 3 through 6 of this appendix. A more detailed development may be found in (Ref. 1).

I.3. INITIAL DOME LOADING AND HEPA FILTER LOADING

In the Mixer Pump SA developed for Tank SY-101, a bounding dome volume loading of 0.64 kg (1.41 lb_m) was used (Ref. 2). A high value was used for SY-101 because that tank used to experience periodic gas-release events (rollovers) that were very energetic and that resulted in considerable surface motion. Thus, there was a plausible mechanism of entraining considerable waste material into the dome space during these rollovers.

SSTs do not exhibit similar energetic rollover behavior. For SSTs, the maximum dome loading may result from some intrusive activities such as rotary-mode core sampling (RMCS). At the end of a drilling step, a maximum value of 0.35 kg (0.77 lb_m) is estimated for the SST dome loading.³

The energetic respirable material release from the failed HEPA filters during a dome burn in Tank SY-101 is estimated to be less than 0.5 kg (1 lb_m).² This number is expected to be less for the SSTs because the number of filters is less and the ventilation flow rates are lower.

In this analysis, it is conservatively assumed that the total respirable release resulting from the initial dome loading and HEPA failures is ≤ 0.6 L or 1 kg (2.2 lb_m). As shown later in this appendix, this release is negligible compared to uncertainty in potential releases during the impact phase of dome collapse.

I.4. MATERIAL ENTRAINED DURING A DEFLAGRATION

The only credible deflagration mechanism believed capable of failing the dome is one where the gas has been released to the dome air space so that the dome failure pressures can be exceeded. The temporary pressures of a postulated gas deflagration in the waste may produce some aerosol, but the available energy is small compared to the waste heat capacity. Even if we assume that 10% of the waste volume is a homogeneous, interconnected stoichiometric hydrogen-nitrous oxide mixture at 1.5 atm., the resulting heat of combustion is 10.85 MJ/m³ of waste. The waste heat capacity in 0.9 m³ is 2.6 MJ/°C. Consequently, combustion of this gas will only raise the average waste temperature by about 4°C. More details are given in Ref. 4.

Based on numerical simulation of the SY-101 dome space deflagration (Ref. 2) and using the entrainment correlation given in Ref. 5, the entrainment from the waste surface is estimated as 5.45 kg (12 lb_m) for a deflagration front moving at ~45 m/s (150 ft/s). This value is believed to be conservative and will be used for SSTs also. Note that the entrainment correlation given in Ref. 5 is derived using dry powders and is possibly more applicable to SST conditions in which the waste surface may be drier than the SY-101 waste. The correlation in Ref. 5 is independent of particle size, and it was taken from experiments ranging from 4.5 μ m to 48 μ m aerodynamic equivalent diameter (AED). Therefore, a respirable fraction (RF) of 0.52 from Ref. 6 was applied to obtain a respirable release of 1.8 L or 2.8 kg. As shown later in this appendix, this release is within the uncertainties of other potential releases during the impact phase of the dome collapse.

L5. MATERIAL RELEASE DURING DOME COLLAPSE

The phenomenology associated with calculating the amount of respirable material released during dome collapse is complex, especially given that the waste material properties are not fully understood. With present algorithms for calculating dome collapse and for the short-term release, the key parameters are the drop height and the associated tank air volume. The larger the drop height, the greater the impact on the waste and the more aerosol that can be produced. Ignoring the watch list tanks that are interim-stabilized, the drop height is bounded by the 30.2 ft (9.2 m) distance in Tank SX-101 and a tank air volume of 115,600 ft³ (3,273 m³). The short-term release is based on these tank dimensions, and it assumes the waste composition can correspond to any watch list SST.

Two difference collapse geometries are perceived to exist, depending on the waste composition. First, if the primary contents of the tank are salt cake or salt cake with some sludge but minimal liquid, the collapsing concrete and soil have limited waste penetration. Second, if the tank contents can be assumed to behave more like a slurry or as salt cake and sludge with extensive interstitial liquid, the fall of material into the tank is assumed to produce a splash and penetrate the waste.

I.5.1. Impact on Salt Cake.

In this situation, the concrete and soil are assumed to collect upon the waste. The aerosol produced is that from the initial impact on each unit area of waste. Additional fragmented waste produced later in the collapse sequence is trapped by the initial soil and concrete debris layer. The analysis objective is to evaluate releases from the initial impacts conservatively. Two numbers are desirable, the airborne release from the tank and the respirable release from the tank. Because of uncertainties in the phenomenology, two methods were used to determine the respirable release.

1.5.1.1. The Airborne Release. In this analysis, an upper limit to the airborne release was determined from an aerosol concentration limit. The appropriate limit is necessarily a transient one. One such limit is -1 kg/m^3 , which is possible to achieve in a dust explosion (Ref. 7). This number is not a hard limit; for example, the Fire Protection Handbook (Ref. 8) does reference some dust explosion experiments performed with 2 kg/m^3 . Nevertheless, it is four orders of magnitude

above a tentative Sutter (Ref. 9, p. 2.66) recommendation of 100 mg/m³ "...as a maximum upper-limit quasi-stable air concentration after an explosive event." The 100 mg/m³ limit appears difficult to justify for a transient situation. Hinds (Ref. 7) states that as a rule of thumb, coagulation is neglected in laboratory experiments and industrial hygiene work if the aerosol number density is less than $10^{6}/\text{cm}^{3}$. Fig. I-1 shows a plot of aerosol concentrations with a number density of 10^{6}



Fig. I-1. Concentration for 1E+6 particles/cm³ using a lognormal distribution.

particles per cm³. Lognormal distributions were assumed with a geometric standard deviation, σ_g , of 2 and a particle density of 1.6 g/cm³. For the 1 kg/m³ lognormal distribution in Fig. I-1, in other words, the point with a mass median diameter (MMD) of 21.8 μ m, the RF is 0.072. These are particles with a AED of less than 10 μ m. For a 3273 m³ tank volume, this RF is not negligible.

To examine the transient situation more closely, scoping calculations of aerosol agglomeration and aerosol scrubbing by terminal velocity soil particles are evaluated in Ref. 1. They suggest that with an overall concentration of 1 kg/m^3 , respirable particles will only exist for a few seconds. A lower concentration limit is believed appropriate. However, at this time, selection and justification of a lower number would require numerical calculations that were beyond the scope of this analysis. The airborne concentration adopted here is used in Sections 5.1.3 and 6.0.

I.5.1.2. Respirable Release Using the MacDougall Equation.¹⁰

The first approach to determine the respirable release involves a formula for a respirable release developed by MacDougall, Scully, and Tillerson (Ref. 10). This formula is independent of the airborne concentration, and is given by

$$V_{R} = 2 \times 10^{-10} \times (E_{imp}/Vol) \times Vol$$
,

where

 V_R = respirable release volume,

 $E_{imp} = impact energy, (J),$

Vol = volume of material that absorbs the impact energy, m³, which is also the volume to which the respirable fraction applies.

Equation (I-1) was derived from a series of Argonne National Laboratory (ANL) experiments (Refs. 11, 12, and 13) involving the brittle fracture of small samples of glass, ceramics, uranium dioxide pellets, and concrete. Most tests were in the energy range, E_{imp}/Vol , of 10⁶ to 10⁷ J/m³. These energy densities were compared to dome collapse energy densities by formulas from two separate approaches: a projectile penetration equation (Ref. 14) and a relationship describing the crater produced by assuming the collapse energy is equivalent to TNT (Ref. ¹⁵). Dome collapse energy densities of from 7.1 x 10⁶ to 3.6 x 10⁷ J/m³ were obtained, suggesting that the ANL energy density range is reasonable to apply to the dome collapse accident.

Available calculations of dome loading (Refs. 16, 17, and 18) illustrate that cracking and failure of the dome proceeds from the center outward. An idealized diagram of this collapse mode is given in Fig. I-2. Because of the difficulty in determining aerosol escape with a coherent collapse, and because of the spreading of soil from previously collapsed segments preventing further direct impact on such covered waste, an evaluation was performed varying the fraction of the dome credited with



Fig. I-2. Schematic of dome collapse on salt cake.

August 8, 1996

(I-1)

collapsing coherently on salt cake that is not covered by previous debris. Because the variable Vol does not influence Eq. (I-1) results, the release depends only on the E_{imp} associated with the collapsing mass. The results of this analysis are given in Fig. I-3,



Fig. I-3. Coherent collapsing mass and associated respirable release.

which assumes that the minimum depth of soil above the dome is 10 ft. Because of coherency limitations caused by the strongly reinforced haunch region of the dome, a conservative estimate was judged to involve coherent collapse of the concrete and soil up a limiting diameter of ~90%. This release is 60 L or 96 kg (with a waste density of 1600 kg/m³) of respirable aerosol.

L5.1.3. Respirable Release Using an Energetics Analogy

Collapse does not have to be so coherent. It could start during tank depressurization after the initial deflagration. Expansion calculations show that the residual gas can remain hot, at 600 K or more, and Siebe (Ref. 19) notes that a 150- μ m particle has a heat-transfer time constant of about 0.1 s. In short, some of the organic carbon could undergo combustion in airborne particulate. To evaluate a bounding case, a respirable fraction for the 1 kg/m³ of particulate in the tank air was calculated using the prescription of Halverson and Mishima (Ref. 20), which relates the heat of combustion to aerosol formation by an explosive source. To be conservative, σ_g was set to 2.5. Results as a function of the heat of combustion are given in Fig. I-4. The heat of combustion, Δ H, upper bound was assumed to be 300 cal/g (1.26 MJ/kg).





Fig. I-4. Respirable release treating a burn using an explosion prescription.

This ΔH gives an RF of 0.0289; application to the assumed 3273 kg of airborne waste in the tank gives a respirable release of 95 kg. This release is highly conservative. As shown in Appendix H of this SA, combustion propagation is very difficult to achieve in dusts produced from simulants of waste in the Hanford SSTs.

I.5.2. Impact on Liquefied Waste

L5.2.1. A Scoping Assessment Simulating Impact As An Equivalent Spill.

For liquefied waste, the MacDougall formula for brittle impact does not apply. A splash should be represented. The approach adopted was to model the splash as an equivalent spill, during which the spill energy is assumed to be equivalent to the dome collapse energy. Eq. (26) of a study by Ballinger et al. (Ref. 21) was used to get the airborne release fraction. The Ballinger correlation is based primarily on the Archimedes number, which represents the ratio of gravitational forces to viscous forces. For the gravitational forces, an effective height was used multiplying the real height by the ratio of the mass of the dome over the mass of the waste layer that absorbed the impact. The collapse mass was assumed to contain all the soil out to the angle of repose (30°) because the initially impacting mass may sink, exposing further waste to impact. At the top of the dome, the minimum soil thickness was 10 ft at a conservative density of 120 lb_m/ft³ (1920 kg/m³). With the addition of a 100 lb_m/ft² external load on the dome, the total collapsing mass was 1.34 x 10⁷ kg. For the viscous forces, the viscosity of the waste was conservatively estimated to be 20 cP, the minimum supernatant liquor viscosity for Tank SY-101.

Results as a function of the waste thickness that is assumed to absorb the energy are given in Fig. I-5. A conservative bound for the waste thickness was judged to be 1 ft.



Fig. I-5. Amount of aerosolized material as a function of an energy-absorbing layer.

This gives a 132-kg (291-lb_m) airborne release. DOE-HDBK-3010-94 (page 3-4, Ref. 22) states that a bounding RF for a 3-m spill is 0.8. Ballinger's correlation for the aerodynamic MMD strongly increases as a function of impacting velocity. Consequently, an RF of 0.8 should be bounding for higher drop heights. Using an RF = 0.8 gives a respirable release of 106 kg.

I.5.2.2. Quasi-Mechanistic Considerations Assuming a Continuous Collapse

The analysis in Sec. 5.2.1 is unrealistic in that the simulated coherent collapse leaves no path through the debris allowing the aerosol to escape. An improved picture is to assume that the dome collapse in incoherent. Such a picture is shown in Fig. I-6. The main effect of the filling of the tank by the collapsing dome is to decrease the effective drop distance. (From the impact equations used to obtain the energy density in Sec. 5.1.2, the ratio of the impacting mass to waste mass may be assumed as approximately constant.) The Ballinger correlation depends on the drop height to the 1.35 power. By assuming the Fig. I-6 incoherent collapse, the correlation can be averaged (integrated) over the drop height, reducing the airborne release by a factor of 2.35. The RF is calculated by assuming 0.8 for fall heights greater than 3 m, and RF = 1.0 for that part of the fall height that is less than 3 m. This gives an average RF of 0.84. The respirable release is $0.84 \times 132/2.35 = 47$ kg.



Fig. I-6. Incoherent model of the collapsing dome.

I.6. DOME COLLAPSE LONG-TERM RELEASES

The main concern is atmospheric entrainment of aerosol before post-accident recovery operations can contain the release. There are three situations that require consideration. First, the salt cake impact has produced airborne waste that may have settled outside the dome and can be resuspended. Second, the logical extension of the model shown in Fig. I-6 results in a pool of liquefied waste that can be entrained. Third, if buried waste has been ignited, the heat can release additional toxic or radioactive gases.

For the salt cake impact, the airborne release is approximately 1 kg/m^3 times the tank air volume of 3273 m³ in Sec. 5.1.1. Conservatively, we could assume that ~3000 kg might settle onsite. Part could be resuspended as respirable particulate. The bounding release rate for aerodynamic entrainment and resuspension is 4E-5/h with an RF = 1.0 from DOE-HDBK-3010-94 (page 5-4, Ref. 22). Multiplication gives 0.12 kg/h, or about 20 kg for a 7-day release.

For the liquefied waste case, the volume of the collapsing soil and concrete is $227,800 \text{ ft}^3 (6451 \text{ m}^3)$. After collapse, only 12 ft (3.66 m) will exist between the bottom of the final soil surface and the ground surface even if collapse occurs into an empty tank. The most conservative configuration for long-term releases is to use the assumption that no waste is absorbed by the soil. Instead, a pool of waste is postulated to form on the top of the collapsed soil. Entrainment can occur from this pool. Assuming the volume of the pool is the volume of the waste in Tank A-101, the volume of the pool is 127,010 ft³ (3597 m³), the radius of the pool is 74.1 ft (22.6 m), the depth of the pool center is 9.18 ft (2.8 m), and the surface of the pool is

2.82 ft (0.86 m) below the surface of the ground. The pool surface area is 17,300 ft² or about 1600 m².

With variable winds and property uncertainties, pool entrainment is a difficult problem. The approach was to take the bounding DOE-HDBK-3010-94 (page 3-5, Ref. 22) entrainment rate of 4×10^{-6} /h for outdoors, assessed as applicable to large pools and ponds with windspeeds to 30 mph (13.4 m/s). The RF is 1.0. This rate applies to the mass in a 1-cm active layer. The respirable aerosol released in 7 days is

 $(1600 \text{ m}^2)(0.01\text{m})(1600 \text{ kg/m}^3)(4 \times 10^{-6}/\text{h})(168 \text{ h}) = 17.2 \text{ kg}.$

Gases released from a buried waste burn could not be assessed fully at this time. An adiabatic temperature rise is $(300 \text{ cal/g})(4.184 \text{ J/cal})/(1.8 \text{ J/g-°C}) \approx 700^{\circ}\text{C}$. Potentially, vaporization of some cesium compounds would be possible. However, condensation of escaping gases on soil particles would appear to preclude release. At this time, the conservatism in the resuspension and entrainment calculations would seem sufficient to allow respirable releases from buried waste to be ignored as long as burial can be maintained.

L7. CONCLUSIONS

During the dome collapse accident, the conservative estimate for the prompt release of respirable aerosol is obtained as follows:

| Initial dome loading: | 0.6 L |
|---------------------------------------|---------|
| Entrainment during deflagration: | 1.8 L |
| Solid respirable aerosol from impact: | ~60.0 L |
| TOTAL | 62.4 L |

Because of the neglect of the rapid reduction phenomena from aerosol agglomeration and from possible aerosol scrubbing by incoming soil during the release from the tank, the prompt release would be bounded by 62.4 L of solid particulate during the dome collapse.

Using the receptor doses for 1 L release given in Appendix R, the onsite receptor dose for the prompt release may be calculated as 17,200 rem. The offsite receptor dose is 9.3 rem.

These numbers are conservative if one considers the maximum concentrations that can be sustained in a plume or a cloud. In order to receive 17,200 rem, the on-site individual must inhale 7.2×10^{-4} L of waste. Sutter⁹ states that the concentrations in dust storms range between 0.5 g/m^3 and 10 g/m^3 . The concentration at the face of a mine would be 0.5 g/m^3 , and in smog it would be 50 mg/m^3 . If one assumes that, by

expansion, dispersion, agglomeration, and deposition, the concentration in the plume drops to the lower dust storm limit of 1 g/m^3 (or $6.3 \times 10^{-4} \text{ L}$ of waste/m³ of air), the on-site individual must inhale ~1 m³ of air. Using a $3.3 \times 10^{-4} \text{ m}^3/\text{s}$ breathing rate, the individual must remain in the dust cloud for ~50 min. It is highly unlikely that an individual at 100 m from a catastrophic accident will remain in the cloud for such an extended period of time without protection or without leaving the premises. These calculations indicate that the doses resulting from the dome collapse accident are conservatively estimated. However, to remove some of the conservatism, a more detailed analysis of entrainment, dispersion, agglomeration, deposition, and resuspension must be performed under different weather conditions.

Releases from liquefied waste are on the same order of magnitude. The doses resulting from liquid releases are about an order of magnitude less than the doses resulting from an equal volume of solid releases (see Appendix R). Thus, the liquid releases are not further considered in the dome collapse accidents. Present analysis suggests that incoherence in the collapse may reduce the magnitude of the liquid release by a factor of 2. However, the extent of and effects from incoherence in a dome collapse pose questions that deserve more study.

The long-term entrainment from an open pool of waste following the dome collapse is bounded by 12.5 L/wk of solid resuspended respirable aerosol.

I.8. REFERENCES

- 1. W. Bohl and C. Unal, "Dome Collapse Accidents For Single-Shell Flammable Watch List Tanks," Los Alamos National Laboratory Calc-Note TSA6-CN-WT-SA-TH-108 (March 1996).
- 2. L. H. Sullivan et al., "A Safety Assessment for the Proposed Pump Mixing Operations to Mitigate Episodic Gas Releases in Tank 241-SY-101: Hanford Site, Richland, Washington," Los Alamos National Laboratory report, LA-UR-92-3196, Rev. 14 (March 31, 1995).
- 3. R. T. Perry, "Aerosol Mass in the Dome," Los Alamos National Laboratory Calc-Note TSA6-CN-WT-SA-TH-096 (December 1995).
- 4. K. O. Pasamehmetoglu, "Additional Dome Collapse Calculations For Single-Shell Flammable Watch List Tanks," Los Alamos National Laboratory Calc-Note TSA6-CN-WT-SA-TH-093 (December 1995).
- 5. J. D. Iverson, "Particulate Entrainment by Wind," Department of Aerospace Engineering, Iowa State University report Project 1699, ISU-ERI-Ames-85010, prepared for Los Alamos National Laboratory under Subcontract No. 9-X54-K9610-1 (July 1984).
- D. R. MacFarlane, D. W. Stack, J. Kindinger, R. K. Deremer, S. R. Medhekar, and T. J. Mikschl, "Probabilistic Safety Assessment for Hanford High-Level Waste Tanks," Los Alamos National Laboratory report, LA-UR-95-1900, TSA-11-94-R110 (November 1995).
- 7. W. C. Hinds, "Aerosol Technology, Properties, Behavior, and Measurement of Airborne Particles" (John Wiley and Sons, New York, NY, 1982).
- 8. R. F. Schwab, "Dusts," Chapter 12, Section 3, from A. E. Cote, P. E., Editor-in-Chief, Fire Protection Handbook (National Fire Protection Association, Quincy, Massachusetts, 1992).
- 9. S. L. Sutter, "Accident Generated Particulate Materials and Their Characteristics—A Review of Background Information," Pacific Northwest Laboratory report NUREG/CR-2651, PNL-4154 (May 1982).
- 10. H. R. MacDougall, L. W. Scully, and J. R. Tillerson, "Preliminary Preclosure Radiological Safety Analysis," Appendix F in "Site Characterization Plan Conceptual Design Report," Sandia National Laboratory report SAND84-2641 (1987).
- 11. W. J. Mecham, L. J. Jardine, R. H. Pelto, G. T. Reedy, and M. J. Steindler, "Interim Report of Brittle-Fracture Impact Studies: Development of

Methodology," Argonne National Laboratory report ANL-81-27 (October 1981).

- 12. L. J. Jardine, W. J. Mecham, G. T. Reedy, and M. J. Steindler, "Final Report of Experimental Laboratory-Scale Brittle Fracture Studies of Glasses and Ceramics," Argonne National Laboratory report ANL-82-39 (October 1982).
- 13. W. J. Mecham, L. J. Jardine, G. T. Reedy, and M. J. Steindler, "General Statistical Description of the Fracture Particulates Formed by Mechanical Impacts of Brittle Materials," Ind. Eng. Chem. Fundam., 22, pp. 384-391 (1983).
- 14. C. W. Young, "Equations for Predicting Earth Penetration by Projectiles: An Update," Sandia National Laboratory report SAND88-0013 (July 1988).
- 15. "Effects of Impact and Explosion," Summary Technical Report of Division 2, NDRC, Volume 1, Copy No. 247, U. S. Government War Department, Washington D. C. (Columbia University, Division of War Research, New York, New York, 1946).
- 16. L. J. Julyk, "Static Internal Pressure Capacity of Hanford Single-Shell Waste Tanks," Westinghouse Hanford Company report, WHC-SD-WM-TI-623 (August 1994).
- 17. E. A. Rodriguez, R. F. Davidson, and M. L. Walsh, "Comparison of ABAQUS and EPIC Nonlinear Transient Dynamic Models for a Hydrogen Burn Event of Hanford Site High-Level-Waste Tank 241-SY-101," Los Alamos National Laboratory report LA-UR-95-3441 (October 1995).
- 18. R. F. Davidson, M. L. Walsh, E. A. Rodriguez, "EPIC Hydrodynamic Code Modeling of Hanford Site HLW Tank 241-SY-101 for Bounding Hydrogen Detonation," Los Alamos National Laboratory draft report (December 1995).
- 19. D. A. Siebe, "Thermal Analysis of Rotary-Mode Core Drilling," Los Alamos National Laboratory Calc-Note TSA10-CN-WT-SA-TH-111 (February 1996).
- M. A. Halverson and J. Mishima, "Initial Concepts on Energetics and Mass Releases During Nonnuclear Explosive Events in Fuel Cycle Facilities," Pacific Northwest Laboratory report PNL-5839, NUREG/CR-4593, US Nuclear Regulatory Commission (August 1986).
- M. Y. Ballinger, J. W. Buck, P. C. Owczarski, and J. E. Ayer, "Methods for Describing Airborne Fractions of Free Fall Spills of Powders and Liquids," Pacific Northwest Laboratory report NUREG/CR-4997, PNL-6300 (October 1987).

22. U. S. Department of Energy, "Airborne Release Fractions/Rates and Respirable Fractions for Nonreactor Nuclear Facilities," U.S. Department of Energy Handbook DOE-HDBK-3010-94 (October 1994).


APPENDIX J

HYDROGEN DIFFUSION INTO DRILL STRING AND BURN/DETONATION ANALYSIS

J.1. INTRODUCTION

During the core-sampling operation, nitrogen will flow through the drill string and out the drill bit. The purpose of the nitrogen flow is to cool the drill bit, to provide the lift necessary to remove drill cuttings from the annulus bore, and to maintain a hydrostatic pressure at the drill bit to keep the waste out of the drill string during sampler change. During the sample retrieval process, the waste and flammable gases such as hydrogen, nitrous oxide, ammonia, and methane will flow into the drill string in case of loss of nitrogen flow. The gas concentration will reach the low flammable-gas limit, and a gas burn will occur if a spark is generated because of friction. The objective of this appendix is two-fold:

- To calculate the potential for flammable-gas accumulation in the drill string when the nitrogen purge system is lost; and
- To consider the possibility of transition from deflagration to detonation and to estimate the over-pressure ratio and the rate-of-pressure rise. The over-pressure ratio and the rate-of-pressure rise will be used as boundary conditions for the structure analysis.

J.2. HYDROGEN DIFFUSION INTO THE DRILL STRING

The following sections provide a summary of the analysis methodology and the results of the calculations for the hydrogen diffusion into the drill string.

J.2.1. Analysis

For constant density and diffusion coefficient and for no convective flow, Reference 1 gives the following governing equation for one-dimensional transient diffusion of species a into species b,

$$\frac{\partial c_a}{\partial t} = D_{ab} \nabla^2 c_a \quad ,$$

(J-1)

where

 $c_a = volumetric concentration of species a, kg-moles/m³.$

 D_{ab} = diffusion coefficient of species a into species b, m²/s.

Given the assumptions above and for only one species diffusing into another, the volumetric concentration of the mixture will also be constant so that Eq. (J-1) can be rewritten in terms of mole fraction or volume fraction, assuming ideal gases.

$$\frac{\partial x_a}{\partial t} = D_{ab} \nabla^2 x_a$$

where

 x_a = mole fraction of species a.

If the flow area varies, but the diffusion process can be assumed to be onedimensional, then Equation (J-2) can be rewritten as

$$\frac{\partial \mathbf{x}_{a}}{\partial t} = \frac{\mathbf{D}_{ab}}{\overline{\mathbf{A}}} \frac{\partial}{\partial z} \left(\mathbf{A} \frac{\partial \mathbf{x}_{a}}{\partial z} \right) , \qquad (J-3)$$

where

A = flow area, which is a function of z, m^2 and

 \overline{A} = average flow area over the volume of interest, m².

For this problem, the initial condition is that the mole fraction of hydrogen in the drill string be zero. The boundary conditions are that the mole fraction of hydrogen at the bottom of the drill string (i.e. z = 0) be one and that the diffusion flux of hydrogen at the top of the drill string be zero, which implies the gradient of the mole fraction of hydrogen at z = L is zero.

Equation (J-3) is solved numerically.² An analytical solution to Eq. (J-2) with the initial conditions and boundary conditions given above exists. Comparison of the analytical solution with the finite-difference solution for constant flow area indicates that because of the numerical diffusion, the finite difference solution diffuses hydrogen faster than the numerical solution on the order of 10% absolute relative error, which is conservative.

The diffusion coefficient for hydrogen in nitrogen was obtained from curve fitting parameters to the CHEMKIN³ database. These curve-fitting parameters indicate that the diffusion coefficient for hydrogen in nitrogen increases from 0.682 cm²/s at 20°C to 1.042 cm²/s at 90°C. A conservative value of 1 cm²/s was chosen for this analysis. The molecular diffusion coefficient varies as the inverse of the absolute pressure. For this analysis, a conservative pressure of 1 atm was chosen.

The diffusion of one species into another because of thermal gradients is governed by the thermal diffusion ratio. The mass flux of species a into species b because of a temperature gradient is given by Eq. (J-4) as,

(J-2)

$$j_{a}^{(t)} = -\left(\frac{c^{2}}{\rho}\right) M_{a} M_{b} D_{ab} k_{T} \nabla \ln T ,$$

where

 $j_a^{(t)}$ = mass flux of species **a** into species **b** caused by a thermal gradient, kg/m²-s,

c = molar concentration of the mixture, kg-moles/ m^3 ,

 ρ = density of the mixture, kg/m³,

 M_a = molecular weight of species a, kg/kg-mole,

 M_b = molecular weight of species **b**, kg/kg-mole,

 k_{T} = thermal diffuse ratio, and

T = gas temperature, K.

From Eq. (J-4) it can be seen that the sign of thermal diffusion ratio determines the direction of the thermal diffusion for a given species. If the thermal diffusion ratio is positive, then species a moves toward the colder temperature; when the thermal diffusion ratio is negative, then species a moves toward the hotter temperature. Experimental values for the thermal diffusion ratio for hydrogen diffusing into nitrogen caused by a thermal gradient are given in Reference 2 for 264 K as -0.0548 to -0.0663, depending upon the hydrogen mole fraction. The negative thermal diffusion ratio indicates that hydrogen diffusion caused by thermal gradients will diffuse toward a higher temperature region and nitrogen will diffuse toward the colder temperature region. Therefore, it is conservative to neglect this effect. Hydrogen diffusion caused by thermal gradients because the waste is anticipated to be at a higher temperature than the nitrogen gas in the drill string. An order of magnitude solution can be obtained based on a steady-state solution of the case in which the molecular diffusion is balanced by the thermal diffusion as,

$$\Delta \mathbf{x}_{\mathbf{a}} = -\mathbf{k}_{\mathrm{T}} \ln \left(\frac{\mathbf{T}_{2}}{\mathbf{T}_{1}} \right) ,$$

where

- T_2 = hot gas temperature, K, and
- $T_1 = cold gas temperature, K.$

August 8, 1996

(J-5)

(J-4)

 Δx_a = difference in steady-state hydrogen mole fraction at the hot temperature minus the hydrogen mole fraction at the cold temperature.

Evaluating Eq. (J-5) with experimental values in Reference 1 yields a difference in hydrogen mole fraction at a hot temperature minus the hydrogen mole fraction at a cold temperature at steady-state to be on the order of 1.4%. Therefore, the magnitude of the effect is not significant.

Based on past analyses, the buoyancy effect of hydrogen gas mixing with air or nitrogen is significant. The actual mixing of hydrogen with the drill-string nitrogen gas may be faster than the conservative diffusion calculation presented in this appendix. The actual mixing of retained gas with the drill-string nitrogen gas may be limited by how fast the retained gas can diffuse from the waste to the drill bit. However, if the drill bit is in a relatively large retained gas bubble, then diffusion and mixing of retained gas with the drill-string nitrogen gas will not be limited by diffusion of retained gas through the waste.

J.2.2. Results

Equation (J-3) was solved for two geometries. Both geometries were assumed to be 320 in., which represents the smallest length for the drill string to still be in contact with the waste. The small flow area geometry assumes that the seal in the bottom of the rotary-core drill string has failed completely and that the available flow area for diffusion of hydrogen into the drill string is an annulus with an o.d. of 1.82 in. and an i.d. of 1.78 in. Drill-string gas volume continues to be an annulus with a varying i.d. until the elevation is above the latching mechanism. For the bit-flow area geometry, the drill-string geometry is a cylinder at all elevations with a diameter of 1.91 in. and with an inlet flow for hydrogen diffusion equal to the available bit-flow The average hydrogen concentration in the rotary-core drill string as a area. function of time is given in Fig. J-1 for the small flow area geometry and the bit-flow area geometry. From Fig. J-1 it can be seen that for the bit-flow area geometry, the average concentration rises faster than for the small flow area geometry. The bitflow area geometry case reaches an average flammable-gas concentration within 25 minutes, while the small-flow area case reaches an average flammable-gas concentration within 180.0 minutes.

Figure J-2 shows the results for hydrogen concentration versus axial level for selected times into the transient for the small-flow area case. You can see that the first few feet of the drill string that is an annulus for this case fills with hydrogen relatively quickly. The discontinuity in the slope of the results occurs when the geometry goes from an annulus to a cylinder. For the bit-flow area case, shown in Figure J-3, there is no discontinuity in the slope of the hydrogen concentration versus z curve, because the geometry is a uniform cylinder. By comparing Figs. J-2 and J-3, it can be seen that the larger inlet area for the bit-area case results in a larger diffusion rate of hydrogen into the drill string. Within 24 hours, both calculations

indicate significant concentrations of hydrogen from the top to the bottom of the drill string.



Time (hrs)

Fig. J-1. Average hydrogen concentration in the drill string.

WHC-SD-WM-SAD-035, Rev. 0-a











The maximum molecular mass-diffusion flux into the drill string occurs at the firsttime step of each transient calculation. The following results were obtained for the maximum mass-diffusion volumetric flow of hydrogen into the rotary-core drill stream: (a) for a small-area case, 0.000122 cfm, (b) for a bit-flow area case, 0.000448 cfm. An order-of-magnitude analysis can be used to verify these numbers. The massdiffusion flux at the entrance to the drill string can be written as

$$j_a = -cM_a D_{ab} \nabla x_a \quad , \tag{I-6}$$

where j_a = mass diffusion flux caused by the concentration gradient, kg/m²-s.

Evaluating Eq. (J-6) for P = 1 atm, T = 303 K, a diffusion coefficient of 1 cm²/s and the gradient of the concentration profile on the order of -10 yields $j_a = 8.064e-05 \text{ kg/m}^2\text{-s}$ or in terms of a volumetric flux yields, $q_a = 0.001 \text{ m/s}$ of hydrogen.

For the small-flow area case, the inlet flow area is 0.000232 m^2 and for the bit-flow area case the inlet flow area is 0.00133 m^2 , which results in a peak volumetric flow of hydrogen into the drill string caused by hydrogen diffusion of 0.000232 liters/s (0.000492 cfm) for the small-flow area case and 0.0009 liters/s (0.00019 cfm) for the bit-flow area case. These order-of-magnitude results are consistent with the numerical solutions to the transient diffusion equation.

Results obtained from Eq. (J-6) are independent of the pressure chosen because the molar concentration for an ideal gas mixture is proportional to the pressure, and the diffusion coefficient is inversely proportional to the pressure. However, the volumetric flux depends upon one over the density, and the density is directly proportional to pressure. Therefore, as pressure increases, density will increase, and the volumetric flow of hydrogen caused by diffusion will decrease.

J.2.3. Summary and Conclusions

A conservative estimate of the diffusion rate of hydrogen into the rotary-core drill string based on two different geometries indicates that within 20 minutes to 180 minutes, depending upon the inlet flow area, there will be a significant concentration of hydrogen within the drill string. It is a conservative assumption that the retained gas, which includes its own oxidizer, will diffuse as fast as hydrogen by itself. Based on this conservative analysis, the average retained-gas concentration within the rotary-core drill string following the failure of sampler chevron seal will exceed 4% within 180 minutes and following a nitrogen purge system failure with the test section removed, will exceed 4% within 20 minutes. Concentrations high enough to support a detonation (assuming an oxidizer is present) are reached within minutes in the first few feet of rotary-core drill string drilling for both cases, and the average concentrations for the rotary-core drill string reach detonation limits again, assuming an oxidizer is present within 3 to 11 hours.

The maximum diffusion flux of hydrogen into the rotary-core drill string occurs at the first-time step into the transient solution. The maximum volumetric flow of hydrogen into the drill string was calculated for the bit-flow area geometry based on an order of magnitude analysis to be 0.00019 cfm.

J.3. BURN AND DETONATION ANALYSIS IN THE DRILL STRING

The assumptions for this analysis are as follows:

- 1. The nitrogen pressurization system failed to operate for more than 10 h;
- 2. A burn is generated by a spark resulting from metal friction in the drill string;
- 3. The gases in the drill string are well mixed; and
- 4. The fuel is hydrogen only; this assumption is reasonable because the hydrogen concentration may be very high, as discussed in Appendix C.

J.3.1. Burn Analysis

Burn can probably occur in two places in the drill string during a loss-of-nitrogenflow accident. One is between the drill bit and the bottom of the universal sampler. The other one is between the top of the universal sampler and the ball valve. However, the geometry at both locations, which are the two closed ends of a pipe, is similar. Therefore, a 1D burn-combustion code can be used to analyze the problem and to estimate the overpressure and the-rate-of-pressure rise, dP/dt, in the drill string.

The problem was solved with GASBURN, which is a derivative of the GASFLOW code.⁴ The code strips the chemical reaction subroutines in GASFLOW and solves the 1D hydrodynamic equations. The result calculated by GASBURN is given in Reference 5. The maximum overpressure is 30%, which is a little above the stoichiometric value of 29.5%.

In Reference 5, the pressure ratio calculated by the GASBURN code is compared with the data obtained by the Bureau of Mines (BOM).⁶ The measured pressure ratio, which is defined as the ratio of maximum absolute explosion pressure to the initial pressure, is in good agreement with that calculated with the adiabatic constant-volume combustion model. If we convert the overpressure calculated by GASBURN to the pressure ratio, the pressure ratio calculated by GASBURN is in good agreement with that calculated by BOM. This comparison shows that the chemical reaction model in GASBURN is properly modeled.

We can either use the overpressure calculated by GASFLOW or the experimental data for the safety analysis. To estimate the overpressure in the drill string, we need to know the hydrogen concentration. In Section 2 of this appendix, we calculate a concentration of 30% when the axial position of the universal sampler is about 10 ft above the rotary drill in about 12 h. The hydrogen concentration of 30% is the point at which maximum pressure occurs. The calculated overpressure is about 6.3 bars or 630 kPa.

For structure analysis, we need to estimate the rate of pressure rise in the drill string. The maximum rate-of-pressure rise, which is defined as $K_G = (dP/dt)V^{1/3}$ bar-m/s, is obtained from the BOM experiment. The maximum rate of pressure rise for 30% hydrogen is about 1120 bar-m/s (from the Appendix, "Detailed Listing of Flammability Test Data").⁶ The i.d. of the test chamber is 2 ft. The volume of the test chamber is 4.1888 ft³ or 0.118613 m³. The rate-of-pressure rise is 2279.5 bar/s. It should be noted that the data of K_G are scattered from 456 to 1120 bar-m/s.

The BOM data were obtained from a spherical chamber. However, the drill string is in cylindrical geometry. The BOM data may not be applicable to drill-string geometry. We used the GASFLOW code to calculate the over-pressure and the rateof-pressure rise in a pipe of 1.91 in. in diameter and 10 ft in length. The overpressure and the rate-of-pressure rise are ~9.0 bar and ~2280 bar/s. The results are similar to those calculated for a spherical chamber as above.

J.3.2. Detonation Analysis

Consider a tube with a very large length-to-diameter ratio (L/D) such as the drill string case containing a combustible mixture whose normal burning velocity is high relative to the velocity of sound of the unburned gas. With these circumstances, ignition at a closed end can cause the flame to accelerate to such an extent that a detonation eventually occurs in the tube. The phenomena of transition from deflagration to detonation is probably the least-understood aspect of detonation theory at this time. The critical parameters for controlling the transition are the tube diameter, the tube length, and fuel concentration. Detailed discussions of these parameters are given in Reference 7 and are summarized below.

The critical cell size in a detonation is a fairly easy quantity to measure. It ties with the chemical reaction rate having the gross macroscopic propagation behavior of detonations. The further a mixture is from stoichiometric, and hence the less energetic the chemical reaction, the larger is the detonation cell size. In Reference 8, the detonation cell size for a hydrogen-to-air mixture at atmospheric pressure is measured.⁶ The minimum detonation cell width is about 0.6 inches. The required diameter for propagating the detonation wave is about one-third of the detonation cell width. Therefore, it only needs a 0.2-in.-diameter tube to propagate a detonation wave generated by a stoichiometric mixture. The i.d. of the drill string is much larger than that value. Hence, if any detonation generates in the drill string, it may propagate in the drill string.

The induction distance (the distances from the igniter to the location in the detonation tube where the flame propagation rate first attains detonation velocity) are an important parameter to control wave propagation. If the tube is too short, then the wave never reaches it maximum velocity. Bollinger performed an experimental study to measure the induction distances in hydrogen nitrous oxide mixtures.⁹ For a 79-mm i.d. tube, the induction distance for 30% hydrogen is about 200 cm. The ratio of induction distance to the tube diameter is about 23. The L/D

for the drill string is about 60, which is much longer than the above ratio. Therefore, in the drill string, the wave speed may reach the detonation wave speed, which is about 2000 m/s for the stoichiometric mixture.

Hydrogen and air mixtures near stoichiometric ($\sim 29.5\%$ hydrogen) are known to be detonable. Mixtures departing from stoichiometric, either in the hydrogen-lean or hydrogen-rich direction are increasingly more difficult to detonate. It has been generally believed that hydrogen-air mixtures with mixture ratios of <18% or >58% hydrogen, could not be detonated. As shown in Section 2 of this appendix, the hydrogen concentration in the drill string reaches 30% in 16 minutes at 2 ft above the rotary drill. One operation of drilling takes about 30 minutes. By that time, the hydrogen concentration is already above 18%. Therefore, some burn in the drill string may detonate.

Akbar and Shepherd performed an experiment to study deflagration-to-detonation transition (DDT) using the gas mixture similar to the 101-SY waste gas mixture (42.8% H₂, 36.4% N₂O, and 28.8% N₂) with various dilutions.¹⁰ This study shows that the waste gas mixture produced DDT up to 50% of dilution with air. The waste gas composition may contain higher concentrations of hydrogen (see Appendix C). Therefore, DDT may be produced in the drill string.

Theoretical detonation pressure for hydrogen-to-air mixtures in a pipe can be calculated through a numerical method by solving the differential equation for isentropic compression in the burn gas. Reference 8 shows the theoretical detonation pressure and temperature, and reflected detonation pressure and temperature for hydrogen-to-air mixtures. The pressure ratio is about 15 at 30% H₂ before the pressure wave is reflected back from the tube end. The final over-pressure is about 15 bars. Using the detonation velocity 2000 m/s (see page 2-50, Ref. 7) and a detonation cell with 10 mm, the rate-of-pressure rise is about 3 x 10⁶ bar/s. It should be noted that we did not use the reflected pressure because an increase in the rate of first compression pressure is high enough to fail the structure of the drill string.

J.3.3. Conclusions

From the above analysis, the overpressure and the rate-of-pressure rise during a burn in the drill string are 630 kPa and 2279.5 bar/s, respectively. During a detonation in the drill string, the overpressure and the rate-of-pressure rise are 15 bars $(1.5 \times 10^6 \text{ Pa})$ and $3 \times 10^6 \text{ bar/s}$, respectively. A DDT may occur in the drill string. Therefore, any burn in the drill string may result in a detonation because the rate of pressure rise is very high $(3 \times 10^6 \text{ bar/s})$ during the detonation. Consequently, the structure of the drill string is conservatively assumed to fail.

J.4. REFERENCES

- 1. R. B. Bird, et.al., "Transport Phenomena," (John Wiley & Sons, New York, NY 1960).
- 2. J. W. Spore, "Bounding Hydrogen Diffusion Analysis in Rotary Core Drill String," Los Alamos National Laboratory TSA-10 Calc-Note TSA-CN-WT-SA-GR-048 (December 1995).
- 3. R. J. Kee and J. A. Miller, "A Structured Approach to the Computational Modeling of Chemical Kinetics and Molecular Transport in Flow Systems," Sandia National Laboratory report SAND86-8841 (December 1993).
- 4. J. R. Travis, K. L. Lam, and T. L. Wilson, "GASFLOW: A Three-Dimensional Finite-Volume Fluid-Dynamics Code for Calculation of the Transport, Mixing, and Combustion of Flammable Gases in Geometrically Complex Domains," Los Alamos National Laboratory report LA-UR-94-2207 (July 11, 1994).
- 5. J. C. Lin, "Drill String Detonation," Los Alamos National Laboratory Calc-Note TSA10-CN-WT-SA-GR-050 (January 1996).
- 6. K. L. Cashdollar, et. al., "Laboratory Flammability Studies of Mixtures of Hydrogen, Nitrous Oxide, and Air," Westinghouse Hanford Company report WHC-SD-WM-ES-219, Rev. 0 (June 1992).
- 7. A. L. Camp, et. al., "Light Water Reactor Hydrogen Manual," Sandia National Laboratories report NUREG/CR-2726, SAND82-1137 (June 1983).
- 8. C. M. Guirao, et. al., "Hydrogen-Air Detonations," presented at the 1982 International Combustion Symposium, Haifa, Israel (1982).
- 9. L. E. Bollinger, J. A. Laughrey, and R. Edse, "Experimental Detonation Velocities and Induction Distances in Hydrogen-Nitrous Oxide Mixtures," ARS Journal, Vol. 32, 81-82 (1962).
- 10. R. Akbar and J. E. Shepherd, "Detonation in N2O-H2-N2-Air Mixtures," Rensselaer Polytechnic Institute, Department of Mechanical Engineering report prepared for Los Alamos National Laboratory under contract C-4836 (June 18, 1993).

APPENDIX K

TOXICOLOGICAL CONSEQUENCE ANALYSIS METHODOLOGY

K.1. INTRODUCTION

This appendix discusses the methodology used in quantifying the toxicological consequences of accidental releases during rotary-mode core sampling (RMCS) operations.

K.2. METHODOLOGY

The methodology provided in References 1 and 2 are adopted for consistency with other safety basis documents used at the Hanford Site Tank Farms.

K.2.1. Release of Toxic Gases

For gaseous releases, the receptor doses are obtained as

$$C = \frac{\frac{\chi}{Q}}{1 + V \times \frac{\chi}{Q}} \times S \times V$$

where C is the receptor dose,

 χ/Q is the atmospheric dispersion factor given in Table K-1,

V is the gas-release volume (puff release) or gas-release volume flow rate (continuous release), and

S is the species concentration at the source.

For $(V)(\chi/Q) \ll 1$, which is the case for the accidents analyzed in this safety assessment (SA), Eq. (K-1) reduces to

$$C = \frac{\chi}{Q} \times S \times V \quad . \tag{K-2}$$

The atmospheric dispersion coefficients are obtained using the computer code GXQ, which uses the methods in the Nuclear Regulatory Commission (NRC) Regulatory Guide 1.145 (Ref. 3). The atmospheric dispersion coefficient is the time-integrated normalized air concentration at the receptor for toxicological releases. For puff releases, the atmospheric dispersion coefficient is the instantaneous maximum concentration at the receptor. Only the plume meander correction was

August 8, 1996

(K-1)

used in the calculations. The coefficients used to calculate dose or concentration are presented in Tables K-1. The acute release values were developed for weather conditions that result in downwind concentrations exceeded only 0.5% of the time in the maximum sector (16 sectors) or 5% of the time for the overall site. The larger of the two values was used as the integrated bounding value for on-site and off-site individuals. The chronic annual average χ/Q values were obtained by calculating chronic annual average values for each sector and using the highest one for calculation purposes. The value is suitable for long-term releases but not for accidents. All the releases analyzed in this safety assessment (SA) are either puff releases or releases that occur in less than 1 h. Thus, the plume meander correction or the chronic annual average values for the atmospheric dispersion coefficients are not used.

Reference 2 suggests that releases with a duration of less than 3.5 s must be treated as puff releases for the on-site receptor. Likewise, releases less than 420 s must be treated as puff releases for the off-site receptor.

This SA is also interested in releases from a 5-m (15-ft) tall stack. The atmospheric dispersion coefficients for this problem are not provided in References 1 and 2. The AI-RISK model,⁴ which was previously used in the Tank 101-SY Mixer Pump SA,⁵ was used to obtain the atmospheric dispersion coefficients for the stack releases.

For a ground point source release with no initial momentum and buoyancy, the 95% atmospheric dilution coefficient for a 100-m receptor is obtained as 3.26 E-02 using AI-RISK.⁶ Thus, the AI-RISK results are in good agreement with the atmospheric dilution coefficient given in Table K-1 for an on-site receptor. There is no concern about off-site values because for toxic-gas releases analyzed in this SA, the off-site consequences are always less than 1 ppm (Sec. 5) and taking credit for the stack release in the calculation is not necessary. Table K-2, obtained from References 6 and 7, summarizes the AI-RISK results for the conditions of interest.

| | Continuous Release | Puff Release | |
|----------|-------------------------|------------------------------|--|
| RECEPTOR | χ/Q (s/m ³) | χ/Q (1/m ³) | |
| On-site | 3.44 E-02 | 9.85 E-03 | |
| Off-site | 1.88 E-05 | 4.45 E-08 | |

TABLE K-1DISPERSION COEFFICIENTS

The receptor doses for a 1 m^3 or 1 m^3 /s release of a pure toxic gas and summarized Table K-3 were obtained using the atmospheric dilution coefficients given in Tables K-1 and K-2.

K.2.2. Release of Toxic Solid/Liquid Waste

To determine the acceptance limits for liquid/solid-waste material, WHC uses different types of composite waste for different tank groupings.² The risk associated with waste releases is divided into three categories: particulate, toxic effects, and corrosive effects. The chemical species in the composite waste are divided into toxic and corrosive bins. Within each bin, the allowable releases are computed using the "sum of the fractions" methodology. The minimum among the three categories (particulate, toxic, and corrosive) is chosen in each frequency range. For further details on the methodology used in determining the maximum acceptable waste releases for single-shell tank liquid and solid releases are given in Table 5-4 in Section 5 of this SA. Note that the values reported in Table 5-4 represent the release quantities at the source because the atmospheric dispersion coefficients already are factored in arriving at these magnitudes.

TABLE K-2 ATMOSPHERIC DISPERSION COEFFICIENTS OBTAINED FROM AI-RISK

| Release Type | χ/Q (s/m ³) |
|--|------------------------------|
| GROUND RELEASE, $D = 0 \text{ m}$, $V = 0.1 \text{ m/s}$, $T = 300 \text{ K}$ | 3.26 E-02 |
| STACK RELEASE (15 ft), $D = 0 \text{ m}$, $V = 0.1 \text{ m/s}$, $T = 300 \text{ K}$ | 5.12 E-03 |
| STACK RELEASE (15 ft), $D = 0$ m, $V = 1$ m/s, $T = 300$ K | 5.12 E-03 |
| STACK RELEASE (15 ft), $D = 0.3 \text{ m}$, $V = 0.1 \text{ m/s}$, $T = 300 \text{ K}$ | 5.03 E-03 |
| STACK RELEASE (15 ft), $D = 0.3 \text{ m}$, $V = 1 \text{ m/s}$, $T = 300 \text{ K}$ | 4.24 E-03 |
| STACK RELEASE (15 ft), $D = 0.3 \text{ m}$, $V = 10 \text{ m/s}$, $T = 300 \text{ K}$ | 1.05 E-03 |
| STACK RELEASE (15 ft), D = 0.1 m, V = 15 m/s, T = 300 K | 2.20 E-03 |
| STACK RELEASE (15 ft), D = 0.1 m, V = 23 m/s, T = 300 K | 1.31 E-03 |

D: source diameter, V: source vertical velocity, T: source temperature

| | ON-SITE (ppm) | | OFF-SITE (ppm) | |
|---|---------------|------|----------------|------|
| RELEASE TYPE | Cont. | Puff | Cont. | Puff |
| Ground, $D = 0$ m, $V = 0.1$ m/s | 34,400 | 9850 | 19 | 5 |
| ck, D = 0 m, V = 0.1 m/s | 5120 | 5120 | 19 | 5 |
| ack, D = 0.3 m, V = 0.1 m/s | 5030 | 5030 | 19 | 5 |
| Stack, $D = 0.3 \text{ m}$, $V = 1 \text{ m/s}$ | 4240 | 4240 | 19 | 5 |
| Stack, $D = 0.3 \text{ m}$, $V = 10 \text{ m/s}$ | 1050 | 1050 | 19 | 5 |
| Stack, $D = 0.1 \text{ m}$, $V = 15 \text{ m/s}$ | 2200 | 4240 | 19 | 5 |
| Stack, D = 0.1 m, V = 23 m/s | 1310 | 1050 | 19 | 5 |

TABLE K-3RECEPTOR DOSES FOR A 1-M3 OR 1-M3/S TOXIC-GAS RELEASE

K.3. REFERENCES

- 1. "Tank Waste Composition and Atmospheric Dispersion Coefficients for Use in Accelerated Safety Analysis Consequence Assessment," Westinghouse Hanford Company report WHC-SD-WM-SARR-016, Rev. 1 (October 1995).
- 2. "Toxic Chemical Considerations for Tank Farm Releases," Westinghouse Hanford Company report WHC-SD-WM-SARR-011, Rev. 1 (October 1995).
- 3. "Atmospheric Dispersion Models for Potential Accident Consequence Assessments at Nuclear Power Plants," US Nuclear Regulatory Commission Regulatory Guide 1.145, (USNRC, Washington, DC 1982).
- 4. Y. Y. Chien, and D. MacFarlane, "AI-RISK: A Computer Program for Calculating Doses and Health Risk from Accidental Release of Radioactive Materials," Los Alamos National Laboratory report LA-UR-92-2636, (July 1992).
- 5. L. H. Sullivan et al., "A Safety Assessment for the Proposed Pump Mixing Operations to Mitigate Episodic Gas Releases in Tank 241-SY-101: Hanford Site, Richland, Washington," Los Alamos National Laboratory report LA-UR-92-3196, Rev. 14 (March 31, 1995).
- 6. C. Unal, "Evaluation of Atmospheric Dispersion Coefficients at Elevated Discharge Temperatures, Velocities, Diameters and Release Heights by Using AI-RISK Computer Program," Los Alamos National Laboratory Calc-Note TSA10-CN-WT-SA-TH-078 (August 1995).
- 7. C. Unal, "Toxicological Consequences of Ammonia and Nitrous Oxide Releases During Rotary Mode Core Sampling," Los Alamos National Laboratory Calc-Note TSA10-CN-WT-SA-TH-104 (January 1996).

APPENDIX L

PROBABILITY OF GAS-RELEASE EVENTS DURING INTRUSION

L.1. BACKGROUND

In this appendix, a probabilistic model is developed to determine the gas-release event (GRE) characteristics during an intrusive operation in flammable-gas single-shell tanks (SSTs). The intrusive operations considered in this SA are rotary-mode and push-mode core sampling in flammable watch list tanks. Undoubtedly, some SSTs may have a GRE likelihood and consequences that are much lower than the ones obtained in this appendix. However, a tank-by-tank analysis was not the intent of this appendix.

As discussed below, there is very limited empirical information to develop such a probabilistic model based on data alone. Consequently, the model incorporates "expert judgment" combined with

- the limited data during intrusive activities in flammable-gas tanks,
- the information gained from laboratory experiments with simulants, and
- a general understanding of gas retention and release mechanisms in SSTs.

Because the approach has limited empirical support, the current model must not be used for eliminating safety features or defense-in-depth controls until further knowledge can be gained. The probabilistic model is merely intended for use in comparing the bounding consequences against the Westinghouse Hanford Company (WHC) Risk Guidelines (RGs) that require a realistic estimate of frequencies and allow for qualitative estimates for phenomena for which there are a lack of data. This safety assessment (SA) was developed using the following step-wise approach:

- 1. The most important issue is to develop a spark-management strategy that is appropriate for a hazardous flammable-gas environment such as the one summarized above and detailed in Appendix B of this SA.
- 2. The quantification of the reliability of the equipment used to protect against burns also must provide reasonable assurance that all practicable preventive measures are taken against burn accidents. It is believed that the equipment failure probabilities on the order of 10⁻⁴ to 10⁻⁵ per operation is acceptable, considering the type of operation. Appendices D and E show that this objective also is achieved in this SA.
- 3. The probabilistic GRE model discussed in this appendix was introduced only after the completion of these first two steps in order to obtain realistic accident frequencies.

The order in which these steps are taken is identical to the order in which this SA was developed. This order should not be reversed until further data and understanding for SST gas retention and release mechanisms are available.

In SSTs, burn accidents result in unacceptable consequences because even a small pressurization in the dome may result in dome failure, potentially followed by a catastrophic collapse. In order to meet the RGs, one must demonstrate that such burn accidents have a frequency of less than 10^{-6} /yr. However, burn accidents are not the only accidents that may result in dome failure. The design basis earthquake (DBE) has a return period of 7500 yr (frequency > 1×10^{-4} /yr). SSTs are designed to withstand a 0.20-g ZPA (zero-period acceleration), but there is high probability that a DBE of magnitude 0.20 g may produce structural failure of the dome because of potential cracks that occurred immediately after back-filling. Thus, the baseline dome collapse frequency for the SSTs is $\sim 10^{-4}$ /yr, and this frequency is independent of rotary-mode core sampling (RMCS). Even if one assumes that the failure frequencies are equivalent to burn and dome collapse frequencies, the dome collapse risk is less than the base-line risk associated with a DBE as discussed above. Introducing the GRE probability only demonstrates that the incremental risk is negligible compared to the base-line risk.

The next stage is to convert the failure probabilities into accident frequencies. The major accidents in the flammable-gas tanks are burn accidents that require the existence of a flammable-gas mixture combined with an equipment failure resulting in a spark. In order to obtain realistic accident frequencies, one needs to know the probability of experiencing flammable-gas atmosphere in the regions where sparks can exist. Unfortunately, the potential gas-release volume, gas-release rate, and gas composition data necessary to develop such a model for SSTs are scarce. Likewise, the gas retention and release mechanisms for SSTs are not fully understood so that one may develop a realistic or bounding model of an expected GRE during waste intrusive activities. On the other hand, through the limited data obtained during similar intrusive activities, through educated estimates of the waste contents and properties pertinent to gas retention and gas release, and through laboratory experiments conducted with simulants, it is possible to obtain an expert opinion on the likelihood of GREs during intrusive activities.

In order to determine the GRE probability model, the recent data available from intrusive operations in flammable-gas SSTs and double-shell tanks (DSTs) was considered. The data and the conclusions are summarized in the next section. In Section 3, the probability distributions are obtained based on the implications of the limited data and "expert interpretation" of the existing knowledge. Section 4 summarizes and concludes this appendix.

L.2. AVAILABLE SOURCE OF INFORMATION

There are some data obtained during intrusive activities in SSTs that can be used to develop likelihood arguments for the GREs. Likewise, there is a limited amount of

modeling performed for the gas retention and release mechanisms in the SSTs. However, given the uncertainty of the physical and chemical properties of the waste, such theories also must be used in accounting for large uncertainties associated with the quantitative results.

L.2.1. Summary of Available Data

WHC Characterization Field Engineering screened the recent flammable-gas concentration data obtained during and after intrusive activities in flammable-gas DSTs and SSTs.¹ A total of 49 intrusive sampling events (core sampling and auger sampling) alone were evaluated for 21 tanks sampled. One possible evidence of a GRE was observed in tank level data (A-103). Waste surface or liquid level drops for 34 of the sampling events showed no GRE. During 12 of the events, the plots had insufficient or non existent data on which to draw a conclusion. During two of the events, minor changes were observed during sampling activities, but the changes were well within the normal data scatter. Only one change for Tank A-103 during March of 1986 was questionable. The waste surface showed a 2.4 in. drop between March 24 and March 31, 1986. The level drop occurred when the lower segment samples were removed on March 25 to March 26, 1986. During the second core taken on March 31 through April 3, no level drop was observed.

No GREs were evident based on temperature data. Twenty-six of the activities' plots showed no evidence of a GRE activity. For 23 events the plots either had either insufficient or nonexistent data on which to draw a conclusion. If a rollover type of GRE were to have occurred, that would have been observed in the temperature data.

Tank vapor space monitoring data indicated three of the seventeen tanks monitored using a combustible gas meter (CGM) or other hydrogen monitoring equipment during waste-intrusive activities showed a very minor increase in the flammable-gas concentration during the sampling event. Three events showed small gas releases estimated to be ~50 ft³. The dome vapor space never exceeded 9% of the lower flammability limit (LFL) during any of these events.

All the sampling activities contained in the database took place in tanks from the Flammable Gas Watch List (FGWL) or tanks recommended for the FGWL. Seventeen of these tanks are SSTs.

Reference 1 also provides observations during 38 liquid observation well (LOW) installations in SSTs in 1984. The database includes LOW installation in all the Flammable Watch List tanks except AX-103, which currently has a waste level of 42 in. During these activities, no flammable-gas monitoring was done. The temperature data are all insufficient to draw any conclusions, or are nonexistent. Waste surface level drop data showed that a GRE did not occur in 23 of the 38 installations. During seven events, the data were insufficient to draw any conclusions. During another seven events, small variations (well within the normal data scatter) were observed. Only one level change for SX-104 on May 24, 1984, was questionable. The waste surface showed a 2.1 in. drop. The data show that the level dropped on 1 to 8 days before the LOW

installation. However, the potential exists that the data logs might have been off for a couple of days, and that the level drop actually occurred during the LOW installation.

Typically, ammonia can be smelled at 20 ppm. At higher concentrations, the smell can cause discomfort. No strong ammonia odor problem was reported during these activities. Because the SSTs typically do not have an active ventilation system, even small concentrations of ammonia in the dome space may result in exceeding the smell detection levels above an open riser. However, considering most of these activities took place more than ten years ago, it is likely that such occurrences (if they existed) are not in the data logs.

Likewise, no complete or partial temperature inversion (typical of rollovers) was observed during these activities. Also, no perturbations in the local temperature that would indicate waste motion were observed.

Unfortunately, these data cannot be formalized into a complete statistical model. The benefit of these data is to suggest that, qualitatively, a large GRE resulting in hazardous flammable-gas concentrations in the dome space is not a likely event during the intrusive activities. Quantitatively, the data may be used in a number of different ways:

It can be stated that 2 out of 87 activities resulted in a detectable GRE, suggesting a frequency of 2.3×10^{-2} . Accounting only for the activities in the SSTs, the probability becomes 2 out of 77 (2.6×10^{-2}). Also subtracting the activities in which the level data were inconclusive, the probability becomes 2 out of 58 (3.4×10^{-2}). However, this approach does not apply to a bounding tank, and at best it provides an overall probability of a GRE in the flammable-gas tanks. The LOWs are installed in A-101 (waste level ~345 in.) and AX-101 (waste level 278 in.), which have the largest waste volumes in the SSTs that are on the original FGWL. Tank A-103, which showed a 2.4 in. level drop between March 25 and March 26, 1986, also has a large waste volume (waste level ~370 in.). The other possible 2-to 3-in. level drop corresponds to Tank SX-104 that has less than 240 in. of waste.

A 3-in. level drop would correspond to 1600 ft^3 of gas release, assuming that the gas is originally compressed at 1.5 atm. Unfortunately, the level measurements were not taken frequently enough to estimate the release rate. No dome pressure data were reported in Reference 1 to assess the gas-release rate based on dome pressurization.

Another pertinent set of data analysis is provided in Reference 2. In this study, all the data from the original FGWL tanks were screened for GRE behavior. The study concluded that the tanks can be categorized in four groups:

Category 1: The tanks that exhibit neither periodic release behavior nor waste growth. Tanks 103-AX, 111-S, 112-S, 103-SX, 106-SX and 109-SX are in this category.

- Category 2: The tanks that do not exhibit waste growth, but periodic release behavior is indeterminate. Tanks 101-AX, 101-SX, 102-SX, 103-SX, 104-SX and 105-SX are in this category.
- Category 3: Tanks that exhibit periodic release behavior. Only Tank 101-A is in this category. These tanks exhibited 2-to 3-in. surface level variations between 1988 and 1993 that may be indicative of slow periodic releases.
- Category 4: Tanks that exhibit waste growth but do not exhibit periodic release behavior. Tanks 102-S, 110-T, 103-U, 105-U, 107-U, 108-U, and 109-U are in this category.

It must be noted, however, that Reference 2 was primarily interested in natural GRE behavior as opposed to a GRE during an intrusion. Recent detailed analysis of the waste level history in Tank A-101 indicates that the observed level fluctuations are possibly not a result of periodic releases. They are likely to be caused by random measurement errors.³

L.2.2. Summary of Available Simulant Experiments

The experience with SST waste simulants is primarily developed at Pacific Northwest National Laboratory (PNNL). The discussion below provides a summary of the PNNL experience to date with SST simulants." The simulants are aimed at simulating the physical characteristics of SST waste (sludge and salt cake) as opposed to chemical properties. The simulants are typically prepared using various clay, bentonite, and water mixtures, and gas is generated with the use of hydrogen peroxide.

Simulant experiments have been conducted to measure the retained fraction of gas generated within a simulant as a function of shear strength. These experiments were conducted without an overlying supernatant layer, which is typical of many SSTs. The results show that essentially all the generated gas is retained until a maximum retention capacity of the sludge is reached. Beyond the point of maximum retention, which is about 35% void, any additional generated gas and some of the retained gas is released. The results show that the maximum gas retention depends on the sludge shear strength. For the strongest sludges, the maximum retention is about 30% void. The maximum retention increases to a peak value of about 40% void as the sludge strength decreases. Together, these data show that sludges are quite capable of retaining a high void fraction of gas, and it can be inferred that it is difficult to release bubbles trapped in the sludge. To date, a few very qualitative experiments have been conducted in which small rods are inserted into the waste to initiate a release.

^{**} Private communications with P. Gauglitz (PNNL) and K. Pasamehmetoglu (LANL) (February 1996) and a desk-top demonstration by P. Gauglitz (PNNL) at the January 17, 1996, Flammable Gas Data Review meeting.

shown that the intrusion releases a small amount of retained gas, and that the release comes out of a region immediately adjacent to the intrusion.

Another interesting set of simulant experiments conducted by PNNL is related to the effect of the supernate thickness on the rollover strength. Some SSTs show a thin layer of supernate on top of the sludge layer. The experiments show that with thick supernate layers the rollover is quite energetic, and a substantial amount of gas is released. On the other hand, the rollovers with thin supernate layers are very unenergetic, and very little gas (if any) is released during the rollover.

While developing a rigorous statistical model based on simulant experience alone is not appropriate, the simulant data are quite valuable in developing a best-estimate prediction for the GRE behavior in SSTs.

L.2.3. Summary of Available Models

The DSTs on the FGWL all indicate that there may be a convective layer over the nonconvective layer where the gas is trapped. The prompt gas-release mechanism is postulated to be the density inversion or the Rayleigh-Taylor instability.⁴ The observation of the axial temperature profiles and the waste types that are in the SSTs do not support the Rayleigh-Taylor instability as being a likely gas-release mechanism. Likewise, the waste-level history of the SSTs does not show rather rapid level increases followed by sudden level drops that were observed, for instance, in Tank 101-SY during and after a rollover.

A necessary condition for a rollover is the existence of a density inversion, where a layer becomes less dense than the layer above it. This implies that the layer above has been able to release gas generated in it, while the layer below is retaining the generated gas. In an SST with a significant supernatant layer or convective layer (CL), then most of the gas generated in that layer will be released. If the layer below the supernatant or CL has sufficient strength to retain most of the generated gas, a density inversion can develop. A rollover will result when the hydrostatic forces between the two layers are sufficient to cause the lower layer to flow as a fluid. This illustrates the second condition necessary for a full tank rollover; the lower layer waste strength must be large enough to retain gas, but not large enough to preclude fluid flow under hydrostatic forces. If the lower layer strength is too large, then a full tank rollover does not occur and gas is released through local eruptions and/or mini-rollovers. Without a significant supernatant level or CL, it is possible in an SST that there be significant variations in waste properties between two layers so that the upper layer releases gas while the lower layer retains gas. However, if this were occurring, then a continuous level growth would be observed until the rollover of the lower layer. Also, even during an unlikely event of a rollover in an SST, the rollover is expected to progress very slowly, and gas-release rates are expected to be much smaller than the gas-release rates observed in DSTs (e.g., 101-SY).

A semi-qualitative evaluation of gas-release mechanisms in SSTs is provided in Reference 5, which suggests that while rollover is possible in the SSTs, it is not a likely

mechanism. Because of the limited volume of the supernatant liquid, during an unlikely rollover event, the gas releases would be small. Also unlikely is a mechanism whereby large bubbles (1 m diameter) are stored in the waste and released during an intrusion. Intrusion into the largest expected bubble based on conservative estimates of waste properties, indicates that 350 ft³ of retained gas could be released over an ~20 minute time period. The total release during this unlikely mechanism also is postulated to be small. The most likely mechanism that could lead to a large gas release appears to be a "mud pot" fed by dendritic bubbles if a chimney is opened to the dendritic bubble region as a result of the intrusion. The calculations reported in Ref. 5 indicate that 700 ft³/h releases may be expected through this "unlikely" mechanism. Finally, the fracture of a dry sludge or dry salt cake are other possible, but unlikely, mechanisms quoted in Ref. 5. Releases through this mechanism are estimated to be at a smaller rate than the 700 ft³/h value obtained for the "mud pot" mechanism.

During RMCS operations, the drill string is rotated at 55 rpm and a nitrogen purge flow of 30 ft³/min is used. If it is assumed that the drill string disturbs a region that is 20 times its diameter (4 in.) and is inserted at 2 in./min, the gas volume disturbed would be ~1 ft³/min (assuming a conservative void of 20% in the waste) resulting in a total release of less than 200 ft³. However, nitrogen purge also may scrub soluble ammonia out of the waste. Conservatively, assuming that the ammonia partial pressure in the waste is 0.6 atm and that ammonia comes in equilibrium with nitrogen instantaneously, an ammonia release rate may be obtained as 45 ft³/min. This value is in agreement with the conservative magnitude obtained in Ref. 6. However, such a high ammonia release rate would be possible until the liquid immediately around the drill string is depleted and a mass transfer resistance to ammonia forms in the liquid. Thus, the total release volume would be < 500 ft³ (Ref. 6). Based on these discussions, one could argue that the release rates are possible for a short period during drilling.

Among the 19 SSTs that are on the original FGWL, A-101 has the largest waste volume. Based on the available data, a data reconciliation analysis for the gas inventory of A-101 was performed.³ The results indicate that the gas inventory estimate for A-101 is 6300 ft³ with a standard deviation of 1300 ft³. Thus, the gas inventory for this tank, at two standard deviations level, would be less than 9000 ft³.

The gas-release mechanisms during salt well pumping (SWP) of A-101 are discussed in Reference 7. During SWP operations, liquid waste will drain from the waste into the salt well and reduce the effective hydrostatic head as seen by the retained gas bubbles. The reduced hydrostatic head and liquid level in the waste can be related to the gas-release volume and rate. Gas-release rates for a given liquid drainage rate depend upon the assumed bubble size and distribution. Based on a range of assumed conditions the calculated gas-release rate was typically less than 1 to 2 ft^3/min .

L.3. PROBABILISTIC MODEL

In this section, the knowledge developed by the limited data, limited modeling results, and the expert-judgment are set into a probabilistic model. In order to develop a probabilistic model of reaching or exceeding hazardous conditions in the dome space, one needs probabilistic models for the gas-release rate, the gas-release volume, and the gas composition. These models may then be combined-into a hydrodynamic model of dome space to compute the consequences of different-sized gas releases.

First, the discussion provided in the previous section is cast into a qualitative model. The qualitative likelihood matrix is provided in Table L-1. In Table L-1 qualitative likelihood definitions included following frequencies; a-Range 1 is defined as an event with a frequency of 1, b-Range 2 is defined as an event with a frequency of 10^{-1} , c-Range 3 is defined as an event with a frequency of 10^{-3} , e-Range 5 is defined as an event with a frequency less than 10^{-3} .

TABLE L-1 QUALITATIVE LIKELIHOOD MATRIX FOR GAS-RELEASE VOLUMES AND RATES

| | Volume (ft ³) | | | |
|-----------------------------|---------------------------|---------|---------|---------|
| Rate (ft ³ /min) | 100 | 500 | 2500 | 10,000 |
| 1 | Range 1 | Range 2 | Range 3 | Range 4 |
| 10 | Range 2 | Range 2 | Range 3 | Range 4 |
| 100 | Range 3 | Range 4 | Range 4 | Range 4 |
| 1000 | Range 4 | Range 4 | Range 5 | Range 5 |

The gas-release volume break-down in Table L-1, is based on unnoticeable level drop (100 ft³), 1 in. level drop (~500 ft³), 5 in. level drop (2500 ft³) and nearly 20 in. level drop (10000 ft³). The data that is considered in Section 2.1. shows that a level drop > 2.5 in. was not observed in any of the intrusive activities covered in the database.

The rate range shown in Table L-1 is obtained by considering the orders-of-magnitude, based on the discussions provided in the previous section. Even at slow release rates, larger release volumes are less likely because:

- 1. Typically release will occur from the waste volume immediately around the intrusive device, and
- 2. Flammable gas monitors will detect the gas release and terminate waste penetration, which is expected to stop the induced gas release.

For release volumes that are never observed in the SSTs, a qualifier of "Range 3" is assigned. The existing models do not predict release volumes corresponding to Range 3 and less likely categories, except for a rollover and seismic event that can liquefy the waste by vigorous shaking. Even for these Range 3 events the release volume is limited by the fraction of total retained gas. The maximum values shown in Table L-1 (10,000 ft³) corresponds to the largest GRE volume observed in Tank 101-SY before mitigation. Similar events have not been observed in any of the SSTs.

The rate range shown in Table L-1 is obtained by considering the orders-of-magnitude, based on the discussions provided in the previous section. Only rollovers and seismic shaking may result in release rates on the order of 100 ft³/min or greater, while resulting in large release volumes. There is a remote possibility of tapping into a high pressure gas region which may result in large release rates but a very small volume. Thus, while small volume releases at high releases rates are believed to be Range 4 events, large volumes with large releases rates are in Range 5 event category.

The qualitative matrix may be quantified for the gas-release volume and gas releases rates assuming that these two are independent parameters. In the quantification process, the following probabilities were assigned: a probability of 1 for Range 1 event, 10⁻¹ for Range 2 event, 10⁻² for Range 3 event and 10⁻³ for Range 4 event. Thus, the quantitative results may be approximated as a logarithmic distribution given by

$$P(Q' \ge Q) = \left(\frac{Q}{100}\right)^{-1.5}$$
 for $Q \ge 100 \text{ ft}^3$. (L-1)

Likewise, the release rate may be approximated by a logarithmic distribution given by

$$P(q' \ge q) = (q)^{-1}$$
 for $q \ge 1 \text{ ft}^3 / \min$. (L-2)

L.3.1. Effect of Release Volume and Release Rate on Dome Concentrations

This section, the relationship between the gas-release rates and the peak dome concentrations are quantified. In order to quantitatively assess the reduction in the flammable-gas inventory, the following simple mathematical model is used for the dome space waste gas concentration: ⁸

$$V\frac{dX}{dt} = q - X \cdot f \quad , \tag{L-3}$$

where

V is the dome volume = $1416 \text{ m}^3 (50,000 \text{ ft}^3)$,

X is the dome-averaged waste gas concentration,

t is the time,

q is the waste gas-release rate into the dome, and

f is the entilation flow rate.

For simplicity, it is assumed that the gas-release rate is constant and the gas release is modeled as a pulse function as illustrated in Fig. L-1. Thus, the gas release is characterized by the following parameters (only two of-them are independent): q: gas-release rate, T: gas release duration, and Q: total gas-release volume ($Q = q \times T$).



Fig. L-1. Gas-release rates as a function of time.

The solution for Eq. (L-3) is obtained as

$$X(t) = \begin{cases} \frac{q}{f} \left[1 - \exp\left(-\frac{f}{V}t\right) \right] & \text{if } 0 \le t \le T \\ \frac{q}{f} \left[1 - \exp\left(-\frac{f}{V}T\right) \right] \exp\left[-\frac{f}{V}(t-T)\right] & \text{if } t > T \end{cases}$$
(L-4)

where it is assumed that the initial concentration in the dome (before the beginning of the gas-release event) is zero. The peak dome concentration is obtained as

$$X_{\max} = \frac{q}{f} \left[1 - \exp\left(-\frac{f}{V}T\right) \right] = \frac{q}{f} \left[1 - \exp\left(-\frac{f}{q}\frac{Q}{V}\right) \right]$$
(L-5)

If q > f (which is always the case when active ventilation is not available), the discharge rate is set equal to the gas-release rate during a gas-release event. Thus, the solution of Eq. (L-3) becomes

$$X(t) = \begin{cases} 1 - \exp\left(-\frac{q}{V}t\right) & \text{if } 0 \le t \le T \\ \left[1 - \exp\left(-\frac{q}{V}T\right)\right] \exp\left[-\frac{f}{V}(t-T)\right] & \text{if } t > T \end{cases}$$
(L-6)

L-10

The peak concentration for this case becomes

$$X_{\max} = 1 - \exp\left(-\frac{q}{V}T\right) = 1 - \exp\left(-\frac{Q}{V}\right)$$
 (L-7)

As shown by Eq. (L-7), when the gas-release rate is greater than the ventilation flow rate, the peak concentration is independent of the release rate and is a function of the release volume only. This is also illustrated in Fig. L-2 that shows the peak dome concentrations for Q = 2000 ft³ (Q/V) = 0.04 as a function of q/f. As shown in this figure, for q/f > 0.4, the peak concentration is nearly independent of the gas-release rate. At q/f < 0.4, the peak concentration rapidly decreases with decreasing q/f. But even at q/f = 0.1, the peak concentration decreases to 3.3% compared to the maximum value of 3.9%.

Based on these discussions, one could argue that, for a ventilation flow rate of 200 ft^3/min , for gas releases > 10 ft^3/min , the gas-release rate has a minimum effect on the final results. On the other hand, for gas releases around 1 ft^3/min , the ventilation flow substantially suppresses the peak concentration for an extended period of time. The peak concentration in the dome for a 1 ft^3/min gas-release rate would be 5000 ppm, regardless of the release volume.

L.3.2. Gas Composition

There is considerable uncertainty in the gas composition as discussed in Appendix C. The maximum hydrogen concentration may be as high as 90% (App. B). However, that estimate assumes that there is no nitrogen, ammonia, or water vapor in the release gas (all of those species are known to exist in the waste gases). Without any ammonia, the LFL for this hydrogen-rich mixture would be 4%. Based on discussions provided in Appendix C, a bounding ammonia fraction of 60% may be used for the SSTs.



Fig. L-2. Peak dome concentration as a function of gas-release rate.

L-11

Adding ammonia (that is also known to exist) reduces the LFL. For a given release vectore, a larger ammonia fraction in the released gas would imply a smaller hydrogen fraction. Thus, a larger release volume is needed to reach the LFL. The net effect is that the most conservative case is the maximum hydrogen in the waste gas even though the LFL for that mixture is the highest. While it is still bounding, the use of 50% hydrogen in the waste gas would be a good estimate for probability calculations. In the subsequent analysis, a uniform distribution is used for hydrogen between 50% and 90% and conservatively assume that the rest of the gas is ammonia. Neglecting the other gas species, which are also known to be generated in the waste, provides an additional degree of conservatism in the above calculations.

L.3.3. Time-at-Risk Estimates

Another important parameter for the burn probability estimates is the time-at-risk. The time-at-risk is the period of time when the dome volume remains above the LFL. The results of the ventilation study documented in Ref. 8 are shown in Fig. L-3 for different release volumes. For this example, it is assumed that the ventilation flow rate is 250 ft^3/min and that the dome volume is 50000 ft^3 . The hydrogen concentration in the release gas is 90%, and the LFL is 4%. Figure L-3 is provided as an example without crediting the expected low release rates. The results of more detailed calculations are given in the next section.

L.4. RESULTS

A code was written to perform the following tasks:

- (1) perform Monte Carlo sampling from probability of frequency distributions for: gas-release rate, gas volume, and gas composition,
- (2) calculate dome gas thermodynamic properties for each set of sampled gas release values, and
- (3) bin the calculated thermodynamic gas properties into various bins to provide probability of frequency distributions for these thermodynamic properties.

Equation L-1 was used for the probability of frequency distribution for the gas-release volume. Equation L-2 was used for the probability of frequency distribution for the gas-release rate. The exhauster was assumed to run continuously through the GRE. Stopping the drilling operations upon detection of flammable gases and potentially stopping the gas release is not modeled. Using information from WHC, the exhauster flow rate was set at 200 ft³/min and the initial tank pressure was set at -0.5 in. w.g. Nitrogen cooling flow to the drill bit was set at 35 ft³/min. The details of the analysis are provided in Ref. 9.



Fig. L-3. Dome space concentration as a function of release volume and time.

A uniform probability density function was used for the fraction of hydrogen in the gas between a lower limit of 0.5 and an upper limit 0.9. The fraction of the gas that was not hydrogen was assumed to be ammonia.

In these calculations, a uniform dome mixing was assumed. Three-dimensional analysis show that for ventilation flow rates ≥ 200 ft³/min, the release gases mix fairly uniformly in the dome space.¹⁰

L.4.1. LFL Results

A sample size of 400,000 was used. The probability that LFL is exceeded in the dome and that the tank pressure is positive was calculated to be 7.0E-5. The probability of exceeding twice the LFL is 3.5E-5.

The mean time that LFL was exceeded in the dome was calculated to be 0.12 minutes. Using a mission time of 144 hours, this implies that the probability that LFL is exceeded during an event that is random in time is: 0.12/(144*60) = 1.4E-5.

In these calculations, a conservative estimate of the LFL is used to define the hazardous conditions because the calculations given in Ref. 10 show that the release gases mix fairly uniformly in the dome space with a ventilation flow rate ≥ 200 ft³/min. Thus, as

long as the ventilation is operable, the probability of a plume burn in or above the dome space is very unlikely. However, if the ventilation system fails, the probability of a plume burn in the dome increases because uniform mixing in the dome may not occur.

There are two accident scenarios for failed ventilation:

- 1. Ventilation fails without tripping the drilling operations that subsequently cause a GRE;
- 2. Ventilation fails during a GRE; and
- 3. Ventilation is intentionally shut down because of high flow (> 250 ft³/min)

The simple calculations given in Appendix C show that the dome pressure limit may be exceeded if a waste gas volume of 1000 ft³ is burned in the dome space. This volume corresponds to a dome concentration greater than 25% of the LFL. Using the above models, the probability of exceeding 25% of the LFL in the dome space is < 5E-2. The combined probability of ventilation failure (conservatively estimated as 0.1/activity) and failure to shut down the drilling operations (1.4E-3) is estimated as 1.4E-4 per activity. Thus, the frequency of exceeding 25% of the LFL during RMCS operations without ventilation is 7E-6/activity, which is an order of magnitude smaller than the frequency of exceeding the LFL.

For the second scenario, the mean time above the LFL and 25% of the LFL are obtained as 0.12 min and 5 min, respectively, during an activity. Using the conservative estimate of 0.1 failure per activity (144 h), the failure probability during the time period while the dome is above the LFL is 1.4E-6. The failure probability while the dome is above 25% of the LFL is 5.8E-5. These numbers are smaller than the probability of exceeding the LFL while the ventilation is operable.

Finally, the intentional exhauster trip is considered in the analysis. The exhauster trips if the flow rate exceeds 250 ft³/min. During a rapid gas-release event, the exhauster flow rate is expected to exceed the nominal 200 ft³/min flow rate. The probability of exceeding the 250 ft³/min flow rate is calculated as 2.5E-3. The probability of exceeding the 250 ft³/min flow rate for more than 1.5 min is calculated as 4E-5. The exhauster trip is delayed by 5 min once the 250 ft³/min limit is exceeded. This design feature minimizes the possibility of losing the exhauster as a result of exceesive flow during a GRE.

L.4.2. Toxic Gas Releases

To provide information for evaluating release of toxic gas, the code was run with a fixed concentration of 60% ammonia and a sample size of 50,000. The results are summarized in Table L-2.

TABLE L-2 PROBABILISTIC RESULTS FOR DOME SPACE AMMONIA CONCENTRATIONS

| Concentration Bins | Positive Dome Pressure | Negative Dome Pressure |
|---------------------|---------------------------|---------------------------|
| 0 - 6000 ppm | 6.2E-03 — | _ 9.7E-01 |
| 6000 - 10,000 ppm | 5.2E-04 | 1.8E-02 |
| 10,000 - 25,000 ppm | 2.4E-04 | 5.9E- 03 |
| 25,000 - 50,000 ppm | 1.0E-04 | 5.6E-04 |
| > 50,000 ppm | 2.0E-05 | 1.0E-04 |

L.4.3. Conclusions

In this appendix, a quantification of the GRE probabilities are presented based on the best-estimate interpretation of the state of knowledge of gas releases during intrusion in the SSTs. The probability values quoted in this appendix may be used in evaluating the best-estimate accident frequencies in comparing the conservative accident consequences with the risk guidelines. An uncertainty analysis is not performed because such an analysis is not required for accident frequency estimates. However, although point estimate GRE probabilities are combined with failure rate frequencies in Section 4 of this SA, the GRE probabilities must be viewed as order-of-magnitude estimates. At least an order-of-magnitude uncertainty must be considered in making decisions critical to the safety of the RMCS operations.

In the accident analysis (Section 4 of the SA), two probability numbers were used. For potentially continuous (if the GRE is undetected) spark sources, the probability of exceeding the LFL is used as 7E-5. For spark sources that are random in time, the ratio of mean time at risk to the total mission time (144 h) is used as 1.4E-5. This analysis is based on the assumption that a waste-intrusive event triggers a large enough GRE to result in exceeding hazardous conditions. Without intrusion, a natural release that exceeds the LFL is assumed to be less likely than an event during intrusion. A detailed analysis of Tank 101-A level data indicates that periodic natural releases in that tank are very unlikely³ and that the maximum level changes are not sufficient to result in a GRE that exceeds the LFL. However, all the tanks are not analyzed to the same level of detail as Tank A-101. Consequently, a GRE probability is a checklist item that must be addressed for each specific tank (see Section 7 of this SA).

L.5. REFERENCES

- 1. J. S. Schoefield, "Evidence of Release Events During Waste Intrusive Activities in Flammable Gas Watch List (FGWL) and Proposed FGWL Tanks," Westinghouse Hanford Company internal memorandum to B. Barton (February 21, 1996).
- 2. B. D. Nichols, S. W. Eisenhawer, and J. W Spore, "Bounding Gas Release Calculations of Flammable Gas Watch List Single-Shell Tanks," Los Alamos National Laboratory report, LA-UR-94-1323 (1994).
- 3. W. L. Kubic, "Summary, Review and Analysis of Data for Tank 241-A-101," Los Alamos National Laboratory Calc-Note, TSA10-CN-WT-SA-DA-001 (January 1996).
- 4. L. H. Sullivan et al., "A Safety Assessment for the Proposed Pump Mixing Operations to Mitigate Episodic Gas Releases in Tank 241-SY-101: Hanford Site, Richland, Washington," Los Alamos National Laboratory report LA-UR-92-3196, Rev. 14 (March 31, 1995).
- 5. R. T. Allemann, "A Discussion of Some Mechanisms for Sudden Gas Release from Single-Shell Waste Tanks," Pacific Northwest Laboratory letter report PNL-WTS-101095 (October 1995).
- 6. J. D. Hudson, B. J. Palmer, and C. W. Stewart, "The Potential for Flammable Concentrations of Ammonia Resulting from Core Drilling Operations," Pacific Northwest National Laboratory letter report PNL-WTS-112295 (November, 1995).
- 7. J. W. Spore, "Conservative Gas Releases for Tank 241-A-101," Los Alamos National Laboratory Calc-Note, TSA10-CN-WT-SA-GR-046 (January 1996).
- 8. K. O. Pasamehmetoglu, "Issues Associated with the Use of An Exhauster During Saltwell Pumping Activities in Single-Shell Tanks that are on the Flammable Gas Watchlist" Los Alamos National Laboratory Calc-Note, TSA10-CN-WT-SA-TH-107 (January 1996).
- 9. J. L. Darby, "Monte Carlo/Thermodynamic Model for Dome Gases", Los Alamos National Laboratory Calc-Note TSA10-CN-WT-SA-TH-114 (February 1996).
- 10. P. Sadasivan, "Effects of Exhauster Flow Rates on Consequences of a GRE in 101-A," Los Alamos National Laboratory Calc-Note, TSA10-CN-WT-SA-GR-054 (February 1996).

APPENDIX M

UNFILTERED RELEASES AND NITROGEN ADDITIONS

M.1. INTRODUCTION

In this appendix, the unfiltered releases are quantified and the effect of nitrogen addition are discussed during rotary-mode core sampling (RMCS) operations. First, in the next section, the amount of aerosol that is suspended in the dome space during RMCS operations is discussed.

M.2. DOME LOADING DURING RMCS OPERATIONS

During drilling operations, aerosols will be carried into the dome. Some will be transported to the high-efficiency particulate air (HEPA) filter, while others will settle out on the crust. In this appendix, an estimate is given of the bounding value for the mass of aerosols to be found in the tank dome during drilling operations.

The TRAC-TE code was developed by Science Applications International Corporation to calculate the mass of aerosols that would be carried to the HEPA filter during drilling operations. Part of the results output from the code is the mass of aerosols found in the dome under various operating conditions. This code and the results of calculations are discussed in Reference 1; however, this code is not available to us at this time. Thus, the analysis presented in these notes is based on the results of analyses using the TRAC-TE code found in Reference 1.

In Appendix G of Reference 1, the results for two relevant bounding cases of salt drilling, calculated using TRAC-TE, are presented. The two relevant cases presented are based on 700 scfm flow. The results are based on a total drilling depth of 266 inches and a schedule of 40 minutes of drilling (19 inches) followed by 60 minutes delay before starting the next drilling period. In Case 1, Appendix G-1 of Reference 1, the exhauster is on only during drilling, but in Case 2, Appendix G-2 of Reference 1, the exhauster is on continually. The results take into account the settling of the aerosol created and the flow of material up the stack to the HEPA filter. The cases are referred to as bounding because they maximize the amount of material transported to the stacks. The results for the first 700 minutes, i.e., seven drilling and waiting periods for these two cases are given here in Table M-1. Note that equilibrium has been reached.

The results in Table M-1 have simply been extracted from Appendix G-1&2 of Reference 1. The results given are the amount of material in the dome, up the stack, or cumulative grams made airborne at the end of the time period given. The amount of material made airborne in any 40 minute drilling period, 3106 grams, is the total amount drilled minus the 1 in. diameter sample. This would be a 2.5 in. diameter and 19 in. long cylinder less the 19 in long and 1 in diameter sample.
TABLE M-1

GRAMS OF AEROSOL IN THE TANK DOME AND GRAMS TRANSPORTED UP TO STACK FOR 700-SCFM FLOW

| Time | Total | Case 1 | Case 2 | Case 1 | Case 2 |
|-------------|----------------|-----------------------|-------------------------|-----------------------|-------------------------|
| (minutes) | (grams made | on during drilling | continuous exhauster | on during drilling | continuous exhauster |
| | airborne) | (grams in dome) | (grams in dome) | (grams up stack) | (grams up stack) |
| | | | | | |
| 40 | 3106 | 249 | 249 | 131 | 131 |
| 100 | 3106 | 44 | 17 | 0 | 59 |
| 140 | 6212 | 262 | 254 | 149 | 138 |
| 200 | 6212 | 52 | 19 | 0 | 62 |
| 240 | 9318 | 26 6 | 255 | 152 | 138 |
| 300 | 9318 | 55 | 19 | 0 | 62 |
| 340 | 12424 | 267 | 255 | 154 | 139 |
| 4 00 | 12424 | 56 | 19 | 0 | 62 |
| 44 0 | 15530 | 267 | 25 5 | 154 | 139 |
| 500 | 15530 | 56 | 19 | 0 | 62 |
| 540 | 18636 | 267 | 255 | 154 | 139 |
| 600 | 18636 | 56 | 19 | 0 | 62 |
| 640 | 21741 | 268 | 255 | 154 | 139 |
| 70 0 | 21741 | 57 | 19 | 0 | 62 |

An exhaust flow of 250 scfm (rather than a flow of 700 scfm) with continuous exhauster flow is closer to actual operating conditions. The amount of material going into the stack over the entire time period, i.e., 1440 minutes, for the 250 scfm case is given in Table 2 of Reference 1, but the amount of material in the dome is not presented. In Table 2 of Reference 1, the same information is given for 700 scfm and has been taken from the results presented in Appendix G-2 of Reference 1. However, it is the equilibrium material in the dome at 250 scfm that is of interest here, and this value will be determined below.

The material given that goes into the stack, according to Reference 1, contains only those particles with a diameter less than 5.5 μ m. Reference 1, page 11, states, "These values assume that the low transport velocities in the exhaust riser can carry only smoke sized particle, 5.5 μ m and less, out the riser." Note that this is conservative since the objective here is to determine the maximum amount of material in the dome.

From Table 2 of Reference 1, the amount of material going up the stack is 901 grams for the 250-scfm case and 1741 grams for the 700-scfm case at the end of 1440 minutes. Thus, when the flow is 250-scfm, only 52% of the material flows out the stack when compared with the flow in the 700-scfm case for continuous exhauster flow.

Referring to Table M-1, note that for Case 2 at equilibrium there are 255 grams in the dome and 139 grams going up the stack. If the flow is reduced to 250 scfm, then it is assumed that 48% (100%-52%) of this 139 grams will remain in the dome. Because some of this material will also settle, a conservative first estimate value of material in the dome is:

 $255g + 0.48 \times 139 g = 322 g.$

Note that there is a compounding factor. Table 2 of Reference 1 is based only on those particles less that 5.5 μ m. However, Table M-1 here assumes that all size particles flow up the stack. Referring to Table G-2 of Reference 1, it is found that 39% of the mass of airborne particles calculated to escape are larger than 5.5 μ m at the end of the 700 minutes. It is expected that this mass to remain in the dome. Thus, as a bounding value, this mass should also be added to the mass in the dome. Thus, the bounding value for the material in the dome is:

 $255 \text{ g} + 0.39 \times 139 \text{ g} + 0.48 (0.61 \times 139 \text{ g}) = 350 \text{ g}.$

This calculation states that 39% of the material that is calculated to go up the stack will not do so because of size, and it is assumed that it will remain in the dome. Of the remaining 61% that would go up the stack in the 700-scfm case, 48% will remain in the dome in the 250-scfm case of interest here. This material that was earlier calculated to go up the stack is assumed not to settle out of the dome.

A less conservative value may be found by assuming a fraction of this material now remaining in the dome will settle out. An estimate of the amount that may settle out is 94%, which is based on the results given in Appendix G-2 in Reference 1. Thus, the less conservative, but more realistic, value is

$$255 \text{ g} + 0.06[(0.39 \times 139 \text{ g} + 0.48(0.63 \times 39 \text{ g})] = 261 \text{ g}.$$

Hereafter, the bounding value of material found in the dome during drilling operations is 350 g for the continuous exhauster operation at 250 scfm. Prior to RMCS operations, there may be additional suspended waste in the dome. To bound the background concentration, 600 g (1.32 lb) of aerosol in the tank dome is used. This number is used in the safety assessment of mixer pump operations² in the 101-SY tank and is based on a very conservative fog limits during a rollover. Typically, quasi-steady particulate concentrations less than 100 mg/m³ are expected for SSTs. For a 1416-m³ (50,000-ft³) dome, the maximum amount would be 142 g (0.31 lb). Note that the unfiltered releases discussed in the next section are inversely proportional to the dome volume. Consequently, they are directly proportional to the concentration and the releases would not be affected by the larger dome volumes. Thus, total aerosol in the dome during RMCS operations may be bounded conservatively by ~0.6 kg (1.3 lb).

M.3. UNFILTERED RELEASES

The following unfiltered releases are considered in this appendix:

- Release as a result of a gas-release event (GRE) during operation or removal (including material entrained from the drill string as a result of decontamination system failure);
- Continuous release from the exhauster after HEPA failure; and
- Continuous releases during open riser period with ventilation failure.

Releases from spill accidents or HEPA filter failures are considered separately.

M.3.1. Unfiltered Releases Caused by a GRE

Unfiltered releases can occur in operation and removal. During installation, before intruding into the waste, a gas release event is not expected; therefore, only the operation and removal phase will involve unfiltered material releases.

One can estimate the amount of material entrained out of the tank during a GRE using the following simple equation:

$$M_{\rm T} = M_0 \frac{Q}{V} \quad , \tag{M-1}$$

August 8, 1996

where Q is the total gas release into the dome, M_0 is the initial dome loading [0.6 kg (1.3 lb)], and V is the dome volume [1415.8 m³ (50,000 ft³) based on Reference 3]. This simple model is based on the following assumptions:

- 1. The aerosol waste generated by the drilling and the waste gas released during a GRE are homogeneously mixed in the dome space;
- 2. The waste gas mixes homogeneously in the dome during a GRE;
- 3. The gas release rates are high enough to pressurize the dome [which is conservative for GREs expected in the single-shell tanks (see App. L)]; and
- 4. All the suspended material flows out of an open riser that is not protected by a HEPA filter (this is also very conservative because the inlet and exhaust headers are equipped with HEPA filters).

Using the probabilistic model developed in Appendix L combined with the conservative assumptions listed above, the resulting waste release from an open riser is tabulated as a function of the gas release as shown in Table M-2.

The release rates are computed for toxicological consequence analysis which require a 15-min average value. Thus, it is assumed that the total release occurs over a 15min period. This assumption is conservative because the release rates for the three cases analyzed about would be greater than 66 ft³/min, 333 ft³/min, and 666 ft³/min, which are comparable to 101-SY rollover releases.² Such high release rates are not expected for the SSTs (see Appendix L).

These releases are applicable to removal and operation time periods. During operation, however, the nitrogen purge system must fail for an unfiltered release to occur through the open riser. Thus, the likelihood of an unfiltered release during operation is much lower than a release during removal.

| Frequency (per activity) | GRE Volume (ft ³) | Material Release (g) | Release Rate (g/s) |
|---------------------------------------|----------------------------------|-------------------------|-----------------------|
| 10 ⁻² (anticipated) | 1000 | 12 g | 0.013 |
| 10 ⁻⁴ (very unlikely) | 5000 | 60 g | 0.07 |
| 10 ⁻⁶ (extremely unlikely) | 10000 | 120 g | 0.13 |

 TABLE M-2

 UNFILTERED AEROSOL RELEASE AS A FUNCTION OF GAS RELEASE

Another scenario is concerned with a GRE during removal with a contaminated drill string. The amount of waste on the drill string (inside or outside) during removal could vary. Under normal conditions, waste should not be inside, and the

outside is decontaminated. If the decontamination is not effective or failed and a GRE occurs during removal, additional waste would be entrained from the contaminated drill string. To estimate the entrainment, it is assumed that there is a 3-mm-thick waste layer or one 19-in. drill rod. Note that each drill rod is removed individually. The total amount of waste on one drill rod becomes 370 g. It is assumed that 1% of this amount would be entrained resulting in an additional 3.7 g. This scenario assumes the failure of decontamination and seal or sampler latching. Thus, the frequency of the additional release is much lower than the frequencies listed in Table M-2. Considering the conservative assumptions made in obtaining the results in Table M-2, any additional releases resulting from decontamination failure combined with a GRE are neglected.

M.3.2. Continuous Release through the Exhauster After HEPA Failure

This scenario is concerned with the releases through the exhauster if the HEPA filters fail. The failure of the HEPA filters will be detected by the high flow or low pressure drop readings. Failure to shut down the exhauster will result in unfiltered material release. The frequency of this accident is dominated by the failure to shut the exhauster down following a HEPA failure, which is in the unlikely to extremely unlikely range. The maximum ventilation flow rate is 250 ft³/min (0.12 m³/s). Based on a 1-kg (2.2-lb) dome loading, a continuous release of 0.08 g/s will be experienced. Thus, the consequences of material release caused by a GRE bounds this accident.

M.3.3. Continuous Release with Ventilation Failure

If the riser is open and the gauge pressure inside the tank becomes positive, an unfiltered release will occur. Ventilation system failure with open-tank conditions has been identified as one of these conditions. Controls are established to help ensure that this kind of release is minimized. No operation can be started if the ventilation system is not working properly. However, the ventilation system were to fail during the periods when the riser is open. If the ventilation system were to fail to maintain pressure inside the tank at less than atmospheric pressure, the continuous generation of gases in the tank would cause a release. In comparison to other releases driven by a GRE, this release is small. The gas release out of an open riser as a result of natural convection and steady-state gas release out of the waste is expected to be bounded by $0.03 \text{ m}^3/\text{s}$ (60 ft³/min), resulting in a material release of 0.02 g/s. The frequency of this accident is also small (in the unlikely to extremely unlikely range). The consequences of material release caused by a GRE bounds this accident.

M.4. NITROGEN ADDITIONS

Nitrogen addition is of concern because it may (1) result in dome pressurization, and/or (2) cause a gas release.

Nitrogen gas is added to the tank during normal RMCS operations. Nitrogen is added to the waste at a rate of 30 to 50 ft³/min. Each sampling may take from 2 to 38 minutes. It is assumed an average drilling period of 10 minutes in the reliability study. This safety assessment (SA) assumes that two full sampling activities will be performed in a tank per year. Each sampling activity would include 11 samples. Thus, the drilling time per year per tank would be 220 minutes. The total nitrogen addition into waste during drilling is 11,000 ft³ per year using a maximum flow rate of 50 ft³/min.

The hydrostatic head also supplies nitrogen with a flow rate of $0.3 \text{ ft}^3/\text{min}$. Removing samples is assumed to take 2 hours. For two samples, the total nitrogen addition per day would be 72 ft³. This is a relatively small amount added during drilling. The riser will be purged with about 5 ft³/min. However, this will be added to the tank dome.

Past experience in air additions to Tank 101-SY can be used in understanding the expected response of the tank caused by nitrogen injection.⁴ From 1987 to 1989, Tank 101-SY was air-lanced periodically. In 1989, the air lancing was performed almost daily. This air-lance data from each air-lancing operation was examined that introduced 28.3 m³ (1000 ft³) of air into the waste. Examination of the level data indicates that there is no systematically observed pattern, in terms of change of waste level, with respect to the air lancing. The level does not always increase or decrease after each air-lancing operation. Generally the level continued to increase (in an average sense) until a small or a big GRE occurred. In some cases, the level tended to stay constant or decreased slightly after several days following the air lancing. However, these small changes were insignificant because they did not change the level-time plot significantly.

The observations above suggest that introducing air in the waste does not immediately cause a rollover with a significant gas release. Adding air to the waste is expected to increase the level growth rate. This conclusion assumes that some fraction of added air will remain entrapped in the waste as bubbles. The added nitrogen can stay in the waste or rise to the surface depending upon the yield stress of the waste. In a worst-case condition, the released nitrogen can form large bubbles (Ref. 5) and gives criteria for a spherical bubble to move.

The bubble diameter-yield stress relation is plotted in Fig. M-1. This figure illustrates that large bubbles can be created if the yield stress of the waste sludge is high (and if it exists). There is a possibility that released nitrogen can be held in the waste, especially at the bottom of the tank where the yield stress is thought to be relatively high. The total amount of nitrogen added during RMCS is significant. If this nitrogen is somewhat retained in the waste, a GRE initiated by a rollover could be observed.



Fig. M-1. Bubble-diameter to yield-stress relation at the initiation of a bubble motion.

However, note that the daily gas-generation rate in a full tank is expected to be around 100 ft³/d using 101-SY data resulting in a rate of level rise of 0.15 to 0.2 in./d. None of the SSTs experience large level growth rates, suggesting that they have reached a steady state whereby the gases that are generated are being released to the dome. Thus, it is very likely that the additional nitrogen also would be released continuously to the dome through the release paths that are available for the generated gas. It is even more likely that the nitrogen would be released from the hole created by the drilling. However, while flowing into the dome space, nitrogen may entrain some of the retained gas. The most likely gas-release mechanism would be the scrubbing of ammonia by the nitrogen flow. Initial ammonia releases as high as 34 ft³/min are predicted by Reference 6. However, Reference 6 also suggests that such high release rates will not last forever as the waste near the drill string will be quickly depleted of its dissolved ammonia. Furthermore, the flammable gas monitors would stop the operations if high ammonia concentrations are detected at the exhauster. Ceasing the nitrogen flow will stop the ammonia release.

One of the purposes of the use of nitrogen is to remove the waste from the drill bit. The nitrogen carries the waste chips. The flutes on the drill string also help the transportation of waste chips. Because of nitrogen addition and aerosol created by drilling, an exhauster is required to be operated during RMCS operations. The exhauster operation will also remove the small amounts of waste gas that may be entrained into the dome during nitrogen purging. It is believed nitrogen remaining trapped in the waste and subsequently triggering a large GRE is an unlikely occurrence.

The nitrogen addition can cause the dome to be pressurized if the exhauster fails. Thus, the nitrogen flow must be stopped upon exhauster failure.

M.5. REFERENCES

- 1. P. M. Francis, "Aerosol Study for the Rotary Mode Exhauster," Westinghouse Hanford Company report WHC-SD-WM-ES-225, Rev. 0 (1993).
- L. H. Sullivan et al., "A Safety Assessment for the Proposed Pump Mixing Operations to Mitigate Episodic Gas Releases in Tank 241-SY-101: Hanford Site, Richland, Washington," Los Alamos National Laboratory report, LA-UR-92-3196, Rev. 14 (March 31, 1995).
- 3. K. O. Pasamehmetoglu, "Single Shell Tanks Dome Volume Calculations," Los Alamos National Laboratory Calc-Note TSA10-CN-WT-SA-TH-086, (November 1995).
- 4. C. Unal, "Concerns with Uncontrolled Reactions Caused by Rapid Pressurization and Depressurization of the Voidmeter Sampler," Los Alamos National Laboratory Calc-Note, TSA6-CN-WT-SA-TH-044, (May, 1994).
- 5. A. N. Beris, J. A. Tsamopoulos, R. C. Armstrong, and R. A. Brown, "Creeping Motion of a Sphere through a Bingham Plastic," J. Fluid Mech. 158 (1985) pp. 219-244.
- 6. J. D. Hudson, B. J. Palmer, and C. W. Stewart, "The Potential for Flammable Concentrations of Ammonia Resulting from Core Drilling Operations," Pacific Northwest National Laboratory letter report PNL-WTS-112295 (November 1995).

APPENDIX N

STRUCTURAL ISSUES ASSOCIATED WITH ROTARY MODE CORE DRILLING OPERATIONS

N.1. INTRODUCTION

In this appendix we discuss structural issues associated with the operation of rotarymode core sampling (RMCS). There are six different issues examined. Some of the issues are related to the ignition hazards, and others are breach of containment issues. They are not discussed in a particular order.

N.2. DOME LOADING

RMCS operations for single-shell tanks will impose an increase in the live weight over the tank dome. Section 2 describes all the equipment that is necessary to perform the sampling operation.

N.2.1. Static Dome Loading

In this section, we consider static and dynamic dome loading. Reference 1 indicates that dome loading must be controlled by the dome-loading limits for single-shell tanks (SSTs) as specified in OSD-T-151-00013 (Ref. 2). The RMCS System Weights Table N-1 lists the weights of the various components that possibly will be imposed simultaneously on the dome surface. This additional tank dome loading is considered to be a live load in the Westinghouse Hanford Company (WHC) evaluation of the tank structural integrity.

The static load capacity of the tank dome is monitored carefully, and an overload state that could precipitate a structural failure must be avoided. The equipment required on the surface of the tank to support RMCS sampling operations qualifies as a live load. The tank loads study permits a 50-ton concentrated live load over a 10-ft radius on the tank dome. Although all equipment listed in Table N-1 weighs more than 50 tons, not all of it is placed on top of the tank. The weight of the RMCS equipment on top of the tank is <50 tons; thus, this limit is not exceeded.

N.2.2. Dynamic Dome Loading

In case a truck were to fall on the tank dome, the dome would be subjected to dynamic dome loads. Reference 2 considers this scenario and analyzes the consequences of the dynamic loading caused by the truck dropping onto the dome. The truck weight is estimated to be 30,000 lb. The drop height for the truck was selected as 3 ft from the raised platform, and the truck is assumed to land on the minimum of 7 ft of fill.

| Component | Weight (lb) |
|--|-------------|
| Core sample truck (includes grapple hoist assembly and shielded receiver assembly) | 30,000 |
| Truck platform | 6,000 |
| Universal sampler (11 @ 10.3 lb) | 113 |
| Drill String (50 ft @ 4 lb/ft) | 200 |
| Change-out assembly | . 45 |
| Riser adapter and drill rod washer | 280 |
| Riser sleeve | 200 |
| Inlet breather filter stack | 2,000 |
| Support truck | 7,000 |
| Cask truck | 8,000 |
| Cask stand | 300 |
| Casks (5 @ 480 lb) | 2,400 |
| Mobile X-ray system | 5,000 |
| Exhauster and flammable gas detection systems | 12,200 |
| Light plants (2 @ 1000) | 2,000 |
| Video vehicle | 5,000 |
| Tent | 7,000 |
| Tent weights | 33,000 |
| People (10 total) | 2,000 |
| Total Potential Weight | 122,738 |

TABLE N-1 RMCS COMPONENT WEIGHT BREAK-DOWN

The analysis examines the worst case in which the platform disappears and the truck lands on its tires, which is conservative because this gives the smallest projected area. (The greater likelihood is that the truck would fall off the platform and land on its side, thereby affecting a greater area.)

The kinetic energy of the dropping mass is assumed to work by the resistance representing the impact. The work is estimated by integrating the load deflection diagram. (Details of calculations are available in Reference 3.) The deflection calculation considered the impact and static contributions. The dynamic load is calculated as 525,000 lb. Assuming that the load is spread over a 60° cone angle as it passes through the minimum 7 ft of soil until it reaches the dome, a projected bearing surface area on the dome is calculated as 29,860 in². The allowable load on the dome is given as 5540 lbf/ft² in Reference 4. Considering 3 in. of water vacuum

in the dome, the net allowable dome load is estimated as $4,680 \text{ lbf/ft}^2$ when the dome pressure is atmospheric.

This conservative analysis shows that the dome will withstand the impact force of the 30,000-lb truck dropping on it from the 3-ft-high platform. The analysis in Reference 3 is conservative in that it uses the smallest conceivable projected area of the dome in the calculation. If the truck were to fall on its five jacks, or on its side, the projected reacting area of the dome would be greater, and the margin of safety larger. Further, it is not likely that the truck will fall at the same time the dome is subjected to a vacuum equivalent to 3 in. of water.

N.3. BUCKLING OF THE DRILL STRING

The failure of the drill string during drilling is evaluated as a fire hazard. The downward force is limited to prevent drill bit overheating. The other limit on the down force should be based on the buckling limit. In this section, we discuss the structural buckling limit under various boundary conditions.

The drill string has an outer diameter (o.d.) of 2.25 in. and an inner diameter (i.d.) of 1.91 in., and these dimensions will be used in all of the calculations discussed in this appendix. The effects of torsion on the drill string have not been considered, and the weakening effects of the multiple threaded joints have been ignored.

Three different cases have been considered. In each case, it is assumed that the upper end of the drill string is fixed. The lower end may be free, pinned, or fixed. The formulas for allowable buckling load for long columns in these three conditions are given in References 3 and 5. Calculated buckling limits are shown in Fig. N-1 as a function of drill string length.

It is difficult to predetermine which set of end conditions should be applied as bounding boundary conditions. There are two different cases to consider; (1) the drill bit entering the crust, and (2) the drill bit penetrating the crust or waste. Reference 6 documents the case of a drill-string failure at a length of 41.7 in. under a vertical load of 2,600 lb (the drill bit was in the waste). This falls exactly on the fixedfixed curve of the above graph. However, it does not necessarily follow that waste entrance should be treated as such. It is conceivable that when a new drill first bites into the hard salt cake surface, the string will behave as fixed-fixed. It is also possible that the drill will wander along the salt cake, and behave as fixed-free.

The first core sampler has a centering spike attached to the bottom of the core sampler. The use of centering spike ensures that the drill bit does not wander along the salt cake as it tries to penetrate. Therefore, it is reasonable to assume that the boundary conditions are fixed-pined as the drill string penetrates the waste or when it is in the waste. We have establish a control requiring that the first core sampler must include the centering spike. Based on the above discussion, the downward force during crust penetration should be evaluated from Figure N-1 using fixed-pinned boundary conditions. We could not put a limit that could bound all lengths. For drill string lengths less than 45 ft, the downward force limit is quite high. However, the downward force in this range is also limited with 750 lbf established by the envelope testing to eliminate the waste ignition hazard. Thus, the downward force will be set to 750 lbf when the drillstring length is less than 45 ft. For drill string lengths higher than 45 ft, this limit should be lowered to 650 lbf because of buckling concerns.



MAXIMUM BUCKLING LIMITS



N.4. MAXIMUM BEARING FORCE CALCULATIONS OF DRILL STRING AGAINST SLEEVE

The major concern is the frictional spark when the drill string rotates in the sleeve and periodically bangs into flutes on the conductive sleeve. In this section, we determine the horizontal force that could be applied to the drill string during contact with the sleeve.

Figure N-2 is an illustration of the operation of the drill string in the conductive sleeve. The objective is to determine the absolute maximum force that the drill string could exert against the sleeve before the failure of the drill string. This maximum force can occur when the end of the sleeve is the shortest distance from the salt cake.

The following assumptions are proposed. The tube material is steel with an ultimate tensile strength of 90 ksi. The weakness caused by the threads is ignored in order to compute the highest possible bearing force. The sleeve is 15 ft long and is totally rigid. Among the tanks considered, Tank U-108 has the shortest distance from the bottom of the sleeve to the top of the salt cake. That distance is 80 in. Upon touching the waste, the drill experiences a lateral force, sufficient to fracture it at the point where it leaves the sleeve.

The bending moment required to fracture the drill string is given in Reference 5 as $M = s_u \times I/c = F \times 80$ in. From this equation, the bearing force is estimated as 610 lb



Fig. N-2. Sketch of a drill string in riser.

against the sleeve before the drill string fails. This particular case will develop the highest possible bearing force. The deflection at this force is found as 5.7 in. in Reference 3. It has been assumed that the sleeve is completely rigid. In reality, some deflection of the sleeve will occur, and the drill string may not necessarily fail where it leaves the sleeve. Furthermore, the particular tank chosen has the shortest distance between the sleeve and the salt cake. These two factors combine to give the maximum contact force of 610 lb. A deflection of 5.7 in. is used to conclude that the drill string could not damage the side walls. The contact load estimates are also used in designing the ignition testing discussed in Appendix T.

N.5. DRILL STRING DROP ACCIDENT

A potential hazard to the tank structural integrity exists if the RMCS drill string were to fall and impact the tank bottom. The drill string is restrained from falling and impacting the tank bottom by the pneumatic foot clamp. After numerous sections of the drill string have been added, the suspended weight will cause the drill string to fall if the clamp is released because the force of gravity exceeds the frictional forces. Initially, the frictional force developed at the riser seal interface exceeds the string weight. The frictional force is produced by the rubber seal that girths the outside diameter of the drill string shell. This constant force eventually is overcome by the column weight as the sections are added. The long drill string extending nominally halfway into the tank poses the largest hazard to the integrity of the tank bottom from an impact. We evaluated the impact force that would occur if the drill string were released.³

Among the tanks considered, Tank AX-103 has been chosen as a worst case because the low level of waste (42 in.) is insufficient to absorb a significant amount of energy from the falling drill string. Consequently, the effects of drag and buoyancy will be ignored.

Assumptions made in evaluating the drop accidents are as follows. The tank has a cylindrical height of 390 in. and is topped by an elliptical dome that is 144 in. high. The minimum waste level is 42 in. (Ref. 7). There are 84 in. of fill on top of the tank. The first rod section of drill weighs 6.3 lb (Ref. 8).

The methodology is based on the ballistic impact on the tank bottom. The drop distance from the ground to the tank bottom is 618 in. (Ref. 8). The impact velocity is 17.6 m/s (57.6 ft/s). This velocity is conservative because the drag force decelerating the motion is not considered. The maximum impact kinetic energy for one drill rod of 6.3 lb becomes 325 ft-lbf. Conservatively, no credit was taken for the concrete beneath the steel liner because over the years the concrete could be displaced.

If the steel bottom corroded 4.5 mils per year for 40 years (Ref. 9), the remaining thickness would be 0.195 in. (Ref. 9). From the analytical techniques of Reference 8, we found t/D = 0.086 < 0.35. The critical velocity at which 50% of the cases would

have penetration is found to be 218 ft/s. This is greater than the 57.6 ft/s calculated above; therefore, penetration is not likely for the single section of drill string.

As another check, it has been customary to convert a body that does not have a ballistic shape to an equivalent body by solving for an effective diameter. This has been considered as discussed in Reference 3 and the same conclusion is found; penetration is not likely if a single drill rod is dropped.

However, if we examine the case where the drill string is 340 in. long, the weight becomes 113 lb. The drop distance then becomes 278 in. Using the same ballistic impact analysis for higher weights, we found the critical velocity at which 50% of the cases would have penetration as 34.6 ft/s. It appears likely that the base would be penetrated if the whole drill string were to be dropped.

Another method is based on stresses induced in the steel liner from the impact as it was done in Reference 8. We equate the work done by the liner in absorbing the kinetic energy of the dropped drill string to the kinetic energy.

The impact velocity of the drill string is expressed as a function of the number of the drill rod (V = $(2gh)^{0.5} = [(2 g (618 - 19n)]^{0.5})$, where n = the number of 19-in. drill sections). The kinetic energy becomes (324n - 10n²) (Ref. 3). The kinetic energy peaks at 16 sections for a string length of 16 x 19 = 304 in.

Taking Poisson's ratio as 0.25 and the radius of the bottom plate that responds to the impact as 60 in., the equivalent spring constant for the plate is found as 6,370 lb per in. The impact force is calculated as 20,100 lb. The deflection at this load is found at 3.15 in. The developed moment can be calculated as 8164 in. lb. This results in a stress level of 1.3E6 psi. It is clear that this is more than an order of magnitude higher than the ultimate strength of the material. Thus, bottom liner penetration could occur if the drill string were dropped.

N.6. STRUCTURAL FAILURE OF THE ROTARY CORE DRILL STRING UNDER PURE TORSION

If the drill string becomes embedded in the waste because of debris in the waste, torque will continue to be applied to the drill string at a constant rate. If such a condition occurs, there is a possibility that the drill string will partially fail. Partial failure of the drill string will then result in a nitrogen flow bypass through the area that is partially failed with a leak path. Providing the necessary nitrogen flow to the drill bit protects the drill bit from overheating. Thus, waste or flammable-gas ignition is prevented. The flow rate is measured in the nitrogen instrumentation box. If a partial drill-string failure occurs because of applied torque, this condition will never be detected. Therefore, the likelihood of ignition in the waste becomes uncertain. In other words, providing nitrogen flow if there is a leak path on the drill string would not prevent the drill bit from overheating.

N-7

In this section we examine the scenario in which the drill string is considered to be torque from the upper end while the rotation of the lower end is not allowed. If the drill string fails in a short period of time under the over-torque conditions, we can conclude that the drill bit does not have enough time to overheat itself. Thus, the purpose of this section is to examine whether or not the drill string will fail in the short period of time when a torque is applied.

This issue is examined in Reference 10. The method applied and results will be briefly summarized here. An analysis in Reference 10 assumes that the drill string is jammed in the waste while torque is applied at a maximum rate of 55 rpm. Linear elastic methods are applied as a first approximation to obtain the lower bound failure estimate. Secondly, strain energy methods are used to upper bound the solution by assuming that the ultimate shear strain in the drill rod is proportional to the shear modulus. This is a very conservative approximation but will provide the upper limit of failure and will estimate the maximum time-tofailure.

Material is manufactured from ASTM A-518 carbon steel (its properties are given in Reference 10). Linear elasticity analysis estimates the shear stress as Tc/J (where T =torque, c = distance from the point of rotation to the maximum outer fiber on the cylinder under shear, and J = polar moment of inertia). The torque results in an angle of rotation for pure torsion as given by θ = TL/IG where, G = shear modulus, θ = angle of rotation, and L = length. The shear modules are calculated as 11.2E6 psi using a Poisson's ratio of 0.3. Assuming the shear stress developed under torsion is brought to the limit of the material, that is, the shear ultimate strength of the pipe, then the maximum angle of rotation can be determined. This is conservative because the theory is applicable to linear elastic bodies, and obviously this assumption does not account for the ductility of the material. The angle of rotation is calculated as 126° in Reference 10. This corresponds to about one-third of a full revolution. In terms of a time-to-failure determination, the rod will fail in 0.38 seconds for a rotational speed of 55 rpm. The time period to failure becomes 2.1 seconds for a rotational speed of 10 rpm and 4.2 seconds for 5 rpm. The lower speeds are not usually selected because sampling becomes poor. However, even in the case of low rotational speed, the drill rod will fail in a few seconds.

The second method that upper-bounds the torsional resistance is based on the strain energy method. Shear failure is assumed to occur when the materials' total strain energy equals the work done by the applied torque. Then, the strain energy can be written in terms of the shear stress and shear strain. However, in this case, the assumption is that the shear strain is the "ultimate" shear strain capacity of the material based on a linear elastic relationship. Using this method, the angular rotation to cause failure is estimated as 5208°. For a rotational speed of 50 rpm, this corresponds to a time period of 15.8 seconds. The time becomes 43.4 seconds for a rotational speed of 20 rpm. This second method is only provided to show an upper bound, and by no means will this be a reasonable failure angle or time-to-failure of the drill string under pure torsion. Because linear-elastic methods do not account for ductility or the strain hardening/softening of materials, the strain energy calculated is grossly conservative. Furthermore, the ultimate strain used in the calculations is also overly conservative because for this specific material, typically, there is virtually no ductility.

In reality, the drill string will fail somewhere between 0.5 and 2 seconds for a drill bit rotational speed of 55 rpm and between 2 to 8 seconds for a rotational speed of 20 rpm. The most important notion to consider is that this material does not have much ductility, as evidenced by the lack of strain hardening. The fact that the yield strength is only 10 ksi lower than the ultimate strength shows negligible strain hardening capability. In fact, one may consider this material somewhat brittle. Therefore, the assumption that the material attains 15% ultimate shear strain at failure is very conservative. In addition, no stress concentration factor (or effective area) is considered for threads between the drill rods. The threaded rods allow the drill string to fail in much smaller time periods.

N.7. DRILL-STRING RESONANCE

A torque and axial load will be applied to the drill string during drilling. Excitation of the drill string with a natural frequency is of concern. In this section, we evaluate the resonance rotational speeds under various boundary conditions. However, the weakening effects of the multiple threaded joints have been ignored.

Consider first the case of torsional resonance. The equation for the first mode of torsional resonance is given in Reference 3. Limiting rotational speed is plotted in Fig. N-3 as a function of the drill string length.

The maximum rotational speed is 55 rpm. From Fig. N-3, it is clear that the torsional resonance speed is extremely high. Figure N-4 plots the first- and second-mode resonance rotational speed as a function of the drill string length for the fixed-free, fixed-pinned, and fixed-fixed boundary conditions. Curves with fixed-pinned boundary conditions are suggested for the cases where the drill bit penetrates the waste and where the drill bit is in the waste. We tabulated the suggested range of speeds in Table N-2.

When the drill string is in the waste, a rotational speed of up to a maximum of 55 rpm can be used when the drill string length is less than 45 ft. Above 45 ft, the speed should be either 40 rpm. The second excitation frequency is much higher than the maximum allowable speed of 55 rpm.

N.8. DRILLING THROUGH THE BOTTOM OF THE TANK

Buckling of steel liner bottom plates in underground single-shell tanks (SSTs) and double-shell tanks (DSTs) has been observed in about three instances and documented in one specific case that we are aware of. DST SY-101 has a known buckle that extends inwardly about 4.5 to 5.0 inches, which was caused by the stress-relieving operations immediately after heat-treating the shell liner to 1100°F. Documentation of this specific case is extensive, also providing contour maps of the actual buckled shape.



Fig. N-3. Resonance speed caused by torsional load.

Other cases in SSTs that are known to exist but have not been documented pose a potential concern when installing equipment through the risers that extend to near the bottom of the tank. The inward deflection created by the buckles may prevent equipment from fully seating on the riser. Other more important concerns are with RMCS equipment, which is required to obtain samples from the complete depth of tank waste. The potential for drilling through the steel liner is evident if the steel bottom plate has a deformation from lateral buckling. Therefore, limiting the drill string height to a safe distance above the bottom plate is justified.





| Drill String Length (ft) | Force (lbf) | Rotational Speed (rpm) |
|---|----------------|---------------------------|
| L<45 | <750 | <55 |
| 45 <l< td=""><td><650</td><td><40</td></l<> | <650 | <40 |

TABLE N-2 ROTATIONAL SPEED LIMITS

This analysis evaluates nominal deformations in the bottom plate caused from effects related to tank construction practices. Because very little is known about the post-fabrication condition of some SSTs, one can only depend on the structural deformation created from the construction phase, such as the loads imposed on the tank walls after soil backfilling and shrinkage loads created during the curing period of the reinforced concrete shell.

N.8.1. Analysis

Using dimensions of Tank A-101 for the typical SST, Figure N-5 shows the approximate soil depth and tank characteristics. The tank liner is 3/8-in. thick carbon steel with a 75-ft diameter and the depth of soil to the tank bottom plate is approximately 54.5-ft. The waste temperature currently is taken as 135°F.

The lateral soil pressure is assumed as the "at-rest" soil pressure for retaining-type tures. The hydrostatic load at the base of the wall is shown^{11,12} as

$$P = \gamma H K_o$$

where,

 γ = Soil density (lb/ft³),

H = Depth of soil (ft), and

 K_o = Coefficient of soil pressure at rest.



Fig. N-5. Cross-sectional view of A-101.

The maximum soil pressure, assuming a moist, dense soil, typical of the Hanford Tank Farm region is

 $P = 6500 \text{ lb/ft}^2$ or P = 45 psi.

The lateral soil force per unit circumference on the 3/8-in. bottom plate is

 $F_{\rm s} = 17 \, {\rm lb/in.}$

Concrete shrinkage and creep is very common, and design practices allow for some shrinkage; however, structures are often slightly deformed because a small margin was allowed for deformation-type strains. For 3000 psi compressive strength concrete, typical of A-101 construction, Ferguson¹³ provides the nominal shrinkage to be approximately 2.0E-4 in./in.

Obviously, there is some appreciable gap between the concrete and the steel liner after construction before shrinkage begins. This gap cannot be fully quantified. Thus, nominally, a 1/4 in. gap is assumed to exist between the steel and concrete immediately after construction.

The bottom plate radial deformation, based on a concrete shrinkage coefficient of 2.0E-4 in./in. is

$$\Delta R = \alpha R$$
,

where,

 $\alpha = 2.0\text{E-4 in/in.},$

R = 450 in., and

 $\Delta R = 0.09$ in.

Thermal expansion deformations of the tank bottom, caused by heat loads from the waste, will comprise about

 $\Delta R_{th} = \alpha_{stl} R \Delta T \; ,$

where,

 $\alpha_{stl} = 6.5$ E-6 in/in/°F,

and the temperature difference is from 75°F to the current waste temperature of 135°F, or $\Delta T = 65$ °F. The 75°F temperature is assumed to exist at the time of construction with no appreciable deformations. The maximum thermal expansion of the tank bottom is

$$\Delta R_{th} = 0.175$$
-in.

Accounting for the 1/4-in. initial gap between the steel and concrete, the maximum lateral deflection of the plate after concrete shrinkage and thermal expansion, would be 0.015 inches. The difference between the radial gap and the concrete shrinkage plus thermal expansion is an interference fit, that is, the radial contraction imposed on the tank plate. Thus, there exists a radial force per unit of circumference such that the proportionality of the following relationship must hold true in the elastic region

$$\frac{\Delta R}{\alpha R} = \frac{\sigma}{\alpha E} \,,$$

where, $\sigma =$ Stress, (psi)

E = Modulus of elasticity, (psi)

or,
$$\Delta R = \frac{\sigma R}{E}$$

The applied stress from the shrinkage and thermal effects is

$$\sigma = 1000$$
-psi,

or a radial force per unit circumference of

$$F_{cs} = 375 \text{-lb/in.}$$

Comparing this value to the critical buckling load of the 450-in. diameter plate assuming a clamped condition $is^{14,15}$

$$\sigma_c = 1.22 \frac{E}{1 - v^2} \left(\frac{t}{a}\right)^2$$
, and

the critical stress is

$$\sigma_c = 28$$
-psi or a radial load of $F_{cs} = 10.5$ -lb/in.

The compressive load from the concrete shrinkage and thermal expansion is much greater than the critical buckling load of the plate. Given these large loads, the plate would have undergone local plastic deformation and therefore would never achieve such high radial forces. Obviously, the actual gaps and conditions are unknown, and any other assumption as to the actual state of construction is an estimate.

Therefore, in order to achieve a physical meaning, a conservatively critical buckling load representative of the circular plate plus additional load from the lateral soil pressure is used in calculating the maximum deflection of the buckled form.

This rationale assumes that soil pressure will always be present on the tank walls, and that thermal and concrete shrinkage loads will cause local plasticity, thus allowing for relieving of strains. Thus, the maximum load is taken as

$$F_t = 27$$
-lb/in.

The out-of-plane deformation of the bottom plate for the compressive load shown above is approximated using a differential strip element of plate along its diameter. This is a conservative assumption because it assumes a rectangular strip of length L and not a radial segment of a circular plate. In order to obtain a buckled shape deformation with a compressive radial load, a very small out-of-plane perturbation load is applied to initiate buckling, where the perturbation load is Q. Let Q equal 1% of the compressive load, and the deformation is found as¹⁴

$$w = \frac{Q}{2Pk} \left(\tan \frac{kL}{2} - \frac{kL}{2} \right),$$

where,

w = Deflection (in.),

P = Lateral compressive force (lb),

$$k=\sqrt{\frac{P}{EI}}\ ,$$

L = Elemental strip length (diameter), and

I = Moment of inertia (in.⁴)

The solution to the maximum deflection is

$$w = 2.2$$
 in.

It should be understood that this calculation only provides an "estimate" of the buckling deformation. It presents a simplistic argument for a complex condition and makes assumptions on the state of construction. There are numerous unknown conditions that may exist, which if conservative assumptions are used, would yield unreasonable results of the buckling deformation. Furthermore, because actual observations of DST buckling are documented (e.g., SY-101) with maximum heights of about 4 to 5 inches, this calculation provides a basis for the order of magnitude.

N.8.2. Conclusions

A 3-in. minimum distance is recommended for limiting the lowering of the drill string to the tank bottom. This limit is established from the best-estimate analyses provided above. One may estimate larger deformations using more conservative assumptions that those used in this analysis. Obviously, the 3-in. limit does not guarantee that the drill bit will never contact the tank bottom. However, it reduces the probability of drilling through the bottom without sacrificing valuable sample data near the bottom of the tanks. As mentioned before, one documented case of bottom buckling is for 101-SY. For that case, the peak deflection is measured as ~4 in., which covers less than a few percent of the total surface area.

N.9. REFERENCES

- 1. N. Milliken, "Safety Analysis for the Push Mode and Rotary Mode Core Sampling," Westinghouse Hanford Company report WHC-SD-WM-SARR-031, Rev. 1 (June, 1995).
- 2. V. C. Boyles, "Operating Specifications for Single-Shell Waste Storage Tanks," Westinghouse Hanford Company report OSD-T-151-00013, Rev. D-4 (1994).
- 3. B. Weinstein, "Structural Analysis of Rotary-Mode Core Sampling Operations," Los Alamos National Laboratory Calc-Note TSA10-CN-WT-SA-ST-086 (December, 1995).
- 4. K. J. Lawrance, "Dome Analysis," Los Alamos National Laboratory Calc-Note N6-CN-WT-SA-ST-024 (August, 1992).
- 5. R. J. Roark and W. C. Young, Formulas for Stress and Strain, 3rd edition, (McGraw Hill Co., New York, New York, 1975).
- 6. H. H. Ziada, "Axial Failure Food Estimates for Push Mode Drill String," Westinghouse Hanford Company memorandum H5-52 (October 18, 1995).
- K. O. Pasamehmetoglu, "Single-Shell Tanks Dome Volume Calculations," Los Alamos National Laboratory Calc-Note TSA10-CN-WT-SA-TH-086 (December 1995).
- 8. W. O. Miller, "Retained-Gas Sampler Drill-String Drop Analysis," Los Alamos National Laboratory Calc-Note TSA6-CN-WT-SA-075 (November 30, 1994).
- 9. J. P. Strehlow, "Drill-Bit Punch Through Single-Shell Tank Liner," Westinghouse Hanford Company memorandum CSA: JPS:ggb:92/32 (1992).
- E. Rodriguez, "Structural Failure of the Rotary-Core Drill String Under Pure Torsion," Los Alamos National Laboratory Calc-Note TSA10-CN-WT-SA-ST-085 (December 1995).
- 11. G. B. Sowers and G. F. Sowers, Introduction to Soil Mechanics, 3rd Edition (Macmillan Publishing Co., New York, New York, 1970).
- 12. J. E. Bowels, Foundations Analysis and Design, 3rd Edition (McGraw-Hill Book Co., New York, New York, 1982).
- P. M. Ferguson, Reinforced Concrete Fundamentals, 3rd Edition (John Wiley & Sons, Inc., New York, New York, 1973).

N-16

August 8, 1996

- 14. S. P. Timoshenko and J. M. Gere, Theory of Elastic Stability, 2nd Edition (McGraw-Hill Book Co., New York, New York, 1961).
- 15. R. J. Roark and W. C. Young, Formulas for Stress and Strain, 5th Edition (McGraw-Hill Book Co., New York, New York, 1975).

August 8, 1996

Ē



APPENDIX O

ENERGY AND WATER ADDITIONS/REMOVALS

0.1. INTRODUCTION

In this appendix, the issues associated with energy and water addition to the tank during sampling are addressed.

O.2. WATER ADDITIONS

These accidents address the possibility of excessive water additions to the tank that would cause the level to rise above that allowed by safety controls. In the limit, flooding could result from the release of tank materials into the environment caused by hydrostatic failure of the tank. This appendix also addresses the effect of the water addition on the waste temperature and gas releases.

The maximum quantity of water available for decontamination during operation and removal is 5.7 m^3 (1500 gal.), which is physically controlled by the size of the water supply. Adding this quantity of water increases the tank level by a maximum of 1.4 cm (0.55 in.). Per activity, water additions are limited to 0.95 m³ (250 gal.). Flooding of the tank by adding a maximum 1500 gal. of water is impossible. Numerous 1500-gal. tanks of water would have to be added in violation of the 1500gal. total addition limit. The maximum quantity of water available for decontamination during operation and removal is 250 gal., which physically is controlled by the size of the water supply. The water addition to an assumed leaky single-shell tank (SST) must be reduced to 100 gal.

One of the reasons for keeping the water addition to minimum amounts is the danger of interim-stabilized tanks jeopardizing the interim-stabilized status. A tank must contain less than 50,000 gal. of drainable interstitial liquid and less than 5000 gal. of supernatant liquid in order to maintain its interim-stabilized status.

These numbers are predicated on the flow rate, duration of operation of the decontamination system, and maximum number of anticipated decontamination operations. Any addition of water beyond this total level must be approved by the test review group (TRG) with consideration for overall effects on the tank waste. To ensure that the water is sufficiently provided to the check valve for the purpose of cleaning, the water level in the water supply tank will be monitored.

Water addition has several more effects: (a) changes the waste temperature, (b) releases gases, (c) changes the waste pH level, and (d) increases the activity of radiolysis.

Farley¹ investigated the water addition effect. Roughly, unit change in waste pH would require an order-of-magnitude change in volume to sufficiently affect the hydronium ion concentration. Assuming water will affect a 1-ft radius, the minimum waste volume can be found as 82 gal. that are calculated for Tank AX-103 that has a minimum waste level of 42 in. A unit change in pH for this tank would require 820 gal. of water. The water addition is limited to 250 or 100 gal. per activity. The small amount of water added to the tank will dissolve salts and will be saturated. The pH change is not expected. The water-addition limit is 100 gal. for assumed leaky tanks to minimize the release to the environment.

Hydrogen is generated by the radiolysis of water. The water addition would reduce radioisotope concentration. However, the volume would be increased by water addition. Therefore, hydrogen generation per unit volume would not be affected with water addition. Small amount of water addition in comparison to the total volume would imply insignificant effects. Generation of other gases would be affected in the same way. Water addition, however, may cause gas release, reducing the gas retention capability. During rotary-mode core sampling operations, water will be added at the waste surface, and a gas release triggered by water addition is not expected.

0.3. STEAM GENERATION CAUSED BY ENERGY ADDITION

In Reference 2, a steam-release accident is postulated for high-heat tanks. High heat was considered to generate and retain steam in the waste. However, an analysis specific to high-heat Tank C-106 demonstrated that the condition of this tank is not sufficient to initiate a steam-release event. This was believed because there is sufficient heat transport to cool or condense the steam bubbles, and the sludge-shear strength was not enough to retain steam bubbles in the saturated region. The steam generation caused by heat generation is not a concern in the single-shell flammable-gas tanks having waste temperatures less than 93°C, and the waste level does not significantly change to indicate a great deal of accumulation.

In this safety assessment (SA), the concern is the generation of steam caused by drill bit frictional heating that could remove moisture from the tank. The energy addition to the tanks is discussed above for two limiting cases: frictional heating with no penetration and efficient cutting with penetration.

The bounding energy input caused by drilling is estimated as 8.66 x 10^6 J for frictional heating and 5.6 x 10^7 J for smooth penetrating drilling operations. Reference 3 gives the amount of water that can be evaporated for the tanks that are on the Flammable Gas Watch List. The maximum amount of water is estimated as 27.3 kg. These results are very conservative because all the drilling energy is assumed to result in evaporation. Nonetheless, the amount of evaporation is negligible and will not change the moisture content and waste level significantly in any of tanks listed in the table. The vapor volume is 22.84 m³ (807 ft³). Considering

that the minimum dome volume is 1416 m^3 (50,000 ft³) and that the steam will be generated over hours of operation, the temperature increase in the dome will not be significant.

0.4. WATER REMOVAL CAUSED BY EXHAUSTER

Another way to remove the moisture in the tank is by operating the exhauster. Among the original Flammable Gas Watch List tanks, the maximum tank dome temperature is found to be $37^{\circ}C$ (98°F) in Tank 101-A. This temperature is used to calculate a conservative moisture removal. The exhauster has a relative humidity requirement of 70%. Above this value, normal exhauster operation is not allowed unless there is a gas-release event.

Reference 3 calculates the time of operation necessary to observe a noticeable wastelevel decrease, 0.5 in., as ~15 days if the relative humidity in the dome were 70%. These results are very conservative because they do not account for the water vapor partial pressure suppression in highly concentrated salt solutions. In reality, the waste (especially if the surface is dry) is likely to act like a desiccant and absorbs moisture from the dome. Nevertheless, the rotary-mode core sampling (RMCS) operation is assumed to involve roughly about 10 days per year per tank in this SA (App. E). This period considers two full samplings each, including 10 to 11 samples. Note that each full sampling may occur at different times. Thus, during RMCS, water removal as a result of continuous ventilation is not a safety concern.

O.5. REFERENCES

- 1. W. G. Farley, "Safety Assessment for Thermocouple Tree System Installation and Operation in Ferrocyanide Waste Tanks," Westinghouse Hanford Company report WHC-SD-WM-SAD-014, Rev.-3 (November 1993).
- 2. N. Milliken, "Safety Analysis for Push-Mode and Rotary Mode Core Sampling," Westinghouse Hanford Company report WHC-SD-WM-SARR-031 (July 1995).
- 3. C. Unal, "Calculation of Water Removal From Single Shell Tanks Due to Exhauster Operation," Los Alamos National Laboratory Calc-Note, TSA10-CN-WT-SA-TH-092 (December 1995).

APPENDIX P

ANALYSIS OF FRICTIONAL HEATING IN ROTARY-MODE CORE SAMPLING

P.1. INTRODUCTION

It is possible that the rotating drill string may contact the conducting sleeve during drilling operations. Because this contact involves potentially high horizontal loads and dry metal-to-metal sliding, an analysis of the temperature rise caused by frictional heating was performed. In addition, a scoping analysis of frictional heating in the event of exhauster-fan-to-case contact was performed, as discussed in Section P-8 of this appendix. The analyses presented in this appendix should be viewed in light of statements in the literature such as

"The actual surface temperatures achieved during dry friction are virtually impossible to measure, and both the temperatures themselves and the methods used to obtain them are matters of continuous controversy."¹¹

Rabinowicz² notes that

"...In consequence of the difficulties involved in this situation, hardly anyone ever tries to calculate the temperature rise produced in sliding. Instead, if the author's experience is typical, reliance is generally placed on one or another of two widely established but mutually exclusive maxims:

- 1. The flash temperature at a sliding surface is usually only a little greater than the average temperature of the contacting surfaces.
- 2. The flash temperature is usually the melting temperature of the lower melting of the two sliding materials."

Because reliance on either of the statements above provides a poor basis for a safety case, an attempt was made to develop a defensible calculation.

The concept of "flash temperature" mentioned by Rabinowicz is important to understanding the phenomenon of frictional heating and the analyses presented in this appendix. In sliding situations, contact is not made over the whole of the apparent contact area, A_a , but over a few isolated junctions (which together form the real contact area, A_r). Because these junctions are small and they receive substantial thermal energy, their temperatures may be much higher than that of neighboring surface regions (that is, the local macroscopic surface temperature). The flash temperature is defined as the hot-spot temperature associated with these junctions. During sliding, junctions continue to be made and broken, and the "hot spots" on the surfaces shift their location throughout the apparent contact area. There seems to be general agreement, however, that the flash temperature reached at any of these hot spots tends to be reasonably constant under constant sliding conditions. Figure P-1 provides a schematic view of an interface, showing the apparent and real areas of contact.

Real Contact Area is Projected Area of Point Contacts



Fig. P-1. Schematic view of a sliding interface between real materials, showing the apparent and real areas of contact.

For a given dry-sliding problem, estimation of peak temperatures is, therefore, the combined problem of calculating the nominal surface temperature distribution resulting from frictional heating over the entire apparent contact area, and the "flash temperature rise," which are then added together. The first difficulties arise in determining appropriate areas and in estimating the flash temperature rise.

The real contact area depends on the characteristics of the surfaces in contact, namely the size and shape of the apparent area of contact, the surface roughness of the two materials, and the way they are placed together. Fortunately, it is possible to estimate the real contact area based on simple limit analyses assuming ideally plastic deformation. If the surfaces that are placed in contact are rough (not prepared bearing surfaces), the contact area of a localized region of material under an imposed load will depend on the stress that such a region can carry without plastic yielding. This property is known as the penetration hardness, and is about three times the yield stress. This argument is plausible based on the similarity between the load carried by a surface high point (called an asperity in the friction literature) and the load imposed by the indenter used in performing the surface hardness test. It also yields results consistent with experimental measurements, such as electrical contact resistance measurements with precious metals. The simple limiting values obtained by this approach are often modified to account for shear forces (always present in sliding) and surface energy considerations.

The apparent contact area consists of the real contact area of the junctions and the area of those regions that appears as if contact might have been made there (but was not). Again, this can be very difficult to estimate for other than simple (for example, planar) geometry. It can be conservatively estimated from the real contact area estimate, based on the generalization in the literature that "the real contact area is *no more than* 10% of the apparent contact area" (emphasis added).

When surfaces slide together, almost all of the energy dissipated in friction appears in the form of heat at (or near) the interface. Historically, it was a friction experiment that proved the equivalence of mechanical energy and heat. As implied above, some of this energy can be deposited just below the surface because the plastic and elastic deformations occur throughout a volume. The material properties and characteristics of the surfaces can change as a result of this surface "working," changing the subsequent frictional characteristics, including heating. Several formulas for use in estimating the *expected* flash temperature rise have been developed based on simplified assumptions. The expected flash temperature rise results from distributed contact over many contact points. The limiting case of a single asperity with an area equal to A_r yields the *worst-case* flash temperature rise.

P.2. DRILL-STRING-TO-SLEEVE FRICTIONAL HEATING

As described in Section 4, the drill string moves within a conducting sleeve placed in the tank penetration. Information provided indicates that the drill string is 1100series carbon steel and the sleeve is stainless steel. The sleeve is a Schedule 80S 3-in. nominal pipe,³ which has an i.d. of 2.90 in. with a wall thickness of 0.3 in. and is nominally 15 ft long. The drill string has a 2.25 in. o.d. and a 1.906 in. i.d. The drill sections themselves are fluted on the o.d. and are nickel plated. It has been reported that the portion of the string not penetrating the waste is not fluted, and it is not assumed to be plated. A scale drawing of the drill-string-in-sleeve cross section is shown in Fig. P-2. Coolant nitrogen for the drill bit is fed through the drill string at a nominal rate of 30 scfm. A nitrogen purge flow is introduced into the annulus between the sleeve i.d. and the drill-string o.d. at a nominal rate of 5 scfm.

Contact between the drill string and sleeve is possible. An analysis provided in Appendix N determined that a horizontal load of up to 610 lbf could be applied before the drill string would break at the point where the string exits the sleeve. The figure in Appendix N illustrating this situation is greatly exaggerated, however. Because the nominal clearance between the drill string and sleeve is small compared to the 15 ft sleeve length, the contact angle is quite shallow (see Fig. P-3). The bounding frictional heating analysis presented here assumes the limiting load of 610 lbf is applied horizontally (incipient drill-string failure). It further assumes a drill rotational speed of 55 rpm and initial temperatures of 38°C. This analysis conservatively assumes contact at only one region (sleeve exit) and that the contact region remains stationary on the sleeve. It would be expected, in reality, that bending would cause the drill string to rotate eccentrically, sweeping the contact region around the sleeve periphery. Bending might distribute real horizontal loads over two regions of contact. Also, a minimum penetration rate must be maintained for continued drilling, which would "feed" colder drill-string material into the contact region.



Fig. P-2. Cross-sectional view of the drill string in the conducting sleeve (to scale).

P.3. ESTIMATING CONTACT AREAS

Rabinowicz² suggests estimating the real contact area from the following relationship based on a limiting plastic-deformation analysis,

$$A_r = \frac{L}{p}, \qquad (P-1)$$

where L is the applied load acting normally and p is the penetration hardness. Because sliding applies shear forces at the surface, the limiting value obtained by Eq. (P-1) is usually modified to account for shear as



Fig. P-3. Vertical section of the drill string in the conducting sleeve (to scale).

$$A_{\rm r} = \frac{L}{p} \sqrt{1 + \alpha f^2} , \qquad (P-2)$$

where f is the coefficient of friction and α is an empirical constant.⁴ When applied to the problem conditions described in Section P-2 above, the real contact area A_r is estimated to be 13.1 mm² (see Reference 5 for calculation details).

The apparent contact area is needed to specify the area over which frictional heating occurs to enable calculation of the nominal surface temperature distribution.

August 8, 1996
Because the problem geometry involves both curved and straight surfaces (see Figs. P-2 and P-3), determination of the apparent area is not straightforward, and a conservative approximation of ten times the real contact area is used (131 mm^2) . The contact area in the problem considered here is probably elliptical.⁶ To convert the area above to an ellipse, the major axis length is assumed to be four times the minor axis height. The resulting elliptical area has a major axis length of 12.9 m m and a minor axis height of 3.2 mm.

P.4. FLASH TEMPERATURE RISE

Rabinowicz² summarizes the state of the art in estimating the flash temperature rise, and offers the following choices:

$$\theta = \frac{v}{2} \pm (a \text{ factor of 3}), \tag{P-3}$$

described as "an order of magnitude realistic" method, where θ is the flash temperature rise (°C) and v is the sliding velocity in cm/s. For our problem, this method yields an estimate of 8°C, with a range of 0°C to 33°C.

For the simplification of one circular junction of diameter $2r_j$ sliding on another material at moderate speed v, the interface attains an equilibrium mean temperature rise θ_m above the nominal surface temperature of

$$\theta_{\rm m} = \frac{f\Delta Lv}{4r_{\rm j}(k_1 + k_2)}, \qquad (P-4)$$

where r_j is the junction radius taken from the literature for dry steel on steel, and L is the applied load per junction, where the number of junctions is obtained by dividing the area of one junction of radius r_j into the real contact area A_r . The thermal conductivities of the two materials in contact are denoted k_1 and k_2 . For this problem, θ_m is estimated by Eq. (P-4) to be 3°C.

Finally, a method is presented based on surface energy considerations, yielding "an order of magnitude" estimate for θ_m of

$$\theta_{\rm m} = \frac{7800 f W_{\rm ab} v}{(k_1 + k_2)},$$
(P-5)

where W_{ab} is the surface energy of adhesion. For our problem, this method yields the most conservative estimate of 46°C. This value was added to the maximum

value calculated for the nominal surface temperature to estimate the peak surface temperature.

P.5. PEAK SURFACE TEMPERATURE CALCULATION

P.5.1. Nominal Surface Temperature Scoping Calculation

A scoping calculation was performed earlier, based on two semi-infinite bodies subjected to a sudden heat flux at the surface. This calculation used a different set of assumptions (fluted drill geometry in contact with the sleeve and a lower horizontal loading), but the results are qualitatively interesting. The calculation, described in Reference 5, concluded that a surface temperature rise on the order of several hundred degrees centigrade would require approximately 200 s. A corresponding steady-state interface temperature was calculated, with some general assumptions about convective heat transfer to the nitrogen drill-bit cooling and sleeve purge flow, at about 350°C.

P.5.2. Detailed Numerical Calculation of Nominal Surface Temperature

A detailed numerical model, with a two-dimensional drill string in contact with a three-dimensional sleeve was developed and applied. The string was modeled in two dimensions because the rotation smears any instantaneous azimuthal differences. The sleeve was modeled in three dimensions to account for the considerable heat capacity of the relatively thick wall and the extended area for convective heat transfer. Axial meshing was the same for both the string and the sleeve in the upper part of the model (through the contact region), and the drill string model extended below the sleeve. Approximately 1.45 m of overall drill string length was modeled, with the sleeve model just over half that length. The model axial ends were assumed adiabatic. Although the outer surface of the sleeve would be cooled by natural convection and could radiate heat to the cold tank interior, the outer surface of the sleeve and the outer surface of the drill string below the sleeve were conservatively considered to be adiabatic.

The finite-difference meshing can be varied, depending on the problem definition and expected thermal gradients. The axial meshing used to date is quite fine in the region of contact, being about one-fourth the thermal diffusion length for one revolution of the string. The axial nodes get courser as they get farther from the contact region. Both the drill string and the sleeve had three radial nodes for the calculations discussed here, with the outer drill-string-node thickness and the inner sleeve-node thickness kept equal. The azimuthal noding of the sleeve was relatively fine, as can be seen in Fig. P-4. The resulting model had 903 separate nodes. Frictional heating was applied to the contact region based on the elliptical contact area described earlier. Because frictional heating occurs throughout the volume "worked" in sliding contact and because the material properties of the sleeve and drill string are similar, half of the heat was applied to each component. The contact region was coupled with a conduction model with contact resistance,

allowing heat to flow across the interface in either direction as dictated by the instantaneous temperature distribution. Convective cooling (or heating farther down the string interior) was included in both the drill string and the sleeve-to-drill annulus. The annulus convection model included the effect of ten different temperature surfaces (nine azimuthal sleeve nodes and the drill string outer surface).

The model was developed and solved within the architecture of the Los Alamos Systems Analysis (LASAN) code package.⁷ LASAN materials properties and convective cooling routines were applied directly. The resulting model had 903 nonlinear ordinary differential equations that were solved simultaneously with a fully implicit integration algorithm. Temperature-dependent material properties were used throughout. The maximum surface temperature and its location were saved at each time step along with user-defined list variables and plot variables. Details of the model can be found in Reference 5.

P.5.3. Peak Surface Temperature Estimates

Figure P-5 presents the estimate for the peak surface temperature as a function of time after initial contact. The lower curve is the maximum temperature calculated with the numerical model described above. The conservative flash temperature rise from Eq. P-5 is added to yield the estimated peak surface temperature as a function of time. This is the peak surface temperature calculated with very conservative assumptions, which should bound the "expected" temperatures. The conservative estimate of the maximum possible "worst-case" surface temperature can be found in the following section.

P.6.WORST-CASE HOT-SPOT TEMPERATURE, AREA, AND LIFETIME

The literature presents an equation for the "worst-case temperature rise." This equation results from applying Eq. (P-4) with a single asperity of area A_r .

Application to our problem results in an estimated worst-case temperature <u>rise</u> above the nominal surface temperature of 577°C. The area of the hot spot is the real contact area, A_r , or 13.1 mm². This small volume would be expected to cool rapidly to the nominal surface temperature. Because the "single asperity" would be plastically deformed in sliding while carrying the *entire* applied load, it is reasonable to assume that this condition could only occur early in the dry sliding contact of two surfaces.

To determine the lifetime, or time-temperature history, of this hot spot, a numerical calculation was performed. It was assumed that the spot was formed on the first revolution in contact, so the nominal surface temperature was the initial temperature (assumed to be 38° C). When this 13.1 mm^2 spot is swept out from

under the contact area, heating stops and cooling begins. Because conduction is expected to dominate, convective and radiative heat removal were ignored.



Fig. P-4. Azimuthal noding of the sleeve. Nine azimuthal nodes are actually calculated (symmetry gives the effect of 17 nodes with an adiabatic boundary opposite the contact region).

The average surface roughness (from ANSI B46.1-1978) for cold rolled or drawn materials ranges from about 1-10 μ m.¹ The asperity height chosen for this analysis (100 μ m) ensures that the contact point is well above the nominal surface. The analysis is quite sensitive to the assumed height of the contacting asperity, because the volume of the hot spot and the conduction length to the cooler surface both depend directly on the height. Using temperature-dependent material properties and the LASAN solver described above, the profile shown in Fig. P-6 was calculated. It can be seen that the hot spot drops below 400°C within 0.6 ms.



Fig. P-5 Estimated peak surface temperature as a function of time after initial contact. The lower curve is calculated with the numerical mode: described. The upper curve results when the conservative "flash temperature rise" is added to the numerical results.

P.7. CONCLUSIONS FOR DRILL-STRING-TO-SLEEVE CONTACT

The analyses above show that frictional ignition from a drill-string-to-sleeve contact event is unlikely. Even for a very conservative set of boundary conditions and loading assumptions (which probably could not be physically sustained for a long period), it takes on the order of 2100 seconds to reach surface temperatures of 400°C (500°C is not reached in 1 hour of contact). Although the possibility of a worst-case hot spot with a rapid, higher temperature rise cannot be precluded, analysis has shown the lifetime above 400°C to be very short, and the area is very small. With the nitrogen purge between the sleeve and the drill string, the presence of a combustible mixture in the vicinity of the hot spot is not likely. Consideration of the minimum surface temperature and the necessary area and induction time for ignition indicate that ignition is not expected even if a combustible mixture happens to be present.



Fig. P-6. Transient cooling curve for a "worst-case" hot spot. It can be seen that the very small (13.1 mm²) hot spot cools to below 400°C in less than 0.6 ms when exposed from under the contact area.

An experimental and theoretical study of the ignition of hydrogen/oxygen/nitrogen mixtures by hot surfaces, done for Canadian (CANDU) reactor licensing, provides methods for estimating both the minimum ignition surface temperature (MIST) and the induction time for ignition by surfaces with temperatures above the MIST.⁸ For "20% and 30% hydrogen in air," the MIST was determined to be 916.2 K (643°C), which is higher than any of the temperatures calculated above. For temperatures below the MIST, the induction time is infinite (that is, ignition is not expected to occur, and did not occur in the experimental program).

However, as noted in the introduction to this appendix, there is considerable controversy about the ability to estimate the temperatures arising from dry metal-tometal sliding. In order to conclude that ignition caused by drill string frictional contact with the sleeve is not likely with a high degree of confidence, it is necessary to obtain experimental verification by performing prototypical ignition testing. As test results are factored in, the predictive capability of the analysis will improve and the associated uncertainty will be reduced.

P.8. FRICTIONAL HEATING IN FAN-TO-CASE CONTACT

P.8.1. Scoping Calculation

A question has been raised about the possibility of frictional ignition by the exhauster fan during rotary-mode core sampling operations. The fan and its case are both made of aluminum. It is possible, using surface energy methods, to estimate the flash temperature rise above the nominal surface temperature resulting from dry metal-to-metal sliding as a function of sliding velocity.

Rabinowicz² develops an equation for the flash temperature rise based on surface energy considerations, presented earlier as Eq. (P-5). It can be noted that all parameters in this expression are material constants, with the exception of the sliding velocity. It is, therefore, particularly attractive for scoping estimates where a specific "event sequence" has not been postulated.

For aluminum, Table 2.1 of Reference 2 gives a value for the surface energy, γ , of 900 mJ/m² (0.9 J/m²). The surface energy of adhesion for two materials in contact, W_{ab}, is defined as²

$$W_{ab} = c_m \left(\gamma_a + \gamma_b \right), \qquad (P-6)$$

where the compatibility parameter, c_m , is 1.0 for identical metals, yielding a value for W_{ab} of 1.8 J/m². The coefficient of friction, f, for dry sliding of aluminum on aluminum is quite high. Marks' Handbook³ gives a value of 1.4, confirmed by Figure 4.16 in Reference 2. The thermal conductivity for "generic" aluminum⁹ is about 200 W/m °C.

Using the values listed above, Eq. (P-5) yields a unit-velocity flash temperature rise of about 49°C per m/s of sliding velocity. This value is only one-sixth of the comparable value for the drill-string-to-sleeve (carbon-steel/stainless-steel) problem above of 288°C per m/s of sliding velocity, but the potential sliding speeds in fan-tocase contact are much higher. Assuming an initial temperature of 38°C, any sliding velocity above about 12.7 m/s will produce melting using this simple model. (The drill-string-to-sleeve contact has a surface sliding velocity of only 0.16 m/s).

The exhauster ventilation fan has a 0.381-m (15-in.) o.d. and a maximum rotational speed of about 2400 rpm (estimated at the 60% power limit, 4.5 hp point — typical operating point is 28% power).¹⁰ These fan characteristics give a tip speed of almost 48 m/s. Together with the unit-velocity flash temperature rise calculated above, this sliding speed gives a temperature well above the melting point of aluminum (conservatively 660°C, reported values for different material specifications range

from 643°C to 657°C). In such cases, the temperature rise is limited to the melt temperature by the phenomenon of melt lubrication, which greatly reduces the coefficient of friction. This reduction in friction reduces the subsequent heating rate and limits the total quantity of material melted.

P.8.2. Discussion of Exhauster Fan-to-Case Frictional Heating

The first thing to remember is that the above estimate is an order of magnitude estimate for the mean flash temperature rise in equilibrium. Here equilibrium presumably means contact and constant sliding speed and because heat capacity does not appear in Eq. (P-5), thermal equilibrium of the "junctions" is expected. However, the junctions involved in the flash temperature *rise* (above the nominal surface) are *very* small and will reach thermal equilibrium quickly (assumed instantaneously in the friction and wear literature). But, a short duration "impact" could be different. Also, Eq. (P-5) gave the highest temperatures, by far, of three methods used in the drill-string-to-sleeve problem.

A second, and perhaps more important, consideration is what conditions are required for ignition of a hydrogen/air mixture by a hot surface. As mentioned in Section P-7 above, an experimental and theoretical study of ignition of hydrogen/oxygen/nitrogen mixtures by hot surfaces, done for Canadian reactor licensing, provides methods for estimating both the minimum ignition surface temperature and the induction time for ignition by surfaces with temperatures above the MIST.⁸ For "20% and 30% hydrogen in air," the MIST was determined to be 916.2 K (643°C), or very close to the melt temperature of aluminum. For temperatures below the MIST, the induction time is infinite (that is, ignition is not expected to occur, and did not occur in the experimental program).

The relationship between the surface temperature, T_s , and induction time, t*, was given as

$$T_{\rm S} = T_{\rm S\infty} e^{A/t^*} , \qquad (P-7)$$

where $T_{s\infty}$ is the MIST and A is an empirical characteristic time constant. For "20% and 30% hydrogen in air," A was reported to be 5.08e-5 s.

Rearranging Eq. (P-7) to solve for t^* given T_s ,

$$t^* = \frac{A}{\ln\left(\frac{T_s}{T_{s\infty}}\right)}.$$
 (P-8)

For T_s at the melting point of aluminum, the induction time t^{*} calculated with Eq. (P-8) is 2.77 ms.

August 8, 1996

10 september 2

To determine the hot-spot lifetime above the MIST, a simple 1D transient heat transfer calculation was performed. The hot-spot thickness was taken to be $20 \,\mu$ m, as measured in experimental studies of methane ignition by a molten metal smear produced by worn coal-mining bits.¹¹ In a one-dimensional calculation, the area drops out if the calculation is based on the enthalpy per unit area (the time derivative is the heat <u>flux</u>). Because conduction was expected to dominate, other heat-transfer mechanisms such as convection to the ventilation flow and radiation to the ambient-temperature environment were ignored. The system was assumed to be initially at 38°C. The thermal conductivity and heat capacity of the melt were assumed to be equal to that of solid aluminum. The simple model was solved numerically with the LASAN code package.⁷

Figure P-7 displays the transient cooling of the melt smear to the ambient temperature surface. The plateau represents the time required to remove the heat of fusion from the melt. It can be readily seen that the hot smear is very short lived, with the time above the MIST temperature less than 4 μ s and the time above 400°C less than 7 μ s. This transient cooling is quite sensitive to the assumed melt-smear thickness, because both the volume of hot material and conduction length to the cooler surface depend directly on the assumed thickness. To assess the sensitivity, the problem was rerun with the assumed melt-smear thickness increased by a factor of 10 to 200 μ m. Results for this parametric sensitivity calculation are shown in Fig. P-8. While the lifetimes are much longer than the base case, they are still short compared to the calculated induction time of 2.77 ms. Examination of Fig. P-8 shows the time above the MIST temperature is less than 0.4 ms, and the time above 400°C is less than 0.65 ms. At a sliding velocity of 48 m/s, this 0.65 ms lifetime would produce a smear above 400°C about 3 cm long (width depends on geometry of contact).

P.8.3. Conclusions for Exhauster Fan-to-Case Frictional Heating

Based on the simple analysis above, it appears unlikely that ignition would result from short-term contact between the exhauster fan and case, even though quite high local temperatures can be generated. The question of surface temperatures resulting from long-term rubbing is more difficult and would require detailed modeling of the fan and case to address. However, it requires a mechanical failure for fan-to-case contact to occur; therefore, sustained powered fan operation with contact is unlikely. If sustained rubbing is determined to be possible, the high thermal conductivity of aluminum provides some time to sense contact and interrupt fan power. Because the hot spots are quite localized, but at high temperatures, an optical temperature measurement (for example infrared) might be possible (depending on case geometry and "view factors"). This analysis is further documented in Reference 12.



Fig. P-7. Transient cooling curve for a 20-µm-thick hot-melt smear as a function of time after the melt is swept out of the contact zone. At that point, heating stops and conduction to the cool surface begins. The plateau represents the time required to remove the heat of fusion from the melt.



Fig. P-8. Transient cooling curve for a 200-µm-thick (10 x expected) hot-melt smear as a function of time after the melt is swept out of the contact zone. At that point, heating stops and conduction to the cool surface begins. The plateau represents the time required to remove the heat of fusion from the melt.

P.9. REFERENCES

- 1. H. E. Boyer and T. L. Gall, editors, *Metals Handbook Desk Edition*, (American Society for Metals, Metals Park, Ohio, 1985).
- 2. E. Rabinowicz, Friction and Wear of Materials, 2nd ed., (John Wiley and Sons, New York, New York, 1995).
- 3. T. Baumeister and L. S. Marks, editors, Standard Handbook for Mechanical Engineers, 7th ed., (McGraw-Hill, New York, New York, 1978).
- 4. J. S. Courtney-Pratt and E. Eisner, "The Effect of a Tangential Force on the Contact of Metallic Bodies," *Proc. Royal Soc.* A, Vol. 238, 529-550 (August 1956).
- 5. K. R. Stroh, "Analysis of Frictional Heating in Drill-String-to-Sleeve Contact," Los Alamos National Laboratory Calc-Note TSA10-CN-WT-SA-ST-092, (March 1996).
- 6. J. Bos and H. Moes, "Frictional Heating of Tribological Contacts," Journal of Tribology, Vol. 117, 171-177, (January 1995).
- K. R. Stroh, G. J. Willcutt, Jr., M. L. Walsh, and A. G. Barker-Wohletz, "LASAN-3 - A General Purpose Code for the Simulation of Large-Scale Complex Systems," Los Alamos National Laboratory report, DRAFT November 1992.
- 8. R. K. Kumar and G. W. Koroll, "Ignitability of Hydrogen/Oxygen/Diluent Mixtures in the Presence of Hot Surfaces," *Nuclear Safety*, Vol. 36, No. 1, (January-June 1995).
- 9. J. P. Holman, <u>Heat Transfer</u>, 3rd Ed., (McGraw-Hill, New York, New York, 1972).
- 10. J. Robinson, Westinghouse Hanford Company, personal communication, February 22, 1996.
- 11. R. Blickensderfer, "Methane Ignition by Frictional Impact Heating," Combustion and Flame, Vol. 25, 143-152, (1975).
- K. R. Stroh, "Analysis of Frictional Heating in Exhauster-Fan-to-Case Contact," Los Alamos National Laboratory Calc-Note TSA10-CN-WT-SA-ST-093, (March 1996).

August 8, 1996



APPENDIX Q

SAMPLE PROBLEM FOR THE COMPUTATION OF THE SITE-WIDE RISK

Q.1. INTRODUCTION

Rotary-mode core sampling (RMCS) operations will be performed in a number of tanks during the course of the year. In this safety assessment (SA), the bounding accident consequences and frequencies are computed. In doing so, a bounding tank was chosen in such a way that both the consequences and frequencies of a given accident were computed conservatively. For instance, for the dome collapse accident (App. I), the maximum drop height that corresponds to a tank that is less than half full is chosen. Likewise, the "super tank" source term that bounds all the single-shell tanks (SSTs) in the tank farm is used in quantifying the radiological consequences (App. R). The major accident scenario that leads to dome collapse is a deflagration in the dome space. In computing the accident frequency, large gas-release volumes that would normally be observed in full tanks (App. L) are used. Likewise, the minimum dome space volume is used in computing the dome concentrations (App. L), assuming a worse-case gas compositions (App. C). The bounding accident consequences are compared with the risk guidelines (RGs) (see Section 5) using the frequencies computed per year per tank basis.

However, it has been suggested that the frequency should be multiplied by the number of tanks that would be sampled during the year. There are reasons to question the validity of this approach. Simply multiplying "super tank" worse case accident frequencies by the number of sampling operations in the farm does not provide a reasonable estimate of the RMCS risk. At best, it provides a bound on the risk. The usefulness of this bound depends on its intended use.

The first reason to question whether this multiplication yields a reasonable estimate is the conservative nature of the assumptions used in deriving the bounding consequences computed in this SA. There is not a one tank that is represented by the super tank, hence there is no idea of how far removed the super tank is from a representative tank in accident frequencies and consequences. Because of the conservative nature of the SA results, no estimate of tank specific accidents frequencies has been made. Without further analysis, it is unknown, how representative the frequencies are to realistic tank conditions. In a similar vein, consequences based on a super tank are only meant to provide an upper bound. Again, it is unclear how far removed tank specific realistic consequences are from the bounding super tank consequences.

The second reason to question whether simple frequency multiplication is reasonable has to do with risk guidelines (RGs) in general, and WHC RGs in particular. Generally speaking, RG arose in the context of Probabilistic Risk Assessment (PRA). The RGs are intended to be used with PRA results, which

August 8, 1996

include conservative assumptions when warranted. However, the PRA results are not considered worse-case or bounding. Hence, RG would not normally be used with bounding results. The fact that bounding analyses yields results below the RGs means only that the realistic case is bounded. It does not provide information what the actual risk really is. More specifically, WHC RGs are not cumulative in nature. In other words, following the WHC RGs, all frequency/consequence point representing accident sequences are compared individually to the RG. It can be seen that any differences between a real tank and the super tank will manifest themselves as different points in frequency/consequence space.

Based on these discussions, we do not believe that the proposed approach of multiplying the frequencies by the number of tanks may not be adequate in obtaining a realistic risk for the RMCS operations. Such an approach may grossly overestimate the risk and result in slowing down progress in the characterization program.

In this appendix, an example problem is provided. In this sample problem, bounding SA consequences acquired using SA methodology are compared with the risk of performing the same activity in a collection of tanks. The 19 single-shell tanks (SSTs) that are on the original Flammable Gas Watch List (FGWL) are chosen as the collection of tanks in this example.

Before the quantitative evaluation of the sample problem is started, the reader must be warned that the objective is not to provide a total realistic site-wide risk for the RMCS operations. The sole objective is to demonstrate that multiplying the risk of a bounding accident obtained using the SA methodology by the number of tanks may grossly overestimate the risk. In the process, an attempt was made to use realistic assumptions and input parameters while linearizing many of the mechanistic models for simplicity.

Q.2. DESCRIPTION OF SAMPLE PROBLEM

For the sample problem, the dome collapse accident is considered. In the following subsections, the mechanistic models used for consequence and frequency analysis are discussed.

Q.2.1. Consequence Model

The model used in Appendix I shows that the respirable solid waste material (M) that becomes aerosol as a result of dome collapse is directly proportional to the impact energy of collapsing concrete and soil. Assuming a constant concrete and soil weight on each tank, the impact energy is proportional to the drop height (D).

The radiological consequences (C) of the release is proportional to the amount of respirable material (M), the atmospheric dispersion coefficient (χ/Q), and the unit dose associated with the waste material(S). Thus,

$$C \propto M \times S \times \left(\frac{\chi}{Q}\right) \propto D \times S \times \left(\frac{\chi}{Q}\right)^{-1}$$
 (Q-1)

For simplicity, the off-site consequences are considered using the approximation that the offsite receptor is at a constant distance and direction from each tank. Thus, the χ/Q for each tank is constant. However, for the bounding consequence analysis, 95% weather conditions are used. But, the tanks will be sampled at different times during the year, and the assumption that 5% weather conditions exist each time is overly conservative in assessing the risk. Assuming that the sampling activities are randomly spread over the year, the use of 50% meteorology is more appropriate for risk assessment. Thus, the ratio of individual tank consequence and the bounding consequence may be computed as

$$\frac{C_{i}}{C_{max}} = \frac{D_{i}}{D_{max}} \times \frac{S_{i}}{S_{max}} \times \frac{(\chi/Q)_{50\%}}{(\chi/Q)_{95\%}} .$$
 (Q-2)

Q.2.2. Frequency Model

For this sample problem, it is assumed that the dome collapse scenario is a dome space deflagration caused by a lightning strike when flammable gases exist in the dome space. Each tank has the same cross-sectional area. Assuming that each tank has equal lightning protection, the lightning strike frequency per tank per year is constant. Thus, the frequency of dome deflagration caused by lightning is directly proportional to time-at-risk (T), which is defined as the time period during which the dome concentration is above flammable conditions. Thus,

F∝T

Note that, for the purpose of this sample problem, lightning may be replaced with any other spark source which randomly occurs in time. To further simplify the problem, the following assumptions are introduced:

- Intrusion by the RMCS device causes a gas-release event (GRE) in each tank; and
- The volume of gas released is directly proportional to the volume of waste in the tank:

$$Q \propto V_w = a \times V_w$$

(Q-3)

where Q is the gas-release volume, a is the proportionality constant and W_w is the waste volume. This assumption may be challenged as not being applicable to all tanks depending upon the waste properties in each tank. On the other hand, it can be argued that (based on the current knowledge of the overall SST waste properties and gas release mechanisms discussed in App. L), releasing a gas volume of 5 volume percent of the waste volume in each RMCS operations conservatively bounds all the tanks.

Assuming homogeneous mixing in the dome, the dome space concentration (X) a_5 a result of the GRE is given by

$$X(t) = \begin{cases} 1 - \exp\left(-\frac{q}{V}t_{o}\right) & \text{if } 0 \le t \le t_{o} \\ \left[1 - \exp\left(-\frac{q}{V}t_{o}\right)\right] \exp\left[-\frac{w}{V}(t - t_{o})\right] & \text{if } t > t_{o} \end{cases}$$
(Q-4)

where q is the gas-release rate, V is the dome volume, t is the time after the GRE, t_0 is the release period, and w is the ventilation flow rate. Assuming an instantaneous release and linearizing the concentration decay curve, Equation (Q-4) yields:

$$X(t) = \frac{Q}{V} \left(1 - \frac{w}{V} t \right) \quad wt < V \quad .$$
 (Q-5)

Defining the lean flammability of the gas mixture as the Low Flammability Limit (LFL), the time-at-risk (T) may be obtained as

$$T = \begin{cases} 0, & \text{if } a \times V_w / V \le LFL \\ \frac{V}{w} \left(1 - \frac{LFL \times V}{a \times V_w} \right), & \text{if } a \times V_w / V > LFL \end{cases}$$
(Q-6)

Thus, the ratio of accident frequency for individual tanks to the bounding accident frequency can be computed as

$$\frac{T_{i}}{T_{max}} = \frac{V_{i}}{V_{max}} \frac{\left(1 - \frac{LFL \times V_{,i}}{a \times V_{w,i}}\right)}{\left(1 - \frac{LFL \times V_{max}}{a \times V_{w,max}}\right)}$$

(Q-7)

Two additional assumptions were introduced in deriving Eq. (Q-7):

- The same ventilation flow rate is used in all the tanks; and
- LFL is the same for all tanks and is equal to 0.04. One could introduce more details into this problem by using individual gas compositions and corresponding LFL values as discussed in Appendix C. However, such an exercise would be academic at best for the purpose of this calculation.

Q.3. RESULTS

For the bounding accident, we use the following parameters,

 $V_{max} = 50000 \text{ ft}^{3} \text{ (App. C)},$ $D_{max} = 30 \text{ ft (App. I)},$ $S_{max} = 2.4 \times 10^{5} \text{ Sv/L (App. R)},$ $(\chi/Q)_{95\%} = 1.9 \times 10^{-5} \text{ s/m}^{3} \text{ (App. R)},$ $V_{w,max} = 130,000 \text{ ft}^{3} \text{ (App. C)}, \text{ and}$ a = 0.05 (conservative estimate based on discussions in App. L).

The 50% atmospheric dispersion coefficient is obtained from Reference 1 as approximately 10% of $(\chi/Q)_{95\%}$. The individual tank data are obtained from Appendix C and listed in Table Q-1. Also shown in Table Q-1 are the individual tank source terms [denoted as (Sa) obtained from Agnew's database].² This database includes only the ¹³⁷Sr, ¹³⁷Cs, and Pu. These nuclides make up ~70% of the bounding source term given in Appendix R. Thus, when using Agnew's data the comparable bounding source term would be $S_{max} = 1.6 \times 10^5 \text{ Sv/L}$ (Ref. 2). The estimates by Agnew are based on historical fill records and a set of defined waste types that are based on many factors. Agnew defines a tank layer model (TLM), supernate mixing model (SMM) and total inventory (TI). In this sample problem, we use the TLM values because they are more representative of the total tank waste, and we used the bounding value where the maximum dose for each nuclide is chosen among the TLM, SMM and TI values. This is similar to the "super tank" approach used for the source terms in Appendix R. Instead of choosing the maximum nuclide for the entire tank farm, the maximum nuclide for a given tank is used. This source term is denoted as $S_{a,max}$ in Table Q-1.

Another alternative to using Agnew's data would be to use the SST flammable gas subset source term developed in Reference 3 for all the tanks. It can be argued that,

Q-5

because the database is small, this subset does not necessarily bound all the flammable-gas tanks. However, for the purpose of this sample problem, a realistic source term for each tank rather than a bounding source term for all the tanks is of interest. The SST flammable-gas subset given in Reference 3 is possibly still conservative for these purposes. The bounding SST flammable-gas source term is given as 6.1×10^3 Sv/L (Ref. 3).

Details of the calculations are given in Reference 4. The results are summarized in Table Q-2.

| | V | Vw | D | Sa | S _{a,max} |
|--------|--------------------|--------------------|------|--------|--------------------|
| TANK | (ft ³) | (ft ³) | (ft) | (Sv/L) | (Sv/L) |
| A-101 | 51,900 | 127,000 | 16 | 800 | 20,910 |
| AX-101 | 76,600 | 102,400 | 21 | 1460 | 38,500 |
| AX-103 | 163,500 | 15,600 | 41 | 5260 | 39,350 |
| S-102 | 65,200 | 75 <i>,</i> 500 | 20 | 400 | 2740 |
| S-111 | 65,200 | 75,500 | 20 | 1360 | 3950 |
| S-112 | 68,100 | 72,500 | 20 | 1170 | 6900 |
| SX-101 | 115,600 | 63,300 | 30 | 1580 | 2040 |
| SX-102 | 103,500 | 75 <i>,</i> 500 | 27 | 730 | 2820 |
| SX-103 | 88,700 | 90,200 | 24 | 1110 | 1510 |
| SX-104 | 91,600 | 87,300 | 25 | 1110 | 2200 |
| SX-105 | 84,300 | 94,600 | 23 | 1300 | 4750 |
| SX-106 | 102,700 | 76,200 | 27 | 660 | 1370 |
| SX-109 | 144,000 | 35,000 | 37 | 2710 | 2710 |
| T-110 | 57,600 | 54,500 | 16 | 60 | 60 |
| U-103 | 50,600 | 61,500 | 15 | 710 | 820 |
| U-105 | 56,900 | 55,200 | 16 | 920 | 960 |

TABLE Q-1 INDIVIDUAL TANK DATA

TABLE Q-1 (cont) INDIVIDUAL TANK DATA

| | v | V _w . | D | Sa | S _{a,max} |
|-------|--------------------|--------------------|------|--------|--------------------|
| TANK | (ft ³) | (ft ³) | (ft) | (Sv/L) | (Sv/L) |
| U-107 | 50,200 | 61,900 | 15 - | 2690 — | 11,270 |
| U-108 | 51,300 | 60,780 | 15 | 1570 | 17 <i>,</i> 500 |
| U-109 | 58,700 | 53,400 | 17 | 1780 | 6080 |

In Table Q-2, c represent the consequence from individual tanks as a percent of the bounding consequence (C_{max}). Thus,

$$c = 100 \frac{C_i}{C_{\text{max}}} \qquad (Q-8)$$

The individual accident frequencies (f) for the tanks are expressed as percent of the accident frequency for the bounding case (F_{max}). Thus,

$$f = 100 \frac{F_i}{F_{max}}$$
 (Q-9)

We define the risk as the product of frequency and consequences. The term r in Table Q-2 is defined as the percent fraction of the bounding risk. Thus,

 $r = 100 \frac{R_i}{R_{max}} = 100 \frac{F_i C_i}{F_{max} C_{max}}$ (Q-10)

As shown in Table Q-2, the total consequences range between 1.4 and 8.4% of the bounding consequences using different source terms. The lowest cumulative consequence is obtained using Agnew's best-estimate source term. The highest consequence results from Agnew's bounding source term. These results suggest that, using conservative tank specific models, the offsite consequences of 19 consecutive dome collapses in a year is less than 10% of the bounding dome collapse consequences computed in this SA.

The major contribution to the overall consequences (>30%) is from Tank AX-103 which has a very small waste volume and a large dome volume. Thus, the deflagration frequency for this tank is 0 resulting in a 0% contribution to the overall risk.

| TANK | c ₁ (%) | c ₂ (%) | c ₃ (%) | f (%) | r ₁ (%) | r ₂ (%) | r3 (%) |
|--------------|--------------------|--------------------|--------------------|-------|--------------------|--------------------|--------|
| A-101 | 0.14 | 0.03 | 0.69 | 100 | 0.14 | 0.03 | 0.70 |
| AX-101 | 0.18 | 0.06 | 1.66 | 89 | - 0.16 | -0.06 | 1.48 |
| AX-103 | 0.35 | 0.44 | 3.32 | 0 | 0.00 | 0.00 | 0.00 |
| S-102 | 0.17 | 0.02 | 0.11 | 58 | 0.10 | 0.01 | 0.07 |
| S-111 | 0.17 | 0.06 | 0.16 | 58 | 0.10 | 0.03 | 0.10 |
| S-112 | 0.17 | 0.05 | 0.28 | 49 | 0.08 | 0.02 | 0.14 |
| SX-101 | 0.25 | 0.10 | 0.13 | 0 | 0.00 | 0.00 | 0.00 |
| SX-102 | 0.24 | 0.04 | 0.16 | 0 | 0.00 | 0.00 | 0.00 |
| SX-103 | 0.20 | 0.05 | 0.07 | 55 | 0.11 | 0.03 | 0.04 |
| SX-104 | 0.21 | 0.06 | 0.11 | 43 | 0.09 | 0.02 | 0.05 |
| SX-105 | 0.20 | 0.06 | 0.22 | 70 | 0.14 | 0.04 | 0.16 |
| SX-106 | 0.23 | 0.04 | 0.08 | 0 | 0.00 | 0.00 | 0.00 |
| SX-109 | 0.31 | 0.21 | 0.21 | 0 | 0.00 | 0.00 | 0.00 |
| T-110 | 0.14 | 0.00 | 0.00 | 26 | 0.04 | 0.00 | 0.00 |
| U-103 | 0.13 | 0.02 | 0.03 | 50 | 0.06 | 0.01 | 0.01 |
| U-105 | 0.14 | 0.03 | 0.03 | 29 | 0.04 | 0.01 | 0.01 |
| U-107 | 0.13 | 0.08 | 0.35 | 51 | 0.07 | 0.04 | 0.18 |
| U-108 | 0.13 | 0.05 | 0.54 | 48 | 0.06 | 0.02 | 0.26 |
| U-109 | 0.14 | 0.06 | 0.21 | 21 | 0.03 | 0.01 | 0.04 |
| TOTAL | 3.6% | 1.5% | 8.4% | 746% | 1.2% | 0.3% | 3.2% |

TABLE Q-2 SAMPLE PROBLEM RESULTS

Subscripts

(1) Flammable Gas Source Term (Ref. 3)
 (2) Agnew's Best-estimate Source Term S_a(Ref. 2)

(3) Agnew's Bounding Source Term S_{a,max} (Ref. 2)

The total relative frequency may be obtained as 7.5 (750% of the bounding tank frequency). The major contributors are tanks A-101 (100%) and AX-101 (~90%) because they have a large waste volume to dome volume ratio compared to other tanks. Thus, for this sample problem, doing RMCS activities in 19 tanks increases the frequency by a factor of 7.5 instead of a factor of 19 that the conservative approach would require.

Finally, this sample problem shows that, using conservative tank-specific parameters, RMCS operations in 19 FGWL tanks would result in a total risk to an off-site individual, which is only 1.2% of the bounding risk computed in this SA (using the flammable-gas source term). Using Agnew's upper-bound source terms for each tank, the total risk becomes 3.2% of the risk obtained from bounding analysis. For the conservative case, nearly 70% of the total risk is from Tanks A-101 and AX-101 whereas the remaining 17 tanks only contribute 30% of the total risk. The maximum risk from an individual tank is from Tank AX-101, which is only 1.5% of the bounding risk.

Q.4. SUMMARY AND CONCLUSIONS

The sample problem discussed in this appendix clearly illustrates that in obtaining the total risk associated with the RMCS operations in multiple tanks over a year, we should not take either the consequences or frequencies obtained for one tank and multiply it by the number of tanks. This approach would be overly conservative. The above sample problem demonstrates that using tank-specific yet conservative estimates, the total risk of dome collapse risk in running the RMCS device in 19 FGWL SSTs is ~3.2% of the bounding risk calculated in this SA per tank. Three big factors that reduce the total risk are as follow:

- The use of tank-specific source terms,
- The use of 50% meteorology in computing the atmospheric dispersion factors, and
- The use of tank-specific gas-release volume and time-at-risk magnitudes derived from a simple scale model.

Based on the analysis provided in Section 5 of this SA, the radiological offsite consequences of a dome collapse accident is < 10 rem. The mitigated frequency of the dome collapse accident for a bounding tank is shown to be < 1.0E-6/yr in this SA. Assuming a frequency of 1.0E-06/yr, the risk is equal to 1.0E-5 rem/yr.

As shown in this sample problem, the worse case consequence among the 19 FGWL tanks analyzed is 0.3 rem. In order to exceed the radiological offsite RGs, the accident frequency must be > 1.0E-2/yr.

In a typical risk assessment, one would complement the above calculations with an uncertainty analysis. The purpose in this appendix is not to provide a full risk assessment. First, a conservative source term is used even for tank specific conditions. Further, the orders of magnitude differences between the proposed approaches demonstrate the point, and further analysis using these linearized simple models will be at most academic at this stage.

If one were interested in obtaining the site-wide risk associated with running the RMCS in all the flammable-gas tanks, an approach similar to the one shown here can be used. The use of the bounding accident consequences and frequencies as being applicable to each tank to estimate the site-wide risk will greatly overestimate the risk and unnecessarily slow down the progress in the tank characterization program.

Q.5. REFERENCES

- 1. C. Unal, "Evaluation of Atmospheric Dispersion Coefficients at Elevated Discharge Temperatures, Velocities, Diameters and Release Heights by Using AI-RISK Computer Program," Los Alamos National Laboratory Calc-Note TSA10-CN-WT-SA-TH-078 (August 1995).
- 2. R. T. Perry, "Bounding Radioactive Source Term for Hanford Tanks," Los Alamos National Laboratory Calc-Note TSA10-CN-WT-SA-RAD-009 (November 1995).
- 3. R. J. Van Vleet, "Radionuclide and Chemical Inventories for the Single-Shell Tanks," Westinghouse Hanford Company report WHC-SD-WM-TI-543 (1993).
- 4. K. P. Pasamehmetoglu, "Computation of the Site-Wide Risk," Los Alamos National Laboratory Calc-Note TSA10-CN-WT-SA-TH-113 (February 1996).

APPENDIX R

SOURCE TERM, CRITICALITY, RADIATION EXPOSURES AND CONSEQUENCE CALCULATION METHODOLOGY

R.1. INTRODUCTION

In this appendix, the following issues are addressed:

- Section 2.0: Bounding source terms applicable to single-shell tank (SST) waste are summarized. The inhalation doses resulting from the bounding source term also are included in this section.
- Section 3.0: The criticality issues associated with the rotary-mode core sampling (RMCS) operations are discussed.
- Section 4.0: The high-efficiency particulate air (HEPA) filter loading and resulting radiation exposure doses are estimated. Radiation doses from the shielded receiver and the drill unit also are discussed in this section.
- Section 5.0: This section summarizes the methodology used for estimating the receptor doses from material releases.

R.2. SST SOURCE TERM

The radionuclides used in this safety assessment (SA) for the source term are based on an evaluation of the data characterizing the tank waste made by Westinghouse Hanford Corporation (WHC) engineers. The results are based upon the information contained in References 1 through 6. In addition, files of sample data collected by the Tank Characterization Program, Tank Characterization Report, and the Tank Contents Database maintained by Pacific Northwest Laboratories for WHC were also used. Scatter plots of activities versus tank and tables containing very high and very low activities were prepared from the reports and data noted. The results are given in Reference 7.

Eleven radionuclides that account for 99.9% of the total dose based on the data from References 1 and 2 were plotted. On each plot, a line was drawn that represented the maximum activity concentrations obtained. When an experimental sample point higher than the line was found, it was reviewed by a panel. If the panel could not technically justify eliminating the point, it was used. Conversely, the panel would lower a point on the line if it could be justified.

After the maximum sample activity concentrations were established, a maximum sample activity composite was developed for nine tank groupings based on the recommended maximum activity concentrations for each of the nuclides. A

August 8, 1996

nuclide is listed if it contributes to 0.1% or more of the activity. These source terms for SST solids and SST liquids are listed in Table R-1. These activities would be used for external exposure calculations.

The unit-liter-dose values that are used to calculate exposure from inhalation are based on the source terms described above and dose conversion factors. The dose conversions factors are those found in Reference 8. Multiplying the dose conversion factors for each nuclide by the source term for each nuclide results in a

mmitted effective dose equivalent (CEDE). These CEDEs are summed for each aposite for a total CEDE. The CEDEs summed with effective dose equivalent (EDE) caused by external dose radiation result in a total effective dose equivalent (TEDE). Here, however, the EDEs are negligible for all accident scenarios except for mose involving a liquid pool. The results here assume that the TEDE is equal to the CEDE. The CEDE for SST liquids and SST solids based on the maximum activities per liter for SST tanks are also given in Table R-1.

R.3. CRITICALITY CAUSED BY RMCS OPERATIONS

The purpose of this section is to address the criticality issues associated with the RMCS operations. It presents calculations and the reasoning made to determine if the single-shell Flammable Gas Watch List tanks are criticality safe under all conditions for the proposed rotary-core drilling.

The maximum plutonium content found in any experimental sample in SST analysis forms the basis for the ²³⁹Pu content. This value is given in Table R-1, which has the maximum value of plutonium found in any sample for SSTs.

For solids in single-shell tanks, the plutonium concentration is given as 4.4e+08 Bq/L. This value contains both the contribution from ²³⁹Pu and ²⁴⁰Pu. Plutonium-240 is not fissionable. However, assuming all activity is caused by the fissionable ²³⁹Pu, this translates to 0.191 g/L, a bounding value for ²³⁹Pu in single-shell tanks. The plutonium concentration demonstrated to be criticality safe has been calculated and reported in References 9 and 10.

The minimum critical plutonium concentration calculated by Rogers⁹ was 2.6 g/L with a 95% confidence level. He used a reasonably conservative mixture of highlevel tank waste, which did not include some significant neutron absorbers known to be in the waste. This analysis is discussed and accepted in Reference 11.

The minimum critical plutonium concentration calculated by Carter¹⁰ was 3 g/L. He states that as long as the concentration in waste remains under this value, criticality will not occur for any conditions of moderation or reflection. This analysis is discussed and accepted in Reference 12.

TABLE R-1 ACCIDENT SOURCE TERM ACTIVITY CONCENTRATIONS AND UNIT-LITER-DOSE VALUES FOR ACCIDENTS INVOLVING RELEASE OF SOLID OR LIQUID TANK WASTE

| | SST Liquid | SST Solid | Conversio | SST Liquid | SST Solid |
|-------------------|------------|------------------|-----------------------|----------------------|------------------|
| Nuclide | Source- | Source- | n Factor | CEDE- | CEDE- |
| | Bq/L | Bq/L | | Sv/L | Sv/L |
| ¹⁴ C | 1.0e+05 | 1. 2e+ 05 | 5.64e-10 | 5.9e-05 | 7.0e-05 |
| ⁶⁰ Co | 1.2e+07 | 5.3e+08 | 5.91e-08 | 7.1e-01 | 3.1e+01 |
| 79 _{Se} | N- | 1.7e+04 | 2.66e-09 | * | 4.5e-05 |
| 90 _{Sr} | 1.1e+10 | 1.7e+12 | 6.47 e- 08 | 7.2e+02 | 1.1 e +05 |
| 90 _Y | 1.1e+10 | 1.7e+12 | 2.28e-09 | 2.5e+01 | 3.9e+03 |
| ⁹⁹ Тс | 1.7e+07 | 1.2e+10 | 2.77e-10 | 4.7e-03 | 3.4e+00 |
| ¹⁰⁶ Ru | 3.3e+03 | 2.4e+0 5 | 1.29e-07 | 4.3e-04 | 3.1e-02 |
| ¹²⁵ Sb | 5.3e+04 | 2.8e+08 | 3.30e-09 | 1.8e-04 | 9.2e-01 |
| 129 _I | 1.0e+04 | 6.4e+08 | 4.69e-08 | 4.7e- 04 | 3 .0e-0 1 |
| ¹³⁴ Cs | 2.1e+05 | 2.6e+06 | 1.25e-08 | 2.6e-03 | 3.2e-02 |
| ¹³⁷ Cs | 2.3e+10 | 7.5e+10 | 8.63e-09 | 2.0e+02 | 6.5e+02 |
| ¹⁴⁴ Ce | 4.3e+01 | 1.6e+03 | 1.01e-07 | 4.3e-06 | 1.6e-04 |
| ¹⁵⁴ Eu | 2.7e+09 | 6.6e+09 | 7.73e-08 | 2.1e+02 | 5.1e+02 |
| ¹⁵⁵ Eu | 7.5e+07 | 6.4e+06 | 1. 12e- 08 | 8. 4e -01 | 7.1e-02 |
| ²³⁷ Np | * | 3.0e+07 | 1.46e-04 | * | 4.7e+03 |
| 238 _{Pu} | 9.3e+04 | 1.9e+08 | 1.0 6e- 04 | 9.8e+00 | 2.0e+04 |

TABLE R-1 (cont) ACCIDENT SOURCE TERM ACTIVITY CONCENTRATIONS AND UNIT-LITER-DOSE VALUES FOR ACCIDENTS INVOLVING RELEASE OF SOLID OR LIQUID TANK WASTE

| | SST Liquid | SST Solid | Conversio | SST Liquid | SST Solid |
|---------|------------------|-----------------|-----------------------|---------------|------------------|
| Nuclide | Source- Bq/L | Source- Bq/L | n Factor [—] | CEDE- Sv/L | CEDE- SvíL |
| 239Pu** | 3.6e+07 | 4.4e+08 | 1.16 e- 04 | 4.2e+03 | 5.2 e+ 04 |
| 241Pu | 2.8e+08 | 3.5e+09 | 2.23e-06 | 6.3e+02 | 7.8e+03 |
| 241Am | 3.7 e +07 | 3.6e+08 | 1.20e-04 | 4.4e+03 | 4. 4e+ 04 |
| | | | | | |
| ULD | | | | 1.0e+04 | 2.4e+ 05 |

*No Data Available. These radionuclides have a negligible impact on the radiological dose evaluations because of their low activity concentrations.

**The ²³⁹Pu nuclide dose contribution also includes ²⁴⁰Pu.

ULD - unit-liter dose. ULD values are given for each composite in terms of committed effective dose equivalent (Sv) per unit-liter of waste inhaled at the location of the maximum on-site/off-site individual.

Because the bounding value of 239 Pu concentration in the tanks is 0.191 g/L, the tanks are currently criticality safe because this value is at least an order of magnitude less than the minimum critical plutonium concentration. The question then is whether there is a possible mechanism resulting from the rotary-core drilling that could concentrate the plutonium into a critical configuration.

The rotary-core drilling will not add any fissionable material to the tank, and in fact, will be removing some material during the process as the drill penetrates the salt cake. Pockets of possible high concentration would not be forced into smaller volumes; they would be removed as the drill passes through. The process should actually decrease the multiplication factor by removing material.

If the drill enters a liquid layer, the material would be displaced without concentration. The plutonium concentration in liquid is less than that found in solids by an order of magnitude or more, according to Table R-1. If the liquid were forced into the salt or sludge, it would serve as a diluent, resulting again in a decrease in the multiplication factor.

If the drill enters the sludge in the bottom of the tank, the sludge would be mixed if the sampling is assumed to cause a local rollover or tank rollover through a gasrelease mechanism, resulting in a mixture with a layer with less concentration of Pu, again acting as a diluent. The above analysis and facts provide assurance that nuclear criticality within the single-shell Flammable Gas Watch List tanks will not occur as a result of RMCS.

R.4. RADIATION EXPOSURE CALCULATIONS

In this section we estimate the radiation exposure from the HEPA filters, shielded receiver, and the drill string. Only gamma exposure—is calculated. The gamma source strength in the solid is from the column labeled "SST Solid" in Table R-1. The only gamma emitter of importance is ¹³⁷mBa, a daughter of the decay of ¹³⁷Cs. The source strength of ¹³⁷Cs from the column "SST Solid" is 7.5e10 Bq/L. The ¹³⁷Cs is in secular equilibrium with ¹³⁷mBa, and 94.6% of the decay from ¹³⁷Cs produces ¹³⁷mBa. Using a salt density of 2.2 grams/cm³, the ¹³⁷mBa source strength is 8.71e-04 Ci/g.

R.4.1. HEPA Filter Loading and Radiation Exposure

The HEPA filter is designed to trap the particle escaping the tank during the drilling operations. This note gives an estimate of the exposure from the filter caused by the capture of the radioactive particles during drilling operations.

In Table 2 of Ref. 13, the total mass out of the riser and to the HEPA filter is 901 g, a value based on continuous operation of the exhauster, a total drilling depth of 266 inches, and a schedule of 40 minutes of drilling (19 inches) followed by a 60-minute delay before starting the next drilling period. Only particles greater than 5.5 μ m were assumed to have been carried up the stack. The total exhaust flow for the case used here is 250 ft³/min. These results were calculated using the TRAC-TE code. This code and the results are discussed in Reference 13. The results take into account the settling of the aerosol created. The time required to reach an equilibrium concentration was not given; however, the time required to reach an equilibrium condition for a 700 scfm with continuous exhauster flow case was given as 400 minutes. For this 700-ft³/min case, the mass flow rate out the stack for the first 100 minutes was 90% of the average mass flow rate. Thus, for the calculations in this note, it is not unreasonable to use an average exhaust rate of 901g/1400 minutes, i.e., 64.4 grams captured in the filter in 100 minutes of operation. This would be one drilling period and one hour's wait. The filter would have 64.4 grams x 8.71E-4 Ci/g loading.

Using the code MicroShield,¹⁴ a filter geometry of 24 in. x 24 in. x 11.5 in. containing SiO_2 at a density of 0.1 g/cm³, we find the exposure at 1 cm from the surface is about 300 mR/h at the end of 100 minutes of operation. This exposure may be scaled with units of 100 minutes of time because it is based on an average mass flow rate.

The maximum allowable dose for a worker is 300 mrem/week. These calculations indicate that there may be cases in which this limit could be exceeded. It is necessary then that health and safety personnel monitor the filter during operation.

Also, the size of the exposure seems to require a prefilter in the limiting cases when the source term is of this size.

R.4.2. Radiation Exposure From Shielded Receiver and Drill Unit

Shielded Receiver. The receiver is modeled as a series of concentric cylinders 19 inches long. The dimensions and materials are given in Table R-2.

The source-containing material is modeled as NaNO3. The ¹³⁷mBa source contained in the NaNO3 is 0.596 curies (7.0874e+07 Bq/cm³). Using the code MicroShield,¹⁴ we find the exposure at 1 cm from the surface is 211 mR/h. The bounding value of exposure calculated is 211 mR/h at 1 cm from the surface of the receiver. It would be expected that there would be no samples that would actually produce this value because the samples would contain liquid that have a lower source strength. In addition, the source strength used here is the maximum found in any sample. However, this value could cause a worker to exceed the limit of 300 mrem/week. It is necessary then that health and safety personnel monitor the receiver during operation.

 TABLE R-2

 MATERIALS, DIMENSIONS, AND DENSITY OF THE SAMPLE RECEIVER

| Material | Density, (g/cm ³) | Maximum Radius (in.) |
|----------|----------------------------------|-------------------------|
| | | |
| NaNO3 | 2.2 | 0.564 |
| Steel | 7.93 | 0.75 |
| Air | 0.00122 | 1.0335 |
| Steel | 7.93 | 1.1875 |
| Lead | 11.34 | 2.013 |
| Steel | 7.93 | 2.25 |

Contaminated Pintle Rod. In this section, the exposure rate from a contaminated pintle rod is determined. The exposures determined are results from the MicroShield code. In these calculations, the equivalent volume of waste that would be found on a contaminated pintle rod is modeled in the center of the quill rod. These approximations are conservative.

The volumes of waste placed in the center of the quill rod are based on a 1-mm and a 3-mm-thick contamination on the pintle rod. The pintle rod is 92.48 cm long with a radius of 0.237 cm. A 1-mm contamination gives an equivalent radius of 0.24 cm of contamination in the center of the quill rod and a 3-mm contamination gives an equivalent radius of 0.482 cm. Using the bounding value of ¹³⁷Cs found in solid waste, the two source strengths are 0.0338 and .137 Ci.

The steel quill rod has an inside radius of 2.42 cm and an outside radius of 2.875 cm. The exposure as determined using MicroShield at 1 cm from the surface of the quill rod for the 1-mm contamination is 0.72 R/h, and for the 3-mm contamination, the exposure at 1 cm for the surface of the quill rod is 2.84 R/h. At 1 meter from the surface of the quill rod, the exposure is 8.6 mR/h for the 1-mm contamination and 35 mR/h for the 3-mm contamination.

The magnitude of these exposures indicates that radiation monitoring of the quill rod is necessary to prevent unnecessary exposure to personnel in the event that the rod becomes contaminated.

R.5. RECEPTOR DOSE CALCULATIONS

In this section, the radiological dose calculation methodology is discussed. The methodology is adopted from Reference 15 for consistency with other safety basis documents employed at the Hanford Site. In Reference 16, this methodology is compared with the AI-RISK methodology¹⁷ previously used in the Tank 101-SY Mixer Pump SA.¹⁸ The receptor dose is calculated as

$$D = Q \times \frac{\chi}{Q'} \times R \times ULD$$
 (R-1)

where

- D is the dose in Sv,
- Q is the material release volume (L),

 χ/Q' is the atmospheric dispersion coefficient (see Tables R-3 and R-4),

R is the breathing rate

= $3.3 \times 10^{-4} \text{ m}^3/\text{s}$ typical acute breathing rate (light activity)

R-7

August 8, 1996

= $2.7 \times 10^{-4} \text{ m}^3/\text{s}$ typical chronic breathing rate (24 h average), and

ULD is the unit liter dose in Sv/L obtained from Table R-1.

The atmospheric dispersion coefficients are obtained using the computer code GXQ, which uses the methods in the Nuclear Regulatory Commission (NRC) Regulatory Guide 1.145 (Ref. 19). The atmospheric dispersion coefficient is the timeintegrated normalized air concentration at the receptor for radiological releases and For puff chemical releases the atmospheric continuous chemical releases. dispersion coefficient is the instantaneous maximum concentration at the receptor. Only the plume meander correction was used in the calculations. The coefficients used to calculate dose or concentration are presented in Tables R-3 through Table R-4. The acute release values were developed for weather conditions that result in downwind concentrations exceeded only 0.5% of the time in the maximum sector (16 sectors) or 5% of the time for the overall site. The larger of the two values was used as the integrated bounding value for the on-site and off-site individuals. The chronic annual average χ/Q values were obtained by calculating chronic annual average values for each sector and using the highest value for calculation purposes. This value is suitable for long-term releases but not for accidents.

TABLE R-3 DISPERSION COEFFICIENTS

| | χ/Q' | χ/Q' with PM* | CAA** χ/Q' |
|----------|---------------------|---------------------|---------------------|
| RECEPTOR | (s/m ³) | (s/m ³) | (s/m ³) |
| Onsite | 3.44 E-02 | 1.13 E-02 | 4.03 E-04 |
| Offsite | 1.88 E-05 | 1.49 E-05 | 9.16 E-08 |

*PM: plume meander correction applied

** CAA: chronic annual average

For a release duration of less than 1 h, the χ/Q' (first column) is used. For a release duration between 1 and 2 h, the plume meander correction is applied, and the χ/Q' with PM (second column) is used. For a release duration greater than 8760 h (1 yr), the CCA χ/Q' (third column) is used. For a release duration between 2 h and 8760 h, a logarithmic interpolation between values in column 2 and column 3 is used. The offsite receptor distances for the 200-Area tank farms obtained from Reference 15 are shown in Table R-4.

| Direction | Distance (km) |
|-----------|----------------|
| S | 16.650 |
| SSW | 16.650 |
| SW | 13.875 |
| WSW | 11.100 |
| W | 11.100 |
| WNW | 11.100 |
| N W | 14.800 |
| NNW | 14.800 |
| N | 16.650 |
| NNE | 23.125 |
| NE | 19.425 |
| ENE | 15.725 |
| Е | 15.725 |
| ESE | 21.275 |
| SE | 24.97 5 |
| SSE | 21.275 |

TABLE R-4SITE BOUNDARY DISTANCES FOR THE TANK FARMS15

One case referred to in this SA requires the evaluation of the dispersion coefficient for a 1-wk (168-hr) duration release. Applying the logarithmic interpolation, the onsite and off-site dispersion coefficients may be obtained as

$$\frac{\log(1.13\times10^{-2}) - \log\left(\frac{\chi}{Q'}\right)}{\log(1.13\times10^{-2}) - \log(4.03\times10^{-4})} = \frac{\log(2h) - \log(168h)}{\log(2h) - \log(8760h)} = \frac{\chi}{Q'} = 1.94\times10^{-3}$$

and

$$\frac{\log(1.49 \times 10^{-5}) - \log\left(\frac{\chi}{Q'}\right)}{\log(1.49 \times 10^{-5}) - \log(9.16 \times 10^{-8})} = \frac{\log(2h) - \log(168h)}{\log(2h) - \log(8760h)} = \frac{\chi}{Q'} = 1.01 \times 10^{-6}$$

respectively.

Table R-5 provides the receptor doses for a 1-L release.

TABLE R-5RECEPTOR DOSES FOR A 1-LITER RELEASE

| | ONSITE | | OFF | SITE | |
|---------------|--------------|--------------|--------------|--------------|--|
| | SST Solid | SST Liquid | SST Solid | SST Liquid | |
| Prompt (< 1h) | 2.76E+00 Sv | 1.15E-01 Sv | 1.49E-03 Sv | 6.20E-05 | |
| | 2.76E+02 rem | 1.15E+01 rem | 1.49E-01 rem | 6.20E-03 rem | |
| 1 week | 1.54E-01 Sv | 6.40E-03 Sv | 8.00E-05 Sv | 3.33E-06 Sv | |
| | 1.54E+01 rem | 6.40E-01 rem | 8.00E-03 rem | 3.33E-04 rem | |
| 1 month | 8.65E-03 Sv | 3.60e-05 Sv | 3.31E-05 Sv | 1.38E-06 Sv | |
| | 8.65E-01 rem | 3.60E-02 rem | 3.31E-03 rem | 1.38E-04 rem | |
| 1 year | 3.20E-03 Sv | 1.33E-05 Sv | 7.26E-06 Sv | 3.02E-07 Sv | |
| | 3.20E-01 rem | 1.33E-02 rem | 7.26E-04 rem | 3.02E-05 rem | |

R.6. REFERENCES

- 1. R. J. Van Vleet, "Radionuclide and Chemical Inventories for the Double Shell Tanks," Westinghouse Hanford Company report WHC-SD-WM-TI-543 (1993).
- 2. R. J. Van Vleet, "Radionuclide and Chemical Inventories for the Single Shell Tanks," Westinghouse Hanford Company report WHC-SD-WM-TI-565 (1993).
- 3. D. J. Braun, et al., "High-Level Waste Tank Subcriticality Safety Assessment," Westinghouse Hanford Company report WHC-SC-WM-SARR-003 (1994).
- 4. F. M. Jungfleisch, "TRAC: Preliminary Estimation of the Waste Inventories in Hanford Tanks Through 1980," Westinghouse Hanford Company report WHC-SD-WM-TI-057 (1984).
- 5. S. R. Agnew, "Estimated Chemical and Radiochemical Inventories Spreadsheet: NE Quadrant, A, AX, B, BX, BY, C Farms," Westinghouse Hanford Company report WHC-SD-WM-TI-628, Rev. 0, (1994).
- 6. S. R. Agnew, "Estimated Chemical and Radiochemical Inventories Spreadsheet: SW Quadrant, S, SX, U Farms," Westinghouse Hanford Company report WHC-SD-WM-TI-631, Rev. 0 (1994).
- 7. W. L. Cowley, "Tank Waste Source Term Inventory Validation," Westinghouse Hanford Company report WHC-SD-WM-ER-400, Rev. 0 (1995).
- 8. K. F. Eckerman, et al., "Limiting Values of Radionuclide Intake and Air Concentration and Dose Conversion Factors for Inhalation, Submersion, and Ingestion," Federal Guidance Report No. 11 (DE89-011065) Office of Radiation Programs, US Environmental Protection Agency (1988).
- 9. C. A. Rogers, "An Analytical Model for Evaluating Subcritical Limits for Waste in Hanford Site Storage Tanks," Westinghouse Hanford Company report WHC-SD-SQA-CSA-20356 (1993).
- 10. R. D. Carter, "CSAR 79-007 Underground Waste Storage Tanks and Associated Equipment," Westinghouse Hanford Company report SD-SQA-CSA-20108 (1979).
- 11. D. J. Braun et al., "High-Level Waste Tank Subcriticality Safety Assessment," Westinghouse Hanford Company report WHC-SD-WM-SARR-003 (1994).
- 12. T. S. Vail, "Justification for Continued Operation of Hanford High-level Waste Tanks Resulting from the Criticality" Westinghouse Hanford Company report USQ, 492-CRITSAF, WHC-SD-WM-JCO-001 Rev. 3 (1993).

August 8, 1996

- 13. P. M. Francis, "Aerosol Study for the Rotary Mode Exhauster," Westinghouse Hanford Company report WHC-SD-WM-ES-225 REV 0 (1993).
- 14. C. A. Engine and G. Work, Micro Shield Version 4, (Grove Engineering, Inc., Rockville, Maryland, 1992).
- 15. "Tank Waste Composition and Atmospheric Dispersion Coefficients for Use in Accelerated Safety Analysis Consequence Assessment," Westinghouse Hanford Company report WHC-SD-WM-SARR-016, Rev. 1 (October 1995).
- 16. K. O. Pasamehmetoglu, "Comparison of the Radiological Consequence Analysis Used in the Mixer Pump SA with the ASA Analysis," Los Alamos National Laboratory Calc-Note TSA10-CN-WT-SA-TH-073 (October 1995).
- 17. Y. Y. Chien, and D. MacFarlane, "AI-RISK: A Computer Program for Calculating Doses and Health Risk from Accidental Release of Radioactive Materials," Los Alamos National Laboratory report LA-UR-92-2636, (July 1992).
- L. H. Sullivan et al., "A Safety Assessment for the Proposed Pump Mixing Operations to Mitigate Episodic Gas Releases in Tank 241-SY-101: Hanford Site, Richland, Washington," Los Alamos National Laboratory report, LA-UR-92-3196, Rev. 14 (March 31, 1995).
- 19. "Atmospheric Dispersion Models for Potential Accident Consequence Assessments at Nuclear Power Plants," Regulatory Guide 1.145, US Nuclear Regulatory Commission, (1982)

APPENDIX S

SPILL RELEASE FRACTIONS

S.1. INTRODUCTION

Several spill scenarios are identified in the hazards evaluation. One of the most important scenarios is a spill from the core barrel when the core barrel is flooded. The other one is a spill from the core sampler.

If waste collects inside the core barrel or the drill rods, handling could cause a release. Even though accumulated waste may be discovered by radiological monitoring, this would only aid workers in preparing to catch the spill that could follow removal.

The length of the core barrel is 1.1 m (40 in.), and the inside diameter is 5.08 cm (2 in.). Thus, the core barrel could contain 2.2 L (0.07 ft³) of waste. Using a waste density of 1.6 kg/L (100 lb/ft³), the corresponding mass is obtained as 3.5 kg (7.8 lb).

The sampler is 48 cm (19 in.) long and has a 1-in. (2.54-cm) diameter. The volume of the sampler is 0.25 L (15 in.³) and would contain 0.4 kg (0.9 lb) of waste.

S.2. ANALYSIS

In order to estimate the aerosol fraction during a spill, the waste may be modeled as liquid or solid.

The airborne material (source term) released from a liquid spill can be evaluated from the following equation, as suggested in the Department of Energy (DOE) Handbook:¹

where

MAR = material at risk,

DR = damage ratio = 1,

F = airborne release fraction,

RF = respirable fraction, and

LPF = leak path factor = 1.

The LPF and DR are conservatively assumed as 1.

(S-1)
S.2.1. Respirable Aerosol Fraction for Liquid Spills

Reference 2 (Sec. 4.4.2.2, p. 4.74) summarizes the experimental release fraction from 1 quid spills. Most of the experimental data are obtained at Pacific Northwest Laboratory (PNL)³ and are for viscous liquids and slurries. The spill height was 3 m (10 ft) in PNL's experiments. Other parameters such as the diameter of particles, densities of liquids and solids, and the viscosity of slurry and liquids were varied. Also included in the database was a slurry containing important species of 101-SY waste. The resulting correlation is given as

$$F = 6.31 \times 10^{-6} Ar^{0.45} \left(\frac{\rho_{air}}{\rho_{liq}}\right)^{2.2} Fr^{0.35}$$

where

F = fraction airborne,

 $\rho_{air} = air density,$

 ρ_{liq} = solution density,

Fr = Froud number, and

Ar = Archimedes number.

Froud and Archimedes numbers are defined as

$$Fr = \frac{V^2}{gR}$$
 and $Ar = \frac{\rho_{liq}^2 h^3 g}{\mu^2}$

where

h = spill height,

g = gravitational constant,

 μ = viscosity of solution,

V = impact velocity, $(2gh)^{1/2}$,

R = radius of liquid drop, $(3/4 \text{ Vol}/\pi)^{1/3}$, and

Vol = volume of solution.

August 8, 1996

(S-3)

(S-2)

Equation (S-2) shows that the respirable fraction is proportional to the drop height and drop velocity and inversely proportional to the volume of the material and viscosity. Using a bounding drop height of 5 m (15 ft), a minimum liquid volume of 0.25 L (7.1 x 10⁻⁶ ft³), a liquid density of 1000 kg/m³ (62.4 lb/ft³), and a liquid viscosity of 1 cP, the respirable fraction (F) is obtained as 1.0×10^{-4} . Note that the liquid density and viscosity used in this calculation correspond to water density and viscosity, which is very conservative for the waste supernate.

S.2.2. Respirable Aerosol Fraction for Solid Waste Spills

To estimate the respirable aerosol fraction of a solid waste spill, we use the formula provided by MacDougall et al.4 given by the following simple relationship:

$$F = 2 \times 10^{-10} \frac{E}{Vol}$$
 , (S-4)

where E is the impact energy in joules and Vol is the material volume in m^3 .

Equation (S-4) was derived from Argonne National Laboratory experiments (Refs. 5, 6, and 7) involving the brittle fracture of small samples of glass, ceramics, uranium dioxide pellets, and concrete. A test consisted of placing a single cylindrical specimen on its side between two hardened tool steel plates inside a sealed chamber. Each specimen received a dynamic diametrical impact by a weight dropped from a preselected height onto the upper plate. The impact energy is dissipated by disintegration, and some aerosol particulate is formed. The net respirable fraction of the brittle material (particulates with a diameter of less than 10 μ m) is given by F, which is found to be linearly proportional to the impact energy. If the energy is set equal to the impact energy of the material being spilled, Eq. (S-4) reduces to

$$F = 2 \times 10^{-10} (\rho g h)$$
 , (S-5)

where ρ is the waste density taken as 1600 kg/m³ (100 lb/ft³) and h is the drop height bounded by 5 m (15 ft). Thus, the respirable fraction becomes 1.6 x 10⁻⁵.

S.3. CONCLUSIONS

In this safety assessment, a conservative value of 2×10^{-4} (Ref. 8) is used for the respirable fraction during a spill accident. The analysis provided in Section 2 of this appendix shows that a release fraction of 2×10^{-4} is bounding for liquid and solid waste spills of interest.

WHC-SD-WM-SAD-035, Rev. 0-a

S.4. REFERENCES

- 1. "Airborne Release Fractions/Rates and Respirable Fractions For Nonreactor Nuclear Facilities" DOE Handbook 3010-94 (October 1994).
- 2. J. E. Ayer et al. "Nuclear Fuel Cycle Facility Accident Analysis Handbook," NUREG-1320, Nuclear Regulatory Commission (April 1988).
- 3. M. Y. Ballinger and W. H. Hodgson, "Aerosols Generated By Spills of Viscous Solutions and Slurries," PNL-5910, NUREG/CR-4658, Nuclear Regulatory Commission (November 1986).
- 4. H. R. MacDougall, L. W. Scully, and J. R. Tillerson, "Preliminary Preclosure Radiological Safety Analysis," Appendix F in "Site Characterization Plan Conceptual Design Report," Sandia National Laboratory report SAND84-2641 (1987).
- W. J. Mecham, L. J. Jardine, R. H. Pelto, G. T. Reedy, and M. J. Steindler, "Interim Report of Brittle-Fracture Impact Studies: Development of Methodology," Argonne National Laboratory report ANL-81-27 (October 1981).
- 6. L. J. Jardine, W. J. Mecham, G. T. Reedy, and M. J. Steindler, "Final Report of Experimental Laboratory-Scale Brittle Fracture Studies of Glasses and Ceramics," Argonne National Laboratory report ANL-82-39 (October 1982).
- 7. W. J. Mecham, L. J. Jardine, G. T. Reedy, and M. J. Steindler, "General Statistical Description of the Fracture Particulates Formed by Mechanical Impacts of Brittle Materials," Ind. Eng. Chem. Fundam., 22, 384-391 (1983).
- 8. R. M. Marusich, "Safety Assessment for Push Mode Core Sampling in Tank 101-SY," Westinghouse Hanford Company report WHC-SD-WM-SAD-008, Rev. 0-A (May 10, 1991).

S-4

APPENDIX T

BUREAU OF MINES IGNITION TEST PROGRAM FUNCTIONAL REQUIREMENTS AND ACCEPTANCE CRITERIA

T.1. INTRODUCTION

The hazard identification process presented in Section 3 of this safety assessment (SA) resulted in numerous burn initiators in the dome, in the drill string, in the waste and aboveground. Section 4 discusses these accidents and their possible initiators. Some of these burn accidents are postulated to occur as a result of mechanical sparks during rotary-mode core sampling (RMCS) normal operations or accidental conditions. The frequencies of these accidents are determined in Appendix E and summarized in Section 4. The discussion in Section 4 shows that, without additional experimental studies, the consequences and frequencies of the burn accidents caused by mechanical sparks are not acceptable.

The key reason to do experiments with prototypical atmospheres is the need for a database to support safety and engineering decisions on a technical basis. Unless there is a reliable and experimentally tested theoretical basis for predicting behavior, one will be reduced to making ad hoc decisions in the absence of experimental data. There is no general and reliable theory of frictional ignition that will enable test results with one mixture and frictional ignition situation to be extrapolated to a very different mixture or frictional stimulus. In the multicomponent mixtures characteristic of the flammable gases in the Hanford tanks, the interactions of the various components are a key aspect of behavior that can only be conclusively resolved in experimental studies.

Table T-1 summarizes the accident scenarios in which the mechanical sparks are initiators for a burn accident and the experimental results are necessary to conclude that the risk is bounded by the guidelines. Also included in Table T-1 is the section number where the bounding tests for each accident is discussed. Note that a separate series of tests for each accident is not needed, and a number of accidents (mainly the drop accidents) in Table T-1 may be bounded by one set of tests.

As a requirement of this SA, an experimental test program is implemented to demonstrate that no sparks capable of igniting a bounding flammable-gas mixture under bounding operating conditions can be generated as a result of the accidents listed above. These experiments are being performed at the U.S. Bureau of Mines (BOM) In some cases, a bounding set of tests are used to cover more than one accident scenario shown in Table T-1. In this appendix, the necessary BOM ignition tests required by this SA, their functional design requirements, and acceptance criteria are described for each ignition test. The results of the tests performed at the BOM is also discussed in this appendix.

TABLE T-1 SUMMARY OF ACCIDENT CASES FOR WHICH TEST RESULTS ARE NEEDED

| Ignition Scenario | Description | Section |
|---|---|----------------------|
| lgnition caused by frictional sparks as a result of the drill bit penetration | During the drill-bit penetration through the crust or waste sludge, a frictional spark can ignite the flammable gas in the vicinity of drill bit. | 2.0 |
| Ignition under the waste surface caused by frictional sparks by the drill bit | The ignition of the flammable gas in the waste by the frictional sparks created by the drill bit is of concern if the drill bit strikes hard materials such as rocks, metals, or hard waste layers. | 2.0 |
| Ignition in the riser caused by a frictional spark | If the conductive sleeve-nitrogen purge system fails, ignition of flammable gases is possible. Another case considered is the ignition of hydrogen in the riser (between the conductive sleeve and riser). | 3.0 |
| Ignition in the drill string caused by assembly/disassembly of drill strings or drill rod-quill rod adapter impact | The ignition of the flammable gas in the drill string caused by impacts that could be created during drill string or drill string quill rod assembly or disassembly are possible. | 4.0, 6. 0 |
| Ignition in the riser caused by the drill string, equipment, or tool drop | The flammable gas in the riser/dome can be ignited if equipment, drill string, or tools are dropped into or onto the riser during the installation or removal phase. | 4.0 and 6.0 |
| Ignition caused by the drill string or tool drop on crust | Dropping the drill string or other tools on the crust may ignite the flammable gas in the vicinity of the crust. | 4.0, 5.0, and 6.0 |
| Ignition in the drill string caused by drops | Ignition in the drill string as a result of frictional sparks, dropping core sampler are possible. | 4.0, 5.0, and 6.0 |
| Ignition in X-ray or cask | The ignition source is the drop of the sampler into the X-ray container or the cask. Low-impact energies and velocities are involved. | 5.0 |
| Ignition in the shielded receiver | The drop of core sampler in the shielded receiver could cause spark. Low-impact energies and velocities are involved. | 5.0 |

T.2. DRILL BIT FRICTIONAL IGNITION TESTS

One of the safety concerns of sampling with rotary mode in single-shell flammablegas tanks is the ignition of the flammable gases retained in the waste. The ignition of flammable gas may occur in several different ways:

- The waste may be hard enough to create a frictional spark as the drill bit penetrates the salt cake or any other hard layer in the waste;
- The drill bit may also strike metal debris and cause a metal-to-metal friction resulting in a spark; and

• Friction on hard debris may cause local hot spot generation during which the temperature exceeds the autoignition temperature of the gas mixture.

The objective of drill bit frictional ignition tests is to <u>demonstrate</u> that the operation of rotary core drilling in a bounding frictional environment and bounding gas composition does not cause an ignition. The following are the sections in which the test parameters for the ignition tests are discussed:

- Downforce (Section 2.1.)
- Rotational Speed (Section 2.2.)
- Nitrogen flow rate (Section 2.3.)
- Gas Temperature (Section 2.4.)
- Gas Composition (Section 2.5.)
- Debris Material and Configuration (Section 2.6.)
- Drill Bit Type (Section 2.8.)
- Number and Duration of Tests (Section 2.7.)
- Autoignition Temperatures (Section 2.9.)

T.2.1. Downforce

The downforce used during tests must be at least 120% of the nominal downforce that will be used during drilling in the actual waste. Currently, the analysis provided in Appendixes F and N determine the maximum allowable downforce. In ignition testing, the downward force must be \geq 120% of the limiting value defined by the envelope tests or structural analysis.

T.2.2. Rotational Speed

Like the downforce, the rotational speed (rpm) must be set higher than the nominal rotational speed that will be used during actual sampling operations, which is 55 rpm. For the tests, the rotational speed must be $\geq 120\%$ of the limiting value defined by the envelope tests or structural analysis (approximately 65 rpm). This margin is believed to be sufficient for protection against accidentally exceeding the nominal speed, considering the other conservative features of the test.

T.2.3. Nitrogen Purge Flow Rate

During ignition tests, the nitrogen purge system will not be used.

T.2.4. Gas Temperature

To bound the maximum waste temperatures in all the flammable-gas tanks (including the measurement uncertainties), the gas mixture must be $\geq 100^{\circ}$ C.

T.2.5. Gas Composition

The initial analyses of the limited data show that the gas composition varies in each tank (App. C). Thus, the frictional ignition testing using actual drill bits must be performed in a bounding gas composition and bounding contact configuration.

It is very difficult to determine a realistic gas composition that bounds all possible combination of species mentioned above for all single-shell tanks. Using the current database, one can only determine major flammable-gas species that may be assumed to exist in all of the single-shell tanks. Based on the analysis provided in Appendix C, the major flammable-gas species are hydrogen (H₂), nitrous oxide (N₂O) which is an energetic oxidizer and NH₃. Based on observation of waste simulants, free oxygen (O₂) is not expected in the waste gas. However, air addition to the waste is possible during the RMCS operation.

In summary, the important gas species to consider are H_2 , NH_3 , N_2O , O_2 , (used to conservatively bound the effect of entrained air). Instead of trying to determine a realistic gas composition that may bound all tanks, the stoichiometric mixtures are chosen. In determining the stoichiometric mixtures, the findings of the previous studies are used.

Krok and Shepherd¹ were unable to ignite a typical Tank 101-SY gas mixture (without ammonia and air) by striking two bars together or creating sparks by grinding. They were able to ignite the 101-SY gas mixture with a grinder striking on a metal piece when it is mixed with air fractions between 75% and 90%. It is concluded that, in the gas mixtures tested, N₂O acts as an ignition inhibitor. References 2 and 3 also confirm that a typical 101-SY gas mixture cannot be ignited with the frictional operation of the drill bit in a simulated waste crust or on a steel plate. Based on these findings, it appears that a stoichiometric H₂-O₂ mixture would be bounding for mechanical sparks.

The evidence about the ability of bronze and brass strikers to cause frictional ignition is controversial. Powell⁴ cites evidence for ignition with hydrogen and methane but not with gasoline fumes in an enriched oxygen atmosphere. The phenomena of frictional ignition is sensitive to surface phenomena such as the oxidation state of the metal and contaminants such as aluminum. Very small amounts of aluminum or an aluminum-containing compound on a surface can

WHC-SD-WM-SAD-035, Rev. 0-a

produce frictional ignition in situations that otherwise would be considered absolutely safe. Ammonia is known to be chemically reactive with copper and other base metals containing compounds and will alter the state of the surface. This clearly raises the issue that the frictional ignition characteristics of a surface may be affected by ammonia. There have been no studies on frictional ignition in the presence of ammonia. Although one may suspect that frictional ignition of ammonia will be difficult because of the low flame speed of ammonia-air mixtures, that is really not a valid reason to rule out ammonia as playing a role in frictional ignition in multicomponent mixtures. Thus, a stoichiometric mixture is introduced (keeping the hydrogen-to-ammonia ratio as 1) of H₂-NH₃ and O₂ in the test matrix.

Finally, although the earlier studies¹⁻³ show that mechanical sparks are not expected to ignite H_2 - N_2O mixtures, the addition of ammonia to this mixture has not been tested. References 5 and 6 indicate that ammonia may be very explosive in a nitrous oxide atmosphere. The studies documented in References 5 and 6 do not include mechanical sparks. However, because of the high explosiveness of this mixture, a stoichiometric mixture of H_2 - NH_3 and N_2O also must be tested.

Thus, as part of the acceptance criteria, three gas mixtures must be tested. These mixtures are summarized in Table T-2.

| Species | Mixture #1 | Mixture #2 | Mixture #3 |
|---------------|------------|------------|------------|
| Hydrogen | 66% | 30% | 20% |
| Oxygen | 34% | 40% | - |
| Ammonia | * | 30% | 20% |
| Nitrous Oxide | • | • | 60% |

TABLE T-2 GAS MIXTURES

T.2.6. Debris Material and Configuration

The condition of waste in terms of hardness is not known before operation. There exists a possibility of penetrating a very hard waste layer in a tank. In addition, there may exist some metal debris lost or dropped from the riser in the past. Hard materials such as rocks also may exist in the waste. Thus, it is likely that the drill bit may strike against metal and other hard objects during the operation.

The possibility of ignition is higher if the drill bit strikes on harder materials such as carbon or stainless steel rather than hard salt cake or crust. Because of this, the use of a simulant to model the frictional sparks during the waste penetration in the proposed testing is not necessary. Instead, the bounding spark initiating contact condition, i.e., striking metal objects or rocks are considered.

The conditions of the metal or hard object and geometry of the contact is another key parameter for ignition testing. The metal objects may already be corroded and may have a rough surface. The drill bit may strike wedge-shaped objects, a flat metal surface, or the edge of an inclined metal sheet or pipe. The configuration of drill bitto-metal contact may play a role in generating a frictional spark that is capable of igniting the flammable-gas mixture. The metal object may be made of soft carbon steel as well as hard carbon steel. Thus, the selection of a bounding drill bit contact configuration and material needs to be addressed by testing.

Two bounding materials are selected for the frictional ignition tests;

- 1. 4140 hardened tool steel shapes, and
- 2. rocks found in the area of the Hanford Tank Farms.

The basis for the selection of 4140 hardened alloy steel is because of its relative incendivity and radiance properties. The relative radiance indicates the ability to create a spark (Ref. 7). Relative incendivity is the parameter to describe the ability to ignite hydrogen-air mixture in an ignition test. Both parameters are defined relative to the properties of pure iron.

Relative incendivity is defined as the inverse of the time to ignition in a given ignition test as described in Reference 7. This parameter is found to be a function of the applied loading pressure and surface speed. The relative incendivity of the hardened 4140 steel was found to be much higher than mild steel, structural steel, and stainless steel.

Relative radiance (as quantified by the measured radiant flux) for 4140 hardened tool steel is also much higher than the carbon steel (1018, 1096, 1030, 1080, 1040, 1060, etc.).

Figure T-1 shows the three sharp-edge shape made of 4140 hardened tool-steel piece before the ignition test. This contact geometry is believed to be bounding to cause frictional sparks. Figure T-2 illustrates the drilling process onto these sharp-edged steel pieces. Figure T-3 is the photograph of the drill bit after drilling on these pieces. As shown, the drill bit surface is quite damaged.

T.2.7. Duration and Number of Tests

For each gas mixture #1 (stoichiometric H_2 - O_2) shown in Table T-2 and for each material (metal and rocks as described previously), a minimum of five tests must be conducted. If there are uncertainties in the bounding configurations, multiple configurations may be tried as part of the minimum number of tests required.

The number of tests for the other two gas mixtures shown in Table T-2 must be ≥ 3 for each material and each gas mixture tested. Different configurations may be used to count towards the requirement of the minimum number of tests for each material.

In the accident analysis, the operators are credited for a few minutes response time in case the penetration rate control is violated. For consistency with that credited control, each test must be run for a duration ≥ 3 minutes.

Tests must be conducted in well-mixed mixtures at elevated temperatures. The test chamber must be purged sufficiently before the flammable gas is introduced into the chamber. The flammable gas must be introduced into the chamber for a sufficient period. The gas composition in the chamber must be verified. To verify that wellmixed flammable gases existed during the test, the gases must be ignited with an alternate ignitor at the end of each test if a frictional ignition does not occur during the test.

T.2.8. Drill Bit Type

The drill bit cutting teeth are made of a sintered bronze with tungsten chips in the bronze matrix. This material can wear down easily when the drill bit strikes metal objects or hard materials. The core sample drill bits used in RMCS are Longyear (trademark of Longyear Incorporated) Parts Numbers 100IVD/8 (currently used) and 9505-15E (new prototype bit). BOM ignition testing also considers a new drill bit (Longyear part number 9505-15E).⁵ This new drill bit is not used during testing to determine the safe operating parameters to prevent waste ignition (Appendix F); therefore, it is not considered in this SA. The purpose of specifying the drill bit is not exactly to indicate that no other drill bit can be used in an RMCS operation. This appendix summarizes the requirements of tests. However, any drill bit that is to be used in RMCS operations must pass both ignition and envelope testing requirements. The safe operating parameters (envelope tests) are obtained from experiments performed with the current drill bit. Therefore, the drill bit model number is specified to indicate that this type is the only one tested in ignition and envelope tests. When the drill bit model is changed or a new one is developed, it must pass the requirements of ignition and safe envelope testing, before it can be used in RMCS operations.

The current and new drill bits, current and new one, include carbon steel pins in the base of the drill teeth. Westinghouse Hanford Company (WHC)⁵ currently is performing ignition tests for both bits to demonstrate that both drill bits do not ignite a flammable-gas mixture that may bound all possible combinations of flammable gases in single-shell Flammable Gas Watch List tanks (SSFGWLT). Tests are designed to simulate the action of a drill bit striking a hard object in the waste, such as a piece of structural steel or a rock, and determine what, if any, core drilling conditions exist that could ignite the flammable-gas mixtures. Tests were conducted in a bounding stoichiometric hydrogen oxygen and ammonia nitrous oxide mixture as required by this SA.

Tests to date with different structurally sharp carbon-steel objects consistently showed no ignition in the H_2 - O_2 mixture. However, drilling on a hard rock resulted

WHC-SD-WM-SAD-035, Rev. 0-a

in ignition of the hydrogen oxygen mixture when the new bit was used. Examination of the new drill bit after the ignition indicated that the teeth were worn and carbon-steel pins were exposed to the rock.

The conclusion of testing is that the ignition of bounding mixtures of hydrogen and oxygen is likely if carbon-steel pins contact steel or rock targets. Because ignition is observed and the reason was proven to be the carbon-steel pins, this SA requires that the carbon steel pins or carbon steel components must be removed from the tooth region of the bit before using them in the tanks. A control also is implemented to replace the drill bit if a trip signal is received on the penetration rate four times with a cumulative penetration of 0.3 in. This control and the requirement to pass the tests listed in this appendix control the spark sources in the drill bit.

When the drill bit is redesigned to include pins with different materials such as stainless steel or brass, a series of ignition testing could be set up. In these tests, ignition tests with 3-min. drilling periods should be performed until the hydrogenoxygen mixture is ignited. After each 3-min period, a waiting period must be implemented to make sure the starting temperatures were the same. The number of tests that give the ignition should be higher than 4; otherwise, control must be changed to half of this new number. If ignition occurs in the first or second test, the material must be re-evaluated.

T.2.9. Autoignition Temperatures

During one of the rock-drilling tests that was run for more than 5 minutes, the hydrogen oxygen mixture was ignited after 6 minutes of testing. It was postulated that the ignition occurred because the autoignition temperature at the teeth surface was reached. The test was repeated, and the ignition was observed at almost the same time. The bit teeth were not worn sufficiently to cause the carbon steel blanks to be exposed. Additional tests without flammable gas were performed to determine the interface temperature. In one of the tests, a thermocouple was placed 1/8 in. beneath the assorted rock. The rock was not worn significantly; therefore, the temperature just beneath the rock could be measured. The rock temperature 1/8 in. beneath the surface was 236° C after 6 minutes of testing. An infrared temperature probe was also used to determine the surface temperature of the teeth. After 4 minutes of testing, temperatures up to 400° C were observed. All of this evidence indicates that the autoignition temperature can be reached if the drilling lasts more than 5 minutes.

To ensure that, during drilling into the rock without purge gas and minimal insertion rate, the temperatures will not reach autoignition temperatures, the following tests must be performed. Ten runs shall be performed while drilling into a rock with gas mixture #1, and with the downforce and rotational speed specified above. The run period shall be 3 minutes. After each 3-minute run, the bit and the rock shall be allowed to cool to approximately ambient temperature. The bit face

and the rock surface shall be photographed after each run. Ignition during any of the runs shall be sufficient cause to disqualify the tested bit design from FG/RMCS.

Envelope Testing

The possibility of causing an exothermic waste reaction exits if the drill bit-waste interface temperature exceeds 160°C. The operating parameters such as downward force, rotational speed, and nitrogen purge flow are necessary to be controlled not to cause unacceptable drill bit/waste temperature during drilling. Appendix F describes the testing requirements for thermal performance of the drill bit in addition to requirements in this appendix for frictional spark issue.

T.3. CONTACT BETWEEN DRILL STRING AND TANK RISER MATERIAL

Upon loss of nitrogen purge flow to the riser (annulus between the drill string and conductive sleeve), frictional sparks can be generated through the drill string and conductive sleeve contact. In order to demonstrate that sparks cannot be generated from drill string and sleeve contact, ignition tests must performed by the BOM. Four different sets of tests must be conducted in a stoichiometric hydrogen-oxygen mixture. A prototype drill rod was rotated within a test chamber with a side load, 200 lbf, pressed against it. The side load acts against the drill string itself and not against the drill bit. The following materials must be used for the drill string:

- 1. A standard uncoated steel drill string rubbing against carbon steel;
- 2. A standard steel drill string with a pipe joint compound on it rubbing against a carbon steel
- 3. A nickel plated and fluted drill string rubbing against carbon steel;
- 4. A fluted drill string with the nickel coating ground off rubbing against carbon steel.

Note that these test parameters are conservative because the conductive sleeve is made of a stainless-steel pipe and if flammable gas exists in the riser, it is a hydrogen-air mixture. The rotational speed is 65 revolutions per minute (rpm). A total number of 12 tests are conducted (three repeat tests for each material discussed above).

The steel selected was 4140 carbon-steel. The 4140 carbon-steel is bounding as discussed previously. If tests are performed with actual configuration (drill string is inside of the sleeve), the annulus must be purged with the well mixed gas mixture. If the drill string is rubbing on the outside surface of a pipe or a flat steel surface, only adequate mixing in the vicinity of the contact point must be provided. The

temperature of the well mixed hydrogen-oxygen should be ~100°C. The purge flow rate should be minimized to provide well-mixed flammable-gas mixture.

The test period must be longer for these tests. The maximum period must be 26 minutes. This number is obtained by dividing the average drill_rod length, 19 in., with the minimum penetration rate of 0.75 in./min.

These tests must show that the rotational motion of a carbon-steel drill string or a nickel-plated fluted drill string cannot create the ignition of a stoichiometric hydrogen oxygen mixture when the rotational speed and normal force are controlled.

T.4. DROP OF A NINETEEN-INCH DRILL STRING SECTION

A spark during disconnecting the quill rod from the drill string has been observed in the field. The quill rod adapter and the drill string were made of carbon steel. Any misalignment between the drill string and the quill rod caused by undesired platform movement or operator errors could create a relatively fast impact between the quill rod and the drill string when the drill string is disconnected. There is no instrument to detect the misalignment or any stress level on the drill string or quill rod adapter. Therefore, it is very difficult to evaluate the condition of drill stringquill rod adapter before disconnecting the drill string. Because a spark is observed in the field operation, this event must be assumed to have a high likelihood.

The DS is assembled by adding the drill rods. The addition of a drill rod could be performed as follows:

- 1. Manual installation by an operator; or
- 2. Manual installation using the lifting bail.

During both modes, drill-rod-to-drill-string impact also is possible. Impact can be caused by a drop of the drill rod or by an operator error. This accident is likely because the drill rods are made of carbon steel and assembling/disassembling is performed for each sample. The cable spray washer is installed after the DS is disconnected from the quill rod adapter. The change out assembly is installed after cable spray washer. If dropped on the DS, the change-out assembly also may cause a spark.

Ignition tests are designed to simulate the impact of dropping a 19-in. drill rod on a vertically oriented DS from a height of 3 feet. A height of 3 feet corresponds to an impact velocity of 14 ft/s. These tests must be performed 30 times because the expectation is that carbon-steel pieces could create sparks when impacted. Provided that these tests conducted with realistic conditions result in no ignition of stoichiometric hydrogen-air mixture, the probability of ignition may be

estimated using binomial theorem. For 30 tests with no ignition, the probability of ignition is $\sim 2.0 \times 10^{-2}$. Considering this probability, the accident frequency becomes on the order of $\sim 10^{-7}$.

A well-mixed gas mixture must be provided to the impact point. This is a key issue in these drop tests. To achieve mixing, the gas mixture must be provided with a minimum flow rate so as not to cause significant convective effects inside the stationary drill string. The purge time and flow rate must account for the volume of the test chamber as well as the drill string volume. Appropriate mixing in the test chamber also must be provided. The gas temperature must be $\geq 100^{\circ}$ C, and the gas mixture must be a stoichiometric mixture of hydrogen-air for these tests.

Tests must not induce ignition the of the stoichiometric hydrogen-air mixture.

T.5. DROP TEST OF A ROTARY MODE SAMPLER ON TO A ROTARY BIT

Drop tests simulating the drop of the core sampler on the drill bit were performed in the following fashion. A prototype core sampler must be dropped from a height of 60 feet through a prototype drill string on the drill bit attached to the lower end of the drill string. The section of the core barrel containing the grooves, or serrated edges, where the quadralatch fingers latch, and the quadralatch fingers and body must be made of 304 stainless steel to reduce the likelihood of a spark. The test chamber and drill string must be filled with a stoichiometric hydrogen-air mixture. Ignition is not expected because the bottom of the core sampler is made of stainless steel. Tests must be repeated ten times to confirm this expectation.

A well-mixed gas mixture at the impact point must be provided. Note that the 60-foot DS may require significantly larger purge periods. The gas mixture must be purged with a minimum flow inside of the stationary DS in all tests. The purge time and flow rate must account for the volume of the test chamber as well as the long DS volume. Mixing in the test chamber should also be provided. The gas temperature must be $\geq 100^{\circ}$ C.

If there are pins inside the current drill bit, they need to be scratched so that the core sampler can impact on carbon steel pins. For this experiment, you must use the current drill bit (Longyear, trademark of Longyear Incorporated) Part Number 100IVD/8). The number of tests must not be less than 10.

These tests must conclude that no ignition of stoichiometric hydrogen-air mixtures is observed in ten core sampler drop tests performed under these conditions.

T.6. QUILL ROD ADAPTOR-DRILL STRING IMPACT TESTS

A misalignment between the quill rod adapter and drill string could cause sparks when the drill string is disconnected from the quill rod adapter. This event can occur during both RMCS. The quill rod adapter impact test used a section of the same type of stainless-steel pipe dropped on its end onto the end of a section of carbon drill string. Calculations (Ref. 8) showed that the maximum kinetic energy during drill-string/quill-rod misalignment is 115 in.-lb. A safety factor of 2 was employed, raising the kinetic energy to 230 in.-lb. A 5.22-lb piece was dropped 44 in. onto the carbon-steel pipe to simulate this impact.

A well-mixed gas mixture of stoichiometric hydrogen and either oxygen or air (66%/34%) with a minimum flow rate must be provided to the impact point. The purge time and flow rate must account for the volume of the test chamber as well as the long DS volume. The gas temperature must be $\geq 100^{\circ}$ C. The gas mixture needs to be supplied both to the drop tube and the test chamber.

These tests must be repeated 10 times and must show no ignition.

T.7. RESULTS OF IGNITION TESTING PERFORMED AT BUREU OF MINES

Witwer⁸ performed the ignition testing required by this appendix at the BOM. The procedure and test setup, including the recipes of the flammable gas described by Witwer in his report, are reviewed and found to be consistent with the requirements of this appendix. Results of each test described in this appendix did not show an ignition of the flammable mixtures of hydrogen-oxygen, hydrogen-ammonia-oxygen and hydrogen-ammonia-nitrous oxide, as reported by Witwer.⁸ Therefore, the accident scenarios, including the ignition of flammable gases caused by mechanical sparks (frictional sparks, sparks resulting from impact), are considered to be not credible for RMCS operations.

T.8. SUMMARY AND CONCLUSIONS

All of the required tests described in this appendix were performed at BOM by Witwer.⁸ All the test procedures and results are documented in Ref. 8. The test procedure and findings of this report have been reviewed, and they were found to be consistent with descriptions given in this appendix. Results showed no ignition in all of the tests described in this appendix. The conclusion of ignition testing therefore is used to address the postulated fire accidents caused by mechanical sparks discussed in Section 4.

The core sample drill bits used in BOM tests were Longyear (trademark of Longyear Incorporated) part numbers 100IVD/8 (currently used) and 9505-15E (new prototype bit). These drill bits are qualified for FG/RMCS based on BOM test results. However, there are no envelope tests for the new prototype bit, Number 9505-15E.



Fig. T-1. The photograph of three sharp, hardened 4140-tool steel shapes before drilling.



Fig. T-2. The photograph illustrating the three sharp, hardened 4140-tool steel shapes during drilling.



Fig. T-3. The photograph of the drill bit after drilling on a hardened 4140-tool steel shape.

The FG/RMCS equipment may be modified in the future. Any modifications made in drill bits, core samplers, drill rods or operating conditions specified in this appendix must be assessed against ignition test results given in this appendix. Proposed design or operating parameter changes which are not bounded by the results of ignition testing discussed in this appendix will require performance of additional bounding tests. The USQ process will be applied to ensure that any design or operational parameter changes are bounded by the results of the ignition testing in this SA.

T.9. REFERENCES

- J. C. Krok and J. E. Shepherd, "Electrical and Frictional Spark Ignition of N2O-H2-N2-air Mixtures," Los Alamos National Laboratory report, LANL-WT-800, (June 1993).
- K. L. Cashdollar, K. S. Witwer, A. Furno, G. M. Green, R. A. Thomas. "Ignitability Testing for Core Drilling Systems," Westinghouse Hanford Company report WHC-SD-WM-TRP-224 (March 1995).
- K. L. Cashdollar, M. Herzberg, I. A. Zlochower, C. E. Lucci, G. M. Green, and R. A. Thomas, "Laboratory Flammability Studies of Mixtures of Hydrogen, Nitrous Oxide, and Air," Westinghouse Hanford Company report WHC-SD-WM-ES-219, Rev. 0 (August 1992).
- 4. M. Hertzberg and I. A. Zlochower, "Explosibility of Nitrous Oxide: The Effect of H-Atom Bearing Impurities," Twenty-Fifth International Symposium on Combustion, University of California, Irvine (July 31– August 5, 1994).
- 5. E. Jones and J. C. Kerr, "Inflammability Limits of Ammonia, Nitrous Oxide and Air," J.S.C.I., Vol. 68 (January 1949), pp. 31–34.
- 6. F. Powell "Ignition of Gases and Vapors," Industrial and Engineering Chemistry, Vol. 61, No. 12, p. 29 (1969).
- R. Blickensderfer, J.E. Kelley, D.K. Deardorff, and M.I. Copeland, "Testing of Coal-Cutter Materials for Incendivity and Radiance of Sparks," Bureau of Mines Report of Investigation, RI-7715 (1972).
- 8. K. S. Witwer, "Test Report for Ignitability Testing," Westinghouse Hanford Company report WHC-SD-WM-TRP-257 (June 1996).

APPENDIX U

FUNCTIONAL REQUIREMENTS FOR FLAMMABLE AND TOXIC GAS (HYDROGEN AND AMMONIA) SENSORS

U.1. INTRODUCTION

The purpose of the sensors is to provide signals that can be used as safety shut-down indicators for both flammability (mainly hydrogen and ammonia) and toxic (mainly ammonia) hazards during rotary-mode core sampling (RMCS) operations. The primary flammability hazard is the release of hydrogen gas, and the primary toxicity hazard is the release of ammonia gas (which is also flammable). The signal will be used to shut down the drill truck and to alert personnel to evacuate the tank farm as a protection against toxic-gas exposure. The objective of this appendix is to specify the functional requirements of the chosen sensors for this applications.

U.2. EVALUATION OF THE SENSOR

At the time of the submittal of this safety assessment (SA), two sensors have been identified and are being designed as part of the ventilation system. The first is a Wittaker Cell hydrogen detection system and the second is an SMC combustible gas sensor.

The Wittaker Cell is an electrochemical cell with a membrane placed between the sample gas and the active element. It is very selective for hydrogen and responds directly to the partial pressure of hydrogen on the other side of the membrane. It is configured with a calibration port that can be used to flood the sensor region with a calibration or zero gas during operation conditions. For the sensor to read out concentration it is essential that the pressure in the sensor region be within the expected tank pressure range during the calibration.

The Sierra Monitor Corp. (SMC) combustible gas sensor uses a catalyst to "burn" the gas and detects the resulting heat release. To increase sensitivity and decrease drift, the heat detection is done by comparing the temperature of a reference (uncatalyzed bead) to that of a signal (catalyzed) bead. The beads are imbedded in a sintered metal housing which prevents the combustion energy from igniting a flammable mixture. It has the advantage of responding to both ammonia and hydrogen.

Westinghouse Hanford Company has considerable experience with the Wittaker Cell, which has been shown to have an adequate response time as given in App. B. The Wittaker cell has been shown to have adequate sensitivity, and experience has shown that it is very reliable and stable in the current tank farm applications. Calibration is required only every three months.

The SMC detector has not been used in the tank farm environment before, but it has

recently been tested as described in a report by Straalsund. The key points of this report were the following:

- The output is noisy, which can be rectified by electrical processing but still is barely adequate for this application.
- The response of the system was very sensitive to the flow rate of the gas by the sensor.
- The response was sensitive to the orientation of the device.
- The sensor responded to ammonia as well as hydrogen with the ammonia response factor being approximately 40% of the hydrogen response.

Based on the information provided, the sensors, if implemented properly, should provide the required protection against flammability and toxicity hazards during a gas-release event (GRE). The SMC sensor, if set to trip at 5000 ppm hydrogen equivalent, should sense both a 5000 ppm hydrogen concentration and a 12000 ppm ammonia concentration for alarm purposes if the gas was purely hydrogen or ammonia, respectively.

U.3. FUNCTIONAL REQUIREMENTS FOR THE SENSORS

The requirements for the Wittaker Cell sensor deployment will be a slightly less stringent than the SMC sensor because the Wittaker cell has better sensitivity and has been demonstrated to be stable and reliable in the tank farm environment.

U.3.1. Wittaker Cell Requirements

- The system must be rated to sample from a Class-I, Division-1, Group-B environment.
- The deployment system must retain the response time requirement of reaching 90% of full scale in less than 2 minutes.
- The configuration must allow for the system to be calibrated in a manner that ensures that zero and calibration gas are within the expected tank pressure range during the calibration. The system must be able to be operated during pressure transients caused by a potential GRE.
- The calibration should be performed at initial deployment and every three months after that point. The calibration should consist of setting the zero with pure air or nitrogen and calibration with a nominal 6000 ppm hydrogen. The calibration procedure with a nominal 6000 ppm

hydrogen should test the shutoff electronics as well as the sensor reading.

U.3.2. SMC Combustible Gas Sensor Requirements

The requirements for the SMC combustible gas sensor are more stringent for several reasons:

- The sensor is sensitive to both flow and pressure.
- The signal-to-noise and lower detection limit seem to be marginal for the application.
- There is no operational history for exposure to tank farm gases.

The following are the functional requirements for the SMC combustible gas sensor:

- The system must be rated to operate in a Class-I, Division-1, Group-B environment.
- In addition to the electrical qualification, it must be documented that the sensor element itself will not ignite a flammable mixture.
- The deployment system must retain the response time requirement of reaching 90% of full scale in less than 2 minutes.
- The design must assure that the system conservatively responds to a pressure or flow rate transient anticipated during a GRE.
- Configuration of both sensors must allow the system to be calibrated in a manner that the pressure in the sensor region is within the expected tank pressure range during the calibration.
- The initial sensor must be calibrated with both hydrogen and ammonia.
- The functional test should be performed at initial setup at each location and every day the system is used after that point. The functional test must consist of setting the zero with pure air or nitrogen and calibration with a nominal 6000 ppm hydrogen. The functional test procedure with a nominal 6000 ppm hydrogen should test the shutoff electronics as well as the sensor reading.
- The sensor must be replaced at least once per month with a new sensor that has been calibrated with both hydrogen and ammonia.

WHC-SD-WM-SAD-035, Rev. 0-a

U.4. REVIEW OF FINAL DESIGN, FUNCTIONAL REQUIREMENTS, PERFORMANCE, AND CALIBRATION OF FLAMMABLE GAS DETECTION SYSTEM

The final design and calibration of the sensors are summarized in Ref. 2 The description of the system is given in Section 2. A brief summary of the system design is given below.

The flammable-gas detection system consists of four primary components: (1) a spool piece with gas sensors to obtain gas samples from the exhaust stream, (2) and (3) two identical, separate, electronic packages, and (4) a power distribution skid with redundant shutoff contactors. The system is powered by the same source as the exhauster. The flexible duct from the waste tank is attached to the spool piece that is bolted directly to the exhauster heater. The ventilation stream passes through the spool piece and into the exhauster. Attached to the spool piece are two separate flammable-gas sensors; a Whittaker hydrogen detector cell and a SMC combustible gas detector. The use of SMC and Wittaker sensors provides a redundancy in hydrogen detection. The detection of ammonia can only be performed by the SMC sensor and therefore is not redundant.

The purpose of the gas sensors on the spool piece is to provide safety shut-down signals for both flammability and toxic hazards during core sampling operations. Out-of-tolerance conditions include concentrations of hydrogen equivalent flammable gas greater than 5000 ppm, or concentration rate increases greater than 100 ppm/s. Upon detection of out-of-tolerance conditions, the interlock will initiate drill rig shutdown and alert personnel to evacuate the tank farm.

Review of Ref. 2 indicates that both the Wittaker cell and SMC combustible gas sensors are qualified to be operated in Class-I, Div.-1, Group-B environments.

The calibration of both sensors consisted of setting the zero with pure air or nitrogen and calibrating with a nominal 6000 ppm of hydrogen. Calibration involved hydrogen concentrations from zero to a nominal 6000 ppm with a step increase in concentration. No unexpected nonlinearity was observed, and calibration results were acceptable.

The response time of the SMC combustible gas sensor was experimentally measured using the orientation of the prototype. These tests are performed with calibrated sensors. The hydrogen concentration was increased from 0 to a nominal 6000 ppm. Both sensors meet the requirement of reaching 90% of the full scale (a nominal 6000 ppm) in less than 2 minutes, which was used in determining the trip set point of the flammable-gas detector.

Configuration of both sensors allows the system to be calibrated in a manner that zero and calibration gas are exposed to sensors within the expected pressure as during the operation mode.

Pressure and flow transient tests simulating a GRE are performed. Results confirm that the sensors adequately operate under an anticipated transient condition.

Both sensors themselves do not ignite a flammable mixture when they are operational.

SMC sensor, if set to trip at a 5000-ppm hydrogen equivalent, should sense both a 5000-ppm hydrogen concentration and a 12000-ppm ammonia concentration for alarm purposes if the gas were purely hydrogen or ammonia, respectively. The set point for a rate of increase of concentration is 100 ppm/s for 10 seconds.

Wittaker cell, if set to trip at a 5000-ppm hydrogen equivalent, should sense a 5000-ppm hydrogen concentration for alarm purposes if the gas were purely hydrogen. The set point for a rate of increase of concentration is also 100 ppm/s for 10 seconds.

A control is established for an SMC sensor to be initially calibrated with hydrogen and ammonia and an SMC sensor to be replaced at least once per month with a new sensor that has been calibrated with both hydrogen and ammonia. A control is also established for the SMC sensor to be calibrated in the following fashion. A functional test shall be performed at the initial setup at each location and every day the system is used after that point. The functional test shall consist of setting the zero with pure air or nitrogen and calibration with a nominal 6000-ppm hydrogen. The functional test procedure with a nominal 6000-ppm hydrogen shall test the shutoff electronics as well as the sensor readings. A functional calibration test shall be performed once every three months for the Whittaker cell.

U.5. CONCLUSIONS

This appendix presents the functional requirements of flammable-gas sensors used in RMCS operations. A reliable, fast, accurate, and redundant flammable-gas detector system is designed to be operated in steady-state exhauster operating conditions as well as during rapid transient surges. They are located in a spool piece upstream of the preheater in the exhauster. Flammable-gas sensors consist of a Wittaker cell and an SMC combustible gas sensor.

Review of the design and results of calibration of both sensors and how the design meets the requirements given in this appendix reveal that the flammable-gas detection system will function as desired to detect GRE and flammability on the tank dome and shut down the drill engine. Requirements listed in this appendix for both Wittaker and SMC combustible gas sensors are concluded as being fulfilled.

The following controls are established for the flammable-gas detection system:

- SMC sensors shall be replaced at least once each month with new sensors,
- New SMC sensors shall be initially calibrated in a laboratory environment using hydrogen and ammonia,

- A functional calibration test shall be performed in the field at the initial setup and then every day for the SMC sensor,
- A functional calibration test shall be performed once every three months for the Whittaker cell,
- The functional calibration test shall consist of setting the zero with pure air or nitrogen and calibration with a nominal 6000-ppm hydrogen for both sensors,
- The functional calibration test procedure with a nominal 6000 ppm hydrogen shall test the shutoff electronics as well as the sensor reading for both sensors.
- The set point for tripping both sensors shall be a 5000-ppm hydrogen equivalent.

The set point for a rate of increase in hydrogen equivalent concentration is 100 ppm/s for 10 seconds.

U.6. REFERENCES

- 1. E. Straalsund, "Combustible Gas Sensor Interference Report," Westinghouse Hanford Company report WHC-SO-WM-TRP-249 (January 1996).
- 2. E. K. Straalsund, Flammable Gas-Detection System and Calibration "Mid-Columbia Engineering report MCE-RPT-001 (June 1996).



| | | | | 96/ | 3861 | |
|---|--|---|---|--|---|--|
| | ENGINEERING C | HANGE NOTICE | Pege | 1 of | 1. ECN 60 Proj. ECN | 999 |
| | | | | | | |
| ECN Category (mark one) | 3. Originator's Name and Telephone No. | e, Organization, MSIN, | 3a. USQ Req | uired? | 4. Date | |
| Supplemental M | R. E. Raymond, | 75200, S7-12 | [] Yes | 🕻 Но | 4/30/96 | |
| Direct Revision | 5. Project Title/No. | /Work Order No. | 6. Bldg./Sy | s./Fac. No. | 7. Approval De | signato |
| Temporary PC [] | Rotary Mode Core S | Sampling System (RMCS) | Tank Farm | Facilities | SQD | |
| Supersedure [] | 8. Document Number | s Changed by this ECN | 9. Related | ECN No(s). | 10. Related PO | No. |
| | WHC-SD-WM-IS | DSR-005, Rev. 0 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 | N, | /A | N/A | |
| 11a. Modification Work | 11b. Work Package | 11c. Modification Work | r Complete | 11d. Restor | ed to Original C | ondi- |
| 1 Yes (fill and | Νο. Ν/Δ | N/Δ | | tion (Temp. N/Δ | or Standby ECN | only) |
| Blk.11b) | | Cog. Engineer Signat | ure & Date | Coa. Ena | ineer Signature 3 | & Date |
| [X] No (NA Blks. 11b, | 4 A | | / _ | | | |
| SU+SM+OSR+OO5, Isteet rev | 15100." | | | | · · · · · · | |
| REFERENCE WHC-IP-0954, Rev. 5, 1 shall have the following : L. M. Co 3.7.1, 3.7.2, 3.7. 5.31.7, 5.31.8, 5.31.9, 5 HM-0SR-005. Rev. 0 and the | "Tank Farms Interim Op statement added to page "The IOSR's for Rotary of the RMCS in verify implementation .3, 3.7.4, 3.7.5, and J .31.10, 5.31.11, and 5 e Table of Contents (Te | erational Safety Requirer e 2, the Executive Summan Mode Core Sampling (RMC flammable gas tanks of this IOSR. Administrative Controls 5 .31.12 (SEE ATTACHED PAGE OC) will be modified to p | nents Complian TY- S) will be ful 5.31.1, 5.31.2 S) will be ad reflect these | ce Implementa ly implement readiness ass , 5.31.3, 5.3 ded to Sectio additional re | ition Plan," Rev. ed prior to depl essment/review w i1.4, 5.31.5, 5.3 ins 3 and 5 of WH cuirements. | 5, oyment ill i.6, IC-SD- |
| REFERENCE *. WHC-IP-0954, Rev. 5, Shall have the following shall have | "Tank Farms Interim Op statement added to page "The IOSR's for Rotary of the RMCS in verify implementation .3, 3.7.4, 3.7.5, and J .31.10, 5.31.11, and 5 e Table of Contents (Te cone) | erational Safety Requirer e 2, the Executive Summar Mode Core Sampling (RMC flammable gas tanks of this IOSR. Administrative Controls 5 .31.12 (SEE ATTACHED PAGE OC) will be modified to r | nents Complian ry. S) will be ful 5.31.1, 5.31.2 S) will be ad reflect these | ce Implementa ly implement readiness ass , 5.31.3, 5.3 ded to Sectio additional re | ition Plan," Rev. ed prior to depl essment/review w 1.4, 5.31.5, 5.3 ins 3 and 5 of WH equirements. | oyment ill i1.6, IC-SD- |
| REFERENCE WRC-IP-0954, Rev. 5, 1 shall have the following : 1.4 LCO 3.7.1, 3.7.2, 3.7. 5.31.7, 5.31.8, 5.31.9, 5. HM-OSR-005, Rev. 0 and the 13a. Justification (mark of Criteria Change [] | "Tank Farms Interim Opstatement added to page "The IOSR's for Rotary of the RMCS in verify implementation .3, 3.7.4, 3.7.5, and J .31.10, 5.31.11, and 5 e Table of Contents (To one) Design Improvement | erational Safety Requirer e 2, the Executive Summan Mode Core Sampling (RMC flammable gas tanks of this IOSR. Administrative Controls 5 .31.12 (SEE ATTACHED PAGE DC) will be modified to r | nents Complian ry. S) will be ful 5.31.1, 5.31.2 S) will be add reflect these | ce Implementa ly implement readiness ass , 5.31.3, 5.3 ded to Sectio additional re Facili | ed prior to depl essment/review w 1.4, 5.31.5, 5.3 uns 3 and 5 of WH equirements. | oyment ill il.6, IC-SD- |
| REFERENCE WHC-IP-0954, Rev. 5, shall have the following shall | "Tank Farms Interim Op statement added to page "The IOSR's for Rotary of the RMCS in verify implementation .3, 3.7.4, 3.7.5, and J .31.10, 5.31.11, and 5 e Table of Contents (To one) Design Improvement Facilitate Const s n to implement the leve in Flammable Gas Sign | erational Safety Requires e 2, the Executive Summan / Mode Core Sampling (RMC flammable gas tanks of this IOSR. Administrative Controls 5 .31.12 (SEE ATTACHED PAGE DC) will be modified to r [X] Environmental [] Const. Error/ el 1 controls established le Shell Tanks. Hanford S | nents Complian ry. S) will be ful 5.31.1, 5.31.2 S) will be add reflect these [] Omission [] d in the Safet site. Richland | ce Implementa ly implement readiness ass , 5.31.3, 5.3 ded to Sectio additional re Facili Design y Assessment Washington | ed prior to depl essment/review w i1.4, 5.31.5, 5.3 ms 3 and 5 of WH equirements. ty Deactivation Error/Omission "A Safety Assess " WHC-SD-WM-SAD- | 5, oyment ill |
| ReFERENCE WHC-IP-0954, Rev. 5, shall have the following shall | "Tank Farms Interim Opestatement added to page "The IOSR's for Rotary of the RMCS in verify implementation .3, 3.7.4, 3.7.5, and J. .31.10, 5.31.11, and 5. e Table of Contents (To one) Design Improvement facilitate Const s n to implement the leve in Flammable Gas Sing name, MSIN, and no. of ent Control, Kel | erational Safety Requirer e 2, the Executive Summary Mode Core Sampling (RMC flammable gas tanks of this IOSR. Administrative Controls 9 .31.12 (SEE ATTACHED PAGE DC) will be modified to r [X] Environmental [] Const. Error/4 el 1 controls established le Shell Tanks, Hanford S f copies) 1 S. Lenz (N4-18) | nents Complian ry. (5) will be ful (5.31.1, 5.31.2 (5) will be addreflect these [] Omission [] d in the Safet Site, Richland | ce Implementa ly implement readiness ass , 5.31.3, 5.3 ded to Sectio additional re Facili Design y Assessment , Washington, | ed prior to depl essment/review w i1.4, 5.31.5, 5.3 ms 3 and 5 of WH equirements. ty Deactivation Error/Omission "A Safety Assess " WHC-SD-WM-SAD- | 5, oyment ill :1.6, IC-SD- [] [] :ment of 035, |
| REFERENCE WRC-IP-0954, Rev. 5, shall have the following 1. Justification (mark of Criteria Change [] As-Found [] 13b. Justification Details The IOSR requires revision Rotary Mode Core Sampling Rev. 0 14. Distribution (include Unclassified Docume | "Tank Farms Interim Operations statement added to page "The IOSR's for Rotary of the RMCS in verify implementation .3, 3.7.4, 3.7.5, and A .31.10, 5.31.11, and 5. e Table of Contents (To one) Design Improvement Facilitate Const son to implement the leve in Flammable Gas Single name, MSIN, and no. of ent Control, Kel | erational Safety Requires e 2, the Executive Summary Mode Core Sampling (RMC flammable gas tanks of this IOSR. Administrative Controls 9 .31.12 (SEE ATTACHED PAGE OC) will be modified to r [X] Environmental [] Const. Error/M el 1 controls established le Shell Tanks, Hanford S f copies)]i S. Lenz (N4-18) | nents Complian ry. S) will be ful 5.31.1, 5.31.2 S) will be addreflect these [] Omission [] d in the Safet Site, Richland | ce Implementa .ly implementa readiness ass , 5.31.3, 5.3 ded to Sectio additional re Facili Design Assessment Washington, | ed prior to depl essment/review w 11.4, 5.31.5, 5.3 ins 3 and 5 of WH equirements. ty Deactivation Error/Omission "A Safety Assess " WHC-SD-WM-SAD- RELEASE STAMP HAXEGRD RELEASE STAMP | 5, oyment ill (1.6, IC-SD- [] [] (] () () () () () () () () () () () () () |
| REFERENCE WRC-IP-0954, Rev. 5, shall have the following defined and the following for the following for the following for the following for the following for the following for the following for the following for the following for the following for the following for the | "Tank Farms Interim Operations statement added to page "The IOSR's for Rotary of the RMCS in verify implementation .3, 3.7.4, 3.7.5, and J .31.10, 5.31.11, and 5. e Table of Contents (To one) Design Improvement Facilitate Const s n to implement the leve in Flammable Gas Sing name, MSIN, and no. or ent Control, Kel | erational Safety Requirer e 2, the Executive Summary Mode Core Sampling (RMC flammable gas tanks of this IOSR. Administrative Controls 9 .31.12 (SEE ATTACHED PAGE OC) will be modified to r [X] Environmental [] Const. Error/4 el 1 controls established le Shell Tanks, Hanford S f copies)]i S. Lenz (N4-18) | nents Complian ry. (5) will be ful (5.31.1, 5.31.2 (5) will be addreflect these (1) (1) (1) (1) (1) (1) (1) (1) (1) (1) | ce Implementa ly implement readiness ass , 5.31.3, 5.3 ded to Section additional re Facilit Design (Assessment Washington, DATE: STA: 4 SEP 0 | ed prior to depl essment/review w 1.4, 5.31.5, 5.3 ins 3 and 5 of WH equirements. ty Deactivation Error/Omission "A Safety Assess " WHC-SD-WM-SAD- HAXFORD RELEASE STAMP HAXFORD RELEASE STAMP | 5, oyment ill :1.6, IC-SD- [] [] :ment of 035, UD: |
| REFERENCE WRC-IP-0954, Rev. 5, shall have the following 1. ICO 3.7.1, 3.7.2, 3.7 5.31.7, 5.31.8, 5.31.9, 5. WH-OSR-005, Rev. 0 and the 13a. Justification (mark of Criteria Change [] As-Found [] 13b. Justification Details The IOSR requires revision Rotary Mode Core Sampling Rev. 0 14. Distribution (include Unclassified Docume | "Tank Farms Interim Operations statement added to page "The IOSR's for Rotary of the RMCS in verify implementation .3, 3.7.4, 3.7.5, and A .31.10, 5.31.11, and 5. e Table of Contents (Te one) Design Improvement Facilitate Const In to implement the leve in Flammable Gas Singi name, MSIN, and no. or ent Control, Kel | erational Safety Requires e 2, the Executive Summan / Mode Core Sampling (RMC flammable gas tanks of this IOSR. Administrative Controls 9 .31.12 (SEE ATTACHED PAGE OC) will be modified to r [X] Environmental [] Const. Error// el 1 controls established le Shell Tanks, Hanford S f copies)]i S. Lenz (N4-18) | nents Complian ry. (S) will be ful (S.31.1, 5.31.2 (S) will be addreflect these (] () () () () () () () () () () () () () | ce Implementa ly implementa readiness ass 5.31.3, 5.3 ded to Section additional re Facili Design Assessment Washington, CATE: STA: 4 SEP 0 | ed prior to depl essment/review w 1.4, 5.31.5, 5.3 ins 3 and 5 of WH equirements. ty Deactivation Error/Omission "A Safety Assess " WHC-SD-WM-SAD- RELEASE STAMP HAXFORD RCLEACE 4 1996 | 5, oyment ill (1.6, IC-SD- [] [] |
| PREFERENCE WRC-IP-0954, Rev. 5, shall have the following define the following define the following define the following define the construction (mark of criteria Change [] As-Found [] 13b. Justification Details The IOSR requires revision Rotary Mode Core Sampling Rev. 0 define the following rev. 0 | "Tank Farms Interim Opestatement added to page "The IOSR's for Rotary of the RMCS in verify implementation .3, 3.7.4, 3.7.5, and J. .31.10, 5.31.11, and 5. e Table of Contents (To one) Design Improvement facilitate Const s n to implement the leve in Flammable Gas Sing name, MSIN, and no. of ent Control, Kel | erational Safety Requirer e 2, the Executive Summary Mode Core Sampling (RMC flammable gas tanks of this IOSR. Administrative Controls 5 .31.12 (SEE ATTACHED PAGE OC) will be modified to r [X] Environmental [] Const. Error/4 el 1 controls established le Shell Tanks, Hanford S f copies) 1 i S. Lenz (N4-18) | nents Complian ry. (5) will be ful (5.31.1, 5.31.2 (5) will be addreflect these (1) (1) Omission [1] (2) In the Safet (3) Site, Richland | ce Implementa ly implement readiness ass , 5.31.3, 5.3 ded to Section additional re Facility Design (Assessment Washington, DATE: STA: 4 SEP 0 | RELEASE STAMP | 5, oyment ill 1.6, IC-SD- [] [] ment of 035, ID: |
| REFERENCE WRE-TP-0954, Rev. 5, Shall have the following shall have the | "Tank Farms Interim Opstatement added to page "The IOSR's for Rotary of the RMCS in verify implementation .3, 3.7.4, 3.7.5, and J. .31.10, 5.31.11, and 5. e Table of Contents (To one) Design Improvement Facilitate Const son to implement the leve in Flammable Gas Single name, MSIN, and no. of ent Control, Kel | erational Safety Requires e 2, the Executive Summary Mode Core Sampling (RMC flammable gas tanks of this IOSR. Administrative Controls 9 .31.12 (SEE ATTACHED PAGE OC) will be modified to r [X] Environmental [] Const. Error/M el 1 controls established le Shell Tanks, Hanford S f copies)]i S. Lenz (N4-18) | nents Complian ry. (S) will be ful (S.31.1, 5.31.2 (S) will be addreflect these (() () () () () () () () () () () () | ce Implementa ly implementa readiness ass , 5.31.3, 5.3 ded to Section additional re Facili Design (Assessment Washington, OATE: STA: 4 SEP 0 | ed prior to depl essment/review w 11.4, 5.31.5, 5.3 ins 3 and 5 of WH equirements. ty Deactivation Error/Omission "A Safety Assess " WHC-SD-WM-SAD- HANFORD RELEASE STAMP HANFORD RELEASE STAMP | <pre>, 5, oyment ill :1.6, IC-SD- [] [] </pre> |



| | | 37 NUTU.7 | 1 | | |
|--|--|---|--|--|---|
| | | | Page 2 of | Z No. 609 | 990 |
| 15. Design Verification | 16. Cost Impact | | | 17. Schedule I | mpact (days) |
| Required | ENGINEERING | | CONSTRUCTION | | |
| [] Yes | Additional [] | \$ None Additional | • [] \$ | Improvement | [] None |
| - [X] NO | Savings [] | \$ Savings | [] \$ | Delay | [] |
| 18. Change Impact Rev | view: Indicate the re | lated documents (other | than the engineering | g documents identif | ied on Side 1) |
| SDD/DD . | fl | Seismic/Stress Analysis | | Tank Calibration Ma | nual [] |
| Functional Design Criteria | [] | Stress/Design Report | [] [] | Health Physics Prod | L] edure |
| Operating Specification | נג רז | Interface Control Drawing | | Spares Multiple Uni | LJ it Listing |
| Criticality Specification | LJ TI | Calibration Procedure | נו | Test Procedures/Sp | ecification [1] |
| Conceptual Design Report | LJ 71 | Installation Procedure | [/] | Component Index | LJ [] |
| Equipment Spec. | | Maintenance Procedure | [] | ASME Coded Item | [] [] |
| Const. Spec. | LJ . [1 | Engineering Procedure | | Human Factor Con | ideration [] |
| Procurament Spec. | . LJ . r1 | Operating Instruction | [] | Computer Software | L] (1 |
| vendor Information | LJ [] | Operating Procedure | LJ | Electric Circuit Sch | LJ F1 edule |
| OM Manual | | Operational Safety Require | IMENT [Y] | ICRS Procedure | L] רז |
| SAR/SAR | | IEFD Drawing | | Process Control Ma | nual/Plan r1 |
| Safety Equipment List | LJ r1 | Cell Amangement Drawing | | Process Flow Chart | L] [1 |
| adiation Work Permit | | Essential Material Specific | ation i [] | i I Purchase Requisitio | נן היי |
| invironmental Impact State | LJ ment [] | Fac. Proc. Samp. Schedul | • [] | Tickler File | [] [] |
| Environmental Report | | Inspection Plan | · · · · · | | [] [] |
| ovimomental Permit | . •• • L] 71 | inventory Adjustment Reg | | | L J 7 1 |
| indicate that the Specific Oper | e signing organization ar/Revision aling Procedures ar | has been notified of o nd Calibration Proce | other affected docume Revision dures to, be deterr | ents listed below. Document Numb Mined. | er Revision |
| indicate that the Specific Oper- | e signing organization arr/Revision aling Procedures ar | has been notified of a | other affected docume Revision dures to be detern | Document Numb | er Revision |
| indicate that the Specific Oper | e signing organization an/Revision aning Procedures ar | has been notified of c nd Calibration Proce | Revision dures to be deterr | ents listed below. Document Numb Mined. | er Revision |
| indicate that the Document Number Specific Oper 20. Approvals | e signing organization ar/Revision ang Procedures ar | has been notified of c nd Calibration Proce | Ather affected docume Revision dures to be deterr | ents listed below. Document Numb mined. | er Revision |
| indicate that the Specific Oper 20. Approvals OPERATIONS AND ENGINE | signing organization aling Procedures ar Signature ERING | has been notified of c nd Calibration Proce | other affected docume Revision dures to be detern , , , , , , , , , , , , , , , , , , , | ents listed below. Document Numb | er Revision Date |
| indicate that the Document Numbr Specific Oper 20. Approvals <u>PPERATIONS AND ENGINE</u> Cog. Eng. J. S. Sc | signing organization antherision ating Procedures an Signature <u>ERING</u> chofield for full | has been notified of c nd Calibration Proce Date | ARCHITECT-ENGINE | ents listed below. Document Numb | er Revision Date |
| indicate that the Document Number Specific Oper 10. Approvals IDERATIONS AND ENGINE IOG. Eng. J. S. Sc IOG. Mgr. D. W. Ha | signing organization and Revision aling Procedures an Signature <u>ERING</u> shofield form full milton (2000 m | has been notified of c nd Calibration Proce Date Date | ARCHITECT-ENGINE | ents listed below. Document Numb | er Revision Date |
| indicate that the Document Number Specific Oper 20. Approvals 20. Approvals | signing organization aling Procedures ar Signature <u>ERING</u> shofield form fut milton the future M. L. McDury | has been notified of a nd Calibration Proce Date Jule <u>1/20/86</u> | other affected docume Revision dures to be deterr <u>ARCHITECT-ENGINE</u> PE QA Safety | ents listed below. Document Numb | er Revision Date |
| indicate that the Document Number Specific Oper 20. Approvals 20. Approvals | signature EERING M. L. M. Curry | has been notified of c nd Calibration Proce Date Jul <u>5/1//96</u> <u>5/1/96</u> <u>5/1/96</u> | ARCHITECT-ENGINE PE QA Safety Design | ents listed below. Document Numb | er Revision Date |
| indicate that the Document Numb Specific Oper 20. Approvals 20. Approvals 20 | signature ERING M. L. McDury | has been notified of a nd Calibration Proce Date Jul 1 5/1/96 5/196 5/196 | ARCHITECT-ENGINE | ents listed below. mined. ignature ER | er Revision Date |
| indicate that the Document Numbs Specific Oper 20. Approvals 20. Approvals 2 | signing organization and Procedures an Signature <u>EERING</u> chofield from from M. L. McDury | has been notified of c nd Calibration Proce Date Jul 5/1/96 <u>9/20/86</u> <u>5/1196</u> | ARCHITECT-ENGINE PE QA Safety Design Other | ents listed below. Document Numb | er Revision Date |
| indicate that the Document Numbs Specific Oper 20. Approvals 20. Approvals 2 | signature EERING M. L. M. C. M. | has been notified of c nd Calibration Proce Date Date <u>Full</u> <u>5/1/96</u> <u>5/196</u> | ARCHITECT-ENGINE ARCHITECT-ENGINE PE QA Safety Design Environ. Other | ents listed below. mined. ignature <u>ER</u> | er Revision Date |
| indicate that the Document Numb Specific Oper 20. Approvals DERATIONS AND ENGINE Cog. Eng. J. S. Sc Cog. Mgr. D. W. Ha DA M. L. McElroy Safety M. N. Islam Environ. Other IT Plant Eng. Mgr. WRS Safety Program | signing organization and procedures an Signature ERING thofield from the milton the M. L. McDury J. W. Lentsch Jury | has been notified of c nd Calibration Proce Date 5/1/96 <u>9/20/86</u> 5/196 | Safety Design Denviron. Other | ents listed below. mined. ignature ER | er Revision Date |
| indicate that the Document Numbs Specific Oper 20. Approvals 20. Approvals 2 | signature EERING M. L. M. Lentsch J. E. Truax | has been notified of c nd Calibration Proce Date $\frac{1}{20/36}$ $\frac{5/1/96}{5/196}$ | ARCHITECT-ENGINE PE QA Safety Design Environ. Other | ERGY | er Revision Date |
| indicate that the Document Numbs Specific Oper 20. Approvals 20. Approvals 2 | signature ERING M. Lentsch Jury J. W. Lentsch Jury the RENT Strang J. W. Lentsch Jury K. Truax | has been notified of c nd Calibration Proce Date full $\frac{511'/96}{\frac{7/20/36}{5/196}}$ $\frac{5/196}{\frac{5/2}{96}}$ $\frac{5}{2}$ Dete | ARCHITECT-ENGINE ARCHITECT-ENGINE PE QA Safety Design Environ. Other DEPARTMENT OF EN Signature or a C | ERGY ERGY Autor ERGY Autor | er Revision Date |
| indicate that the Document Number Specific Oper 20. Approvals 20. Approvals | signing organization aling Procedures ar Signature <u>ERING</u> thofield from from M. L. McCurry J. W. Lentsch from L. Truax Mission the fill Worke from 100 M | has been notified of c nd Calibration Proce Date full 5/11/96 7/20/86 5/1196 5/1196 5/1196 5/1196 5/1196 5/1196 5/1196 5/1196 5/1196 5/1196 5/1196 5/1196 5/1196 5/1196 5/1196 | ARCHITECT-ENGINE ARCHITECT-ENGINE PE QA Safety Design Environ. Other DEPARTMENT OF EN Signature or a C tracks the Approv | ERGY Automoter that values and the signature | er Revision Date |
| indicate that the Document Number Specific Oper 20. Approvals 20. Approvals | signature EERING Chofield form for M. L. McCurry J. W. Lentsch for E. Truax for for the full Works for for for the full works for for for the full works for for for for for for the full works for | has been notified of c nd Calibration Proce Date Date $\frac{1}{5/1/96}$ $\frac{5/1/96}{5/196}$ $\frac{1}{5/1/96}$ $\frac{1}{5/1/96}$ $\frac{5/196}{5/2/96}$ J Decknow $\frac{5/2}{96}$ | ADDITIONAL | ERGY Out (terms of the second | er Revision Date |
| indicate that the Document Numbs Specific Oper 20. Approvals DERATIONS AND ENGINE Cog. Eng. J. S. Sc Cog. Mgr. D. W. Ha DA M. L. McElroy M. N. Islam Environ. Dther M. N. Islam Environ. Dther M. Plant Eng. Mgr. (WRS Safety Program IFTP Deputy Dir. J. SEAC R. M. Marusic Auth. Basis J. J. K Design Authority | signature ERING J. W. Lentsch K. J. Blanchard | has been notified of c nd Calibration Proce Date Date $\frac{1}{20/36}$ $\frac{5/1/96}{5/196}$ $\frac{5/2}{96}$ 3-0ecchoos 5/2/96 $\frac{5/2}{96}$ $\frac{5/2}{96}$ | ADDITIONAL | ERGY Out (terms of the second | er Revision Date |
| indicate that the Document Number Specific Oper 20. Approvals 20. Approvals | signing organization and procedures an Signature ERING thofield from from M. Lentsch from L. W. Lentsch from the fill worke from the | has been notified of c nd Calibration Proce Date Date $\frac{1}{196}$ | ADDITIONAL | ERGY Qui Here o ontrol Number that val Signature | er Revision Date |
| indicate that the Document Numbs Specific Oper 20. Approvals 20. Approvals 2 | signature ERING Signature ERING thofield form for M. Lentsch for L. W. Lentsch for R. J. Blanchard for R. J. Blanchard | has been notified of c nd Calibration Proce Date Date $\frac{1}{5/11/96}$ $\frac{5/11/96}{5/1196}$ $\frac{5/11/96}{5/21/96}$ To be chosen $5/12/96$ To be chosen $5/12/96$ | ADDITIONAL | ERGY Poul (Hern a ontrol Number that val Signature | er Revision Date |
| indicate that the Document Numb Specific Oper 20. Approvals 20. Approvals 20 | signing organization and the second | has been notified of c nd Calibration Proce Date Date $\frac{1}{\sqrt{20/36}}$ $\frac{5/1/96}{5/196}$ $\frac{5/2/96}{5/2/96}$ $\frac{5/2/96}{5/2/96}$ $\frac{5/2/96}{5/2/96}$ | ADDITIONAL | ERGY Out (tern a ontrol Number that val Signature ER Mumber that val Signature Komer Kome | er Revision Date |
| indicate that the Document Number Specific Oper 20. Approvals 20. Approvals | signing organization and procedures ar Signature <u>ERING</u> thofield from from M. L. M. Corry J. W. Lentsch E. Truax the RW Works from thos M. M. M. M. | has been notified of c nd Calibration Proce Date Date $\frac{1}{196}$ | ADDITIONAL | ERGY Qui Here o ontrol Number that val Signature | er Revision Date are |

.

٠

.

•

ł

.

•

.

- -



| RELEASE INSTRUCTIONS (RI) RETURN RECEIPT REQUIRED | | | | UMENT NO.: IC-SD-WM- PAGE | ISB-00 1 of | 1. Vol. 1 4 |
|--|----------------------------------|--|---|---------------------------------|----------------|----------------|
| TO: | TITLE: RELEASE N DATE PREP | Hanford Sit Interim Sat Do.: O-K PARED: O7/18 | ce Tank Fa Fety Basis /96 | arm Fac | cilities | |
| I have entered this release into the document instructions. | t per | lf you h contact D. W. H Phone: | ave any ques : Hamilton 372-1130 | stions abou | ut this r | elease |
| VERIFIED RECO 1. REMOVE AND/OR INSERT INDICATED SECTIONS INTO 2. UPDATE THE RELEASE RECORD AT THE FRONT OF THE 2. DIDATE THE RELEASE RECORD AT THE FRONT OF THE | EIPT CO | PY INST | | | | |
| 3. SIGN INTS FORM AND RETORN IT TO D. W. RAMITCO | | REMOVE INSERT | | | | |
| SECTION NO. AND TITLE(S) | PAGES | REV | DATE | PAGES | REV | DATE |
| 6.0 ISB Requirements for Single Shell Tanks | 6-9 | 0-J | 6/7/96 | 6-9 | 0-K | 7/19/96 |
| | | | | | | |
| | | - | | | | |

IMPLEMENTATION NOTICE

6.0, Rev. O-K, "ISB Requirements for Single Shell Tanks"

.

-

This page implements the controls established by A Safety Assessment of Rotary Mode Core Sampling in Flammable Gas Single Shell Tanks.

When document is no longer needed, please recycle/destroy contents and binder. Sign and date the UDC Unclassified Coversheet and return to UDC, A3-95, indicating "DESTROYED".

| | RETUR | N TO | |
|---------------------------|--------------------------|---------|------|
| POSTAL MA | (L: | PLANT N | AAL: |
| Westinghouse Har | Westinghouse Hanford Co. | | |
| P.O. Box 1970/S7-12 | | S7-12 | |
| Richland, WA 99352 | | 372-11 | 30 |
| A-6001-683 (05/96) GEF307 | FOLD TO DOTTED LINE | | |
| | Return To: | \$7-12 | |

| | (1) Document Number | |
|--------------------|--------------------------|--------|
| RECORD OF REVISION | WHC-SD-WM-ISB-001,Vol. 1 | Page 1 |

(2) Title

Hanford Site Tank Farm Facilities Interim Safety Basis

| CHANGE CONTROL RECORD | | | | | |
|---|-----|--|---|---------------------------------------|--|
| (3) Revision | (4) | Description of Change - Replace Add. and Delete Pages | Authorized for Release | | |
| | ļ | | (5) Cog. Engr. | (6) Cog. Mgr. Date | |
| 0-K | (7) | A fourth line added to table to implement rotary mode core sampling requirements | J. S. Schofield J. L. Schofield 7/149 | D. W. Hamilton | |
| | | | | | |
| | | | | <u> </u> | |
| | | | | | |
| | | | | | |
| | | | | | |
| | | | | | |
| | | | | | |
| | | | | | |
| | | | | | |
| | | | | | |
| | | | | | |
| | | | | | |
| | | | | · | |
| | | | | | |
| | | | | | |
| | | | | | |
| | | | | | |
| | | | | | |
| | | | | | |
| | | | | | |
| a se | | | | | |
| | | | | | |
| | | | | | |
| | | | | | |
| | | | | | |
| | | | | | |
| | | | | | |
| | | | | · · · · · · · · · · · · · · · · · · · | |
| | | | | | |
| | | | | | |

Hanford Site Tank Farm Facilities Interim Safety Basis

J. W. Lentsch Westinghouse Hanford Company, Richland, WA 99352 U.S. Department of Energy Contract DE-AC06-87RL10930

| EDT/ | /ECN: | 609990 | UC: | |
|------|-------|--------|--------------|-------|
| Org | Code: | W75230 | Charge Code: | N4HBB |
| B&Ř | Code: | | Total Pages: | 2 |

Key Words: rotary mode, sample truck, flammable gas tank, hydrogen

Abstract: This page implements the controls established by A Safety Assessment of Rotary Mode Core Sampling in Flammable Gas Single Shell Tanks.

TRADEMARK DISCLAIMER. Reference herein to any specific commercial product, process, or service by trade name, trademark, manufacturer, or otherwise, does not necessarily constitute or imply its endorsement, recommendation, or favoring by the United States Government or any agency thereof or its contractors or subcontractors.

Printed in the United States of America. To obtain copies of this document, contact: WHC/BCS Document Control Services, P.O. Box 1970, Mailstop H6-08, Richland WA 99352, Phone (509) 372-2420; Fax (509) 376-4989.

Release Approval

Date

Release Stamp

Approved for Public Release

A-6400-073 (10/95) GEF321



ISB REQUIREMENTS FOR SINGLE-SHELL TANKS

| | | | Page 1 of 1 |
|---|----------------|--------------------|---|
| CONTROL TOPIC(S) | AFFECTED | REQUIREMENT | ISB REQUIREMENT |
| | FACILMES | SOLACE | |
| Interim Operational Safety Requirements | AII SSTs | WHC-SD-WM-OSR-005, | The Single Shell Tank Interim Operational Safety Requirements |
| | | Latest Revision | are adopted in their entirety as ISB Requirements. |
| Overground Transfer Lines | All SSTs | WHC-SD-WM-SAR-034; | The Operational Salety Requirements from Chapter 11.0 of the |
| | | Revision 0-A | "Overground Transfer Line Addendum to the Single-Shell Tank |
| | | | Salety Analysis Report," WHC-SD-WM-SAR-034, Revision O-A |
| | | | are adopted in their entirety as ISB Regulrements. |
| Nuclear Criticality | SSTs | WHC-SD-WM-OSR-005, | The criticality USQ was approved for closure via the |
| | | AC 5.12, Latest | Requirement Source Memo. Refer to WHC-SD-WM-OSR-005, |
| | | Revision | Administrative Control (AC) 5.12 for requirements. |
| | | | |
| | | DOE-HQ Momo, | · |
| | í í | T.P. Grumbly, HQ, | |
| | | to Manager, RL | |
| | | dated March 14, | |
| | · | 1994 | h. |
| Rotary Mode Core Sampling | Flammable Gas | WHC-SD-WM-OSR-005, | Rotary Mode Core Sampling shall be controlled by Sections |
| | Watch List | Latest Revision | 3.7 and 5.31 of WHC-SD-WM-OSR-005, Latest Revision |
| | SSTs specified | WHC-SD-WM-SAD-035, | |
| | in Requirement | Latest Revision | 2 · · · · · · · · · · · · · · · · · · · |
| | Source | | <u> </u> |

١,

المراجع المراجع الع


| | RECORD OF REVISION | (1) Document Num | ber | | |
|--------------|--|-------------------|-------------|---------------------------------------|--|
| | RECORD OF REVISION | WHC-SD-WM-OSR-005 | | Page 1 | |
| (2) Title | | · | | | |
| Single Shel | 1 Tank Interim Operational Safety Requirements | S | | * | |
| | | | | | |
| | | Authori | zed for Rel | ease | |
| (3) Revision | (4) Description of Change - Replace, Add, and Delete Pages | (5) Cog. Engr. | (6) Cog. | Mgr. Date | |
| 0-E | (7) LCO 3.7 and Administrative Controls | J. S. Schofield | D. W., 4 | amilton | |
| | 5.31. Rotary Mode Core Sampling | John 1/18196 | Anton | m> | |
| | · · · | | | | |
| | | | | <u></u> | |
| | | | | | |
| | | | [| | |
| | | | | | |
| | | ļ | ļ | | |
| | | | | | |
| | | | | | |
| | | | | | |
| | | | | | |
| | | | | | |
| | | | | | |
| | | - | | | |
| | | | | | |
| | | | | <u></u> | |
| | | | | | |
| | ······································ | | | | |
| | | | | | |
| | | | | | |
| | ······································ | | | | |
| | | | | · · · · · · · · · · · · · · · · · · · | |
| | | | | | |
| | | | | | |
| | | | | <u></u> | |
| | | | | | |
| | | | | | |
| | | | | | |
| | | | | , | |
| | | | | | |
| | | | | | |
| | | | | | |
| | | | | | |
| | | | ····· | | |
| | | | | <u> </u> | |
| L | | [| | | |
| • | | 1 | | | |

A-7320-005 (08/91) WEF168

.

Single Shell Tank Interim Operational Safety Requirements

J. W. Lentsch Westinghouse Hanford Company, Richland, WA 99352 U.S. Department of Energy Contract DE-AC06-87RL10930

| EDT/ECN: | 609990 | UC: | |
|-----------|--------|--------------|-------|
| Org Code: | W75230 | Charge Code: | N4HBB |
| B&R Code: | | Total Pages: | 53 |

Key Words: rotary mode, sample truck, flammable gas tank, hydrogen

Abstract: These pages implement the controls established by A Safety Assessment of Rotary Mode Core Sampling in Flammable Gas Single Shell Tanks.

TRADEMARK DISCLAIMER. Reference herein to any specific commercial product, process, or service by trade name, trademark, manufacturer, or otherwise, does not necessarily constitute or imply its endorsement, recommendation, or favoring by the United States Government or any agency thereof or its contractors or subcontractors.

Printed in the United States of America. To obtain copies of this document, contact: WKC/BCS Document Control Services, P.O. Box 1970, Mailstop H6-08, Richland WA 99352, Phone (509) 372-2420; Fax (509) 376-4989.

Release Approval

Date

Release Stamp

Approved for Public Release

A-6400-073 (10/95) GEF321

Section 3 Rotary Mode Core Sampling System LCO 3.7

3.7 ROTARY MODE CORE SAMPLING SYSTEM

3.7.1 Flammable Gas Detection System

- LCO 3.7.1 The flammable gas detection system shall be operable with trip setpoints at 5000 ppm hydrogen concentration equivalent and > 100 ppm/s rate of equivalent hydrogen concentration increase over a 10 sec period.
- APPLICABILITY: OPERATION (RMCS waste intrusive operations in Flammable Gas Watch List (FGWL) Tanks or those tanks recommended by the contractor to be included on the FGWL).

ACTIONS

| CONDITION | | | REQUIRED ACTION | COMPLETION TIME |
|-----------|--|-----|--|---|
| Α. | Automatic trip occurs. | A.1 | Stop RMCS activities and evacuate all personnel within a 100-meter radius from the edge of tank. | Immediately |
| | | AND | | |
| | | A.2 | Re-enter within 100 meter radius in accordance with the Tank Farm Health and Safety Plan WHC-SD-WM- HSP-002 | In accordance with the Tank Farm Health and Safety Plan WHC-SD- WM-HSP-002 |
| В. | Flammable Gas Detection System Inoperable | B.1 | Stop RMCS waste intrusive operations | Immediatély |
| | | AND | | i i |
| | · · | B.2 | Restore Flammable Gas Detection System Operability | Prior to resuming RMCS Waste Intrusive Operations |

.

_ .

SURVEILLANCE REQUIREMENTS

| | FREQUENCY | | |
|----|-----------|--|---|
| SR | 3.7.1.1 | Two redundant channels shall be operable. Calibration shall be checked per required frequency and a trip test performed. | 6 months |
| SR | 3.7.1.2 | Hydrogen Detector (Whittaker Cell Sensor) | <u> </u> |
| | | • The system must retain the response time requirement of reaching 90% of full scale in less than 2 min. | Initial setup and every three months thereafter |
| SR | 3.7.1.3 | Flammability Detector (SMC Sensor) | |
| | | • The functional test shall be performed at the specified frequency. The test procedure shall test the shutoff electronics as well as the sensor readings. | Initial setup and daily thereafter |
| | | The system must retain the response time requirement of reaching 90% of full scale in less than 2 min. | monthly |
| | | The sensor must be replaced at the required frequency with a new sensor that has been calibrated. | monthly |

3.7 ROTARY MODE CORE SAMPLING SYSTEM

3.7.2 Tank Gas Pressure Detection System

- LCO 3.7.2 The gas pressure detection system shall be operable and capable of detecting an increase in tank pressure greater than 2 in. w.g. in any 5 min period.
- APPLICABILITY: OPERATION (RMCS waste intrusive operations in Flammable Gas Watch List (FGWL) Tanks or those tanks recommended by the contractor to be included on the FGWL).

Section 3 Rotary Mode Core Sampling System LCO 3.7

| | | ~ • • | - |
|--------|---|----------------|----|
| A (17 | | (NN I | C. |
| - AI | | 1 11/1 | |
| ()) (| ÷ | \sim \cdot | ~ |

| | | | • | LUU 3.7 |
|-------|--|-----|--|---|
| ACTI | ONS | | | |
| ••••• | CONDITION | | REQUIRED ACTION | COMPLETION TIME |
| Α. | Automatic trip occurs. | A.1 | Stop RMCS activities and evaluate all personnel within a 100-m radius from the edge of tank. | Immediately |
| | | AND | | |
| | | A.2 | Re-enter within 100 m radius in accordance with the Tank Farm Health and Safety Plan WHC-SD-WM-HSP-002. | In accordance with the Tank Farm |
| | | AND | | Safety Plan WHC-SD-WM- HSP-002 |
| | | A.3 | A minimum 10 minute wait period shall be imposed following an automatic drill rig shut down due to GRE or other event prior to resuming operations. | > 10 min |
| Β. | Pressure Detection System Inoperable | B.1 | Stop RMCS Waste Intrusive Operations | Immediately |
| | | AND | | |
| | | B.2 | Restore Pressure Detection System Operability | Prior to resuming RMCS waste intrusive operations |

SURVEILLANCE REQUIREMENTS

| | | SURVEILLANCE | FREQUENCY |
|----|---------|--|-----------|
| SR | 3.7.2.1 | Two redundant channels shall be operable. Calibration shall be checked per required frequency and a functional test performed. | 6 months |

Section 3 Rotary Mode Core Sampling System LCO 3.7

- 3.7 ROTARY MODE CORE SAMPLING SYSTEM
- 3.7.3 Exhauster Induced Tank Pressure
- LCO 3.7.3 Exhauster shall be operable and shall maintain tank pressure less than atmospheric pressure and greater or equal to a negative 3 in. w.g.
- APPLICABILITY: OPERATION (one hour prior, during, and sixteen hours following waste intrusive operations in Flammable Gas Watch List (FGWL) Tanks or those tanks recommended by the contractor to be included on the FGWL).

ACTIONS

| | CONDITION | | REQUIRED ACTION | COMPLETION TIME |
|----|--|-----|--|---|
| Α. | Tank pressure exceeds limits. | A.1 | Restore pressure to acceptable limits. If pressure cannot be restored, shut down drill rig engine. | Immediately |
| Β. | Exhauster shutdown due to automatic | B.1 | Cease RMCS operations. | Immediately |
| L | trip. | ANU | | |
| | | B.2 | A 10 min minimum waiting period is required prior to resuming operations. | > 10 min |
| С. | Exhauster inoperable. | C.1 | Stop RMCS Waste Intrusive Operations. | Immediately |
| | | AND | | |
| | | C.2 | Restore Exhauster operability. | Prior to resuming RMCS waste intrusive operations |

SURVEILLANCE REQUIREMENTS

| | | SURVEILLANCE | FREQUENCY |
|----|---------|---|-----------|
| SR | 3.7.3.1 | The exhauster pressure switch shall be calibrated periodically. | 6 months |

Section 3 Rotary Mode Core Sampling System LCO 3.7

| | | • | LCO 5.7 |
|------|---------------|---|---|
| SURV | EILLANCE REQU | JIREMENTS | |
| | | SURVEILLANCE | FREQUENCY |
| SR | 3.7.3.2 | Prior to initiating RMCS operations and periodically during exhauster operations, the pressure shall be verified to be in limits. | Prior to initiating operations and then once every 24 hours. |
| SR | 3.7.3.3 | All exhauster shutdown indication elements shall be calibrated with independent verification and tested with indication of all failures. | 6 months |

.

Section 3 Rotary Mode Core Sampling System LCO 3.7

3.7 ROTARY MODE CORE SAMPLING SYSTEM

3.7.4 <u>Nitrogen Purge System</u>

- LCO 3.7.4 The Nitrogen Purge System shall be operable and able to:
 - a. supply the drill string at a rate of \geq 30 scfm.
 - b. supply nitrogen to the drill string at a temperature \searrow > 10°F and < 140°F.
- APPLICABILITY: OPERATION (RMCS waste intrusive operations in Flammable Gas Watch List (FGWL) Tanks or those tanks recommended by the contractor to be included on the FGWL).

ACTIONS

| | CONDITION | | REQUIRED ACTION | COMPLETION TIME |
|----|--------------------------------------|------------|---|---|
| A. | Automatic trip | A.1 | Cease RMCS operations. | Immediately |
| | occurs. | AND | | |
| | | A.2 | A 10 min minimum waiting period is required prior to resuming operations. | > 10 min |
| Β. | Nitrogen temperature | B.1 | Stop drill rig engine. | Immediately |
| | | AND | | |
| | | B.2 | Continue nitrogen purge. | Immediatoly |
| | | <u>AND</u> | | Immediately |
| | | B.3 | Do not resume drilling until temperature is within normal range. | > 10 min |
| C. | Nitrogen purge system inoperable. | C.1 | Stop RMCS waste intrusive operations. | Immediately |
| | | AND | | |
| | | C.2 | Restore Nitrogen Purge System Operability | Prior to resuming RMCS operating |

3-29

| | | WHC-S | SST SD-WM-OSR-005 Rev. 0-E Core S | Section 3 Rotary Mode ampling System LCO 3.7 |
|----|---|-------|---|---|
| D. | Nitrogen temperature monitor inoperable. | D.1 | Stop RMCS waste intrusive operations. | Immediately |
| | | AND | | |
| | | D.2 | Restore Nitrogen temperature monitor operability. | Prior to resuming RMCS operating |

.

Section 3 Rotary Mode Core Sampling System LCO 3.7

SURVEILLANCE REQUIREMENTS

| | | SURVEILLANCE | FREQUENCY |
|----|---------|--|-----------|
| SR | 3.7.4.1 | The purge system shall be tested for bypass leakage periodically. Testing shall be independently verified with indication of failures. Leak rate shall be limited to the uncertainty of the system or less than 2% of the required flow. | 6 months |
| SR | 3.7.4.2 | Flow monitoring and automatic shutdown system will be calibrated periodically and verified as capable of automatically sending a shutdown signal to the drill rig engine immediately upon receipt of a valid shutdown signal of detecting nitrogen flow less than the required flow (2 of 3 control channels). | 6 months |
| SR | 3.7.4.3 | Temperature indicator and alarm shall be calibrated periodically. | 6 months |

Section 3 Rotary Mode Core Sampling System LCO 3.7

3.7 ROTARY MODE CORE SAMPLING SYSTEM

3.7.5 <u>Rotary Drilling Parameters</u>

- LCO 3.7.5 The RMCS equipment shall be operable and shall:
 - a. not be operated with a down force on the drill bit > 750 lbf.
 - b. not operate at a drill string rotation > 55 rpm.
 - c. not be operated when the penetration rate is <0.75 in/min for a cumulative time of 60 sec in any 3 min period.
- APPLICABILITY: OPERATION (RMCS waste intrusive operations in Flammable Gas Watch List (FGWL) Tanks or those tanks recommended by the contractor to be included on the FGWL).

ACTIONS

| CONDITION | | REQUIRED ACTION | | COMPLETION TIME |
|-----------|---|-------------------|--|--|
| Α. | Automatic trip. | A.1 | A minimum 10 min waiting period shall be imposed following an automatic trip due to down force. rpm, or penetration rate prior to resumption of RMCS operations. | > 10 min |
| Β. | Walkdown Function, hydraulic bottom detector, Down Force. Rotary RPM Measurement or | B.1 <u>AND</u> | Stop RMCS waste intrusive operations. | Immediately |
| | penetration rate system inoperable. | B.2 | Restore system operability. | Prior to resuming RMCS operations |
| С. | Grapple load exceeds 250 lb. | C.1 | Stop electric motor driving the grapple hoist (or verify automatic stop). | Immediately |

.

Section 3 Rotary Mode Core Sampling System LCO 3.7 _ .

SURVEILLANCE REQUIREMENTS

| | | SURVEILLANCE | FREQUENCY |
|----|---------|---|-----------|
| SR | 3.7.5.1 | RPM measurement and alarm trip equipment shall be calibrated and functionally tested on the stated frequency. | 6 months |
| | | There shall be dual RPM measurement sensors, both of which shall be operable when rotating the drill bit below the waste surface. The drill string rotation and penetration shall cease upon alarm or failure of either sensor circuits. | |
| | | Penetration rate measurement system and alarm trip equipment shall be calibrated and functionally tested on the stated frequency. | 6 months |
| | | Down force measurement system and alarm trip equipment shall be calibrated and functionally tested on the stated frequency. | 6 months |
| | | Down force and alarm trip equipment shall be operable when rotating the drill bit below the waste surface. | |
| | | Upon a valid high RPM, high penetration rate or high down force alarm, the system shall be capable of ceasing drill string rotation and penetration immediately. | |
| | | The walkdown function and hydraulic bottom detector shall be verified operational. | 6 months |
| | | The walkdown function shall be operable for all samples except the last, when the hydraulic bottom detector shall be used. | |
| | | Grapple hoist cable shall be inspected periodically. | 6 months |
| | | Grapple load cell shall be calibrated. | 6 months |

Bases Rotary Mode Core Sampling System B 3.7

B 3.7 ROTARY MODE CORE SAMPLING SYSTEM

B 3.7.1 Flammable Gas Detection System

BASES BACKGROUND The flammable gas detection system is connected between the riser and exhauster housing with local system readouts and alarmed setpoints. During drilling exceeding the setpoints will trip the drill rig engine and close the nitrogen purge flow solenoid operated valve. APPLICABLE Analysis performed in Section 4 and Appendix B of SAFETY ANALYSES Reference 1 to allow safe operation of the RMCS sets two limits to protect against a GRE and possible fire or explosion. This mitigates unacceptable consequences resulting from a dome collapse. LCO A trip setpoint of 5000 ppm hydrogen concentration equivalent assures that the tank in guestion does not exceed 25% of Lower Flammability Limit (LFL) during a GRE. An addition rate trip of 100 ppm hydrogen concentration equivalent increase over a 10 sec. period protects against exceeding the 25% LFL and gives sufficient notice of an impending GRE to allow proper precautions to be taken. APPLICABILITY OPERATION (RMCS waste intrusive operations in Flammable Gas Tanks).

Bases Rotary Mode Core Sampling System B 3.7

| BASES | |
|---------------|---|
| ACTIONS | <u>A.1</u> |
| | These actions were established in Reference 1 as the minimum personnel requirements to protect plant personnel < upon exceeding the setpoints established. |
| | <u>A.2</u> |
| | Re-entry into the 100 meter will be in accordance with the Tank Farm Health and Safety Plan, WHC-SD-WM-HSP-002. |
| | <u>B.1</u> |
| | RMCS waste intrusive operations are not allowed without an operable Flammable Gas Detection System. |
| | <u>B.2</u> |
| | Flammable Gas Detection System operability must be restored prior to resuming RMCS waste intrusive operations. |
| | <u>SR 3.7.1.1</u> |
| REQUIREMENTS | A 6-month trip test and calibration of the redundant trip circuitry provides assurance that system performance has not degraded. |
| SURVEILLANCE | <u>SR 3.7.1.2</u> |
| REQUIREMENTS | Initial and 3 month check required by Reference 1 for Whittaker Cell. |
| | The calibration should be performed at initial setup at each location and every 3 months after that point. The calibration should consist of setting the zero with pure air or nitrogen and calibration with 6000 ppm hydrogen. The calibration procedure with 6000 ppm hydrogen should test the shutoff electronics as well as the sensor reading. |
| SURVEILLANCE | <u>SR 3.7.1.3</u> |
| KEQUIKEREN IS | Initial and daily check required by Reference 1 for SMC Sensor. |

Bases Rotary Mode Core Sampling System B 3.7

| BASES | | | |
|------------|----|--|----|
| REFERENCES | 1. | LANL, 1996, A Safety Assessment of Rotary Mode Co Sampling in Flammable Gas Single Shell Tanks, Hanford Site, Washington, WHC-SD-WM-SAD-035, Rev. O-a, Los Alamos National Laboratory, Los Alamos, New Mexico. | re |

Bases Rotary Mode Core Sampling System B 3.7

B 3.7 ROTARY MODE CORE SAMPLING SYSTEM

B 3.7.2 Tank Gas Pressure Detection System

| BASES | · · · · · · · · · · · · · · · · · · · |
|-------------------------------|--|
| BACKGROUND | The purpose of this system is to prevent burn in and out of dome by detecting gas release rates > $1000 \text{ ft}^3/\text{min}$. |
| | It also provides protection for fire and toxic hazards, reduces the likelihood of spark in flammable gas atmosphere, and provides protection for exposure to toxic gas releases. |
| | This mitigates unacceptable offsite consequences resulting from a dome_collapse. |
| APPLICABLE SAFETY ANALYSES | Safety analyses were performed in Sections 4. 4.2. and Appendix B of Reference 1. |
| LCO | A pressure pulse of 50.8 mm (2 in.) w.g. would correspond to a 0.5% increase in the dome pressure. Using a dome volume of 1416 m ³ (50,000 ft ³), the ideal gas law and adiabatic compression, a 5.1 m ³ (180 ft ³) sudden release into the dome is sufficient to generate a 2-in. w.g. pressure pulse. The purpose of the 2" w.g. trip is to protect against a release rate of > 1000 ft ³ /min. |
| APPLICABILITY | OPERATION (RMCS waste intrusive operations in Flammable Gas Tanks). |

Bases Rotary Mode Core Sampling System B 3.7

| BASES | |
|--------------|--|
| ACTIONS | <u>A.1</u> |
| | These actions were established in Reference 1 as the minimum personnel requirements to protect plant personnel \smallsetminus upon exceeding the pressure setpoint. |
| | <u>A.2</u> |
| | Re-entry into the 100 meter will be in accordance with approved Tank Farm procedures. |
| | <u>A.3</u> |
| | The 10 min wait allows sufficient time for the drill bit and associated waste to adequately cool down. |
| | <u>B.1</u> |
| | Waste intrusive operation are not allowed without an operable pressure detection system. |
| | <u>B.2</u> |
| | Pressure detection system operability must be restored prior to resuming RMCS waste intrusive operations. |
| SURVEILLANCE | <u>SR 3.7.2.1</u> |
| | A 6-month trip test and calibration is necessary to ensure that the equipment will perform as required by the safety assessment. A channel is considered to be the sensor and associated circuitry and components up to the input to the alarm and trip circuit. |
| REFERENCES | LANL, 1996. A Safety Assessment of Rotary Mode Core Sampling in Flammable Gas Single Shell Tanks, Hanford Site, Washington, WHC-SD-WM-SAD-035, Rev. O-a. Los Alamos National Laboratory, Los Alamos, New Mexico. |

Bases Rotary Mode Core Sampling System B 3.7

B 3.7 ROTARY MODE CORE SAMPLING

B 3.7.3 Exhauster Induced Tank Pressure

| BASES | |
|-------------------------------|--|
| BACKGROUND | Excessive vacuum in the dome is another accident identified in Reference 1 as a result of exhauster operations. The exhauster design prevents the occurrence of excessive vacuum because the shutoff head is -14 in. w.g. A dome collapse would not occur until -15 in. w.g. Also, the inlet high-efficiency particulate air (HEPA) filter has a vacuum breaker to mitigate excessive negative tank pressure. The vacuum breaker is set at about -4 in. w.g. |
| APPLICABLE SAFETY ANALYSES | The analysis was performed in Section 4.6.1 of Reference 1. |
| LCO | The range established by the LCO assures that tank pressure is less than atmospheric and greater than that which would open the vacuum breaker or collapse the tank dome. |
| APPLICABILITY | OPERATION (RMCS waste intrusive operations in Flammable Gas Tanks). |
| | |

Bases Rotary Mode Core Sampling System B 3.7

| BASES | |
|------------------------------|---|
| ACTIONS | <u>A.1</u> |
| | Pressure change during exhauster operation is relatively slow. In any case cessation of RMCS operations will restore a serious negative pressure condition. |
| | <u>B.1</u> |
| | RMCS operations are not allowed without the exhauster in operation. |
| | <u>B.2</u> |
| | The minimum 10 min waiting period upon an exhauster automatic trip is required to ensure adequate cool down time for the drill bit and associated waste. |
| | <u>C.1</u> |
| | RMCS waste intrusive operations are not allowed without an operable exhauster. |
| | <u>C.2</u> |
| | Exhauster operability must be restored prior to resuming RMCS waste intrusive operations. |
| SURVEILLANCE | <u>SR 3.7.3.1</u> |
| | A 6-month system test and calibration will assure that the exhauster is performing as designed and meeting the requirements assumed in the safety assessment. |
| SURVEILLANCE REQUIREMENTS | <u>SR_3.7.3.2</u> |
| | The pressure switch requires periodic surveillance to ensure that pressure requirements are met. |
| | <u>SR 3.7.3.3</u> |
| | A test of all exhauster shutdown system elements ensures system reliability. |

Bases Rotary Mode Core Sampling System B 3.7

B 3.7 ROTARY MODE CORE SAMPLING SYSTEM

B 3.7.4 <u>Nitrogen Purge System</u>

BASES

| BACKGROUND | Nitrogen is supplied for five different functions during RMCS operations: (1) the Drill String purge gas system used during RMCS drilling; (2) the purge through the riser sleeve annulus, (3) the hydrostatic head in the drill string and (4) in the shielded receiver, and (5) the Z- purge (NFPA 496) in the SR weather cover. The systems provide: (1) drill bit cooling and cleaning during rotary drilling. (2) help prevent waste flooding in the drill string, and (3) prevent gas accumulation. |
|-------------------------------|---|
| APPLICABLE SAFETY ANALYSES | Through use of the purge system, drill bit overheating and waste ignition is prevented. Control prevents local exothermic chemical reactions as well as a possible ignition of flammable gas in the waste. Analyses to establish this LCO were performed in Sections 4.4.1.4.4.4. and 4.6.3 of Reference 1. |

Bases Rotary Mode Core Sampling System B 3.7

| LCOEnvelope testing determined that 30 scfm was the minimum flow which. in combination with other parameters. would provide adequate cooling when coupled with a limit on nitrogen inlet temperature.The two-second requirement for a trip to occur is based on the safety assessment assumption that the determination of a valid alarm signal requires approximately 2 seconds.Immediately is used as a special completion time. In this case. the Required Action is to be commenced without delay and continuously pursued in a controlled manner until complete.While the minimum purge flow must be greater than 30 scfm: however, it is possible that necessary cooling to the drill bit would not be provided if there were a leak from the nitrogen purge system between the flow measurement location and the drill bit. WHC determined that the leak from the truck is within the uncertainty range of instrumentation. As indicated in Table 4.12 of Reference 1. the leak rate from the nitrogen system must be measured once every 6 months. This control requires that the leak rate must be within the uncertainty range of instrumentation or < 2% of the nominal flow.APPLICABILITYOPERATION (RMCS waste intrusive operations in Flammable Gas Tanks). | BASES | |
|---|---------------|---|
| The two-second requirement for a trip to occur is based on the safety assessment assumption that the determination of a valid alarm signal requires approximately 2 seconds.Immediately is used as a special completion time. In this case, the Required Action is to be commenced without delay and continuously pursued in a controlled manner until complete.While the minimum purge flow must be greater than 30 scfm: however, it is possible that necessary cooling to the drill bit would not be provided if there were a leak from the nitrogen purge system between the flow measurement location and the drill bit. WHC determined that the leak from the truck is within the uncertainty range of instrumentation. As indicated in Table 4.12 of Reference 1, the leak rate from the nitrogen system must be measured once every 6 months. This control requires that the leak rate must be within the uncertainty range of instrumentation or < 2% of the nominal flow. | LCO | Envelope testing determined that 30 scfm was the minimum flow which. in combination with other parameters. would provide adequate cooling when coupled with a limit on nitrogen inlet temperature. |
| Immediately is used as a special completion time. In this case, the Required Action is to be commenced without delay and continuously pursued in a controlled manner until complete.While the minimum purge flow must be greater than 30 scfm: however, it is possible that necessary cooling to the drill bit would not be provided if there were a leak from the nitrogen purge system between the flow measurement location and the drill bit. WHC determined that the leak from the truck is within the uncertainty range of instrumentation. As indicated in Table 4.12 of Reference 1, the leak rate from the nitrogen system must be measured once every 6 months. This control requires that the leak rate must be within the uncertainty range of instrumentation or < 2% of the nominal flow. | | The two-second requirement for a trip to occur is based on the safety assessment assumption that the determination of a valid alarm signal requires approximately 2 seconds. |
| While the minimum purge flow must be greater than 30 scfm: however, it is possible that necessary cooling to the drill bit would not be provided if there were a leak from the nitrogen purge system between the flow measurement location and the drill bit. WHC determined that the leak from the truck is within the uncertainty range of instrumentation. As indicated in Table 4.12 of Reference | | Immediately is used as a special completion time. In this case, the Required Action is to be commenced without delay and continuously pursued in a controlled manner until complete. |
| APPLICABILITY OPERATION (RMCS waste intrusive operations in Flammable Gas Tanks). | | While the minimum purge flow must be greater than 30 scfm; however, it is possible that necessary cooling to the drill bit would not be provided if there were a leak from the nitrogen purge system between the flow measurement location and the drill bit. WHC determined that the leak from the truck is within the uncertainty range of instrumentation. As indicated in Table 4.12 of Reference 1, the leak rate from the nitrogen system must be measured once every 6 months. This control requires that the leak rate must be within the uncertainty range of instrumentation or < 2% of the nominal flow. |
| | APPLICABILITY | OPERATION (RMCS waste intrusive operations in Flammable Gas Tanks). |

Bases Rotary Mode Core Sampling System B 3.7

BASES

ACTIONS

RMCS operations are not allowed without an Nitrogen Purge System in operation.

<u>A.2</u>

A.1

Provided that the drill string rotation is not re-started for a minimum period of 10 min, it is possible to resume RMCS operations. This is the minimum time to allow drill bit and waste cooling.

<u>B.1</u>

RMCS operations are not allowed with the Nitrogen temperature out of range.

<u>B.2</u>

A continued purge is required to assist drill bit cool down.

<u>B.3</u>

A temperature outside the bounds of the safety assessment must be corrected prior to resuming operations.

<u>C.1</u>

RMCS waste intrusive operations are not allowed without an operable nitrogen purge system.

<u>C.2</u>

Nitrogen purge system must be restored prior to resuming RMCS waste intrusive operations.

<u>D.1</u>

RMCS waste intrusive operations are not allowed without an operable nitrogen temperature monitor.

<u>D.2</u>

Nitrogen temperature monitor must be restored prior to resuming RMCS operations.

Bases Rotary Mode Core Sampling System B 3.7

| BASES | |
|--------------|---|
| SURVEILLANCE | <u>SR 3.7.4.1</u> |
| | The 6-month testing and calibration requirement was established in Reference 1 to ensure equipment reliability and operability. |
| SURVEILLANCE | <u>SR 3.7.4.2</u> |
| | The 6-month testing and calibration requirement was established in Reference 1 to ensure equipment reliability and operability. A channel is considered to be the sensor and associated circuitry and components up to the input to the alarm and trip circuit. Two of the three channels must alarm in order for the trip to occur. If one channel is taken out of service then it shall be tripped and one more channel alarm will cause a trip condition. |
| SURVEILLANCE | <u>SR 3.7.4.3</u> |
| | Temperature sensor calibration and alarm test assures proper operator response to out of normal condition. |
| REFERENCES | LANL, 1996. A Safety Assessment of Rotary Mode Core Sampling in Flammable Gas Single Shell Tanks. Hanford Site, Washington, WHC-SD-WM-SAD-035, Rev. 0-a, Los Alamos National Laboratory, Los Alamos, New Mexico. |

.

Bases Rotary Mode Core Sampling System B 3.7 17

B 3.7 ROTARY MODE CORE SAMPLING SYSTEM

B 3.7.5 Rotary Drilling Parameters

BASES

| BACKGROUND | Behavior of the drill string and, in particular the drill bit, was a high focus area in the safety assessment (Reference 1). Many accident scenarios were analyzed to determine performance requirements. All the requirements established were to limit the heat generating capacity of the drilling operation. |
|-------------------------------|--|
| APPLICABLE SAFETY ANALYSES | Wastes including mixtures of sodium nitrate and sodium nitrite with organic compounds can produce violent exothermic reactions (Appendix G Reference 1). Increasing the temperature of the waste in the vicinity of the drill bit can cause a thermal runaway. There are several hazards that are associated with a local thermal runaway. and they are discussed in Appendix G of Reference 1. Two major important hazards are the ignition of the flammable gas and the ignition of a self-propagating exothermic reaction in the waste. Reactions in mixtures containing relatively small amounts of organic compounds can result in temperatures greater than the autoignition temperature of hydrogen mixtures, so the ignition of flammable gases is the more limiting condition. |
| | Because the possibility of flammable-gas mixture cannot be eliminated. the approach used is to take all practical measures to eliminate ignition sources. A local runaway reaction is a potential ignition source, so the requirement that there be no local runaway reaction is consistent with the philosophy used. Preventing a local thermal runaway is also protection against a propagating exothermic reaction, and it eliminates the possibility of generating additional flammable gas as a result of elevated temperatures. Appendix G of Reference 1 discusses runaway reactions and waste ignition in great detail. Basic conclusions of Appendix G of Reference 1 are that local runaway reactions can be prevented by establishing waste temperature limits. The following temperature limits are established: |
| | The temperature of small waste fragments produced at the drill tip must not exceed 180°C. |
| | The temperature of the drill bit and the average temperature of the waste affected by drilling must not exceed 160°C for more than 10 min. |

.

Bases Rotary Mode Core Sampling System B 3.7

| BASES | |
|--|---|
| APPLICABLE SAFETY ANALYSES (continued) | New envelope testing has been performed by WHC to determine the operating parameters, rotational speed, down forces, and nitrogen purge flow to comply with the safety criteria given above. Appendix F of Reference 1 discusses the details of testing and the results obtained. |
| | As a summary, results of envelope testing and their analysis showed that the drill bit surface temperature and the waste substrate temperature can be kept below 160°C, including an uncertainty of 10°C, if the following limits are applied: Down force < 750 lbf, rotational speed < 55 rpm, minimum nitrogen flow > 30 scfm, and penetration rate > 0.75 in./min. The chip temperatures under these conditions are also limited to 180°C as required. If a trip is initiated when one of the set points for these four parameters is exceeded, drilling must be stopped. After a shutdown there must be a waiting period of 10 min before drilling can continue. The waiting period of 10 min is based on the experimentally determined cooling time. The testing and the analysis included plugged holes on the drill bit. |
| | The drill bit shall be replaced if drilling is shut down four times consecutively as a result of low penetration rate and if the cumulative penetration is < 0.3 in. for the last three attempts. |
| LCO | The requirements of 3.7.5 (a), (b), and (c) were established by envelope testing documented in Reference 1. The time requirement in the action statement is based on the approximate time it takes to establish a valid trip signal determination. |
| APPLICABILITY | OPERATION (RMCS waste intrusive operations in Flammable Gas Tanks). |

B-64-M

•

Bases Rotary Mode Core Sampling System B 3.7

.

| BASES | | | |
|------------------------------|--|--|--|
| ACTIONS | <u>A.1</u> | | |
| | The minimum 10 min waiting period is to provide adequate drill bit and associated waste cool down. | | |
| | <u>B.1</u> | | |
| | RMCS waste intrusive operations are not allowed without an operable Down Force. Rotary RPM Measurement, penetration rate system, walkdown function, and hydraulic bottom detector. | | |
| | <u>B.2</u> | | |
| | Operability must be restored prior to resuming RMCS waste intrusive operations. | | |
| | <u>C.1</u> | | |
| | Action statement prevents load exceeding those specified in Reference 1. | | |
| SURVEILLANCE REQUIREMENTS | <u>SR 3.7.5.1</u> | | |
| | A 6-month test and calibration was required by Reference 1 to ensure system reliability and performance. | | |
| | Controlled loads on the grapple hoist prevent grapple drop consequences and potential for fire (Sections 4.3.2 and 4.3.6 of Reference 1). | | |
| REFERENCES | LANL, 1996, A Safety Assessment of Rotary Mode Core Sampling in Flammable Gas Single Shell Tanks. Hanford Site, Washington, WHC-SD-WM-SAD-035. Rev. 0-a, Los Alamos National Laboratory, Los Alamos, New Mexico. | | |

Section 5 Rotary Mode Core Sampling AC 5.31

Section 5 ADMINISTRATIVE CONTROLS (ACs)

5.31 ROTARY MODE CORE SAMPLING

5.31.1 Control Applicability

For single-shell tanks (SSTs) on the flammable Gas Watch List (FGWL), or those tanks recommended to be on the watch list, an evaluation checklist shall be completed per Section 7.0 of Reference 1.

Checklist items will include:

- (1) Tank Specific Hazards / Other Watch lists
- (2) Flammable Gas Composition
- (3) Toxic Gas Composition
- (4) Waste Temperature
- (5) Waste Energetics
- (6) Likelihood of Gas Release Event (GRE)

Successful completion of the checklist is required for RMCS operations to proceed. This means that no adverse items remain unresolved.

5.31.2 Ignition and Envelope Testing

Ignition and envelope test requirements and acceptance criteria shall be used to verify drill bit and material performance. Required testing has been completed.

5.31.3 Flammable Gas Detection System Verification

Functional requirements and performance acceptance criteria provided in Reference 1 shall be used to verify the performance of the flammable gas detection system. This is a one time check for the Whittaker Cell and the SMC sensor and need not be repeated. Should a new sensor be selected for use. its performance shall be verified against Appendix U of Reference 1.

5.31.4 Tank Dome Activities

Controls shall be in place that prevent other activities on a specific tank during RMCS waste intrusive activities.

These controls will allow simultaneous dome-intrusive activities if:

- the equipment is qualified for operation in a Class I, Division 1, Group B environment, and,
- operation is based on its own safety assessment, and,
- operation does not physically interact with the drill string, and is not waste-intrusive.

5.31.5 Gas Leak Paths

A formal tank walkdown procedure shall be developed and implemented prior to waste-intrusive activities that:

- assesses the general condition of risers,
- identifies observable leaks,
- as a minimum, documents identified leaks ≥ 1 in. (equivalent diameter), and,
- as a minimum, seals or adds deflectors to identified leaks with an equivalent leak diameter ≥ 1 in. (equivalent diameter).

5.31.6 Open Riser Exclusion Zone

An exclusion zone shall be established with a radius of 36-riserdiameters around any open riser during waste-intrusive RMCS activities.

5.31.7 Energized Equipment in the Dome and Open Riser Exclusion Zones

During waste-intrusive operations, all energized equipment exposed to the tank dome vapor space being sampled (as defined in approved contractor safety documentation) shall be rated for operations in Class I. Division 1, Group B environment or Class I, Division 2, Group B environment with automatic shut down for flammable gas concentrations ≥ 25% LFL.

All existing energized equipment not meeting the above control shall be de-energized.

This also applies to open riser exclusion zones (exclusion zone is defined as an area with a radius of 36-riser-diameters around open risers during waste-intrusive activities).

5.31.8 Tank Loading

Loading on each tank shall comply with IOSR requirements for simultaneous static and dynamic loading for each specific FG/RMCS tank.

5.31.9 Truck Position

A procedure shall be developed and implemented that, prior to wasteintrusive operations:

- prevents positioning the sampling truck over an open riser, and
- seals any risers under the truck, and

Section 5 Rotary Mode Core Sampling AC 5.31

• raises the truck a minimum of 36 in. between any potential ignition source on the truck and the top of any riser or pit over which the truck is positioned.

5.31.10 Portable Inlet Stack

Prior to waste-intrusive activities, it shall be verified the inlet breather filter effluent (in the event of tank pressurization) is directed vertically to a height of at least 15 ft. above ground level.

5.31.11 Drill Bit, Core Barrel, and Drill Rods

The FG/RMCS drill bit, core barrel, and drill rods shall be of the configuration and material tested by the Bureau of Mines and performance evaluated in Reference 1.

5.31.12 RMCS Operations

It shall be verified that the exhauster is fully operational:

- 1 hr before the nitrogen purge flow to the Drill String is established \geq 30 scfm.
- the flammable gas concentration in the tank vapor space shall be <1000 ppm before starting RMCS operations to obtain the initial segment in any core in a given riser.
- during all rotary drilling operations. and
- for a cumulative 16 hrs following termination of nitrogen purge flow to the Drill String.

When rotary drilling operations are resumed within any 16-hr waiting period. a new 16-hr period will be initiated following termination of Drill String nitrogen purge flow of \geq 30 scfm.



Bases RMCS B 5.31

B 5.0 ADMINISTRATIVE CONTROLS (ACs) - GENERAL

B 5.31 Rotary Mode Core Sampling

BASES

| BACKGROUND | Reference 1 addresses each of the required elements associated with the installation, operation, and removal of a rotary-mode core sampling (RMCS) device in flammable- gas single-shell tanks (SSTs). The RMCS operations are needed in order to retrieve waste samples from SSTs with hard layers of waste for which push-mode sampling is not adequate for sampling. | | | |
|-------------------------------|---|---|--|--|
| | In Reference proposed ac systematica could resul were identi assessed. changes req hazards wer | e 1, potenti tion were id lly. Severa t in radiolc fied and ana Administrati uired to eli e identified | al hazards associated with the dentified and evaluated al potential accident cases that ogical or toxicological gas releases alyzed and their consequences ve controls, procedures and design minate or reduce the potential of a | |
| APPLICABLE SAFETY ANALYSES | Results of have been i analyses ar | the safety a ncorporated e as follows | issessment performed in Reference 1 into this document. Specific | |
| | (1) (2) (3) | B 5.31.1 B 5.31.2 B 5.31.3 | Control Applicability Ignition and Envelope Testing Flammable Gas Detection System | |
| | (4) (5) (6) (7) | B 5.31.4 B 5.31.5 B 5.31.6 B 5.31.7 | Tank Dome Activities Gas Leak Paths Open Riser Exclusion Zone Energized Equipment in the Dome and Open Riser Exclusion Zones | |
| | (8) (9) (10) (11) | B 5.31.8 B 5.31.9 B 5.31.10 B 5.31.11 | Tank Loading Truck Position Portable Inlet Stack Drill Bit, Core Barrel, and Drill Rods | |
| | (12) | B 5.31.12 | RMCS Operations | |

Bases RMCS B 5.31

B 5.0 ADMINISTRATIVE CONTROLS (ACs) - GENERAL

B 5.31.1 Control Applicability

BASES

| BACKGROUND | The Safety Assessment of Rotary Mode Core Sampling in Flammable Gas Single Shell Tanks: Hanford Site, Richland, Washington, WHC-SD-WM-SAD-035, Rev. O-a, June 7, 1996, hereafter designated as Reference 1 was developed using bounding assumptions. However, these assumptions are not verified against each one of the existing or potential flammable gas tanks (see Section 1 of Reference 1). Thereafter the following applies to RMCS waste intrusive operations in flammable gas tanks (FGT's) |
|------------|---|
| | Operacions in rialinable gas carks (rai s). |

11.
| Ŀ | Bases. | |
|---|--------|--|
| | RMCS | |
| В | 5.31 | |

BASES

APPLICABLE The following assumptions must be verified before RMCS SAFETY ANALYSES waste intrusive operations in any FGT.

CHECKLIST ITEMS

Tank Specific Hazards / Other Watch lists

If a given tank has a specific hazard or accident initiator that is not analyzed in Sections 3 and 4 of Reference 1, the analysis must be supplemented to cover the tank specific conditions. For instance, Reference 1 does not address ferrocyanide issues even though some of the FGTs may also be on the ferrocyanide watch list. This checklist item is especially important for tanks that are on multiple watch lists (in addition to flammable gas watch list.

Flammable Gas Composition

Reference 1 assumes that 25% of the LFL is greater than 5000 ppm hydrogen based on the analysis provided in Appendix C of Reference 1. The only flammable gas species considered are hydrogen and ammonia with small amounts of methane. If new information (information that is not cited in Appendix C of Reference 1) reveals that, for a given tank, there are other flammable gas species and/or the assumed value of the LFL is not conservative, the analysis in Appendix C must be revised to incorporate the new data.

Toxic Gas Composition

For toxic effects, the gas composition in a given GRE is assumed to be 60% ammonia or 75% nitrous oxide. If any evidence before the FG/RMCS operation exists to indicate that these values (especially the ammonia fraction) may be exceeded in one of the SSTs as a result of new analysis or data, or if they are not conservative, the consequence that these values (especially the ammonia fraction) may be exceeded in one of the SSTs as a result of new analysis or data, or if they are not conservative, the consequence that or if they are not conservative, the consequence analysis must be re-evaluated.

Also, the results of vapor space sampling program were reviewed. Major toxic gases that are found in the dome

Bases RMCS B 5.31

BASES

(continued) space of the presently defined flammable-gas tanks are ammonia and nitrous oxide. Other gases are found in trace quantities and do not pose a concern. However, it was recognized that the existing data are limited and all tanks of interest are not covered. Thus, if new data reveal that toxic gases in excess of the hazardous limits are detected in a given tank, the consequence analysis must be reevaluated. The reevaluation may be done by simply scaling the toxic gas fraction and the guidelines against the ammonia fraction and the associated RGs.

Waste Temperature

The best available tank temperature data must show that the peak waste temperature (considering uncertainties) must be < 90°C. If the peak waste temperature is \geq 90°C. the envelope testing results discussed in Appendix F of Reference 1 must be re-evaluated.

Waste Energetics

The Safety Assessment assumption in regard to tank specific parameters such as gas and waste composition, gas release probability etc. may become non-conservative by a new analysis or data for a specific tank considered to be sampled by RMCS prior operations.

See Section 7.0 and Appendix G of Reference 1.

| BASES | |
|-------------|---|
| (continued) | Likelihood of Gas Release Events |
| | The GRE probability includes statistical distributions for gas-release amounts and rates that are based on limited data and expert judgment. If additional data or analyses exist for a specific tank to indicate that the GRE probabilities used in Reference 1 are not conservative, the accident frequencies need to be re-evaluated for that tank. |
| | In general, before the RMCS operation starts on a given tank, the best available tank specific data for gas inventory and gas release evidence must be evaluated to confirm that the statistical model for gas-release amounts and rates used in Reference 1 are still conservative. In general, if one or more of the following conditions are observed for a given tank, the GRE probability model given in Appendix L must be re-evaluated. |
| | Periodic level drops and level swells in excess of ± 3 in. Level drop ≥ 3 in. during or after an intrusive event Dome concentration measurements ≥ 25% of the LFL before, during or after a waste intrusive event. A well defined nonconvective layer (parabolic temperature profile) below a supernate or convective layer (flat temperature profile) that would be indicative of potential rollovers. Retained gas inventory estimates (via level swell, fill history, etc) is > 20% of the available dome space volume. |
| | If the re-evaluation indicates that the existing GRE model is not conservative for a given tank, a revision to the Safety Assessment will be necessary. |

Bases RMCS B 5.31

B 5.0 ADMINISTRATIVE CONTROLS (ACs) - GENERAL

B 5.31.2 Ignition and Envelope Testing

BASES

| BACKGROUND | Fire hazards, a significant part of the safety assessment, are considered to be beyond an extremely unlikely event based on ignition testing. |
|-------------------------------|--|
| APPLICABLE SAFETY ANALYSES | |
| | Appendix F "Thermal Analysis of Rotary Drilling". Appendix G "Waste and Crust Ignition", and Appendix T "Bureau of Mines Ignition Test Program Functional Requirements and Acceptance Criteria, all located in Reference 1 shall be used to verify drill bit and material performance. |
| | The material cited above provides a safety basis for RMCS operations. The summary report required by Appendix T must be reviewed and approved by the PRC and verification of same must be acknowledged as a prerequisite to RMCS operations. |
| - | Any subsequent change in drill bit or material would require revalidation per Appendix T requirements. |

Bases RMCS B 5.31

B 5.0 ADMINISTRATIVE CONTROLS (ACs) - GENERAL

B 5.31.3 Flammable Gas Detection System Verification

BASES

| BACKGROUND | The purpose of the sensors is to provide signals that can be used as safety shutdown indicators for both flammability and toxic hazards during rotary-mode core sampling (RMCS) waste intrusive operations. The primary flammability hazard is the release of hydrogen gas. and the primary toxicity hazard is the release of ammonia gas (which is also flammable). The signal will be used to shut down the drill truck and to alert personnel to evacuate the tank farm as a protection against toxic-gas exposure. |
|------------|---|
| APPLICABLE | |

SAFETY ANALYSES

Bases RMCS B 5.31

BASES

The use of a redundant and adequate flammable gas detector is a key assumption to provide safe shutdown of RMCS operations during a GRE. This reduces the likelihood of a spark in flammable gas atmosphere and also provides protection for toxic gas exposures. This mitigates unacceptable offsite consequences resulting from a dome collapse.

Two sensors have been identified and are being designed as part of the ventilation system. The first is a Whittaker Cell hydrogen detection system and the second is an SMC combustible gas sensor.

The Whittaker Cell is an electrochemical cell with a membrane placed between the sample gas and the active element. It is very selective for hydrogen and responds directly to the partial pressure of hydrogen on the other side of the membrane. It is configured with a calibration port that can be used to flood the sensor region with a calibration zero gas during operation conditions. For the sensor to read out concentration it is essential that the pressure in the sensor region be identical during the calibration as it is during actual operations.

The SMC combustible gas sensor uses a catalyst to "burn" the gas and detects the resulting heat release. To increase sensitivity and decrease drift, the heat detection is done by comparing the temperature of a reference (uncatalyzed bead) to that of a signal (catalyzed) bead. The beads are imbedded in a sintered metal housing which prevents the combustion energy from igniting a flammable mixture. It has the advantage of responding to both ammonia and hydrogen.

Westinghouse Hanford Company has considerable experience with the Whittaker Cell, which has been shown to have an adequate response time (Appendix C Reference 1). The Whittaker Cell has been shown to have adequate sensitivity, and experience has shown that it is very reliable and stable in the current tank farm applications. Calibration is required only every three months.

The SMC sensor has recently been tested and should provide the required protection against flammability and toxicity hazards during a gas-release event (GRE).

Bases RMCS B 5.31

B 5.0 ADMINISTRATIVE CONTROLS (ACs) - GENERAL

B 5.31.4 Tank Dome Activities

BASES

BACKGROUND The tank dome is considered to be Class I. Division 1. Group B environment because RMCS can induce a GRE. Sparks in the dome or volumes having a connecting path to the tank dome must be controlled. Control prevents ignition of flammable gas by non-FG/RMCS-related ignition sources.

This mitigates unacceptable offsite consequences resulting from a dome collapse.

APPLICABLE SAFETY ANALYSES

> This event considers the possibility of an ignition caused by the existence of energized equipment in the dome or domes of connected tanks or connecting ventilation systems. Any activity in the connecting tanks may initiate a fire that may propagate into the tank being sampled. Therefore, it is required that all equipment in the dome be rated for operations in a Class-I. Division-1. Group-B environment or a Class-I, Division-2, Group-B environment with automatic shutdown. Any equipment that does not meet this requirement must be deenergized during RMCS operations. No other activities in the connecting tanks or on the same tank are allowed during RMCS operations. These controls reduce the likelihood of ignition caused by existing equipment in the tank dome and the domes of connecting tanks. Violation of this control may result in an unanalyzed initiator.

> Class-I. Division-2. Group-B equipment is capable of sparking upon single failure. Consequently, such equipment must be protected by a reliable shut-down circuit. Furthermore, the background concentrations of flammable gases must be measured and shown to be less than 25% of the lower flammability limit (LFL) before energizing this equipment. In order to cause a spark, a sequence of double independent failures are necessary (the shut-down circuit must fail, and the Class-I, Division-2, Group-B equipment must also fail simultaneously).

Bases RMCS B 5.31

B 5.0 ADMINISTRATIVE CONTROLS (ACs) - GENERAL

B 5.31.5 Gas Leak Paths

BASES

BACKGROUND Above ground fires that could propagate into the dome are managed by controlling position of the non-qualified equipment. Since sparks from non-qualified equipment can not be eliminated, a safe distance criteria is developed. This criteria requires that possible leak diameter is to be known and controlled.

This mitigates unacceptable offsite consequences resulting from a dome collapse.

APPLICABLE SAFETY ANALYSES

Bases RMCS B 5.31

1

·

| BASES | |
|-------------|---|
| (continued) | Flammable-Gas Release Through Open Riser (Or Possible Leak Paths) Driven By Gas Release Event And Burn (Operation and Removal) |
| | This event postulates that a large GRE occurs, releasing flammable gas from openings in the tank, exposing the flammable gas to equipment with possible spark sources, resulting in a fire on top of the tank. This event is of concern during operation and removal phases of RMCS waste intrusive operations. None of the auxiliary support equipment on top of the tank is qualified to operate in a flammable-gas environment. |
| · | Preventing flammable-gas exposure to this equipment is managed by inspection of the tank top for leaks and the repair of all leaks with an effective diameter of ≥ 1 in. Open paths from the tank dome to the tank top include the exhauster stack, the breather inlet riser, open risers. drill string riser, and other possible tank leak sources (unsealed risers, pits, etc.). It is assumed that inspection of tank top penetrations for potential leak paths will find leak paths with nominal leak diameters > 1 in. It is assumed that undetected leak paths with a nominal 1-in. diameter could go undetected. Therefore, the top of the tank must be examined to identify leaks, and when identified, the leaks must be ≤ 1 in equivalent diameter (reference paragraph 4.1.2.1 of WHC-SD-WM-SAD- 035, revision 0-A). |
| | A portable stack over the breather inlet HEPA system will be used during waste-intrusive operations. The portable stack is 15 ft tall, has an upper 4-in. diameter, is sealed at the ground level, and is grounded. The purpose of using a portable stack over the breather inlet is two- fold: |
| | The gas release would be released through stack, resulting in increased atmospheric dilution and reducing the toxicological consequences of a GRE. |
| | Any non-qualified equipment on the top of the tank around the breather inlet HEPA system would be protected from flammable-gas exposures. |
| | It is required that any non-qualified electrical equipment |

Bases RMCS B 5.31

| R | Δ¢ | CF | 2 |
|---|----|----|---|
| 0 | n. | ັ | 0 |

(continued)

riser during waste-intrusive operations. If a GRE occurs, immediate personnel evacuation is required.

Another control also requires that when the drill truck needs to be parked over a riser or pit, the riser or pit must be sealed. Note that the riser or pit considered here is not the riser being sampled, but the one that the front part of the truck is parked on. If the truck is parked over an unused riser or pit, the potential spark location is not considered random, and no credit can be taken for the probability of a random placement. However, the leak size from pit or riser is assumed to be no bigger than 1 in. There is at least 3 ft between the top of the pit/riser and any potential ignition sources on the truck. A control was established to make sure the distance between potential ignition sources on the truck and the top of the pit/riser is \geq 36 in. Combining the failure probability to seal the riser/pit and violate the 36-in. distance criteria and the GRE probability that makes the dome pressure positive, the accident frequency is determined as 2.1E-8/yr. This frequency is low. However. the unmitigated frequency becomes 1.4E-4/yr. if the control to seal the riser/pit and 36-in, distance criteria are not implemented.

The flammable-gas release could occur from other unused risers if they have undetected leak paths. The control requiring the examination of risers before operations reduces the probability of have an unknown open path. It is assumed that leaks from threaded junctions, flanges, and cover plates could be identified with an equivalent leak diameter > 1 in. If a GRE occurs and non-qualified equipment is located close to these unknown openings, the accident frequency of a above-tank fire becomes 1.8E-7/yr. This frequency includes the probability of a GRE based on exposure time (Appendix L of Reference 1) and the probability of a temporal random spark. It also assumes that 50% of all risers on the top the tank leak after the initial inspection is performed. The existence of a spark on the equipment located around risers is also assumed. The unmitigated accident frequency is 2.8E-5/yr for this accident scenario; therefore, the control requiring limiting the leaks before the RMCS operation is important.

The last accident scenario includes the ignition of a flammable

Bases RMCS B 5.31

| R | Δ | C | F | ς |
|----|---|---|----|--------|
| υ. | | - | ц. | \sim |

(continued)

not to place any non-qualified equipment within 36 diameters of the riser being opened during waste-intrusive operations. Considering this control and a GRE probability based on exposure time including a temporal random spark, the mitigated accident frequency becomes 4.2E-8/yr. This number is conservative because it was not considered that the riser may be open for only a fraction of the mission time. The unmitigated accident frequency is 1.4E-5/yr.

The combined mitigated frequency of these three events is 2.4E-7/yr. Note that the fire propagation into the dome conservatively is assumed to be 1.0. This number is conservative because it is assumed that non-qualified equipment does include a spark source and the probability of a random spark in time is based on a conservative dome concentration.

Bases RMCS B 5.31

B 5.0 ADMINISTRATIVE CONTROLS (ACs) - GENERAL

B 5.31.6 Open Riser Exclusion Zone

BASES

 BACKGROUND
 Minimizes ignition sources in proximity with flammable

 gas.

 Minimizes gas leakage in proximity with ignition sources or unqualified equipment.

 This mitigates unacceptable offsite consequences resulting from a dome collapse.

 APPLICABLE SAFETY ANALYSES

See B 5.31.5

Bases RMCS B 5.31

B 5.0 ADMINISTRATIVE CONTROLS (ACs) - GENERAL

B 5.31.7 Energized Equipment in the Dome and Open Riser Exclusion Zones

BASES

| BACKGROUND | This control prevents ignition sources in proximity with \uparrow flammable gases. |
|-------------------------------|--|
| | This mitigates unacceptable offsite consequences resulting from a dome collapse. |
| APPLICABLE SAFETY ANALYSES | |
| | See B 5.31.5 |

Bases RMCS B 5.31

B 5.0 ADMINISTRATIVE CONTROLS (ACs) - GENERAL

B 5.31.8 Tank Loading

BASES

BACKGROUND An administrative limit on tank dome loading can prevent collapse.

This mitigates unacceptable offsite consequences resulting from a dome collapse.

APPLICABLE SAFETY ANALYSES

The static load capacity of the tank dome is monitored carefully and an overload state that could collapse the dome must be avoided. The equipment required on the surface of the tank to support FG/RMCS sampling operations qualifies as a live load (see Appendix N of Reference 1). The tank loads study permits a 50-ton live load on the tank dome. Even though the weight of all FG/RMCS equipment exceeds 50 tons, not all of the equipment is placed on the tank dome at one time. Therefore, the dome loading must be controlled by the dome loading limits for SSTs as specified in OSD-T-151-00013.

The dome would be subjected to dynamic dome loads if a truck were to fall from the hydraulic jack or from a platform. Appendix N considers this scenario and analyzes the consequences of the dynamic loading caused by dropping the truck. It is concluded that the dome could withstand the impact force of the 30.000-lb truck dropping on it from the 3-ft-high platform.

Bases RMCS B 5.31

B 5.0 ADMINISTRATIVE CONTROLS (ACs) - GENERAL

B 5.31.9 <u>Truck Position</u>

BASES

| BACKGROUND | The RMCS truck when parked over an unused riser must be protected from flammable gas releases. This requires controlling the distance between the leak source and truck, reducing above tank fire frequencies by mitigating |
|------------|--|
| | human error and minimizing ignition sources. |

APPLICABLE SAFETY ANALYSES

| It is important that when the RMCS truck needs to be parked over an unused, closed riser or pit, the riser or pit must be sealed. Note that the riser or pit considered here is not the riser being sampled, but the one that the front part of the truck is parked on. If the truck is parked over an unused riser or pit, the potential spark location is not considered random, and no credit can be taken for the probability of a random placement. However, the leak size from pit or riser is assumed to be no bigger than 1 in. There is a least 3 ft between the top of the pit/riser and any potential ignition sources on the truck. A control was established to make sure the distance between potential ignition sources on the truck and the top of the pit/riser is ≥ 36 in. Combining the failure probability to seal the riser/pit and violate the 36-in. distance criteria and the GRE probability that makes the dome pressure positive, the accident frequency is determined as 2.1E-8/yr. This frequency is low. However, the unmitigated frequency becomes $1.4E-4/yr$ if the control to seal the riser/pit and 36-in, distance criteria are not |
|---|
| to seal the riser/pit and 36-in. distance criteria are not implemented. |

Bases RMCS B 5.31

B 5.0 ADMINISTRATIVE CONTROLS (ACs) - GENERAL

B 5.31.10 Portable Inlet Stack

BASES

| BACKGROUND | Directing the breather inlet HEPA effluent 15 ft | | | |
|-------------------------------|--|---|--|--|
| | • | allows atmospheric dispersion coefficients to decrease so that toxicological acceptance guidelines at the onsite boundary can be met. | | |
| | • | prevents electrically unqualified equipment in proximity with flammable gas. | | |
| APPLICABLE SAFETY ANALYSES | | | | |
| | A por be us porta diame The p inlet | table stack over the breather inlet HEPA system will ed during waste-intrusive FG/RMCS operations. The ble stack is at least 15 ft tall, has an upper 4-in. eter. is sealed at the ground level, and is grounded. purpose of using a portable stack over the breather is two-fold: | | |
| | • | The gas release would be released through stack, resulting in increased atmospheric dilution and reducing the toxicological consequences of a GRE. | | |
| | • | Any non-qualified equipment on the top of the tank around the breather inlet HEPA system would be protected from flammable-gas exposures. | | |

Bases RMCS B 5.31

B 5.0 ADMINISTRATIVE CONTROLS (ACs) - GENERAL

B 5.31.11 Drill Bit. Core Barrel, and Drill Rods

BASES

.

| BACKGROUND | Results from envelope testing and ignition testing are only valid for equipment of the design tested. | |
|-------------------------------|---|--|
| | This mitigates unacceptable offsite consequences resulting from a dome collapse. | |
| APPLICABLE SAFETY ANALYSES | | |

Bases RMCS B 5.31

BASES

One of the safety concerns of sampling with the rotarymode drill in flammable gas tanks is the ignition of the flammable gas in the waste by the frictional sparks created by the drill bit. The condition of waste in terms of hardness is not known before the operation. A possibility of penetrating a very hard waste layer in a tank exists. In addition, there may be some metal debris lost or dropped from the riser in the past. Hard materials such as rocks or metals can also exist in the waste. Thus, it is likely that the drill bit may strike against metal and other hard objects during drilling.

Ignition caused by frictional sparks is evaluated by performing ignition testing in bounding conditions. Appendix T (Reference 1) discusses the ignition testing requirements and acceptance criteria. The objective of the tests is to demonstrate that the operation of rotary core drilling in a bounding frictional environment with bounding gas composition does not cause an ignition. Testing is being performed by the WHC and BOM personnel.

The conditional frequencies of a fire accident resulting from exceeding the rotational speed and down force are estimated as 1.4E-5 and 9.4E-5/yr as indicated in Table 4.12. Controls are established to trip the drilling operation when the rotational speed and down force exceed 55 rpm and 750 lbf. There is no delay time for the trip except the delay time from the data acquisition system. If needed, the alarm points will be set at lower values. However, drilling must stop when the trip value is reached. The RMCS operations must use only the drill-bit type tested according to the requirements listed in Appendix T (Reference 1).



B 5.0 ADMINISTRATIVE CONTROLS (ACs) - GENERAL

B 5.31.12 RMCS Operations

BASES

. .

| BACKGROUND | Flammable gas concentrations are mixed, dispersed, and reduced with exhauster preparation. Accumulation of aerosols, particulates, and flammable gases following waste intrusive activities must be minimized. | |
|------------|--|--|
| | This mitigates unacceptable offsite consequences resulting from a dome collapse. | |

Bases RMCS B 5.31

BASES

APPLICABLE SAFETY ANALYSES Wastes including mixtures of sodium nitrate and sodium nitrite with organic compounds can produce violent exothermic reactions. Increasing the temperature of the waste in the vicinity of the drill bit can cause a thermal runaway. There are several hazards that are associated with a local thermal runaway, and they are discussed in Appendix G (Reference 1). Two major important hazards are in ignition of the flammable gas and the initiation of a self-propagating exothermic reaction in the waste. Reactions in mixtures containing relatively small amounts of organic compounds can result in temperatures greater than the autoignition temperature of hydrogen mixtures, so the ignition of flammable gases is the more limiting condition. However, a self-propagating reaction would produce very high temperatures, which would cause structural damage to the tank. The consequences of a self-propagating reaction could be severe.

Because the possibility of flammable-gas mixture cannot be eliminated, the approach used is to take all practical measures to eliminate ignition sources. A local runaway reaction is a potential ignition source, so the requirement that there be no local runaway reaction is consistent. Preventing a local thermal runaway is also protection against a propagating exothermic reaction, and it eliminates the possibility of generating additional flammable gas as a result of elevated temperatures. Appendix G of Reference 1 discusses runaway reactions and waste ignition in great detail. Basic conclusions of Appendix G are that local runaway reactions can be prevented by establishing waste temperature limits. The following temperature limits are established:

- The temperature of small waste fragments produced at the drill tip must not exceed 180°C.
- The temperature of the drill bit and the average temperature of the waste affected by drilling must not exceed 160°C for more than 10 min.

Because the consequences of a propagating exothermic reaction are severe, FG/RMCS should not be performed in tanks in which a propagating exothermic reaction may occur.

Bases RMCS B 5.31

BASES

(continued)

force. and nitrogen purge flow to comply with the safety criteria given above. Appendix F (Reference 1) discusses the details of testing and the results obtained.

As a summary, results of envelope testing and their analysis showed that the drill bit surface temperature correspondingly the waste substrate temperature can be kept below 160°C, including an uncertainty of 10°C, if the following limits are applied: Down force < 750 lbf, rotational speed < 55 rpm, minimum nitrogen flow > 30 scfm, and penetration rate > 0.75 in./min. The chip temperatures under these conditions are also limited to 180°C as required. If a trip is initiated when one of the set points for these four parameters is exceeded, drilling must be stopped. After a shutdown there must be a waiting period of 10 min before drilling can continue. The waiting period of 10 min. is based on the experimentally determined cooling time. The testing and the analysis included plugged holes on the drill bit.

The minimum purge flow must be > 30 scfm: however, it is possible that necessary cooling to the drill bit would not be provided if there were a leak from the nitrogen purge system between the flow measurement location and the drill bit. WHC determined that the leak from the truck is within the uncertainty range of instrumentation. This control requires that the leak rate must be within the uncertainty range of instrumentation or < 2% of the nominal flow.

Drill rods are threaded to each other. An O-ring is used to provide a seal. The leaks are possible if the O-rings are left out. WHC determined the possible leak rates could not be higher than 0.3 scfm when the O-rings are not used. This is < 1% of a nominal flow of 30 scfm and negligible. With the use of O-rings, the leak rate also was measured and was shown to be negligible. Therefore. O-rings on the drill rod are not required, and the nitrogen purge flow for drill bit cooling is sufficient when set to a minimum of 30 scfm.

There is one event that would include an unknown leak path as a result of failure of the drill string during drilling. If the drill bit or string becomes embedded in the waste momentarily because of debris in the waste.

Bases RMCS B 5.31

| BASES | |
|-------------|--|
| (continued) | torque could continue to be applied to the drill string at a constant rate. If such a condition occurs, there is a possibility that the drill string could partially fail. Continuing to operate with a partially failed drill string could result in nitrogen flow bypass through the failed area. This concern is assessed below. |
| | Appendix N (Reference 1) examines the possibility of over- torquing the drill string. The drill string is considered as having torque applied from the upper end, but the rotation of the lower end is not allowed. Appendix N presents two methods to determine the time necessary to break the drill string. Linear elastic methods are applied as a first approximation to obtain the lower bound failure estimate. Second, strain-energy methods are used to determine an upper bound by assuming that the ultimate shear strain in the drill rod is proportional to the shear modules. It is estimated that failure would occur in less than 15 sec for all rotational speeds. Note that Appendix N did not take any credit for the threaded drill rods. Experience shows that the drill string always fails at the threaded sections. The real failure time is expected to be in a few seconds because of the stress concentration factor for threaded sections. Therefore, it is concluded that a drill string tear without a break is very unlikely. |
| | Envelope testing measured the rate of increase in the drill-bit surface temperature when nitrogen flow is terminated at steady-state operating conditions. The test results are summarized in Appendix F (Reference 1). Results showed that an average heat-up rate of 2°C/s is observed in the time period of 0 to 20 sec after the nitrogen flow is shut down. This rate corresponds to a temperature increase of 30°C in 15 sec in which the drill string would be broken when overtorqued. Envelope testing established the operating parameters so that the drill bit and waste temperature is less than 150°C. Considering a 30°C heat-up of the drill bit for this accident. the drill bit/waste temperature would be 180°C. Appendix G (Reference 1) argues that the waste temperature would be allowed to be at the minimum exothermic-reaction temperature of 180°C for a short period of time because the induction time of reaction is expected to be much larger than 10 to 20 sec. Therefore, it is concluded that if the drill fails because it is over-torqued, it would |

| E | 3ases | |
|---|-------|--|
| | RMCS | |
| В | 5.31 | |

| BASES | | · · · · · · · · · · · · · · · · · · · |
|-------------|----------------|--|
| (continued) | vicin react | ity of the drill bit would not experience runaway ions. |
| | • | RMCS operations must be shut down if the dome flammable gas concentration is above 5000 ppm hydrogen equivalent. Based on currently available calibration data, this set point gives protection for ammonia concentrations of ~10.000 ppm if the gas is purely ammonia. |
| | • | Evacuation is necessary if a gas release occurs. |
| REFERENCES | 1. | LANL, 1996, A Safety Assessment of Rotary Mode Core Sampling in Flammable Gas Single Shell Tanks, Hanford Site, WHC-SD-WM-SAD-035, Rev. 0-a, Los Alamos National Laboratory, Los Alamos, New Mexico. |

•



CHARACTERIZATION EQUIPMENT ENGINEERING Characterization Field Engineering Characterization Equipment Design

To: N. Morcos

From: Dennis Hamilton

Phone #:

Phone #:(509) 372-1130

Fax #: (208) 526-9267

Location: Idaho

Location: Richland, WA

Fax #: (509) 373-4238

MESSAGE:

• • ~ · ·

REVISIONS MADE TO ROTARY-MODE CORE SAMPLING SAFETY ASSESSMENT REVISION-0a AS A RESULT OF INEL REVIEW

Changes are underlined.

Page 1-2. Section 1.2, 1st paragraph

.... parameters of interest. The total waste stored in Tank A-101 is 953 Kgal and is mostly consisted of salt cake (950 Kgal). Tank A-101 waste is classified as double-shell slurry feed (DSSF). Estimated radionuclides in Tank A-101 are Cesium (Cs-137), Strontium (S-90), Plutonium (Pu), and Uranium (U).

Page 2.2, Section 2.2.1

The DOE Orders cited in Table 2-1, are presently applicable to the design of the rotary-mode core sampling equipment. They are helpful in developing the criteria outlined in Sections 2.1.2, 2.1.3, 2.1.4, and 2. <u>The risk criteria are given in Section 5 of this SA</u>.

Page 2-4, Section 2.1.4.

... The safety criteria are based on a set of tests in which dry sodium acetate nitrate/nitrite mixtures exhibited propagating behavior at about 300°C (572°F) with a TOC value greater than 6 wt%. Appendix G evaluates this criteria in detail for each SST.

Page 2-8, Section 2.3

.... Of the three possible tank configurations, the 1,000,000-gal.-capacity tank used in Farms A, AX, and SX is schematically shown in Fig. 2-1. The other tank configurations are similar to that of given in Fig. 2-1. The BY-, TX- and S-Tank Farm has tanks with 758,000-gal. capacity, but tanks in Farms B, BX, C, T and U have a capacity of 530,000 gallons.

Page 2-13, Section 2.4.1.4, 1st component

Drill Bit. The drill bit rotationally bores into the waste to produce a nominal 1-in.-diameter core sample, and acts as the leading tip of the drill string. The bit has a hollow-cored center section surrounded by cutting teeth and holes on the drilling surface for nitrogen purge flow. The commercially available unit (nominally 2.5-in. o.d.) is made of copper-based (sintered bronze) material with teeth designed to "smear" when they come into contact with the bottom of the tank to prevent penetration. Appendices T and F addresses the safety concerns in regard to material properties.

1

帧

Page 20, Section 2.4.2, 3rd item

....Vendor information indicates that the HEPA filter performance is undetermined at a differential pressure of 10 in. w.g. after 15 minutes, but there are no relief valves or vacuum breakers installed to protect the HEPA from excessive delta pressure. In order to protect against filter collapse, the blower is limited to 9 inches of water static pressure.

Page 2-31, Section 2.5.3

Heading of 2.5.4 is eliminated and the remaining of text is renumbered.

Page 5-10, Section 5.4.1. 4th paragraph

The dome peak ammonia and nitrous oxide concentrations then become 1.2%, 6%, and 12% for the anticipated, very unlikely, and extremely unlikely gas-release events <u>defined based on frequency ranges and expected gas release</u> <u>amounts and marked as GRE-1. GRE-2 and GRE-3 in Table 5-7. These three</u> <u>GRE, GRE-1, GRE-2, and GRE-3, categories will be used in Tables given in the</u> <u>rest of this section....</u>

| Event Probability | Event Category | Q-Prompt Release, (ft ³) | Peak NH ₃ Conc. (%) |
|-------------------|----------------|---|--------------------------------|
| ≥1.0 E-2 | <u>GRE-1</u> | ≤1000 | 1.2 |
| ≥1.0 E-4 | <u>GRE-2</u> | ≤ 5000 | 6 |
| ≥1.0 E-6 | GRE-3 | ≤ 10,000 | 12 |

TABLE 5-1 ANTICIPATED AND UNLIKELY GAS-RELEASE EVENTS

TABLE 5-2 SUMMARY OF GAS-RELEASE ACCIDENTS

| Accident Condition | Frequency (yr) | Gas Concentration & Release Rate |
|--|---------------------------|-------------------------------------|
| 1- <u>GRE-1</u> and open riser (removal) | 1.1E-3 (0.01x5.6E-2x2) | 1.2% NH3 in the dome, 66 scfm* |
| 2- <u>GRE-2</u> and open riser (removal) | 1.1E-5 (1E-4x5.6E-2x2) | 6% NH3 in the dome, 333 scfm* |

| 3- <u>GRE-3</u> and open riser (removal) | 1.1E-7 | 12% NH ₃ in the dome, |
|--|------------------------------------|-----------------------------------|
| | (1E-6x5.6E-2x2) | 666 scfm* |
| | Beyond cxtremely unlikely | |
| 4-GRE-1 and open frisbee (operation)(nitrogen | 3.2E-6 | 1.2% NH ₃ in the dome, |
| purge to riser sleeve fails) | (2x0.01x1.6E-4) | 66 scfm* |
| 5- <u>GRE-2</u> and open frisbee (operation) (nitrogen | 3.2E-8 | 6% NH ₃ in the dome, |
| purge to riser sleeve fails) | (2x1.0E-4x1.6E-4) | 333 scfm* |
| | beyond extremely unlikely event | |
| 6-GRE-3 and open frisbee (operation) (nitrogen | 3.2E-10 | 12% NH, in the dome, |
| purge to riser slocve fails) | (2x1.0E-6x1.6E-4) | 666 scfm* |
| | beyond extremely unlikely event | |
| 7-GRE-1 and drill string open at the top with | 2.6E-5 | 1.2% NH, in the dome, |
| sampler in the drill string | (2x0.01x1.3E-3) | 66 scfm* |
| 8-GRE-2 and drill string open at the top with | 2.0E-7 | 6% NH ₃ in the domc, |
| sampler in the drill string | (2x1.0E-4x1.3E-3) | 333 scfm* |
| | beyond extremely unlikely ovent | |
| 9-GRE-2 and drill string open at the top with | 2.0E-9 | 12% NH, in the dome, |
| sampler in the drill string | (2×1.0E-6×1.3E-3) | 666 scfm* |
| | beyond extremely unlikely event | |
| 10- GRE-1 from exhaust and inlet stack | 0.02 | 1.2% NH ₃ in the dome, |
| (operation) | (2x0.01) | 66 scfm* |

TABLE 5-8 (cont) GAS-RELEASE ACCIDENTS

| Accident Condition | Frequency (yr) | Gas Concentration & Release Rate | |
|--|-------------------|-------------------------------------|--|
| <u>11-GRE-2 from exhaust and inlet stack</u> | 2.0E-4 | 6% NH, in the dome, | |
| (operation) | (2x1.0E-4) | 333 scfm* | |
| 12-GRE-3 from exhaust and inlet stack | 2.0E-6 | 12% NH ₃ in the dome, | |
| (operation) | (2x1.0E-6) | 666 scfm* | |
| 13- <u>Continuous releases from exhauster after an</u> | 0.02 | 1.2% NH, in the dome, | |
| <u>GRE-1</u> | (2x0.01) | 250 scfm** | |
| 14-Continuous releases from exhauster after an | 2.0E-4 | 6% NH3 in the dome, | |
| <u>GRE-2</u> | (2x1.0E-4) | 250 scfm** | |

| 15-Continuous releases from exhauster after an | 2.0E-6 | 12% NH ₃ in the dome, |
|--|------------|----------------------------------|
| <u>CRE-3</u> | (2x1.0E-6) | 250 scfm** |

* Averaged over 15-minute period Page 6-1

(Note that all tables in Section 5 are updated).

Page 6-1, Section 6.1, 2nd paragraph and reference 4

Westinghouse Hanford Company (WHC) standard controls include a series of WHC documents that define the safety envelope for the tank farm, waste transfer activities, and waste storage activities. The primary document is the WHC Health and Safety Plan¹although other documents include the Safety Assessment for Push- and Rotary-Mode Core Sampling in Ferrocyanide Tank,² Safety Analysis for the Push Mode and Rotary Mode Core Sampling and the Interim Operational Safety Requirements for Rotary Mode Core Sampling in Flammable Gas Single Shell Tanks.4 During the development of the procedures for each of the activities, the current operational safety requirements (OSRs), interim operational Safety requirements (IOSRs), and operational safety documents (OSDs) must be considered (refer IOSR⁴ for restart requirements). The safety envelope established by the analyses shall not be changed unless approved by the Department of Energy (DOE). The controls provided in this section can be modified if the appropriate organization grants approval

<u>Ref. 4</u>.

WHC-SD-WM-OSR-OOS Rev O' Single Shell Tank Interim Operational Safety Requirements," Author L.F. Daughorty

Page C-4

...This model is based on the assumption that the species ratios obtained from the dome space measurements are the same for the gas bubbles that exist in the waste. Thus, it is assumed that

- The mass transfer (including the molecular diffusion out of the • waste) from the waste surface is negligible, and
- The species ratios are established by equilibrium in the waste prior initiation of GRE.

Page C-6, last paragraph

Based on this analysis, the composition obtained for Tank AX-101 results in the maximum equivalent fuel being 103% H_2 and appears to be the limiting composition for the flammability analysis at the conditions prior GRE.

Page C-11, 2nd paragraph

... Thus, even small volumes of gas burned in the dome space may result in catastrophic dome failure. Note that the gas release volume of 96 m³ is not 100% hydrogen. It was assumed that there is 60% ammonia in the released gas. Remaining 40% includes 88.1% hydrogen (Tank AX-101) as indicated in Table C-3. Thus, the hydrogen concentration becomes 35% in the released gas. This results in 2/4% hydrogen concentration in the dome.

Page G-5, Section G.4 definitions

- U = heat transfer coefficient for the fragment,
- T_{o} = initial temperature of the fragment,
- $T_s = temperature surrounding the fragment,$
- r = density of the fragment,
- cp = heat capacity of the fragment, and
- \vec{D} = diameter of the fragment,
- k = thermal conductivity.

Page G-15, Section G.5.2.3

Summary

The second method of analysis is more restrictive than the event-tree analysis given in Section 5.2.1

Page J-11. 2nd paragraph

Hydrogen and air mixtures near stoichiometric (~ 29.5% hydrogen) are known to be detonable. Mixtures departing from stoichiometric, either in the hydrogen-lean or hydrogen-rich direction are increasingly more difficult to detonate. It has been generally believed that hydrogen-air mixtures with mixture ratios of <18% or >58% hydrogen, could not be detonated. As shown in Section 2 of this appendix, the hydrogen concentration in the drill string reaches 30% in 16 minutes at 2 ft above the rotary drill. One operation of drilling takes about 30 minutes. By that time, the hydrogen concentration is already above 18%. Therefore, some burn in the drill string may detonate.

Page J-11, Section J.3.3

Conclusions

From the above analysis, the overpressure and the rate-of-pressure rise during a burn in the drill string are 630 kPa and 2279.5 bar/s, respectively.

During a detonation in the drill string, the overpressure and the rate-ofpressure rise are 15 bars (1.5×10^6 Pa) and 3×10^6 bar/s, respectively. A DDT may occur in the drill string. Therefore, any burn in the drill string may result in a detonation because the rate of pressure rise is very high (3×10^6 bar/s) during the detonation. Consequently, the structure of the drill string is conservatively assumed to fail

Page L-1, Section L.1, 3rd bullet

The probabilistic GRE model discussed in this appendix was introduced only after the completion of these first two steps in order to <u>obtain</u> realistic accident frequencies.

Page L-4, 5th paragraph

It can be stated that 2 out of 87 activities resulted in a detectable GRE, suggesting a frequency of 2.3×10^{-2} . Accounting only for the activities in the SSTs, the probability becomes 2 out of 77 (2.6×10^{-2}). Also subtracting the activities in which the level data were inconclusive, the probability becomes 2 out of 58 (3.4×10^{-2}). However, this approach does not apply to a bounding tank, and at best it provides an overall probability of a GRE in the flammable-gas tanks. The <u>LOWs</u> are installed in A-101 (waste level ~345 in.) and AX-101 (waste level 278 in.), which have the largest waste volumes in the SSTs that are on the original FGWL. Tank A-103, which showed a 2.4 in. level drop between March 25 and March 26, 1986, also has a large waste volume (waste level ~370 in.). The other possible 2-to 3-in. level drop corresponds to Tank SX-104 that has less than 240 in. of waste.

Page L-8, Section L.3, 2nd paragraph

First, the discussion provided in the previous section is cast into a qualitative model. The qualitative likelihood matrix is provided in Table L-1. In Table L-1 qualitative likelihood definitions included following frequencies: a-Range 1 is defined as an event with a frequency of 1. b-Range 2 is defined as an event with a frequency of 10^{-2} , d-Range 4 is defined as event with a frequency of 10^{-3} , e-Range 5 is defined as an event with a frequency less than 10^{-3} .

TABLE L-1 QUALITATIVE LIKELIHOOD MATRIX FOR GAS-RELEASE VOLUMES AND RATES

| | Volun | ie ((13) | |
|-----|-------|----------|--------|
| 100 | 500 | 2500 | 10,000 |

| | Range 1 | <u>Range 2</u> | Range 3 | Range 4 |
|------|---------|----------------|---------|---------|
| 1 | | | | |
| | Range 2 | Range 2 | Range 3 | Range 4 |
| 10 | | | | |
| | Range 3 | Range 4 | Range 4 | Range 4 |
| 100 | | , | | |
| | Range 4 | Range 4 | Range 5 | Range 5 |
| 1000 | | | | |

Page L-9, 2-4th paragraph

For release volumes that are never observed in the SSTs, a qualifier of "<u>Range 3</u>" is assigned. The existing models do not predict release volumes corresponding to <u>Range 3</u> and less likely categories, except for a rollover and seismic event that can liquefy the waste by vigorous shaking. Even for these <u>Range 3</u> events the release volume is limited by the fraction of total retained gas. The maximum values shown in Table L-1 (10,000 ft³) corresponds to the largest GRE volume observed in Tank 101-SY before mitigation. Similar events have not been observed in any of the SSTs.

The rate range shown in Table L-1 is obtained by considering the orders-ofmagnitude, based on the discussions provided in the previous section. Only rollovers and seismic shaking may result in release rates on the order of 100 ft^3 /min or greater, while resulting in large release volumes. There is a remote possibility of tapping into a high pressure gas region which may result in large release rates but a very small volume. Thus, while small volume releases at high releases rates are <u>believed to be Range 4 events</u>. large volumes with large releases rates are in Range 5 event category.

The qualitative matrix may be quantified for the gas-release volume and gas releases rates assuming that these two are independent parameters. In the quantification process, the following probabilities were assigned: a probability of 1 for Range 1 event, 10⁻¹ for Range 2 event, 10⁻² for Range 3 event and 10⁻³ for Range 4 event. Thus, the quantitative results may be approximated as a logarithmic distribution given by....

Page R-5, Section R.4.1, 2nd paragraph

.. containing SiO_2 at a density of 0.1 g/cm³, we find the exposure at 1 cm from the surface is <u>about 300</u> mR/h at the end of 100 minutes of operation. This exposure may be scaled with units of 100 minutes of time because it is based on an average mass flow rate.

Page U-1, Section U-1

APPENDIX U FUNCTIONAL REQUIREMENTS FOR FLAMMABLE AND TOXIC <u>GAS (HYDROGEN AND</u> <u>AMMONIA)</u> SENSORS

INTRODUCTION

The purpose of the sensors is to provide signals that can be used as safety shut-down indicators for both <u>flammability (mainly hydrogen and ammonia)</u> and toxic (mainly ammonia) hazards during

Page U-2, Section U-2, 2nd paragraph

.... operation conditions. For the sensor to read out concentration it is essential that the pressure in the sensor region be within the expected tank pressure range during the calibration.

Page U-2, Section U.3.1, 3rd and 4th bullets

The configuration must allow for the system to be calibrated in a manner that ensures that zero and calibration gas are within the <u>expected tank pressure</u> range during the calibration. The system must be able to be operated during pressure transients caused by a potential GRE.

The calibration should be performed at initial deployment and every three months after that point. The calibration should consist of setting the zero with pure air or nitrogen and calibration with a nominal 6000 ppm hydrogen. The calibration procedure with a nominal 6000 ppm hydrogen should test the shutoff electronics as well as the sensor reading.

Page U-3, Section U.3.2., 5th and 8th bullets

•In addition to the electrical qualification, it must be documented that the sensor element itself will not ignite a flammable mixture.

•Configuration of both sensors must allow the system to be calibrated in a manner that the pressure in the sensor region is within the expected tank pressure range during the calibration

Page U-4, Section U.4

...; a Whittaker hydrogen detector cell and a SMC combustible gas detector. The use of SMC and Wittaker sensors provides a redundancy in hydrogen detection. The detection of ammonia can only be performed by the SMC sensor and therefore is not redundant.

.....Both sensors themselves do not ignite a flammable mixture when they are operational.

SMC sensor, if set to trip at a 5000-ppm hydrogen equivalent, should sense both a 5000-ppm hydrogen concentration and a 12000-ppm ammonia concentration for alarm purposes if the gas were purely hydrogen or ammonia, respectively. The set point for a rate of increase of concentration is 100 ppm/s.

Wittaker cell, if set to trip at a 5000-ppm hydrogen equivalent, should sense a 5000-ppm hydrogen concentration for alarm purposes if the gas were purely hydrogen. The set point for a rate of increase of concentration is also 100 ppm/s.

A control is established for an SMC sensor to be initially calibrated with hydrogen and ammonia and an SMC sensor to be replaced at least once per month with a new sensor that has been calibrated with both hydrogen and ammonia. A control is also established for the SMC sensor to be calibrated in the following fashion. A functional test shall be performed at the initial setup at each location and every day the system is used after that point. The functional test shall consist of setting the zero with pure air or nitrogen and calibration with a nominal 6000-ppm hydrogen. The functional test procedure with a nominal 6000-ppm hydrogen shall test the shutoff electronics as well as the sensor readings. A functional calibration test shall be performed once every three months for the Whittaker cell.

Page U-5, 3rd paragraph

The following controls are established for the flammable-gas detection system:

- SMC sensors shall be replaced at least once each month with new sensors,
- New SMC sensors shall be initially calibrated in a laboratory environment using hydrogen and ammonia,
- A functional calibration test shall be performed in the field at the initial setup and then every day for the SMC sensor.
- <u>A functional calibration test shall be performed once every three</u> months for the Whittaker cell.

• The functional calibration test shall consist of setting the zero with pure air or nitrogen and calibration with a nominal 6000-ppm hydrogen for both sensors,

Appendix A

The second column of Table A-5 named as Hazard. We agreed to change this $\$ title as Hazard/Accident. It, however, is very difficult to revise this table because it is scanned from a MS-DOS based-document. Therefore, we suggest that we make this change in Revision 1 if INEL agrees.


96/3861 LOCKHEED MART

Lockheed Idaho Technologies Company P. O. Box 1625 Idaho Falls, ID 83415

Mr. R.R.McNulty, ESQ DOE-RL TWRS Management System Division P.O.Box 550 Richland, WA 99352

Date: July 18, 1996 From: N. Morcos, MS 2114

Subject: Third Tier Review Safety Evaluation Report of the Safety Assessment document titled "A SAFETY ASSESSMENT OF ROTARY MODE CORE SAMPLING IN FLAMMABLE GAS SINGLE SHELL TANKS: HANFORD SITE, RICHLAND; WASHINGTON."

Attached is the Third Tier Independent Review Team's (IRT) Safety Evaluation Report (SER) for the Hanford Safety Assessment (SA) for the document titled "A SAFETY ASSESSMENT OF ROTARY MODE CORE SAMPLING IN FLAMMABLE GAS SINGLE SHELL TANKS: HANFORD SITE, RICHLAND, WASHINGTON" (RMCS). The SA review was performed in response to a request from the DOE-RL Management Systems Division (MSD) independent review coordinator (April 26, 1996). The review was performed in accordance with DOE-RL's DOCUMENT REVIEW PLAN FOR SAFETY ASSESSMENTS FOR SAT-WELL PUMPING AND ROTARY MODE CORE SAMPLING IN FLAMMABLE GAS TANKS (Appendix 1 of SER). The IRT members were selected in accordance with directions in the review charter and submitted for approval by DOE-RL TWRS MSD on the basis of their qualifications and their expertise. The list of approved reviewers is shown in Appendix 2 of the SER.

Based on the independent review, the IRT recommends approval of the SA titled "A SAFETY ASSESSMENT OF ROTARY MODE CORE SAMPLING IN FLAMMABLE GAS SINGLE SHELL TANKS: HANFORD SITE, RICHLAND, WASHINGTON" together with it's attached ISORs (contained in engineering change notice, ECN 609990) without any conditions or limitations except for the provision that any changes made to the operations or associated equipment as described in the SA should be followed by a careful USQ assessment process

The independent review process is summarized in the Executive Summary in the SER. The associated details and documentation are available in the RMCS SA file as submitted to DOE-RL TWRS MSD.

Please do note hesitate to call me at 208-526-4926 if I can be of any additional assistance

Sincerely,

a te

N. Morcos, Ph.D. IRT Chairman.



RMCS SAFETY EVALUATION REPORT

RMCS (SA)

SAFETY EVALUATION REPORT

Prepared for:

U.S. Department of Energy Richland Operations Office Tank Waste Remediation System

Prepared by:

The INEL Independent Review Team

| N. Morcos | nethuren |
|----------------|----------------|
| G. Dinneen | Geneld Durnean |
| K. Moore | X Amoon |
| J. Prendergast | Domender For |
| G. Beitel | Sunge 9. Bit |
| | |

Lockheed Martin Idaho Technologies P.O. Box 1625 Idaho Falls, Idaho 83415

Work Order MSIDL DOE 1

Idaho Technologies Company

P.O. Box 1625

idaho Falls, ID 83415

ABSTRACT

This Safety Evaluation Report (SER) documents the INEL Independent Review Team's (IRT's) assessment of the (SA) A SAFETY ASSESSMENT of ROTARY MODE CORE SAMPLING in FLAMMABLE GAS SINGLE SHELL TANKS: HANFORD SITE, RICHLAND, WASHINGTON" (RMCS)

The SA addresses each of the required elements associated with the installation, operation and removal of the rotary mode core sampling (RMCS) device in Flammable Gas Watch List (FGWL) Single-Shell tanks (SSTs) located within the 200 Area in the Hanford Site, Richland, Washington. The RMCS operations are needed to retrieve waste samples from SSTs with hard layers of waste for which push-mode sampling is not adequate for sampling. Potential hazards associated with the proposed activities were identified and evaluated systematically. Potential accident cases that could result in radiological or toxicological gas releases were identified and analyzed, and their consequences analyzed. Administrative controls, procedures, and necessary design changes to eliminate or reduce the potential hazards were also identified.

The SA was prepared using the "Interim Guidance for Preparing Safety Assessments", (rev 2, March 6, 1992), developed by SAIC for DOE as directed by a letter from Len Duffy in a memorandum dated May 6, 1992 which included in Appendix 3. The SA addresses most of the elements required in the Department of Energy (DOE) Standard DOE-STD-3011-94 "Guidance for preparation and Submittal of Basis Interim Operation (BIO) for DOE Nonreactor Nuclear Facilities." DOE Standard DOE-STD-3009-94 "Preparation Guide for U.S. Department of Energy Nonreactor Nuclear Facility Safety Analysis Reports" was used as guidance to perform the Hazard Analysis.

The independent review was performed in accordance with a DOE-RL charter and is intended to assess compliance of the RMCS SA with applicable safety-related statutes, rules, DOE Orders, and Standards and, in particular with the requirements of DOE Order 5480.23 as delineated in related DOE standards. Additional clarification and guidance regarding these requirements were obtained from DOE-STD-1027-92, "Hazard Categorization and Accident Analysis Techniques for Compliance with DOE Order 5480.23, Nuclear Safety Analysis", DOE-STD-3009-94, "Preparation Guide for U.S. Department of Energy Nonreactor Nuclear Facility Safety Analysis Reports", DOE-STD-3011-94 "Guidance for preparation and Submittal of Basis Interim Operation (BIO) for DOE Nonreactor Nuclear Facilities", and the DOE-RL approved Quality Assurance Plan. The review was also performed on the basis of six broad standards of performance: accuracy, completeness, traceability, consistency, readability, and functionality.

The IRT members were selected in accordance with the review charter submitted for approval by DOE-RL TWRS PRI.

The review was conducted in two stages. The first stage was initiated upon receipt of the document on May 6, 1996. The INEL review comments were forwarded to DOE-RL on May 10, 1996 for evaluation by DOE and WHC. Responses to our comments were received from WHC on June 10, 1996. The response were evaluated and formally returned to DOE-RL on June 17, 1996 requesting incorporation of the changes agreed upon to resolve the open issues. Finally, a meeting was held with WHC, Los Alamos staff, and DOE-RL on July 2 and 3, 1996 to review the final document draft and resolve/close all remaining open issues. The final draft of the ISORs was also reviewed on July 15, 1996.

This document summarizes the evaluation of the revised revision 0 of the RMCS SA and the final draft of the IOSRs as received from WHC. The evaluation entailed assessment of the modifications to the documents as agreed upon during the July 2 and 3, 1996 meeting to resolve the identified IRT concerns. All open issues and areas of concern have been clarified, addressed and resolved. There remains no open issues of concern to the IRT after reviewing the changes incorporated into the final document revision as reviewed. A copy the Review Comment Record sheets used in the July 2-3, 1996 meeting for closing all open issues is included as Appendix 4.

The DOE-RL, TWRS recommends approval of the SA titled "A SAFETY ASSESSMENT of ROTARY MODE CORE SAMPLING in FLAMMABLE GAS SINGLE SHELL TANKS: HANFORD SITE, RICHLAND, WASHINGTON" (WHC-SD-WM-SAD-035, Rev. 0-a); with the provision that any changes to the operations or the equipment as described in the document should be followed by a careful USQ assessment process. This SA is a stand-alone safety analysis supporting a temporary activity. It systematically identifies the hazards of the RMCS operations, describes and analyzes measures taken to eliminate, control and mitigate identified hazards, and analyzes potential accidents together with their associated risks.

The SA is organized in accordance with the "Interim Guidance for Preparing Safety Assessments", (rev 2, March 6, 1992), as shown in Appendix 3.

The authorization basis consists of this SER, the Safety Assessment # WHC-SD-SAD-035, Rev 0-a; the WHC letter # 9653662 R2, dated August 30, 1996, and the IOSRs identified in

ECN # 609990. Rotary mode core sampling will be limited to the following four (4) tanks: AX-101, AX-103, BX-110, and TY-102.

Paul Hemang

Paul Hernandez, DOE-RL, TWRS

TABLE OF CONTENTS

ABSTRACT

| EXE | CUTIVE SUMMARY Description Evaluation Criteria Evaluation Findings Conclusions | | • • • | |
|----------|---|---------|-------|---|
| 1.0 | SCOPE1.1Description1.2Evaluation Criteria1.3Evaluation Findings1.4Conclusions | | | |
| 2.0 • | DESCRIPTION OF ACTION 2.1 Description 2.2 Evaluation Criteria 2.3 Evaluation Findings 2.4 Conclusions | × | | · |
| 3.0 | IDENTIFICATION OF HAZARDS 3.1 Description 3.2 Evaluation Criteria 3.3 Evaluation Findings 3.4 Conclusions | · | | |
| 4.0 | HAZARDS ANALYSIS 4.1 Description 4.2 Evaluation Criteria 4.3 Evaluation Findings 4.4 Conclusions | | | |
| 5.0 | CONSEQUENCES OF ACCIDENTS 5.1 Description 5.2 Evaluation Criteria 5.3 Evaluation Findings 5.4 Conclusions | | | |
| 6.0 | CONTROLS 6.1 Description 6.2 Evaluation Criteria 6.3 Evaluation Findings 6.4 Conclusions | | | |

Lockheed Idaho Technologies Company

Idaho Falis, ID 83415

i

APPENDICES

| APPENDIX 1 | CHARTER OF SAFETY ANALYSIS, INDEPENDENT REVIEW TEAM |
|------------|---|
| PPENDIX 2 | INDEPENDENT REVIEW TEAM MEMBERS |
| APPENDIX 3 | GUIDANCE FOR PREPARATION OF SAFETY ASSESSMENTS |
| APPENDIX 4 | COMPILATION OF IRT COMMENTS AND CONCERNS |
| APPENDIX 5 | QUALITY ASSURANCE PLAN |

Idaho Technologies Company

 \bigcirc

P.O. Box 1625

÷

EXECUTIVE SUMMARY

Description

This Safety Evaluation Report (SER) documents the results of a third tier independent review of the Safety Assessment (SA) document titled "A SAFETY ASSESSMENT OF ROTARY MODE CORE SAMPLING IN FLAMMABLE GAS SINGLE SHELL TANKS: HANFORD SITE, RICHLAND, WASHINGTON " (RMCS) for the Hanford Tank Waste Remediation System (TWRS) Program tank farms. The independent review was performed to assess compliance of the RMCS SA with the requirements of DOE Order 5480.23, DOE-STD-3009-94, DOE-STD-3011-94, "Interim Guidance for Preparing Safety Assessments", (REV 2, March 6, 1992), developed by SAIC for DOE as directed by a letter from Leo Duffy in a memorandum dated May 6, 1992 (Appendix 3), and DOE-STD-1027-92.

The SA is intended to address the safety basis for each of the required elements associated with the installation, operation, and removal of a rotary mode core sampling (RMCS) device in Flammable Gas Watch List (FGWL) Single-Shell tanks (SSTs) located within the 200 Area in the Hanford Site, Richland, Washington. Potential hazards associated with the proposed activities are identified and evaluated systematically. Potential accident cases that could result in radiological or toxicological gas releases are identified and analyzed, and their consequences analyzed. Administrative controls, procedures, and necessary design changes to eliminate or reduce the potential hazards are also identified.

The review was performed by an Independent Review Team (IRT) at the Idaho National Engineering Laboratory (INEL) and in accordance with a charter issued by DOE-RL (Appendix 1). The charter specified the review criteria and the rules of independence and qualifications of the reviewers. The list of qualified reviewers is shown in Appendix 2 together with their respective qualifications.

Evaluation Criteria

The independent review team used the following six broad performance standards to guide their review: accuracy, completeness, traceability, consistency, readability, and functionality. The primary focus of the review was to compare the draft document against the governing requirements documents:

DOE Order 5480.23, Nuclear Safety Analysis.

DOE-STD-1027-92, "Hazard Categorization and Accident Analysis Techniques for Compliance with DOE Order 5480.23, Nuclear Safety Analysis."

DOE- STD-3009-94, "Preparation Guide for U.S. Department of Energy Non-reactor Nuclear Facility Safety Analysis Reports."

DOE-STD-3011-94 "Guidance for preparation and Submittal of Basis Interim Operation (BIO) for DOE Nonreactor Nuclear Facilities."

lockheed Idaho Technologies Company

P.O. Box 1625

Idaho Falls, ID 83415

"Interim Guidance for Preparing Safety Assessments", (REV 2, March 6, 1992), developed by SAIC for DOE as directed by a letter from Leo Duffy in a memorandum dated May 6, 1992 (Appendix 3).

Evaluation Findings

An evaluation summary of the final version of revision 0 of the RMCS SA and the final draft of the IOSRs as received from WHC is presented. The evaluation entailed assessment of the modifications to the documents as agreed upon during the July 2, 1996 meeting to resolve the identified IRT concerns. All open issues and areas of concern have been clarified, addressed and resolved for both the SA and associated IOSR documents. There remains no open issues of concern to the IRT after reviewing the changes incorporated into the final document revision as reviewed. A copy the Review Comment Record sheets used in the July 2, 1996 meeting for closing all open issues is included as Appendix 4.

Conclusions

The IRT recommends approval of the SA titled "A SAFETY ASSESSMENT OF ROTARY MODE CORE SAMPLING IN FLAMMABLE GAS SINGLE SHELL TANKS: HANFORD SITE, RICHLAND, WASHINGTON " (RMCS) with the provision that any changes to the operations or the equipment as described in the document should be followed by a careful USQ assessment process.

Idaho Technologies Company

P.O. Box 1625

1.1 Description

Section 1 describes the action that requires the SA. It provides a discussion of the activity, and the action required by the purpose of the SA. The action for Rotary Mode Core Sampling is well described together with it's identified needs. Alternative courses for sampling and the consequences of "no action" are described and assessed.

1.2 Evaluation Criteria

"Interim Guidance for Preparing Safety Assessments", (REV 2, March 6, 1992), developed by SAIC for DOE as directed by a letter from Leo Duffy in a memorandum dated May 6, 1992 (Appendix 3).

1.3 Evaluation Findings

There are no open issues with respect to the RMCS SA SECTION 1

1.4 Conclusions

This section adequately describes Rotary Mode Core Sampling, alternative methods, and consequences of "no action".

2.0 SECTION 2 - DESCRIPTION OF ACTION

2.1 Description

Section 2 identifies feasible alternatives to implement Rotary Mode Core Sampling, and establishes the criteria for evaluating alternative modes. The alternative modes are assessed, and the RMCS is justified. Maps, layouts, drawings and procedures are provided to identify hazard elements that may be encountered during RMCS.

2.2 Evaluation Criteria

DOE- STD-3009-94, DOE-STD-3011-94, "Interim Guidance for Preparing Safety Assessments", (REV 2, March 6, 1992), developed by SAIC for DOE as directed by a letter from Leo Duffy in a memorandum dated May 6, 1992 (Appendix 3), and DOE Order 5480.23



2.3 Evaluation Findings

There are no open issues with respect to the SA Section 2

2.4 Conclusions

Section 2 of the RMCS adequately meets the requirements of the evaluation criteria.

3.0 SECTION 3 - IDENTIFICATION OF HAZARDS

3.1 Description

Section 3 of the SA clearly identifies all potential hazards related to the RMCS activities. The functional analysis of the RMCS activities, environmental materials fabrication, degradation processes are addressed. This section also addresses inspection and monitoring consideration of equipment and facilities used in RMCS processes. Hazards identification is based on task analyses related to the installation, operation, and removal of equipment and materials required for RMCS.

3.2 Evaluation Criteria

DOE- STD-3009-94, DOE-STD-3011-94, "Interim Guidance for Preparing Safety Assessments", (REV 2, March 6, 1992), developed by SAIC for DOE as directed by a letter from Leo Duffy in a memorandum dated May 6, 1992 (Appendix 3), and DOE Order 5480.23

3.3 Evaluation Findings

There are no open issues with respect to the SA Section 3

3.4 Conclusions

Section 2 of the RMCS adequately meets the requirements of the evaluation criteria.

Idaho Technologies Company

P.O. Box 1625

4.1 Description

Section 4 provides hazards analyses for hazards identified in Section 3 as related to their potential severity including potential for occurrence. Environmental, materials, fabrication, degradation processes, inspection and monitoring practices as they could impact failure modes of system components are considered.

4.2 Evaluation Criteria

DOE- STD-3009-94, DOE-STD-3011-94, "Interim Guidance for Preparing Safety Assessments", (REV 2, March 6, 1992), developed by SAIC for DOE as directed by a letter from Leo Duffy in a memorandum dated May 6, 1992 (Appendix 3), and DOE Order 5480.23

4.3 Evaluation Findings

There are no open issues with respect to the SA Section 4

4.4 Conclusions

Section 4 of the SA adequately meets the requirements of the criteria listed above.

5.0 SECTION 5 - CONSEQUENCES OF ACCIDENTS

5.1 Description

Section 5 describes accident consequences and considers safety and system effects. It also describes radiological and toxicological consequences of accidents to both the workers and the public.

5.2 Evaluation Criteria

DOE- STD-3009-94, DOE-STD-3011-94, "Interim Guidance for Preparing Safety Assessments", (REV 2, March 6, 1992), developed by SAIC for DOE as directed by a letter from Leo Duffy in a memorandum dated May 6, 1992 (Appendix 3), and DOE Order 5480.23

Lockneed Idaho Technologies Company

P.O. Box 1625

idaho Falis, ID 83415

5.3 Evaluation Findings

There are no open issues with respect to the SA Section 5

5.4 Conclusions

Section 5 of the SA adequately meets the requirements of the criteria listed above.

6.0 SECTION 6 - CONTROLS

6.1 Description

Section 6 provides the identified and necessary controls required to perform the operations required by RMCS while ensuring that the risk acceptance guidelines are maintained.

6.2 Evaluation Criteria

DOE- STD-3009-94, DOE-STD-3011-94, "Interim Guidance for Preparing Safety Assessments", (REV 2, March 6, 1992), developed by SAIC for DOE as directed by a letter from Leo Duffy in a memorandum dated May 6, 1992 (Appendix 3), and DOE Order 5480.23

6.3 Evaluation Findings

There are no open issues with respect to the SA Section 6

6.4 Conclusions

Section 6 of the SA adequately meets the requirements of the criteria listed above.

7.0 SECTION 7 - CHECKLIST ITEMS

Section 7 of the RMCS SA provides a checklist of assumptions used as basis for developing the SA. Therefore, any parameters and/or limits that vary from the checklist should trigger re-evaluation of the conclusions made in this SA, probably via a USQ process.

Idaho Technologies Company

P.O. Box 1625

Idaho Falls, ID 83415

5.3 Evaluation Findings

There are no open issues with respect to the SA Section 5

5.4 Conclusions

Section 5 of the SA adequately meets the requirements of the criteria listed above.

6.0 SECTION 6 - CONTROLS

6.1 Description

Section 6 provides the identified and necessary controls required to perform the operations required by RMCS while ensuring that the risk acceptance guidelines are maintained.

6.2 Evaluation Criteria

DOE-STD-3009-94, DOE-STD-3011-94, "Interim Guidance for Preparing Safety Assessments", (REV 2, March 6, 1992), developed by SAIC for DOE as directed by a letter from Leo Duffy in a memorandum dated May 6, 1992 (Appendix 3), and DOE Order 5480.23

6.3 Evaluation Findings

There are no open issues with respect to the SA Section 6. Rotary Mode Core Sampling activities will be initially limited to the following four tanks BX-110, AX-101, AX-103 and TY-102 with the proviso that the requirements of the checklist in Section 7 of the SA are met.

6.4 Conclusions

Section 6 of the SA adequately meets the requirements of the criteria listed above.

7.0 SECTION 7 - CHECKLIST ITEMS

Section 7 of the RMCS SA provides a checklist of assumptions used as basis for developing the SA. Therefore, any parameters and/or limits that vary from the checklist should trigger re-evaluation of the conclusions made in the SA, probably via a USQ process.

Lockheed IdahoTechnologies Company

P.O. Box 1625

APPENDIX 1

DOE-RL REVIEW CHARTER

LOCKheed Idaho Technologies Company

Ĵ

P.O. Box 1625

DOCUMENT REVIEW PLAN

.

for

SAFETY ASSESSMENTS

for

SALT-WELL PUMPING AND ROTARY MODE CORE SAMPLING IN FLAMMABLE GAS TANKS

DOCUMENT REVIEW PLAN

SALT-WELL PUMPING AND RMCS IN FLAMMABLE GAS TANKS

SAFETY ASSESSMENTS

PURPOSE

The purpose of this Document Review Plan is to establish requirements and responsibilities for the Independent (3rd tier) review of the Safety Assessments for Salt-well Pumping and Rotary Mode Core Sampling (RMCS) in Flammable Gas Tanks per *Tank Waste Remediation Systems Procedure*, TWRS 08-01, "Safety Documentation Review and Approval".

BACKGROUND

A major safety issue at Hanford has been the generation, retention and sudden release of flammable gases in large waste tanks (500,000 gallons and larger) containing substantial amounts of sludge and saltcake. An Unreviewed Safety Question was declared and various double shell (DSTs) and single shell (SSTs) tanks were put on a "Watch List". Restrictions were placed on activities involving intrusion into the dome space and waste in tanks on the Flammable Gas Watch List (FGWL). As a result, neither Rotary Mode Core Sampling (RMCS) for characterization nor Salt-well pumping for interim stabilization is currently within the Authorization Basis for tanks on the FGWL. Several Tri-Party Agreement (TPA) milestones are affected by the restrictions on activities in FGWL tanks. Additionally, the potential exists for any SST to leak and contaminate the soil around the tank unless the tank is interim stabilized, since most SSTs are beyond their design life. In view of the fact that more than 60 SSTs are known or suspected to be leaking, there is substantial risk in delaying interim stabilization. Characterization also needs to be performed, especially to facilitate Privatization.

 \checkmark The purpose of the two SAs to be reviewed is to establish a sound technical basis for authorizing waste-intrusive activities for tanks on the FGWL, without undue risk to the public, workers and the environment. The immediate risk of a deflagration or detonation accident caused by waste-intrusive activities needs to be balanced against the long-term risk of large quantities of liquid waste releases to the environment, with the potential for reaching the surface and causing harm to workers and the public.

Substantial progress has been made in the study and understanding of the physical and chemical phenomena associated with the retention of flammable gases and their sudden release ("burp") since the declaration of a USQ and establishment of the FGWL. However, as evidenced by the recent imposition of controls related to flammable gases on all 177 tanks, understanding of the phenomena is not complete. Analyses are currently being performed with a view to removing controls for flammable gases on tanks which are not on the FGWL.

In view of the foregoing, the question of whether waste-intrusive activities should be authorized in FGWL tanks is a serious one. An Independent (3rd tier) Review by recognized experts not associated with RL-TWRS is, therefore, considered necessary, even though the activities under consideration do not involve permanent changes to the tank configuration and are of a short duration per tank.

REVIEW PLAN

SCOPE

The primary objective of this review is to ensure that the document complies with the criteria and guidelines established in DOE Order 4700.1, <u>Project Management</u> <u>System</u>, DOE Order 5400.5, <u>Radiation Protection of the Public and the Environment</u>, 40 CFR 835, <u>Occupational Radiation Protection</u>, DOE Order 6430.1A, <u>General Design</u> <u>Criteria</u>, and DOE Order 5480.23, <u>Nuclear Safety-Analysis Reports</u> (as augmented by DOE STD 3009) as applicable to the grading designated below. The review will determine if the criteria and guidelines for these limited duration activities have been followed; and, if the conclusion that the salt-well pumping and RMCS activities can be performed without undue risk to public, workers and the environment, is well supported and presented.

LIMITATIONS

This review covers all sections of the Safety Assessment. The activities for which authorization is being sought, viz., Salt-Well Pumping and RMCS, involve the introduction of equipment into the tank waste for a sort duration, and their removal at the conclusion of the activity. Due to the short-term nature of the proposed activities, only the following sections of a Safety Analysis Report (SAR) are included in this SA (deviation authority is Leo Duffy letter dated May 6, 1992):

- Executive Summary
- Description of Activity
- Hazard Identification
- Hazards Analysis
- Accident Analysis
- .Required Controls

REVIEW TEAM

The DOE-RL TWRS Management Systems Division (MSD) with the concurrence of the Environmental, Safety and Health Division of the Richland Operations Office (ES&H) will commission, coordinate, and support the Independent Review Team (IRT) review of the SAs. The IRT shall report to the MSD through the IRT Chairman. The MSD, as an independent oversight division of TWRS, has no responsibility for tank farm operations and did not draft any portion of the SAs.

IRT members must be experienced in one or more of the safety related subject matters addressed in the SAs. They must have sufficient experience with safety analyses similar to the SAs to be considered industry experts. Members are qualified for the panel based on their industry experience and backgrounds. In particular, the following specific experience is required in the IRT (though not required of each IRT member):

- hazard identification, methodology, and assessment, including event trees
- hazard classification as described in DOE Order 5480.23, "Safety Analysis Reports"

- hazard and accident analysis format and content, as stipulated in DOE Order 5480.23, "Safety Analysis Reports", and DOE-STD-3009.
- accident selection from hazard analysis
- radiological and toxicological accident assessment and methodology.
- knowledge of hazards associated with the storage and transportation of radioactive and mixed wastes at Defense nuclear facilities

Each IRT member participating in review of the SAs should have the following minimum pualifications:

- Minimum Bachelor of Science degree in engineering/physical science (preferably Masters)
- Ten to Twenty years experience in the commercial or federal nuclear industry or non-reactor nuclear industry. A minimum of 5 years involvement with deterministic and/or probabilistic accident safety analyses.

• Knowledge in one or more of the following areas:

- Heat transfer
- Chemistry/chemical reactions
- Thermodynamics
- HVAC
- Criticality
- Source term dispersion mechanisms, including airborne release fractions, release rates, respirable fractions, etc.
- structural analysis (concrete/steel), including seismic

A chairman of the IRT review team will be appointed by MSD with the concurrence of DOE-RL ES&H.

Proposed members of the IRT may be suggested by DOE-RL or DOE-HQ. However, the IRT chairman is under no obligation to follow these suggestions. Selection of review team members are subject to the concurrence of MSD and DOE-RL ES&H. The review chairman shall submit individual qualification statements for each team member to the MSD for DOE-RL review and concurrence.

The IRT chairman shall insure that each reviewer is independent of:

- project activities sponsoring the document;
- project activities that result in direct involvement with document development, and;
- other reviews of the document

REVIEW ASSIGNMENTS

The IRT chairman shall identify the scope and extent of each individual reviewer's review assignment(s).

SCHEDULE

The substantive independent review shall be completed and all review comments shall be transmitted to MSD within two weeks following actual receipt of the document by the IRT chairman. For planning purposes, the Salt Well Pumping SA will be available for third tier review on or about April 15th. The RMCS SA will be available on or about March 20th. The IRT chairman shall be allowed 10 working days from the receipt of the revised SA to verify final comment closure and issue a draft Safety Evaluation Report (SER) to MSD for administrative review. The final SER shall be provided to MSD within two working days of receipt of MSD's administrative concurrence with the draft SER.

GENERAL REQUIREMENTS

All review comments are to be documented on the Review Comment Record (RCR) Form (Attachment A or electronic Form WEF Oll on HLAN). Satisfactory disposition of each individual reviewer's comments shall be confirmed independently by the individual reviewer's signature or initials on the disposition record.

The IRT shall conduct one round of comment closure reviews following submission of the initial IRT review comments. This review shall evaluate the acceptability and adequacy of each proposed comment closure. A written response shall be provided MSD identifying all disposition comments found to be unacceptable by the IRT and the technical basis for reject. Final closure of comment responses which require a modification of SA text shall not occur until the IRT chairman verifies that the required text modification(s) have been incorporated into the SA and have been found acceptable (redline/strikeout verification of SA is allowed).

The IRT chairman shall submit a final SER, which at a minimum, shall include the material content identified in attachment B. A separate SER shall be produced for each SA reviewed. Additionally, each SER shall include one copy of the SA as last reviewed by the IRT for the purpose of establishing a baseline document control.

The IRT chairman shall submit a quality assurance plan to MSD which details the method of review, conflict/comment resolution process, reviewer qualifications and review assignment(s), administrative controls, and review criteria (see Safety Assessment Checklist below).

REVIEW CRITERIA

The review criteria have been established by the guidance and requirements provided in the aforementioned DOE Orders (see SCOPE SECTION), DOE guidance, and related federal environmental, safety, and health laws. A Safety Assessment Checklist (Attachment C) has been developed for optional use to aid in addressing the applicable criteria. Major elements of the established criteria consist of description of the safety analysis/evaluation process, description of the hazard categorization, accident analysis and risk acceptance, safety equipment and features, safety classification, and general document quality. The IRT chairman may modify Attachment C or develop an independent checklist as part of the IRT quality assurance plan.

REFERENCES

- 1. Tank Waste Remediation System Procedure 08-01, <u>Safety Documentation</u> <u>Review and Approval</u>, November 1995.
- 2. DOE Order 4700.1, Project Management System.
- 3. DOE Order 6430.1A, <u>General Design Criteria</u>.
- 4. WHC-CM-4-46, Nonreactor Nuclear Facility Safety Manual.

| | | | | | | | | • | |
|-----------------------------|--|---|----------------------|------------|-------------|---------------------------------------|----------|------------------|------------|
| - | | | | | 1. D | ate | 2. | Review No. | |
| | | MENT RECORD (RCR) | | | 3. P | roject No. | 4. | Page 1 of | 1 |
| 5. Document Num | mber(s)/Title(\$) | 6. Program/Project/ 7 Building Number | '. Reviewer | P | | 8. Organization/Group | | 9. Location/F | hone |
| . Comment Subm | nittal Approval: | 10. Agreement with indicat | ed comment | dispositio | en(s) | 11. CLOSED | | L: | • |
| Organization | Manager (Optional) | Review Date | er/Point of | Contact | | Date | Rev | iewer/Point of (| Contact |
| | | Author | /Originator | · | | | Aut | hor/Originator | |
| 12. 13. Comment a resolve t | ment(s)/Discrepancy(s) (Provid and detailed recommendation of the discrepancy/problem indica | e technical justification for the the action required to correct/ | 14. Hold Point | 15. Disp | position | (Provide justification i | f NOT | accepted.) | 16. Sta |
| | | | | | | | · | | |
| | ····· | | | <u> </u> | | | | | + |
| | · · | · · · · · · · · · · · · · · · · · · · | | | | · · · · · · · · · · · · · · · · · · · | | | + |
| | | ······································ | | | | ····· | | | |
| | <u></u> | | | | <u> </u> | · | <u>.</u> | _ \ | |
| | | - | | | | | | | + |
| | | · · · · · · · · · · · · · · · · · · · | | | | · · · · · · · · · · · · · · · · · · · | | | |
| | | · | | | | | | · | <u> </u> |
| | | | | | | | | | ╉── |
| | | | | | | | <u> </u> | | 1 |
| | · · · · · · · · · · · · · · · · · · · | | | | | | | | \square |
| | | | | | | | | | ╂ |
| | | ······ | | | | | | | _ |

A-6400-090.1 (03/92) WEF011

| TITLE: REVIEW AND APPROVAL OF NUC SAFETY DOCUMENTS | LEAR | Number: Effective Date: Exhibit: | RLP 5480.23 DRAFT H 8.1 - sht 1 of 2 | | | | | | |
|--|--|---|---|--|--|--|--|--|--|
| | | Page: | 16 of 43 | | | | | | |
| E – Recommended Fo | EXHIBIT 8.1 - Recommended Format and Content of SERs - | | | | | | | | |
| Overall content of the SER shall b DOE 5480.23. An SER being written of SAR development, USQ, or JCO s related Safety Documentation. An S attachment to a formal DOE transmi DOE transmittal. | Overall content of the SER shall be based on a graded approach as described in DOE 5480.23. An SER being written in response to an individual phase or stage of SAR development, USQ, or JCO shall be commensurate with the scope of the related Safety Documentation. An SER may be in the form of a formal report, an attachment to a formal DOE transmittal, or may be part of the text of a formal DOE transmittal. | | | | | | | | |
| In general, an SER is expected to f | ollow the form | at and conten | t provided below: | | | | | | |
| a. COVER/CONCURRENCE PAGES signatures. Non-concurr review team members sho differing opinions shoul | - Title o rence or diffe ould be indica d be documente | f document ering profess ated (non-cor d in an Appen | and concurrence ional opinion by currences and/or dix to the SER). | | | | | | |
| b. EXECUTIVE SUMMARY (option overall summary of the f this section is a crucial the DOE review and theref SER is of relatively sho summary may not be warra | hal) - This sec acility and th part of the SE ore should be ort and of lim nted. | tion is inten ne associated ER in presenti largely self c nited complex | ded to provide an risks. As such, ing the results of contained. If the ity, an executive | | | | | | |
| Reason for change, Brief description o Description of DOE Important findings Description of additany restrictions on A statement of adeq non-concurrence or Recommendations of a Facility can operate safety of the public | revision, or n f facility, in review process and conclusion tional controls operations uacy. Include differing prof acceptability, safely, witho c, workers, an | ew document cluding histo s , conditions discussion o essional opin including a s ut undue risk d the environ | ry if pertinent for approval, and f any team member ion tatement that the to the health and ment. | | | | | | |
| c. Table of Contents | | • . | | | | | | | |
| d. Remaining sections of the author of SER, typical subject Safety Document. review team is typically first chapter/topic. T chapter/topic: | e SER should be y reflecting Description presented as he following | e based on for the chapters of the revie part of the should be ad | mat determined by /sections of the w methodology and discussion of the dressed for each | | | | | | |
| Brief description o Presentation and di Conclusions and con activity or operati of facility worke | f intent and c scussion of fi nditions for a on in terms of rs, co-located | ontent of the ndings pproval (acco risk to the d workers, a | chapter/topic eptability of the health and safety and the public). | | | | | | |

r

()

| Recommended Format and Content of SERs (Cont'd) - Include any non-concurrence or differing professional opinion review team members (non-concurrences and/or differing opinio should be documented in an Appendix). Summary of any conditions for approval, including: Clarification of the safety basis as described in the Safe Document; Requirement for preparation and submittal of additional analys or data; Commitment to system design changes or additional operation limitations; Additional programmatic or institutional commitments to certa administrative controls, including maintenance, testin surveillance, and training provisions; Changes and/or additions to OSR/TSR. Appendices Acronyms Review team members and resumes or biographies Review plan/review basis Team member concurrence/non-concurrence sheets (team members w have differing professional opinion are expected to docume concerns and disagreements; these should be acknowledged in t conclusions section of the SER and in the Executive Summary). Attachments Review comments and resolutions (RCRs are adequate). | TITLE: R S | EVIEW AND APPROVAL OF NUCLEAR SAFETY DOCUMENTS RLP 5480. Effective Date: DRAFT H Exhibit: 8.1 - sht Page: 17 of 43 | 23 2 of 2 |
|--|---------------|--|--|
| Include any non-concurrence or differing professional opinion review team members (non-concurrences and/or differing opinio should be documented in an Appendix). e. Summary of any conditions for approval, including: Clarification of the safety basis as described in the Safe Document; Requirement for preparation and submittal of additional analys or data; Commitment to system design changes or additional operation limitations; Additional programmatic or institutional commitments to certa administrative controls, including maintenance, testin surveillance, and training provisions; Changes and/or additions to OSR/TSR. f. Appendices Acronyms Review team members and resumes or biographies Review plan/review basis Team member concurrence/non-concurrence sheets (team members w have differing professional opinion are expected to docume concerns and disagreements; these should be acknowledged in t conclusions section of the SER and in the Executive Summary). g. Attachments Review comments and resolutions (RCRs are adequate). | <u> </u> | - Recommended Format and Content of SERs (Cont'd) - | |
| e. Summary of any conditions for approval, including: Clarification of the safety basis as described in the Safe Document; Requirement for preparation and submittal of additional analys or data; Commitment to system design changes or additional operation limitations; Additional programmatic or institutional commitments to certa administrative controls, including maintenance, testin surveillance, and training provisions; Changes and/or additions to OSR/TSR. f. Appendices Acronyms Review team members and resumes or biographies Review plan/review basis Team member concurrence/non-concurrence sheets (team members w have differing professional opinion are expected to docume concerns and disagreements; these should be acknowledged in t conclusions section of the SER and in the Executive Summary). g. Attachments Review comments and resolutions (RCRs are adequate). | - | Include any non-concurrence or differing professional opi review team members (non-concurrences and/or differing o should be documented in an Appendix). | nion by pinions |
| Clarification of the safety basis as described in the Safe Document; Requirement for preparation and submittal of additional analys or data; Commitment to system design changes or additional operation limitations; Additional programmatic or institutional commitments to certa administrative controls, including maintenance, testin surveillance, and training provisions; Changes and/or additions to OSR/TSR. f. Appendices Acronyms Review team members and resumes or biographies Review plan/review basis Team member concurrence/non-concurrence sheets (team members w have differing professional opinion are expected to docume concerns and disagreements; these should be acknowledged in t conclusions section of the SER and in the Executive Summary). g. Attachments Review comments and resolutions (RCRs are adeguate). | e. | Summary of any conditions for approval, including: | |
| f. Appendices Acronyms Review team members and resumes or biographies Review plan/review basis Team member concurrence/non-concurrence sheets (team members w have differing professional opinion are expected to docume concerns and disagreements; these should be acknowledged in t conclusions section of the SER and in the Executive Summary). g. Attachments Review comments and resolutions (RCRs are adequate). | • | Clarification of the safety basis as described in the Document; Requirement for preparation and submittal of additional a or data; Commitment to system design changes or additional oper limitations; Additional programmatic or institutional commitments to administrative controls, including maintenance, t surveillance, and training provisions; Changes and/or additions to OSR/TSR. | Safety nalyses ational certain esting, |
| Acronyms Review team members and resumes or biographies Review plan/review basis Team member concurrence/non-concurrence sheets (team members w have differing professional opinion are expected to docume concerns and disagreements; these should be acknowledged in t conclusions section of the SER and in the Executive Summary). g. Attachments Review comments and resolutions (RCRs are adequate). | f. | Appendices | |
| g. AttachmentsReview comments and resolutions (RCRs are adequate). | | Acronyms Review team members and resumes or biographies Review plan/review basis Team member concurrence/non-concurrence sheets (team memb have differing professional opinion are expected to d concerns and disagreements; these should be acknowledged conclusions section of the SER and in the Executive Sumn | pers who locument l in the lary). |
| Review comments and resolutions (RCRs are adequate). | g. | Attachments | |
| | · • | Review comments and resolutions (RCRs are adequate). | |

()

SAFETY ASSESSMENT CHECKLIST

Some General Considerations, Merits, and Reservations

The attached checklist, developed by the Department of Energy (DOE) Richland Operations Office (RL) Tank Waste Remediation System (TWRS) Program Integration Division (PRI), and Westinghouse Hanford Company (WHC) Safety Analysis and Engineering (SA&E) is intended to be used as an aid in preparing and reviewing Safety Assessments (SAs) prepared by TWRS contractor/participants pursuant to criteria and guidelines established in DOE Order 4700.1, Project Management System, DOE Orders 6430.1A, General Design Criteria, and WHC-CM-4-46, Nonreactor Nuclear Facility Safety Manual, and are based on many DOE Orders, DOE guidance, and related federal environmental, safety, and health laws.

The checklist provides columns for "yes," "no," and not applicable ("N/A") responses. If desired, notes on document adequacy and other comments can also be entered. Generally, the questions are phrased so that a "yes" answer is preferable to a "no" answer. Note that not all questions will apply to all SAs; the questions should be adapted to the particular circumstances presented by the proposed action. Consider also the use of the "graded approach"- depending on the extent and potential hazard of a given facility, some reduction in coverage under each area or combination of subjects may be acceptable in the SA.

In providing this product, RL-TWRS recognized that a document checklist, while it may have value to the contractor/participants, will also have limitations. A carefully applied checklist may help experienced preparers and reviewers avoid overlooking required or recommended items. It may help preparers and reviewers identify needed analyses and discussions. It may provide a record of an internal review. On the other hand, each DOE proposal presents unique circumstances and potential impacts, and the preparation and review of SAs does not reduce to a single formula or checklist. No document checklist can be universally comprehensive or complete. No document checklist should be relied upon as the only way to build quality into a DOE safety analysis process. No checklist can be a substitute for the original laws, regulations, and guidance documents. No checklist can ensure that the SA will be adequate under, and in full compliance with, DOE Order 5480.23 and associated standards. The RL-TWRS contractors/participants should also bear in mind that addressing generic items on a checklist alone may not lead to a sufficiently rigorous analysis of potential impacts for a specific safety issue. Further, checklist items are not always of equal importance or weight. If a known, obvious hazard is not addressed, a SA is generally inadequate; however, if full accident analyses are omitted, it is not a critical matter in terms of document adequacy. In short, a checklist should not be a replacement for good judgment.

Finally, this SA checklist is not intended to promote the rote generation of standardized documentation, nor promote an ethic focused on minimal compliance with DOE Orders and standards. It cannot induce or measure whether the methods of safety analysis are systematically applied or whether resources are appropriately allocated, or whether management/decision-makers utilized SA results in decisions and whether those decisions result in the protection of the worker, public, and environment. Generation of safety documents will not by itself foster daily excellence in worker, public, and environmental safety. In the long run, the focus should be on the ultimate "product" of the safety analysis process: quality decisions, stewardship of resources, and safety to the total community.

DOE RL-TWRS

February 16, 1996

| Document Number: Office/Phone: Document Date: Tate: Ves No N/A Page Adequacy Evaluation and Comments SUMMARY Does the summary address the entire SA? (Consideration by the reviewers must be given to status in the project life cycle - DOE Order 4700.1) Image: Consequences of the segin basis accident (DBAs), and risk-dominant accident secarios that have been analyzed, and the measures taken to eliminate, control, or mitigate the consequences of these accidents? (Does this include a discussion or statement specific to criticality?) Image: Control or mitigate the consequences of these accidents? (Does this include a discussion or statement specific to criticality?) Image: Control or mitigate the consequences of these accidents? (Does this include a discussion or statement specific to criticality?) Image: Control or mitigate the consequences of these accidents? (Does this include a discussion or statement specific to criticality?) Image: Control or mitigate the facility, including its physical and institutional boundaries, distinguishing if from other DOE facilities or operations outside the scope of the analysis and clearly define those external functions, such as utilities or external support organizations. Image: Control or statement of purpose and need discuss: Image: Control or statement of purpose and need discuss: Image: Control or statement of purpose and need discuss: Image: Control or statement of purpose and need discussion or consequences, or otherwise, alter the ext | Document Title: Re | eviewed By: | | | | | |
|--|---|--------------|----|-----|------|---------------------------------------|---|
| Document Date: Date: Ves No N/A Page Adequacy Evaluation and Comments SUMMARY Does the summary address the entire SA? (Consideration by the reviewers must be given to status in the project life cycle - DOE Order 4700.1) Image: Consideration by the reviewers must be given to status in the project life cycle - DOE Order 4700.1) Image: Consideration by the reviewers must be given to status in the project life cycle - DOE Order 4700.1) Image: Consideration by the reviewers must be given to status in the project life cycle - DOE Order 4700.1) Does the summary discuss the design basis accidents (DBAs), and risk-dominant accident scenarios that have been analyzed, and the measures taken to eliminate, control, or mitigate the consequences of these accidents? (Does this include a discussion or statement specific to criticality?) Image: Consequences of these accidents? Does the summary describe the potential hazards that have been addressed? Image: Consequence of the analysis and clearly define the facility, including its physical and institutional boundaries, distinguishing it from other DOE facilities or operations outside the scope of the analysis and clearly define those external support organizations. Image: Consequence of the statement of purpose and need discuss: any new types of accidents which were not possible before? Image: Consequences, or otherwise, alter the scenario for accidents merious with were not possible before? | Document Number: Of | ffice/Phone: | | | | | • |
| Yes No | Document Date: Da | ate: | | | | <u></u> | |
| SUMMARY Does the summary address the entire SA? (Consideration by the reviewers must be given to status in the project life cycle - DOE Order 4700.1) Does the summary discuss the design basis accidents (DBAs), and risk-dominant accident scenarios that have been analyzed, and the measures taken to eliminate, control, or mitigate the consequences of these accidents? (Does this include a discussion or statement specific to criticality?) Does the summary describe the potential hazards that have been addressed? Does the summary clearly define the facility, including its physical and institutional boundaries, distinguishing it from other DOE facilities or operations outside the scope of the analysis and clearly define those external functions, such as utilities or external support organizations. In projects which modify or interface with existing facilities, does the statement of purpose and need discuss: any new types of accidents which were not possible before? any synergistic effects? any increased probabilities or consequences, or otherwise, alter the scenario for accidents reviewel | | Yes | No | N/A | Page | Adequacy Evaluation and Comments | |
| Does the summary address the entire SA? (Consideration by the reviewers must be given to status in the project life cycle - DOE Order 4700.1) Does the summary discuss the design basis accidents (DBAs), and risk-dominant accident scenarios that have been analyzed, and the measures taken to eliminate, control, or mitigate the consequences of these accidents? (Does this include a discussion or statement specific to criticality?) Does the summary describe the potential hazards that have been addressed? Does the summary clearly define the facility, including its physical and institutional boundaries, distinguishing it from other DOE facilities or operations outside the scope of the analysis and clearly define those external functions, such as utilities or external support organizations. In projects which modify or interface with existing facilities, does the statement of purpose and need discuss: any new types of accidents which were not possible before? any synergistic effects? any increased probabilities or consequences, or otherwise, alter the sscenario for accidents or exter the sscenario for accidents or exter the sscenario for accidents or exter the sscenario for accidents or external functions, content projects which mode the facility of the facility of the facilities or consequences, or otherwise, alter the sscenario for accidents or external functiones content of the scenario of the projects of the facilities or consequences of the facilities or consequences of accidents which were not possible before? any synergistic effects? | | SUMMA | RY | | • | | |
| Does the summary discuss the design basis accidents (DBAs), and risk-dominant accident scenarios that have been analyzed, and the measures taken to eliminate, control, or mitigate the consequences of these accidents? (Does this include a discussion or statement specific to criticality?) Does the summary describe the potential hazards that have been addressed? Does the summary clearly define the facility, including its physical and institutional boundaries, distinguishing it from other DOE facilities or operations outside the scope of the analysis and clearly define those external functions, such as utilities or external support organizations. In projects which modify or interface with existing facilities, does the statement of purpose and need discuss: any new types of accidents which were not possible before? any synergistic effects? any increased probabilities or consequences, or otherwise, alter the scenario for accidents nerviously | Does the summary address the entire SA? (Consideration by the reviewers must be given to status in the project life cycle - DOE Order 4700.1) | n r | | | | · · · · · · · · · · · · · · · · · · · | |
| Does the summary describe the potential azards that have been addressed? | Does the summary discuss the design basis accidents (DBAs), and risk-dominant accident cenarios that have been analyzed, and the neasures taken to eliminate, control, or mitig he consequences of these accidents? (Does th nclude a discussion or statement specific to criticality?) | ate is | · | | | | |
| Does the summary clearly define the facility, ncluding its physical and institutional woundaries, distinguishing it from other DOE acilities or operations outside the scope of the malysis and clearly define those external unctions, such as utilities or external support organizations. Image: Comparison of the malysis and clearly define those external unctions, such as utilities or external support organizations. n projects which modify or interface with existing facilities, does the statement of purpose and need discuss: Image: Comparison of the facility of the facility of the facilities of the | Does the summary describe the potential mazards that have been addressed? | | | | | • | |
| In projects which modify or interface with existing facilities, does the statement of purpose and need discuss: any new types of accidents which were not possible before? any synergistic effects? any increased probabilities or consequences, or otherwise, alter the scenario for accidents previously | Does the summary clearly define the facility, ncluding its physical and institutional boundaries, distinguishing it from other DOE facilities or operations outside the scope of the analysis and clearly define those external functions, such as utilities or external support organizations. | c | | | | | |
| any new types of accidents which were not possible before? | In projects which modify or interface with existing facilities, does the statement of purpo and need discuss: | ose . | | | | | |
| any synergistic effects? any increased probabilities or consequences, or otherwise, alter the scenario for accidents previously | any new types of accidents which were not possible before? | | | | | | |
| any increased probabilities or consequences, or otherwise, alter the scenario for accidents previously | any synergistic effects? | | | | | | |
| analyzed? | any increased probabilities or consequences, or otherwise, alter th scenario for accidents previously analyzed? | ıc | | | | | |

8

| | Yes | No | N/A | Page | Adequacy Evaluation and Comments | | |
|--|-------|--------|--------|---------|-------------------------------------|--|--|
| DESCRIPTION OF THE SAFET | Y ANA | LYSIS/ | EVALU. | ATION I | PROCESS | | |
| The safety analysis process is iterative and is used to establish and maintain the safety basis throughout the project life cycle. Does the safety analysis process used: | | | 1 | - | | | |
| describe the design to systematically identify necessary systems, structures, and components (SSCs) that prevent or mitigate unacceptable releases of radioactive and/or hazardous material; | | | | | | | |
| analyze a new or existing facility to verify that the design adequately performs the identified safety functions during normal and accident conditions; | | | | | | | |
| systematically determine the safety function(s) of SSCs; and | | | | | | | |
| identify as non-binding, preliminary technical safety requirements and/or administrative controls as possible mitigations to consequences | | | - | | | | |
| Does the SA aid management in insuring that major safety considerations (technical and compliance) are incorporated early into the project life cycle? | | | | | | | |
| Does the SA evaluate the functional design criteria (FDC) and is it issued concurrently with the conceptual design report (CDR) as part of the budget validation package for the project? | | | | | | | |
| Is the need for new safety documentation and modifications to the existing safety documentation identified in the SA? | | | | | | | |
| DESCRIPTION OF THE HAZARDS | | | | | | | |
| Was the operation broken down into components or modes to be analyzed? This can be SSCs, processes, operating modes, etc. | | | | | | | |
| Were proposed non-binding safety class SSCs and their basis for designation determined? (i.e., SEL) | | | | | | | |
| Were the hazards and/or hazardous conditions present described? The following descriptions should be considered: | | | | | | | |

| | Yes | No | N/A | Page | Adequacy Evaluation and Comments |
|--|---------|----|-----|------|-------------------------------------|
| Energy sources (these sources include energy forms which could contribute to the release and exposure of radiological and/or otherwise hazardous materials resulting in biological damage) - natural phenomenon, electrical, gravitational, pressure, ionizing radiation, non- ionizing radiation, linear, rotational, potential, kinetic, radiant, fire/explosion, criticality, airborne acoustic radiation, noise, static magnetic fields, subfrequency static electric fields, etc.) | | | | | |
| Hazardous Materials - those materials that are toxic, explosive, flammable, corrosive, or otherwise physically or biologically threatening. | | | - | | |
| <u>Typical Classes of Hazards</u> - fire/explosion, loss of containment/confinement, injury to personnel, criticality, interfaces between steps, components, systems, and nearby facilities, maintenance, natural phenomena, exposure to radiation and/or other hazardous materials, operational failures and their impact on the facility. | | | | | |
| <u>Consequence</u> - state in qualitative terms, the impacts to the health and safety of the workers, members of the public, and the environment. | | | | | |
| Engineered Features and Administrative Controls - identify and document any engineered features (providing safety function) and/or administrative controls for which credit may be taken to detect, prevent, reduce the frequency of, control, and/or mitigate the stated consequences. | с, (| | | | - |
| Does the SA identify the need for additional, detailed analysis for each item identified in the hazards identification process above using the following: | | | | | |
| Designate scenarios that have an annual probability of less than 10 ⁶ as "beyond extremely likely". | | | | | |
| Identify events with programmatic impact only or with no significant safety consequence. | | | | | · . |

 \bigcirc

| 1 | Yes | No | N/A | Page | Adequacy Evaluation and Comments |
|---|--------|------|--------|------|-------------------------------------|
| Identify credible accident scenarios which must be evaluated to assure that risks presented are adequately analyzed and managed. | | | | | |
| Segregate the remaining event sequences according to the frequency range. Within each category, group common types of events. Select the scenarios which are expected to result in the most significant consequences. These scenarios are selected for further evaluation in the accident analysis. | | | 1 | | |
| stage of facility design, and the types of hazards being analyzed. | | | | | |
| ACCIDENT ANALYS | IS AND | RISK | ACCEPT | ANCE | |
| For each event sequence: | | | | | |
| Have all postulated sequences of events which could lead to the most significant consequences been evaluated? | | | | | |
| Have bounding consequence values been identified? | | | | | |
| Have the quantity and characteristic of radionuclides and hazardous materials available for release been defined to acceptable standards and guidelines? | | | | | |
| Have all assumptions for the mechanisms and extent of release through the mitigating barriers been defined? | • | | | | |
| Have the potential consequences to onsite and offsite receptors been determined? | | | | | |
| Have the frequencies of the stated scenario event sequences been evaluated? | | | | | |
| Have the environmental impacts been determined? | | | | | |
| Has the adequacy of the design to effectively perform the identified function(s) been demonstrated? | | | | | |
| Has the acceptability of the risks (consequences) been evaluated? | | | | | |

()

| • | Yes | No | N/A | Page | Adequacy Evaluation and Comments | | | | |
|---|-----|----|-----|------|-------------------------------------|--|--|--|--|
| Has the accident analysis process, basis, and assumptions and conclusions been appropriately critiqued to ensure that the project is representative of the work to be performed? | | | 4 | | | | | | |
| SAFETY EQUIPMENT AND FEATURES | | | | | | | | | |
| Does the SA refer to safety class items and their design to withstand the effect and be compatible with, the environmental conditions associated with operation, maintenance, shutdown, testing, and accidents? | | | | | | | | | |
| Does the SA demonstrate how the design of the facility shall reduce the consequences of normal and DBA events by incorporating ALARA design concepts? | | | | | | | | | |
| Does the SA provide an assessment of the design against potential nuclear criticality? | | | | | | | | | |
| SAFETY CLASSIFICATION ¹ | | | | | | | | | |
| Have safety class designations been made in conjunction with the analysis of consequences from hazards, natural phenomena, and postulated accident scenarios? | | | | | | | | | |
| Have the safety class designations been assigned the highest applicable safety class? | | | | | | | | | |
| Has the safety classification criteria determination (at the system and major-structure level) been included in the preliminary safety documentation (e.g., the SA)? | | | | | | | | | |
| Have determinations for safety classification been based on the nature of the safety or environmental protection function? | -1 | | | | | | | | |
| Have determinations for safety classification been based on the requirements of DOE Order 6430.1A Section 1300-3.2 Safety Class Items? | | | | | | | | | |

¹ Applicable only as determined by current preliminary design requirements.

| | Yes | No | N/A | Page | Adequacy Evaluation and Comments |
|--|-----|----|-----|------|-------------------------------------|
| Safety class items are systems, components, and structures, including portions of process systems, whose failure could adversely affect the environment or the safety and health of the public. Specifically, safety class items are those SSCs, with the following characteristics: | | | | | |
| Those whose failure would produce exposure consequences that would exceed the guidelines in DOE Order 6430.1A, Section 1300-1.4, Guidance on Limiting Exposure of the Public, at the site boundary or nearest point of public access | | | | | |
| Those required to maintain operating parameters within the safety limits specified in the Operational Safety Requirements (OSR), Interim Safety Basis (ISB), Interim OSR (IOSR) during normal operations and anticipated operational occurrences | 、 | | | | |
| Those required for nuclear criticality safety | | | | | |
| Those required to monitor the release of radioactive materials to the environment during and after a DBA | | | • | | |
| Those required to achieve and maintain the facility in a safe shutdown condition | | | | | |
| Those that control the safety class items described above | | , | | | |
| DOE/TIC 11603, Rev. 1, presents examples of safety classification of plant systems, structures, and components in its appendices; however, for comparable sections in DOE/TIC 11603, Rev. 1, and DOE 6430.1A, the design criteria in DOE 6430.1A shall govern. | ., | | | | |
| Are the Safety class items subject to appropriately higher-quality design, fabrication, and industrial test standards and codes such as those specified in Section 0106, Regulatory Requirements, and Section 0109, Reference Standards and Guides, to increase the reliability of the item and allow credit to be taken for its capabilities in a safety analysis. Safety class items shall be designed to the American Society of Mechanical Engineers (ASME) Boiler and Pressure Vessel Code (Section III) or to other comparable safety-related codes and standards that are appropriate for the system being designed. | | | | | |

| | Yes | No | ·N/A | Page | Adequacy Evaluation and Comments |
|--|-----|----|------|------|-------------------------------------|
| Safety class and non-safety class items shall comply with Section 0140, Quality Assurance. The design of systems, components and structures that are not safety class items shall, as a minimum, be subject to conventional industrial design standards, codes, and quality standards. Failure of these items shall not adversely affect the environment or the safety and health of the public. In addition, their failure shall not prevent safety class items from performing their required functions. | | | | | |
| Does the design ensure that a single failure does not result in the loss of capability of a safety class system to accomplish its required safety functions? To protect against single failures, the design shall include appropriate redundancy and shall consider diversity to minimize the possibility of concurrent common-mode failures of redundant items. | | | | | |
| Safety class items shall be designed to withstand the effects of, and be compatible with, the environmental conditions associated with operation, maintenance, shutdown, testing, and accidents. The environmental capability of equipment shall be demonstrated by appropriate testing, analysis, and operating experience, or other methods that can be supported by auditable documentation, or a combination of these methods. | | | | | |
| Equipment qualification shall provide assurance that safety class items will be capable of performing required safety functions under DBA conditions. The qualification shall demonstrate that the equipment can at least perform for the period of time that its safety functions are required. Subsequent equipment failure, after its safety function is no longer required, may be allowable. | | | | | |
| Safety class items shall be designed to allow inspection, maintenance, and testing to ensure their continued functioning, readiness for operation, and accuracy. Ancillary equipment, such as pumps, blowers, motors, compressors, gear trains, and controls, shall be located in an area least likely to be contaminated. | | | | | |
| The design shall include provisions for periodic testing of monitoring, surveillance, and alarm systems. In addition, the design shall provide the capability to test periodically, under simulated emergency conditions, safety class items that are required to function under emergency conditions. | | | | | - |

Ð

| | Yes | No | N/A | Page | Adequacy Evaluation and Comments |
|---|---|----|-----|------|-------------------------------------|
| All system for which credit is taken to meet the criteria of DOE Order 6430.1A, Section 1300-1.4.2, Accidental Releases, shall be in-place testable in terms of pressure, filtration or removal efficiency, alarm capability, leak resistance, and the like. Safety class items shall be designed to be testable on a regular schedule. | | | 1 | • | |
| Structures, systems, and components that provide for nuclear criticality safety shall be designed as safety class systems and be capable of performing their criticality safety functions during and following design basis accidents and events. | | | | | |
| Risk and vulnerability analyses can be used to identify targets that are essential to ensure the operability of safety-class items and the security of critical programs or facilities. In addition, cost-benefit analyses can be conducted to identify efficient and cost-effective measures to meet site-specific safeguards and security requirements. Targets shall be prioritized so as to determine those to be afforded the greatest level of security in accordance with the graded safeguards and security approach. | | | | | |
| FORMAT, GENERAL DOCUM | FORMAT, GENERAL DOCUMENT QUALITY, USER-FRIENDLINESS | | | | |
| Is the SA written precisely and concisely, using plain language, and without jargon? | | | | | |
| Is the SI system of units used (with English units in parentheses) to the extent possible? | - | • | | | |
| If scientific notation is used, is an explanation provided? | | | | | |
| Does the SA use appropriate significant figures? | | | | | |
| Are the units consistent throughout the document? | | | | | |
| If regulatory terms are used, are they consistent with their regulatory definitions? | | | | | |
| Are visual aids used whenever possible to simplify the SA? | | | | | |
| Are technical terms defined? | | | , | | |
| Is the SA grammatically correct? | | | | | |
| Are abbreviations and acronyms defined the first time they are used? | | | | | |
| Is the use of abbreviations minimized to the extent practical? | | | | | |

 \bigcirc

| | Yes | No | N/A | Page | Adequacy Evaluation and Comments |
|---|-----|----|-----|------|-------------------------------------|
| Do the appendices support the content and conclusions contained in the main body of the SA? Is information in the appendix consistent with information in the main body of the SA? | | | | | |
| Is information in tables and figures consistent with information in the text and appendices? | | | - | • | |
| OVERALL CONSIDERATIONS/INCORPORATION OF SA VALUES | | | | | |
| If the Environmental Assessment (EA) adopts, in whole or in part, this SA document prepared by Westinghouse Hanford Company, has DOE independently evaluated the information used? | | | | 6 | |
| Does the SA appropriately use incorporation by reference? Is/are the incorporated document(s) up-to-date? | | | | | |
| Do the conclusions regarding potential impacts support the information and analyses presented in the EA? | | | | | |
| Although not required, will there be follow-up to determine if the analyses were accurate and that the mitigation measure are implemented? | | | | | |
| PROCEDURAL CONSIDERATIONS | | | | | |
| Does the SA make reference to a more comprehensive safety assessment/analysis as more definitive design information is available? | | | | | |

}

APPENDIX 2 SA IRT Members

| DOE: | R. McNulty, DOE-RL, a | and C. Noble, DOE-ID |
|-------|-----------------------|---|
| ID | NAME | QUALIFICATION & AREA OF EXPERTISE |
| BEI | G. A. Beitel | Ph.D Physics, Explosives, Waste Management |
| din | G. A. Dinneen | MS Mechanical Engineering/Nuclear Science, Safety Analysis |
| GUZ - | R. C. Guenzler | MS Civil Engineering, Structural Analysis |
| NM | N. Morcos | Ph.D Radiochemistry, Nuclear Chemistry, Radiological Risk Assessment, Safety Analysis, Commercial Nuclear Power, Commercial Radioisotope Production. |
| ŴJP | W.J. Prendergast | BS Chemical Engineering; Diplomat in Environmental Engineering; Professional Engineer; Plant Design and Operations Analysis. |
| KSM | K.S. Moor | MS Zoology; Environmental Monitoring and Assessment |

Idaho Technologies Company

 \bigcirc

P.O. Box 1625

APPENDIX 3

GUIDANCE FOR PREPARATION OF SAFETY ASSESSMENTS

Lockheed Idaho Technologies Company

Idaho Falls, ID 83415

· HAturs

United States Government

008 F 1310.3 - /**8-83:** EFG (07-60)

memorandum

DATE: MAY 6 1992 REPLY TO ATTN OF: EM-36:Calley:FTS 233-7417

SUBJECT: Guidance for Preparation of Safety Assessments

To: Manager, DOE Richland Field Office

Attached for your use is interim guidance for the preparation of safety assessments to support proposed actions to address unreviewed safety questions (USQs) and other priority safety issues related to high-level radioactive waste storage.

Historically, these safety assessments have been untimely and of inconsistent quality. In an effort to improve them, the EM staff has prepared the attached "Interim Guidance for Preparing Safety Assessments," Revision [2], dated March 6, 1992. This guidance was originally prepared in draft form in September 1991, and reviewed by the DOE Richland Field Office (RL), Westinghouse Hanford Company (WHC); Office of Environmental, Safety and Health (EH); and the Office of Nuclear Safety (NS). The views and comments of those organizations have been incorporated into the interim guidance. We view this as a dynamic document and plan to make whatever future changes are necessary to make it compatible with our experience in applying the guidance to the safety assessment preparation and approval process.

It is our intent that this guidance will provide the needed continuity in the preparation and review of safety assessments. You are directed to use this guidance in the preparation of all safety assessments related to highlevel waste tank safety. My staff contact for this action is Harry Calley (FTS 233-7417).

no anagina del manima calendaria finanziana di manima di manima di manima di manima di manima di manima di mani

Leo P. Duffy

Assistant Secretary for Environmental Restoration and Waste Management

Attachment

cc w/attachment: .R. Gerton, DOE-RL

4 ME A.

.

- ---

Department of Energy

RE COMMITMENT

CONTROL PHONE 5-8537 CONTROL NO

1008.0

ISSTORED IN

DISTAIBUTICHE

<u>mik</u> Erp Erd

OCC TSD

whid

DEP

2 Commitment Control

MAY 1 2 1992

Ruhland Operations Office

The off the last state of the s
INTERIM GUIDANCE

FOR

PRÉPARING SAFETY ASSESSMENTS

(

Developed for U. S. Department of Energy High-Lavel Radioactive Waste Tanks Task Force

Revision [2]

March 6, 1992

C. C. Herrington J. R. Fearring

Science Applications International Corporation

mana and and and

100

The second s

INTERIM GUIDANCE FOR PREPARING A SAFETY ASSESSMENT

INTRODUCTION

123456

As is indicated by the title, development of this paper, has continued toward improving the guidance for Safety Assessments. This revision of the "INTERIM GUIDANCE" incorporates new thinking that has developed and addresses the expectations of the SA reviewers who are ultimately responsible for concurrence or approval of the assessments. In addition past guidance related to addressing the Scope of the action addressed by the SA, has been expanded to more fully treat the need for completing an Engineering Evaluation of Alternatives prior to beginning the preparation of a SA.

The guidance information presented below is cryanized as follows:

- 1. "General Guidance" related to the overall document
- "Specific Guidance" organized according to the SA content topics which remain unchanged. They are:

| .0 | SCOPE |
|-----|---------------------------|
| .0 | DESCRIPTION OF ACTION |
| .0 | IDENTIFICATION OF HAZARDS |
| .0 | HAZARD ANALYSIS |
| . 0 | CONSEQUENCE OF ACCIDENTS |
| .0 | CONTROLS |

GENERAL GUIDANCE

.

In addition to being technically accurate and complete, the SA must be written with the audience in mind. The purpose of the SA is to convince the decision makers that the proposed course of action is necessary, and that you have logically and systematically identified and addressed the safety issues involved.

A Safety Assessment is not intended to convince the reviewer that the subject situation is the best action, activity or modification to accomplish its goals whether the goals be economic, production related or improving safety. Nor is a Safety Assessment intended to be an engineering study where various options are evaluated and the best selected. The engineering study addressing the options and the selection of the action, activity, or modification best suited to meet the needs of the program is entitled, an Engineering Evaluation of Alternatives. It is usually completed prior to the SA and referenced in the SA. A Safety Assessment is a safety analysis to support a temporary activity or to cover a small situation until the safety analysis/SAR is undated to envelop it. As such, a Safety Assessment has similar goals and should utilize an approach similar to that of a safety analysis/SAR. The goal is to systematically identify the hazards of a situation, describe and analyze measures taken to eliminate, control or mitigate identified hazards and to analyze potential accidents and their associated risks (DOE Order 5480.5 (1)).

When possible, the SA should be developed as a stand-alone document. When using references, avoid the use of internal memos which are not readily available to the reader. When possible, information necessary to support the conclusions presented should be attached as appendices.

SPECIFIC GUIDANCE

. 1

1.0 SCOPE

The section on SCOPE describes the action that requires the safety assessment (SA) to be written. The section is prefaced by a brief discussion of the EACKGROUND of the activity, telling why this particular subject requires attention (e.g., HLW tanks are generating hydrogen) and the action to be addressed by this SA [e.g., Means should be in place to ensure that the concentration of hydrogen in the tank remains lower than the lower flammability limit].

All of the information that identifies the need for the specific proposed action and defines the action itself, should be included in order to provide the full scope of consideration needed by reviewers in evaluating the SA. Alternative courses of action to accomplish the desired

result must be considered. The alternative of no action should be considered and addressed as appropriate. The methodology used in identifying alternatives and the bases for selecting the preferred course of action to be proposed in the SA should be documented at the time that an Engineering Evaluation of Alternatives (EEA) is prepared.

Ideally, the FFA should be completed prior to the preparation of the SA. The various sections of the EEA can then be used in developing related sections of the SA and the Environmental Assessment (EA) (See Attachment 1). An: effective EFA will contain information that is needed by the SA reviewers to judge the adequacy of the total SA. It will identify the problem to be addressed by the proposed action. It will demonstrate that the basic safety mechanisms related to the problem are understood and that specific safety mechanisms requiring controls have been identified. An effective FFA will present criteria for choosing among alternatives. Criteria related to the areas of Safety, Environmental, Schedule, and Technology considerations and constraints should be included as appropriate. All viable alternatives for action as presented by the problem should be identified. The identified alternatives should be analyzed to determine the potential for hazards which could lead to the release of radioactive material, containment degradation, or toxic gas releases. Finally an effective EEA will demonstrate that the comparison of alternatives was based upon consideration of the identified criteria.

2.0 DESCRIPTION OF ACTION

W.A.H.

ŝ

The section on DESCRIPTION OF ACTION identifies feasible alternatives to implement the selected course of action, and establishes the criteria for evaluating alternatives.

The SA should then evaluate the implementing alternatives and justify the one selected . This section should detail the steps necessary to accomplish the selected implementing alternative, including preparation, execution, and recovery. Factors to consider in the description include program and process requirements and the physical and schedule constraints likely to be met.

The degree of detail in this section must be sufficient to identify all of the elements of any potential hazard that is described in Section 3.0. Maps, layouts, drawings and procedures should be used to clarify items described in this section.

3.0 IDENTIFICATION OF HAZARDS

 (\cdot) :

÷ -

-

The section on IDENTIFICATION OF HAZARDS should clearly identify all potential hazards related to the proposed activity. The hazards should be based on a functional analysis of the processes involved, and on environmental, materials fabrication, degradation processes, and inspection and monitoring considerations related to equipment and facilities.

Identification of Hazards should be based on an analysis of the tasks involved in installation, operation, and removal (as appropriate) of equipment and/or material in the facility. The analysis should be started after reviewing the work plan and understanding the possible effects and interactions of the activities and the physical and chemical phenomena in the facility being analyzed. Hazards to be considered include:

- release of radioactive material due to a spill or pressurization
- generation or release of flammable gases with potential for ignition
- pressurization of portions of the facility caused by exothermic reactions or other means (e.g., steam explosions)
- breaching of a tank caused by corrosion, embrittlement, deformation, reopening leak pathways, or combinations of the above
- reduction of facility structural integrity impacting safety
- creating and releasing toxic gases
- application of external influences such as:
 - addition of energy to the tank systems, including impact, friction, agitation, ultrasound, ignition energy
 - creation of chemical or physical reaction by additions or intermixing of tank contents

- • seismic hazards, as appropriate
- changing the physical or chemical phenomena occurring in the tank to initiate any of the hazards identified

The hazard identification technique used and the results should be described. The hazards typically associated with

high-level waste tank safety should be given extra consideration when identifying pertinent hazards. In doing so, consideration should be given to the following:

Tank contents

- burning flammable_gases or organic vapors
- explosion of gas mixtures
- explosion caused by liquid chemical reactions, / radiolytic reactions, and steam generation
- temperature excursions

Tank structural integrity .

• corrosion, embrittlement, deformation, re-opening leak pathways, and multiple failure considerations.

Hazard-initiators

\$2.5

-

- addition of energy to the tank systems, including impact, friction, agitation, ultrasound, ignition energy
- creation of chemical or physical reaction by additions or intermixing of tank contents
- Seismic hazards, as appropriate

4.0 HAZARD ANALYSIS

1

The section on HAZARD ANALYSIS should address each of the hazards identified in Section 3.0 relative to their potential severity. Factors include potential for occurrence (probability and magnitude of threat and jeopardy of vital equipment). Consideration should be given to environmental, materials, fabrication, and degradation processes, and inspection and monitoring practices as they impact failure modes and fragility analyses of system components, and the total system.

5.0 CONSEQUENCE OF ACCIDENTS

n a second de la constante de l La constante de la constante de

The section on CONSEQUENCE OF ACCIDENTS should consider safety and system effects caused by accidents related to the potential hazards. Appropriate use may be made of computer codes which have been benchmarked, verified, and validated. The consequence evaluation addressing exposure of site workers and the public to radiation should be expressed in terms of committed effective dose equivalence. The consequence evaluation addressing toxic environments should

5

の感謝論 Constantion

be expressed in terms of immediate danger to life and health.

6.0 CONTROLS

1

The section on CONTROLS should specify the controls determined by the hazard analysis that are necessary and sufficient to eliminate or reduce the consequences of the; hazard.

Controls should be described as follows:

- Give the basis for the selection of controls, including consideration of alternatives and the factors that resulted in the final decision. The Safety Assessment should document the analysis that demonstrates that the chosen controls are effective in controlling risk.
- Describe special operational requirements, the basis for special needs, and the means for their implementation. This category may include identification of hold points for administrative, safety, or quality assurance control.
- Describe any special equipment requirements identified by the assessment. Include design considerations to ensure its contribution to safety during the activity.
- For each potential hazard having significant consequences, indicate the means to be utilized to monitor the efficacy of the controls that are necessary for prevention and mitigation of the hazards associated with operation safety requirements with their associated safety limits and limiting conditions for operation.
- Provide a means for evaluating the status of parameters related to the control.
- Consider the potential for interaction between potential hazards and their individual means of control that may increase or decrease safety.

6

SA/EA DEVELOPMENT GUIDANCE I - ENGINEERING EVALUATION OF ALTERNATIVES

Assess problem
 Bevelop selection criteria
 Identify need for action
 Identify alternatives
 Identify alternatives
 Bescribe selected alternative

4. Identify hazards for alternatives

. - -

-

()

. **'**, 1 a te.

the shift of the

÷

躗

SA/EA DEVELOPMENT GUIDANCE II - SAFETY ASSESSMENT

- 1. Identify scope (EEA 1,2,3)*
- 2. Describe action (EEA 7)
- 3. Identify hazards (EEA 4)
- 4. Perform hazard analysis
- 5. Analyze consequences
- 6. Identify Controls

11 **1** 1

i i i

(Parens identify relevant sections of the Engineering Evaluation of Alternatives (EEA) development)

ï

SA/EA DEVELOPMENT GUEDANCE III - ENVIRONMENTAL AS ESSMENT

<u>،</u>

)

1.121, 221

•

- 1. Prepare introduction SA 1, 3; SA 1)*
- 2. Identify need for action (EEA 2)
- 3. Describe selected alternative (EEA 7)
- 4. Identify other alternatives considered (EEA 3)
- 5. Identify affected environment (From Safety Analysis Report)

· (Parens identify relevant sections of the Engineering Evaluation of Alternatives (EEA) and Safety Assessment (SA) development)

IRT COMMENTS

Lockheed Idaho Technologies Company

Idaho Falls, ID 83415

PRELIMINARY FINAL

LANL AND DOE COMMENTS HAVE BEEN INCORPORATED

 \bigcirc

FnelVersion used in closure (7-2-96) of Hanford.

JUNE 7, 1996

| | |) ⁷ • • |
|-----------------------------|-----------------|--------------------|
| REVIEW COMMENT RECORD (RCR) | 1. Date 5/10/96 | 3. Review No. |
| | 2. Project No. | 4. Page 1 |

A SAFETY ASSESSMENT OF ROTARY MODE CORE SAMPLING IN FLAMMABLE GAS SINGLE SHELL TANKS

HANFORD SITE, RICHLAND WASHINGTON

| 12. Item | 13. Comment(s)/Discrepancy(s) (Provide technical justification for the comment and detailed recommendation of the action required to correct/ resolve the discrepancy /problem indicated.) | 14. Hold Point | 15. Disposition (Provide justification if NOT required.) | 16. Status |
|--------------------------|--|----------------------|---|---------------|
| din 1 xvi gen E | Several of the Acronyms are not consistently presented or have editorial problems with them. These include the following; | | Accept - text will be scanned and acronyms made consistent. | |
| (| <u>minimum ignition</u> surface temperature permiss <i>i</i> ble exposure limit time-weighted average <i>unreviewed</i> safety question | | (| |

0.K.

1107 CHANGED IN TEXT PSXXVII - XXVIII

3. Review No. REVIEW COMMENT RECORD (RCR) 1. Date 5/10/96 2. Project No. 4. Page 2 NM 1 Administrative controls and The Accidents analysis section of this document is very Gen well developed. However, key issues such as mitigating mitigating procedural S procedures, controls, and safety systems and components requirements necessary to meet have not been developed. This approach at presenting a guidelines are identified in safety document for final approval and expecting the Section 6.0. completion of its essential components through the USQ process primarily circumvents and defeats the purpose of The purpose of the SA was to developing and writing a Safety Analysis Document. As establish a proper authorization such, the SA is incomplete, and suggests an improper basis for RMCS operations. application of the USQ process. Design features necessary to meet THE SSES GEENIA TON risk guidelines are identified in IDMINIFIED IN KEEL IS OFFICIEL Section 6.0. The Safety Equipment List (SEL), in WHC-SD-WM-SEL-032. reflects the requirements of Section 6.0 and identifies the safety classification of credited design features. The SEL (WHC-SD-WM-SEL-032) will be added as a reference to the SA. The SA itself states that adding a tank not currently considered 51 Should in lande a telle of Tacompleted issues Ingetter melle a commitment for "Choin Tec per the approved list will require a revision to the SA. This does not make the SA incomplete nor does it improperly apply the USQ process.

Comment Coding S Significant O Optional. E Editorial

A STREET

| <u></u> | | 1 | | ······································ |
|---------------------------------|--|--|--|--|
| | REVIEW COMMENT RECORD (RCR) | 1. Date 5/10/96 | 3. Review No. | |
| | | 2. Project No. | 4. Page 3 | |
| | | | | |
| NH 1 Gen S (continued) | NOTE: SER WILL MERICATE JOTE: SER WILL MERICATE SAFETY SIGNIKICANT OPEN JITEMIS | For those tar list in Apper the authoriza which future including USC based. The approach (1) perform a that establis necessary and and includes open issues. Level 1 contr the appropria implement Lev in OSD's and resolve all items per di stated in the an Operation and (6) obta prior to ini | hks on the approved ndix g. this SA is ation basis against operations, constant operations, constant operations, constant operations, constant operations, as a fety assessment shes all required d sufficient controls any unresolved or (2) implement rols via change to ate IOSR, (3) vel 2 and 3 controls procedures, (4) open or unresolved rections clearly e SA, and (5) perform al Readiness Review in DOE-RL approval tiating operations. | |
| | | • | | • |
| | | | | |
| | | | | |
| | | L | | |

.

Comment Coding S Significant, O Optional, E Editorial

1000 (1000) 1000 (1000) 1000 (1000) 1000 (1000) 1000 (1000) 1000 (1000)

| | | | | | | | · · · · |
|---|--|---|-------------|-----------|---|---|---------|
| | | REVIEW COMMENT RECORD (RCR) | 1. | Date 5 | 5/10/96 | 3. Review No. | |
| | | • | 2. | Project (| ¥o. | 4. Page 4 | |
| Π | | | | | | · · · · · · · · · · · · · · · · · · · | |
| F | NM 2 Exec Summary GEN S | The SA must identify the mechanism(s) by which the open items such as undeveloped new procedures. unidentified Safety Systems and Components. and specific test results (Bureau of Mines) will be evaluated with respect to DOE Orders and Standards to become part of this SA and support the TWRS ISB. As such, the ASA is incomplete, and it is an improper application of the USQ process. See <i>mayform</i> to $MM - 1$ by M Molecos | | | Items 7.1 throug complete. Section consists only of operational evan Mechanisms are p adequacy. Action up to and include evaluations, or SA. Currently f Mines (BOM) test completed and the monitors are quarequirements. | th 7.7 are now ion 7.0 now f a pre- luation checklist. provided to verify ons are identified ding new analysis. revisions to the the Bureau of ts have been he hydrogen alified per SA | |
| | NM 3 1-2 1.2 0 See Tevel Change | This section states that a representative tank with bounding conditions was chosen by performing a preliminary screening process. The criteria for decision making should be included in this description. | on re-Jan (| e t | Bounding tank se discussed in App mention of crite to 1.2. Following text "The screening important tank retained-gas ame dome flammable- concentrations, anticipated gas and the waste t SSTs on the FGW found to maximi of interest. | election is pendix C. Brief eria will be added added: process considered parameters such as ount. measured and toxic-gas the observed or -release amount. ype. Among the L Tank A-101 was ze the parameters | her |
| | | Sa | | | • | | |

•4

•

| (| | | | | | ·• ·, |
|---------------------------------------|---|-------|-----------|--|---|-------|
| | REVIEW COMMENT RECORD (RCR) | 1. | Date 5 | /10/96 | 3. Review No. | |
| · | | 2. | Project N | lo. | 4. Page 5 | |
| | | 1 | T | | | |
| din 2 2-2 2.1.1 Tbl 2-1 0 | DOE Orders 5480.7 and 6430.1A have been superseded by Order 420.1. The SA should reflect the current DOE Orders even though they have not been incorporated in new contract with the Contractor. | / DOE | | Excerpting from Statement DOE P "Policy Statement Identification Compliance with Safety, and Hea | Interim Policy 450.2 (9-29-95) nt on the Implementation and Environment, 1th Requirements": | |
| | | | | "Transition To H Orders | Rules And Revised | |
| | | | | The Department number of its Or rules and revise resulting trans managed so as to protection throu with maintaining protection, cost should be conside appropriately. | is replacing a rders with new ed Orders. The ition must be o assure adequate ughout: consistent g adequate ts and benefits dered | |
| | | | | Even though many be canceled as of rules and revise issued, cancella Orders does not modify or otherw contractual obli- the canceled Ord Requirements in which are incorp implemented in a remain in effect contract is modi- those requirement | y ES&H Orders will corresponding ed Orders are ation of these , by itself, wise affect any igation based on ders. canceled Orders porated and a contract will t until the ified to delete nts. | |

| (| \sim | | | | ; ; ; |
|--|---|------------------------|--|--|-------------|
| | REVIEW COMMENT RECORD (RCR) | 1. Date | 5/10/96 | 3. Review No. | |
| | | 2. Project | No. | 4. Page б | |
| din 2 2-2 2.1.1 Tb1 2-1 0 (continued) | O.U. TEXT NOT NO AL | | Also these are that were applidesign at the t clarified in the Section 2.0.2 wincluding Table | the DOE orders cable to the ime. This will be e text. as modified, (NEM 2-1. | |
| BEI 1 2-2 2.1.1 5 5 2.1.1 5 5 2.1.1 5 5 2.1.1 5 7 2.1.1 5 7 2.1.1 5 7 2.1 2 2 2.1.1 5 7 2.1 2 9 7 2.1 2 9 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 | A list of DOE Orders is NOT "principal health and safety criteria". They may contain design criteria, safety criteria, and so on. some relevant, some not. The principal health and safety criteria are those criteria which will be used in this document to determine which risks and consequences from normal, abnormal, and accident conditions are acceptable, and which require safety class (class 1, 2, or 3) fixes. NEED TO NECUDE IN TUXT "INFESTERY REEDS, REEMENT INCLUDES DOSIGN To THE FISE CRITERIA GIVEN IN WAC-CM-4. AS DISCUSSED IN SECTION 5. | 6. K. Elly I C S | Accept - Retitle Safety and Design and prior to Tal sentence. These helpful in deve criteria outline 2.1.3, 2.1.4, and The risk criter WHC-CM=4-46_and Section 5.0 of the title will | e 2.1.1 as DOE gn Requirements ble 2-1 add the e orders were loping the ed in 2.1.2, nd 2.2. ia are given in discussed in the SA. Modifying avoid confusion. | |
| BEI 2 2-3 2.1.3 5 | None of the bullets listed under the opening sentence an criteria. Please provide the criteria promised. | oF | Accept - The se revised to: "b considering the Quantitative cr given. In this present general WHC. Some of t the rest of the for example 25% if there is a f concern. | ntence will be y qualitatively following." iteria are also section we criteria used by hese are used in document as is, of LFL to decide lammability | |

•4

| C | | 0 | |
|---|-----------------------------|-----------------|---------------|
| | REVIEW COMMENT RECORD (RCR) | 1. Date 5/10/96 | 3. Review No. |
| | | 2. Project No. | 4. Page 7 |

| BEI 3 2-3 2.1.3 E | The first sentence in the second paragraph seems to be at odds with Tank Farm Safety Criteria. It is apparent from many other tank farm documents that one of the principal safety criteria is that the tank dome vapor space always have less than LEL for H ₂ and I believe it is <2%. Anything higher is an off-normal or accident mode. This sentence says that the Safety criteria is that it have less than 25% H ₂ , which says that 24% H ₂ is considered safe. Please fix this sentence. | 01C | Accept - sentence should state less than 25% LFL hydrogen equivalent. | |
|-------------------------------------|--|---|---|--------|
| BEI 4 2-4 2.1.3 E | The sentence at the top is a judgmental statement under a Criteria Section. Please delete. | SF. | The observations were important in establishing the 25% LFL criteria and need to remain as explanatory text. | |
| ONE: 4 Sec. 1241- Aprenolines | All of the material on page 2-4, including ¶2.1.3 and ¶2.1.4 have promised safety criteria, but the only thing that appears is criteria related to establishing a tank classification scheme. Whereas that is interesting, and of value somewhere else, it does not help with this /document if one were to look for the principal health and safety criteria, which are missing. | NO SWERNS SECTIONS SECTIONS INITIONS | This Section (now 2.0.3 and 2.0.4) have been revised. Criteria are now referenced but bulleted items at beginning of 2.0.3 are now considered "qualitative observations", 17,5 As it evises, 18,000 | 4 rent |
| din 3 2-5 2.2 Tb1 2-2 0 | The safety criteria of organic watch list tanks specifies that the level 3 tanks (unsafe) are defined as not meeting level 2 requirements. This implies there are three variables that could be judged to define a level 3 tank which are: less than 5 wt% TOC, less than 17 wt% moisture, and greater than 90°C. I don't think that it is intended that all three criteria be used to judge the level 3 Watch List tanks. Level 1 tanks are defined as having less than 5 wt% TOC. The specific criteria used to define the level 3 tanks should be specifically defined. | | Table cited was not germane and has been deleted. SAFETY TANKS SHOWED WATCH LIST TANKS SHOWED WATCH LIST TANKS SA BE | K |

.

| \mathcal{O} | | | |
|-----------------------|----------|--------------|---------------|
| REVIEW COMMENT RECORD | (RCR) 1. | Date 5/10/96 | 3. Review No. |
| | 2. | Project No. | 4. Page 8 |

| BE1 6 2-5 2.2.1 0 | These four categories have no category for tanks that exhibit episodic gas-release behavior. | e¥- | The third item considers tanks that potentially exhibit episodic gas release. There is only one SST in this category (A-101). There is no data to suggest that SSTs exhibit episodic releases. There is however a concern that they could. See Appendix L also. |
|-----------------------------|---|-----|--|
| BE1 7 2-5 2.2.2 E | The first two sentences tell the reader that there is "some very scarce, limited." data available. That seems to have more qualifiers in a row than necessary. | 514 | Accept - "Very scarce" will stay, "limited" will be deleted. |
| BE1 8 2-6 2.2.3 E | "free-standing" liquid. This would imply an unsupported column of water, last seen when Moses crossed the Red Sea. I suspect that some term from the hydrologists, like "perched" supernatant would be more accurate and descriptive. | OK | Accept - The sentence will be reworded. |
| BE1 9 2-7 2.2.4 E | On page 2.7, the heat of combustion is referred to as heat of combustion, on the next page it is called heat of reaction. The term heat of combustion is correct. | or | Accept - heat of reaction on page 2-8 will be changed to heat of combustion. |
| BEI 10 2-8 2.2.4 E | line 9 from top of page. It is not "potential hazards". but "potential consequences" which are increased. | σK | Accept - Hazard will be changed to consequences. |
| BEI 11 2-8 2.3 E | Is it correct that 6.5 feet of earth covering was used for "Heat dissipation"??? | эК | Accept - Yes. Primary Reason: shielding Secondary Reason: heat dissipation |

-4

.

120

.

ial

| <u> </u> | | | <u> </u> |
|----------------|--------------|-----------------|---------------|
| REVIEW COMMENT | RECORD (RCR) | 1. Date 5/10/96 | 3. Review No. |
| | | 2. Project No. | 4. Page 9 |

| BEI 12 2-9 2.3 Fig. 2.1 E See flue | Please insert a drawing that an engineer could find acceptable. This tank is out of scale, lop-sided, and probably inaccurate. (rabute /outer | OK. 147202 IT FROM 185 - | Note Figure 2-1 is a schematic. A schematic is a broad, diagrammatic presentation. A detailed civil engineering drawing was not intended. However, an improved schematic was added with representative dimensions. The engineering drawings are available for the reviewers |
|---|---|-----------------------------------|--|
| din 4 2-8 2.3 last 0 | This section describes that the HEPA filters are protected by a small loop seal to protect against a plugged filter. This is also potential for an unmitigated release of radioactive and hazardous material if the loop seal is unintentionally voided of the seal material or needed to relieve the pressure from a plugged filter. This potential accident scenario should be included in the safety analysis of system operation. | 8-1. | The loop filter seal is on the tank breather inlet HEPA. The presence of the RMCS does not increase the probability of loop seal failure and reduces the probability of unmitigated release (due to the presence of the exhauster) in the event of a loop seal failure. |
| | | | - A FAILUZE OF THE EXHAUSTER CONNECDENT WITH LOOP SEAC FAILURE OF SHOW O BE EVALUATED FAILURE OF SHOW O BE EVALUATED AS A POTENTIAL ADDIDENT. |

4

Comment Coding S Significant.

5

O Optional. E Editorial
 REVIEW COMMENT RECORD (RCR)
 1. Date
 3. Review No.

 2. Project No.
 4. Page

 10

| din 5 2-9 2.4 gen 5 5 6 6 7 7 7 8 8 13 2-11 2.4.1 F | From the description of the various components associated with the core sampling operation, it is difficult to understand how the components of the drill string interacts with each other. This information is essential for the reviewer to determine the potential accident that can occur with the drilling operations. Detailed diagrams/drawings of the component interfaces should be provided. Examples of the interfaces needing further definitions include the quadralatch, the pintle rod, the pintle rod/rotary valve, the grapple clasp. the mechanical remote latch, etc. For example, it is not clear from the discussion/drawings, how the mechanical remote latch is used to interface with the sampler to insert/remove the new samplers as the drilling progresses or to retrieve the samplers following the completion of the core sample drilling. | O'K OK | The difficulty in understanding this system is understood. Section 2.0 and Appendix A provide some detailed drawings, however. without the use of video tape or direct field observation it is difficult to fully comprehend the operation of the RMCS system. WHC staff would welcome the opportunity to show the reviewer(s) how the system performs. Section 2.0 tries to give minimum design information and defines the system in a general sense. References and adequate WHC design documentation are available to the reviewer. WHC will assemble a drawing set for use by reviewers. Accept - 'but' will be changed to 'and'. | |
|---|---|--------|---|--|
| BEI 14 2-11 2.4.1.2(3) E | The sentence that says the load cell is designed to shut off the motor with a "maximum load" of 250 lb. This is grammatically incorrect. That sentence says it would not shut it off with a load of say 300 lb. Recommend, " if the load equals or exceeds 250 lb." | œ | Accept - will strike "with a maximum load of" and replace with "if the load equals or exceeds 250 lbs." | |

Comment Coding S Significant,

O Optional, E Editorial

| | (j) | <u>}</u> |
|-----------------------------|-----------------|---------------|
| REVIEW COMMENT RECORD (RCR) | 1. Date 5/10/96 | 3. Review No. |
| | 2. Project No. | 4. Page 11 |

| BEI 15 2. 9. 1. 4 (2) S O D Add S O D Add S O D A D S O D S | The drill teeth are a safety feature of the system. However the teeth are "proprietary" NOT OK! This is the Safety Analysis document. If the Safety Analysts don't know what it is made of. I find it hard to believe that it is acceptable. Being proprietary does not mean that we should be commensurably ignorant! Provide either the composition of the material, proprietary or not, or a complete property data sheet, showing as a minimum hardness, tensile and compressive strength, melting point, and sparking potential. | (O.K | The material is now identified in 24.1.2 NO SUCH SUCTION 1 NO SUCH SUCTION 1 PROVIDE IN SA. | |
|---|--|------|---|--|
| $\begin{array}{c} \text{BEI 16} \\ 2-13 \\ 2.4.1.4(4) \\ \text{S} \end{array}$ | The description of the drill rods is unacceptable. The hydrogen burn and explosion are based on the volume and strength of the drill rod. Although the OD is given, the volume can not be calculated and neither can the burst strength. Please add also the ID, the nominal wall thickness, the minimum wall thickness (in the vicinity of the threads), and the tensile strength and fracture toughness. | | Accept - The volume of the drill string is important but we always assumed that if there is a spark in the drill string no matter what the volume and gas composition is the flame will propagate into the waste. This is conservative. However, the requested data will be added (now Section 2.3.1.2). | |

NO SUCH SECTION IN REPORT TWAILAISLE WHEN AVAILAISLE STILL WEN

-4

~~

| | <u> </u> | <u> </u> |
|-----------------------------|-----------------|-----------------|
| REVIEW COMMENT RECORD (RCR) | 1. Date 5/10/96 | 3. Review No. |
| | 2. Project No. | 4. Page - 12 |

.

| din 6 2-19 2.4.2 2nd P S Ad Leven Ad Leven Ad Leven Ad Leven Ad Leven | This section describes the exhauster system which includes an air preheater to maintain the humidity of the air to be no greater than 80%. What controls are placed on the operation of the preheater for the operation of the exhauster system? If the preheater fails, does this imply that the exhauster must also be shutdown? The implications associated with a system failure should be described in this section or provide a reference to where the operational limitations are provided. Loss of preheater is not one of the items identified in Section 2.5.5.3 that would cause an automatic termination of the exhauster. A failure of the preheater could be either in leak of the HX to the filter system which could result in plugging the filter or a failure of the HX to operate that would result in excess tank humidity getting to the filter, thus causing the filter to plug. The controls for the preheater should be discussed. | 6.4 | Failure of the water tube preheater does not result in a hot spot and cannot cause ignition of flammable gases. Failure of the preheater could result in saturated HEPA filters. Airstream humidity is constantly monitored and an alarm is generated when the level exceeds 80%. To further protect against filter collapse, the blower is limited to nine inches of water static pressure. The HEPA filter manufacturer states that the filters will remain intact and maintain effectiveness at 10 inches of water pressure differential. No preheater controls are required. | |
|--|--|-----|---|---|
| BEI 25 2-29 2.5.2 Bullet 1 0 | After a detailed description of the equipment in Section 2.4x, there was no mention of a "bonded lifting bail." Is this "bonded" as in insured or guaranteed (to compensate for the proprietary drill bit) or is it bonded as in glued?, and if glued, what kind of glue. Or is it somehow electrically bonded? The statement obviously raises more questions than it answers. | 014 | Accept - Clarification will be added that the electrically bonded lifting bail is a device that supports the drill string and never enters the tank. | 8 |

Comment Coding

| C | | | | | | · · · · |
|---|---|----------------------|-----------------|--|--|---------|
| | REVIEW COMMENT RECORD (RCR) | 1. | Date 5 | /10/96 | 3. Review No. | |
| | 17 | 2. | Project N | ło. | 4. Page 13 | |
| | IN REVISED LOPA IN REVISED LOPA SECTION 2.5.3 IS IN SAME AS | z,5 | \$ \ | | | |
| BEI 26 2-30 2.5.3.1 Bullet 1 S | A key component is the gas monitor. The system can not be operated without it. Later, (table 4-10), it is identified as a credited control. There is no mention of the gas monitor in the system description, or in the diagram (Fig 2-13) accompanying this paragraph. | of | ð. | Accept - A brief the gas monitor Section 2.0 with Appendix U (adde The gas monitor are discussed in | f description of will be added to reference to ed as 2.3.3). design criteria Appendix U. | |
| din 7 2-36 2.5.5.2.2 S | This section specifies the limits on the rotational spec of the drill at 55 rpm and the down force limit at 750 lbf. This is not consistent with the limits specified in Appendix N which specifies the rotational limit of 40 rp and the down force limit of 650 lbf for drill strings of greater than 45 ft. The limits should be consistent throughout the report. This same comment applies to the information presented in Table 2-7. | ed in Dom F | | These limits are problems. 55 rpm and 750 l with purge gas f control drill b (Level 1 Hardwar 6.6.4.1). The 40 rpm and 6 compensate for a buckling problem drill string (Le Administrative 6 | e for different lbf in conjunction flow are to it temperature re Control 650 lbf are to a potential m with a long evel 3 Control 6.9.12.4). | đ |
| NM 4 2-37 2.5.5.2.4 2.5.5.2.5 S | The document should specify which procedures and controwill ensure implementation of the necessary actions outlined in these sections. | | ALLED. | The document spender of the document spender of the second | ecifies all ols in Section | |
| Cristing of the second | This section states that "Restart following off-normal incidents should be performed in a way consistent with the requirements of the Interim Safety Basis." Even though this is necessary, it is not a sufficient statement. The applicable and approved requirements in the ISB should be identified in the section in table format |) | BEST. | Applicable and requirements ar Section 6.0. This SA and IOS implemented via ISB. | approved e identified in R's will be a revision to the | |

•

Comment Coding

ŝø,

 $\begin{array}{c} L_{1} < 1 \\ L_{2} < 1 \\ \ell < 0 \\$

| | <u></u> | <u> </u> |
|-----------------------------|-----------------|---------------|
| REVIEW COMMENT RECORD (RCR) | 1. Date 5/10/96 | 3. Review No. |
| | 2. Project No. | 4. Page 14 |

| BEI 17 3-2 Table 3-1 E | For chemistry consistency, recommend that Oxygen be added to the list of flammable gases, since it is abundantly present; then, under the right hand column group first the "flammable fuel gas" and then the oxidizers. | ØK | Flammable gases, in the context of the safety assessment, are mixtures of H_2 and oxides. Oxygen will be added as an Oxidizer. | • |
|--|--|---|--|---|
| NM 6 3-3 (Methodotogy) S V | See NM 1 Gen and NM 2 Exec Sum Gen Above. This approach circumvents the Safety Analysis Review process by omitting essential issues from the SA and leaving them for review at a later date under a USQ. As such, the SA is incomplete, and it is an improper application of the USQ process. | See . Nill file. | No circumvention of Safety Analysis Review process is Intended. See response to NM 1 GEN and NM 2 Exec Sum GEN. | |
| NM 7 3-4 5 3.2 | This section states that "the hazards associated with transportation of the rotary-core drilling unit or its auxiliary equipment from the tank farm or its ultimate decontamination and disposal are not considered. Transportation of the cask where core samples are stored is also not considered." The SA should identify which approved document evaluates these activities and conditions. Otherwise this presents a problem with "Interfacing" between activities and facilities, and consequently introduces non-evaluated scenarios and associated risks. | REFERENCE 12 SALESTON 12 SALESTON 12 SALESTON 12 SALESTON | Accept - Appropriate reference documents will be cited. Transportation of this equipment outside the tank farm boundary is clearly outside the purview of the safety assessment. | - |
| NM 8 3-5 3.2 S/ | First Paragraph, states "frequency determination did not include a detailed failure-rate evaluation. but qualitative frequency estimates are provided." This is a legitimate approach, however, the SA should provide a logical derivation of the values presented more than "engineering assessment". | | Data used in the estimation of reliabilities are obtained from standard industry literature and documented in (Appendix D and E) and provide a logical derivation to the approach. They are available to the reviewer. | |

| - | Ú. | (| | |) |
|---|-------------|------------------|---------------|---------------|---|
| | REVIEW COMM | ENT RECORD (RCR) | 1. Date 5/ | 3. Review No. | |
| | | | 2. Project No | 4. Page 15 | |
| • | | | | | |

.

•

4

.

| BEI 27 3-8 Table 3-3 Row 6 0 | Comment BEI 15 was made in response to a failure to identify the drill bit material. In this table the teeth are incorrectly identified as (brass-based). Since brass is a generic name for a wide variety of copper-zinc alloys. it is non-scientific to refer to a brass-based alloy. This table refers to \P 4.2.6, which correctly refers to the drill bits as bronze (a correct and valid term) and also equally correctly as copper-based. It is recommended that a proper description (sintered bronze with tungsten bits) be incorporated into \P 2.4.2.4, and thereafter use either bronze or copper based unless more detail is warranted. | SK | Accept - It will be changed to copper-based. | |
|---|--|-----|--|--|
| 8EI 28 3-10 Table 3-4 Row 5, C3 E | "maintain continuous in contact" ? How should it read ? | aK | Accept - Sentence revised to "maintain continuous contact". | |
| din 8 4-6 4.1.1.3 0 | This section specifies that the exhauster air stream is preheated with a hot water heat exchanger to preclude condensation from occurring on the HEPA exhaust filter. It is not apparent that the increased likelihood of the plugging of the HEPA filter due to leaks from the heat exchanger has been taken into consideration. (See comment din 6). | or | Yes. this has been taken into consideration. The benefit of a non-electrical water heater far exceeds plugging the HEPA filter, which event would cause exhauster shutdown, an uncommon but anticipated event. | |
| din 9 4-10 4.1.3 E | This section specifies that leak size for pit to riser is assumed to be no bigger than 1 in. It should be clarified what is meant by 1 in. In Table 4-3, it states that "less than 1 ineffective leak diameter". If this is what is meant, then it should be used throughout the text. | oic | Accepted - text will be scanned for consistency and changed to 1" effective leak diameter. | |

Comment Coding

| | 0 | <u> </u> |
|-----------------------------|-----------------|-------------------|
| REVIEW COMMENT RECORD (RCR) | 1. Date 5/10/96 | 3. Review No. |
| | 2. Project No. | 4. Page <u>16</u> |

| din 10 4-21 4.2.3 S | Several of the frequencies presented in this section are not supported by the logic or assumptions used to derive the frequency values. Differing frequency values can be found in some of the Appendices (D and E). It is recommended that the detailed information and assumptions be provided in the SA. Some of the probability numbers appear to be extremely small. Justification of these numbers needs to be provided. | 01 | Accept - Appendix D and E will be reexamined and appropriate justification provided as necessary. Ecosoria et soctron 4.2.3 | |
|------------------------------|--|-------------------------------|---|--|
| | This section states that aluminum, bronze, and teflon are known to be incompatible with some of the wastes. In Section 4.2.6, it states that the drill bit is made of sintered bronze. What compatibility tests have been run to determine the compatibility between the waste and the drill bit? This information should be supplied in the SA. Chapter 2 should provide a description of material used for the construction of the various components so that the reviewer can determine the compatibility of components has been adequately addressed. | 15, 15/500 105 ARIFAED. | As stated on page 4-25 if the Bureau of Mines (BOM) tests indicate no ignition, then the accident frequency with this proprietary material is 'considered zero. BOM tests are now complete. Material compatibility has also been demonstrated by the evaluation of historical information. | |
| din 12 4-51 4.8.1 S | The assumptions used to determine the maximum release from a HEPA filter failure is based on the loading from the drilling operations associated with one tank. This implies that the exhaust HEPA filter must be changed out after each tank drilling operation to maintain the safety analysis within limits. This requirement should be incorporated into the administrative section of the IOSR to ensure that it is accomplished. | of | As long as filter housing contact dose rate is less than the 100mR limit (specified in 6.9.6.5), the value for the maximum release as postulated in the SA would not be exceeded. There is no need to change out filters until the 100mR limit is reached or the pressure differential limit (5.9 inches w.g.) across the filter bank is exceeded. | |

. . . **.**

Comment Coding

O Optional, E Editorial

| (| | | | | ÿ |
|--|---|--------------------------|--|---|------|
| | REVIEW COMMENT RECORD (RCR) | 1. Date | 5/10/96 | 3. Review No. | |
| | | 2. Projeç | t No. | 4. Page 17 | |
| | | | | | |
| BEI 29 4-46 4.6.2(1) 0 | "The rubber seal that girths" is not shown/discussed in the system description section ¶2.xx. | ok | Accept - It is ((page 2-21), TI will be changed | described in 2.4.3 he word 'seal' to 'frisbee'. | |
| BEI 30 4-46 4.6.2(2) E | The low frequency does not impact the hazard. Change in frequency changes the Risk. |) 014 | Accept - change | hazards to risks. | |
| BEI 31 4-48 4.6.4(2) E | "The stress from this impact" can not trace the impact being addressed. | ØK | Last sentence o paragraph discu driven into the | f previous sses risers being tanks. | |
| | | | The stress refe result of these | rence to is the | |
| | This section specifies the Radiological Risk Guidelines that were used in the evaluation of the safety or the Core Drilling Operations. The most restrictive of the risk guidelines cited were used in the safety evaluation However, some more restrictive limits are being proposed primarily for the onsite workers. How much impact on th safety analysis results if the more stringent limits were imposed? | n. d. Xr- ne re | As stated in 5. Revision 3 of t were used per t DOE-RL. More s guidelines have analyzed. Dog | 1.1. he Risk Guidelines he direction of tringent not been HAS MADE | - |
| •••••••••••••••••••••••••••••••••••••• | | ••• | A RULIN FISE GU WILST IN S 914 RET OF | LA ON WHICH MDELINES BE USED A-ZS. S Soction 5 Hoc S Soction 5 Hoc CELT BULLENST S LISE GUIDEL | ATUS |
| | | | • | | |

-4

. .

ł .

.

Comment Coding

n Ontional

F Editorial

| | (| <u>(</u>) | | |
|--|---|-----------------------------|-----------------|---------------|
| | | REVIEW COMMENT RECORD (RCR) | 1. Date 5/10/96 | 3. Review No. |
| All and a second | | | 2. Project No. | 4. Page 18 |
| | | | | |

| din 14 5-8 tbl 5-7 S Juger Jug Juger Juger Jug Juger Juger Juger J | The selection of the size of the releases during drilling operations is based on event probability: smaller releases having a higher probability of occurrence. The event probability associated with the size of the release is provided in Table 5-7. However, in Table 5-8 the events are classified as "anticipated", "very unlikely", etc., that do not correspond to the calculated frequency of occurrence should be provided. An explanation of the event classification and how it corresponds with the calculated frequency of occurrence. One would expect that the classification and the calculated frequency of occurrence would be in direct correlation with one another. It is difficult to rationalize an anticipated if event having a frequency of occurrence on the order of 2E-6/yr. | A A A A A A A A A A A A A A A A A A A | Frequency Category vs. Frequency Range was discussed on page 5-3. The words "anticipated", "unlikely" etc. are used to define the gas release event during the accident itself. There are other events that must occur to have a radiological release. These events are discussed in the text and their failure probabilities are given in the table. | | |
|--|--|---|--|----------|----|
| NM 9 7-1 7.0, 7.1, 7.2 S | See NM 1, NM 2, and NM 6 $0 \cdot 1$. | 1, 1 1 × 101 1, 1 1 × 10 1 5 5 × 1, 1 | See NM 1. NM 2. | | |
| NM 10 7-2 7.7 S | This section requires that "If additional Level 1 Controls are found to be necessary as a result of actions needed to close the open items, the SA must be reviewed and approved prior to operations." 1) How will this be implemented, and which authority/expertise will make this determination ? 2) See NM 1, NM 2, and NM 6 above. This de facto makes this document incomplete for the planned activity. | | If changes to the SA are required, review and approval will be by DOE-RL. Implementing the Section 6.0 controls and completing the Chapter 7 items makes the SA complete for the tanks specified in Appendix G. | | |
| Comment Codir | ng the Optional E Editorial IE | PILUT PILUT S.A. AP S.A. AP S.A. AP S.A. THIC ZESTRIC ZESTRIC ZESTRIC | PROVINCE AN ON UT PROVINCE AN ON UT PROVINCE AN OF SEATLON OT ON OF SEATLON TON AREA | . | 4. |

| | (| <u>()</u> | |
|---|-----------------------------|-----------------|---------------|
| | REVIEW COMMENT RECORD (RCR) | 1. Date 5/10/96 | 3. Review No. |
| - | | 2. Project No. | 4. Page 19 |

6.5.2.3 NM 11 reader can infer that an SA must be performed for each Ans Barry performed Considering the checklist shown in this section, the The checklist itself will 7-3 ROUSED D stipulate the mechanism for 78 FG/RCMS activity on any other tank, or treat these adding an additional tank. This mechanism is covered in 4.2.3 of 'Appendix G (page G-15) and Administrative Control 5.31.1 of the IOSR. INIKS. However 480.23 NM 12 See NM 1, NM 2, and NM 6. The SSCs, preventive and See NM 1 and NM 2 (at a SA. 1 A-2 commercial operating nuclear (5460.03) TOO SHOWED Derson and Prover GEN .-power facility, abnormal (1st Para) operating and emergency response procedures are not in integral part of the safety assessment). first Bullet. The type and quantity of hazards are not Appendix A is used for screening/ NM 13 Waste" do 1 for The Contract of Norm M, Mandem M & Emillerizes (OPERATIONDS, (ATTACH. 1 ITEM 13). purposes. Table 3-1 summarizes A Wyshown in Table A-5. "Chemical Reactions" and "Tank Waste" do not describe type or quantity of a hazard. THE HARARDS IDENTIFIED IN THELE A-S DO NOT FOLLOW THE DEFINITION OF HARARD DEFINED IN SECTION A.I.Z. NETTHER DOUS TABLE 3-1 DEFINIS THE SPECIFIC ITA BARDS ASSOCIATED WITH " CITEMICAL REACTION" OVE. TANK WASTE Comment Coding

| | | <u></u> | | <u> </u> |
|-------------------------------|---|--|---|--|
| | REVIEW COMMENT RECORD (RCR) | 1. | Date 5/10/96 | 3. Review No. |
| | | 2. | Project No. | 4. Page 20 |
| | | T | | |
| NM 14 A-6 2.0 S | This section concludes that "bump drilling mode we eliminated from the operation." The SA should de the controls or procedures that implement this example. | vas elineate kclusion. | Bump drilling m safety envelope the SA. | ode is outside the established by |
| | | | These modes of proposed. They physically poss controls and ad requirements. | operation were are not ible due to design ministrative |
| | 1 | | Section 2.0 int not mention bum rotation modes identified duri hazards identif They are not go Therefore, they in Section 2.0. addresses the o in Section 2.0. | entionally does p mode or slow that were ng the initial ication study. ing to be used. are not defined This SA only perations defined |
| BE1 35 B-04 3.3(2) E | There appears to be either an error in this paramit is simply unclear. If the definition of "dian remains constant, than it is impossible to have concentration drop from 8% to 2.4% to 0.6% and t increase to 1.2% at distances of 0, 18, 2.2, and source diameters, respectively. A sketch could most of the mis-understanding. | graph or meters" the hen 36 solve | Accept - The ce concentration d 1.2%. The conc is at a radial axial distance. will be revised concentration a distance drops radial distance jet centerline | nterline rops from 2.4% to entration of 0.6% distance at 18 D The sentence I to"the it the 18 D axial to 0.6% at a e of 2.2D from the " |

•

4

.

| <u> </u> | | | | | ·· | <u> </u> |
|--------------------------|---|--------------------------|------------------------|--|--|-------------------------|
| | REVIEW COMMENT RECORD (RCR) | 1. Da | ate 5/ | /10/96 | 3. Review No. | |
| | | 2. Pr | roject No | D. EEQUIREM | 4. Page | |
| | | He | UST of | Filte REQUIED | N CHAP 4/11/ EMENT ARE | NOT |
| NM 15 B-5 3.3 | This section describes and calculates consequence to equipment exposed to a flammable atmosphere. The second paragraph on page B-5 stipulates positioning of equipmer during_the activity. This preventive approach is legitimate, however, the SA should also indicate the controls and procedures which will ensure and implement this preventive approach. | PLE PLE PLE PLE | ASE ZOUNDE MONAU | The appropriate Deen stated in S 4:0. Section 6. will be added to then put into 10 and procedures. | requirements h Sections 6.0 ar 0 requirements 5 the IOSR and 5 wer tier contr | nave id rols |
| NM 16 B-8 4.1 S | The SA should provide the rational and basis for assumin that the releases from tank 101-SY are bounding for the FG/RMCS activities. | 19 There Here | Jaco , | The rationale ar assuming that ga Tank 101-SY are contained in App The worst gas re in tank 101-SY. Appendix L. SST expected to be slower. Based of have the boundin in the appendix | nd basis for as releases fro bounding are bendix L. elease is obser As discussed gas releases a much smaller ar on the data we ng number is g | om ived are ad |

۰.

•

-4

Comment Coding

1973

0 Ontional

¢

al E Editorial

| | REVIEW COMMENT RECORD (RCR) | 1. Date | 5/10/96 | 3. Review No. | |
|---|---|---|---|---|---|
| | | 2. Projec | t No. | 4. Page 22 | |
| | | NIGED DAMS | DEFINITION OF WEEKS, YEARS | "stow", etc. | |
| NM 17 C-3 3.1 S | The second bullet states " The species ratios do n change with time or bubble location in the waste." is correct only under steady state conditions but after a GRE since the production rate for each of species is unique and different. Consequently, th assumption that they are in at constant proportion also not always true. The SA should address this concern. | ot This not these e added s is | Gas release ev fast. Chemica effecting gas <u>are slow</u> ? The during a GRE stored in the GRE. Therefor of stored gas of the compos a GRE. The con the GRE is the | vent is relatively al processes generation rates e gas released is the gas that is waste prior to the re, the composition is a good measure ition of gas during mposition prior to a concern. | |
| ·0. U | THERE IS ALD ANDI NAVEGJILIBEILM WILL EXIST! | ionitor 101 | During a GRE stored gas is the gas as it RMCS operation contemplated | the release of the the concern and not is being generated. ns were not during transient | • |
| NM 18 C-5 Table C-3 S (), K | The discussion, methodology, and results summarize Table C-3 are based on Steady-State conditions. T not necessarily represent conditions during and af GRE. The SA should evaluate this concern. | d in hese do ter a | See response data are used bound the com that is store long time. | to NM 17. The SST conservatively to position of the gas d in the waste for a | |
| NM 19 C-6 3.1 S (; // | The last statement in this section concerning AX-1 should be re-evaluated in view of comment NM 18 ab | 01 ove. | See response (also the ver assuring hydr concentration | to NM 17 and 18 ification step of ogen equivalent gas is < 1000 ppm prior CS operations | |

•

4

Comment Coding

O Optional. E Editorial

| <u> </u> | | | | | | · · · |
|---|--|------------------|---------|---|--|-------|
| | REVIEW COMMENT RECORD (RCR) | 1. | Date | 5/10/96 | 3. Review No. | |
| | | 2. | Project | No. | 4. Page 23 | |
| | | | | | ۰ ۰ | |
| BEI 36 C-10 4.0 S | Sustained reactions in a gas are a function of composition, pressure and temperature. The over-pressure is also a function of temperature. Please provide temperature information, initial and calculated final temperatures, and ignition temperatures. SEE AMACHED TEXT AND CALS. THE TEXT INDICATES THAT THE IGNITION TEM PETERTURE IS A FEINIERTON OF PRESS Smm Hy CORRESPONDS TO IGNIFTED OF PRESS Smm Hy CORRESPONDS TO IGNIFTED TEMP O ~ 400°L. AT ATMOS PRETIC PRESSURE, THE IGNITION TEMP IS ~ 550°C. THIS INFOR SHOWLD BE INCORPORTATED INTO Hy EVAL | re A | rod | Calculations are conservative was Pressure does no adiabatic burn because specific gas is only fun- temperature. The pressure was atte ignition temper in this calcula is assumed, a s ability to igni regardless of i Ignition temper concern when the somewhere in the | e based on ste temperatures. ot effect the calculation c heat of ideal ction of he initial dome mospheric. The ature is not used tion. When a burn park with an te is considered ts origin. ature is only a ere is a hot spot e system. | 1 |
| BEI 37 C-10 4.0(3.4) S J:- CJW J-4 , W J-4 W J-4 | The data in ¶ 3 and 4 can not be reconciled. In ¶ 4 it states that 96 m ³ of hydrogen is equivalent to 2.4%. In ¶ the dome volume is assumed to be 1415 m ³ ; but, 96/1415 = 6.78%. IHE EXPLANATION OF THE MADE $FI = 6.78%$. IHE EXPLANATION OF THE MADE $FI = 6.78%$. | 1 5 4 - | OF | 96 m3 is the re and is not 100% assumed that th in the released 40% includes 88 indicated in AX hydrogen concen 35%. If 96/141 by 35% we have the dome. | leased gas volume hydrogen. It was ere is 60% ammonia gas. Remaining 1% hydrogen as 101. Thus, tration becomes 5.8 is multiplied 2.4% hydrogen in | S |
| | WOULD BE USEFUL. | | | include in | Turk. | |

ч

.

Comment Coding

0 Ontional E Editorial

| REVIEW COMMENT RECORD (RCR) | 1. Date 5/10/96 | 3. Review No. |
|-----------------------------|-----------------|---------------|
| | 2. Project No. | 4. Page 24 |

| BE1 38 C-10 4.0(4) S | It is stated that the adiabatic burn will give a pressure of 35 psia in a 2.4% H ₂ burn. Using 57.8 kcal/mole (241 kJ/mole as on page C-6, 8.8, 6.9, and 7.07 cal/mole•K for the specific heat of water, nitrogen, and oxygen, respectively, and 6.6% by volume hydrogen, and an initial dome air-space temperature of 50 C, this reviewer calculates a peak temperature of 590 C (the ignition temperature of hydrogen in air from the Handbook of Chemistry and Physics) and an final pressure of 37 psia. Using similar assumptions, a 4% mixture would result in a pressure of 30 psia, still enough to fail the tank. The issue is not whether or not an over-pressure sufficient to fail the dome is possible, but in reproducible and | Section C.4 (page C-16) was largely rewritten. SFE ATTACHED TEVT AND CALS. MAKO STATEM ONTS WITEXT DEFENSITS LE |
|-------------------------------|---|--|
| | issue is not whether or not an over-pressure sufficient to fail the dome is possible, but <u>in reproducible and</u> traceable statements in the text. | |

Comment Coding
| <u>(</u> | <u> </u> |
|-----------------------------|-------------------------------|
| REVIEW COMMENT RECORD (RCR) | 1. Date 3. Review No. 5/10/96 |
| | 2. Project No. 4. Page 25 |
| | |

| | | Following the above comment. the flame/gas-explosion literature was revisited. It is very clear that the mathematics of flames and explosions is far more complicated that the description in Appendix C. Clearly, one must be careful in describing terms and conditions. And the state variables are pressure, volume (or mole- percent), and temperature. The resolution of the problem discussed in the previous comment is that ignition temperature is a function of pressure. At a pressure of about 5 torr, the ignition temperature is 400 C, that is the flowest temperature at which a hydrogen oxygen flame can propagate. At 1 atmosphere, the ignition temperature is 550 to 590 C. The 4% LEL is not energetic enough to raise the temperature falls in an oxygen rich (air) atmosphere. The 4% and 400 C are safe limits under essentially any conditions. However, one can not use the minimum ignition temperature of 590 C, and the concentration that will produce that temperature to calculate an over-pressure and attribute it to an administratively low 2.4% hydrogen concentration. Please include a few standard graphs which show ignition temperature as a function of pressure as a function of hydrogen concentrations such that the document will be technically accurate. (See attached Reference Figure) | , ruents | In order to cause a burn we need three things, fuel. oxygen. and energy (spark) to ignite. Spark can be created by electrical, mechanical, etc. devices. In this SA we are mostly concerned with the frictional sparks, electrical sparks, impact related sparks, etc. These have enough energy to ignite hydrogen. One needs only 0.01 mJ (electrical device) to ignite H2. One other way to start combustion is to exceed autoignition temperature as the reviewer points out. The autoignition temperature is only concerned when there is a hot spot created by friction or something else. Autoignition temperature is pressure dependent but also changes with composition. Autoignition temperatures may be around 400°C (See Appendix G) for H2 and N20 mixtures. The pressure effect is somewhat irrelevant because we have always atmospheric pressures or slightly less than this value in the dome. The flame propagation after ignition is mostly concentration dependent. There are only a few accidents in the SA concerning the autoignition temperature (frictional heating of the riser | |
|-------------------|-----------------|--|----------|---|---|
| Comment S Sign | Codin ifican | g It O Ontional. E Editorial | ` A | nto ignition at 100°C occurs |) |

| Ć | \bigcirc | | | | | · · · · |
|--|---|-----|-----------|---|---|---------|
| | REVIEW COMMENT RECORD (RCR) | 1. | Date 5 | /10/96 | 3. Review No. | |
| | | 2. | Project I | No. | 4. Page 26 | |
| T | | | | | ······································ | |
| BEI 39 C-10 4.0(4)b S (cont inued) | Sec allached copies of pages from KUO, "principles of Combustion" The conflict has been and remains That Ti = 400°C @ Low presence (5m Ti = ,580°C @ Atmosphere. | (F) | { | hot spots are co prototypical exp no ignition due frictional spart autoignition ter flammability lin different for m down with addit Additional clar provided through most of Appendia information pre- accurate. | boled and beriments showed to both ks and mperature. Lower mits is also ixtures. It goes ion of NH3. ity has been h a re-write of x C. The sented is | |

It is agreed that gas and water

distributions are poorly

This section refers to the

interstitial gas and liquid

stored in the gas retaining layer, and does not address the

macro stratification in the tank. This section estimates the total

volume of gas available to burn,

understood.

5K

por

and that propagation is a function of the Tintial and Concentration, and present PV = nRT.

Comment Coding Significant

NM 20

15.0

C-12-

0 Optional F Editorial

over" or GREs observed in tank 101-SY.

The next to last paragraph of this section assumes that

stratified in the tanks, and that the gas is collected below a stratified layer and the supernatant liquid

assumption cannot/be true since we know that the waste is

(water) is above this layer. Case in point is the "roll

"gas and water are/homogeneously distributed." This

| | (5) | | | | | ' • |
|--|---|------------------|--|--|---|------------|
| | REVIEW COMMENT RECORD (RCR) | 1. | Date 5 | /10/96 | 3. Review No. | |
| | | 2. | Project N | ło. | 4. Page <u>27</u> | |
| | | | | | | r |
| din 15 D-6 Tbl D-1 S | The results in this section specifies some results fo the failure to trip drill string on excessive rpm. I not clear if this includes the dual trip value of 55 for less than 45 ft and 40 rpm for DS lengths greater than 45 ft. This should be clarified in the SA. | r t is rpm | or | 55 rpm relates t drill bit (Level Control 6.9.12.3 40 rpm relates t buckling (Level Control 6.9.12.4 | to temperature of 1 Hardware 3). to drill string 3 Administrative 4). | |
| | 1 | | | Structural limit controlled autor manually using t controls as ind 6.0 and 4.0. | ts are not matically but by administrative icated in Sections | |
| din 16 D&E gen | There are twelve items that were modeled to perform a systems reliability analysis. The results of the analysis is provided in Table D-1 for the frequency of the initiating event and the probability of subsequen failures. However, there are no details or assumption used in deriving the results. For the SA reviewer to assess the results, these details are needed to be included in the SA description. Some of the results appears to be questionable on the low side and needs be substantiated with more information. Appendix E j repeats the same results. | to pe u | 0 - K 540000 145 1450 1450 10 11 1450 10 1450 10 1450 1450 1450 1450 1450 1450 1450 1450 | Appendix D estin based on fault f references. Ap reliability num the frequency of fevent trees per Calculations are peviewer and in necessary inform | mates reliability trees given in SA bendix E uses the bers and estimates f accident using activity. e available to the nclude all the mation. | |
| din 17 E-4 Tbl E-1 row 3 E | This row discusses the leaking riser penetration releases. Under the controls credited for mitigative action, it states "limit leakage from all unused risers/pits to less than 1 in". Is this a hole size, area (units), or the amount of vacuum needed to be maintained? The control measure needs to be clarifie | an . | 14 | Accept - text w "less than 1" e diameter." | ill be amended to quivalent | |

4

| Ú., | | | | | ٠. |
|--|---|--------------|--|--|-----------|
| | REVIEW COMMENT RECORD (RCR) | l. Date | 5/10/96 3 | . Review No. | |
| | | 2. Project i | No. 4 | I. Page 28 | |
| | | | | · · · · · · · · · · · · · · · · · · · | |
| BEJ 40 G-02 2.0 Bullet 2 S | As stated in Comment BEI 39, the auto-ignition temperature is a function of pressure. The 400 C temperature is applicable to about a 5 torr pressure system. The atmospheric pressure auto-ignition temperature is above 500 C. | | See response to BI SEE EBPONGET BEIGHT FOR | EI 39. | (K. |
| art G-06 3.7 el Eq G-4k | The paragraph following Eq-G4 is unsupported. Biot is undefined, k is undefined, and time constant is, undefined. $K = \pi m e$ constant Arce Nor Defined. | O.K | They are defined | in the text. | |
| BEI 18 J-02 Para 3 E | This ¹ sentence needs help. " the mole fraction be zero" is not good English. Elsewhere in this section "be" is used incorrectly. | or | Accept - change " be " to mole frac be one " to " is | mole fraction tion is " and " one ". | |
| BEI 19 J-03 Para 3 E | Middle sentence " hydrogen caused by thermal gradients " should have the word "diffusion" after "hydrogen" | OK | Accept. | | |
| BEI 20 J-10 Para 3 O | Agree. Independent calculation by this reviewer confirms the calculated over-pressure. | or. | No action require | d. | · |
| BEI 21 J-10 3.2(3) E | a problem with plural "distance" | OK | Accept - will cha | nge to singular. | |
| BEI 22 J-11 (2) E | "beyond" is a poor choice of preposition, since 20% is "beyond 18%" and that is not what is meant. | . 6K | Replace 'beyond' than' and changed 58%. | with 'greater to < 18% and > | |

.

•4

• •

Comment Coding S Significant,

0 Optional. E Editorial

| \bigcirc | | <u> </u> |
|-----------------------------|--------------------|---------------|
| REVIEW COMMENT RECORD (RCR) | 1. Date 5/10/96 | 3. Review No. |
| | 2. Project No. | 4. Page 29 |

BE1 23 The logic supporting the conclusion that "any burn in the Last sentence in paragraph states J-11 drill string may detonate" is inadequate. There is a "therefore any burn in the drill (2) problem with an undistributed middle. The argument string may detonate." ς presented is: A- Concentrations between 18% and 58% O.K hydrogen detonate. (agree) B - Hydrogen concentration, Detonable concentration is can reach 18% in 30 minutes. (agree) Therefore, C - All obtainable in the drill string if burns will detonate. (disagree!) B has nothing to do there is a waiting time. The with A. This faulty argument is then made the e minimum time to reach the minimum detonable concentration is conclusion of the Appendix - see \P 3.3 below (BEI 24). determined. and a control is provided to eliminate the flammable gas from drill string. Therefore, there is a relation between the time and concentration. (NOT HUE BURNS MANY DOTON HTE. (NOT HUE BURNS MANY DOTON HTE. (Strictly amilted of longic 1071 çan't Putono Suggisted wordend Therefore, the pome burns may land to detoration. 5870 MAJD Donotu/ All Burns

Comment Coding

O Optional, E Editorial





| <u>(</u> | \bigcirc | <u> </u> |
|-----------------------------|--------------------|---------------|
| REVIEW COMMENT RECORD (RCR) | 1. Date 5/10/96 | 3. Review No. |
| | 2. Project No. | 4. Page 30 |

BEI 24 The conclusion section contains 6 conclusions. The first Fifth Conclusion J-11 four address over-pressure and rate of pressure rise. 3.3 S This reviewer has no argument with those conclusions. May detonate is not unsupported The fifth conclusion is that "any burn may detonate and the unknowns associated with because the rate of pressure rise is very high during the FGT's make it a prudent detonation". That is an unsupported statement and is conclusion until more data can be worse logic than made in the body of the Appendix (see obtained. comment BEI 23 above. The final conclusion is totally unsupported and appears for the first time in the Final Conclusion Supported by conclusion paragraph - Consequently, the structure of the last paragraph of 3.2 drill string may fail and cause partial dome failure ! The document should present facts and documents to The relation between drill string support the statement that a 15 bar over-pressure in a detonation is established in 'oik confined volume of approximately, 20 liters in a 4 Ml Appendix I. first bullet under chamber would collapse the dome // It has not even been 2.1. / This is the Portion of the commute that and Should be addressed Should be shown, in this appendix that a hydrogen detonation would cause failure of the drill string // The last sentence will be revised and "partial dome collapse" will be eliminated. Section 3.2 presents arguments (Ref. 9. 10) why detonation is likely. The geometry (diameter and length) is suitable for the wave speed to reach detonation wave speed (Ref. 9). Ref. 10 presents studies detonation issue in H2-N20 mixtures to determine necessary concentration etc. The words "and cause partial dome failure" have been deleted from the text.

| \bigcirc | | 0 | | ······································ |
|---|-----------------------------|---------|---|--|
| | REVIEW COMMENT RECORD (RCR) | 1. Date | 5/10/96 | 3. Review No. |
| | | 2. Praj | ect No. | 4. Page 31 |
| | | | | |
| BEI 24 J-11 3.3 S (continued) | | Ov | The main conclu there is enough the drill strin detonation "may accident was us bounding releas controls were a prevent hydroge the drill strin There was also detonation wave the waste as a deflagration wa it is agreed th structural dama without further analysis. Howe in the drill st mitigated regar extent of the s | sion is that if concentration in g and spark "occur. This ed to estimate the es. The necessary lso determined to n accumulation in g. a concern about a propagating into detonation or we. Nonetheless, at a dome ge is a step taken structural ever, a detonation ring must be dless of the structural damage. |

.

4

Comment Coding

O Optional,

l, E Edi**t**orial

| | REVIEW COMMENT RECORD (RCR) | 1. Date | 5/10/96 | 3. Review No. | |
|---------------------------------------|--|--------------------------------------|---|--|--|
| | | 2. Project | No. | 4. Page 32 | |
| NM 21 L-1 1.0 (Item 3.) S | This section claims to derive "realistic accident frequencies". It should include a description/methodology on how they could be derived without full identification of the required mitigatic controls, procedures, and of SSCs. | ve = " 7 5 OK | The actual stat Appendix L is " model is merely in comparing th consequences ag Guidelines that realistic estim and allow for c estimates for p there are a lac Derive is not t here. Appendix L defi probability of | ement in 1.0 of The probabilistic intended for use e bounding ainst the WHC Risk require a hate of frequencies ualitative henomena for which k of data" he operative word nes the a GRE and of | |
| NM 22 L-4 2.1 0 | The fifth paragraph of this section postulates that the retained gas volume is proportional to the waste volume" and concludes " no GRE during intrusion into tanks would be interpreted as encouraging." The ret gas volume is actually proportional to 1) the concentration of beta and alpha emitters in the wast layer adjacent to water and radiolytically decomposi waste, 2) the water volume, and 3) the layering characteristics of sediments both above and below th water-Sr-90 and Cs-17 interface. The Sr-90/Water interface produces hydrogen gas. The Sr-90, Cs-137 interface with the balance of the waste produces the | " o the ained ng (0 K ne | exceeding the L The frequency of determined in S While the rate is a function of load, the gas n of a function of properties. For properties, lan would retain la gas. | FL in the dome. FL in the dome. of accidents are Section 4.0. of gas generation of the radiolytic retention is more of the physical or a given set of rger waste volumes arger amounts of | |

.

Comment Coding S Significant O Optional, E Editorial

| <u></u> | | | |) | <u></u> |
|--|---|--|---|--|---------------------------------------|
| | REVIEW COMMENT RECORD (RCR) | 1. Date | 5/10/96 | 3. Review No. | |
| | | 2. Project | No. | 4. Page 33 | |
| | | | | | |
| din 18 L-8 3.0 gen S O V V V V V V V V V V V V V | A quantitative likelihood for a gas release volume and gas release rates are discussed in this section. There are two comments regarding this discussion. 1) the terms used to describe the likelihood have established frequency of occurrences assigned to them f radiological risk guidelines (See Section 5.5.1). and t frequency of occurrence used in the gas release likelihood does not use the same frequency of occurrenc ranges. Wherever these terms are used, they should hav the same definition. Either make the frequency of occurrences consistent or reclassify the events using "other" descriptive terms. 2) the largest gas release that has been observed in o of the tanks is 10,000 cubic feet and this release has been assigned a frequency of occurrence of less than 1E-6. If an event has occurred, then it is difficult t classify the event as having such a low frequency of occurrence without providing a better justification for the event. | 1. A Constrant Alonical CLATORE SHELL 1. A Constrant Alonical CLATORE SHELL 1. A Constrant Alonical CLATORE SHELL 1. ACCARTENT Alonical CLATORE SHELL 1. A Constrant Shere Shere D 1. A Constrant Shere Shere Shere D 1. A Constrant Shere Shere Shere Shere Shere D 1. A Constraint Shere Sher | Table L-1 gives bins. The word qualitative sen numerical impli- page gives the and indicates t 10,000 ft ³ relea In this appendi quantified base probabilities a (as done in Sec NOTE: Tank 101- list of t SA is cur (that is 10,000 ft Such a re an incred | the preliminary s are used in a se rather than cations. Next probability curves hat probability of ase is 1E-3. x, the bins are d on the nd not frequencies tion 5.0). SY is not on the anks for which the rently applicable the tank where was observed). lease in an SST is ible event. | 2. Provbe 735 704815 5 Just flog Tron |
| NM 23 L-12 3.2 0 | The second paragraph of this section states "Adding ammonia (that is also known to exist) reduces the LFL b reduces the amount of hydrogen volume in the release." This statement would be true if the production rates of the two gases were inter-dependent, and the "gas bubble is at atmospheric pressure. However, the pressure in t "gas bubble" continues to increase until buoyancy is achieved, and the total production of each gas is cumulative (total pressure is equal to the sum of their partial pressures). The conclusions in this section should be consistent with this principle. | he | Accept - The te clarified for a volume, saying ammonia fractio gas would imply fraction. Thus volume is neede LFL. | ext will be given release that a larger in the release smaller hydrogen a larger release d to reach the | |

•

щ,

÷

| <u> </u> | | | | <u> </u> | • • |
|---|---|------------------------|---|--|--------|
| | REVIEW COMMENT RECORD (RCR) | 1. Date | 5/10/96 | 3. Review No. | |
| | | 2. Project | No. | 4. Page 34 | |
| | I | | | | · |
| din 19 N-1 2.1 E O O N-11 Tb1 N-2 S | This section specifies that the 50 ton live load limit is not exceeded by the RMCS equipment. However, on the nex page, Table N-1, a weight break-down is provided that indicates that the total potential weight exceeds the limit by about 9 ton. On page 2-25, a similar table indicates that the loads would exceed the limit by about 10 ton. Some rationale should be provided to indicate the acceptability of the loads and the discrepancy between the tables should be eliminated. There we the force limits as determined by the drill string analysis. The limits are a function of the drill string length. These limits should be reflected in Table 2-7 that provides the operational parameters for the drilling operation. Only the maximum limits are provided and not the limits as a function of drill string length. For example, the limits for drill string greater than 45 ft. would be less than 650 lbs force and less than 40 rpm. The limits on the force and rpm should be included in Section 2 consistent with that provided in the table at the end of Section 6, pages 6-34, -35 for operation of the drill string greater than 45 ft. | st Turo VISE SA- | Accept - Not al listed in Table tank. See 4.6. discrepancy betw will be elimina Only one table additional expl provided under loading. <u>PZ</u> - Accept - line w Table 2-7. | l of the equipment N-1 is on the 1. The ween the tables ted. is now used and anation is static dome $Z_T = P N - Z - P$ ill be added to | |
| BE1 32 R-4 4.0 E | line 7, the 7 is dropped off in ^{137m} Ba | er | Accept - text w | ill be modified. | |
| BEI 33 R-5 4.1 E | Last sentence in $\P1$. the power of the number 8.71 has been dropped. (8.71 E-4 Ci/g). | af | Accept - text w | nill be modified. | |

.

•

Comment Coding

O Ontional E Ec

F Fditorial

| REVIEW COMMENT RECORD (RCR) 1. Date 3. Review No. BEI 34 This reviewer has performed an independent calculation for the radiation at the surface of the filter housing and agrees with the magnitude of the number. However the filed at the surface is given to 3 significant figures (317 Mr/hr). This implied accuracy greatly exceeds the uniformity of the filter geometry. Fields can be expected to vary by as much as a factor of 10 (greater or less than the 317 mR/hr). depending on the uniformity of deposition and the exact region of the filter one is 1 cm from. On the Scingle Digit Accuracy in this section the functional requirements for the filtements for the filter gas sensors is described. The sensors were not intended to detect toxic gases. The sensors were not intended to the presence of combustible gas and would be minimally effective in detecting the presence of toxic gases. The SMC detector appears to marginally satisfy the functional requirements for the selection of this detector should be provided in this Appendix. It would seem prudent to In this septendix. It would seem prudent to | <u> </u> | \bigcirc | ···· |
|---|--|--|--|
| 2. Project No. 4. Page 8F1 34 R-5 4.1 0 This reviewer has performed an independent calculation for the radiation at the surface of the filter housing and agrees with the magnitude of the number. However the field at the surface is given to 3 significant figures (317 Mr/hr). This implied accuracy greatly exceeds the uniformity of the filter geometry. Fields can be expected to vary by as much as a factor of 10 (greater or less than the 317 mR/hr). depending on the uniformity of deposition and the exact region of the filter one is 1 cm from. Both results are above allowable limits (100 mr/h). Therefore. accuracy considering uncertainty does not change the conclusion. The necessary controls to mitigate the event have been established and are specified in Section 6.0. din 21 U-gen gen gen workertid wurker flammable and toxic gas sensors is described. The sensors described include the Wittaker Cell and the SMC combustible gas sensor. These sensors would only detect the presence of combustible gas sensor. The rationale for the selection of this detector should be provided in this Appendix. It would seem prudent to The sensors were not intended to detect or appears to marginally satisfy the functional requirements for the combustible gas sensor. The rationale for the selection of this detector should be provided in this Appendix. It would seem prudent to | REVIEW COMMENT RECORD (RCR) | 1. Date 5/10/96 | 3. Review No. |
| BEI 34 R-5 4.1 0 This reviewer has performed an independent calculation for the radiation at the surface of the filter housing for the radiation at the surface of the filter housing adgrees with the magnitude of the number. However the field at the surface is given to 3 significant figures (317 Mr/hr). This implied accuracy greatly exceeds the uniformity of the filter geometry. Fields can be expected to vary by as much as a factor of 10 (greater or less than the 317 mR/hr), depending on the uniformity of deposition and the exact region of the filter one is 1 cm from. Both results are above allowable limits (100 mr/h). Therefore, accuracy considering uncertainty does not change the conclusion. The necessary controls to mitigate the event have been established and are specified in Section 6.0. din 21 U-gen gen gen o for the presence of combustible gas sensors. The gen sensors described include the Wittaker Cell and the SMC combustible gas sensor. These sensors would only detect the presence of combustible gas and would be minimally effective in detecting the presence of toxic gases. The SMC detector appears to marginally satisfy the functional requirements for the combustible gas sensor. The rationale for the selection of this detector should be provided in this Appendix. It would seem prudent to Not sensors would not be provided in this Appendix. It would seem prudent to | | 2. Project No. | 4. Page 35 |
| BEI 34 R-5 a)This reviewer has performed an independent calculation for the radiation at the surface of the filter housing and agrees with the magnitude of the number. However the field at the surface is given to 3 significant figures (317 Mr/hr). This implied accuracy greatly exceeds the uniformity of the filter geometry. Fields can be expected to vary by as much as a factor of 10 (greater or less than the 317 mR/hr). depending on the uniformity of deposition and the exact region of the filter one is 1 cm from.Both results are above allowable limits (100 mr/h). Therefore, accuracy considering uncertainty does not change the conclusion. The necessary controls to mitigate the event have been established and are specified in Section 6.0.din 21 u-gen gen gen o VIn this section the functional requirements for the flammable and toxic gas sensors is described. The sensors described include the Wittaker Cell and the SMC combustible gas sensor. These sensors would only detect the presence of combustible gas and would be minimally effective in detecting the presence of toxic gases. The SMC detector appears to marginally satisfy the functional requirements for the combustible gas sensor. The rationale for the selection of this detector should be provided in this Appendix. It would seem prudent toBoth results are above allowable limits (100 mr/h). Therefore, accuracy considering uncertainty does not change the conclusion. The necessary controls to mitigate the event have been established and are specified in Section 6.0.din 21 u-gen gen o u-fortIn this section the functional requirements for the sensors described include the Wittaker Cell and the SMC combustible gas and would be minimally effective in detecting the presence of toxic gases. The <br< th=""><th></th><th></th><th></th></br<> | | | |
| din 21 U-gen gen 0 1 1 1 1 1 1 1 1 1 1 1 1 1 | BEI 34 R-5 4.1 0 1 1 1 1 1 1 1 1 1 1 1 1 1 | culationBoth reshousinglimits (However theaccuracyfiguresdoes notceeds theThe necebemitigate(greater orestablisformity ofSectionone is 1 cmwarta | sults are above allowable (100 mr/h). Therefore. (considering uncertainty) change the conclusion. essary controls to the event have been shed and are specified in 6.0. |
| reassess the use of the SMC detector to satisfy the flammable gas detection functional requirements. | din 21 U-gen gen 0 1 1 1 1 1 1 1 1 1 1 1 1 1 | r the The The d the SMC nly detect inimally ases. The functional The hould be ent to y the S. | sors were not intended to toxic gases. The sensors e requirements of Appendix |

THE TITLE OF APPENDIX & IS "Functions the Rodon Remains For FLAMMABLE AND TOXIC GAS SENSORS THIS SECTION Show PROVIDE THE "BEDRUKEMENTS" FOR THE TOXIC GAS SENSORS AND A DESCREPTION OF HOW THOSE FOODUREMENTS ARE IMPLEMENTED. IT STATES THAT THE MAJOR TOXIC GAS IS AMMONIN.

Comment Coding

O Optional E Editorial

| | | <u> </u> |
|-----------------------------|--------------------|---------------|
| REVIEW COMMENT RECORD (RCR) | 1. Date 5/10/96 | 3. Review No. |
| | 2. Project No. | 4. Page 1 |

INTERIM OPERATIONAL SAFETY REQUIREMENTS FOR ROTARY MODE CORE SAMPLING IN FLAMMABLE GAS SINGLE SHELL TANKS

| Hanford | SITE, | RICHLAND | WASHINGT | ON |
|---|--|-----------------------------------|----------|----|
| and the second secon | the second s | فيعجب المستحد والشيادي ومستحد الأ | | |

| 12. Item | 13. Comment(s)/Discrepancy(s) (Provide technical Justification for the comment and detailed recommendation of the action required to correct/ resolve the discrepancy /problem indicated.) | 14. Hold Point | 15. Disposition (Provide justification if NOT required.) | 16. Status |
|-----------------------------|--|----------------------|---|-----------------|
| din-1 SR 3.7.1.2 S | A calibration of the hydrogen detector should be performed at initial setup at each location and every month thereafter. The calibration should test the shutoff electronics as well as the sensor reading. This includes both the total concentration limit as well as the rate of concentration increase limit. | Sparyra 17-16-16 | Per Section 6.0 of the SA Wittaker Sensor - initial and monthly SMC Sensor - initial and daily RMCS Instrumentation- every six months (including rate of rise in hydrogen concentration) - Contact of the subscription | or_ cuperted |
| din-2 SR 3.7.1.3 S | The system response time for the Flammability. Detector should be tested the same as is required for the Hydrogen Detector. | See Charge | Only one system response time is required for both sensors. Both detectors will be tested and calibrated to the same very standard. | ्रत्वमंडच्च |
| din-3 SR 3.7.4.1 S | The nitrogen purge flow is supplied to the drill string purge gas, the riser sleeve annulus, the hydrostatic head of the drill string, the shielded receiver, and the sample receiver weather cover. Bypass to the purge flow is influenced by each individual tank setup for the riser sleeve and the drill string purge flow. A surveillance test for bypass to the purge flow should be performed after each setup and not wait for 6 months for determining the bypass flow to these features. This will ensure that the purge flow is installed correctly for each tank setup. | | A 6 month test was specified after examining data on purge gas system components and function. The physical design of the system prohibits an incorrect hookup. The hookup involves different size Hansen quick disconnect fittings. Only one arrangement is physically possible. This is a design safety feature. Shourd BE DESCRED IN S.A. | |

Comment Coding

f .

 $\frac{C_{2}(\alpha_{1})}{\frac{1}{2}} \frac{C_{2}(\alpha_{1})}{C_{1}} \frac{C_{2}(\alpha_{1})}{C_$

RESPONDETIME FOR FRAMMAERING DETECTOR NEEDS TO BE IDENTIFIED

| | \mathbf{C} \mathbf{O} |) |) ` • • • |
|----------------------------|---|---|------------------|
| | REVIEW COMMENT RECORD (RCR) | 1. Date 3. Review No. 5/10/96 | |
| | | 2. Project No. 4. Page 2 | |
| | | | |
| din-4 LCO 3.7.5 S | The Rotary Drilling Parameters were selected to prevent an over temperature condition from occurring during the waste drilling operation. Other limitations are defined in Appendix N which need to be factored into establishing limits on these parameters. Buckling of the drill string when the drilling greater than 45 ft, puts a limit on the drill force of less than 650 lbf rather than 750 lbf. Drill string resonance when drilling greater than 45 ft. imposes a rotational frequency limit of 40 rpm rather than the 55 rpm specified. The conditional limits for drilling greater than 45 ft should be factored into the LCO. The LCO on the penetration rate should be clarified to indicate that the penetration rate commences when the drill string is at the surface of the waste in the tank. | Temperature is not the primary concern for the drill string over 45 ft. The requirements for a drill string over 45 ft. are Level 3 (item 6.9.12.2 and 6.9.12.14) and only Level 1 controls are in this LCO. Applicability statement is RMCS Waste Intrusive Operations which occurs at the surface of the waste. (See also previous DIN-4 comment in the SA comments) | 01K- |
| BEI-1 3.7.5 S | Action A. Required Action. It appears that the required action does not prevent the restart into a potentially dangerous situation. The 10 minutes is OK, but it seems as if an additional check, verification, or other evaluation should be made prior to resumption of operations. | Automatic trips due to the varied nature of the waste are expected, the 10 min. wait is to assure bit and associated waste have properly cooled. On a restar operators will be expected to exercise all required monitoring and precautions. No other evaluations are necessary to operate within the safety envelope established by the credited design features and administrative controls. | t or |

-4

Comment Coding

 \sim

A Contraction of the second se

0-+10-1

E Editorial

| <u> </u> | | <u>`````</u> |
|-----------------------------|-----------------|----------------|
| REVIEW COMMENT RECORD (RCR) | 1. Date 5/10/96 | 3. Review No. |
| | 2. Project No. | 4. Page . 3 |

| the second s | | ······································ | | |
|--|--|--|--|------|
| din-5 SR 3.7.5.1 S | The surveillance section should specify how many measurements are required to be operable for each of the drill string parameter measurements. This includes the RPM, the force, and the penetration rate measurements. Is only one required for each? | | RPM - SR 3.7.5.1 states: "There shall be dual RPM measurement sensors both of which shall be operable when rotating the drill bit below the waste surface." | |
| | 1 | | Down force and penetration rate: Only one sensor is required. During the calibration process for RPM, downforce. and penetration rate sensors are set at zero and full scale and then one or more intermediate points are measured. Therefore, a minimum of 3 points are measured for each sensor. | σĽ |
| din-6 LCO B3.7.1 E | The first sentence should be revised as follows "tank in question is does not exceed 25%" | | Accept. | oi.C |
| din-7 B3.7.3 A.1 E O.K | The negative pressure in the tank can only be restored by the cessation of the exhauster. The other portions of the RMCS operation will tend to reduce the negative pressure, such as the purge flow. This should be clarified in the writeup. Is only one vacuum breaker required to be operational on the tanks during drilling operations? | | Accept - words will be added to $B_{3.7.3}$. Only 1 vacuum breaker required. $B_{3.7.3}$ | |
| | | | THE NO BE NOT NO THE . Were SHALTINGTER. | |

.

Comment Coding

0 Outsional E Editorial

| | \mathbf{C} | <u>)</u> | | \sim | 14 - 14 - |
|------------------------------|--|----------|---|---|-----------|
| | REVIEW COMMENT RECORD (RCR) | 1. | Date 5/10/96 | 3. Review No. | |
| | | 2. | Project No. | 4. Page 4 | |
| ······ | p | | | | |
| din-8 B3.7.4 LC0 0 | Because the tank purge flow bypass is dependent on the setup with each tank, the leak rate should be tied to each tank setup rather than just every 6 months. A faulty setup could significantly increase the purge flow bypass. | | The unique hookup featur din-3 prevents faulty ho Accept addition of words | re discussed in pokups of site of the site of the site of the site of the site of the site | |
| 0.0~ | B.3 (E) Change this sentence to read: "A temperature outside the bounds of the safety assessment must be corrected prior to resuming operations." | | | • • | |
| din-9 B3.7.5 APPL 0 | This section should describe the "other" restrictions that must be imposed on the Rotary Drilling Parameters". This includes the restrictions imposed by the drill string buckling and resonance when drilling greater than 45 ft. These parameter restrictions are just as important as the ones specified. | | Those are Level 3 requirent IOSR changes im 1 changes. Level 3 con lesser importance that Level 3 requirements prodepth and will be implein operating procedures. | rements. The plement the Level trols are of Level 1 controls. ovide defense in mented into | 84 |
| din-10 5.31.10 S | It is not clear why the restrictions are imposed on the vertical height of the exhauster inlet and not the exhauster outlet. The height of both of the stacks are important for the proper dispersal of the potentially radioactive and combustible gases and the potential for igniting of the combustible gases. | | The exhauster outlet he design/configuration co The restriction propose of the SA is only on th inlet that could become various accident scenar | ight is fixed by ntrol. d by Section 6.0 e breather filter the outlet during ios. | ฮาL |

• .

.

4

:

.

Comment Coding

0 Ontional

F Editorial

| | ý | <u>(</u> |) | | | | , *• * • |
|---|-----------------------------|--|---|--------|---|---|-----------------|
| | | REVIEW COMMENT RECORD (RCR) | | 1. Dat | e 5/10/96 | 3. Review No. | |
| | | | | 2. Pro | ject No. | 4. Page 5 | |
| - | | | | | | · · · · · · · · · · · · · · · · · · · | |
| | din-11 5.31. gen S | The administrative controls for the Rotary Mode Core Sampling should include the additional requirements that must be met prior to satisfy the assumptions cited in the SA: 1) a changeout of the HEPA filter on the exhauster outlet following each tank drilling operations (page 4-51) and 2) the checking on the thunderstorms or lightning within the 50 miles of the planned drilling site for the potential of drilling operation shutdown (page 4- 57). | | 1. | 6.9.6.5 requires of HEPA filter ho changeout at a co 100 mr/hr. 6.6.10.3 covers 1 thunderstorms. These are Level 3 The current IOSR reflect Level 1 c per the SA, all c implemented via p RMCS operations. | daily monitoring using and ntact reading of ighting and controls. modifications only ontrols. However, ontrols will be rocedures prior to | DT |

• .

٠

4

.

•

O Optional E Editorial

B37-36 To = 300°K TF Tr= 400°C = 400+300 = 700°K AR = 700 × 14.7psi = 3575i 300 × 14.7psi = 3575i This assume 14.775i SAR used 400° cas 1 gnitin Temp, 1 not 580° C

Principles of Combustion

 $\left(f_{ij}^{min}\right)$

Kenneth Kuan-yun Kuo

Distinguished Alumni Professor Department of Mechanical Engineering The Pennsylvania State University University Park, Pennsylvania

A WILEY-INTERSCIENCE PUBLICATION JOHN WILEY & SONS New York • Chichester • Brisbane • Toronto • Singapore

BE1-30

Review of Chemical Kinetics

or

144

 $t = 64 \times 10^{-8} \sec \approx 10^{-6} \sec = 1 \,\mu \sec$

This is certainly a very rapid combustion process. In an actual combustion process, not all reactions are chain-branching. However, the reaction rate is still very fast even for a very small portion of chain-branching reactions. For a combustion process in which 1% ($\alpha' = 1.01$) of the reactions are chain-branching, the time required for all the molecules in the volume to react would be only

$$\frac{\alpha'^{N+1} - 1}{\alpha' - 1} = \frac{1.01^{N+1} - 1}{0.01} = 10^{19} \text{ molecules/cm}^3$$

$$N = 3934$$

$$t = 3934 \times 10^{-8} \sec \approx 40 \,\mu \text{scc}$$

This is still a very fast reaction.

In general, branched-chain reactions and explosions can be studied by considering the following chemical kinetics:

$$\begin{array}{c}
M \xrightarrow{k_1} R & \text{chain-initiating} \\
R + M \xrightarrow{k_2} \alpha' R + M^* & \text{chain-branching} \\
R + M \xrightarrow{k_3} P & \\
R \xrightarrow{k_4} M & \\
\text{(on wall)} & \\
R \xrightarrow{k_5} \text{ nonreactive species}
\end{array}$$
chain-terminating (2-58)

The rate equation (applying the steady-state assumption) is

$$\frac{dC_R}{dt} = 0 = k_1 C_M + (\alpha' - 1) k_2 C_R C_M - k_3 C_R C_M - k_4 C_R - k_5 C_R \quad (2-59)$$

Solving for C_R gives

Explosion Limits

145

The rate of change of the product concentration is given as

$$\frac{dC_{\rho}}{dt} = k_{3}C_{R}C_{M} = \frac{k_{1}k_{3}C_{M}^{2}}{k_{3}C_{M} + k_{4} + k_{5} - k_{2}(\alpha' - 1)C_{M}}$$
(2-61)

The quantity $k_2(\alpha' - 1)C_{\mu}$ is positive; as its value increases it tends to decrease the denominator in Eq. (2-61). The critical value of α' is given as

$$\alpha'_{\text{critical}} = 1 + \frac{k_3 C_M + k_4 + k_5}{k_2 C_M}$$
(2-62)

and we have

 $\alpha' \ge \alpha'_{\text{critical}} \implies \text{chain-branching explosion}$ $\alpha' < \alpha'_{\text{critical}} \implies \text{no explosion}$

However, it is important to note that for some actual explosion processes, because the concentration of R does not remain small, the steady-state approximation may not be valid. Other reaction steps may also become important. The postulated reaction kinetics in Eq. (2-58) may not always be applicable during an explosion.

8 EXPLOSION LIMITS

8.1 $H_2 = O_2$ System

It is experimentally observed that a pressure vessel containing hydrogen and oxygen under the conditions shown in Fig. 2.11 will explode as the pressure is raised. Intuitively, one would assume that as the pressure is raised, the concentration of free radicals would be increased, which would lead to an explosion. However, an explosion is also experimentally observed as the pressure is lowered.



 $k_1 C_M$

(7 60)



The existence of explosion limits in a closed vessel can be understood very simply from qualitative considerations of competition between chain-breaking and chain-branching reactions on surfaces and in the gas phase. Typical experimental results for hydrogen-oxygen mixtures are plotted schematically in Fig. 2.12. The first, or lower, explosion limit occurs at roughly the same pressure over a relatively large temperature range. The lower explosion limit is determined by a balance between the removal of chain carriers on the surface (wall effect) and production of chain carriers by gas-phase reactions. In this low pressure range, the number of collisions and the rate of production of chain carriers are both low. From Eq. (2-62) we know that

$$\alpha'_{\text{critical}} = (1 + C_1) + \frac{C_2}{C_M} \text{ and } C_M \propto p$$

The lower the pressure, the larger the value of $\alpha'_{critical}$, and hence the smaller the chance for explosion. However, none of these analyses can predict the exact location of the explosion limits; they can only explain the mechanism of reaction in each region.

As the pressure is raised, the rate of production of chain carriers by gas-phase reactions increases to the point at which surface destruction is no longer sufficient to prevent a branching explosion. The lower explosion limit defines the condition at which chain branching in the gas phase is balanced by chain breaking at the surface.

Explosion Limits

kinetics:

Recommended by a number of investigators in the 1920s:

$$H_2 \rightarrow 2H + 106 \text{ kcal/mole}$$
 (dissociation) (A)

Suggested by Lewis and von Elbe:6

$$H_2 + O_2 + M \rightarrow H_2O_2 + M^*$$

 $\downarrow 2OH + 51 \text{ kcal/mole} (B)$

and after the OH radical is generated,

 $OH + H_2 \rightarrow H_2O + H - 15 \text{ kcal/mole}$ $H + O_2 \rightarrow OH + O + 16 \text{ kcal/mole}$ $O + H_2 \rightarrow OH + H + 2 \text{ kcal/mole}$

The reaction proposed in (A) is more endothermic than the reaction in (B). However, reaction (B) requires a third-body reaction which is not as likely as the dissociation reaction. At low temperatures, therefore, reaction (B) is more likely, and at high temperatures reaction (A) is more likely.

At high pressures the second explosion limit is approached. The existence of the second explosion limit is readily explained if the three-body reaction

$$H + O_2 + M \rightarrow HO_2 + M \tag{2-63}$$

is added to the scheme. In this reaction the symbol M denotes any third molecule that stabilizes the combination of H and O_2 . Because the metastable intermediate hydroperoxide radical (HO_2) is thought to be relatively unreactive, it is able to diffuse to the wall. HO₂ becomes a vehicle for the destruction of free valences, and therefore the above reaction is considered a chain-breaking reaction. With increasing pressure, the frequency of ternary collisions $H + O_2 + M$ increases relative to the frequency of binary collisions $H + O_2$. There is therefore a pressure above which the rate of removal of free valences exceeds the rate of formation of free valences by chain-branching reactions, and the second explosion limit is established. The destruction of the HO₂ molecule on the wall can be expressed by the reaction

$$HO_2 \xrightarrow{\text{wall}} \frac{1}{2}H_2 + O_2$$
$$HO_2 \xrightarrow{\text{wall}} \frac{1}{2}H_2O + \frac{1}{4}O_2$$

the second second

Review of Chemical Kinetics

At some pressure above the second explosion limit, however, HO_2 is assumed to participate in the chain propagation process according to the reaction

$$HO_2 + H_2 \rightarrow H_2O_2 + H$$

Therefore, above a critical pressure, there is a rapid increase in the number of radicals. This critical pressure defines the third explosion limit. Now H_2O has a bond frequency very close to that of HO_2 , the structures being

so H_2O is an excellent third body for the reaction given in Eq. (2-63). The region for pronounced effect of H_2O and HO_2 is indicated in the explosion-limit diagram. It is useful to note that for $T > 600^{\circ}$ C, HO_2 cannot be stabilized and therefore explosion is observed at all pressures.

The treatment of explosion limits in flow systems can be worked out through an extension of the methods developed for closed reaction vessels.

8.2 CO - O₂ System

As shown in Fig. 2.13, mixtures of carbon monoxide and oxygen also exhibit the phenomenon of explosion limits. The chain-initiating reaction is

 $CO + O_2 \rightarrow CO_2 + O$, $\Delta H_r = -9 \text{ kcal/mole}$ (exothermic)

This initiating reaction is hard to achieve without H₂. Lewis and von Elbe⁶



Figure 2.13 Explosion limits of a stoichiometric

Rate Laws for isothermal Reactions Utilizing Dimensionless Parameters 149 have suggested that the explosion limit is essentially controlled by the reactions

$$M + CO + O \rightarrow CO_2 + M$$

$$M + O + O_2 \rightarrow O_3 + M^* \quad \text{(exothermic)}$$

$$O_3 + CO \rightarrow CO_2 + 2O \quad \text{(very rapid)}$$

$$O_3 + CO + M \rightarrow CO_2 + O_2 + M^*$$

It should be noted that the behavior of the CO-O₂ system is changed radically by the admixture of small amounts of H₂ or H₂O, and that the rate-controlling reaction mechanism now involves H, OH, H₂, HO₂, and H₂O as well as O, O₂, CO, CO₂, and O₃.

The water-gas reaction is most probably surface-catalyzed:

$$CO + H_2O \Rightarrow CO_2 + H_2$$
 (surface)

and should be followed by the surface reaction in the hydrogen-oxygen reaction

$$H_2 + O_2 \rightleftharpoons H_2O_2$$
 (surface)

Chain carriers are then provided by the dissociation at H_2O_2 in the gas phase:

 $H_2O_2 \rightarrow 2OH$ OH + CO $\rightarrow CO_2 + H$ H + O₂ $\rightarrow OH + O$

9 RATE LAWS FOR ISOTHERMAL REACTIONS UTILIZING DIMENSIONLESS PARAMETERS³

For the sake of brevity, the following discussion will be restricted to a pair o opposing chemical reactions. Generalization to chain reactions can be made without difficulty.

For the most general opposing chemical reactions,

$$\sum_{i=1}^{N} v_i' M_i \rightleftharpoons_{k_k}^{k_f} \sum_{i=1}^{N} v_i'' M_i$$

As seen in Section 2.5, the net rate of production of species M, is given by U

BEI -34

Sheet1

| Ignition Temp | erature, Hai | ndbook of P | hysics and | chemistry : | = 580 - 590 | C | | |
|---------------|--------------|-------------|------------|-------------|--------------|------------|-----------|-----------------|
| | | | | | | | | |
| Air + 2H2 + C |)2> Air + 2 | 2H2O + HE | AT | | | | <u> </u> | |
| 2% H2 | | <u> </u> | | | <u> </u> | | | |
| | <u> </u> | | | | | · . | air | |
| Reactants | Ср | Hfg | Hv | m | mol wt | vol % | mol % | |
| | cal/mol/K | kcal/mol | kcal/mol | mol | ļ | | 1 | |
| | | | ļ | | L | | ļ | |
| N2 | 7.28 | 0 | | 78 | 28 | | ļ | 78 |
| 02 | | 0 | <u> </u> | 21 | 32 | | ļ | 21 |
| H2 | | 0 | | 2 | 2 | 0.019802 | ļ | . <u> </u> |
| | | [| sum | 101 | ļ | | | |
| PRODUCTS | | | | | | | | |
| PRODUCIS | | | <u> </u> | | | | | |
| H2O g | 8.82 | 57.8 | 9.72 | 2 | 18 | | | |
| N2 | 6.9 | 0 | | 78 | 28 | | 1 | |
| 02 | 7.07 | 0 | | 20 | 32 | | | |
| | delta H = - | e16*c16 | | 115.6 | | | · · | |
| | | | | | | | | |
| | delta T = d | elta H/sum(| Cp*m) | 165.7966 | | | | |
| | | e21/(b16:1 | 8 *e16:18) | | | | | |
| Air + 2H2 + 0 | 2> Air + 2 | 2H2O + HE | <u>АТ</u> | | | | | |
| 4% H2 | | | | | | | | |
| | 1 | | | | | | air | |
| Reactants | Ср | Hf g | Ηv | m | moi wt | vol % | mol % | |
| | cal/mol/K | kcal/mol | kcal/mol | mol | | | | |
| N2 | 7 28 | 0 | | | 28 | | | 78 |
| 02 | 1.20 | 0 | | 0 061538 | 20 | | | $\frac{70}{21}$ |
| H2 | | 0 | | 2.001000 | 2 | 0.040848 | | <u> </u> |
| | | | sum | 48,96154 | | 0.010010 | | |
| | | | | | | | | |
| PRODUCTS | | | | | | | | |
| H2O g | 9 | -57.8 | 9.72 | 2 | 18 | | | |
| N2 | 6.916 | 0 | | 37 | 28 | | | |
| 02 | 7.072 | 0 | | 8.96 | 32 | | | |
| | delta H = -e | e16*c16 | | 115.6 | | | | |
| | | | | | Ignition Ter | np = 580 C | - <u></u> | |
| | delta T = de | elta H/sum(| Cp*m) | 342.7652 | | | | |
| | | e21/(b16:1) | 8 *e16:18) | | | | | |

٩

 \bigcirc

Ł

Sheet1

| [| 1 | 1 | 1 | 1 | 1 | 1 | <u></u> | _ |
|--|---------------|--------------|-----------------|----------|--------------|------------|---------------------------------------|----|
| · · · · · · · · · · · · · · · · · · · | + | <u> </u> | <u> </u> | 1 | † | | <u> </u> | |
| Air + 2H2 + C |)2 -> Air + : | | AT | | | | <u> </u> | |
| 30 % H2 | T | 1 | Ţ | <u> </u> | 1 | | 1 | |
| ************************************** | 1 | 1 | | | | | air | |
| Reactants | Cp | Hf g | Hv | m | mol wt | vol % | mol % | |
| | cal/mol/K | kcal/mol | kcal/mol | mol | | 1 | | |
| | | | | | | | | |
| N2 | 7.28 | 0 | | 3.714286 | 28 | | | 7 |
| 02 | | 0 | | 1 | 32 | | | 2 |
| H2 | | 0 | <u> </u> | 2 | 2 | 0.297872 | | |
| | | | sum | 6.714286 | | | | |
| PRODUCTS | | | | | | [| | |
| | 1 | [| | | 1 | · · · | | |
| H2O g | 8.2 | -57.8 | 9.72 | 2 | 18 | | | |
| N2 | 7 | 0 | | 3.71 | · 28 | | | |
| 02 | 7.36 | 0 | | 0 | 32 | | | |
| | | | | 5.71 | | | | |
| | | | | | | | | |
| ····· | delta H = - | e16*c16 | | 115.6 | | | · · · | |
| | dalta T = d | | [| 2720.246 | | | | |
| | | | Op III) | 2120.340 | | | | |
| | | ez 11(010.1 | <u>6 e10.10</u> | | | | | |
| ····· | Vol Change | P = (mol2/m) | ol1) (delta 1 | [| 7 734182 | | | |
| | T of ondige | | | | 1.704102 | | | |
| | <u> </u> | | | | | | | |
| Air + 2H2 + O | 2> Air + 2 | 2H2O + HE/ | Α.Τ | | | | | |
| 6% H2 | | | | · · · | | | | |
| | | | | | | | air | |
| Reactants | Ср | Hfa | Hv | m . | mol wt | vol % | mol % | _ |
| | cal/mol/K | kcal/mol | kcal/mol | mol | | | | |
| | | | | | | | | |
| N2 | 7.28 | 0 | | 25 | 28 | | | 78 |
| 02 | | 0 | | 6.730769 | 32 | | | 21 |
| H2 | | 0 | | 2 | 2 | 0.059293 | | |
| | | | sum | 33.73077 | | | · · · · · · · · · · · · · · · · · · · | |
| PRODUCTS | | | | | | | | |
| | | | · | | | | | |
| H2O g | 9 | -57.8 | 9.72 | 2 | 18 | | | |
| N2 | 6.916 | 0 | | 25 | 28 | | | |
| 02 | 7.072 | 0 | | 5.7 | 32 | | | |
| | | | | | | | | |
| | delta H = -e | e16*c16 | | 115.6 | | | | |
| | | | | | Ignition Ter | np = 580 C | | |
| | delta T = de | elta H/sum(| Cp*m) | 499.9775 | | | | |

L

A 24

Page 2

Sheet1

| [| <u> </u> | e21/(b16:1 | 8 *e16:18) | 1 | | | | |
|------------------|--------------|---------------------|--------------|------------|--------------|------------|----------|----|
| | | | | 1 | | | <u> </u> | |
| | <u> </u> | | | | | | 1 | |
| Air + 2H2 + 0 | 2> Air + 2 | 2H2O + HE | AT | + | | | | |
| 8% H2 | | | 1 | 1 | | | İ | |
| | · · | | | [| | | air | |
| Reactants | Ср | Hfa. | Hv | m | mol wt | vol % | mol % | |
| | cal/mol/K | kcal/mol | kcal/mol | mol | 1 | | | |
| | | | | 1 | | | | |
| N2 | 7.28 | 0 | | 22 | 28 | | | 78 |
| 02 | | 0 | | 5,923077 | 32 | | | 21 |
| H2 | | 0 | | 2 | 2 | 0.066838 | | |
| ·· | | | sum | 29,92308 | | | | |
| | | | | | | | | |
| PRODUCTS | | | | | | | | |
| | | | | | | | | |
| H2O a | 9 | -57.8 | 9 72 | 2 | 18 | | | |
| N2 | 6.916 | 0 | | 22 | 28 | | | |
| 02 | 7.072 | 0 | | 4 92 | 32 | | | |
| | | | | | | | | |
| | | | | | | | • | |
| | delta H = - | 16*c16 | | 115.6 | | | | |
| | | | | 1.0.0 | Ignition Ter | np = 580 C | | |
| | delta T = de | elta H/sum(| Cp*m) | 564 0504 | -ginton rei | | | |
| | | e21/(b16:1) | 8 *e16:18) | 00 1.000 1 | | | | - |
| | | 02(5 .0.1 | | | | | | |
| At 4% H2 the | molar chan | ne is 49 mo | les anes to | 48 moles | | | | |
| | | <u>ge ie ie iie</u> | | | | | | |
| Delta T = 343 | | | | | | | | |
| | | - <u></u> | | | | | | |
| Pf/Pi = Mf/Mi | * Tf/Ti | = | 2 01984 | | | | | |
| | | | 20 60164 | nei | | | | |
| | | | 23.03104 | psi ., | | | | · |
| | | | | | | | | |
| If Luse ingintio | n temperati | ire of delta | T = 564 K | = 580 C | | | | |
| then I must use | 6 6% H2 | | 1 - 304 K, - | - 560 C | | | | |
| | 5 0.0 /0 HZ | | ~ <u></u> | | | | | |
| Df/Di = Mf/M | * Tf/Ti | | 2 552544 | | | | | |
| | 1//11 | | 27 5002044 | noi | | | | |
| | | | 31.32239 | USI | i | | | |

l

Page 3

APPENDIX 5

QUALITY ASSURANCE PLANS

Lockheed Idaho Technologies Company

P.O. Box 1625

Idaho Falls, ID 83415

Quality Assurance Plan for Safety Documents Independent Review

1.0 INTRODUCTION

This document outlines the quality assurance plan and methods of review for the independent review team (IRT). This team is formed to independently review safety related documents and procedures to assure compliance with safety standards and requirements. Consequently, the IRT is independent of influence from authorship or prior review of documents or parts of documents presented for its review, analysis, or commentary.

The (IRT) shall be formed in accordance with the charter of the requesting organization. The charter may specify the scope of the document to be reviewed, the review criteria, and the review team qualifications. The documents presented to the IRT will be reviewed in the breadth and depth necessary to provide an opinion on whether significant hazards have been identified, appropriate mitigating features have been provided, risks were properly evaluated, and compliance with the applicable requirements as outlined in the appropriate DOE Orders and Standards has been achieved. The review will also be performed in accordance with a set of review standards and criteria described below.

2. REVIEW STANDARDS

The review process is intended to verify that the safety document reviewed ensures compliance with applicable safety-related statutes, rules, and DOE Orders and Standards, and verify safe operations where applicable. Consequently, documents shall be reviewed on the basis of six broad standards of performance: accuracy, completeness, traceability, consistency, readability, and functionality. However, design basis accidents are omitted from the independent review of some safety documents because their associated facility design basis pre-existed the requirements of the DOE Order.

Accuracy means that the document is technically and grammatically accurate. Completeness means that all requirements are addressed, promised statements are provided, and the document is complete from an independent view. Traceability means that the references used in the document exist and contain the promised information, and that the assumptions and bases made re supported by the references as required. Consistency means that the statements made and parameters used in the document agree throughout the document. Readability means that statements made must be understandable by an educated but novice reader and that the document is grammatically adequate. Functionality means that the document satisfies the functional objective of a Safety Analysis Review by establishing a safety envelope, and justifies

the conclusions made therein. These six standards are to be applied by the reviewers in preparing their comments.

3. CRITERIA FOR REVIEWER INDEPENDENCE

IRT reviewers shall be independent of the document preparation. Reviewers shall not have specified a singular approach, ruled out a specific approach, participated in writing the document, performed analyses included in the document or established any input into the document presented for review. However, a reviewer with input into part of the document may evaluate another part of the document where the reviewer's technical qualifications allow such activity.

4. **RESPONSIBILITIES**

The responsibility of the IRT members include:

- 4.1 Review safety analyses and changes in safety analysis documents to ensure that, where applicable:
 - a. Appropriate TSRs have been developed, based upon the described basis and analysis results documented in the SAR, that will ensure safe operations of the facility.
 - b. Compliance with applicable guides, codes, and standards is demonstrated.
 - c. Deviations from current DOE design criteria are evaluated and documented.
 - d. Potential hazards and energy sources are identified and that the facility classification as Category 1, 2, or 3 is appropriate.
 - e. Potential consequences are adequately analyzed.
 - f. The proposed operation will limit risks to the health and safety of the public and the employees, and will adequately protect the environment.
- 4.2 Assess changes involving criticality and verify the system remains within the operational safety envelope and satisfies all applicable DOE requirements.
- 4.3 Determine if the hazard classification category remains correct in document revisions and addenda.
- 4.4 Determine if restart plans are consistent with the applicable ASA, SA, SAR, FSAR, TSRs, and USQDs.

4.5 Assess if deficiencies in design or operation of structures, systems, or components affect risk.

5. REVIEW COMMENT FORMAT AND COMMENT REQUIREMENTS

Comments made by the reviewer(s) shall be technically and grammatically accurate, complete, traceable, consistent with other comments made by the reviewer(s), readable by the document's author(s), and shall assist the author(s) in making the document comply with the requirements imposed on the document. The document's author(s) should be able to understand from the comment the requirement or technical content that has not been met by the document, and should include guidance and suggestions to the authors for correcting the document accordingly.

Comments shall be prepared using word processing software, and provided in the format shown in Figure 1. Reviewers shall use WordPerfect macros provided by the IRT chairman. This will ensure that comments will be in the required format.

For each comment, the reviewer shall enter their initials, the Chapter, page, and paragraph numbers to which the specific comment applies, comment code, and the comment. The initials will be used to identify individual reviewers after all review comments have been combined into a single document and sorted. The chapter, page, and paragraph numbers serve to locate the place in the document text where the comment applies, and provide a means of sorting and organizing the combined comments of all reviewers. The comment code and comments are discussed in the following paragraphs.

The comment code indicates the level of concern that the comment should generate. Comments shall be classified as significant (S), optional (O), or editorial (E). No other codes or notations may be used to classify comments.

Significant comments have safety significance related to the technical content of the document and require additional explanation or description to ensure an adequate safety basis. Significant comments will be resolved before the IRT will issue a final safety evaluation report (SER) without comment. The author(s) response to significant comments will be returned to the original reviewer for evaluation of the response. If there is no agreement reached on a specific comment and resolution thereto, then the IRT may issue the SER noting an exception to the comment's resolution.

Optional comments by their nature do not raise a significant issue to the safety basis or the communication of the safety basis. They are related to the technical content of the document and require additional explanation or description based on the

reviewer's background and experience. Author's responses to optional comments may be accepted by the IRT Chair without further consultation with the original IRT reviewer.

Editorial comments identify grammatical or writing structures that are either incorrect or awkward. Comments shall not be classified as editorial if the grammatical or structural errors generate a significant misunderstanding or ambiguity to the safety basis.

6. PERSONNEL ASSIGNMENTS

Table 1 provides a matrix of review personnel and the chapters assigned for review to the individual reviewers. The assignments are made to obtain both technical depth and breath in the review.

7. REVIEW SCHEDULE AND ACTIVITIES

7.1 Author Briefing

Review of the document will commence with an authors and IRT representative briefing. At this briefing, the authors are expected to brief the IRT representative on their graded approach, and any commitments or agreements made with DOE for the document format and content.

7.2 Initial Reviewers' Meeting - Training

Review of the document will begin after a formal meeting with the IRT. The chairman will ensure documentation of the attending IRT members. He will brief the assembled IRT on the document to be reviewed and request members who may have a conflict of interest in reviewing the document to identify themselves.

The chairman will then present the document to the IRT. He will proceed chapter by chapter through the document asking for and providing comments as areas in question arise. Refer to the printed list of reviewer chesklist to raise issues for discussion (Appendix A).

7.3 First Review Meeting

A meeting of the IRT will convene at mid-review schedule to discuss preliminary review comments. The comments, at this meeting, are expected to be limited to those considered to be significant or fatal to the safety basis. These consequences will be transmitted to the author(s) after an appropriate IRT peer review. This will provide the IRT members and author(s) with an early trend of the comments, allow the author(s) to give early

feedback to the IRT, and allow the author(s) time to prepare for further discussion at the final review meeting, and allow the reviewers to exchange information. Reviewers will provide the IRT administrator with electronic copies of their preliminary comments three working days prior to this first meeting.

7.4 Final Review Meeting

The final review meeting will be held at the end of the review. The reviewers will provide the IRT administrator with electronic copies of their final comments three working days prior to the final meeting. In this final meeting, the reviewers will review all comments, evaluate all the comments for technical validity and evaluate the proposed corrections included in each comment. Rejected comments or proposed corrections will be noted as such and eliminated from the comments compilation which will be forwarded to the author(s) for evaluation and disposition. Reasons for rejecting comments will be documented and recorded in meeting minutes. The author(s) will be briefed on the compiled comments, and after receiving their evaluation, the SER will be drafted incorporating their evaluation and disposition of the compiled comments.

7.5 Adherence to Schedule

The IRT staff is expected to adhere to the review schedule as published. If unforeseen circumstances dictate a change in schedule, IRT members will be notified of the change.

7.6 IRT Review Document Schedule

Meeting minutes will be published for review by the IRT members within 3 working days following the first and final meetings. IRT members shall return their comments on the draft meeting minutes to the IRT chairman within five working days of the draft issuance. The final meeting minutes will be published with the SER. The SER will be published within five working days of final comment resolution or agreement that resolution cannot be achieved on all comments.

8.0 REVIEW CRITERIA

The review criteria for the document are provided in an outline format as a series of questions the document should answer with amplifying information. The review criteria have been extracted from the requirements of DOE Order 5480.23 as annotated by the charter and as pertaining to hazards analysis and classification of the facility, and accident analysis. Some clarifications have been added in some instances.

8.1 Hazards Analysis

Does the document identify the inventory of hazardous materials such as radioactive materials, chemicals, explosives, radiation sources etc. as necessary to support the safety analysis?

_ Hazard classification for each major hazard.

Overall facility hazard classification.

Analyses of Normal, Abnormal, and Accident Conditions Should be Based on Cited Hazards.

Hazards Table

Hazards for each facility covered by the document should be listed in a separate table. A complete list of all hazards must be given in the document.

Each facility-specific table should contain the following columns, but not necessarily in the order given below:

- 1. Hazard Category such as Radiological, electrical, chemical, etc.
- 2. Hazard Description. Identify some specific feature about the hazard. For example, specific quantities of radionuclides, specific quantities of hazardous chemicals, specific electrical hazards, etc.
- 3. Hazards Assessment. Assessment is a judgement about how the hazard could contribute to an accident. For example, information about methods to avoid an incident such as spills, corrosion, leaks, explosions, etc.
- 4. Effects or Consequences. Consequences estimate the potential loss. For example, injury or death, equipment loss, environmental damage, etc.
- 5. **Preventive Measures**. Preventive measures make specific commitments which, when enforced, will help forestall the postulated incidents.

Each facility-specific table of hazards should list the hazards categories by level of risk (Hazard Classification). The classification criteria are derived from DOE-STD-1027-92, and 29 CFR-1910.

- Hazards that are standard industrial hazards (SIH) or that are routinely accepted by the public (RABP) need

only be identified in the document, and OSHA controls are acceptable.

- Hazards that lie between RABP/SIH and category 3 hazards need to be identified in the document, and the safety commitments shall be discussed and analyzed in the document.
- Category 3 and above hazards need to be identified in the document and are expected to have technical specification requirements TSRs associated with them to provide controls.
- Any administrative controls relied on for the categorization shall be identified and summarized for the derivation of the TSRs.
- Tables 1 and 2 provide examples of hazard identification and classification for a hypothetical above ground storage tank containing radionuclides in an aqueous solution and in a sludge at the bottom of the tank.

| Table 1. | Table 1. Example for Hazard Identification and Classification for Tank Sludge. | | | | | | | | |
|--------------------|--|-----------------------------|----------------------------------|--|--|--|--|--|--|
| Hazard Category | Hazard Description | Hazard Assessment | Effects or Consequences | Preventative Measures | | | | | |
| | | At least Catego | ory 3 | | | | | | |
| Rádiological | 1000 Ci ⁹⁰ Sr | spills, leaks, corrosion | area and worker contamination | Tank and spill containment design, Inservice tank inspection. | | | | | |
| Electrical | None | | | | | | | | |
| Chemical | 100 ± 1 lbs mercury | spills, leaks, corrosion | area and worker contamination | Tank and spill containment design, Inservice tank inspection. | | | | | |
| | | Less than catego | ory 3 | | | | | | |
| Radiological | 5 ± 1 g Pu as PuO, | spills, leaks, corrosion | area and worker contamination | Tank and spill containment design, Inservice tank inspection. | | | | | |
| Electrical | None | | | | | | | | |
| Chemical | 1 ± 0.1 lb As | spills, leaks, corrosion | area and worker contamination | Tank and spill containment design, Inservice tank inspection. | | | | | |
| | Routine | ly Accepted by the | Public (RABP) | | | | | | |
| Radiological | < 2 nCi/g ¹³⁷ Cs | | | | | | | | |
| Electrical | 480 V electrical supply to tank heater | Electrical shock | Worker injury and death | OSHA compliance programs. | | | | | |
| Chemical | None | | | | | | | | |

••

· "`\

| Table 2. Example for Hazard Identification and Classification for Tank Liquid. | | | | |
|--|-------------------------------|-----------------------------|----------------------------------|--|
| Hazard Category | Hazard Description | Hazard Assessment | Effects or Consequences | Preventative Measures |
| At least Category -3 | | | | |
| Radiological | 500 ± 50 Ci ¹³⁷ Cs | spills, leaks, corrosion | area and worker contamination | Tank and spill containment design, Inservice tank inspection. |
| Electrical | None | | | |
| Chemical | 2000 ± 100 lb Cl | spills, leaks, corrosion | area and worker contamination | Tank and spill containment design, Inservice tank inspection. |
| Low Temperature | Freezing | Liquid freezes | Tank Failure | Tank design, Class 1E power supply for tank heater. |
| Less than category 3 | | | | |
| Radiological | None | | | |
| Electrical | None | | | |
| Chemical | None | | | |
| Routinely Accepted by the Public (RABP) | | | | |
| Radiological | None | | | |
| Electrical | None | | | |
| Chemical | None | | | |
| Drowning | Tank liquid | Worker falls in tank | Worker injury and death | OSHA compliance programs. |

8.2 Accident Analysis

The document shall describe a thorough analyses of the postulated events that could lead to a breach of the principal health and safety criteria. The document shall also identify the frequency at which these events may occur. It shall establish the design basis accidents where these have been identified, and identify the safety systems that must be operable to prevent or mitigate each identified design basis accident. It shall demonstrate that the principal health and safety criteria can be met by the facilities evaluated for all identified design basis accidents, and to establish the safe operating envelope for the facilities evaluated. The accident analysis should be based on the facility description and operation, the identified hazards analysis and classification of the facility, and the identified principal health and safety criteria. The accident analysis should form the basis of the TSR (or ISORs), and may also provide the basis for personnel training materials and facility management and organization. Questions that must be addressed are:

Does the document provide a listing of possible event scenarios that could lead to releases of radioactive or hazardou's materials?

- Are scenarios provided for each identified hazard?
 - Are the identified scenarios appropriate for the facilities evaluated?
 - Are all possible scenarios identified?
- Do the scenarios challenge the release barriers identified for the facilities evaluated such as waste containers and buildings?
- Does the document provide event frequencies for each of the hypothetical events?

Are the event frequencies supported by appropriate analyses or other evidence?

Does the document identify the structures, systems, and components that are relied upon to reduce the frequency of occurrence (preventive measures)?

Although design basis accidents are not relevant to this document, they are included here for completeness.

DABs are accidents postulated for the purpose of establishing functional requirements for safety-significant structures, systems, components, and equipment. They are usually used to demonstrate that the design, when challenged by a postulated

10 ,

bounding accident, will not violate the principal health and safety criteria.

Are design basis accidents (DBA) identified?

- Does the document identify phenomenologically similar events bounded by a specific DBA? For example, explosions may be bounded by an explosion initiated by a leaking propane heater.
- Are phenomenologically dissimilar events covered by different Dabs? For example, while a propane leak might bound explosions, fires might be bounded by a diesel transport fuel tank leaking.
 - Is each DA supported as a bounding event by appropriate evidence or reasoning?

Does the document provide a deterministic analysis for each hypothetical event classified as an operational or design basis accident?

- Are the following issues identified for each analysis in the document?
 - Initial conditions
 - Boundary conditions
 - Key assumptions

 $\widehat{}$

(

- Identification of structures, systems, and components that are assumed to be operational such as fire suppression.
- Facility design data and information available.
- Reviewers shall review referenced documents such as engineering design files.
- Do the analyses results demonstrate that the principal health and safety criteria will be met?
 - Are the structures, systems and components required to be operational so as to meet the principal health and safety criteria consistent with the identification of engineered safety features in the facility description and operation?
 - Do the analyses results indicate any limits (e.g. a maximum operating temperature) must be satisfied to meet the principal health and safety criteria? Are the failure conditions for engineered safety features identified?

والمرجوع والربي البردي الجيبر والمعصر مرمعور بالمراجع والمتعاد والمراجع
9. MECHANICS OF THE REVIEW PROCESS

9.1 IRT Chairman Role

The IRT Chairman shall base the review and concurrence on an integrated systems concept. The chairman shall also provide for group discussion among reviewers on significant issues regarding the document undergoing review.

9.2 IRT Members Role

- a. Perform their review in accordance with relevant DOE Orders and standards which include references 1-7.
- b. Follow the review guidance provided in the Reviewer Checklist (Appendix A) and the review plan provided with the document at the time it is distributed. Additionally, feel free to make any pertinent comment on any item in the document being reviewed.
- c. Review the parent document (ASA, SA, SAR, FSAR, or TSR) where available, for compatibility during the review.
- d. Visit the site where the proposed involvement will occur, if necessary and when possible, to better understand the activity under review and to obtain a proper frame of reference for comments.
- e. Provide written comments (disk copy or electronic file transfer) to the IRT Chairman, normally three days prior to the scheduled review meeting. Where possible, comments shall be written using WordPerfect software in a format specified by the IRT Chairman.
- f. Notify the IRT Chairman if absence from the scheduled review meeting is anticipated so that provision for absentia comments or alternate reviewer arrangements can be made.
- g. Review for factual accuracy and make any necessary corrections to the draft meeting minutes.
- h. Provide written comments on draft meeting minutes to the IRT chairman.
- i. Review the author responses to comments, as requested by the IRT chairman.
- j. Serve, if assigned, as safety process observer during Title 1 design or readiness reviews.

9.3 Document Review Process

When a document is received by the IRT for review, the Chairman screens and checks if the appropriate signatures are attached to the document, the attached graphics are readable, maps are in appropriate scale, the document is accurate, legible, its format is correct format, and if there are obvious technical errors or shortcomings.

If deficiencies are found in the document, it is returned to the author to correct the deficiencies. After all deficiencies are corrected and the document is returned to the IRT, the Chairman shall:

- a. Determine if the document involves fissionable materials, if so, arrange for review by a member of the Criticality Subcommittee.
- b. Determine if the review request is complex. If so, arrange for the IRT to be briefed on the document by the authors, normally before scheduling for review.
- c. Schedule a review meeting normally two weeks from the date the document is received. Factors that could modify this two week schedule include: the document's size or complexity, special scheduling arrangements agreed upon between the IRT Chairman and the document's manager, or the IRT document review load.
- d. Prepare a review plan, the scope of which may range from simply establishing the schedule and appointing the reviewers to determining specific review requirements for each reviewer.
- e. Prepare a cover letter appointing the reviewers, setting the review meeting time and location, and specifying the document review plan. Distribute copies of the cover letter, review plan, and documents to be reviewed to the assigned committee members. Send a copy of the cover letter to the document author(s).

The IRT Chairman orchestrates the document review by using the following steps to the extent applicable for the subject document:

- a. Personally review the document and prepare written comments.
- b. Normally, collect all reviewer comments three days before review meeting. All comments are to be in electronic format, when possible.
- c. Compile by chapter and page all reviewer comments and make 12 printed copies and one electronic disc copy.

- d. Provide a printed and an electronic disc copy of compiled comments to the document author, normally two days before to the review meeting.
- e. Convene and preside over the document review meeting. If unavailable, appoint the alternate chairman or, if alternate chairman is unavailable, another qualified person to preside. Conduct the review meeting and ensure traceability and control of the meeting.
- f. Moderate the discussion and record all significant comments raised during the meeting not already on the compiled comment list. At the end of the review meeting, collect any reviewer comments not previously submitted.

The following actions provide closure to the review process:

- a. Compile review comments and publish draft meeting minutes documenting the total IRT review comments.
- b. Distribute draft copy of IRT document review meeting minutes to committee members and document author(s) for their review. (Author(s) will write a response to each comment and provide these written comments plus a copy of the revised document indicating by side-bar where changes have been incorporated, or a complete rewrite if needed, to the IRT chairman when the IRT comments have been addressed.)
- c. Submit the author's responses to the reviewers for comment if the author's responses to the reviewer's comments significantly change the document.
- d. Prepare the evaluation report recommending approval or disapproval of the document.
- e. Distribute copies of final meeting minutes, the evaluation report, and transmittal letter to all affected persons.
- f. File a copy of meeting minutes, the evaluation report, and transmittal letter to the in-process file and close the review file. A copy of all document review correspondence will be maintained on file.

Addenda, or documents which have been updated, and resubmitted for review, shall be handled as original documents.

10. REFERENCES

| 1 | DOE Order 5480.5, "Safety of Nuclear Facilities." |
|---|--|
| 2 | DOE-ID Order 5480.5A, "Safety of Nuclear Facilities." |
| 3 | DOE Order 5480.21, "Unreviewed Safety Questions." |
| 4 | DOE Order 5480.22, "Technical Safety Requirements." |
| 5 | DOE Order 5480.23, "Nuclear Safety Analysis Reports." |
| 6 | DOE Order 5481.1B, "Safety Analysis and Review System." |
| 7 | DOE-ID Order 5481.1B, "Safety Analysis and Review System." |

 \sim

APPENDIX A (Page 1 of 3)

REVIEWER CHECKLIST BY DISCIPLINE

Note Reviewers shall consider but not be limited to the following items.

1. Industrial Safety

٦ ۱

()

- 1.1 Understand and apply the concept of passive barriers versus active barriers for protection and mitigation against accidents and energy sources.
- 1.2 Ensure that all energy sources (electrical, pneumatic, chemical, mechanical, etc.) have been analyzed for accident potential and that all credible potential accidents have been completely analyzed.
- 1.3 Determine if the TSR relates directly to an accident analysis and that recovery actions and time limits are included.

2. Industrial Hygiene

- 2.1 Evaluate control techniques or mitigative actions specified in the document to assess their viability and feasibility in reducing or eliminating the potential health hazards.
- 2.2 Identify chemical, physical, biological, and ergonomic stress agents; inadequate operating conditions and procedures; and faulty facilities and equipment which may pose a potential health hazard to on-site and off-site personnel.
- 2.3 Recommend controls (engineering, administrative, and personal protective equipment) that will reduce or eliminate the potential health hazards identified by this review.
- 2.5 Assess risk to on-site and off-site personnel by:
 - a. Verifying exposure level calculations contained in the document.
 - b. Evaluating exposure levels based upon known toxicological data and anticipated exposure potential.

April 19, 1995

IRT DOCUMENT REVIEW PROCESS

Document Cycle

Procedures:

A new document arrives, copies are sent to reviewers, review comments are received electronically or on magnetic diskettes. They are compiled and sent to reviewers the day before the meeting. The first meeting is attended and draft meeting minutes are issued. Comments and resolutions are received back. Resolutions are sent to those reviewers with "S" comments for review. Final meeting minutes are sent out with the evaluation letter, PC file and hardcopy file are closed.

<u>A New Document Arrives</u>

A new document for review arrives with 11 copies from the author.

- 1. Stamp in date
- 2. Set up files (hardcopy file, PC, and diskette)
- 3. Place one copy of the report in file
- 4. Enter in document file schedule for review and first meeting date.
- 5. Notify IRT members of initial meeting date and time.
- 6. Prepare sufficient copies of the document one for each IRT member.
- 7. Place on tracking schedule

To set up a file folder with 4 folders labeled as follows:

- 1 REVIEW DOCUMENT
- 2 MEETING MINUTES
- 3 RESOLUTIONS
- 4 FINAL EVALUATION REPORT

20

an an season enclose parts the second of T

Receiving and Processing Review Comments

TO PREPARE COMMENTS FOR DISTRIBUTION AT REVIEW MEETING

Reviewers will provide their comments on magnetic diskettes approximately 1-2 days prior to the meeting. Comments shall be written using WordPerfect and the provided macros.

- 1. File a copy of each reviewers comments in the PC directory which you set up for the document.
- 2. When all comments are received, compile the comments into one document. Review comments should be sorted by page and paragraph and have the reviewers ID added as well as page numbering.

Use the following procedure to sort the comments:

1. Sort using the keys:

N121 A131 N141 A151

This will sort the type of comment within the numeric order of the comment.

- 2. Make sufficient copies of the compiled comments (IRT members + 2) and give to the Chairman before the committee meeting. Make copies for the authors as needed (usually 2 to 5 copies).
- 3. File one copy of the compiled comments in the file folder.
- 4. Provide the author of the document with their copies of the compiled comments two days prior to the review meeting.

Draft Meeting Minutes

Issued after the first meeting of the committee for document review

- 1. Make changes to the Review Comments as indicated by the Chairman/Members. (These Review Comments will be labeled attachment 2)
 - a. Count the comments per reviewer.
 - b. Count comments by "S" "E" and "O."

21

- 2.
- a. Fill in the subject
- b. Fill in the introduction paragraph
- c. List all attendees from the sign up sheet
- d. List those committee members not present but who have forwarded comments
- e. List those committee members not present
- f. Fill in the comments sections

Dist: See master distribution list

- 3. Give disk and one hardcopy to the chairman for review.
- 4. When the revised draft meeting minutes are received back from the chairman:
- 5. Finalize the draft meeting minutes (put DRAFT heading on top of each page) (IRT FILE NAME DATE DRAFT)
- 6. Obtain necessary signatures
- 7. Distribute copies
- 8. File copy of Draft Meeting Minutes in letterlog.

Resolutions to Meeting Minutes

Resolutions are received back from the authors.

- 1. Stamp in.
- 2. Follow chairman's direction as to action.
- 3. If resolutions are approved by chairman and approval letter sent out or placed in file Note on resolution document this information.

Closed Document Back for Minor Revisions

- 1. Stamp in document
- 2. One copy to IRT Chairman
- 3. Follow Chairman's direction

USQDs

- 1. Document (Unreviewed Safety Question Determination) will be received by Chairman and reviewed.
- 2. Chairman will write Evaluation Letter.
- 3. Set up file, Prepare Evaluation Letter. The Distribution will be determined by Chairman.

To Close Out A Report - Evaluation Report:

- 1. Type the Evaluation Report
- 2. Type draft meeting minutes to final, remove DRAFT header and change the date but not the letter number. Incorporate any changes from comments

Dist: See master distribution list

3. Label attachments according to:

Meeting Minutes - Attachment 1 Comments/Resolutions - Attachment 2 Signature sign-off sheet - Attachment 3 The IRT Chairman may have several attachments to the final evaluation report. These will be provided by the Chairman as he desires.

- 4. File the attendance rosters with the evaluation report
- 5. IRT Chairman signs final meeting minutes AND evaluation report
- 6. Make copies and distribute
- 7. File Evaluation letter and all attachments under the Evaluation Letter Number
- 8. File a copy of the Meeting Minutes under the Meeting Minutes Letter Number (do not remove previous DRAFT copy of the Meeting Minute from the letterlog.

<u>Letters</u>

receive draft w/disk get letter # from letterlog file make corrections and finalize spell check save on hard disk obtain signature make copies distribute file IRT copy in letterlog book

밖에서 이렇게 잘 잘 하는 것을 하는 것 것 같아요. 이렇게 가지 않는 것 같아요. 승규가 많은 것을 물러 가지 않는 것을 다 있다. 것을 하는 것을 가지 않는 것을 가지 않는 것을 하는 것을 수 있다. 이렇게 말했다. 아님에 말했다. 이렇게 말했다. 이 하

| NAME | SECTION/CHAPTER OF DOCUMENT | | | | | | | | |
|------------------------|-----------------------------|--|--|---|----------|--|--|---|--|
| H. Worle | | | | | | | | | |
| B. Meale | | | | | <u> </u> | | | | |
| E. Hochhalter | • | | | | | | | | |
| W. Lussie | | | | | | | | [| |
| G. Dinneen | | | | | | | | | |
| G.Beitel | | | | | | | | | |
| R. Smith | | | | | | | | | |
| N. Morcos | | | | | | | | | |
| Seismic/Struc tural | | | | 1 | | | | | |
| Toxicology | | | | | | | | | |
| | | | | | | | | | |
| | | | | | | | | | |

Q

1

Name Charge Account Numbers Phone # H. Worle B. Meale 6-8963 6-9978 E. Hochhalter 6-1038 W. Lussie 6-1659 G. Dinneen 6-6318 G. Beitel R. Smith N. Morcos 6-0042 •• 6-9345 6-4926

凝白語

EXECUTIVE CORRESPONDENCE



Department of Energy

Richland Operations Office P.O. Box 550 Richland, Washington 99352

AUG 3 0 1996

96-0SH-042

23 05 05 DNF SAFETY LOAN 70 <u>__</u> , and the second $\langle \rangle$ ni a jela Januari Lanada Lanada Lanada Lanada Lanada Lanada Lanada

96/3861

Dr. A. L. Trego, President Westlinghouse Hanford Company Rich and, Washington

Dear Dr. Trego:

AUTHORIZATION OF THE SAFETY ASSESSMENT (SA) OF ROTARY MODE CORE SAMPLING (RMCS) IN FLAMMABLE GAS SINGLE-SHELL TANKS, WHC-SD-WM-SAD-035, REV Oa AND INTERIM OPERATIONAL SAFETY REQUIREMENTS (IOSR)

The Safety Evaluation Report (SER) provided to your staff documents the results of the RL review of the SA of RMCS in Flammable Gas Single-Shell Tanks. This document is recognized to have been prepared as a graded approach to compliance with DOE Order 5480.23, and has been determined to be acceptable for operation of the facility.

By this letter, RL is authorizing the SER and these documents as the nuclear safety authorization basis for RMCS in Flammable Gas Single-Shell Tanks at the Hanford Site.

WHC is expected to implement the requirements of the authorized SA and IOSRs as soon as is reasonably achievable from the date of this letter. Full compliance with the SA and IOSRs is expected and required. The documents are also expected to be updated and revised to meet changing facility conditions and operational requirements. You should coordinate your implementation schedule and plans with the TWRS management.

Please direct any inquiries to me, or your staff may contact, Jackson E. Kinzer, Assistant Manager, TWRS, on 376-7591.

Sincerely.

John D. Wagoner Manager

H. J. Hatch, FDH

DO NOT WRITE ON THIS COVER AS IT IS INTENDED FOR RE-USE RETURN IT WITH THE FILE COPIES TO ORIGINATING OFFICE